ANALYSIS & PDE Volume 18 No. 6 2025



Analysis & PDE

msp.org/apde

EDITOR-IN-CHIEF

Clément Mouhot Cambridge University, UK c.mouhot@dpmms.cam.ac.uk

BOARD OF EDITORS

Massimiliano Berti	Scuola Intern. Sup. di Studi Avanzati, Italy berti@sissa.it	William Minicozzi II	Johns Hopkins University, USA minicozz@math.jhu.edu
Zbigniew Błocki	Uniwersytet Jagielloński, Poland zbigniew.blocki@uj.edu.pl	Werner Müller	Universität Bonn, Germany mueller@math.uni-bonn.de
Charles Fefferman	Princeton University, USA cf@math.princeton.edu	Igor Rodnianski	Princeton University, USA irod@math.princeton.edu
David Gérard-Varet	Université de Paris, France david.gerard-varet@imj-prg.fr	Yum-Tong Siu	Harvard University, USA siu@math.harvard.edu
Colin Guillarmou	Université Paris-Saclay, France colin.guillarmou@universite-paris-saclay.fr	Terence Tao	University of California, Los Angeles, USA tao@math.ucla.edu
Ursula Hamenstaedt	Universität Bonn, Germany ursula@math.uni-bonn.de	Michael E. Taylor	Univ. of North Carolina, Chapel Hill, USA met@math.unc.edu
Peter Hintz	ETH Zurich, Switzerland peter.hintz@math.ethz.ch	Gunther Uhlmann	University of Washington, USA gunther@math.washington.edu
Vadim Kaloshin	Institute of Science and Technology, Austria vadim.kaloshin@gmail.com	András Vasy	Stanford University, USA andras@math.stanford.edu
Izabella Laba	University of British Columbia, Canada ilaba@math.ubc.ca	Dan Virgil Voiculescu	University of California, Berkeley, USA dvv@math.berkeley.edu
Anna L. Mazzucato	Penn State University, USA alm24@psu.edu	Jim Wright	University of Edinburgh, UK j.r.wright@ed.ac.uk
Richard B. Melrose	Massachussets Inst. of Tech., USA rbm@math.mit.edu	Maciej Zworski	University of California, Berkeley, USA zworski@math.berkeley.edu
Frank Merle	Université de Cergy-Pontoise, France merle@ihes.fr		

PRODUCTION

production@msp.org Silvio Levy, Scientific Editor

Cover image: Eric J. Heller: "Linear Ramp"

See inside back cover or msp.org/apde for submission instructions.

The subscription price for 2025 is US \$475/year for the electronic version, and \$735/year (+\$70, if shipping outside the US) for print and electronic. Subscriptions, requests for back issues from the last three years and changes of subscriber address should be sent to MSP.

Analysis & PDE (ISSN 1948-206X electronic, 2157-5045 printed) at Mathematical Sciences Publishers, 798 Evans Hall #3840, c/o University of California, Berkeley, CA 94720-3840, is published continuously online.

APDE peer review and production are managed by EditFlow[®] from MSP.

PUBLISHED BY

mathematical sciences publishers

nonprofit scientific publishing

http://msp.org/

© 2025 Mathematical Sciences Publishers



DAMPED STRICHARTZ ESTIMATES AND THE INCOMPRESSIBLE EULER–MAXWELL SYSTEM

DIOGO ARSÉNIO AND HAROUNE HOUAMED

Euler–Maxwell systems describe the dynamics of inviscid plasmas. We consider an incompressible twodimensional version of such a system and prove the existence and uniqueness of global weak solutions, uniformly with respect to the speed of light $c \in (c_0, \infty)$, for some threshold value $c_0 > 0$ depending only on the initial data. In particular, the condition $c > c_0$ ensures that the velocity of the plasma nowhere exceeds the speed of light and allows us to analyze the singular regime $c \to \infty$.

The functional setting for the fluid velocity lies in the framework of Yudovich's solutions of the twodimensional Euler equations, whereas the analysis of the electromagnetic field hinges upon the refined interactions between the damping and dispersive phenomena in Maxwell's equations in the whole space. This analysis is enabled by the new development of a robust abstract method allowing us to incorporate the damping effect into a variety of existing estimates. The use of this method is illustrated by the derivation of damped Strichartz estimates (including endpoint cases) for several dispersive systems (including the wave and Schrödinger equations), as well as damped maximal regularity estimates for the heat equation. The ensuing damped Strichartz estimates supersede previously existing results on the same systems.

1. Introduction		1310
1.1.	Main results	1311
1.2.	Other models of incompressible plasmas	1314
1.3.	Strategy of proof	1316
1.4.	Notation	1319
2. T	he effect of damping on semigroup flows	1319
2.1.	The damping lemma	1319
2.2.	Damped parabolic estimates	1321
2.3.	Damped Strichartz estimates	1330
3. P	erfect incompressible two-dimensional plasmas	1348
3.1.	Dimensional analysis	1351
3.2.	Approximation procedure and stability	1352
3.3.	Paradifferential calculus and the normal structure	1353
3.4.	Controlling the vorticity	1355
3.5.	Control of high-frequency damped electromagnetic waves	1357
3.6.	Control of low-frequency damped electromagnetic waves	1359
3.7.	Almost-parabolic estimates on the magnetic field	1363
3.8.	Proof of Theorem 3.1	1364
3.9.	Proof of Theorem 3.2	1370
3.10.	Proof of Theorem 3.3	1374
3.11.	Uniform bounds for fixed initial data	1375

Houamed would like to thank Taoufik Hmidi for the fruitful discussions on many related subjects. *MSC2020:* 35A01, 35A02, 35Q35.

Keywords: damped Strichartz estimates, perfect incompressible two-dimensional fluids, Maxwell's system, plasmas, Yudovich's theory, maximal parabolic estimates.

© 2025 MSP (Mathematical Sciences Publishers). Distributed under the Creative Commons Attribution License 4.0 (CC BY). Open Access made possible by subscribing institutions via Subscribe to Open.

Appendix A. Littlewood–Paley decompositions and Besov spaces	1377
A.1. Littlewood–Paley decompositions	1377
A.2. Besov spaces	1378
A.3. Chemin–Lerner spaces	1379
A.4. Embeddings	1379
A.5. Paradifferential product estimates	1380
Appendix B. Oscillatory integrals and dispersion	1381
References	1383

1. Introduction

We are concerned with the existence and uniqueness of solutions to the incompressible Euler–Maxwell system

$$\begin{cases} \partial_t u + u \cdot \nabla u = -\nabla p + j \times B, & \text{div } u = 0 \quad \text{(Euler's equation)}, \\ \frac{1}{c} \partial_t E - \nabla \times B = -j, & \text{div } E = 0 \quad \text{(Ampère's equation)}, \\ \frac{1}{c} \partial_t B + \nabla \times E = 0, & \text{div } B = 0 \quad \text{(Faraday's equation)}, \\ j = \sigma(cE + P(u \times B)), & \text{div } j = 0 \quad \text{(Ohm's law)} \end{cases}$$
(1-1)

for some initial data $(u, E, B)|_{t=0} = (u_0, E_0, B_0)$, with the two-dimensional normal structure on the vector fields

$$u(t,x) = \begin{pmatrix} u_1(t,x) \\ u_2(t,x) \\ 0 \end{pmatrix}, \quad E(t,x) = \begin{pmatrix} E_1(t,x) \\ E_2(t,x) \\ 0 \end{pmatrix} \quad \text{and} \quad B(t,x) = \begin{pmatrix} 0 \\ 0 \\ b(t,x) \end{pmatrix}, \tag{1-2}$$

where $(t, x) \in [0, \infty) \times \mathbb{R}^2$ and $P = \text{Id} - \Delta^{-1} \nabla$ div denotes Leray's projector onto divergence-free vector fields. We will later see that the normal structure (1-2) is propagated by the flow and is therefore persistent.

Taking the divergence of Maxwell's system, which is made up of Ampère and Faraday's equations, notice that the divergence-free conditions div E = 0 and div B = 0 are also propagated by the evolution of the system, provided they hold initially. (In fact, notice that the condition div B = 0 is a trivial consequence of the normal structure (1-2). Nevertheless, it is physically relevant, since magnetic fields are solenoidal.)

This model describes the evolution of a plasma, i.e., a charged gas or an electrically conducting fluid, subject to the self-induced electromagnetic Lorentz force $j \times B$. Here, the field u denotes the velocity of the fluid, E and B are the electric and magnetic fields, respectively, whereas j denotes the electric current. Moreover, the positive constants c and σ represent the speed of light and the electrical conductivity, respectively. We refer to [Biskamp 1993; Davidson 2001] for details about the physical principles behind the modeling of plasmas.

It is readily seen that any smooth solution $(u, E, B) \in C_c^1([0, \infty) \times \mathbb{R}^2)$ of (1-1) satisfies the energy inequality

$$\|u(t)\|_{L^{2}}^{2} + \|E(t)\|_{L^{2}}^{2} + \|B(t)\|_{L^{2}}^{2} + \frac{2}{\sigma} \int_{0}^{t} \|j(\tau)\|_{L^{2}}^{2} d\tau \le \mathcal{E}_{0}^{2}$$
(1-3)

for all $t \ge 0$, where we define

$$\mathcal{E}_0 := \|(u_0, E_0, B_0)\|_{L^2}.$$

(Observe that (1-3) actually holds with an equality sign for smooth functions, but this will not be used.) This is the only known global a priori estimate for solutions of (1-1), and the ensuing natural bound

$$(u, E, B) \in L^{\infty}([0, \infty); L^2(\mathbb{R}^2))$$

is insufficient to guarantee the existence of global weak solutions to (1-1). At least, no known method has so far been able to build such solutions, and the same holds for the classical two-dimensional incompressible Euler system

$$\partial_t u + u \cdot \nabla u = -\nabla p, \quad \text{div} \, u = 0,$$
(1-4)

which corresponds to the case (E, B) = 0. This is due to the fact that the nonlinear terms in (1-1) and (1-4) are, in general, not stable under weak convergence of solutions.

1.1. *Main results.* Our main result on the Euler–Maxwell system (1-1) establishes the global existence and uniqueness of weak solutions for any initial data in suitable spaces, provided the speed of light *c* is sufficiently large. Note that this is seemingly the only known global existence result for incompressible Euler–Maxwell systems. It reads as follows.

Theorem 1.1. Let p and s be any real numbers in $(2, \infty)$ and $(\frac{7}{4}, 2)$, respectively. For any initial data

$$(u_0, E_0, B_0) \in ((H^1 \cap \dot{W}^{1,p}) \times H^s \times H^s)(\mathbb{R}^2),$$

with div $u_0 = \text{div } E_0 = \text{div } B_0$ and the two-dimensional normal structure (1-2), there is a constant $c_0 > 0$ such that, for any speed of light $c \in (c_0, \infty)$, there is a global weak solution (u, E, B) to the two-dimensional Euler–Maxwell system (1-1), with the normal structure (1-2), satisfying the energy inequality (1-3) and enjoying the additional regularity

$$u \in L^{\infty}(\mathbb{R}^+; H^1 \cap \dot{W}^{1,p}), \quad (E, B) \in L^{\infty}(\mathbb{R}^+; H^s),$$

(cE, B) $\in L^2(\mathbb{R}^+; \dot{H}^1), \quad cE \in L^2(\mathbb{R}^+; \dot{H}^s), \quad (E, B) \in L^2(\mathbb{R}^+; \dot{W}^{1,\infty}).$ (1-5)

It is to be emphasized that the bounds in (1-5) are uniform in $c \in (c_0, \infty)$ for any given initial data.

If, furthermore, the initial vorticity $\omega_0 := \nabla \times u_0$ belongs to $L^{\infty}(\mathbb{R}^2)$, then the solution enjoys the global bound

$$\omega := \nabla \times u \in L^{\infty}(\mathbb{R}^+; L^{\infty}),$$

and it is unique in the space of all solutions $(\bar{u}, \bar{E}, \bar{B})$ to the Euler–Maxwell system (1-1) satisfying the bounds, locally in time,

$$(\bar{u}, \bar{E}, \bar{B}) \in L^{\infty}_t L^2_x, \quad \bar{u} \in L^2_t L^{\infty}_x, \quad \bar{j} \in L^2_{t,x}.$$

and having the same initial data.

Theorem 1.1 is a simple and more accessible reformulation of the results from Section 3, which are stated therein in full detail in the setting of Besov and Chemin–Lerner spaces (see Appendix A for a precise definition of these spaces). Indeed, it is readily seen that Theorem 1.1 follows directly from the combination of Theorems 3.1, 3.2, 3.3 and Corollary 3.13 with straightforward embeddings of functional spaces. The respective proofs of these results are also provided in complete detail in Section 3.

One should note that the constant c_0 in the above statement depends on norms of the initial data. Thus, for any given c > 0, the condition $c_0 < c$ can be interpreted, in a fully equivalent way, as a smallness condition on the initial data. In fact, a careful inspection of (3-4) in the statement of Theorem 3.3 readily provides an explicit expression for c_0 in terms of the norms of (u_0, E_0, B_0) in $(H^1 \cap \dot{W}^{1,p}) \times H^s \times H^s$. More specifically, for any given initial data, one could set, for example,

$$c_0 = \max\{1, (\|u_0\|_{H^1 \cap \dot{W}^{1,p}} + \|(E_0, B_0)\|_{H^s})Ce^{C\mathcal{E}_0^5}\}$$

for some suitable large constant C > 0 which only depends on p and s, and is independent of the initial data. Then, with this definition of c_0 , it is straightforward to show that the condition $c > c_0$ implies the validity of (3-4). In particular, for a given speed of light c, we observe that the existence of solutions is a consequence of the smallness of the initial data. Finally, we also note that it is not difficult to provide sharper formulas for c_0 , with increasing complexity.

A detailed scaling analysis of solutions to the Euler–Maxwell system (1-1) is conducted in Section 3.1, which further clarifies the significance of the initial conditions of our main results and their dependence on the physical constants c and σ .

We have already emphasized that the bounds (1-5) on the solutions of the incompressible Euler-Maxwell system (1-1) are uniform with respect to the speed of light $c > c_0$. This crucial feature allows us to deduce a simple but powerful convergence result in the asymptotic regime $c \to \infty$, which is of particular interest. We refer to [Arsénio et al. 2015] for a thorough discussion of this regime in the context of incompressible Navier-Stokes-Maxwell systems.

Generally speaking, the physical relevance of the regime $c \to \infty$ in Euler–Maxwell systems stems from the fact that the limiting magnetohydrodynamic systems are suitable to describe the behavior of flows which are influenced by self-induced magnetic fields. This is the case, for instance, of the terrestrial magnetic field, which is sustained by the earth's core through the dynamo effect, or the solar magnetic field, which is responsible for sunspots, or the galactic magnetic field, which plays a role in the formation of stars. We refer to [Davidson 2001] for more details on the physical background of magnetohydrodynamic systems.

The next result follows directly from Theorem 1.1 and establishes a magnetohydrodynamic system by taking the limit of the Euler–Maxwell system (1-1) in the singular regime $c \to \infty$. Observe that it recovers the classical Yudovich theorem for the incompressible Euler system (1-4) by setting $B \equiv 0$.

Corollary 1.2. For any given initial data (u_0, E_0, B_0) as in Theorem 1.1 (for some $p \in (2, \infty)$ and $s \in (\frac{7}{4}, 2)$), consider the global solution (u^c, E^c, B^c) constructed therein for each $c \in (c_0, \infty)$. Then, the set of solutions $\{(u^c, E^c, B^c)\}_{c>c_0}$ is relatively compact in $L^2_{t,x,\text{loc}}$. In particular, for any sequence $\{(u^{c_n}, E^{c_n}, B^{c_n})\}_{n \in \mathbb{N}}$, with $c_n \to \infty$, there is a convergent subsequence (which we do not distinguish, for simplicity)

 $(u^{c_n}, E^{c_n}, B^{c_n}) \xrightarrow{n \to \infty} (u, 0, B) \quad in \ L^2_{t,x, \text{loc}}, \tag{1-6}$

where $(u, B) = ((u_1, u_2, 0), (0, 0, b))$ has the normal structure (1-2) and is a global weak solution of the system

$$\begin{cases} \partial_t u + u \cdot \nabla u = -\nabla p, & \text{div } u = 0, \\ \partial_t b - \frac{1}{\sigma} \Delta b + u \cdot \nabla b = 0, \end{cases}$$
(1-7)

with the bounds

$$u \in L^{\infty}(\mathbb{R}^+; H^1 \cap \dot{W}^{1,p}), \quad b \in L^{\infty}(\mathbb{R}^+; H^1) \cap L^2(\mathbb{R}^+; \dot{H}^1 \cap \dot{H}^2 \cap \dot{W}^{1,\infty})$$

If, furthermore, the initial vorticity ω_0 belongs to $L^{\infty}(\mathbb{R}^2)$, then the solution (u, b) to (1-7) satisfies the additional bound $\omega \in L^{\infty}(\mathbb{R}^+; L^{\infty})$ and is unique in the space of all solutions (\bar{u}, \bar{b}) satisfying the bounds, locally in time,

$$\bar{u} \in L^{\infty}_t L^2_x \cap L^2_t L^{\infty}_x, \quad \bar{b} \in L^{\infty}_t L^2_x \cap L^2_t \dot{H}^1_x,$$

and having the same initial data. Moreover, one has the convergence

$$(u^c, E^c, B^c) \xrightarrow{c \to \infty} (u, 0, B) \quad in L^2_{t,x, \text{loc}},$$
 (1-8)

without extraction of subsequences.

Remark. Note that (1-7) is a simple form of a magnetohydrodynamic system. Indeed, the equations for u and b are not genuinely coupled, for the incompressible Euler equation does not contain an external magnetic force. This can be interpreted as a consequence of the two-dimensional normal structure (1-2). More specifically, whenever the electric current is given by $j = \nabla \times B$, a straightforward calculation exploiting (1-2) shows that the Lorentz force satisfies

$$j \times B = (\nabla \times B) \times B = -\frac{1}{2}\nabla(b^2),$$

which can be absorbed in the pressure gradient. In particular, since u is independent of b in this regime, there can be no Alfvén waves (see [Davidson 2001] for an introduction to Alfvén waves). Therefore, in this case, the limiting magnetohydrodynamic system loses the feature of some important physical effects (such as Alfvén waves). This suggests that extending the results of the present article beyond the two-dimensional normal structure (1-2) is of particular interest and significance.

Proof. We begin by showing the relative compactness of the set of solutions $\{(u^c, E^c, B^c)\}_{c>c_0}$ in $L^2_{t,x}(K)$ for any compact set $K \subset \mathbb{R}^+ \times \mathbb{R}^2$. To that end, note that the energy inequality (1-3) and the global bounds (1-5) on the solutions hold uniformly in *c*. In particular, it is readily seen that $E^c \to 0$ in $L^2_{t,x,\text{loc}}$ as $c \to \infty$. Therefore, we only need to focus on $\{(u^c, B^c)\}_{c>c_0}$.

Now, one can show from (1-5) that $u^c \in L^{\infty}_{t,x}$ and $B^c \in L^2_t L^{\infty}_x$ (for instance, using the Gagliardo– Nirenberg convexity inequality (3-16), which is recalled later on). It therefore follows directly from (1-9) that $\partial_t u^c = P(j^c \times B^c) - P(u^c \cdot \nabla u^c)$ is uniformly bounded in $L^1_{t,\text{loc}} L^2_x$. Similarly, it is readily seen from Faraday's equation $\partial_t B^c = -c\nabla \times E^c$ that $\partial_t B^c$ is uniformly bounded in $L^2_{t,x}$. Then, further combining these controls of $\partial_t u^c$ and $\partial_t B^c$ with the uniform bound $(u^{c_n}, B^{c_n}) \in L^{\infty}_t H^1_x$ and the compactness of the embedding $H^1_{\text{loc}} \subset L^2_{\text{loc}}$, we deduce that $\{(u^c, B^c)\}_{c>c_0}$ is relatively compact in the topology of $L^2_{t,x,\text{loc}}$ by a classical compactness result by Aubin and Lions. (See [Simon 1987] for a thorough discussion of such compactness results and, in particular, Section 9 therein, for convenient results which are easily applicable to our setting.) Next, for any convergent subsequence (1-6), employing Ohm's law to substitute $c_n E^{c_n}$ into Faraday's equation in (1-1), observe that we only have to pass to the limit in the system

$$\begin{cases} \partial_t u^{c_n} + u^{c_n} \cdot \nabla u^{c_n} = -\nabla p^{c_n} + j^{c_n} \times B^{c_n}, & \operatorname{div} u^{c_n} = 0, \\ \frac{1}{c_n} \partial_t E^{c_n} - \nabla \times B^{c_n} = -j^{c_n}, & \operatorname{div} B^{c_n} = 0, \\ \partial_t B^{c_n} + \frac{1}{\sigma} \nabla \times j^{c_n} + u^{c_n} \cdot \nabla B^{c_n} = 0. \end{cases}$$
(1-9)

Moreover, up to further extraction of subsequences, it is also possible to assume that one has the weak convergence

$$j^{c_n} \rightarrow j \quad \text{in } L^2_{t,x}$$

All in all, passing to the limit $n \to \infty$ in (1-9) in the sense of distributions and exploiting the strong convergence (1-6), we find that

$$\begin{cases} \partial_t u + u \cdot \nabla u = -\nabla p + j \times B, & \text{div } u = 0, \\ \nabla \times B = j, & \text{div } B = 0, \\ \partial_t B + \frac{1}{\sigma} \nabla \times j + u \cdot \nabla B = 0. \end{cases}$$

Then, recalling the vector identity $\nabla \times (\nabla \times B) = \nabla (\operatorname{div} B) - \Delta B$ and noticing that $(\nabla \times B) \times B = -\frac{1}{2} \nabla (b^2)$, we conclude that (u, b) is a solution of (1-7).

Finally, if we further assume the pointwise boundedness of the initial vorticity ω_0 , then $\omega^c = \nabla \times u^c$ remains uniformly bounded in $L_{t,x}^{\infty}$, thereby yielding a similar bound for the limiting system (1-7). These bounds then fall in the framework of Yudovich's uniqueness theorem (see [Majda and Bertozzi 2002, Section 8.2.4], for instance), which guarantees the uniqueness of the solution u to the incompressible two-dimensional Euler system. Alternatively, one can also deduce the uniqueness of u by reproducing the arguments from Section 3.9 below, by setting (E, B) = 0. As for the uniqueness of b, it easily follows from classical energy estimates on the heat equation.

At last, the uniqueness of the limit point (u, 0, B) allows us to deduce the validity of (1-8), which completes the proof of the corollary.

1.2. *Other models of incompressible plasmas.* The Euler–Maxwell system (1-1) can be seen as the inviscid version of the Navier–Stokes–Maxwell system given by

$$\begin{cases} \partial_t u + u \cdot \nabla u - v \Delta u = -\nabla p + j \times B, & \text{div } u = 0, \\ \frac{1}{c} \partial_t E - \nabla \times B = -j, & \text{div } E = 0, \\ \frac{1}{c} \partial_t B + \nabla \times E = 0, & \text{div } B = 0, \\ j = \sigma(cE + P(u \times B)), & \text{div } j = 0, \end{cases}$$
(1-10)

where $\nu > 0$ denotes the viscosity of the fluid.

The derivation of (1-10) has been established rigorously in [Arsénio and Saint-Raymond 2019] through the analysis of the viscous incompressible hydrodynamic regimes of Vlasov–Maxwell–Boltzmann systems. In particular, it follows from the results therein that (1-10) can be obtained by letting $\delta \rightarrow 0$, with $\delta > 0$,

in the more complete system

$$\begin{cases} \partial_t u + u \cdot \nabla u - v \Delta u = -\nabla p + \delta cnE + j \times B, & \operatorname{div} u = 0, \\ \frac{1}{c} \partial_t E - \nabla \times B = -j, & \operatorname{div} E = \delta n, \\ \frac{1}{c} \partial_t B + \nabla \times E = 0, & \operatorname{div} B = 0, \\ j - \delta nu = \sigma \left(-\frac{c}{\delta} \nabla n + cE + u \times B \right), \end{cases}$$
(1-11)

which takes the Coulomb force nE into account, where n is the electric charge density.

The work performed in [Arsénio and Saint-Raymond 2019] addresses the viscous incompressible regimes of Vlasov–Maxwell–Boltzmann systems only. However, inviscid incompressible regimes can also be achieved as an asymptotic limit of collisional kinetic equations. For instance, the incompressible Euler limit of the Boltzmann equation has been established in [Saint-Raymond 2003; 2009b]. (A general discussion of hydrodynamic regimes of the Boltzmann equation can also be found in [Saint-Raymond 2009a].) Similarly, in the vein of the results from [Arsénio and Saint-Raymond 2019], it is possible to derive (1-1) by considering the incompressible Euler regime of Vlasov–Maxwell–Boltzmann systems, at least formally. However, this remains to be done rigorously.

The well-posedness theory established in this article only concerns (1-1) and does not encompass the inviscid version of (1-11) (i.e., the corresponding Euler–Maxwell system obtained by setting $\nu = 0$ in (1-11)). However, we are hopeful that some adaptation of our results can be implemented to show the existence and uniqueness of solutions to (1-11), with $\nu = 0$. Nevertheless, for the sake of simplicity, we are going to stick to (1-1).

It turns out that there is yet another version of incompressible Navier–Stokes–Maxwell systems which is commonly found in the literature. It reads

$$\begin{cases} \partial_t u + u \cdot \nabla u - v \Delta u = -\nabla p + j \times B, & \operatorname{div} u = 0, \\ \frac{1}{c} \partial_t E - \nabla \times B = -j, & \operatorname{div} B = 0, \\ \frac{1}{c} \partial_t B + \nabla \times E = 0, \\ j = \sigma (cE + u \times B), \end{cases}$$
(1-12)

and a corresponding incompressible Euler–Maxwell system is given by setting $\nu = 0$. We refer to [Arsénio and Gallagher 2020; Germain et al. 2014; Masmoudi 2010] for details on the construction of global solutions to (1-12), with $\nu > 0$.

Unlike (1-10) and (1-11), it is to be emphasized that this model is not obtained as an asymptotic regime of Vlasov–Maxwell–Boltzmann systems, as shown in [Arsénio and Saint-Raymond 2019]. Furthermore, when compared to (1-10) and (1-11), it has the major drawback of not providing a strong control of div *E*. For this reason, we do not make any claim concerning the extension of our work to the above model. It would, however, be interesting to clarify the well-posedness of the nonviscous version of (1-12).

Finally, we observe that there is also a rich family of compressible Euler–Maxwell systems which are commonly used to model the behavior of plasmas. The study of such systems is challenging, and corresponding results tend to focus on the stability of smooth solutions near specific equilibrium states. We refer to [Germain and Masmoudi 2014; Guo et al. 2016] for foundational results on three-dimensional

compressible Euler–Maxwell systems. We note that the results therein do not require any specific vector structure, such as the normal structure (1-2). However, they are sensitive to the speed of light c and, therefore, may not provide uniform bounds as c tends to infinity.

1.3. *Strategy of proof.* We lay out now the strategy and the key ideas leading to the proof of Theorem 1.1, which will be implemented in Section 3 to establish the more precise Theorems 3.1, 3.2 and 3.3.

Observe first that, even if we add a dissipation term $-\Delta u$ to the first equation of (1-1), thereby yielding the incompressible Navier–Stokes–Maxwell system (1-10), it is still unknown whether or not global weak solutions do exist when the initial data are only square-integrable. This is due to the lack of strong compactness (or regularity) in electromagnetic fields (E, B), combined with the lack of stability of the source term $j \times B$ in weak topologies (see [Arsénio and Gallagher 2020] for further details). The same difficulty persists in the inviscid version of the same system, which stems from the propagation of singularities in Maxwell's system, as a result of its hyperbolic nature. The construction of solutions in L^2 to (1-1) is thus highly challenging — all the more so than in the viscous case.

One should therefore treat this system in some higher-regularity spaces. To this end, inspired by known results on the well-posedness of the two-dimensional Euler system (1-4), we shall look at the equivalent vorticity formulation of (1-1), which reads as

$$\begin{cases} \partial_t \omega + u \cdot \nabla \omega = -j \cdot \nabla B, & \text{div } u = 0, \\ \frac{1}{c} \partial_t E - \nabla \times B + \sigma c E = -\sigma P(u \times B), & \text{div } E = 0, \\ \frac{1}{c} \partial_t B + \nabla \times E = 0, & \text{div } B = 0, \\ j = \sigma(cE + P(u \times B)), & \text{div } j = 0, \end{cases}$$
(1-13)

where $\omega := \nabla \times u$ and u can be reconstructed from ω through the Biot–Savart law

$$u = -\Delta^{-1} \nabla \times \omega. \tag{1-14}$$

Observe that the normal structure (1-2) has been used in (1-13) to write $\nabla \times (j \times B) = -j \cdot \nabla B$. This is crucial.

Much of our analysis of (1-13) will hinge on the dispersive properties of the damped Maxwell system

$$\frac{1}{c}\partial_t E - \nabla \times B + \sigma c E = -\sigma P(u \times B),$$

$$\frac{1}{c}\partial_t B + \nabla \times E = 0,$$

div $u = \text{div } E = \text{div } B = 0.$
(1-15)

This will require us to interpret the role of the velocity field u in (1-15), in the spatial variable x, as that of a coefficient in the algebra $L_x^{\infty} \cap \dot{H}_x^1$ (or some weaker variant), thereby allowing us to view (1-15) as a linear system in (E, B) and produce closed estimates on the electromagnetic field.

To be precise, the treatment of the source term $-\sigma P(u \times B)$ in (1-15) will necessitate the control of the velocity field u in a suitable algebra acting on \dot{H}_x^s for appropriate values of s. In particular, according to the classical paradifferential product law

$$\|fg\|_{\dot{H}^{s}(\mathbb{R}^{d})} \lesssim \|f\|_{L^{\infty} \cap \dot{B}^{d/2}_{2,\infty}(\mathbb{R}^{d})} \|g\|_{\dot{H}^{s}(\mathbb{R}^{d})},$$

which holds for any $s \in \left(-\frac{1}{2}d, \frac{1}{2}d\right)$ and $d \ge 1$, it will be natural to seek the control of u (in the space variable) in the weaker algebra $L^{\infty} \cap \dot{B}_{2,\infty}^{1}(\mathbb{R}^{2})$; see Appendix A for a definition of Besov spaces.

In the context of two-dimensional viscous flows, such a control is expected in view of the strong bounds provided by the energy dissipation inequality. For example, in [Arsénio and Gallagher 2020, Theorem 1.2], the existence of weak solutions to a two-dimensional incompressible Navier–Stokes–Maxwell system was established by proving a uniform control of the velocity field in the algebra $L_x^{\infty} \cap \dot{H}_x^1(\mathbb{R}^2)$. More precisely, by building upon the methods from [Masmoudi 2010], it was shown therein (see [Arsénio and Gallagher 2020, Proposition 2.1]) that the control of the velocity field in the space $L_t^2(L_x^{\infty} \cap \dot{H}_x^1)$ was sufficient to propagate some \dot{H}^s -regularity, with -1 < s < 1, in Maxwell's equations (1-15), uniformly as $c \to \infty$.

In the setting of two-dimensional incompressible electrically conducting ideal fluids (i.e., plasmas), which is the focus of our work, global energy estimates are nowhere near as good as their viscous counterpart and, thus, fail to yield the control of u in a useful algebra. Instead, we need to take Yudovich's approach of propagating the $L_x^2 \cap L_x^p$ -norm of the vorticity ω , for some given p > 2, by exploiting the transport equation

$$\partial_t \omega + u \cdot \nabla \omega = -j \cdot \nabla B, \tag{1-16}$$

thereby providing a bound on u in the algebra $L_t^{\infty}(L_x^{\infty} \cap \dot{H}_x^1)$, by classical Sobolev embeddings combined with standard estimates on the Biot–Savart law (1-14). We refer to [Bahouri et al. 2011, Section 7.2] for a modern treatment of global existence results for two-dimensional perfect incompressible fluids and the Yudovich theorem.

In particular, elementary estimates on transport equations, which are performed in detail in Section 3.4, show that the control of ω in $L_t^{\infty} L_x^p$ follows from the control of the initial vorticity ω_0 and the nonlinear source term $j \cdot \nabla B$ in L_x^p and $L_t^1 L_x^p$, respectively. Since j is naturally bounded in $L_{t,x}^2$, by virtue of the energy inequality (1-3), we conclude that ∇B should be controlled in $L_t^2 L_x^{\infty}$.

Now, experience shows that such a Lipschitz bound on *B* cannot easily follow from energy estimates on the wave system (1-15). Indeed, energy estimates on hyperbolic systems are typically performed in L_x^2 . Therefore, in order to control ∇B in L_x^∞ , an energy estimate on (1-15) would lead us, in view of classical Sobolev embeddings, to seek a bound of *B* in $H_x^{2+\delta}$, with a small parameter $\delta > 0$. To that end, the source term $-\sigma P(u \times B)$ in (1-15) would also need to be controlled in $H_x^{2+\delta}$. However, employing paradifferential calculus to control $u \times B$ would require that ∇u be bounded in $L_x^\infty \cap \dot{H}_x^1$ at least. Unfortunately, such uniform bounds on perfect incompressible two-dimensional flows are largely out of reach in our context. This is where the damped dispersive properties of (1-15), on the whole Euclidean plane \mathbb{R}^2 , come into play.

Maxwell's system (1-15) can be rewritten as a system of wave equations (more on this later on, see (1-17)). Thus, heuristically, one expects to be able to employ Strichartz estimates for the wave equation to control the electromagnetic field (E, B). In particular, by paying close attention to the admissibility criteria of functional spaces in Strichartz estimates (see [Bahouri et al. 2011, Section 8.3] or [Keel and Tao 1998]), one observes that it is possible to control the Lipschitz norm of a solution to a two-dimensional wave equation, provided one can bound $\frac{7}{4}$ derivatives of the initial data and the source term in some appropriate

functional spaces (in some Besov spaces, for instance) of L^2 space-integrability. (For simplicity, we have omitted here the consideration of time integrability in Strichartz estimates and focused solely on space regularity and integrability.) Loosely speaking, such an estimate is better than a Sobolev embedding, which would require the control of over two derivatives in $L^2(\mathbb{R}^2)$ in order to bound a Lipschitz norm. This should give the reader some intuition concerning the special role played by the regularity parameter $s = \frac{7}{4}$ in Theorem 1.1.

Thus, so far, our strategy seems to yield some promising closed estimates. Indeed, on the one hand, the transport equation (1-16) gives us a bound on the $L_t^{\infty}(L_x^2 \cap L_x^p)$ -norm of the vorticity ω provided ∇B is controlled in $L_t^2 L_x^{\infty}$, while, on the other hand, a control of ∇B in $L_t^2 L_x^{\infty}$ can be achieved through dispersive estimates on the wave system (1-15) if the velocity field u is sufficiently smooth (at least $L_t^{\infty}(L_x^{\infty} \cap \dot{H}_x^1)$, say).

However, such a roadmap may not lead to global estimates in time. To see this, we need to take a closer look at the temporal norms associated with our strategy. Specifically, it is important to note that the classical Strichartz estimates for the two-dimensional wave equation do not actually give a global control of ∇B in $L_t^2 L_x^\infty$. Instead, they only allow us to control ∇B in $L_t^4 L_x^\infty$ globally, which then leads to a control in $L_t^2 L_x^\infty$ locally in time. This difficulty is solved by complementing our strategy with a careful study of the damping phenomenon in (1-15) produced by the term $\sigma c E$. To that end, we provide, in Section 2, a robust analysis of the damping effect on general semigroup flows, which is formulated in precise terms in Lemma 2.1 (the damping lemma). We also give applications of the damping lemma to parabolic and dispersive equations in Sections 2.2 and 2.3, respectively.

Concerning Maxwell's system (1-15), the ensuing time decay of the electromagnetic field is encapsulated in Corollary 2.12. It is shown therein that (1-15) forms a damped hyperbolic system which is best understood by decomposing the frequencies of the solutions relative to the magnitude of the speed of light c > 0.

Indeed, by appropriately combining Ampère's equation and Faraday's equation from (1-15) and using that $\nabla \times (\nabla \times B) = -\Delta B$, observe that *B* solves the damped wave equation

$$\frac{1}{c^2}\partial_t^2 B + \sigma \partial_t B - \Delta B = -\sigma \nabla \times (u \times B), \qquad (1-17)$$

where the damping term $\sigma \partial_t B$ comes from the term $\sigma c E$ in (1-15).

Heuristically, since waves described by (1-17) typically propagate with a characteristic speed c, it is then natural to expect a consistent hyperbolic behavior of the solutions of (1-15) on the range of frequencies larger than a suitable multiple of the speed of light c. In particular, Corollary 2.12 will confirm that solutions to (1-17) enjoy dispersive properties for those high frequencies, which are analogous to the nondamped case (obtained by setting $\sigma = 0$ in (1-17)) with drastically improved long-time integrability.

On the remaining range of frequencies, i.e., on frequencies slower than c, the same result will establish that the behavior of solutions to (1-15) is largely dictated by the heat equation

$$\sigma \partial_t B - \Delta B = -\sigma \nabla \times (u \times B),$$

which is formally achieved in the asymptotic regime $c \to \infty$ from (1-17).

All in all, the application of the sharp damped dispersive estimates from Section 2 to Maxwell's equations (1-15) will allow us to obtain closed estimates on the incompressible Euler–Maxwell system (1-1) which hold globally and lead to Theorem 1.1. The precise nonlinear analysis of (1-1) is detailed in Section 3 with complete proofs of our main theorems.

It is difficult to pinpoint the exact source of the breakdown of our proofs for small values of light velocity c. However, one can argue that the degeneracy of Maxwell's system in the limit $c \rightarrow 0$ results in a loss of the damped dispersive properties which are central to our nonlinear analysis. We believe that this provides some evidence that our method cannot be extended to the whole range of c > 0. Nevertheless, we are hopeful that other techniques may be used to construct solutions in the remaining range of light velocities.

1.4. *Notation.* Allow us to clarify some notation which will be used repeatedly throughout this article.

First of all, for clarity and convenience, note that all relevant functional spaces of Besov and Chemin– Lerner types are introduced in precise detail in Appendix A.

Next, Leray's projector

$$P: L^2(\mathbb{R}^3; \mathbb{R}^3) \to L^2(\mathbb{R}^3; \mathbb{R}^3)$$

onto divergence-free vector fields, which is used in (1-1), and the corresponding orthogonal projector $P^{\perp} = \text{Id} - P$ onto conservative fields are given by

$$P = \operatorname{Id} - \Delta^{-1} \nabla \operatorname{div}, \quad P^{\perp} = \Delta^{-1} \nabla \operatorname{div}.$$

Finally, when necessary, we will employ the letter *C* to denote a generic constant, which is allowed to differ from one estimate to another, and we will resort to the use of indices to distinguish specific constants. We will also often write $A \leq B$ to denote $A \leq CB$ for some positive constant *C* which only depends on fixed parameters, and $A \sim B$ whenever $A \leq B$ and $B \leq A$ are simultaneously true.

2. The effect of damping on semigroup flows

Here, we analyze the effect of damping on evolution flows, which are generally described by semigroups. More specifically, in Section 2.1, we begin by establishing a robust and general result—called the damping lemma—showing how damping terms act on integral operators. Then, in Sections 2.2 and 2.3, this result is applied to the context of damped parabolic and Strichartz estimates, which will be crucial to our analysis of Maxwell's system in Section 3. In particular, in Section 2.3, we give complete and sharp formulations of Strichartz estimates for the damped Schrödinger, half-wave, wave and Maxwell equations in Euclidean spaces.

2.1. *The damping lemma.* The result below provides a general and robust principle allowing us to take into account the influence of a damping term $e^{-\alpha t}$, with $\alpha > 0$, on an integral operator.

Lemma 2.1 (the damping lemma). Let X and Y be Banach spaces and, for each $s, t \in [0, T)$, with T > 0, let $K(t, s) : X \to Y$ be an operator-valued kernel from X to Y such that

$$K(t,s) \in L^{1}([0,T) \times [0,T); \mathcal{L}(X,Y)),$$

where $\mathcal{L}(X, Y)$ denotes the Banach space of bounded linear operators from X to Y. Further suppose that there are $0 < p_0 \le q_0 \le \infty$, with $q_0 \ge 1$, and a constant A > 0 such that the estimate

$$\left\| \int_{0}^{T} \chi(t,s) K(t,s) f(s) \, ds \right\|_{L^{q_0}([0,T);Y)} \le A \| f \|_{L^{p_0}([0,T);X)}$$
(2-1)

holds for all $f \in L^{p_0}([0,T); X)$ and any $\chi(t,s) \in L^{\infty}([0,T)^2; \mathbb{R})$, with $\|\chi\|_{L^{\infty}} \leq 1$.

Then, for any $\alpha \ge 0$ and $p_0 \le p \le q \le q_0$, with $q \ge 1$, one has the damped estimate

$$\left\|\int_{0}^{T} e^{-\alpha|t-s|} \chi(t,s) K(t,s) f(s) \, ds\right\|_{L^{q}([0,T);Y)} \le C_{\beta} A\left(\frac{T}{1+\alpha T}\right)^{\beta} \|f\|_{L^{p}([0,T);X)}$$

for all $f \in L^p([0,T); X)$ and any $\chi(t,s) \in L^{\infty}([0,T)^2; \mathbb{R})$, with $\|\chi\|_{L^{\infty}} \leq 1$, where $\beta \geq 0$ is defined by

$$\beta = \frac{1}{q} - \frac{1}{q_0} + \frac{1}{p_0} - \frac{1}{p}$$

and $C_{\beta} > 0$ only depends on β .

Proof. For $\alpha = 0$, the result follows straightforwardly from Hölder's inequality on the domain [0, T) for all integrability parameters merely satisfying $0 < q \le q_0 \le \infty$ and $0 < p_0 \le p \le \infty$. We assume now that $\alpha > 0$ and $0 < p_0 \le p \le q \le q_0 \le \infty$, with $q \ge 1$.

For convenience of notation, we extend the definition of the kernel K and the functions χ and f to all real values of t and s by setting them equal to zero whenever t or s fall outside of the interval [0, T).

We begin with the use of a partition

$$\mathbb{1}_{\{t \neq s\}} = \sum_{j \in \mathbb{Z}} \mathbb{1}_{\{2^j \le |t-s| < 2^{j+1}\}}$$

to deduce that

$$\left\| \int_{0}^{T} e^{-\alpha |t-s|} \chi(t,s) K(t,s) f(s) \, ds \right\|_{L^{q}(\mathbb{R};Y)} \leq \sum_{\substack{j \in \mathbb{Z} \\ 2^{j} < T}} e^{-\alpha 2^{j}} \left\| \int_{0}^{T} \chi_{j}(t,s) K(t,s) f(s) \, ds \right\|_{L^{q}(\mathbb{R};Y)}, \quad (2-2)$$

where we have defined

$$\chi_j(t,s) = \mathbb{1}_{\{2^j \le |t-s| < 2^{j+1}\}} e^{-\alpha(|t-s|-2^j)} \chi(t,s).$$

Observe that $\|\chi_j\|_{L^{\infty}} \leq 1$.

Then, we further decompose the domain of t into the disjoint union

$$\bigcup_{k \in \mathbb{Z}} \{ 2^j k \le t < 2^j (k+1) \}$$

to write

$$\left\| \int_{0}^{T} \chi_{j}(t,s) K(t,s) f(s) ds \right\|_{L^{q}(\mathbb{R};Y)} = \left\| \left\| \int_{0}^{T} \chi_{j} Kf(s) ds \right\|_{L^{q}([2^{j}k,2^{j}(k+1));Y)} \right\|_{\ell^{q}(k\in\mathbb{Z})} \le 2^{j\left(\frac{1}{q}-\frac{1}{q_{0}}\right)} \left\| \left\| \int_{0}^{T} \chi_{j} Kf(s) ds \right\|_{L^{q_{0}}([2^{j}k,2^{j}(k+1));Y)} \right\|_{\ell^{q}(k\in\mathbb{Z})}, \quad (2-3)$$

where we employed Hölder's inequality.

Now, notice that

$$2^{j}(k-2) < t - |t-s| \le s \le t + |t-s| < 2^{j}(k+3)$$

whenever $2^j \le |t-s| < 2^{j+1}$ and $2^j k \le t < 2^j (k+1)$. In particular, using (2-1), it follows that

$$\begin{split} \left\| \int_{0}^{T} \chi_{j} Kf(s) \, ds \right\|_{L^{q_{0}}([2^{j}k, 2^{j}(k+1)); Y)} &\leq \left\| \int_{0}^{T} \chi_{j} Kf(s) \mathbb{1}_{\{2^{j}(k-2) < s < 2^{j}(k+3)\}} \, ds \right\|_{L^{q_{0}}(\mathbb{R}; Y)} \\ &\leq A \sum_{n=-2}^{2} \| f \|_{L^{p_{0}}([2^{j}(k+n), 2^{j}(k+1+n)); X)} \\ &\leq A 2^{j\left(\frac{1}{p_{0}} - \frac{1}{p}\right)} \sum_{n=-2}^{2} \| f \|_{L^{p}([2^{j}(k+n), 2^{j}(k+1+n)); X)}, \quad (2-4) \end{split}$$

where we applied Hölder's inequality again.

All in all, combining (2-2), (2-3) with (2-4), and recalling that $\ell^p \subset \ell^q$ because $p \leq q$, we infer that

$$\begin{split} \left\| \int_{0}^{T} e^{-\alpha |t-s|} \chi(t,s) K(t,s) f(s) \, ds \right\|_{L^{q}(\mathbb{R};Y)} \\ & \leq 5A \sum_{\substack{j \in \mathbb{Z} \\ 2^{j} < T}} e^{-\alpha 2^{j}} 2^{j \left(\frac{1}{q} - \frac{1}{q_{0}} + \frac{1}{p_{0}} - \frac{1}{p}\right)} \| \|f\|_{L^{p}([2^{j}k, 2^{j}(k+1));X)} \|_{\ell^{q}(k \in \mathbb{Z})} \\ & \leq 5A \|f\|_{L^{p}(\mathbb{R};X)} \sum_{\substack{j \in \mathbb{Z} \\ 2^{j} < T}} e^{-\alpha 2^{j}} 2^{j \left(\frac{1}{q} - \frac{1}{q_{0}} + \frac{1}{p_{0}} - \frac{1}{p}\right)}. \end{split}$$
(2-5)

It only remains to evaluate the constant resulting from the above sum in $j \in \mathbb{Z}$. If $p = p_0$ and $q = q_0$, the lemma trivially holds and there is nothing to prove. Thus, we may assume that $\beta > 0$, thereby ensuring that the sum converges.

Now, observing that the function $e^{-x}(1+x)^{1+\beta}$ reaches its maximum on $[0,\infty)$ at $x=\beta$, we obtain

$$e^{\beta}(1+\beta)^{-(1+\beta)} \sum_{\substack{j \in \mathbb{Z} \\ 2^{j} < T}} e^{-\alpha 2^{j}} 2^{j\beta} \leq \sum_{\substack{j \in \mathbb{Z} \\ 2^{j} < T}} \frac{2^{j\beta}}{(1+\alpha 2^{j})^{1+\beta}} \leq \sum_{\substack{j \in \mathbb{Z} \\ 2^{j} < T}} \int_{j-1}^{j} \frac{2^{(1+u)\beta}}{(1+\alpha 2^{u})^{1+\beta}} \, du$$
$$\leq \frac{2^{\beta}}{\log 2} \int_{0}^{T} \frac{x^{\beta-1}}{(1+\alpha x)^{1+\beta}} \, dx = \frac{2^{\beta}}{\beta \log 2} \Big(\frac{T}{1+\alpha T}\Big)^{\beta}. \tag{2-6}$$

Therefore, incorporating (2-6) into the estimate (2-5) concludes the proof of the lemma.

2.2. *Damped parabolic estimates.* Let us consider the general solution w(t, x) of a damped heat equation on the Euclidean space \mathbb{R}^d for any dimension $d \ge 1$

$$\begin{cases} \partial_t w + \alpha w - \Delta w = f, \\ w|_{t=0} = w_0, \end{cases}$$
(2-7)

where $(t, x) \in [0, T) \times \mathbb{R}^d$, with T > 0 $(T = \infty$ may also be considered), the damping constant satisfies $\alpha \ge 0$, the right-hand side f(t, x) is a source term and $w_0(x)$ is an initial datum.

Such equations naturally appear in dissipative physical systems. For instance, the heat equation (2-7) provides the linear structure of the damped incompressible Navier–Stokes equations, which arise from hydrodynamic regimes of inelastic particle systems.

Using standard semigroup notation, the solution w(t, x) can be represented as

$$w(t) = e^{-t(\alpha - \Delta)} w_0 + \int_0^t e^{-(t-s)(\alpha - \Delta)} f(s) \, ds.$$
(2-8)

We are now going explore the jungle of parabolic smoothing estimates in Besov spaces for (2-8) by first reviewing the available results for the case $\alpha = 0$ and, then, extending these results to the setting $\alpha > 0$.

When $f \equiv 0$ and $\alpha = 0$, direct parabolic estimates on the semigroup $e^{t\Delta}$ yield the following result.

Proposition 2.2. Let $\sigma \in \mathbb{R}$, $p \in [1, \infty]$ and $q \in [1, \infty]$. If $\alpha = 0$, w_0 belongs to $\dot{B}_{p,q}^{\sigma}$ and $f \equiv 0$, then the solution of the heat equation (2-7) satisfies

$$\|e^{t\Delta}w_0\|_{L^{\infty}([0,\infty);\dot{B}^{\sigma}_{p,q})} \lesssim \|w_0\|_{\dot{B}^{\sigma}_{p,q}}.$$

Furthermore, if $q < \infty$ *, one also has the estimate*

$$\|e^{t\Delta}w_0\|_{L^q([0,\infty);\dot{B}_{p,1}^{\sigma+2/q})} \lesssim \|w_0\|_{\dot{B}_{p,q}^{\sigma}}.$$

Remark. The above result somewhat reinforces the estimate

$$\|e^{t\Delta}w_0\|_{L^q([0,\infty);L^p)} \lesssim \|w_0\|_{\dot{B}^{-2/q}_{p,q}}$$

for any $1 \le p, q \le \infty$, which is commonly found in the literature; see [Bahouri et al. 2011, Theorem 2.34], for instance.

Remark. Note that taking p = q = 2 in the above proposition yields the estimate

$$\|e^{t\Delta}w_0\|_{L^2([0,\infty);\dot{B}_{2,1}^{\sigma+1})} \lesssim \|w_0\|_{\dot{H}^{\sigma}},$$

where we used that $\dot{H}^{\sigma} = \dot{B}_{2,2}^{\sigma}$ (see Appendix A for a precise definition of all relevant homogeneous spaces).

Remark. Throughout this section, we will routinely use the basic estimate

$$\|e^{t\Delta}\Delta_{k}u\|_{L^{p}} \le Ce^{-C_{*}t2^{2k}}\|\Delta_{k}u\|_{L^{p}}$$
(2-9)

for any t > 0, $p \in [1, \infty]$ and any dyadic block Δ_k , with $k \in \mathbb{Z}$, where *C* and *C*_{*} are positive independent constants. We refer to [Bahouri et al. 2011, Lemma 2.4] for a justification of (2-9).

Proof. The first part of the statement is a straightforward consequence of the definition of the homogeneous Besov norm. More precisely, using (2-9), we obtain

$$\|e^{t\Delta}w_{0}\|_{\dot{B}^{\sigma}_{p,q}} = \left(\sum_{k\in\mathbb{Z}} (2^{k\sigma}\|e^{t\Delta}\Delta_{k}w_{0}\|_{L^{p}})^{q}\right)^{\frac{1}{q}} \lesssim \left(\sum_{k\in\mathbb{Z}} e^{-C_{*}t2^{2k}} (2^{k\sigma}\|\Delta_{k}w_{0}\|_{L^{p}})^{q}\right)^{\frac{1}{q}} \lesssim \|w_{0}\|_{\dot{B}^{\sigma}_{p,q}},$$

which, upon taking the supremum in t > 0, concludes the justification of the first estimate.

The second part of the statement is more subtle. Indeed, assuming now that $q < \infty$ and using (2-9), we find that

$$\|e^{t\Delta}w_{0}\|_{\dot{B}^{\sigma+2/q}_{p,1}} = \sum_{k\in\mathbb{Z}} 2^{k(\sigma+\frac{2}{q})} \|e^{t\Delta}\Delta_{k}w_{0}\|_{L^{p}} \lesssim \sum_{k\in\mathbb{Z}} e^{-C_{*}t2^{2k}} 2^{k(\sigma+\frac{2}{q})} \|\Delta_{k}w_{0}\|_{L^{p}}$$

Next, further employing Hölder's inequality and taking a fixed positive value $\lambda > 0$ such that $(q-1)\lambda < 1$, we infer that

$$\|e^{t\Delta}w_{0}\|_{\dot{B}^{\sigma+2/q}_{p,1}} \lesssim \left(\sum_{k\in\mathbb{Z}} (t2^{2k})^{\lambda} e^{-C_{*}t2^{2k}}\right)^{\frac{q-1}{q}} \left(\sum_{k\in\mathbb{Z}} \frac{1}{t} e^{-C_{*}t2^{2k}} (t2^{2k})^{1-\lambda(q-1)} (2^{k\sigma}\|\Delta_{k}w_{0}\|_{L^{p}})^{q}\right)^{\frac{1}{q}}.$$
 (2-10)

Now, for any positive t, considering the unique $j \in \mathbb{Z}$ such that $2^{2j} \leq t < 2^{2(j+1)}$, we find, since $\lambda > 0$, that

$$\sup_{t>0} \sum_{k\in\mathbb{Z}} (t2^{2k})^{\lambda} e^{-C_* t2^{2k}} \le 2^{2\lambda} \sup_{j\in\mathbb{Z}} \sum_{k\in\mathbb{Z}} (2^{2(j+k)})^{\lambda} e^{-C_* 2^{2(j+k)}} = 2^{2\lambda} \sum_{k\in\mathbb{Z}} (2^{2k})^{\lambda} e^{-C_* 2^{2k}} < \infty, \quad (2\text{-}11)$$

whereas, since $\lambda(q-1) < 1$, we evaluate

$$\int_0^\infty e^{-C_*t2^{2k}}(t2^{2k})^{1-\lambda(q-1)}\frac{dt}{t} = \int_0^\infty e^{-C_*t}t^{-\lambda(q-1)}\,dt < \infty.$$

Therefore, integrating (2-10) in time, we finally arrive at the estimate

$$\begin{aligned} \|e^{t\Delta}w_{0}\|_{L^{q}_{t}\dot{B}^{\sigma+2/q}_{p,1}} \\ &\lesssim \sup_{t>0} \left(\sum_{k\in\mathbb{Z}} (t2^{2k})^{\lambda} e^{-C_{*}t2^{2k}}\right)^{\frac{q-1}{q}} \left(\sum_{k\in\mathbb{Z}} \int_{0}^{\infty} e^{-C_{*}t2^{2k}} (t2^{2k})^{1-\lambda(q-1)} \frac{dt}{t} (2^{k\sigma} \|\Delta_{k}w_{0}\|_{L^{p}})^{q}\right)^{\frac{1}{q}} \\ &\lesssim \|w_{0}\|_{\dot{B}^{\sigma}_{p,q}}, \end{aligned}$$

which concludes the proof of the proposition.

In view of the preceding result, the effect of the damping term $e^{-\alpha t}$ on the initial data can be taken into account through a straightforward application of Hölder's inequality, thereby providing the following corollary.

Corollary 2.3. Let $\sigma \in \mathbb{R}$, $p \in [1, \infty]$ and $q \in [1, \infty]$. If $\alpha \ge 0$, w_0 belongs to $\dot{B}_{p,q}^{\sigma}$ and $f \equiv 0$, then the solution of the heat equation (2-7) satisfies

$$\|e^{-t(\alpha-\Delta)}w_0\|_{L^m([0,T);\dot{B}^{\sigma}_{p,q})} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{m}} \|w_0\|_{\dot{B}^{\sigma}_{p,q}}$$

for every $0 < m \le \infty$. Furthermore, if $0 < m \le q < \infty$, one also has the estimate

$$\|e^{-t(\alpha-\Delta)}w_0\|_{L^m([0,T);\dot{B}_{p,1}^{\sigma+2/q})} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{m}-\frac{1}{q}} \|w_0\|_{\dot{B}_{p,q}^{\sigma}}.$$

Proof. A direct use of Hölder's inequality followed by an application of Proposition 2.2 yields

$$\|e^{-t(\alpha-\Delta)}w_0\|_{L^m([0,T);\dot{B}^{\sigma}_{p,q})} \le \|e^{-t\alpha}\|_{L^m([0,T))}\|e^{t\Delta}w_0\|_{L^{\infty}([0,T);\dot{B}^{\sigma}_{p,q})} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{m}}\|w_0\|_{\dot{B}^{\sigma}_{p,q}}$$

for all $0 < m \le \infty$ and

$$\begin{aligned} \|e^{-t(\alpha-\Delta)}w_0\|_{L^m([0,T);\dot{B}^{\sigma+2/q}_{p,1})} &\leq \|e^{-t\alpha}\|_{L^{(1/m-1/q)-1}([0,T))} \|e^{t\Delta}w_0\|_{L^q([0,T);\dot{B}^{\sigma+2/q}_{p,1})} \\ &\lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{m}-\frac{1}{q}} \|w_0\|_{\dot{B}^{\sigma}_{p,q}} \end{aligned}$$

for all $0 < m \le q < \infty$, which completes the proof.

Parabolic estimates are more involved when one includes a nonzero source term f. The coming results contain a wide range of smoothing estimates for the inhomogeneous heat equation. In preparation of these results, in order to reach a broader range of applicability, we are now going to introduce symbols

$$a(t,s,\xi) \in L^{\infty}([0,T) \times [0,T) \times \mathbb{R}^{d}),$$

which act as multipliers on the Fourier variable $\xi \in \mathbb{R}^d$ and are dependent on the time variables $t, s \in [0, T)$, thereby leading to time-dependent Fourier multipliers a(t, s, D).

Definition. For a given $1 \le p \le \infty$, we say that a(t, s, D) is *bounded* if there is a constant $C_a > 0$, independent of t and s, such that

$$\|a(t,s,D)f\|_{\dot{B}^{0}_{p,\infty}} \le C_{a} \|f\|_{\dot{B}^{0}_{p,\infty}}$$
(2-12)

for every $f \in \dot{B}_{p,\infty}^0(\mathbb{R}^d)$ and almost every $(t,s) \in [0,T)^2$. That is, the multiplier a(t,s,D) is bounded if it is bounded over the Besov space $\dot{B}_{p,\infty}^0(\mathbb{R}^d)$, uniformly in *t* and *s*. The *norm* of a(t,s,D), which we denote by

$$\|a(t,s,D)\|_{M_p},$$

is defined as the smallest possible constant $C_a > 0$ that fits in (2-12).

Remark. Equivalently, it is readily seen that (2-12) holds if and only if there is a constant $C'_a > 0$, independent of t and s, such that

$$\|a(t,s,D)\Delta_k f\|_{L^p} \le C'_a \|f\|_{L^p}$$
(2-13)

for every $k \in \mathbb{Z}$, $f \in L^p(\mathbb{R}^d)$ and almost every $(t, s) \in [0, T)^2$.

Remark. Observe that (2-12) and (2-13) hold if and only if one has

$$\|a(t,s,D)f\|_{\dot{B}^{\sigma}_{p,q}} \leq C_{a,\sigma,q} \|f\|_{\dot{B}^{\sigma}_{p,q}},$$

with $C_{a,\sigma,q} > 0$, for all $\sigma \in \mathbb{R}$, $q \in [1,\infty]$ and every $f \in \dot{B}_{p,q}^{\sigma}(\mathbb{R}^d)$.

Since the space of Fourier multipliers over $L^2(\mathbb{R}^d)$ is isomorphic to $L^{\infty}(\mathbb{R}^d)$, it is readily seen, when p = 2, that proving (2-12) and (2-13) is equivalent to establishing a bound

$$a(t, s, \xi) \in L^{\infty}([0, T) \times [0, T) \times \mathbb{R}^d).$$

More generally, when $p \neq 2$, in order to ensure that (2-12) or (2-13) hold, it is sufficient to require that

$$\mathcal{F}^{-1}[a(t,s,\xi)\varphi(2^{-k}\xi)] \in L^1(\mathbb{R}^d),$$

uniformly in t, s and k, where $\varphi(2^{-k}\xi)$ is a smooth compactly supported cutoff function used to define a Littlewood–Paley dyadic decomposition (see Appendix A). Therefore, in view of the straightforward classical estimate

$$\begin{split} \|\mathcal{F}^{-1}[a(t,s,\xi)\varphi(2^{-k}\xi)](x)\|_{L^{1}_{x}} &\lesssim 2^{-k\frac{d}{2}} \|(1+2^{k}|x|)^{N} \mathcal{F}^{-1}[a(t,s,\xi)\varphi(2^{-k}\xi)](x)\|_{L^{2}_{x}} \\ &\lesssim 2^{-k\frac{d}{2}} \sum_{\substack{\alpha \in \mathbb{N}^{d} \\ |\alpha| \leq N}} \|2^{k|\alpha|} \partial_{\xi}^{\alpha}[a(t,s,\xi)\varphi(2^{-k}\xi)]\|_{L^{2}_{\xi}} \\ &\lesssim 2^{-k\frac{d}{2}} \sum_{\substack{\alpha,\beta \in \mathbb{N}^{d} \\ |\alpha|+|\beta| \leq N}} \|2^{k|\alpha|} \partial_{\xi}^{\alpha}a(t,s,\xi)(\partial^{\beta}\varphi)(2^{-k}\xi)\|_{L^{2}_{\xi}}, \end{split}$$

where we have used Plancherel's theorem and N is any integer larger than $\frac{1}{2}d$, we see that (2-12) and (2-13) both hold as soon as $a(t, s, \xi)$ is sufficiently differentiable in ξ (except possibly at the origin $\xi = 0$) and satisfies the estimate

$$\||\xi|^{|\alpha|}\partial_{\xi}^{\alpha}a(t,s,\xi)\|_{L^{\infty}_{t,s,\xi}} < \infty$$
(2-14)

for every multi-index $\alpha \in \mathbb{N}^d$ with $|\alpha| \leq \left[\frac{1}{2}d\right] + 1$. Observe that the above criterion establishes the boundedness of a(t, s, D) over $\dot{B}_{p,\infty}^0(\mathbb{R}^d)$, uniformly in *t* and *s*, for all values of $1 \leq p \leq \infty$, including the endpoints. Later on, we will be making use of (2-14) to show the boundedness of multipliers.

We return now to the smoothing estimates for the heat equation with a nontrivial source term f. The next result provides a large array of such estimates in the classical case $\alpha = 0$.

Proposition 2.4. Let $\sigma \in \mathbb{R}$, $1 < r < m < \infty$ and $p \in [1, \infty]$. If f belongs to $L^r([0, T); \dot{B}_{p,\infty}^{\sigma+2/r})$ and $w_0 = 0$, then the solution of the heat equation (2-7), with $\alpha = 0$, satisfies

$$\left\|\int_{0}^{t} e^{(t-s)\Delta} a(t,s,D) f(s) \, ds\right\|_{L^{m}([0,T), \dot{B}_{p,1}^{\sigma+2+2/m})} \lesssim \|a\|_{M_{p}} \|f\|_{L^{r}([0,T), \dot{B}_{p,\infty}^{\sigma+2/r})}$$

for any Fourier multiplier a(t, s, D).

Remark. We refer to [Arsénio 2019, Lemma 2] for a complete justification of the preceding proposition in the case a(t, s, D) = Id. A straightforward adaptation of this proof readily extends the result to nontrivial multipliers a(t, s, D).

Remark. The endpoint case r = m above corresponds formally to a maximal gain of two derivatives on the solution of the heat equation. However, the method of proof of this result relies on the Hardy– Littlewood–Sobolev inequality, which typically falls short for endpoint settings. It is therefore not possible to extend the proof of [Arsénio 2019, Lemma 2] to the case r = m. The next result generalizes Proposition 2.4 to incorporate the action of a damping term.

Proposition 2.5. Let $\sigma \in \mathbb{R}$, $p \in [1, \infty]$ and

$$1 \le r \le m \le \infty$$
, $0 < \theta < 1 + \frac{1}{m} - \frac{1}{r} \le 1$,

or

$$1 < r < m < \infty$$
, $0 < \theta = 1 + \frac{1}{m} - \frac{1}{r} < 1$

Then, for any $\alpha \geq 0$ *, one has the estimate*

$$\left\| \int_{0}^{t} e^{-(t-s)(\alpha-\Delta)} a(t,s,D) f(s) \, ds \right\|_{L^{m}([0,T),\dot{B}_{p,1}^{\sigma+2\theta})} \lesssim \left(\frac{T}{1+\alpha T} \right)^{1+\frac{1}{m}-\frac{1}{r}-\theta} \|a\|_{M_{p}} \|f\|_{L^{r}([0,T),\dot{B}_{p,\infty}^{\sigma})}$$
(2-15)

for any f in $L^r([0,T); \dot{B}^{\sigma}_{p,\infty})$ and any Fourier multiplier a(t,s,D).

Remark. We emphasize that any implicit constant involved in the estimate of Proposition 2.5 is independent of T and α . Moreover, it is permitted to set $T = \infty$ and $\alpha > 0$ therein, in order to deduce a global estimate.

Remark. Observe that, choosing any $1 \le r \le m \le \infty$, $1 \le p, q \le \infty$ and $\sigma \in \mathbb{R}$, one has the simple estimate

$$\begin{split} \left\| \int_{0}^{t} e^{-(t-s)(\alpha-\Delta)} a(t,s,D) f(s) \, ds \right\|_{L^{m}([0,T),\dot{B}^{\sigma}_{p,q})} &\lesssim \|a\|_{M_{p}} \left\| \int_{0}^{t} e^{-\alpha(t-s)} \|f(s)\|_{\dot{B}^{\sigma}_{p,q}} \, ds \right\|_{L^{m}([0,T))} \\ &\lesssim \left(\frac{T}{1+\alpha T}\right)^{1+\frac{1}{m}-\frac{1}{r}} \|a\|_{M_{p}} \|f\|_{L^{r}([0,T),\dot{B}^{\sigma}_{p,q})} \end{split}$$

for all $\alpha \ge 0$, which corresponds to the case $\theta = 0$ in the previous proposition.

Proof in the case $1 < r \le m < \infty$. First of all, notice that the case

$$1 < r < m < \infty$$
, $0 < \theta = 1 + \frac{1}{m} - \frac{1}{r} < 1$

follows from a direct application of Proposition 2.4 by absorbing the damping term $e^{-\alpha(t-s)}$ into the multiplier a(t, s, D).

In order to treat the remaining case

$$1 < r \le m < \infty, \quad 0 < \theta < 1 + \frac{1}{m} - \frac{1}{r} \le 1,$$

we introduce auxiliary parameters

$$1 < r_0 < r \le m < m_0 < \infty$$

such that

$$\theta = 1 + \frac{1}{m_0} - \frac{1}{r_0}.$$

In particular, in view of Proposition 2.4, we have

$$\left\|\int_0^t e^{(t-s)\Delta}a(t,s,D)f(s)\,ds\right\|_{L^{m_0}([0,T),\dot{B}_{p,1}^{\sigma+2\theta})} \lesssim \|a\|_{M_p}\|f\|_{L^{r_0}([0,T),\dot{B}_{p,\infty}^{\sigma})}$$

Then, an application of the damping lemma (Lemma 2.1) implies, for any $\alpha \ge 0$, that (2-15) holds, thereby concluding the proof.

For the sake of completeness, since the preceding proof fails to treat the cases r = 1 and $m = \infty$, we provide now an alternative justification of Proposition 2.5, based on the proof of Lemma 2 from [Arsénio 2019], which works in full generality.

General proof. Following [Arsénio 2019], we begin by using (2-9) and (2-13) to deduce the existence of an independent constant $C_* > 0$ such that

$$\left\|\Delta_k \int_0^t e^{-(t-s)(\alpha-\Delta)} a(t,s,D) f(s) \, ds\right\|_{L^p} \lesssim \int_0^t e^{-(t-s)(\alpha+C_*2^{2k})} \|\Delta_k f(s)\|_{L^p} \, ds$$

For simplicity, we omit the norm $||a||_{M_p}$, which we absorb in the implicit constants. It then follows that

$$\begin{split} \left\| \int_0^t e^{-(t-s)(\alpha-\Delta)} a(t,s,D) f(s) \, ds \right\|_{\dot{B}^{\sigma+2\theta}_{p,1}} \lesssim \int_0^t \sum_{k \in \mathbb{Z}} e^{-(t-s)(\alpha+C_*2^{2k})} 2^{k(\sigma+2\theta)} \|\Delta_k f(s)\|_{L^p} \, ds \\ \lesssim \int_0^T |t-s|^{-\theta} e^{-\alpha(t-s)} \|f(s)\|_{\dot{B}^{\sigma}_{p,\infty}} \, ds, \end{split}$$

where we have used (2-11), with the assumption that $\theta > 0$, to deduce that

$$\sum_{k\in\mathbb{Z}} 2^{2k\theta} e^{-C_*(t-s)2^{2k}} \lesssim |t-s|^{-\theta}.$$

Next, if $\theta = 1 + 1/m - 1/r$, by virtue of the Hardy–Littlewood–Sobolev inequality, which holds because $0 < \theta < 1$ and $1 < m, r < \infty$, we infer that

$$\left\|\int_{0}^{t} e^{-(t-s)(\alpha-\Delta)}a(t,s,D)f(s)\,ds\right\|_{L^{m}\dot{B}^{\sigma+2\theta}_{p,1}} \lesssim \left\|\int_{0}^{T} |t-s|^{-\theta}\|f(s)\|_{\dot{B}^{\sigma}_{p,\infty}}\,ds\right\|_{L^{m}} \lesssim \|f\|_{L^{r}\dot{B}^{\sigma}_{p,\infty}}.$$

Similarly, if $0 < \theta < 1 + 1/m - 1/r \le 1$, we deduce from Young's convolution inequality that

$$\begin{split} \left\| \int_{0}^{t} e^{-(t-s)(\alpha-\Delta)} a(t,s,D) f(s) \, ds \right\|_{L^{m} \dot{B}^{\sigma+2\theta}_{p,1}} \lesssim \left\| \int_{0}^{T} |t-s|^{-\theta} e^{-\alpha(t-s)} \| f(s) \|_{\dot{B}^{\sigma}_{p,\infty}} \, ds \right\|_{L^{m}} \\ \lesssim \left(\int_{0}^{T} (t^{-\theta} e^{-\alpha t})^{\left(1+\frac{1}{m}-\frac{1}{r}\right)^{-1}} dt \right)^{1+\frac{1}{m}-\frac{1}{r}} \| f \|_{L^{r} \dot{B}^{\sigma}_{p,\infty}} \\ \lesssim \left(\frac{T}{1+\alpha T} \right)^{1+\frac{1}{m}-\frac{1}{r}-\theta} \| f \|_{L^{r} \dot{B}^{\sigma}_{p,\infty}}, \end{split}$$

which concludes the proof of the proposition.

The shortcomings of Propositions 2.4 and 2.5 in the case r = m, with $\theta = 1$, naturally bring the question of the maximal regularity of the Laplacian in Banach spaces.

For a given Banach space X such that the Laplacian operator Δ is defined on a dense subspace of X, we say that the Laplacian (or another elliptic operator) has maximal L^p -regularity on [0, T) for some $1 if the solution (2-8) of the heat equation (without damping, i.e., <math>\alpha = 0$) for a null initial data, i.e., $w_0 = 0$, is differentiable almost everywhere in t, takes values almost everywhere in the domain of Δ and satisfies the estimate

$$\|\partial_t w\|_{L^p([0,T);X)} + \|\Delta w\|_{L^p([0,T);X)} \le C_p \|f\|_{L^p([0,T);X)}$$

for any source term $f \in L^p([0, T); X)$. We refer to [Kunstmann and Weis 2004] for an introduction to the theory of maximal L^p -regularity for parabolic equations.

The next important result, extracted from [Arsénio and Gallagher 2020], establishes the maximal regularity of the Laplacian in all homogeneous Besov spaces $\dot{B}_{p,q}^{\sigma}$. In particular, this result provides the basis which will allow us (in Section 3.6, for instance) to obtain stronger estimates, with sharp gains of parabolic regularity, by avoiding the use of Chemin–Lerner spaces.

Proposition 2.6 [Arsénio and Gallagher 2020]. Let $\sigma \in \mathbb{R}$, $p, q \in [1, \infty]$ and $r \in (1, \infty)$. If f belongs to $L^r([0, T); \dot{B}_{p,q}^{\sigma})$ and $w_0 = 0$, then the solution of the heat equation (2-7), with $\alpha \ge 0$, satisfies

$$\left\|\int_0^t e^{-(t-s)(\alpha-\Delta)}a(t,s,D)f(s)\,ds\right\|_{L^r([0,T),\dot{B}^{\sigma+2}_{p,q})} \lesssim \|a\|_{M_p}\|f\|_{L^r([0,T),\dot{B}^{\sigma}_{p,q})}$$

for any Fourier multiplier a(t, s, D). The result remains valid if r = q = 1 or $r = q = \infty$.

Remark. Again, it is to be emphasized that any implicit constant involved in the above estimate is independent of T and α . In particular, one can set $T = \infty$ therein.

Remark. We refer to [Arsénio and Gallagher 2020, Proposition 3.1] or [Arsénio 2019, Lemma 3] for a proof of Proposition 2.6 in the case a(t, s, D) = Id and $\alpha = 0$. The original proof from [Arsénio and Gallagher 2020] deals first with the case q = 1 and then relies on an interpolation argument. The proof from [Arsénio 2019], however, offers a self-contained approach which avoids interpolation altogether.

Remark. In fact, the original statements of Proposition 3.1 in [Arsénio and Gallagher 2020] and Lemma 3 in [Arsénio 2019] only cover the range of parameters $1 \le q \le r < \infty$. Nevertheless, it is readily seen that the corresponding proofs can be used mutatis mutandis to show identical bounds on the adjoint operator, which is defined by

$$\int_t^T e^{-(s-t)(\alpha-\Delta)} \overline{a(s,t,D)} g(s) \, ds.$$

It then follows from a standard duality argument that these results hold for values $1 < r \le q \le \infty$ as well.

For the sake of completeness and clarity, we provide now a full justification of Proposition 2.6, based on a straightforward adaptation of the proof of [Arsénio 2019, Lemma 3].

Proof. First, noticing that the damping term $e^{-\alpha(t-s)}$ can be absorbed into the bounded multiplier a(t, s, D), we assume, without loss of generality, that $\alpha = 0$. Then, we follow the proof of Lemma 3 from [Arsénio 2019].

We start by considering the case $1 \le q \le r < \infty$. By duality, it is enough to prove that, if g is a nonnegative function in $L^{b'}([0,T))$ with $b = r/q \ge 1$ and 1/b + 1/b' = 1, then

$$\int_0^T g(t) \left\| \int_0^t e^{(t-s)\Delta} a(t,s,D) f(s) \, ds \right\|_{\dot{B}^{\sigma+2}_{p,q}}^q dt \lesssim \|a\|_{M_p}^q \|f\|_{L^r([0,T),\dot{B}^{\sigma}_{p,q})}^q \|g\|_{L^{b'}([0,T))}^q dt$$

To this end, using (2-9) and (2-13), we deduce the existence of a constant $C_* > 0$ such that

$$\begin{split} \int_{0}^{T} g(t) \left\| \int_{0}^{t} e^{(t-s)\Delta} a(t,s,D) f(s) \, ds \right\|_{\dot{B}_{p,q}^{\sigma+2}}^{q} dt \\ &= \sum_{k \in \mathbb{Z}} \int_{0}^{T} g(t) \left\| \int_{0}^{t} e^{(t-s)\Delta} a(t,s,D) \Delta_{k} f(s) \, ds \right\|_{L^{p}}^{q} 2^{k(\sigma+2)q} \, dt \\ &\lesssim \|a\|_{M_{p}}^{q} \sum_{k \in \mathbb{Z}} \int_{0}^{T} g(t) \left(\int_{0}^{t} e^{-C_{*}(t-s)2^{2k}} \|\Delta_{k} f(s)\|_{L^{p}} \, ds \right)^{q} 2^{k(\sigma+2)q} \, dt \\ &\lesssim \|a\|_{M_{p}}^{q} \sum_{k \in \mathbb{Z}} \int_{0}^{T} g(t) \int_{0}^{t} e^{-C_{*}(t-s)2^{2k}} \|\Delta_{k} f(s)\|_{L^{p}}^{q} \, ds \, 2^{k(\sigma+2)q} \, dt. \end{split}$$

For simplicity, we omit the norm $||a||_{M_p}$ in the remaining estimates.

Next, we define a maximal operator by

$$Mg(s) = \sup_{\rho > 0} \rho \int_0^T e^{-\rho|s-t|} |g(t)| \, dt.$$

Classical results from harmonic analysis (see [Grafakos 2014, Theorems 2.1.6 and 2.1.10]) establish that *M* is bounded over $L^{c}([0, T))$ for any $1 < c \le \infty$. One can then write

$$\begin{split} \int_{0}^{T} g(t) \left\| \int_{0}^{t} e^{(t-s)\Delta} a(t,s,D) f(s) \, ds \right\|_{\dot{B}^{\sigma+2}_{p,q}}^{q} dt \\ \lesssim \sum_{k \in \mathbb{Z}} \int_{0}^{T} \left[2^{2k} \int_{s}^{T} g(t) e^{-C_{*}(t-s)2^{2k}} \, dt \right] \|\Delta_{k} f(s)\|_{L^{p}}^{q} 2^{k\sigma q} \, ds \\ \lesssim \sum_{k \in \mathbb{Z}} \int_{0}^{T} Mg(s) \|\Delta_{k} f(s)\|_{L^{p}}^{q} 2^{k\sigma q} \, ds = \int_{0}^{T} Mg(s) \|f(s)\|_{\dot{B}^{\sigma}_{p,q}}^{q} \, ds \end{split}$$

Therefore, by the boundedness properties of Mg and Hölder's inequality, we conclude that

$$\int_0^T g(t) \left\| \int_0^t e^{(t-s)\Delta} a(t,s,D) f(s) \, ds \right\|_{\dot{B}^{\sigma+2}_{p,q}}^q dt \lesssim \|Mg\|_{L^{b'}} \|f\|_{L^r \dot{B}^{\sigma}_{p,q}}^q \lesssim \|g\|_{L^{b'}} \|f\|_{L^r \dot{B}^{\sigma}_{p,q}}^q,$$

which completes the proof of the proposition in the case $1 \le q \le r < \infty$.

Now, observe that the exact same proof applies to the adjoint operator

$$\int_t^T e^{(s-t)\Delta} \overline{a(s,t,D)} f(s) \, ds,$$

thereby leading to the estimate

$$\left\|\int_{t}^{T} e^{(s-t)\Delta} \overline{a(s,t,D)} f(s) \, ds\right\|_{L^{r}([0,T),\dot{B}_{p,q}^{\sigma+2})} \lesssim \|f\|_{L^{r}([0,T),\dot{B}_{p,q}^{\sigma})}$$

whenever $1 \le q \le r < \infty$. Then, a standard duality argument establishes that

$$\left\|\int_0^t e^{(t-s)\Delta}a(t,s,D)f(s)\,ds\right\|_{L^{r'}([0,T),\dot{B}_{p',q'}^{-\sigma})} \lesssim \|f\|_{L^{r'}([0,T),\dot{B}_{p',q'}^{-(\sigma+2)})}$$

for parameter values in the range $1 < r' \le q' \le \infty$. Therefore, replacing p, q, r and σ by p', q', r' and $-(\sigma + 2)$, respectively, shows the proposition in the case $1 < r \le q \le \infty$, which concludes the proof. \Box

2.3. *Damped Strichartz estimates.* We focus now on the interaction between damping and dispersion. More precisely, we are going to explore how the damping lemma (Lemma 2.1) applies to Strichartz estimates. To that end, we first recall the general result on Strichartz estimates for abstract semigroups from [Keel and Tao 1998]. We also refer to [Bahouri et al. 2011, Chapter 8] for a comprehensive exposition of Strichartz estimates.

Proposition 2.7 [Keel and Tao 1998]. Let *H* be a Hilbert space and (X, dx) be a measure space. For each $t \in [0, T)$, with T > 0, let $U(t) : H \to L^2(X)$ be an operator such that

$$U(t) \in L^{\infty}([0,T); \mathcal{L}(H, L^{2}(X)))$$

and, for some $\sigma > 0$,

$$||U(t)U(s)^*g||_{L^{\infty}(X)} \lesssim \frac{1}{|t-s|^{\sigma}} ||g||_{L^1(X)}$$

for all $t, s \in [0, T)$, with $t \neq s$, and all $g \in L^1(X) \cap L^2(X)$. Then, the estimate

$$\|U(t)f\|_{L^q_t L^r_x} \lesssim \|f\|_H$$

and its dual version

$$\left\| \int_0^T U(t)^* g(t) \, dt \right\|_H \lesssim \|g\|_{L_t^{q'} L_x^{r'}}$$

hold for any exponent pair $(q, r) \in [2, \infty]^2$, which is admissible in the sense that

$$\frac{1}{q} + \frac{\sigma}{r} = \frac{\sigma}{2}$$
 and $(q, r, \sigma) \neq (2, \infty, 1).$

Furthermore, if $(\tilde{q}, \tilde{r}) \in [2, \infty]^2$ *is also an admissible exponent pair, then the estimate*

$$\left\|\int_{0}^{T} \chi(t,s)U(t)U(s)^{*}g(s)\,ds\right\|_{L_{t}^{q}L_{x}^{r}} \lesssim \|\chi\|_{L^{\infty}}\|g\|_{L_{t}^{\bar{q}'}L_{x}^{\bar{r}'}}$$

holds for any $\chi(t,s) \in L^{\infty}([0,T)^2;\mathbb{R})$.

Remark. In fact, the statement of the result from [Keel and Tao 1998] only considers the function $\chi(t, s) = \mathbb{1}_{\{s \le t\}}$. However, a straightforward alteration of the proof from [Keel and Tao 1998] easily shows that the result actually holds for all $\chi(t, s) \in L^{\infty}([0, T)^2; \mathbb{R})$. A detailed proof valid for all $\chi(t, s)$ can also be found in Section 8.2 of [Bahouri et al. 2011].

By combining the damping lemma with the preceding proposition, we obtain the damped Strichartz estimates, which are stated in precise terms in the next result.

Proposition 2.8 (damped Strichartz estimates). Let *H* be a Hilbert space and (X, dx) be a measure space. For each $t \in [0, T)$, with T > 0, let $U(t) : H \to L^2(X)$ be an operator such that

$$U(t) \in L^{\infty}([0,T); \mathcal{L}(H, L^{2}(X)))$$

and, for some $\sigma > 0$,

$$||U(t)U(s)^*g||_{L^{\infty}(X)} \lesssim \frac{1}{|t-s|^{\sigma}} ||g||_{L^1(X)}$$

for all $t, s \in [0, T)$, with $t \neq s$, and all $g \in L^1(X) \cap L^2(X)$.

Then, for any $\alpha \geq 0$ *, the estimate*

$$\|e^{-\alpha t}U(t)f\|_{L^{q}_{t}L^{r}_{x}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{\sigma}{r}-\frac{\sigma}{2}}\|f\|_{H}$$

and its dual version

$$\left\|\int_0^T e^{-\alpha t} U(t)^* g(t) \, dt\right\|_H \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{\sigma}{r}-\frac{\sigma}{2}} \|g\|_{L_t^{q'} L_x^{T}}$$

hold for any exponent pair $(q, r) \in [1, \infty] \times [2, \infty]$, which is admissible in the sense that

$$\frac{1}{q} + \frac{\sigma}{r} \ge \frac{\sigma}{2}, \quad \frac{1}{2} + \frac{\sigma}{r} \ge \frac{\sigma}{2} \quad and \quad (r, \sigma) \ne (\infty, 1).$$

Furthermore, if $(\tilde{q}, \tilde{r}) \in [1, \infty] \times [2, \infty]$ *is also an admissible exponent pair such that*

$$\frac{1}{q} + \frac{1}{\tilde{q}} \le 1,$$

then the estimate

$$\left\|\int_{0}^{T} e^{-\alpha|t-s|}\chi(t,s)U(t)U(s)^{*}g(s)\,ds\right\|_{L_{t}^{q}L_{x}^{r}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{q}+\sigma\left(\frac{1}{r}+\frac{1}{p}\right)-\sigma} \|\chi\|_{L^{\infty}} \|g\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}$$

holds for any $\chi(t,s) \in L^{\infty}([0,T)^2;\mathbb{R})$.

Proof. First of all, observe that all hypotheses of Proposition 2.7 are satisfied by U(t). Then, introducing the parameter $q_0 \in [2, \infty]$ by setting

$$\frac{1}{q_0} = \sigma\left(\frac{1}{2} - \frac{1}{r}\right),\tag{2-16}$$

we see that the exponent pair $(q_0, r) \in [2, \infty]^2$ is admissible for Proposition 2.7. Therefore, it follows from Proposition 2.7, with an application of Hölder's inequality, that

$$\|e^{-\alpha t}U(t)f\|_{L^{q}_{t}L^{r}_{x}} \leq \|e^{-\alpha t}\|_{L^{(1/q-1/q_{0})^{-1}}([0,T))}\|U(t)f\|_{L^{q_{0}}_{t}L^{r}_{x}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}-\frac{1}{q_{0}}}\|f\|_{H},$$

which establishes the first estimate of the proposition. The second estimate then ensues from a dual reformulation of the first estimate.

It only remains to justify the validity of the third estimate. To that end, employing (2-16), we introduce auxiliary parameters $q_0, \tilde{q}_0 \in [2, \infty]$, so that the exponent pairs $(q_0, r), (\tilde{q}_0, \tilde{r}) \in [2, \infty]^2$ are admissible for Proposition 2.7. In particular, it follows that

$$\left\|\int_0^1 \chi(t,s)U(t)U(s)^*g(s)\,ds\right\|_{L_t^{q_0}L_x^r} \lesssim \|\chi\|_{L^\infty} \|g\|_{L_t^{\tilde{q}'_0}L_x^{\tilde{r}'}}$$

for any $\chi(t,s) \in L^{\infty}([0,T)^2;\mathbb{R})$. Therefore, noticing that $\tilde{q}'_0 \leq \tilde{q}' \leq q \leq q_0$, we conclude from an application of Lemma 2.1 that

$$\left\|\int_0^T e^{-\alpha|t-s|}\chi(t,s)U(t)U(s)^*g(s)\,ds\right\|_{L^q_t L^r_x} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}-\frac{1}{q_0}+\frac{1}{\tilde{q}_0'}-\frac{1}{\tilde{q}'}} \|\chi\|_{L^\infty} \|g\|_{L^{\tilde{q}'}_t L^{\tilde{r}'}_x},$$

which completes the proof.

Remark. We do not make any claim of optimality of Proposition 2.8. It would be interesting, though, to test the sharpness of the admissibility criteria for the exponent pairs (q, r) and (\tilde{q}, \tilde{r}) in connection with the sensitivity in T and α of the estimates.

We proceed now to specific formulations of the damped Strichartz estimates for the Schrödinger and wave equations, as well as for Maxwell's system.

Corollary 2.9 (damped Schrödinger equation). Let $d \ge 1$ and consider a solution u(t, x) of the damped Schrödinger equation

$$\begin{cases} (\partial_t + \alpha - i\Delta)u(t, x) = F(t, x), \\ u(0, x) = f(x), \end{cases}$$

with $\alpha \ge 0$, $t \in [0, T)$ and $x \in \mathbb{R}^d$.

For any exponent pairs $(q, r), (\tilde{q}, \tilde{r}) \in [1, \infty] \times [2, \infty]$ which are admissible in the sense that

$$\frac{2}{q} + \frac{d}{r} \ge \frac{d}{2}, \quad 1 + \frac{d}{r} \ge \frac{d}{2} \quad and \quad (r, d) \neq (\infty, 2),$$

and similarly for (\tilde{q}, \tilde{r}) , and such that

$$\frac{1}{q} + \frac{1}{\tilde{q}} \le 1,$$

one has the estimate

$$\|u\|_{L^{q}_{t}L^{r}_{x}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|f\|_{L^{2}_{x}} + \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{\tilde{q}}+\frac{d}{2}\left(\frac{1}{r}+\frac{1}{\tilde{r}}-1\right)} \|F\|_{L^{\tilde{q}'}_{t}L^{\tilde{r}'}_{x}}.$$

Proof. The solution u(t, x) can be expressed by Duhamel's representation formula as

$$u(t) = e^{-\alpha t} U(t) f + \int_0^t e^{-\alpha (t-s)} U(t) U(s)^* F(s) \, ds = e^{-\alpha t} U(t) f + \int_0^t e^{-\alpha (t-s)} U(t-s) F(s) \, ds$$

where

$$U(t) = e^{it\Delta}$$
 and $U(t)^* = e^{-it\Delta}$.

 \square

In particular, one has the explicit formula (see Section 8.1.2 in [Bahouri et al. 2011])

$$U(t)f(x) = \frac{1}{(4\pi i t)^{d/2}} \int_{\mathbb{R}^d} e^{-\frac{|x-y|^2}{4it}} f(y) \, dy,$$

which readily implies that U(t) satisfies all hypotheses of Proposition 2.8 with $H = L^2(\mathbb{R}^d)$, $X = \mathbb{R}^d$ and $\sigma = \frac{1}{2}d$. Therefore, we conclude that the corollary follows from a direct application of the damped Strichartz estimates of Proposition 2.8.

Corollary 2.10 (damped half-wave equation). Let $d \ge 2$, and consider a solution u(t, x) of the damped half-wave equation

$$\begin{cases} (\partial_t + \alpha \mp i |D|)u(t, x) = F(t, x), \\ u(0, x) = f(x), \end{cases}$$

with $\alpha \geq 0$, $t \in [0, T)$ and $x \in \mathbb{R}^d$.

For any exponent pairs $(q, r), (\tilde{q}, \tilde{r}) \in [1, \infty] \times [2, \infty]$ which are admissible in the sense that

$$\frac{2}{q} + \frac{d-1}{r} \ge \frac{d-1}{2}, \quad 1 + \frac{d-1}{r} \ge \frac{d-1}{2} \quad and \quad (r,d) \neq (\infty,3),$$

and similarly for (\tilde{q}, \tilde{r}) , and such that

$$\frac{1}{q} + \frac{1}{\tilde{q}} \le 1,$$

one has the estimate

$$2^{-j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}u\|_{L_{t}^{q}L_{x}^{r}} \\ \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|\Delta_{j}f\|_{L_{x}^{2}} + \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{r}-1\right)} 2^{j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}$$

for all $j \in \mathbb{Z}$.

Remark. If $\tilde{q}' \leq p \leq q$, then further multiplying the preceding estimate by $2^{j\sigma}$ for some $\sigma \in \mathbb{R}$ and summing over $j \in \mathbb{Z}$ in the ℓ^p -norm leads to

$$\begin{aligned} \|u\|_{L^{q}_{t}}\dot{B}^{\sigma-((d+1)/2)(1/2-1/r)}_{r,p,x} \\ \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|f\|_{\dot{B}^{\sigma}_{2,p,x}} + \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{\tilde{q}}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{\tilde{r}}-1\right)} \|F\|_{L^{\tilde{q}'}_{t}}\dot{B}^{\sigma+((d+1)/2)(1/2-1/\tilde{r})}_{r,p,x} \end{aligned}$$

Proof. The solution u(t, x) can be expressed by Duhamel's representation formula as

$$u(t) = e^{-\alpha t} e^{\pm it|D|} f + \int_0^t e^{-\alpha(t-s)} e^{\pm i(t-s)|D|} F(s) \, ds$$

For each $j \in \mathbb{Z}$, we introduce now the flow

$$U_j(t)f(x) := e^{\pm it|D|}\psi\left(\frac{D}{2^j}\right)f(x),$$

where $\psi(\xi)$ is a smooth compactly supported function such that $0 \notin \text{supp } \psi$ and $\psi \equiv 1$ on $\{\frac{1}{2} \le |\xi| \le 2\}$. In particular, if Δ_j is the Littlewood–Paley frequency cutoff operator, defined in Appendix A, which localizes frequencies to $\{2^{j-1} \le |\xi| \le 2^{j+1}\}$, one has the representation

$$\Delta_j u(t) = e^{-\alpha t} U_j(t) \Delta_j f + \int_0^t e^{-\alpha (t-s)} U_j(t-s) \Delta_j F(s) ds$$
$$= e^{-\alpha t} U_j(t) \Delta_j f + \int_0^t e^{-\alpha (t-s)} U_j(t) U_j(s)^* \Delta_j F(s) ds.$$

We are now going to apply Proposition 2.8 to the operator $U_0(t)$ in order to control $\Delta_0 u(t)$.

Classical results, based on the stationary phase method, establish that

$$\|U_0(t)U_0(s)^*f\|_{L^{\infty}(\mathbb{R}^d)} \lesssim \frac{1}{|t-s|^{(d-1)/2}} \|f\|_{L^1(\mathbb{R}^d)}$$

for all $t \neq s$ and $f \in L^1(\mathbb{R}^d)$. (Proposition 8.15 from [Bahouri et al. 2011] contains a precise justification of the preceding dispersive estimate, and we further refer to Section 8.1.3 from the same work for more details on the stationary phase method.) It therefore follows that $U_0(t)$ satisfies all hypotheses of Proposition 2.8 with $H = L^2(\mathbb{R}^d)$, $X = \mathbb{R}^d$ and $\sigma = \frac{1}{2}(d-1)$. Hence, we conclude that

$$\|\Delta_{0}u\|_{L^{q}_{t}L^{r}_{x}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|\Delta_{0}f\|_{L^{2}_{x}} + \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{\tilde{q}}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{\tilde{r}}-1\right)} \|\Delta_{0}F\|_{L^{\tilde{q}'}_{t}L^{\tilde{r}'}_{x}}$$
(2-17)

for all admissible exponent pairs.

In order to recover an estimate for all components $\Delta_j u$, where $j \in \mathbb{Z}$, we conduct a simple scaling argument by introducing

$$u_j(t,x) := u\left(\frac{t}{2^j}, \frac{x}{2^j}\right), \quad F_j(t,x) := \frac{1}{2^j} F\left(\frac{t}{2^j}, \frac{x}{2^j}\right), \quad f_j(x) := f\left(\frac{x}{2^j}\right).$$

Noticing that

$$\Delta_0 u_j(t,x) = (\Delta_j u) \left(\frac{t}{2^j}, \frac{x}{2^j}\right), \quad \Delta_0 F_j(t,x) = \frac{1}{2^j} (\Delta_j F) \left(\frac{t}{2^j}, \frac{x}{2^j}\right), \quad \Delta_0 f_j(x) = (\Delta_j f) \left(\frac{x}{2^j}\right)$$

and that $u_i(t, x)$ solves

$$\begin{cases} (\partial_t + 2^{-j}\alpha \mp i |D|)u_j(t, x) = F_j(t, x), \\ u_j(0, x) = f_j(x) \end{cases}$$

on $[0, 2^{j}T)$, we obtain, applying (2-17) to u_{j} ,

$$2^{j\left(\frac{1}{q}+\frac{d}{r}\right)} \|\Delta_{j}u\|_{L_{t}^{q}L_{x}^{r}}$$

$$= \|\Delta_{0}u_{j}\|_{L_{t}^{q}L_{x}^{r}}$$

$$\lesssim \left(\frac{2^{j}T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|\Delta_{0}f_{j}\|_{L_{x}^{2}} + \left(\frac{2^{j}T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{r}-1\right)} \|\Delta_{0}F_{j}\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}$$

$$= \left(\frac{2^{j}T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} 2^{j\frac{d}{2}} \|\Delta_{j}f\|_{L_{x}^{2}} + \left(\frac{2^{j}T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{r}-1\right)} 2^{j\left(-\frac{1}{q}+d\left(1-\frac{1}{r}\right)\right)} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}$$

Finally, reorganizing the terms above, we deduce that

$$\begin{split} \|\Delta_{j}u\|_{L_{t}^{q}L_{x}^{r}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} 2^{j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}f\|_{L_{x}^{2}} \\ &+ \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{\tilde{q}}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{\tilde{r}}-1\right)} 2^{j\frac{d+1}{2}\left(1-\frac{1}{r}-\frac{1}{\tilde{r}}\right)} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}, \end{split}$$

which concludes the proof.

Corollary 2.11 (damped wave equation). Let $d \ge 2$, and consider a solution u(t, x) of the damped wave equation

$$\begin{cases} (\partial_t^2 + \alpha \partial_t - \Delta)u(t, x) = F(t, x), \\ u(0, x) = f(x), \\ \partial_t u(0, x) = g(x), \end{cases}$$

with $\alpha \ge 0$, $t \in [0, T)$ and $x \in \mathbb{R}^d$.

For any exponent pairs $(q, r), (\tilde{q}, \tilde{r}) \in [1, \infty] \times [2, \infty]$ which are admissible in the sense that

$$\frac{2}{q} + \frac{d-1}{r} \ge \frac{d-1}{2}, \quad 1 + \frac{d-1}{r} \ge \frac{d-1}{2} \quad and \quad (r, d) \neq (\infty, 3),$$

and similarly for (\tilde{q}, \tilde{r}) , and such that

$$\frac{1}{q} + \frac{1}{\tilde{q}} \le 1,$$

one has the high-frequency estimate

$$2^{-j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}(\partial_{t}u,\nabla u)\|_{L_{t}^{q}L_{x}^{r}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|\Delta_{j}(g,\nabla f)\|_{L_{x}^{2}} + \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{r}-1\right)} 2^{j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}$$

for all $j \in \mathbb{Z}$ with $2^j \ge \alpha$, and the low-frequency estimates

$$2^{-jd\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}\partial_{t}u\|_{L_{t}^{q}L_{x}^{r}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}} \|\Delta_{j}g\|_{L_{x}^{2}} + \frac{1}{\alpha} \left(\frac{\alpha 2^{2j}T}{\alpha+2^{2j}T}\right)^{\frac{1}{q}} 2^{j\left(1-\frac{2}{q}\right)} \|\Delta_{j}\nabla f\|_{L_{x}^{2}} + \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{q}} 2^{jd\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}$$

and

$$2^{-j(d(\frac{1}{2}-\frac{1}{r})-\frac{2}{q})} \|\Delta_{j}\nabla u\|_{L_{t}^{q}L_{x}^{r}} \lesssim \frac{1}{\alpha} \left(\frac{\alpha 2^{2j}T}{\alpha+2^{2j}T}\right)^{\frac{1}{q}} 2^{j} \|\Delta_{j}g\|_{L_{x}^{2}} + \left(\frac{\alpha 2^{2j}T}{\alpha+2^{2j}T}\right)^{\frac{1}{q}} \|\Delta_{j}\nabla f\|_{L_{x}^{2}} + \frac{1}{\alpha} \left(\frac{\alpha 2^{2j}T}{\alpha+2^{2j}T}\right)^{\frac{1}{q}+\frac{1}{q}} 2^{j(1+d(\frac{1}{2}-\frac{1}{r})-\frac{2}{q})} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}$$

for all $j \in \mathbb{Z}$ with $2^j \leq \alpha$.

Remark. Summing the preceding inequalities in *j* easily leads to damped Strichartz estimates in Besov spaces with summability in ℓ^p for any $\tilde{q}' \leq p \leq q$. However, due to the dichotomy of the statement of Corollary 2.11 into high and low frequencies, the resulting estimates cannot be stated with homogeneous Besov spaces in a unified format, which is natural because the damped wave equation does not enjoy any

scaling invariance. Observe, though, that the high- and low-frequency estimates match in the borderline case $\alpha = 2^j$, with $r = \tilde{r} = 2$.

Remark. Corollary 2.11 is optimal in the loose sense that it provides a similar result as Corollary 2.10 for high frequencies. Moreover, it recovers the optimal Strichartz estimates for the classical wave equation in the limit $\alpha \rightarrow 0$. Corollary 2.11 also displays optimality in its control of low frequencies. Indeed, let us consider a solution $u_c(t, x)$ for each c > 0 of the damped wave equation

$$\begin{cases} (c^{-2}\partial_t^2 + \alpha \partial_t - \Delta)u_c(t, x) = F(t, x), \\ u_c(0, x) = f(x), \\ \partial_t u_c(0, x) = g(x) \end{cases}$$

on $t \in [0, T)$. In particular, it follows that $\tilde{u}_c(t, x) := u_c(c^{-1}t, x)$ solves

$$\begin{cases} (\partial_t^2 + c\alpha \partial_t - \Delta) \tilde{u}_c(t, x) = F(c^{-1}t, x), \\ \tilde{u}_c(0, x) = f(x), \\ \partial_t \tilde{u}_c(0, x) = c^{-1}g(x) \end{cases}$$

on $t \in [0, cT)$. Therefore, applying Corollary 2.11 to \tilde{u}_c , with $r = \tilde{r} = 2$, yields the low-frequency estimates

$$\begin{split} \|\Delta_{j}\partial_{t}u_{c}\|_{L_{t}^{q}L_{x}^{2}} &\lesssim \frac{1}{c^{2}} \Big(\frac{T}{1+c^{2}\alpha T}\Big)^{\frac{1}{q}} \|\Delta_{j}g\|_{L_{x}^{2}} + \frac{1}{\alpha} \Big(\frac{\alpha 2^{2j}T}{\alpha+2^{2j}T}\Big)^{\frac{1}{q}} 2^{j\left(1-\frac{2}{q}\right)} \|\Delta_{j}\nabla f\|_{L_{x}^{2}} \\ &+ c^{2} \Big(\frac{T}{1+c^{2}\alpha T}\Big)^{\frac{1}{q}+\frac{1}{\tilde{q}}} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{2}} \end{split}$$

and

$$2^{j\frac{2}{q}} \|\Delta_{j} \nabla u_{c}\|_{L_{t}^{q}L_{x}^{2}} \lesssim \frac{1}{c^{2}\alpha} \left(\frac{\alpha 2^{2j}T}{\alpha + 2^{2j}T}\right)^{\frac{1}{q}} 2^{j} \|\Delta_{j}g\|_{L_{x}^{2}} + \left(\frac{\alpha 2^{2j}T}{\alpha + 2^{2j}T}\right)^{\frac{1}{q}} \|\Delta_{j}\nabla f\|_{L_{x}^{2}} \\ + \frac{1}{\alpha} \left(\frac{\alpha 2^{2j}T}{\alpha + 2^{2j}T}\right)^{\frac{1}{q} + \frac{1}{q}} 2^{j\left(1 - \frac{2}{q}\right)} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{2}}$$

for all $j \in \mathbb{Z}$ with $2^j \le c\alpha$. Finally, letting *c* tend to infinity and denoting the limit of u_c (in the sense of distributions) by *u*, we obtain the estimates

$$\|\Delta_{j}\partial_{t}u\|_{L_{t}^{q}L_{x}^{2}} \lesssim \frac{1}{\alpha} \left(\frac{\alpha 2^{2j}T}{\alpha + 2^{2j}T}\right)^{\frac{1}{q}} 2^{j2(1-\frac{1}{q})} \|\Delta_{j}f\|_{L_{x}^{2}} + \frac{1}{\alpha} \|\Delta_{j}F\|_{L_{t}^{q}L_{x}^{2}}$$

and

$$2^{j\frac{2}{q}} \|\Delta_{j}u\|_{L_{t}^{q}L_{x}^{2}} \lesssim \left(\frac{\alpha 2^{2j}T}{\alpha + 2^{2j}T}\right)^{\frac{1}{q}} \|\Delta_{j}f\|_{L_{x}^{2}} + \frac{1}{\alpha} \left(\frac{\alpha 2^{2j}T}{\alpha + 2^{2j}T}\right)^{\frac{1}{q} + \frac{1}{\bar{q}}} 2^{-j\frac{2}{\bar{q}}} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{2}}$$

for all $j \in \mathbb{Z}$ and every $q, \tilde{q} \in [1, \infty]$ such that $1/q + 1/\tilde{q} \leq 1$, which are optimal parabolic estimates for the heat equation

$$(\alpha \partial_t - \Delta)u = F$$

with initial data u(0, x) = f(x).

Remark. Other attempts at establishing Strichartz estimates for the damped wave equation can be found in [Inui 2019; Inui and Wakasugi 2021]. However, the results obtained therein are suboptimal. Indeed, Corollary 2.11 supersedes the results from those two works in both the breadth of the range of integrability parameters that it handles and the sharpness of regularity gains that it produces.

Proof. We begin by introducing

$$e(t,x) := \partial_t u(t,x)$$
 and $b(t,x) := i |D| u(t,x)$.

It then follows that (e, b) is a solution of the system

$$\partial_t e - i |D| b + \alpha e = F,$$

 $\partial_t b - i |D| e = 0,$

with initial data (e(0, x), b(0, x)) = (g(x), i | D | f(x)). This system is reminiscent of Maxwell's equations, which are studied in Section 3, and it can be recast as

$$\partial_t \begin{pmatrix} e \\ b \end{pmatrix} = \mathcal{L} \begin{pmatrix} e \\ b \end{pmatrix} + \begin{pmatrix} F \\ 0 \end{pmatrix}$$

where

$$\mathcal{L} = \mathcal{L}(D) := \begin{pmatrix} -\alpha & i |D| \\ i |D| & 0 \end{pmatrix}.$$

Now, a straightforward computation shows that $\mathcal{L}(\xi)$, where $\xi \in \mathbb{R}^d$, has the eigenvalues

$$\lambda_{\pm}(\xi) = -\frac{1}{2}\alpha \pm \sqrt{\frac{1}{4}\alpha^2 - |\xi|^2}$$
(2-18)

in the complex field \mathbb{C} . Moreover, exploiting the trivial identities $\lambda_+ + \lambda_- = -\alpha$ and $\lambda_+ \lambda_- = |\xi|^2$, one can readily verify that

$$\binom{e}{b} = P_+ \binom{e}{b} + P_- \binom{e}{b}$$

provides an eigenvector decomposition, where

$$P_{+}\begin{pmatrix}e\\b\end{pmatrix} = \frac{1}{\lambda_{+} - \lambda_{-}} \begin{pmatrix}\lambda_{+}e + i | D | b\\i | D | e - \lambda_{-}b\end{pmatrix} \quad \text{and} \quad P_{-}\begin{pmatrix}e\\b\end{pmatrix} = \frac{1}{\lambda_{-} - \lambda_{+}} \begin{pmatrix}\lambda_{-}e + i | D | b\\i | D | e - \lambda_{+}b\end{pmatrix}$$

are the projections onto the eigenspaces associated with $\lambda_+(D)$ and $\lambda_-(D)$ (when these eigenvalues are distinct, i.e., when $|\xi| \neq \frac{1}{2}\alpha$), respectively. In particular, this decomposition allows us to deduce the representation formula

$$\begin{pmatrix} e \\ b \end{pmatrix}(t) = P_{+} \begin{pmatrix} e \\ b \end{pmatrix}(t) + P_{-} \begin{pmatrix} e \\ b \end{pmatrix}(t)$$

$$= \begin{pmatrix} \frac{e^{t\lambda_{+}} + e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} g - \frac{e^{t\lambda_{+}} - e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} |D|^{2} f \\ \frac{e^{t\lambda_{+}} - e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} i |D|g - \frac{e^{t\lambda_{+}} + e^{t\lambda_{-}} + e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} i |D|f \end{pmatrix} + \int_{0}^{t} \begin{pmatrix} \frac{e^{(t-s)\lambda_{+}} + \lambda_{+} - e^{(t-s)\lambda_{-}} - \lambda_{-}}{\lambda_{+} - \lambda_{-}} F \\ \frac{e^{(t-s)\lambda_{+}} - e^{(t-s)\lambda_{-}}}{\lambda_{+} - \lambda_{-}} i |D|F \end{pmatrix} ds.$$
(2-19)

We want now to use (2-19) to deduce an estimate on $\Delta_0 e$ and $\Delta_0 b$, which, by a scaling argument in the spirit of the proof of Corollary 2.10, will then result in a control on each dyadic component $\Delta_j e$ and $\Delta_j b$, with $j \in \mathbb{Z}$. Note, however, that the eigenvalues $\lambda_{\pm}(\xi)$ are of a fundamentally different nature depending on the relative size of the frequencies $\{\frac{1}{2} \le |\xi| \le 2\}$ with respect to $\alpha \ge 0$, which leads us to consider several cases.

More specifically, on the one hand, when $\lambda_{\pm} \in \mathbb{R}$ (i.e., when $|\xi| \leq \frac{1}{2}\alpha$), we are going to employ the elementary controls

$$-\alpha \le \lambda_{-} \le -\frac{\alpha}{2} \le -\frac{2|\xi|^{2}}{\alpha} \le \lambda_{+} \le -\frac{|\xi|^{2}}{\alpha},$$

$$0 \le \frac{e^{t\lambda_{+}} - e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} = \frac{\int_{\lambda_{-}}^{\lambda_{+}} te^{ts} \, ds}{\lambda_{+} - \lambda_{-}} \le te^{t\lambda_{+}} \le te^{-t\frac{|\xi|^{2}}{\alpha}},$$
(2-20)

while, on the other hand, when $\lambda_{\pm} \in \mathbb{C} \setminus \mathbb{R}$ (i.e., when $|\xi| \ge \frac{1}{2}\alpha$), we are going to use the properties

$$\begin{aligned} |\lambda_{\pm}| &= |\xi|, \quad |e^{-t\lambda_{\pm}}| = e^{-\frac{\alpha}{2}t}, \\ \frac{e^{t\lambda_{\pm}} - e^{t\lambda_{\pm}}}{\lambda_{\pm} - \lambda_{\pm}} \bigg| &= e^{-\frac{\alpha}{2}t} \frac{|\sin(t\sqrt{|\xi|^2 - \frac{1}{4}\alpha^2})|}{\sqrt{|\xi|^2 - \frac{1}{4}\alpha^2}} \le t e^{-\frac{\alpha}{2}t}. \end{aligned}$$
(2-21)

Considering that the dyadic operator Δ_0 localizes frequencies to $\{\frac{1}{2} \le |\xi| \le 2\}$, we will then distinguish three cases:

- The complex case, where $0 \le \alpha \le \frac{1}{2}$, so that λ_{\pm} and $(\lambda_{+} \lambda_{-})^{-1}$ are complex and smooth on $\{\frac{1}{2} \le |\xi| \le 2\}$.
- The degenerate case, where $\frac{1}{2} < \alpha < 5$ and the eigenvalues may be equal.
- The real case, where α ≥ 5, which implies that the eigenvalues are real and the damping phenomenon dominates the behavior of solutions on {¹/₂ ≤ |ξ| ≤ 2}.

The complex case. We begin by considering the range $0 \le \alpha \le \frac{1}{2}$. In this setting, it is readily seen that the functions $\lambda_+(\xi)$, $\lambda_-(\xi)$ and $(\lambda_+(\xi) - \lambda_-(\xi))^{-1}$, as well as any number of their derivatives, are uniformly bounded on $\{\frac{1}{3} < |\xi| < 3\}$, uniformly in $\alpha \in [0, \frac{1}{2}]$. In particular, by virtue of the criterion (2-14) for the boundedness of multipliers, further introducing a smooth cutoff function $\psi(\xi)$ compactly supported inside $\{\frac{1}{3} < |\xi| < 3\}$ and such that $\psi \equiv 1$ on $\{\frac{1}{2} \le |\xi| \le 2\}$, it follows that $\lambda_+(\xi)\psi(\xi)$, $\lambda_-(\xi)\psi(\xi)$ and $(\lambda_+(\xi) - \lambda_-(\xi))^{-1}\psi(\xi)$ are the symbols of bounded Fourier multipliers over any homogeneous Besov space. Therefore, we deduce from (2-19) that

$$\begin{split} \|\Delta_{0}(e,b)\|_{L_{t}^{q}L_{x}^{r}} & \lesssim \sum_{\pm} \|e^{t\lambda_{\pm}}\Delta_{0}(f,g)\|_{L_{t}^{q}L_{x}^{r}} + \left\|\int_{0}^{t} e^{(t-s)\lambda_{\pm}}\Delta_{0}F\,ds\right\|_{L_{t}^{q}L_{x}^{r}} \\ & = \sum_{\pm} \|e^{-\frac{\alpha}{2}t}e^{\pm it\delta(D)}\Delta_{0}(f,g)\|_{L_{t}^{q}L_{x}^{r}} + \left\|\int_{0}^{t} e^{-\frac{\alpha}{2}(t-s)}e^{\pm i(t-s)\delta(D)}\Delta_{0}F\,ds\right\|_{L_{t}^{q}L_{x}^{r}}, \quad (2-22) \end{split}$$

where we introduced the notation

$$\delta(\xi) := \sqrt{|\xi|^2 - \frac{1}{4}\alpha^2}$$

for convenience.

Now, classically, the stationary phase method can be used (see [Bahouri et al. 2011, Proposition 8.15], for instance) to show that

$$\left| \int_{\mathbb{R}^d} e^{ix \cdot \xi} e^{\pm it |\xi|} \psi(\xi) \, d\xi \right| \le \frac{C_{\psi}}{t^{(d-1)/2}}$$

for all t > 0, where the constant $C_{\psi} > 0$ is independent of t and x, and ψ is any smooth compactly supported function whose support does not contain the origin. A similar estimate holds, uniformly in $\alpha \in [0, \frac{1}{2}]$, if one replaces $|\xi|$ by $\delta(\xi)$ and if the support of ψ is disjoint from the closed ball $\{|\xi| \le \frac{1}{2}\alpha\}$. More precisely, we claim that

$$\left| \int_{\mathbb{R}^d} e^{ix \cdot \xi} e^{\pm it\delta(\xi)} \psi(\xi) \, d\xi \right| \le \frac{C_{\psi}}{t^{(d-1)/2}} \tag{2-23}$$

whenever supp $\psi \subset \{|\xi| > \frac{1}{4} \ge \frac{1}{2}\alpha\}$, where $C_{\psi} > 0$ is independent of t > 0, $x \in \mathbb{R}^d$ and $\alpha \in [0, \frac{1}{2}]$. For the sake of completeness, we provide a justification of (2-23) in Appendix B.

Therefore, introducing the flow

$$U(t) f(x) := e^{\pm it\delta(D)} \psi(D) f(x)$$

for some fixed compactly supported cutoff $\psi(\xi)$ such that $\sup \psi \subset \{|\xi| > \frac{1}{4} \geq \frac{\alpha}{2}\}$ and $\psi \equiv 1$ on $\{\frac{1}{2} \leq \xi \leq 2\}$, we see, in view of (2-23), that U(t) satisfies all hypotheses of Proposition 2.8 with

$$H = L^2(\mathbb{R}^d), \quad X = \mathbb{R}^d \text{ and } \sigma = \frac{1}{2}(d-1).$$

Hence, we conclude from (2-22) that

$$\|\Delta_{0}(e,b)\|_{L_{t}^{q}L_{x}^{r}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|\Delta_{0}(f,g)\|_{L_{x}^{2}} + \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{\bar{r}}-1\right)} \|\Delta_{0}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}$$
(2-24)

for all admissible exponent pairs, when $0 \le \alpha \le \frac{1}{2}$.

The degenerate case. We are now looking at the range $\frac{1}{2} < \alpha < 5$. This case is easily settled by the use of (2-20) and (2-21), which allows us to deduce, whenever $\frac{1}{2} \le |\xi| \le 2$, that

$$\begin{split} \left\| \frac{e^{t\lambda_{+}} - e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} \right\|_{L_{t}^{c}} &\leq \|te^{-\frac{t}{20}}\|_{L_{t}^{c}} \lesssim \left(\frac{T^{2}}{1+T^{2}}\right)^{\frac{1}{c}} \leq 2^{\frac{1}{c}} \left(\frac{T}{1+T}\right)^{\frac{2}{c}} \lesssim \left(\frac{T}{1+T}\right)^{\frac{1}{c}} \\ \left\| \frac{e^{t\lambda_{+}} \lambda_{+} - e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} \right\|_{L_{t}^{c}} &\leq \|e^{t\lambda_{+}}\|_{L_{t}^{c}} + \left\|\lambda_{-}\frac{e^{t\lambda_{+}} - e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} \right\|_{L_{t}^{c}} \lesssim \left(\frac{T}{1+T}\right)^{\frac{1}{c}}, \\ \left\| \frac{e^{t\lambda_{+}} \lambda_{-}e^{t\lambda_{-}} \lambda_{+}}{\lambda_{+} - \lambda_{-}} \right\|_{L_{t}^{c}} &\leq \|e^{t\lambda_{+}}\|_{L_{t}^{c}} + \left\|\lambda_{+}\frac{e^{t\lambda_{+}} - e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} \right\|_{L_{t}^{c}} \lesssim \left(\frac{T}{1+T}\right)^{\frac{1}{c}}. \end{split}$$

for any $1 \le c \le \infty$. Indeed, incorporating these controls into (2-19) and recalling that the space of Fourier multipliers on $L^2(\mathbb{R}^d)$ is isomorphic to $L^{\infty}(\mathbb{R}^d)$ leads to

$$\begin{split} \|\Delta_{0}(e,b)\|_{L_{t}^{q}L_{x}^{r}} \\ \lesssim \|\Delta_{0}(e,b)\|_{L_{t}^{q}L_{x}^{2}} \\ \lesssim \left(\frac{T}{1+T}\right)^{\frac{1}{q}} \|\Delta_{0}(f,g)\|_{L_{x}^{2}} + \left(\frac{T}{1+T}\right)^{\frac{1}{q}+\frac{1}{q}} \|\Delta_{0}F\|_{L_{t}^{\tilde{q}'}L_{x}^{2}} \\ \lesssim \left(\frac{T}{1+T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|\Delta_{0}(f,g)\|_{L_{x}^{2}} + \left(\frac{T}{1+T}\right)^{\frac{1}{q}+\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{r}-1\right)} \|\Delta_{0}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}} \quad (2-25) \end{split}$$

for all admissible exponent pairs, when $\frac{1}{2} < \alpha < 5$.

The real case. In the remaining case, we assume that $\alpha \ge 5$. In particular, when $\frac{1}{2} \le |\xi| \le 2$,

$$\lambda_+ - \lambda_- = \sqrt{\alpha^2 - 4|\xi|^2} \ge \sqrt{\alpha^2 - 16} \ge \frac{3}{5}\alpha.$$

Furthermore, employing (2-20), one finds that

$$\|e^{t\lambda_+}\|_{L^c_t} \le \|e^{-\frac{t}{4\alpha}}\|_{L^c_t} \lesssim \left(\frac{\alpha T}{\alpha+T}\right)^{\frac{1}{c}} \quad \text{and} \quad \|e^{t\lambda_-}\|_{L^c_t} \le \|e^{-\frac{\alpha t}{2}}\|_{L^c_t} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{c}}$$

for any $1 \le c \le \infty$. Therefore, we deduce from (2-19) and (2-20) that

$$\begin{split} \|\Delta_{0}e\|_{L_{t}^{q}L_{x}^{r}} &\lesssim \|\Delta_{0}e\|_{L_{t}^{q}L_{x}^{2}} \\ &\lesssim \left(\frac{\alpha^{1-2q}T}{\alpha+T} + \frac{T}{1+\alpha T}\right)^{\frac{1}{q}} \|\Delta_{0}g\|_{L_{x}^{2}} + \left(\frac{\alpha^{1-q}T}{\alpha+T} + \frac{\alpha^{-q}T}{1+\alpha T}\right)^{\frac{1}{q}} \|\Delta_{0}f\|_{L_{x}^{2}} \\ &+ \left(\frac{\alpha^{1-2(1/q+1/\tilde{q})^{-1}}T}{\alpha+T} + \frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{\tilde{q}}} \|\Delta_{0}F\|_{L_{t}^{\tilde{q}'}L_{x}^{2}} \\ &\lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}} \|\Delta_{0}g\|_{L_{x}^{2}} + \left(\frac{\alpha^{1-q}T}{\alpha+T}\right)^{\frac{1}{q}} \|\Delta_{0}f\|_{L_{x}^{2}} + \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{\tilde{q}}} \|\Delta_{0}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}} \quad (2-26) \end{split}$$

and

$$\begin{split} \|\Delta_{0}b\|_{L_{t}^{q}L_{x}^{r}} &\lesssim \|\Delta_{0}b\|_{L_{t}^{q}L_{x}^{2}} \\ &\lesssim \left(\frac{\alpha^{1-q}T}{\alpha+T} + \frac{\alpha^{-q}T}{1+\alpha T}\right)^{\frac{1}{q}} \|\Delta_{0}g\|_{L_{x}^{2}} + \left(\frac{\alpha T}{\alpha+T} + \frac{\alpha^{-2q}T}{1+\alpha T}\right)^{\frac{1}{q}} \|\Delta_{0}f\|_{L_{x}^{2}} \\ &+ \left(\frac{\alpha^{1-(1/q+1/\tilde{q})^{-1}}T}{\alpha+T} + \frac{\alpha^{-(1/q+1/\tilde{q})^{-1}}T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{\tilde{q}}} \|\Delta_{0}F\|_{L_{t}^{\tilde{q}'}L_{x}^{2}} \\ &\lesssim \left(\frac{\alpha^{1-q}T}{\alpha+T}\right)^{\frac{1}{q}} \|\Delta_{0}g\|_{L_{x}^{2}} + \left(\frac{\alpha T}{\alpha+T}\right)^{\frac{1}{q}} \|\Delta_{0}f\|_{L_{x}^{2}} + \left(\frac{\alpha^{1-(1/q+1/\tilde{q})^{-1}}T}{\alpha+T}\right)^{\frac{1}{q}+\frac{1}{\tilde{q}}} \|\Delta_{0}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}} \tag{2-27}$$

for all admissible exponent pairs, whenever $\alpha \ge 5$.

1341

Scaling argument and conclusion of proof. We are now in a position to conclude the justification of the corollary. In order to deduce an estimate on $\Delta_i(e, b)$ for all $j \in \mathbb{Z}$ from (2-24)–(2-27), we conduct now a scaling analysis in the spirit of the proof of Corollary 2.10. To that end, we introduce

$$(e_{j}, b_{j})(t, x) := (e, b) \left(\frac{t}{2^{j}}, \frac{x}{2^{j}}\right), \quad F_{j}(t, x) := \frac{1}{2^{j}} F\left(\frac{t}{2^{j}}, \frac{x}{2^{j}}\right),$$
$$f_{j}(x) := 2^{j} f\left(\frac{x}{2^{j}}\right), \qquad g_{j}(x) := g\left(\frac{x}{2^{j}}\right)$$

and observe that (e_i, b_i) solves

$$\partial_t e_j - i |D| b_j + 2^{-j} \alpha e_j = F_j,$$

 $\partial_t b_j - i |D| e_j = 0$

on $[0, 2^{j}T)$, with initial data $(e_{i}(0, x), b_{i}(0, x)) = (g_{i}(x), i|D|f_{i}(x)).$

Then, noticing that

$$\begin{split} \|\Delta_{j}(e,b)\|_{L_{t}^{q}L_{x}^{r}} &= 2^{-j\left(\frac{1}{q}+\frac{d}{r}\right)} \|\Delta_{0}(e_{j},b_{j})\|_{L_{t}^{q}L_{x}^{r}},\\ \|\Delta_{j}g\|_{L_{x}^{2}} &= 2^{-j\frac{d}{2}} \|\Delta_{0}g_{j}\|_{L_{x}^{2}},\\ \|\Delta_{j}f\|_{L_{x}^{2}} &= 2^{-j\left(1+\frac{d}{2}\right)} \|\Delta_{0}f_{j}\|_{L_{x}^{2}},\\ \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}} &= 2^{j\left(\frac{1}{q}-d\left(1-\frac{1}{\tilde{r}}\right)\right)} \|\Delta_{0}F_{j}\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}} \end{split}$$

and applying (2-24) and (2-25) to (e_i, b_i) yields the estimate

$$\begin{split} \|\Delta_{j}(e,b)\|_{L_{t}^{q}L_{x}^{r}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} 2^{j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}(g,\nabla f)\|_{L_{x}^{2}} \\ &+ \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{\bar{r}}-1\right)} 2^{j\frac{d+1}{2}\left(1-\frac{1}{r}-\frac{1}{\bar{r}}\right)} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}} \end{split}$$

whenever $2^j \ge \alpha$.

Similarly, if $2^j \leq \alpha$, then, applying (2-25), (2-26) and (2-27) to (e_j, b_j) leads to the controls

$$\begin{split} \|\Delta_{j}e\|_{L_{t}^{q}L_{x}^{r}} \lesssim \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}} 2^{jd\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}g\|_{L_{x}^{2}} + \frac{1}{\alpha} \left(\frac{\alpha T}{\alpha+2^{2j}T}\right)^{\frac{1}{q}} 2^{j\left(1+d\left(\frac{1}{2}-\frac{1}{r}\right)\right)} \|\Delta_{j}\nabla f\|_{L_{x}^{2}} \\ + \left(\frac{T}{1+\alpha T}\right)^{\frac{1}{q}+\frac{1}{q}} 2^{jd\left(1-\frac{1}{r}-\frac{1}{r}\right)} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}} \end{split}$$

and

$$\begin{split} \|\Delta_{j}b\|_{L_{t}^{q}L_{x}^{r}} &\lesssim \frac{1}{\alpha} \Big(\frac{\alpha T}{\alpha + 2^{2j}T}\Big)^{\frac{1}{q}} 2^{j\left(1 + d\left(\frac{1}{2} - \frac{1}{r}\right)\right)} \|\Delta_{j}g\|_{L_{x}^{2}} + \Big(\frac{\alpha T}{\alpha + 2^{2j}T}\Big)^{\frac{1}{q}} 2^{jd\left(\frac{1}{2} - \frac{1}{r}\right)} \|\Delta_{j}\nabla f\|_{L_{x}^{2}} \\ &+ \frac{1}{\alpha} \Big(\frac{\alpha T}{\alpha + 2^{2j}T}\Big)^{\frac{1}{q} + \frac{1}{q}} 2^{j\left(1 + d\left(1 - \frac{1}{r} - \frac{1}{r}\right)\right)} \|\Delta_{j}F\|_{L_{t}^{\tilde{q}'}L_{x}^{\tilde{r}'}}. \end{split}$$
which concludes the proof of the corollary.

which concludes the proof of the corollary.

Corollary 2.12 (damped Maxwell equations). Let d = 2 or d = 3 and consider a solution (E, B)(t, x): $[0, T) \times \mathbb{R}^d \to \mathbb{R}^6$ of the damped Maxwell system

$$\begin{cases} \frac{1}{c}\partial_t E - \nabla \times B + \sigma c E = G, \\ \frac{1}{c}\partial_t B + \nabla \times E = 0, \\ \operatorname{div} B = 0 \end{cases}$$

for some initial data $(E, B)(0, x) = (E_0, B_0)(x)$, where $\sigma \ge 0$ and c > 0.

For any exponent pairs $(q, r), (\tilde{q}, \tilde{r}) \in [1, \infty] \times [2, \infty]$ which are admissible in the sense that

$$\frac{2}{q} + \frac{d-1}{r} \ge \frac{d-1}{2}, \quad 1 + \frac{d-1}{r} \ge \frac{d-1}{2} \quad and \quad (r,d) \neq (\infty,3),$$

and similarly for (\tilde{q}, \tilde{r}) , and such that

$$\frac{1}{q} + \frac{1}{\tilde{q}} \le 1,$$

one has the high-frequency estimate

 $d \pm 1 (1 - 1)$

$$2^{-j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}(PE,B)\|_{L_{t}^{q}([0,T);L_{x}^{r})} \\ \lesssim c^{\frac{d-1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)-\frac{2}{q}} \left(\frac{c^{2}T}{1+\sigma c^{2}T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|\Delta_{j}(PE_{0},B_{0})\|_{L_{x}^{2}} \\ + c^{1+\frac{d-1}{2}\left(1-\frac{1}{r}-\frac{1}{\bar{r}}\right)-\frac{2}{q}-\frac{2}{\bar{q}}} \left(\frac{c^{2}T}{1+\sigma c^{2}T}\right)^{\frac{1}{q}+\frac{1}{\bar{q}}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{\bar{r}}-1\right)} 2^{j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{\bar{r}}\right)} \|\Delta_{j}PG\|_{L_{t}^{\tilde{q}'}([0,T);L_{x}^{\tilde{r}'})}$$

for all $j \in \mathbb{Z}$ with $2^j \ge \sigma c$, and the low-frequency estimates

$$2^{-jd\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j} PE\|_{L_{t}^{q}\left([0,T\right);L_{x}^{r}\right)} \lesssim \left(\frac{T}{1+\sigma c^{2}T}\right)^{\frac{1}{q}} \|\Delta_{j} PE_{0}\|_{L_{x}^{2}} + \frac{1}{\sigma c} \left(\frac{\sigma 2^{2j}T}{\sigma+2^{2j}T}\right)^{\frac{1}{q}} 2^{j\left(1-\frac{2}{q}\right)} \|\Delta_{j} B_{0}\|_{L_{x}^{2}} + c \left(\frac{T}{1+\sigma c^{2}T}\right)^{\frac{1}{q}+\frac{1}{q}} 2^{jd\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j} PG\|_{L_{t}^{\tilde{q}'}\left([0,T\right);L_{x}^{\tilde{r}'}\right)}$$

and

$$2^{-j\left(d\left(\frac{1}{2}-\frac{1}{r}\right)-\frac{2}{q}\right)} \|\Delta_{j}B\|_{L_{t}^{q}\left([0,T\right);L_{x}^{r}\right)} \lesssim \frac{1}{\sigma c} \left(\frac{\sigma 2^{2j}T}{\sigma + 2^{2j}T}\right)^{\frac{1}{q}} 2^{j} \|\Delta_{j}PE_{0}\|_{L_{x}^{2}} + \left(\frac{\sigma 2^{2j}T}{\sigma + 2^{2j}T}\right)^{\frac{1}{q}} \|\Delta_{j}B_{0}\|_{L_{x}^{2}} + \frac{1}{\sigma} \left(\frac{\sigma 2^{2j}T}{\sigma + 2^{2j}T}\right)^{\frac{1}{q} + \frac{1}{\tilde{q}}} 2^{j\left(1 + d\left(\frac{1}{2} - \frac{1}{\tilde{r}}\right) - \frac{2}{\tilde{q}}\right)} \|\Delta_{j}PG\|_{L_{t}^{\tilde{q}'}\left([0,T\right);L_{x}^{\tilde{r}'}\right)}$$

for all $j \in \mathbb{Z}$ with $2^j \leq \sigma c$.

Remark. Corollary 2.12 only provides estimates of the magnetic field *B* and the divergence-free part of the electric field *PE*. Notice, though, that the divergent component $P^{\perp}E$ can also be estimated directly from Maxwell's system. Indeed, applying the projector P^{\perp} to Ampère's equation yields

$$\partial_t P^\perp E + \sigma c^2 P^\perp E = c P^\perp G,$$

which leads to the representation formula

$$P^{\perp}E(t) = e^{-\sigma c^2 t} P^{\perp}E_0 + c \int_0^t e^{-\sigma c^2(t-s)} P^{\perp}G(s) \, ds.$$
A direct estimate then easily gives

$$2^{-jd\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j} P^{\perp} E\|_{L_{t}^{q} L_{x}^{r}} \lesssim \|\Delta_{j} P^{\perp} E\|_{L_{t}^{q} L_{x}^{2}} \lesssim \left(\frac{T}{1+\sigma c^{2}T}\right)^{\frac{1}{q}} \|\Delta_{j} P^{\perp} E_{0}\|_{L_{x}^{2}} + c\left(\frac{T}{1+\sigma c^{2}T}\right)^{\frac{1}{q}+\frac{1}{q}} \|\Delta_{j} P^{\perp} G\|_{L_{t}^{\tilde{q}'} L_{x}^{2}} \lesssim \left(\frac{T}{1+\sigma c^{2}T}\right)^{\frac{1}{q}} \|\Delta_{j} P^{\perp} E_{0}\|_{L_{x}^{2}} + c\left(\frac{T}{1+\sigma c^{2}T}\right)^{\frac{1}{q}+\frac{1}{q}} 2^{jd\left(\frac{1}{2}-\frac{1}{\tilde{r}}\right)} \|\Delta_{j} P^{\perp} G\|_{L_{t}^{\tilde{q}'} L_{x}^{\tilde{r}'}}$$

for any $q, \tilde{q} \in [1, \infty]$ and $r, \tilde{r} \in [2, \infty]$, with $1/q + 1/\tilde{q} \le 1$.

Proof. Since B(t, x) is a solenoidal field, we begin by introducing a vector potential A(t, x), with $t \in [0, cT)$ and $x \in \mathbb{R}^d$, such that

$$B(t,x) = \nabla \times A(ct,x). \tag{2-28}$$

Faraday's equation $c^{-1}\partial_t B + \nabla \times E = 0$ then implies that $(\partial_t A)(ct, x) + E(t, x)$ must be curl-free, whereby there exists a scalar potential $\varphi(t, x)$, with $t \in [0, cT)$ and $x \in \mathbb{R}^d$, such that

$$E(t, x) = \nabla \varphi(ct, x) - (\partial_t A)(ct, x).$$
(2-29)

Observe that A and φ are not uniquely determined. Indeed, for any scalar-valued potential $\psi(t, x)$, it is possible to apply the transformations

$$A(t, x) \mapsto A(t, x) + \nabla \psi(t, x),$$

$$\varphi(t, x) \mapsto \varphi(t, x) + \partial_t \psi(t, x)$$
(2-30)

to produce new potentials representing the same electromagnetic field (E, B). Any particular choice of A and φ is called a gauge.

Different choices of gauge lead to different insights into Maxwell's equations. It is therefore important to carefully select the properties fixing the gauge. A standard example of gauge fixing is the Coulomb gauge, which merely requires that A be solenoidal, i.e., div A = 0. The Lorenz gauge, which imposes the condition

div
$$A(t, x) = \partial_t \varphi(t, x)$$

is another classical example with the property that it produces decoupled wave equations on A and φ when there is no damping, i.e., $\sigma = 0$.

Here, we introduce a damped Lorenz gauge by selecting potentials A and φ solving

$$\operatorname{div} A(t, x) = \partial_t \varphi(t, x) + \sigma c \varphi(t, x).$$
(2-31)

Observe that it is always possible to find a damped Lorenz gauge. Indeed, starting from any other gauge (A, φ) , one can apply the transformations (2-30) with any solution $\psi(t, x)$ of the damped wave equation

$$\partial_t^2 \psi + \sigma c \partial_t \psi - \Delta \psi = \operatorname{div} A - \partial_t \varphi - \sigma c \varphi,$$

thereby producing new potentials satisfying (2-31).

Now, by inserting (2-28) and (2-29) into Ampère's equation and then employing (2-31), a straightforward calculation shows that the damped Lorenz gauge is a solution of the damped wave system

$$(\partial_t^2 + \sigma c \partial_t - \Delta) A(t, x) = -G(c^{-1}t, x)$$
(2-32)

on $t \in [0, cT)$.

Therefore, applying the Strichartz estimates for damped wave equations from Corollary 2.11 to this system, we find, concerning high frequencies, that

$$\begin{split} c^{\frac{1}{q}} 2^{-j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}(PE,B)\|_{L_{t}^{q}([0,T);L_{x}^{r})} \\ &\lesssim 2^{-j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}(\partial_{t}PA,\nabla PA)\|_{L_{t}^{q}([0,cT);L_{x}^{r})} \\ &\lesssim \left(\frac{cT}{1+\sigma c^{2}T}\right)^{\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}-\frac{1}{2}\right)} \|\Delta_{j}(PE_{0},B_{0})\|_{L_{x}^{2}} \\ &+ c^{1-\frac{1}{q}} \left(\frac{cT}{1+\sigma c^{2}T}\right)^{\frac{1}{q}+\frac{1}{q}+\frac{d-1}{2}\left(\frac{1}{r}+\frac{1}{r}-1\right)} 2^{j\frac{d+1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}PG\|_{L_{t}^{\tilde{q}'}([0,T);L_{x}^{\tilde{r}'})} \end{split}$$

for all $j \in \mathbb{Z}$ with $2^j \ge \sigma c$.

As for low frequencies, i.e., when $j \in \mathbb{Z}$ with $2^j \leq \sigma c$, we obtain similarly from Corollary 2.11 that

$$\begin{aligned} c^{\frac{1}{q}} 2^{-jd\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j} PE\|_{L_{t}^{q}([0,T);L_{x}^{r})} \\ &= 2^{-jd\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j} \partial_{t} PA\|_{L_{t}^{q}([0,cT);L_{x}^{r})} \\ &\lesssim \left(\frac{cT}{1+\sigma c^{2}T}\right)^{\frac{1}{q}} \|\Delta_{j} PE_{0}\|_{L_{x}^{2}} + \frac{1}{\sigma c} \left(\frac{\sigma c 2^{2j}T}{\sigma + 2^{2j}T}\right)^{\frac{1}{q}} 2^{j\left(1-\frac{2}{q}\right)} \|\Delta_{j} B_{0}\|_{L_{x}^{2}} \\ &+ c^{1-\frac{1}{q}} \left(\frac{cT}{1+\sigma c^{2}T}\right)^{\frac{1}{q}+\frac{1}{q}} 2^{jd\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j} PG\|_{L_{t}^{\tilde{q}'}([0,T);L_{x}^{\tilde{r}'})} \end{aligned}$$

and

$$\begin{split} c^{\frac{1}{q}} 2^{-j(d(\frac{1}{2}-\frac{1}{r})-\frac{2}{q})} \|\Delta_{j}B\|_{L_{t}^{q}([0,T);L_{x}^{r})} \\ &\lesssim 2^{-j(d(\frac{1}{2}-\frac{1}{r})-\frac{2}{q})} \|\Delta_{j}\nabla PA\|_{L_{t}^{q}([0,cT);L_{x}^{r})} \\ &\lesssim \frac{1}{\sigma c} \Big(\frac{\sigma c 2^{2j}T}{\sigma + 2^{2j}T}\Big)^{\frac{1}{q}} 2^{j} \|\Delta_{j}PE_{0}\|_{L_{x}^{2}} + \Big(\frac{\sigma c 2^{2j}T}{\sigma + 2^{2j}T}\Big)^{\frac{1}{q}} \|\Delta_{j}B_{0}\|_{L_{x}^{2}} \\ &+ c^{1-\frac{1}{q}} \frac{1}{\sigma c} \Big(\frac{\sigma c 2^{2j}T}{\sigma + 2^{2j}T}\Big)^{\frac{1}{q}} 2^{j(1+d(\frac{1}{2}-\frac{1}{r})-\frac{2}{q})} \|\Delta_{j}PG\|_{L_{t}^{\tilde{q}'}([0,T);L_{x}^{\tilde{r}'})} \end{split}$$

which concludes the proof of the corollary.

The global low-frequency estimates from Corollaries 2.11 and 2.12 can be refined by considering the maximal regularity of the heat equation (without damping) discussed in Section 2.2. The next two results provide such low-frequency parabolic estimates for the wave equation and Maxwell's system, respectively.

 \square

Proposition 2.13. Let $d \ge 2$, and consider a solution u(t, x) of the damped wave equation

$$\begin{aligned} (\partial_t^2 + \alpha \partial_t - \Delta) u(t, x) &= F(t, x), \\ u(0, x) &= f(x), \\ \partial_t u(0, x) &= g(x), \end{aligned}$$

with $\alpha > 0, t \in [0, T)$ and $x \in \mathbb{R}^d$.

For any $\chi \in C_c^{\infty}(\mathbb{R}^d)$ and $\sigma \in \mathbb{R}$, one has the low-frequency estimates

$$\|\chi(\alpha^{-1}D)\partial_t u\|_{L^m_t([0,T);\dot{B}^{\sigma+2/m}_{2,q})}$$

$$\lesssim \alpha^{-\frac{1}{m}} \|g\|_{\dot{B}^{\sigma+2/m}_{2,q}} + \alpha^{\frac{1}{m}-1} \|f\|_{\dot{B}^{\sigma+2}_{2,m}} + \alpha^{-(1+\frac{1}{m}-\frac{1}{r})} \|F\|_{L^{r}_{t}([0,T);\dot{B}^{\sigma+2/m}_{2,q})}$$

for any $1 < r \le m < \infty$ and $1 \le q \le \infty$, as well as

$$\|\chi(\alpha^{-1}D)\nabla u\|_{L_{t}^{m}([0,T);\dot{B}_{2,1}^{\sigma+2/m})} \lesssim \alpha^{\frac{1}{m}-1} \|g\|_{\dot{B}_{2,m}^{\sigma+1}} + \alpha^{\frac{1}{m}} \|f\|_{\dot{B}_{2,m}^{\sigma+1}} + \alpha^{\frac{1}{m}-\frac{1}{r}} \|F\|_{L_{t}^{r}([0,T);\dot{B}_{2,\infty}^{\sigma-1+2/r})}$$

for any $1 < r < m < \infty$, and

$$\|\chi(\alpha^{-1}D)\nabla u\|_{L_{t}^{m}([0,T);\dot{B}_{2,q}^{\sigma+2/m})} \lesssim \alpha^{\frac{1}{m}-1} \|g\|_{\dot{B}_{2,m}^{\sigma+1}} + \alpha^{\frac{1}{m}} \|f\|_{\dot{B}_{2,m}^{\sigma+1}} + \|F\|_{L_{t}^{m}([0,T);\dot{B}_{2,q}^{\sigma-1+2/m})}$$

for any $1 < m < \infty$ and $1 \le q \le \infty$.

Proof. Following the proof of Corollary 2.11, we consider

$$e(t, x) := \partial_t u(t, x)$$
 and $b(t, x) := i |D| u(t, x)$

In particular, one has the representation formula (2-19), which, for any given choice of $0 < A < \frac{1}{2}$, can be recast as

$$e(t) = e^{\frac{A}{\alpha}t\Delta}m_{2}(t,D)g - e^{\frac{A}{\alpha}t\Delta}m_{1}(t,D)|D|^{2}f + \int_{0}^{t} e^{\frac{A}{\alpha}(t-s)\Delta}m_{2}(t-s,D)F \, ds$$

$$= (e^{\frac{A}{\alpha}t\Delta}|D|^{2}m_{2}^{+}(t,D) - e^{-At\alpha}m_{2}^{-}(t,D))g - e^{\frac{A}{\alpha}t\Delta}m_{1}(t,D)|D|^{2}f$$

$$+ \int_{0}^{t} (e^{\frac{A}{\alpha}(t-s)\Delta}|D|^{2}m_{2}^{+}(t-s,D) - e^{-A(t-s)\alpha}m_{2}^{-}(t-s,D))F \, ds,$$

$$(2-33)$$

$$h(t) = e^{\frac{A}{\alpha}t\Delta}m_{1}(t,D)|D|g - e^{\frac{A}{\alpha}t\Delta}m_{2}(t,D)|D|f + \int_{0}^{t} e^{\frac{A}{\alpha}(t-s)\Delta}m_{2}(t-s,D)|D|F \, ds,$$

$$b(t) = e^{\frac{A}{\alpha}t\Delta}m_1(t,D)i|D|g - e^{\frac{A}{\alpha}t\Delta}m_3(t,D)i|D|f + \int_0^t e^{\frac{A}{\alpha}(t-s)\Delta}m_1(t-s,D)i|D|F\,ds,$$

with the time-dependent Fourier multipliers

$$m_{1}(t,\xi) = \frac{e^{t\lambda_{+}} - e^{t\lambda_{-}}}{\lambda_{+} - \lambda_{-}} e^{At \frac{|\xi|^{2}}{\alpha}}, \qquad m_{2}(t,\xi) = \frac{e^{t\lambda_{+}} \lambda_{+} - e^{t\lambda_{-}} \lambda_{-}}{\lambda_{+} - \lambda_{-}} e^{At \frac{|\xi|^{2}}{\alpha}},$$

$$m_{2}^{+}(t,\xi) = \frac{e^{t\lambda_{+}} \lambda_{+}}{|\xi|^{2}(\lambda_{+} - \lambda_{-})} e^{At \frac{|\xi|^{2}}{\alpha}}, \qquad m_{2}^{-}(t,\xi) = \frac{e^{t\lambda_{-}} \lambda_{-}}{\lambda_{+} - \lambda_{-}} e^{At\alpha},$$

$$m_{3}(t,\xi) = \frac{e^{t\lambda_{+}} \lambda_{-} e^{t\lambda_{-}} \lambda_{+}}{\lambda_{+} - \lambda_{-}} e^{At \frac{|\xi|^{2}}{\alpha}}, \qquad (2-34)$$

where the eigenvalues $\lambda_+(\xi)$ and $\lambda_-(\xi)$ are defined in (2-18).

Then, making use of the elementary controls (2-20), one can show that

$$\begin{split} \|m_{1}\mathbb{1}_{\{|\xi| \leq \frac{1}{4}\alpha\}}\|_{L^{\infty}_{t,\xi}} &\lesssim \alpha^{-1}, \quad \|m_{2}^{+}\mathbb{1}_{\{|\xi| \leq \frac{1}{4}\alpha\}}\|_{L^{\infty}_{t,\xi}} \lesssim \alpha^{-2}, \\ \|m_{2}^{-}\mathbb{1}_{\{|\xi| \leq \frac{1}{4}\alpha\}}\|_{L^{\infty}_{t,\xi}} &\lesssim 1, \quad \|m_{3}\mathbb{1}_{\{|\xi| \leq \alpha R\}}\|_{L^{\infty}_{t,\xi}} \lesssim 1. \end{split}$$

In particular, since the space of Fourier multipliers over $L^2(\mathbb{R}^d)$ is isomorphic to $L^{\infty}(\mathbb{R}^d)$, we conclude that

$$\mathbb{1}_{\{|D|\leq\frac{1}{4}\alpha\}}m_1, \quad \mathbb{1}_{\{|D|\leq\frac{1}{4}\alpha\}}m_2^{\pm} \quad \text{and} \quad \mathbb{1}_{\{|D|\leq\frac{1}{4}\alpha\}}m_3$$

are bounded in the sense that they satisfy (2-12) for p = 2.

Therefore, applying Propositions 2.2 and 2.4 to the representation formulas (2-33) by suitably scaling time by A/α , we obtain the estimates

$$\begin{aligned} \|\mathbb{1}_{\{|D|\leq\frac{1}{4}\alpha\}}e\|_{L_{t}^{m}\dot{B}_{2,q}^{\sigma+2/m}} &\lesssim \alpha^{\frac{1}{m}-2}\|\mathbb{1}_{\{|D|\leq\frac{1}{4}\alpha\}}g\|_{\dot{B}_{2,m}^{\sigma+2}} + \alpha^{-\frac{1}{m}}\|g\|_{\dot{B}_{2,q}^{\sigma+2/m}} + \alpha^{\frac{1}{m}-1}\|f\|_{\dot{B}_{2,m}^{\sigma+2/m}} \\ &+ \alpha^{\frac{1}{m}-\frac{1}{r}-1}\|\mathbb{1}_{\{|D|\leq\frac{1}{4}\alpha\}}F\|_{L_{t}^{r}\dot{B}_{2,m}^{\sigma+2/r}} + \alpha^{-(1+\frac{1}{m}-\frac{1}{r})}\|F\|_{L_{t}^{r}\dot{B}_{2,q}^{\sigma+2/m}} \\ &\lesssim \alpha^{-\frac{1}{m}}\|g\|_{\dot{B}_{2,q}^{\sigma+2/m}} + \alpha^{\frac{1}{m}-1}\|f\|_{\dot{B}_{2,m}^{\sigma+2}} + \alpha^{-(1+\frac{1}{m}-\frac{1}{r})}\|F\|_{L_{t}^{r}\dot{B}_{2,q}^{\sigma+2/m}} \tag{2-35} \end{aligned}$$

and

$$\|\mathbb{1}_{\{|D| \le \frac{1}{4}\alpha\}} b\|_{L^{m}_{t} \dot{B}^{\sigma+2/m}_{2,1}} \lesssim \alpha^{\frac{1}{m}-1} \|g\|_{\dot{B}^{\sigma+1}_{2,m}} + \alpha^{\frac{1}{m}} \|f\|_{\dot{B}^{\sigma+1}_{2,m}} + \alpha^{\frac{1}{m}-\frac{1}{r}} \|F\|_{L^{r}_{t} \dot{B}^{\sigma-1+2/r}_{2,\infty}}$$
(2-36)

for any $1 < r < m < \infty$ and $1 \le q \le \infty$.

If, instead of Proposition 2.4, one uses Proposition 2.6, then one arrives at the estimates

$$\|\mathbb{1}_{\{|D| \le \frac{1}{4}\alpha\}} e\|_{L_{t}^{m} \dot{B}_{2,q}^{\sigma+2/m}} \lesssim \alpha^{-\frac{1}{m}} \|g\|_{\dot{B}_{2,q}^{\sigma+2/m}} + \alpha^{\frac{1}{m}-1} \|f\|_{\dot{B}_{2,m}^{\sigma+2}} + \alpha^{-1} \|F\|_{L_{t}^{m} \dot{B}_{2,q}^{\sigma+2/m}}$$
(2-37)

and

$$\|\mathbb{1}_{\{|D| \le \frac{1}{4}\alpha\}} b\|_{L_{t}^{m} \dot{B}_{2,q}^{\sigma+2/m}} \lesssim \alpha^{\frac{1}{m}-1} \|g\|_{\dot{B}_{2,m}^{\sigma+1}} + \alpha^{\frac{1}{m}} \|f\|_{\dot{B}_{2,m}^{\sigma+1}} + \|F\|_{L_{t}^{m} \dot{B}_{2,q}^{\sigma-1+2/m}}$$
(2-38)

for any $1 < m < \infty$ and $1 \le q \le \infty$.

In order to handle frequencies lying in the range $\{\frac{1}{4}\alpha < |\xi| \le \alpha R\}$ for any choice of parameter R > 1 with $2AR^2 < 1$, we employ (2-20) and (2-21) to deduce that the multipliers in (2-33) satisfy

$$\begin{split} \|m_{1}\mathbb{1}_{\left\{\frac{1}{4}\alpha<|\xi|\leq\alpha R\right\}}\|L_{t,\xi}^{\infty} \lesssim \alpha^{-1}, \\ \|m_{2}\mathbb{1}_{\left\{\frac{1}{4}\alpha<|\xi|\leq\alpha R\right\}}\|L_{t,\xi}^{\infty} \leq \|\lambda_{-}m_{1}\mathbb{1}_{\left\{\frac{1}{4}\alpha<|\xi|\leq\alpha R\right\}}\|L_{t,\xi}^{\infty} + \|e^{t\left(\lambda_{+}+A\frac{|\xi|^{2}}{\alpha}\right)}\mathbb{1}_{\left\{\frac{1}{4}\alpha<|\xi|\leq\alpha R\right\}}\|L_{t,\xi}^{\infty} \lesssim 1, \\ \|m_{3}\mathbb{1}_{\left\{\frac{1}{4}\alpha<|\xi|\leq\alpha R\right\}}\|L_{t,\xi}^{\infty} \leq \|\lambda_{+}m_{1}\mathbb{1}_{\left\{\frac{1}{4}\alpha<|\xi|\leq\alpha R\right\}}\|L_{t,\xi}^{\infty} + \|e^{t\left(\lambda_{+}+A\frac{|\xi|^{2}}{\alpha}\right)}\mathbb{1}_{\left\{\frac{1}{4}\alpha<|\xi|\leq\alpha R\right\}}\|L_{t,\xi}^{\infty} \lesssim 1. \end{split}$$

Therefore, as previously, by the boundedness of multipliers and the fact that $||e^{-at}||_{L_t^p([0,\infty))} = a^{-1/p}$ for any a > 0 and $1 \le p \le \infty$ (no need to use Propositions 2.2, 2.4 or 2.6, here), we conclude from (2-33) that

$$\begin{aligned} \|\mathbb{1}_{\{\frac{1}{4}\alpha < |D| \le \alpha R\}} e^{\|L_{t}^{m} \dot{B}_{2,q}^{\sigma+2/m}} \\ &\lesssim \alpha^{-\frac{1}{m}} \|g\|_{\dot{B}_{2,q}^{\sigma+2/m}} + \alpha^{-\frac{1}{m}-1} \|\mathbb{1}_{\{|D| \le \alpha R\}} f\|_{\dot{B}_{2,q}^{\sigma+2+2/m}} + \alpha^{-\left(1+\frac{1}{m}-\frac{1}{r}\right)} \|F\|_{L_{t}^{r} \dot{B}_{2,q}^{\sigma+2/m}} \\ &\lesssim \alpha^{-\frac{1}{m}} \|g\|_{\dot{B}_{2,q}^{\sigma+2/m}} + \alpha^{\frac{1}{m}-1} \|f\|_{\dot{B}_{2,m}^{\sigma+2}} + \alpha^{-\left(1+\frac{1}{m}-\frac{1}{r}\right)} \|F\|_{L_{t}^{r} \dot{B}_{2,q}^{\sigma+2/m}} \tag{2-39}$$

and

for any $1 \le r \le m < \infty$ and $1 \le q \le \infty$.

All in all, combining (2-35), (2-37) and (2-39), we obtain

$$\|\mathbb{1}_{\{|D| \le \alpha R\}} e\|_{L_{t}^{m} \dot{B}_{2,q}^{\sigma+2/m}} \lesssim \alpha^{-\frac{1}{m}} \|g\|_{\dot{B}_{2,q}^{\sigma+2/m}} + \alpha^{\frac{1}{m}-1} \|f\|_{\dot{B}_{2,m}^{\sigma+2}} + \alpha^{-(1+\frac{1}{m}-\frac{1}{r})} \|F\|_{L_{t}^{r} \dot{B}_{2,q}^{\sigma+2/m}}$$

for any $1 < r \le m < \infty$ and $1 \le q \le \infty$. Similarly, combining (2-36), (2-38) and (2-40), we deduce that

$$\|\mathbb{1}_{\{|D| \le \alpha R\}} b\|_{L_t^m \dot{B}_{2,1}^{\sigma+2/m}} \lesssim \alpha^{\frac{1}{m}-1} \|g\|_{\dot{B}_{2,m}^{\sigma+1}} + \alpha^{\frac{1}{m}} \|f\|_{\dot{B}_{2,m}^{\sigma+1}} + \alpha^{\frac{1}{m}-\frac{1}{r}} \|F\|_{L_t^r \dot{B}_{2,\infty}^{\sigma-1+2/r}}$$

for any $1 < r < m < \infty$ and

$$\|\mathbb{1}_{\{|D| \le \alpha R\}} b\|_{L_t^m \dot{B}_{2,q}^{\sigma+2/m}} \lesssim \alpha^{\frac{1}{m}-1} \|g\|_{\dot{B}_{2,m}^{\sigma+1}} + \alpha^{\frac{1}{m}} \|f\|_{\dot{B}_{2,m}^{\sigma+1}} + \|F\|_{L_t^m \dot{B}_{2,q}^{\sigma-1+2/m}}$$

for any $1 < m < \infty$ and $1 \le q \le \infty$.

Finally, selecting *R* large enough that supp $\chi \subset \{|\xi| \le R\}$ yields the desired estimates on $\chi(\alpha^{-1}D)(e, b)$, thereby concluding the proof.

Remark. The preceding proof raises a question — is it possible to extend the statement of Proposition 2.13 from the L^2 -setting (in space integrability) to a general L^p -setting, with $p \neq 2$? Such an extension would require dealing with the boundedness of the multipliers defined in (2-34) over L^p . This is related to the boundedness of the Bochner–Riesz multiplier $(1 - |\xi|^2)_+^{1/2}$, which is notoriously challenging and remains unsettled in general dimensions. We will therefore not be going into further detail on this subject.

Corollary 2.14. Let d = 2 or d = 3, and consider a solution $(E, B)(t, x) : [0, T) \times \mathbb{R}^d \to \mathbb{R}^6$ of the damped Maxwell system

$$\begin{cases} \frac{1}{c}\partial_t E - \nabla \times B + \sigma c E = G, \\ \frac{1}{c}\partial_t B + \nabla \times E = 0, \\ \operatorname{div} B = 0 \end{cases}$$

for some initial data $(E, B)(0, x) = (E_0, B_0)(x)$, where $\sigma > 0$ and c > 0.

For any $\chi \in C_c^{\infty}(\mathbb{R}^d)$ and $s \in \mathbb{R}$, one has the low-frequency estimates

$$\begin{aligned} \|\chi(c^{-1}D)PE\|_{L^m_t([0,T);\dot{B}^{s+2/m}_{2,q})} \\ \lesssim c^{-\frac{2}{m}} \|PE_0\|_{\dot{B}^{s+2/m}_{2,q}} + c^{-1} \|B_0\|_{\dot{B}^{s+1}_{2,m}} + c^{-1+\frac{2}{r}-\frac{2}{m}} \|PG\|_{L^r_t([0,T);\dot{B}^{s+2/m}_{2,q})} \end{aligned}$$

for any $1 < r \le m < \infty$ and $1 \le q \le \infty$, as well as

$$\|\chi(c^{-1}D)B\|_{L^m_t([0,T);\dot{B}^{s+2/m}_{2,1})} \lesssim c^{-1} \|PE_0\|_{\dot{B}^{s+1}_{2,m}} + \|B_0\|_{\dot{B}^s_{2,m}} + \|PG\|_{L^r_t([0,T);\dot{B}^{s-1+2/r}_{2,\infty})}$$

for any $1 < r < m < \infty$, and

$$\|\chi(c^{-1}D)B\|_{L^{m}_{t}([0,T);\dot{B}^{s+2/m}_{2,q})} \lesssim c^{-1} \|PE_{0}\|_{\dot{B}^{s+1}_{2,m}} + \|B_{0}\|_{\dot{B}^{s}_{2,m}} + \|PG\|_{L^{m}_{t}([0,T);\dot{B}^{s-1+2/m}_{2,q})}$$

ny 1 < m < \propto and 1 < a < \propto.

for any $1 < m < \infty$ and $1 \le q \le \infty$.

Proof. We follow the steps of the proof of Corollary 2.12, which involves first fixing a damped Lorenz gauge satisfying (2-28), (2-29) and (2-31). Then, applying Proposition 2.13 instead of Corollary 2.11 to the damped wave system (2-32), we find, for any $\chi \in C_c^{\infty}(\mathbb{R}^d)$ and $s \in \mathbb{R}$, that

$$\begin{aligned} \|\chi(c^{-1}D)PE\|_{L_{t}^{m}([0,T);\dot{B}_{2,q}^{s+2/m})} &= c^{-\frac{1}{m}} \|\chi(c^{-1}D)\partial_{t}PA\|_{L_{t}^{m}([0,cT);\dot{B}_{2,q}^{s+2/m})} \\ &\lesssim c^{-\frac{2}{m}} \|PE_{0}\|_{\dot{B}_{2,q}^{s+2/m}} + c^{-1} \|B_{0}\|_{\dot{B}_{2,m}^{s+1}} + c^{-1+\frac{2}{r}-\frac{2}{m}} \|PG\|_{L_{t}^{r}([0,T);\dot{B}_{2,q}^{s+2/m})}, \end{aligned}$$

for any $1 < r \le m < \infty$ and $1 \le q \le \infty$. We also obtain

$$\begin{aligned} \|\chi(c^{-1}D)B\|_{L_{t}^{m}([0,T);\dot{B}_{2,1}^{s+2/m})} &\lesssim c^{-\frac{1}{m}} \|\chi(c^{-1}D)\nabla PA\|_{L_{t}^{m}([0,cT);\dot{B}_{2,1}^{s+2/m})} \\ &\lesssim c^{-1} \|PE_{0}\|_{\dot{B}_{2,m}^{s+1}} + \|B_{0}\|_{\dot{B}_{2,m}^{s}} + \|PG\|_{L_{t}^{r}([0,T);\dot{B}_{2,\infty}^{s-1+2/r})}, \end{aligned}$$

for any $1 < r < m < \infty$, whereas the limiting case $1 < r = m < \infty$ yields

$$\begin{aligned} \|\chi(c^{-1}D)B\|_{L^m_t([0,T);\dot{B}^{s+2/m}_{2,q})} &\lesssim c^{-\frac{1}{m}} \|\chi(c^{-1}D)\nabla PA\|_{L^m_t([0,cT);\dot{B}^{s+2/m}_{2,q})} \\ &\lesssim c^{-1} \|PE_0\|_{\dot{B}^{s+1}_{2,m}} + \|B_0\|_{\dot{B}^s_{2,m}} + \|PG\|_{L^m_t([0,T);\dot{B}^{s-1+2/m}_{2,q})} \end{aligned}$$

for any $1 \le q \le \infty$, which concludes the proof.

3. Perfect incompressible two-dimensional plasmas

We are now going to apply the damped Strichartz estimates for Maxwell's system, established in the preceding section, to the analysis of the two-dimensional incompressible Euler–Maxwell system (1-1). The main goal of this section is to establish Theorems 3.1, 3.2 and 3.3 below.

In order to conveniently state the results, recall first that we denote the initial energy by

$$\mathcal{E}_0 := \|(u_0, E_0, B_0)\|_{L^2}.$$

For ease of notation, we also introduce a natural frequency decomposition of Besov and Chemin–Lerner spaces with respect to the speed of light c > 0. More precisely, we define the Besov seminorms

$$\|f\|_{\dot{B}^{s}_{p,q,<}} := \left(\sum_{\substack{k \in \mathbb{Z} \\ 2^{k} < \sigma c}} 2^{ksq} \|\Delta_{k}f\|_{L^{p}}^{q}\right)^{\frac{1}{q}} \quad \text{and} \quad \|f\|_{\dot{B}^{s}_{p,q,>}} := \left(\sum_{\substack{k \in \mathbb{Z} \\ 2^{k} \ge \sigma c}} 2^{ksq} \|\Delta_{k}f\|_{L^{p}}^{q}\right)^{\frac{1}{q}},$$

as well as the Chemin-Lerner seminorms

$$\|f\|_{\tilde{L}^{r}_{t}\dot{B}^{s}_{p,q,<}} := \left(\sum_{\substack{k \in \mathbb{Z} \\ 2^{k} < \sigma c}} 2^{ksq} \|\Delta_{k}f\|_{L^{r}_{t}L^{p}_{x}}^{q}\right)^{\frac{1}{q}} \quad \text{and} \quad \|f\|_{\tilde{L}^{r}_{t}\dot{B}^{s}_{p,q,>}} := \left(\sum_{\substack{k \in \mathbb{Z} \\ 2^{k} \ge \sigma c}} 2^{ksq} \|\Delta_{k}f\|_{L^{r}_{t}L^{p}_{x}}^{q}\right)^{\frac{1}{q}}$$

for any $s \in \mathbb{R}$ and $0 < p, q, r \le \infty$ (with obvious modifications if q is infinite), where the constant $\sigma > 0$ is the electrical conductivity used in the original Euler–Maxwell system (1-1).

Theorem 3.1. Let p and ε be any real numbers in $(2, \infty)$ and (0, 1), respectively. There is a constant $C_* > 0$ such that, if the initial data (u_0, E_0, B_0) , with div $u_0 = \text{div } E_0 = \text{div } B_0$, has the two-dimensional normal structure (1-2) and belongs to $((H^1 \cap \dot{W}^{1,p}) \times (H^1 \cap \dot{B}_{2,1}^{7/4})^2)(\mathbb{R}^2)$ with

$$(\mathcal{E}_{0} + \|u_{0}\|_{\dot{H}^{1} \cap \dot{W}^{1,p}} + \|(E_{0}, B_{0})\|_{\dot{H}^{1}} + c^{-\frac{3}{4}} \|(E_{0}, B_{0})\|_{\dot{B}^{7/4}_{2,1}} C_{*} e^{C_{*} \mathcal{E}_{0}^{4+\varepsilon}} < c,$$
(3-1)

where c > 0 is the speed of light, then there is a global weak solution $(u, E, B) \in L^{\infty}(\mathbb{R}^+; L^2)$ to the two-dimensional Euler–Maxwell system (1-1), with the normal structure (1-2), satisfying the energy inequality (1-3) and enjoying the additional regularity

$$u \in L^{\infty}(\mathbb{R}^{+}; \dot{H}^{1} \cap \dot{W}^{1,p}), \quad (E,B) \in L^{\infty}(\mathbb{R}^{+}; \dot{H}^{1}), \quad c^{-\frac{3}{4}}(E,B) \in \tilde{L}^{\infty}(\mathbb{R}^{+}; \dot{B}^{7/4}_{2,1}),$$

$$(cE,B) \in L^{2}(\mathbb{R}^{+}; \dot{H}^{1}), \quad B \in L^{2}(\mathbb{R}^{+}; \dot{B}^{2}_{2,1,<}), \quad (3-2)$$

$$(E,B) \in \tilde{L}^{2}(\mathbb{R}^{+}; \dot{B}^{1}_{\infty,1,>}), \quad c^{\frac{1}{4}}E \in \tilde{L}^{2}(\mathbb{R}^{+}; \dot{B}^{7/4}_{2,1}), \quad c^{\frac{1}{4}}B \in \tilde{L}^{2}(\mathbb{R}^{+}; \dot{B}^{7/4}_{2,1,>}).$$

It is to be emphasized that the bounds in (3-2) *are uniform in any set of initial data such that the left-hand side of* (3-1) *remains bounded.*

Remark. For any fixed initial data (u_0, E_0, B_0) satisfying the requirements of Theorem 3.1, it is possible to improve the uniform controls (3-2) by showing that the bound

$$(E,B) \in \widetilde{L}^{\infty}(\mathbb{R}^+; \dot{B}_{2,1}^{7/4})$$

holds uniformly as $c \to \infty$. This is clarified in Section 3.11 below.

Remark. Note that we do not make any claim concerning the uniqueness of solutions produced by Theorem 3.1. However, Theorem 3.2 below strengthens the statement of Theorem 3.1 by achieving such uniqueness, provided the initial vorticity is bounded pointwise.

Remark. Employing the bounds (3-2), one can easily show that $(\nabla E, \nabla B) \in L^2(\mathbb{R}^+; L^\infty)$. Indeed, making use of straightforward embeddings in Besov and Chemin–Lerner spaces, we deduce that

$$\begin{aligned} \|\nabla E\|_{L^{2}(\mathbb{R}^{+};L^{\infty})} &\leq \|\nabla E\|_{\tilde{L}^{2}(\mathbb{R}^{+};\dot{B}_{\infty,1}^{0})} \lesssim \|E\|_{\tilde{L}^{2}(\mathbb{R}^{+};\dot{B}_{\infty,1,<}^{1})} + \|E\|_{\tilde{L}^{2}(\mathbb{R}^{+};\dot{B}_{\infty,1,>}^{1})} \\ &\lesssim \|cE\|_{\tilde{L}^{2}(\mathbb{R}^{+};\dot{B}_{\infty,\infty,<}^{0})} + \|E\|_{\tilde{L}^{2}(\mathbb{R}^{+};\dot{B}_{\infty,1,>}^{1})} \\ &\lesssim \|cE\|_{\tilde{L}^{2}(\mathbb{R}^{+};\dot{B}_{2,\infty,<}^{1})} + \|E\|_{\tilde{L}^{2}(\mathbb{R}^{+};\dot{B}_{\infty,1,>}^{1})} \end{aligned}$$

and

$$\begin{aligned} \|\nabla B\|_{L^{2}(\mathbb{R}^{+};L^{\infty})} &\leq \|\nabla B\|_{L^{2}(\mathbb{R}^{+};\dot{B}^{0}_{\infty,1})} \lesssim \|B\|_{L^{2}(\mathbb{R}^{+};\dot{B}^{1}_{\infty,1,<})} + \|B\|_{\tilde{L}^{2}(\mathbb{R}^{+};\dot{B}^{1}_{\infty,1,>})} \\ &\lesssim \|B\|_{L^{2}(\mathbb{R}^{+};\dot{B}^{2}_{2,1,<})} + \|B\|_{\tilde{L}^{2}(\mathbb{R}^{+};\dot{B}^{1}_{\infty,1,>})}. \end{aligned}$$
(3-3)

Remark. The initial condition (3-1) can be interpreted as a mere strengthening of the property that the velocity of the fluid cannot exceed the speed of light, i.e.,

$$\|u\|_{L^{\infty}_{t,x}} \leq c.$$

On purely physical grounds, this condition seems therefore quite reasonable and is not very restrictive.

The previous result only covers the case $\omega_0 \in L^2 \cap L^p$ in the range of parameters $p \in (2, \infty)$. The next result strengthens Theorem 3.1 by assuming that $\omega_0 \in L^2 \cap L^\infty$.

Theorem 3.2. If, in addition to all hypotheses of Theorem 3.1, for some given $p \in (2, \infty)$, one also assumes that $\omega_0 \in L^{\infty}$ (but not necessarily $\nabla u_0 \in L^{\infty}$), then the solution produced by Theorem 3.1 satisfies the additional bound $\omega \in L^{\infty}(\mathbb{R}^+; L^{\infty})$ and is unique in the space of all solutions $(\bar{u}, \bar{E}, \bar{B})$ to the Euler–Maxwell system (1-1) satisfying the bounds, locally in time,

$$(\bar{u}, \bar{E}, \bar{B}) \in L^{\infty}_t L^2_x, \quad \bar{u} \in L^2_t L^{\infty}_x, \quad \bar{j} \in L^2_{t,x},$$

and having the same initial data.

We address the propagation of regularity in the Euler–Maxwell system (1-1) with the following result. **Theorem 3.3.** Consider parameters $p \in (2, \infty)$, $\varepsilon \in (0, 1)$, $s \in (\frac{7}{4}, 2)$ and $n \in [1, \infty]$. There is a constant $C_{**} > 0$ such that, if the initial data (u_0, E_0, B_0) , with div $u_0 = \text{div } E_0 = \text{div } B_0$, has the two-dimensional normal structure (1-2) and belongs to $((H^1 \cap \dot{W}^{1,p}) \times (H^1 \cap \dot{B}_{2,n}^s)^2)(\mathbb{R}^2)$ with

$$(\mathcal{E}_{0} + \|u_{0}\|_{\dot{H}^{1} \cap \dot{W}^{1,p}} + \|(E_{0}, B_{0})\|_{\dot{H}^{1}} + c^{1-s}\|(E_{0}, B_{0})\|_{\dot{B}^{s}_{2,n}})C_{**}e^{C_{**}\mathcal{E}_{0}^{4+\varepsilon}} < c, \qquad (3-4)$$

where c > 0 is the speed of light, then there is a global weak solution $(u, E, B) \in L^{\infty}(\mathbb{R}^+; L^2)$ to the two-dimensional Euler–Maxwell system (1-1), with the normal structure (1-2), satisfying the energy inequality (1-3) and enjoying the additional regularity

$$u \in L^{\infty}(\mathbb{R}^{+}; \dot{H}^{1} \cap \dot{W}^{1,p}), \quad (E, B) \in L^{\infty}(\mathbb{R}^{+}; \dot{H}^{1}), \quad c^{1-s}(E, B) \in \tilde{L}^{\infty}(\mathbb{R}^{+}; \dot{B}^{s}_{2,n}), \\ (cE, B) \in L^{2}(\mathbb{R}^{+}; \dot{H}^{1}), \quad B \in L^{2}(\mathbb{R}^{+}; \dot{B}^{2}_{2,1,<}), \qquad (3-5)$$

$$c^{\frac{7}{4}-s}(E, B) \in \tilde{L}^{2}(\mathbb{R}^{+}; \dot{B}^{s-3/4}_{\infty,n,>}), \quad c^{2-s}E \in \tilde{L}^{2}(\mathbb{R}^{+}; \dot{B}^{s}_{2,n}), \quad c^{2-s}B \in \tilde{L}^{2}(\mathbb{R}^{+}; \dot{B}^{s}_{2,n,>}).$$

It is to be emphasized that the bounds in (3-5) are uniform in any set of initial data such that the left-hand side of (3-4) remains bounded.

Remark. As in the case of Theorem 3.1, for any fixed initial data (u_0, E_0, B_0) satisfying the requirements of Theorem 3.3, it is possible to improve the uniform controls (3-5) by showing that the bound

$$(E, B) \in \widetilde{L}^{\infty}(\mathbb{R}^+; \dot{B}^s_{2,n})$$

holds uniformly as $c \to \infty$. This is clarified in Section 3.11 below.

The remainder of this section builds up to the proofs of Theorems 3.1, 3.2 and 3.3 by implementing the strategy discussed in Section 1.3. The proofs of the theorems per se are given in Sections 3.8, 3.9 and 3.10, respectively.

3.1. *Dimensional analysis.* Prior to discussing specific elements of the proofs of the above theorems, we provide here a dimensional analysis of the Euler–Maxwell system (1-1), which, we hope, will shed light on the initial conditions (3-1) and (3-4).

Specifically, assuming that (u, E, B) is a solution of (1-1) for some fixed light velocity c > 0 and initial data (u_0, E_0, B_0) , we observe, defining

$$u^{\lambda}(t,x) = \lambda u(\lambda^2 t, \lambda x), \quad E^{\lambda}(t,x) = \lambda E(\lambda^2 t, \lambda x), \quad B^{\lambda}(t,x) = \lambda B(\lambda^2 t, \lambda x), \quad (3-6)$$

for any $\lambda > 0$, that $(u^{\lambda}, E^{\lambda}, B^{\lambda})$ also solves (1-1) with a rescaled speed of light $c_{\lambda} = \lambda c$ (the electrical conductivity σ remains unchanged) and for the initial data

$$u_0^{\lambda}(x) = \lambda u(\lambda x), \quad E_0^{\lambda}(x) = \lambda E(\lambda x), \quad B_0^{\lambda}(x) = \lambda B(\lambda x).$$

In particular, we readily compute

$$\begin{aligned} \|(u_0^{\lambda}, E_0^{\lambda}, B_0^{\lambda})\|_{L^2} + \|u_0^{\lambda}\|_{\dot{W}^{1,p}} + \|(u_0^{\lambda}, E_0^{\lambda}, B_0^{\lambda})\|_{\dot{H}^1} + c_{\lambda}^{-3/4} \|(E_0^{\lambda}, B_0^{\lambda})\|_{\dot{B}^{7/4}_{2,1}} \\ &= \lambda(\lambda^{-1}\|(u_0, E_0, B_0)\|_{L^2} + \lambda^{1-\frac{2}{p}} \|u_0\|_{\dot{W}^{1,p}} + \|(u_0, E_0, B_0)\|_{\dot{H}^1} + c^{-\frac{3}{4}} \|(E_0, B_0)\|_{\dot{B}^{7/4}_{2,1}}, \end{aligned}$$

which, by an optimization procedure in λ , implies that Theorem 3.1 still holds if one replaces assumption (3-1) with the weaker inequality

$$(\mathcal{E}_{0}^{(p-2)/(2p-2)} \| u_{0} \|_{\dot{W}^{1,p}}^{p/(2p-2)} + \| (u_{0}, E_{0}, B_{0}) \|_{\dot{H}^{1}} + c^{-\frac{3}{4}} \| (E_{0}, B_{0}) \|_{\dot{B}^{7/4}_{2,1}} C_{*} e^{C_{*} \mathcal{E}_{0}^{4+\varepsilon}} < c, \qquad (3-7)$$

where the independent constant $C_* > 0$ may take a different value. Note that this inequality is now invariant with respect to the parabolic scaling (3-6). Similarly, the same procedure can be used to optimize (3-4) and replace it with a scaling invariant assumption.

It turns out that the parabolic scaling (3-6) is the only available invariant dilation which leaves the electrical conductivity σ unchanged. However, if one allows σ to be redefined according to the dilation, then other scalings become available. For example, introducing the hyperbolic scaling

$$u^{\lambda}(t,x) = u(\lambda t,\lambda x), \quad E^{\lambda}(t,x) = E(\lambda t,\lambda x), \quad B^{\lambda}(t,x) = B(\lambda t,\lambda x)$$
(3-8)

for any $\lambda > 0$, we see that $(u^{\lambda}, E^{\lambda}, B^{\lambda})$ now solves (1-1) with a rescaled electrical conductivity $\sigma_{\lambda} = \lambda \sigma$ (the speed of light *c* remains unchanged) and for the initial data

$$u_0^{\lambda}(x) = u(\lambda x), \quad E_0^{\lambda}(x) = E(\lambda x), \quad B_0^{\lambda}(x) = B(\lambda x).$$

In particular, by setting $\lambda = \sigma^{-1}$, it is now possible to deduce how the constants C_* and C_{**} in (3-1) and (3-4), respectively, depend on σ . More precisely, this process allows us to show that Theorem 3.1 holds if one further replaces (3-7) by the assumption

$$(\mathcal{E}_{0}^{(p-2)/(2p-2)} \|u_{0}\|_{\dot{W}^{1,p}}^{p/(2p-2)} + \|(u_{0}, E_{0}, B_{0})\|_{\dot{H}^{1}} + (\sigma c)^{-\frac{3}{4}} \|(E_{0}, B_{0})\|_{\dot{B}^{7/4}_{2,1}} C_{*} e^{C_{*} \sigma^{4+\varepsilon} \mathcal{E}_{0}^{4+\varepsilon}} < c \quad (3-9)$$

for some constant $C_* > 0$ which is now independent of the electrical conductivity σ . Observe that (3-9) is now invariant with respect to both the parabolic scaling (3-6) and the hyperbolic scaling (3-8). As previously noted, the same process can be applied to improve (3-4).

3.2. *Approximation procedure and stability.* The proofs of Theorems 3.1, 3.2 and 3.3 proceed by compactness arguments. More specifically, they follow the standard procedure of first considering smooth solutions to a regularized approximation of the original system (1-1), where all formal estimates can be conducted with full rigor, and then showing the stability of the approximation as it converges towards the original system.

Such approximation procedures are absolutely classical in the field of fluid dynamics. We are therefore only going to outline an example of approximation which can be used here to conveniently establish our results. Specifically, for any integer $n \ge 1$, we consider the unique solution (u_n, E_n, B_n) to the approximate Navier–Stokes–Maxwell system

$$\begin{cases} \partial_t u_n + (S_n u_n) \cdot \nabla u_n - \frac{1}{n} \Delta u_n = -\nabla p_n + (S_n j_n) \times B_n, & \text{div } u_n = 0, \\ \frac{1}{c} \partial_t E_n - \nabla \times B_n = -j_n, & \text{div } E_n = 0, \\ \frac{1}{c} \partial_t B_n + \nabla \times E_n = 0, & \text{div } B_n = 0, \\ j_n = \sigma (cE_n + S_n P(u_n \times B_n)), & \text{div } j_n = 0, \end{cases}$$
(3-10)

for the initial data $(u_n, E_n, B_n)|_{t=0} = S_n(u_0, E_0, B_0)$, with the two-dimensional normal structure (1-2), where S_n denotes the Fourier multiplier operator defined in Appendix A which restricts frequencies to the domain $\{|\xi| \le 2^n\}$. The construction of the solution (u_n, E_n, B_n) is a standard procedure. One can, for instance, follow and adapt the steps detailed in [Lemarié-Rieusset 2016, Section 12.2].

Note that other approximation schemes can be employed. In particular, the dissipation term $-(1/n)\Delta u_n$ is not essential. However, as a matter of convenience, the use of this term allows us to comfortably construct the approximate solution (u_n, E_n, B_n) by relying on methods from the analysis of the incompressible Navier–Stokes system.

Observe that the corresponding energy inequality

$$\frac{1}{2}(\|u_n(t)\|_{L^2}^2 + \|E_n(t)\|_{L^2}^2 + \|B_n(t)\|_{L^2}^2) + \int_0^t \left(\frac{1}{n}\|\nabla u_n(\tau)\|_{L^2}^2 + \frac{1}{\sigma}\|j_n(\tau)\|_{L^2}^2\right) d\tau$$

$$\leq \frac{1}{2}\|S_n(u_0, E_0, B_0)\|_{L^2}^2 \leq \frac{1}{2}\|(u_0, E_0, B_0)\|_{L^2}^2$$

for all $t \ge 0$ is now fully justified and, since the initial data is smooth, it is possible to show that (u_n, E_n, B_n) remains smooth for all times, albeit not uniformly in n.

The above energy inequality only allows us to deduce the uniform bounds

$$(u_n, E_n, B_n) \in L^{\infty}_t L^2_x$$
 and $j_n \in L^2_{t,x}$,

which are insufficient to establish the stability of the nonlinear terms

$$(S_n u_n) \cdot \nabla u_n$$
, $(S_n j_n) \times B_n$ and $S_n P(u_n \times B_n)$

in the limit $n \to \infty$. The general strategy is therefore to show that the bounds and properties stated in Theorems 3.1, 3.2 and 3.3 can be fully justified on the smooth system (3-10) uniformly in *n*. Such uniform bounds are then sufficient to show the strong relative compactness of $\{(u_n, E_n, B_n)\}_{n=1}^{\infty}$ in $L_{t,x,\text{loc}}^2$, which then allows us to take the limit $n \to \infty$ (up to extraction of subsequences) and establish the asymptotic stability of (3-10), thereby yielding suitable solutions of the original Euler–Maxwell system (1-1).

In what follows, for the sake of simplicity, keeping in mind that all computations can be fully justified on the approximate system (3-10), we shall perform all estimates formally on the original system (1-1). In particular, we emphasize that, even though, strictly speaking, the dissipation operator $-(1/n)\Delta$ cannot be ignored, it is self-adjoint and therefore will not impact the energy estimates which are performed in the proofs below.

3.3. *Paradifferential calculus and the normal structure.* The use of Besov and Chemin–Lerner spaces in the analysis of nonlinear systems often requires a careful use of product estimates. Such results from paradifferential calculus are rather standard, but their applicability is limited. In the following paradifferential lemma, we show how the normal structure (1-2) can be exploited to extend the range of applicability of classical product estimates. This plays a central role in our analysis of (1-1).

Lemma 3.4. Let $F, G : \mathbb{R}_t \times \mathbb{R}_x^2 \to \mathbb{R}^3$ be solenoidal vector fields with the normal structure

$$F(t, x_1, x_2) = \begin{pmatrix} F_1(t, x_1, x_2) \\ F_2(t, x_1, x_2) \\ 0 \end{pmatrix} \quad and \quad G(t, x_1, x_2) = \begin{pmatrix} 0 \\ 0 \\ G_3(t, x_1, x_2) \end{pmatrix}.$$
(3-11)

Further consider integrability parameters in $[1, \infty]$ *such that*

$$\frac{1}{a} = \frac{1}{a_1} + \frac{1}{a_2}$$
 and $\frac{1}{c} = \frac{1}{c_1} + \frac{1}{c_2}$.

Then, recalling that $P = (-\Delta)^{-1}$ curl curl denotes Leray's projector onto solenoidal vector fields, one has the product estimate

$$\|P(F \times G)\|_{\tilde{L}^{a}_{t}\dot{B}^{s+t-1}_{2,c}(\mathbb{R}^{2})} \lesssim \|F\|_{\tilde{L}^{a_{1}}_{t}\dot{B}^{s}_{2,c_{1}}(\mathbb{R}^{2})}\|G\|_{\tilde{L}^{a_{2}}_{t}\dot{B}^{t}_{2,c_{2}}(\mathbb{R}^{2})}$$
(3-12)

for any $s \in (-\infty, 1)$ and $t \in (-\infty, 2)$, with s + t > 0. Furthermore, in the endpoint case s = 1, one has

$$\|P(F \times G)\|_{\tilde{L}^{a}_{t}}\dot{B}^{t}_{2,c}(\mathbb{R}^{2})} \lesssim \|F\|_{L^{a_{1}}_{t}L^{\infty}_{x}(\mathbb{R}^{2}) \cap \tilde{L}^{a_{1}}_{t}}\dot{B}^{1}_{2,\infty}(\mathbb{R}^{2})}\|G\|_{\tilde{L}^{a_{2}}_{t}}\dot{B}^{t}_{2,c}(\mathbb{R}^{2})}$$
(3-13)

for any $t \in (-1, 2)$.

Remark. The significance of the preceding lemma lies in the fact that it allows us to consider parameters in the range $t \in [1, 2)$. Without the normal structure (3-11), we would be restricted to values t < 1.

Remark. A straightforward simplification of the proof below yields a corresponding result in Besov spaces for vector fields independent of the time variable *t*.

Proof. We are going to use the paradifferential decomposition

$$F \times G = T_F G - T_G F + R(F, G),$$

where the paraproducts are defined by

$$T_F G = \sum_{j \in \mathbb{Z}} S_{j-2} F \times \Delta_j G,$$

$$T_G F = \sum_{j \in \mathbb{Z}} S_{j-2} G \times \Delta_j F = -\sum_{j \in \mathbb{Z}} \Delta_j F \times S_{j-2} G$$

and the remainder is given by

$$R(F,G) = \sum_{\substack{j,k \in \mathbb{Z} \\ |j-k| \le 2}} \Delta_j F \times \Delta_k G.$$

In particular, by virtue of the solenoidal and normal structures of F and G, one has

$$\nabla \times (F \times G) = F \times (\nabla \times G),$$

which leads to the identity

$$\nabla \times (F \times G) = \nabla \times T_F G - T_{\nabla \times G} F + \nabla \times R(F, G).$$

We therefore conclude, by standard embeddings of Besov spaces, that

$$\begin{aligned} \|P(F \times G)\|_{\tilde{L}^{a}_{t}\dot{B}^{s+t-1}_{2,c}} &\lesssim \|\nabla \times (F \times G)\|_{\tilde{L}^{a}_{t}\dot{B}^{s+t-2}_{2,c}} \\ &\lesssim \|T_{F}G\|_{\tilde{L}^{a}_{t}\dot{B}^{s+t-1}_{2,c}} + \|T_{\nabla \times G}F\|_{\tilde{L}^{a}_{t}\dot{B}^{s+t-2}_{2,c}} + \|R(F,G)\|_{\tilde{L}^{a}_{t}\dot{B}^{s+t}_{1,c}} \end{aligned}$$

for every $s, t \in \mathbb{R}$.

Next, employing the classical paradifferential estimates (A-6) and (A-7) presented in the appendix and further exploiting standard embeddings of Besov spaces, we find that

$$\begin{split} \|P(F \times G)\|_{\widetilde{L}^{a}_{t}} \dot{B}^{s+t-1}_{2,c} \\ \lesssim \|F\|_{\widetilde{L}^{a_{1}}_{t}} \dot{B}^{s-1}_{\infty,c_{1}}} \|G\|_{\widetilde{L}^{a_{2}}_{t}} \dot{B}^{t}_{2,c_{2}}} + \|F\|_{\widetilde{L}^{a_{1}}_{t}} \dot{B}^{s}_{2,c_{1}}} \|\nabla \times G\|_{\widetilde{L}^{a_{2}}_{t}} \dot{B}^{t-2}_{\infty,c_{2}}} + \|F\|_{\widetilde{L}^{a_{1}}_{t}} \dot{B}^{s}_{2,c_{1}}} \|G\|_{\widetilde{L}^{a_{2}}_{t}} \dot{B}^{t}_{2,c_{2}}} \\ \lesssim \|F\|_{\widetilde{L}^{a_{1}}_{t}} \dot{B}^{s}_{2,c_{1}}} \|G\|_{\widetilde{L}^{a_{2}}_{t}} \dot{B}^{t}_{2,c_{2}}} \end{split}$$

for any s < 1 and t < 2, such that s + t > 0. This establishes (3-12).

As for the endpoint case s = 1, if, in addition to (A-6), one also uses (A-8), then similar bounds lead to

$$\begin{split} \|P(F \times G)\|_{\tilde{L}^{a}_{t}\dot{B}^{t}_{2,c}} \\ \lesssim \|F\|_{L^{a_{1}}_{t}L^{\infty}_{x}} \|G\|_{\tilde{L}^{a_{2}}_{t}\dot{B}^{t}_{2,c}} + \|F\|_{\tilde{L}^{a_{1}}_{t}\dot{B}^{1}_{2,\infty}} \|\nabla \times G\|_{\tilde{L}^{a_{2}}_{t}\dot{B}^{t-2}_{\infty,c}} + \|F\|_{\tilde{L}^{a_{1}}_{t}\dot{B}^{1}_{2,\infty}} \|G\|_{\tilde{L}^{a_{2}}_{t}\dot{B}^{t}_{2,c}} \\ \lesssim \|F\|_{L^{a_{1}}_{t}L^{\infty}_{x}} \cap \tilde{L}^{a_{1}}_{t}\dot{B}^{1}_{2,\infty}} \|G\|_{\tilde{L}^{a_{2}}_{t}\dot{B}^{t}_{2,c}} \end{split}$$

for any -1 < t < 2, thereby establishing (3-13) and concluding the proof of the lemma.

The following ad hoc variant of a paraproduct estimate will be useful when handling vorticities which are bounded pointwise.

Lemma 3.5. Let $F, G : \mathbb{R}^2_x \to \mathbb{R}^3$ be solenoidal vector fields with the normal structure (3-11). Then, one has the product estimate

$$\|P(F \times G)\|_{\dot{B}^{1}_{2,1}(\mathbb{R}^{2})} \lesssim \|F\|_{L^{2}(\mathbb{R}^{2})} \|G\|_{\dot{B}^{1}_{\infty,1}(\mathbb{R}^{2})} + \|F\|_{\dot{H}^{1}(\mathbb{R}^{2})} \|G\|_{\dot{H}^{1}(\mathbb{R}^{2})}.$$
(3-14)

Remark. The above statement is phrased in terms of mere Besov spaces. As usual, a straightforward extension of the same result to Chemin–Lerner spaces also exists.

Proof. We employ the method of proof of Lemma 3.4. In particular, we obtain

$$\|P(F \times G)\|_{\dot{B}^{1}_{2,1}} \lesssim \|T_{F}G\|_{\dot{B}^{1}_{2,1}} + \|T_{\nabla \times G}F\|_{\dot{B}^{0}_{2,1}} + \|R(F,G)\|_{\dot{B}^{1}_{2,1}}.$$

Then, by virtue of the classical paradifferential estimates (A-6), (A-7) and (A-8) , we arrive at

$$\begin{split} \|P(F \times G)\|_{\dot{B}^{1}_{2,1}} &\lesssim \|F\|_{L^{2}} \|G\|_{\dot{B}^{1}_{\infty,1}} + \|\nabla \times G\|_{\dot{B}^{-1}_{\infty,2}} \|F\|_{\dot{B}^{1}_{2,2}} + \|F\|_{\dot{B}^{0}_{2,\infty}} \|G\|_{\dot{B}^{1}_{\infty,1}} \\ &\lesssim \|F\|_{L^{2}} \|G\|_{\dot{B}^{1}_{\infty,1}} + \|G\|_{\dot{B}^{0}_{\infty,2}} \|F\|_{\dot{B}^{1}_{2,2}}. \end{split}$$

Finally, an application of the two-dimensional embedding $\dot{B}_{2,2}^1 \subset \dot{B}_{\infty,2}^0$ concludes the proof.

3.4. *Controlling the vorticity.* In order to carry out our strategy, previously laid out in Section 1.3, we need to control the vorticity ω in L_x^p , with $p \ge 2$, by exploiting Yudovich's approach of the two-dimensional incompressible Euler equations (1-4). The following basic lemma provides us with a simple tool to do so.

Lemma 3.6. Let $(u, E, B) \in C^1([0, T) \times \mathbb{R}^2) \cap L^\infty([0, T); H^1(\mathbb{R}^2))$ be a smooth solution to (1-1) for some T > 0, with the two-dimensional normal structure (1-2). Then, for all $t \in (0, T)$,

$$\begin{split} \|\omega(t)\|_{L^{2}_{x}} &\leq \|\omega(0)\|_{L^{2}_{x}} + \|j\|_{L^{2}([0,t);L^{2}_{x})} \|\nabla B\|_{L^{2}([0,t);L^{\infty}_{x})}, \\ \|\omega(t)\|_{L^{p}_{x}} &\lesssim \|\omega(0)\|_{L^{p}_{x}} + \|j\|_{L^{2}([0,t);\dot{B}^{0}_{2,\infty})}^{2/p} \|j\|_{L^{2}([0,t);\dot{B}^{1}_{2,\infty})}^{1-2/p} \|\nabla B\|_{L^{2}([0,t);L^{\infty}_{x})} \\ \|\omega(t)\|_{L^{\infty}_{x}} &\lesssim \|\omega(0)\|_{L^{\infty}_{x}} + \|j\|_{L^{2}([0,t);L^{\infty}_{x})} \|\nabla B\|_{L^{2}([0,t);L^{\infty}_{x})} \\ &\lesssim \|\omega(0)\|_{L^{\infty}_{x}} + \|j\|_{L^{2}([0,t);\dot{B}^{1}_{2,1})} \|\nabla B\|_{L^{2}([0,t);L^{\infty}_{x})} \\ &\lesssim \|\omega(0)\|_{L^{\infty}_{x}} + \|j\|_{L^{2}([0,t);\dot{B}^{1}_{2,1})} \|\nabla B\|_{L^{2}([0,t);L^{\infty}_{x})} \end{split}$$

for any 2 .

Proof. By slight abuse of language, we assume here that the vorticity ω is defined as the scalar function $\omega = \partial_1 u_2 - \partial_2 u_1$ and that

$$u = \begin{pmatrix} u_1 \\ u_2 \end{pmatrix}$$
 and $j = \begin{pmatrix} j_1 \\ j_2 \end{pmatrix}$

In particular, a straightforward computation shows that the transport equation (1-16) can then be recast as

$$\partial_t \omega + u \cdot \nabla \omega = -j \cdot \nabla b, \tag{3-15}$$

where b(t, x) is the third component of the magnetic field B(t, x) as defined in (1-2).

Next, supposing, for simplicity, that p is finite and introducing the test function $\varphi(x) = e^{-|x|^2}$, we multiply the above vorticity transport equation by $p\omega|\omega|^{p-2}\varphi(\varepsilon x)$, where $0 < \varepsilon < 1$, and then integrate in space. This yields, since u is divergence-free,

$$\begin{split} \frac{d}{dt} \int_{\mathbb{R}^2} |\omega|^p \varphi(\varepsilon x) \, dx &\leq p \int_{\mathbb{R}^2} |j| |\nabla b| \, |\omega|^{p-1} \varphi(\varepsilon x) \, dx + \varepsilon \int_{\mathbb{R}^2} |\omega|^p u \cdot \nabla \varphi(\varepsilon x) \, dx \\ &\leq p \|j\|_{L^p_x} \|\nabla b\|_{L^\infty_x} \left(\int_{\mathbb{R}^2} |\omega|^p \varphi(\varepsilon x) \, dx \right)^{\frac{p-1}{p}} + \varepsilon \|\omega\|_{L^\infty_x}^{p-1} \|\omega\|_{L^2_x} \|u\|_{L^2_x} \|\nabla \varphi\|_{L^\infty_x}, \end{split}$$

whereby, if ω is nontrivial,

$$\frac{d}{dt} \left(\int_{\mathbb{R}^2} |\omega|^p \varphi(\varepsilon x) \, dx \right)^{\frac{1}{p}} \le \|j\|_{L^p_x} \|\nabla b\|_{L^\infty_x} + \frac{\varepsilon \|\omega\|_{L^\infty_x}^{p-1} \|\omega\|_{L^2_x} \|u\|_{L^2_x} \|\nabla \varphi\|_{L^\infty_x}}{p \left(\int_{\mathbb{R}^2} |\omega|^p \varphi(\varepsilon x) \, dx \right)^{(p-1)/p}}$$

Finally, further integrating in time and letting ε tend to zero, we conclude that

$$\|\omega(t)\|_{L^p_x} \le \|\omega(0)\|_{L^p_x} + \|j\|_{L^2([0,t);L^p_x)} \|\nabla b\|_{L^2([0,t);L^\infty_x)}$$

for any $t \in (0, T)$. A classical modification of this argument gives the same result for an infinite value of the parameter p.

Now, if p = 2, the proof is finished. If 2 , we further employ the following convexity inequality (see [Bergh and Löfström 1976] for details on interpolation theory)

$$\|j\|_{L^{2}([0,t);\dot{H}^{1-2/p})} \lesssim \|j\|_{L^{2}([0,t);\dot{B}^{0}_{2,\infty})}^{2/p} \|j\|_{L^{2}([0,t);\dot{B}^{1}_{2,\infty})}^{1-2/p},$$

in combination with the classical two-dimensional Sobolev embedding $\dot{H}^{1-2/p} \subset L^p$, to conclude that

$$\|\omega(t)\|_{L^p_x} \lesssim \|\omega(0)\|_{L^p_x} + \|j\|_{L^2([0,t);\dot{B}^0_{2,\infty})}^{2/p} \|j\|_{L^2([0,t);\dot{B}^1_{2,\infty})}^{1-2/p} \|\nabla b\|_{L^2([0,t);L^\infty_x)}.$$

Finally, the case $p = \infty$ is settled with an application of the continuous embeddings $\dot{B}_{2,1}^1 \subset \dot{B}_{\infty,1}^0 \subset L^\infty$, valid in two dimensions, thereby completing the proof of the lemma.

Remark. Recall that the approximation procedure presented in Section 3.2 relies on a viscous approximation of the Euler system. It is therefore important to emphasize here that the method of proof of the preceding lemma also applies to viscous approximations of the transport equation (3-15). Indeed, observe that multiplying the transport-diffusion equation

$$\partial_t \omega + u \cdot \nabla \omega - \Delta \omega = -j \cdot \nabla b$$

by $\beta'(\omega)$, for some convex nonnegative renormalization $\beta \in C^2(\mathbb{R})$ with $\beta(0) = 0$, and then integrating in space leads, at least formally, to the estimate

$$\frac{d}{dt} \int_{\mathbb{R}^2} \beta(\omega) \, dx \le \frac{d}{dt} \int_{\mathbb{R}^2} \beta(\omega) \, dx + \int_{\mathbb{R}^2} |\nabla \omega|^2 \beta''(\omega) \, dx = \int_{\mathbb{R}^2} (j \cdot \nabla b) \beta'(\omega) \, dx.$$

This observation can be exploited to derive the estimates stated in Lemma 3.6 in spite of the diffusion term. Thus, we conclude that the approximated system (3-10) is well-suited for an application of the estimates from Lemma 3.6.

Recall now that the space $\dot{H}^1(\mathbb{R}^2)$ barely fails to embed into $L^{\infty}(\mathbb{R}^2)$. Thus, the bounds produced in the preceding lemma will be very useful to recover, by interpolation, an L^{∞} -bound on the velocity field u, as explained in the following simple result.

Lemma 3.7. Let (u, E, B) be a smooth solution to (1-1), with the two-dimensional normal structure (1-2). *Then, for all* T > 0,

$$\|u(T)\|_{L^{\infty}_{x}} \lesssim \|u(T)\|_{L^{2}_{x}}^{(p-2)/(2(p-1))} (\|u(0)\|_{\dot{W}^{1,p}_{x}} + \|j\|_{L^{2}_{t}\dot{B}^{0}_{2,\infty}}^{2/p} \|j\|_{L^{2}_{t}\dot{B}^{1}_{2,\infty}}^{1-2/p} \|\nabla B\|_{L^{2}_{t}L^{\infty}_{x}})^{\frac{p}{2(p-1)}}$$

for any 2 , where all time-norms are taken over the interval <math>[0, T).

Proof. This result follows from the classical Gagliardo–Nirenberg interpolation inequality in two spacedimensions. For the sake of convenience, we provide a brief justification of the precise inequality which is employed here.

Specifically, for any $k \in \mathbb{Z}$ and $2^k \leq R < 2^{k+1}$, we estimate that

$$\|u\|_{L^{\infty}} \leq \|S_{k}u\|_{L^{\infty}} + \sum_{i=k}^{\infty} \|\Delta_{i}u\|_{L^{\infty}} \leq 2^{k} \|u\|_{L^{2}} + \sum_{i=k}^{\infty} 2^{i\left(\frac{2}{p}-1\right)} \|\nabla u\|_{L^{p}} \leq R \|u\|_{L^{2}} + R^{-\left(1-\frac{2}{p}\right)} \|\nabla u\|_{L^{p}},$$

which yields, upon optimization of the interpolation parameter value R > 0, the Gagliardo–Nirenberg inequality

$$\|u\|_{L^{\infty}(\mathbb{R}^{2})} \lesssim \|u\|_{L^{2}(\mathbb{R}^{2})}^{(p-2)/(2(p-1))} \|\nabla u\|_{L^{p}(\mathbb{R}^{2})}^{p/(2(p-1))}$$
(3-16)

for any 2 .

Then, combining this convexity inequality with the estimates from Lemma 3.6 and recalling the equivalence $\|\nabla u\|_{L^p} \sim \|\omega\|_{L^p}$, because *u* is divergence-free, concludes the proof of the lemma.

Remark. Note that the proof of (3-16) can be adapted to establish, for any divergence-free field u, that

$$\|u\|_{L^{\infty}(\mathbb{R}^{2})} \lesssim \|u\|_{L^{2}(\mathbb{R}^{2})}^{(p-2)/(2(p-1))} \|\nabla \times u\|_{L^{p}(\mathbb{R}^{2})}^{p/(2(p-1))}$$
(3-17)

for all values 2 . Indeed, this follows from the observation that

$$\|\Delta_i u\|_{L^{\infty}} \lesssim 2^{-i} \|\Delta_i \nabla \times u\|_{L^{\infty}} \lesssim 2^{i\left(\frac{2}{p}-1\right)} \|\nabla \times u\|_{L^p},$$

provided div u = 0.

3.5. *Control of high-frequency damped electromagnetic waves.* The following result follows from a simple but careful combination of the damped Strichartz estimates for high electromagnetic frequencies, established in Section 2.3, with the paradifferential product estimates from Lemma 3.4.

Lemma 3.8. Let d = 2. Assume that (E, B) is a smooth solution to (1-15) for some initial data (E_0, B_0) and some divergence-free vector field u, with the normal structure (1-2).

Then, for any exponents $1 \le p \le q \le \infty$, $2 \le r \le \infty$ and $1 \le n \le \infty$ which are admissible in the sense that

$$\frac{2}{q} + \frac{1}{r} \ge \frac{1}{2},$$
 (3-18)

one has the high-frequency estimate over any time interval [0, T) for any $0 < \alpha < 1$ and s < 2, with $\alpha + s > 0$,

$$\begin{aligned} \|(E,B)\|_{\tilde{L}^{q}_{t}\dot{B}^{s+\alpha-7/4+3/(2r)}_{r,n,>}} \\ \lesssim c^{\frac{1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)-\frac{2}{q}} \|(E_{0},B_{0})\|_{\dot{B}^{s+\alpha-1}_{2,n,>}} + c^{\frac{1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)+2\left(\frac{1}{p}-\frac{1}{q}\right)-1} \|u\|_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}}^{1-\alpha} \|u\|_{L^{\infty}_{t}\dot{B}^{1}_{2,\infty}}^{\alpha} \|B\|_{\tilde{L}^{p}_{t}\dot{B}^{s}_{2,\infty}}. \end{aligned}$$

In the endpoint cases $\alpha = 1$ and $\alpha = 0$, one also has the respective estimates

$$\| (E,B) \|_{\tilde{L}_{t}^{q}} \dot{B}_{r,n,>}^{s-(3/2)(1/2-1/r)} \lesssim c^{\frac{1}{2}(\frac{1}{2}-\frac{1}{r})-\frac{2}{q}} \| (E_{0},B_{0}) \|_{\dot{B}_{2,n,>}^{s}} + c^{\frac{1}{2}(\frac{1}{2}-\frac{1}{r})+2(\frac{1}{p}-\frac{1}{q})-1} \| u \|_{L_{t,x}^{\infty} \cap L_{t}^{\infty}} \dot{B}_{2,\infty}^{1} \| B \|_{\tilde{L}_{t}^{p}} \dot{B}_{2,n}^{s}$$
(3-19)

for any -1 < s < 2 and

$$\|(E,B)\|_{\tilde{L}^{q}_{t}\dot{B}^{s-7/4+3/(2r)}_{r,n,>}} \lesssim c^{\frac{1}{2}(\frac{1}{2}-\frac{1}{r})-\frac{2}{q}} \|(E_{0},B_{0})\|_{\dot{B}^{s-1}_{2,n,>}} + c^{\frac{1}{2}(\frac{1}{2}-\frac{1}{r})+2(\frac{1}{p}-\frac{1}{q})-1} \|u\|_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}} \|B\|_{\tilde{L}^{p}_{t}\dot{B}^{s}_{2,n}}$$
for any $0 < s < 2$.

Proof. Considering Maxwell's system (1-15) and applying Corollary 2.12, with $\tilde{r} = 2$ and $\tilde{q} = p'$, yields the high-frequency estimate

$$2^{-j\frac{3}{2}\left(\frac{1}{2}-\frac{1}{r}\right)} \|\Delta_{j}(E,B)\|_{L_{t}^{q}L_{x}^{r}} \lesssim c^{\frac{1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)-\frac{2}{q}} (\|\Delta_{j}(E_{0},B_{0})\|_{L_{x}^{2}} + c^{\frac{2}{p}-1} \|\Delta_{j}P(u\times B)\|_{L_{t}^{p}L_{x}^{2}})$$

for all $j \in \mathbb{Z}$ with $2^j \ge \sigma c$, where $1 \le p \le q \le \infty$ and $2 \le r \le \infty$ must satisfy (3-18). It is to be emphasized that, thanks to the damping phenomenon in (1-15), all estimates here hold uniformly over any time interval [0, T), where $T = \infty$ is allowed.

Next, summing the preceding estimate in j and utilizing the paradifferential product law (3-13), we deduce that

for any -1 < s < 2. If, instead of using (3-13), one employs the paradifferential product law (3-12), then one arrives at the estimate

$$\begin{split} \|\mathbb{1}_{\{2^{j} \geq \sigma c\}} 2^{j(s+\alpha-\frac{7}{4}+\frac{3}{2r})} \|\Delta_{j}(E,B)\|_{L_{t}^{q}L_{x}^{r}}\|_{\ell^{n}} \\ &\lesssim c^{\frac{1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)-\frac{2}{q}} \|(E_{0},B_{0})\|_{\dot{B}_{2,n,>}^{s+\alpha-1}} + c^{\frac{1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)+2\left(\frac{1}{p}-\frac{1}{q}\right)-1} \|P(u\times B)\|_{\tilde{L}_{t}^{p}}\dot{B}_{2,n}^{s+\alpha-1} \\ &\lesssim c^{\frac{1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)-\frac{2}{q}} \|(E_{0},B_{0})\|_{\dot{B}_{2,n,>}^{s+\alpha-1}} + c^{\frac{1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)+2\left(\frac{1}{p}-\frac{1}{q}\right)-1} \|u\|_{\tilde{L}_{t}^{\infty}}\dot{B}_{2,1}^{\alpha}\|B\|_{\tilde{L}_{t}^{p}}\dot{B}_{2,\infty}^{s} \\ &\lesssim c^{\frac{1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)-\frac{2}{q}} \|(E_{0},B_{0})\|_{\dot{B}_{2,n,>}^{s+\alpha-1}} + c^{\frac{1}{2}\left(\frac{1}{2}-\frac{1}{r}\right)+2\left(\frac{1}{p}-\frac{1}{q}\right)-1} \|u\|_{\tilde{L}_{t}^{\infty}}\dot{B}_{2,\infty}^{0}} \|u\|_{\tilde{L}_{t}^{\infty}}\dot{B}_{2,\infty}^{1}} \|B\|_{\tilde{L}_{t}^{p}}\dot{B}_{2,\infty}^{s}, \end{split}$$

which is valid for any $0 < \alpha < 1$ and s < 2, with $\alpha + s > 0$, where we exploited the fact that $\tilde{L}_t^{\infty} \dot{B}_{2,1}^{\alpha}$ is an interpolation space between $\tilde{L}_t^{\infty} \dot{B}_{2,\infty}^0$ and $\tilde{L}_t^{\infty} \dot{B}_{2,\infty}^1$ (see [Bergh and Löfström 1976] for details on interpolation theory).

Similarly, the case $\alpha = 0$ yields

for any 0 < s < 2. This completes the justification of the high-frequency estimates.

3.6. *Control of low-frequency damped electromagnetic waves.* The statement of Corollary 2.12 has clearly emphasized how solutions to the damped Maxwell system enjoy intrinsically different properties on the distinct ranges of low and high frequencies.

In particular, the combination of Corollary 2.12 with the paradifferential Lemma 3.4 resulted in the nonlinear high-frequency estimates of Lemma 3.8. A similar strategy based on the same corollary could now be employed to deduce suitable nonlinear low-frequency estimates.

However, in the next lemma, we are instead going to exploit the refined estimates established in Corollary 2.14, which are a consequence of the maximal parabolic regularity studied in Section 2.2, to obtain a sharper control of low frequencies. This will lead to stronger statements of our main theorems.

Lemma 3.9. Let d = 2. Assume that (E, B) is a smooth solution to (1-15) for some initial data (E_0, B_0) and some divergence-free vector field u, with the normal structure (1-2).

Then, for any exponents $1 and <math>1 \le n \le \infty$, one has the following low-frequency estimates over any time interval [0, *T*). For any $0 < \alpha < 1$ and s < 2, with $\alpha + s > 0$,

$$\begin{split} \|E\|_{L_{t}^{q}\dot{B}_{2,n,<}^{s+\alpha-1}} &\lesssim c^{-\frac{d}{q}} \|E_{0}\|_{\dot{B}_{2,n,<}^{s+\alpha-1}} + c^{-1} \|B_{0}\|_{\dot{B}_{2,q,<}^{s+\alpha-2/q}} \\ &+ c^{2\left(\frac{1}{p} - \frac{1}{q}\right) - 1} \|u\|_{L_{t}^{\infty}\dot{B}_{2,\infty}^{0}}^{1-\alpha} \|u\|_{L_{t}^{\infty}\dot{B}_{2,\infty}^{1}}^{\alpha} \|B\|_{L_{t}^{p}\dot{B}_{2,\infty}^{s}}, \\ \|B\|_{L_{t}^{q}\dot{B}_{2,1,<}^{s+\alpha+2/q-2/p}} &\lesssim c^{-1} \|E_{0}\|_{\dot{B}_{2,q,<}^{s+\alpha+1-2/p}} + \|B_{0}\|_{\dot{B}_{2,q,<}^{s+\alpha-2/p}} \\ &+ \|u\|_{L_{t}^{\infty}\dot{B}_{2,\infty}^{0}}^{1-\alpha} \|u\|_{L_{t}^{\infty}\dot{B}_{2,\infty}^{1}}^{\alpha} \|B\|_{L_{t}^{p}\dot{B}_{2,\infty}^{s}}. \end{split}$$
(3-20)

In the endpoint case $\alpha = 1$, with -1 < s < 2, one also has the estimates

 $\|E\|_{L^{q}_{t}\dot{B}^{s}_{2,n,<}} \lesssim c^{-\frac{2}{q}} \|E_{0}\|_{\dot{B}^{s}_{2,n,<}} + c^{-1} \|B_{0}\|_{\dot{B}^{s+1-2/q}_{2,q,<}} + c^{2\left(\frac{1}{p}-\frac{1}{q}\right)-1} \|u\|_{L^{\infty}_{t,x}\cap L^{\infty}_{t}\dot{B}^{1}_{2,\infty}} \|B\|_{L^{p}_{t}\dot{B}^{s}_{2,n}},$ (3-21) as well as, if p < q,

$$\|B\|_{L^q_t \dot{B}^{s+1+2/q-2/p}_{2,1,<}} \lesssim c^{-1} \|E_0\|_{\dot{B}^{s+2-2/p}_{2,q,<}} + \|B_0\|_{\dot{B}^{s+1-2/p}_{2,q,<}} + \|u\|_{L^\infty_{t,x} \cap L^\infty_t \dot{B}^1_{2,\infty}} \|B\|_{L^p_t \dot{B}^s_{2,\infty}}$$

and, if $p = q$,

$$\|B\|_{L^{q}_{t}\dot{B}^{s+1}_{2,n,<}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+2-2/q}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s+1-2/q}_{2,q,<}} + \|u\|_{L^{\infty}_{t,x}\cap L^{\infty}_{t}\dot{B}^{1}_{2,\infty}} \|B\|_{L^{q}_{t}\dot{B}^{s}_{2,n}}.$$
(3-22)

Finally, in the remaining endpoint case $\alpha = 0$, with 0 < s < 2,

$$\|E\|_{L^{q}_{t}\dot{B}^{s-1}_{2,n,<}} \lesssim c^{-\frac{2}{q}} \|E_{0}\|_{\dot{B}^{s-1}_{2,n,<}} + c^{-1} \|B_{0}\|_{\dot{B}^{s-2/q}_{2,q,<}} + c^{2(\frac{1}{p}-\frac{1}{q})-1} \|u\|_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}} \|B\|_{L^{p}_{t}\dot{B}^{s}_{2,n}},$$

as well as, if $p < q$,

$$\|B\|_{L^{q}_{t}\dot{B}^{s+2/q-2/p}_{2,1,<}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+1-2/p}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s-2/p}_{2,q,<}} + \|u\|_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}} \|B\|_{L^{p}_{t}\dot{B}^{s}_{2,\infty}}$$

and, if p = q,

$$\|B\|_{L^{q}_{t}\dot{B}^{s}_{2,n,<}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+1-2/q}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s-2/q}_{2,q,<}} + \|u\|_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}} \|B\|_{L^{q}_{t}\dot{B}^{s}_{2,n}}.$$
 (3-23)

Remark. The devil is in the details. In the statement of Lemma 3.9 and its corresponding proof below, we have paid painstaking attention to the summability index of Besov spaces, occasionally referred to as the third index. Thus, in the above statement, sometimes the third index is q, other times it is n, 1 or ∞ . Either way, we believe that these defining values of Besov spaces are optimal, which is a crucial step to reach sharp statements of our main results.

Proof. Let us consider some fixed regularity parameters $s \in \mathbb{R}$ and $0 \le \alpha \le 1$.

On the one hand, an application of Corollary 2.12 with $r = \tilde{r} = 2$ (note that this is the best possible choice for r and \tilde{r} , since all other estimates for values $2 \le r, \tilde{r} \le \infty$ follow from the case $r = \tilde{r} = 2$ by Sobolev embeddings) and $\tilde{q} = p'$ gives

$$2^{j(s+\alpha-1)} \|\Delta_j E\|_{L^q_t L^2_x} \lesssim c^{-\frac{2}{q}} \|E_0\|_{\dot{B}^{s+\alpha-1}_{2,\infty,<}} + c^{-1} \|B_0\|_{\dot{B}^{s+\alpha-2/q}_{2,\infty,<}} + c^{2(\frac{1}{p}-\frac{1}{q})-1} \|P(u\times B)\|_{L^p_t \dot{B}^{s+\alpha-1}_{2,\infty}}$$

and

$$2^{j(s+\alpha+\frac{2}{q}-\frac{2}{p})} \|\Delta_{j}B\|_{L^{q}_{t}L^{2}_{x}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+\alpha+1-2/p}_{2,\infty,<}} + \|B_{0}\|_{\dot{B}^{s+\alpha-2/p}_{2,\infty,<}} + \|P(u\times B)\|_{L^{p}_{t}\dot{B}^{s+\alpha-1}_{2,\infty,<}}$$

for all $j \in \mathbb{Z}$ with $2^j < \sigma c$, where $1 \le p \le q \le \infty$.

On the other hand, employing Corollary 2.14 yields

$$\begin{aligned} \|\mathbb{1}_{\{2^{j} < \frac{1}{2}\sigma c\}} 2^{j(s+\alpha-1)} \|\Delta_{j} E\|_{L^{2}_{x}} \|_{L^{q}_{t}\ell^{n}_{j}} \\ \lesssim c^{-\frac{2}{q}} \|E_{0}\|_{\dot{B}^{s+\alpha-1}_{2,n,<}} + c^{-1} \|B_{0}\|_{\dot{B}^{s+\alpha-2/q}_{2,q,<}} + c^{2\left(\frac{1}{p}-\frac{1}{q}\right)-1} \|P(u\times B)\|_{L^{p}_{t}} \dot{B}^{s+\alpha-1}_{2,n} \end{aligned}$$

for any $1 and <math>1 \le n \le \infty$, as well as

$$\begin{aligned} \|\mathbb{1}_{\{2^{j} < \frac{1}{2}\sigma c\}} 2^{j(s+\alpha+\frac{2}{q}-\frac{2}{p})} \|\Delta_{j}B\|_{L^{2}_{x}} \|_{L^{q}_{t}\ell^{1}_{j}} \\ \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+\alpha+1-2/p}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s+\alpha-2/p}_{2,q,<}} + \|P(u\times B)\|_{L^{p}_{t}\dot{B}^{s+\alpha-1}_{2,\infty}} \end{aligned}$$

for any 1 , and

$$\|\mathbb{1}_{\{2^{j} < \frac{1}{2}\sigma c\}} 2^{j(s+\alpha)} \|\Delta_{j}B\|_{L^{2}_{x}} \|_{L^{q}_{t}\ell^{n}_{j}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+\alpha+1-2/q}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s+\alpha-2/q}_{2,q,<}} + \|P(u \times B)\|_{L^{q}_{t}\dot{B}^{s+\alpha-1}_{2,n}}$$

for any $1 < q < \infty$ and $1 \le n \le \infty$

for any $1 < q < \infty$ and $1 \le n \le \infty$.

On the whole, combining the above estimates, we arrive at the conclusion that

$$\|E\|_{L^{q}_{t}\dot{B}^{s+\alpha-1}_{2,n,<}} \lesssim c^{-\frac{2}{q}} \|E_{0}\|_{\dot{B}^{s+\alpha-1}_{2,n,<}} + c^{-1} \|B_{0}\|_{\dot{B}^{s+\alpha-2/q}_{2,q,<}} + c^{2(\frac{1}{p}-\frac{1}{q})-1} \|P(u\times B)\|_{L^{p}_{t}\dot{B}^{s+\alpha-1}_{2,n}}$$

for any $1 and <math>1 \le n \le \infty$, as well as

$$\|B\|_{L^{q}_{t}\dot{B}^{s+\alpha+2/q-2/p}_{2,1,<}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+\alpha+1-2/p}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s+\alpha-2/p}_{2,q,<}} + \|P(u \times B)\|_{L^{p}_{t}\dot{B}^{s+\alpha-1}_{2,\infty}}$$

for any 1 , and

$$\|B\|_{L^{q}_{t}\dot{B}^{s+\alpha}_{2,n,<}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+\alpha+1-2/q}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s+\alpha-2/q}_{2,q,<}} + \|P(u \times B)\|_{L^{q}_{t}\dot{B}^{s+\alpha-1}_{2,n}}$$

for any $1 < q < \infty$ and $1 \le n \le \infty$.

We are now going to apply the paradifferential product inequalities from Lemma 3.4 to the three preceding controls. More precisely, setting $\alpha = 1$, restricting the range of *s* to (-1, 2) and utilizing the product law (3-13) to handle the nonlinear term $u \times B$, we obtain

$$\|E\|_{L^{q}_{t}\dot{B}^{s}_{2,n,<}} \lesssim c^{-\frac{2}{q}} \|E_{0}\|_{\dot{B}^{s}_{2,n,<}} + c^{-1} \|B_{0}\|_{\dot{B}^{s+1-2/q}_{2,q,<}} + c^{2(\frac{1}{p}-\frac{1}{q})-1} \|u\|_{L^{\infty}_{t,x}\cap L^{\infty}_{t}\dot{B}^{1}_{2,\infty}} \|B\|_{L^{p}_{t}\dot{B}^{s}_{2,n}}$$

for any $1 and <math>1 \le n \le \infty$, as well as

$$\|B\|_{L^{q}_{t}\dot{B}^{s+1+2/q-2/p}_{2,1,<}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+2-2/p}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s+1-2/p}_{2,q,<}} + \|u\|_{L^{\infty}_{t,x}\cap L^{\infty}_{t}\dot{B}^{1}_{2,\infty}} \|B\|_{L^{p}_{t}\dot{B}^{s}_{2,\infty}}$$

for any 1 , and

$$\|B\|_{L^{q}_{t}\dot{B}^{s+1}_{2,n,<}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+2-2/q}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s+1-2/q}_{2,q,<}} + \|u\|_{L^{\infty}_{t,x} \cap L^{\infty}_{t}\dot{B}^{1}_{2,\infty}} \|B\|_{L^{q}_{t}\dot{B}^{s}_{2,n}}$$

for any $1 < q < \infty$ and $1 \le n \le \infty$.

Similarly, choosing parameters $0 < \alpha < 1$ and s < 2, with $\alpha + s > 0$, utilizing (3-12) instead of (3-13), and exploiting again the fact that $L_t^{\infty} \dot{B}_{2,1}^{\alpha}$ is an interpolation space between $L_t^{\infty} \dot{B}_{2,\infty}^0$ and $L_t^{\infty} \dot{B}_{2,\infty}^1$, we find that

$$\begin{split} \|E\|_{L^{q}_{t}\dot{B}^{s+\alpha-1}_{2,n,<}} \lesssim c^{-\frac{2}{q}} \|E_{0}\|_{\dot{B}^{s+\alpha-1}_{2,n,<}} + c^{-1} \|B_{0}\|_{\dot{B}^{s+\alpha-2/q}_{2,q,<}} \\ + c^{2\left(\frac{1}{p}-\frac{1}{q}\right)-1} \|u\|_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}}^{1-\alpha} \|u\|_{L^{\infty}_{t}\dot{B}^{1}_{2,\infty}}^{\alpha} \|B\|_{L^{p}_{t}\dot{B}^{s}_{2,\infty}} \end{split}$$

for any $1 and <math>1 \le n \le \infty$, as well as

$$\|B\|_{L^{q}_{t}\dot{B}^{s+\alpha+2/q-2/p}_{2,1,<}} \lesssim c^{-1}\|E_{0}\|_{\dot{B}^{s+\alpha+1-2/p}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s+\alpha-2/p}_{2,q,<}} + \|u\|^{1-\alpha}_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}} \|u\|^{\alpha}_{L^{\infty}_{t}\dot{B}^{1}_{2,\infty}} \|B\|_{L^{p}_{t}\dot{B}^{s}_{2,\infty}}$$

for any 1 .

Finally, the case $\alpha = 0$, with 0 < s < 2, is handled with the same product estimate (3-12) and results in

$$\|E\|_{L^{q}_{t}\dot{B}^{s-1}_{2,n,<}} \lesssim c^{-\frac{2}{q}} \|E_{0}\|_{\dot{B}^{s-1}_{2,n,<}} + c^{-1} \|B_{0}\|_{\dot{B}^{s-2/q}_{2,q,<}} + c^{2\left(\frac{1}{p}-\frac{1}{q}\right)-1} \|u\|_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}} \|B\|_{L^{p}_{t}\dot{B}^{s}_{2,n,<}}$$

for any $1 and <math>1 \le n \le \infty$, as well as

$$\|B\|_{L^q_t \dot{B}^{s+2/q-2/p}_{2,1,<}} \lesssim c^{-1} \|E_0\|_{\dot{B}^{s+1-2/p}_{2,q,<}} + \|B_0\|_{\dot{B}^{s-2/p}_{2,q,<}} + \|u\|_{L^\infty_t \dot{B}^0_{2,\infty}} \|B\|_{L^p_t \dot{B}^s_{2,\infty}}$$

for any 1 , and

$$\|B\|_{L^{q}_{t}\dot{B}^{s}_{2,n,<}} \lesssim c^{-1} \|E_{0}\|_{\dot{B}^{s+1-2/q}_{2,q,<}} + \|B_{0}\|_{\dot{B}^{s-2/q}_{2,q,<}} + \|u\|_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}} \|B\|_{L^{q}_{t}\dot{B}^{s}_{2,n}}$$

for any $1 < q < \infty$ and $1 \le n \le \infty$, thereby completing the justifications of all low-frequency estimates. \Box

Note that the estimates from Lemma 3.9 do not include the value $q = \infty$. Rather than providing a technical extension of the preceding proof to incorporate the value $q = \infty$, we show in the next result a simple energy estimate on Maxwell's system (1-15), which corresponds to the case $q = \infty$ in Lemma 3.9. This energy estimate allows us to propagate the \dot{H}_x^1 -norm of electromagnetic fields, which will come in handy in the proof of Theorem 3.1 below.

Lemma 3.10. Let d = 2. Assume that (E, B) is a smooth solution to (1-15) for some initial data (E_0, B_0) and some divergence-free vector field u, with the normal structure (1-2).

Then, one has the estimates

$$\begin{aligned} \|(E,B)\|_{L_{t}^{\infty}\dot{H}_{x}^{1}} + c \|E\|_{L_{t}^{2}\dot{H}_{x}^{1}} &\lesssim \|(E_{0},B_{0})\|_{\dot{H}_{x}^{1}} + \|u\|_{L_{t}^{\infty}L_{x}^{2}} \|\nabla B\|_{L_{t}^{2}L_{x}^{\infty}}, \\ \|(E,B)\|_{L_{t}^{\infty}\dot{H}_{x}^{1}} + c \|E\|_{L_{t}^{2}\dot{H}_{x}^{1}} &\lesssim \|(E_{0},B_{0})\|_{\dot{H}_{x}^{1}} + \|u\|_{L_{t}^{\infty}\dot{B}_{2,\infty}^{0}}^{1-\alpha} \|u\|_{L_{t}^{\infty}\dot{B}_{2,\infty}^{1-\alpha}}^{\alpha} \|B\|_{L_{t}^{2}\dot{B}_{2,\infty}^{2-\alpha}}, \\ \|(E,B)\|_{L_{t}^{\infty}\dot{H}_{x}^{1}} + c \|E\|_{L_{t}^{2}\dot{H}_{x}^{1}} &\lesssim \|(E_{0},B_{0})\|_{\dot{H}_{x}^{1}} + \|u\|_{L_{t,x}^{\infty}\cap L_{t}^{\infty}\dot{B}_{2,\infty}^{1-\alpha}}^{1-\alpha} \|B\|_{L_{t}^{2}\dot{H}_{x}^{1}} \end{aligned}$$
(3-24)

over any time interval [0, T) for any $0 < \alpha < 1$.

Proof. We perform a classical energy estimate on (1-15). More precisely, defining

$$(\tilde{E}, \tilde{B}) := (\nabla \times E, \nabla \times B),$$

we observe from (1-15) that (\tilde{E}, \tilde{B}) solves the system

$$\begin{cases} \frac{1}{c}\partial_t \widetilde{E} - \nabla \times \widetilde{B} + \sigma c \widetilde{E} = -\sigma \nabla \times (u \times B) = \sigma (u \cdot \nabla) B - \sigma (B \cdot \nabla) u, \\ \frac{1}{c}\partial_t \widetilde{B} + \nabla \times \widetilde{E} = 0, \end{cases}$$

where we employed the fact that u and B are divergence-free fields. In fact, the preceding step holds in any dimension d = 2 or d = 3.

However, restricting ourselves to the two-dimensional setting and assuming that the field (u, E, B) satisfies the structure (1-2) allows us to discard the term $(B \cdot \nabla)u$. To be precise, we now have that (\tilde{E}, \tilde{B}) solves

$$\begin{cases} \frac{1}{c}\partial_t \widetilde{E} - \nabla \times \widetilde{B} + \sigma c \widetilde{E} = -\sigma u \times \widetilde{B}, \\ \frac{1}{c}\partial_t \widetilde{B} + \nabla \times \widetilde{E} = 0. \end{cases}$$

Thus, multiplying the first equation by \tilde{E} , the second by \tilde{B} and integrating in time and space, we deduce that

$$\begin{split} \frac{1}{2c} \| (\tilde{E}, \tilde{B})(T) \|_{L_x^2}^2 + \sigma c \| \tilde{E} \|_{L_t^2([0,T);L_x^2)}^2 \\ & \leq \frac{1}{2c} \| (\tilde{E}, \tilde{B})(0) \|_{L_x^2}^2 + \sigma \| u \times \tilde{B} \|_{L_t^2([0,T);L_x^2)} \| \tilde{E} \|_{L_t^2([0,T);L_x^2)} \\ & \leq \frac{1}{2c} \| (\tilde{E}, \tilde{B})(0) \|_{L_x^2}^2 + \frac{\sigma}{2c} \| u \times \tilde{B} \|_{L_t^2([0,T);L_x^2)}^2 + \frac{\sigma c}{2} \| \tilde{E} \|_{L_t^2([0,T);L_x^2)}^2 \end{split}$$

Further employing the fact that

$$\|(E,B)\|_{\dot{H}^1_x} = \|(\nabla E,\nabla B)\|_{L^2_x} = \|(\tilde{E},\tilde{B})\|_{L^2_x},$$

we arrive at the estimate

$$\|(E,B)(T)\|_{\dot{H}_{x}^{1}}^{2} + \sigma c^{2} \|E\|_{L^{2}_{t}([0,T);\dot{H}_{x}^{1})}^{2} \leq \|(E_{0},B_{0})\|_{\dot{H}_{x}^{1}}^{2} + \sigma \|u \times \widetilde{B}\|_{L^{2}_{t}([0,T);L^{2}_{x})}^{2}.$$

Therefore, the proof will be completed upon controlling the nonlinear term $\|u \times \widetilde{B}\|_{L^2([0,T);L^2_X)}^2$.

To this end, an elementary application of Hölder's inequality first leads to

$$\|u\times\widetilde{B}\|_{L^2_{t,x}}\leq\|u\|_{L^\infty_tL^2_x}\|\nabla B\|_{L^2_tL^\infty_x},$$

thereby completing the justification of the first estimate of the lemma.

Alternatively, following the proof of Lemma 3.9, one can use the paraproduct estimates from Lemma 3.4 again. More specifically, utilizing the product law (3-13), we obtain

$$\|u \times \widetilde{B}\|_{L^{2}_{t,x}} = \|P(u \times B)\|_{L^{2}_{t}\dot{H}^{1}_{x}} \lesssim \|u\|_{L^{\infty}_{t,x} \cap L^{\infty}_{t}\dot{B}^{1}_{2,\infty}} \|B\|_{L^{2}_{t}\dot{H}^{1}_{x}}$$

Similarly, choosing a parameter $0 < \alpha < 1$, utilizing (3-12) instead of (3-13), and exploiting the fact that $L_t^{\infty} \dot{B}_{2,2}^{\alpha}$ is an interpolation space between $L_t^{\infty} \dot{B}_{2,\infty}^0$ and $L_t^{\infty} \dot{B}_{2,\infty}^1$, we finally infer that

$$\|u \times \widetilde{B}\|_{L^{2}_{t,x}} = \|P(u \times B)\|_{L^{2}_{t}\dot{H}^{1}_{x}} \lesssim \|u\|_{L^{\infty}_{t}\dot{B}^{\alpha}_{2,2}} \|B\|_{L^{2}_{t}\dot{B}^{2-\alpha}_{2,\infty}} \lesssim \|u\|_{L^{\infty}_{t}\dot{B}^{0}_{2,\infty}}^{1-\alpha} \|u\|_{L^{\infty}_{t}\dot{B}^{1}_{2,\infty}}^{\alpha} \|B\|_{L^{2}_{t}\dot{B}^{2-\alpha}_{2,\infty}},$$

which concludes the proof.

3.7. *Almost-parabolic estimates on the magnetic field.* In the singular limit $c \to \infty$, Maxwell's equations (1-15) formally converge towards the parabolic system

$$\partial_t B + (u \cdot \nabla) B - \frac{1}{\sigma} \Delta B = (B \cdot \nabla) u,$$

where we employed the fact that u and B are divergence-free. This holds in both dimensions d = 2 and d = 3. Further assuming the two-dimensional normal structure (1-2), the preceding system reduces to the simple transport-diffusion equation

$$\partial_t b + u \cdot \nabla b - \frac{1}{\sigma} \Delta b = 0$$

which satisfies the energy inequality

$$\frac{1}{2} \|b(T)\|_{L^2_x}^2 + \frac{1}{\sigma} \|\nabla b\|_{L^2_x}^2 \le \frac{1}{2} \|b(0)\|_{L^2_x}^2$$
(3-25)

for all T > 0, at least formally.

The estimates provided by Lemma 3.9 fail to fully capture this asymptotic parabolic behavior of Maxwell's equations, because they always contain a control of the nonlinear term $u \cdot \nabla b$, whereas this term does not contribute to the energy dissipation inequality (3-25).

The next result establishes a singular almost-parabolic energy estimate for Maxwell's system (1-15) which recovers the classical a priori estimate (3-25) (up to multiplicative constants) for the heat equation in the limit $c \rightarrow \infty$. This is crucial to our work, as it will allow us to construct solutions to the Euler–Maxwell system (1-1) for arbitrarily large initial data as the speed of light *c* tends to infinity.

Lemma 3.11. Let d = 2. Assume that (E, B) is a smooth solution to (1-15) for some initial data (E_0, B_0) and some divergence-free vector field u, with the normal structure (1-2).

Then, one has the estimate

$$\|B\|_{L^{\infty}_{t}L^{2}_{x}} + \|\nabla B\|_{L^{2}_{t}L^{2}_{x}} \lesssim \|B_{0}\|_{L^{2}_{x}} + c^{-1}\|(E_{0}, B_{0})\|_{\dot{H}^{1}_{x}} + c^{-1}\|u\|_{L^{\infty}_{t}L^{2}_{x}}\|\nabla B\|_{L^{2}_{t}L^{\infty}_{x}}$$
(3-26) over any time interval $[0, T)$.

Proof. First, we observe that taking the curl of Ampère's equation in (1-15) and then employing Faraday's equation to substitute $\nabla \times E = -(1/c)\partial_t B$, when necessary, leads to the system

$$\partial_t B + (u \cdot \nabla) B - \frac{1}{\sigma} \Delta B = (B \cdot \nabla) u + \frac{1}{\sigma c} \partial_t (\nabla \times E),$$

where we have also used that u and B are divergence-free. This holds in any dimension d = 2 or d = 3. Then, further assuming that (u, E, B) has the two-dimensional normal structure (1-2) yields

$$\partial_t b + u \cdot \nabla b - \frac{1}{\sigma} \Delta b = \frac{1}{\sigma c} \partial_t (\operatorname{curl} E).$$

Now, the elementary observation that

$$\int_{\mathbb{R}^2} (u \cdot \nabla b) b \, dx = \frac{1}{2} \int_{\mathbb{R}^2} u \cdot \nabla (b^2) \, dx = -\frac{1}{2} \int_{\mathbb{R}^2} (\operatorname{div} u) b^2 \, dx = 0,$$

which is a consequence of the incompressibility of u, allows us to deduce the parabolic energy estimate

$$\frac{1}{2}\frac{d}{dt}\int_{\mathbb{R}^2} b^2 dx + \frac{1}{\sigma}\int_{\mathbb{R}^2} |\nabla b|^2 dx = \frac{1}{\sigma c}\frac{d}{dt}\int_{\mathbb{R}^2} (\operatorname{curl} E)b \, dx + \frac{1}{\sigma}\int_{\mathbb{R}^2} (\operatorname{curl} E)^2 \, dx, \qquad (3-27)$$

where we used Faraday's equation again to substitute $\partial_t b = -c \operatorname{curl} E$.

Then, integrating in time, we infer that

$$\begin{split} &\frac{1}{2} \|b(T)\|_{L_x^2}^2 + \frac{1}{\sigma} \|\nabla b\|_{L_t^2([0,T);L_x^2)}^2 \\ &\leq \frac{1}{2} \|b(0)\|_{L_x^2}^2 + \frac{1}{\sigma c} \int_{\mathbb{R}^2} \operatorname{curl} E(T)b(T) \, dx - \frac{1}{\sigma c} \int_{\mathbb{R}^2} \operatorname{curl} E(0)b(0) \, dx + \frac{1}{\sigma} \|\operatorname{curl} E\|_{L_t^2([0,T);L_x^2)}^2 \\ &\leq \frac{1}{2} \|b(0)\|_{L_x^2}^2 + \frac{1}{4} (\|b(0)\|_{L_x^2}^2 + \|b(T)\|_{L_x^2}^2) + \frac{1}{\sigma^2 c^2} (\|E(0)\|_{\dot{H}_x^1}^2 + \|E(T)\|_{\dot{H}_x^1}^2) + \frac{1}{\sigma} \|E\|_{L_t^2([0,T);\dot{H}_x^1)}^2, \end{split}$$

which leads to

$$\frac{1}{4} \|B(T)\|_{L_{x}^{2}}^{2} + \frac{1}{\sigma} \|\nabla B\|_{L_{t}^{2}([0,T);L_{x}^{2})}^{2} \\
\leq \frac{3}{4} \|B_{0}\|_{L_{x}^{2}}^{2} + \frac{1}{\sigma^{2}c^{2}} \|E_{0}\|_{\dot{H}_{x}^{1}}^{2} + \frac{1}{\sigma^{2}c^{2}} (\|E(T)\|_{\dot{H}_{x}^{1}}^{2} + \sigma\|cE\|_{L_{t}^{2}([0,T);\dot{H}_{x}^{1})}^{2}). \quad (3-28)$$

Finally, combining (3-28) with the energy estimates (3-24) shows that (3-26) holds, thereby reaching the conclusion of the proof.

Remark. The identity (3-27) contains the crucial calculation which enables us to extract a uniform bound on *B* in $L_t^2 \dot{H}_x^1$, with remainder terms of order c^{-1} . This estimate will play an important role in the control of the low frequencies of nonlinear source terms in the Euler–Maxwell system.

3.8. *Proof of Theorem 3.1.* We proceed to the proof of our first main result — Theorem 3.1. Recall that we are taking the liberty of assuming, for simplicity, to be dealing with a smooth solution (u, E, B) of (1-1) for some smooth initial data (u_0, E_0, B_0) , and that the justification of the theorem is only fully completed by relying on the approximation procedure laid out in Section 3.2.

Our proof hinges upon the preliminary lemmas established in Sections 3.4–3.7. Accordingly, we begin by carefully gathering all the relevant estimates on an arbitrary time interval [0, T). We will then move on to construct the energy functional which will produce the uniform bounds we are seeking.

Control of velocity field. The control of *u* is obtained from Lemmas 3.6 and 3.7. Recalling the equivalence $\|\nabla u\|_{L^a_x} \sim \|\omega\|_{L^a_x}$ for any value $1 < a < \infty$, these lemmas provide the estimates

$$\begin{aligned} \|u\|_{L_{t}^{\infty}\dot{H}_{x}^{1}} &\lesssim \|u_{0}\|_{\dot{H}_{x}^{1}} + \|j\|_{L_{t,x}^{2}} \|\nabla B\|_{L_{t}^{2}L_{x}^{\infty}}, \\ \|u\|_{L_{t}^{\infty}\dot{W}_{x}^{1,p}} &\lesssim \|u_{0}\|_{\dot{W}_{x}^{1,p}} + \|j\|_{L_{t,x}^{2}}^{2/p} \|j\|_{L_{t}^{2}\dot{H}_{x}^{1}}^{1-2/p} \|\nabla B\|_{L_{t}^{2}L_{x}^{\infty}}, \\ \|u\|_{L_{t,x}^{\infty}} &\lesssim \|u\|_{L_{t}^{\infty}L_{x}^{2}}^{(p-2)/(2(p-1))} \|u\|_{L_{t}^{\infty}\dot{W}_{x}^{1,p}}^{p/(2(p-1))} \\ &\lesssim \|u\|_{L_{t,x}^{\infty}L_{x}^{2}}^{p-2/(2(p-1))} (\|u_{0}\|_{\dot{W}_{x}^{1,p}} + \|j\|_{L_{t,x}^{2}}^{2/p} \|j\|_{L_{t}^{2}\dot{H}_{x}^{1}}^{1-2/p} \|\nabla B\|_{L_{t}^{2}L_{x}^{\infty}})^{p/(2(p-1))}, \end{aligned}$$
(3-29)

where 2 is a fixed value.

Control of high electromagnetic frequencies. The control of high frequencies of (E, B) is obtained from Lemma 3.8. Specifically, setting $s = \frac{7}{4}$ in (3-19) allows us to deduce that

$$c^{-\frac{3}{4}} \| (E,B) \|_{\tilde{L}^{\infty}_{t} \dot{B}^{7/4}_{2,1,>}} + c^{\frac{1}{4}} \| (E,B) \|_{\tilde{L}^{2}_{t} \dot{B}^{7/4}_{2,1,>}} + \| (E,B) \|_{\tilde{L}^{2}_{t} \dot{B}^{1}_{\infty,1,>}}$$

$$\lesssim c^{-\frac{3}{4}} \| (E_{0},B_{0}) \|_{\dot{B}^{7/4}_{2,1,>}} + c^{-\frac{3}{4}} \| u \|_{L^{\infty}_{t,x} \cap L^{\infty}_{t} \dot{H}^{1}_{x}} \| B \|_{\tilde{L}^{2}_{t} \dot{B}^{7/4}_{2,1,>}}$$

Then, further decomposing high and low frequencies in the last term above, we obtain

$$\begin{split} \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{7/4}_{2,1}} &\lesssim \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{7/4}_{2,1,<}} + \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{7/4}_{2,1,>}} \lesssim \|B\|^{1/4}_{\tilde{L}^{2}_{t}\dot{B}^{1}_{2,\infty,<}} \|B\|^{3/4}_{\tilde{L}^{2}_{t}\dot{B}^{2}_{2,\infty,<}} + \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{7/4}_{2,1,>}} \\ &\lesssim \|B\|^{1/4}_{L^{2}_{t}\dot{H}^{1}_{x}} \|B\|^{3/4}_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} + \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{7/4}_{2,1,>}}, \end{split}$$

which yields the estimate

$$c^{-\frac{3}{4}} \| (E,B) \|_{\tilde{L}_{t}^{\infty} \dot{B}_{2,1,>}^{7/4}} + c^{\frac{1}{4}} \| (E,B) \|_{\tilde{L}_{t}^{2} \dot{B}_{2,1,>}^{7/4}} + \| (E,B) \|_{\tilde{L}_{t}^{2} \dot{B}_{\infty,1,>}^{1}} \\ \lesssim c^{-\frac{3}{4}} \| (E_{0},B_{0}) \|_{\dot{B}_{2,1,>}^{7/4}} + c^{-\frac{3}{4}} \| u \|_{L_{t,x}^{\infty} \cap L_{t}^{\infty} \dot{H}_{x}^{1}} (\| B \|_{L_{t}^{2} \dot{H}_{x}^{1}}^{1/4} \| B \|_{L_{t}^{2} \dot{B}_{2,1,<}}^{3/4} + \| B \|_{\tilde{L}_{t}^{2} \dot{B}_{2,1,>}^{7/4}}).$$
(3-30)

Observe that the choice of regularity parameter $s = \frac{7}{4}$ is dictated by the need to control ∇B in $L_t^2 L_x^\infty$, as anticipated in the strategy presented in Section 1.3. Further notice that it is therefore crucial to be able to set a regularity parameter *s* with a value greater than one in (3-19). This flexibility comes from the use of the normal structure (1-2), which we exploited in the product estimates established in Section 3.3.

Control of low electromagnetic frequencies. The control of low frequencies of (E, B) is deduced from Lemmas 3.9 and 3.10. Here also, the choice of parameters is dictated by the need to control ∇B in $L_t^2 L_x^\infty$. Thus, since $\dot{B}_{2,1}^2(\mathbb{R}^2)$ is contained in $L^\infty(\mathbb{R}^2)$, one could, for instance, set s = 2 in (3-23), which gives

$$\|B\|_{L^2_t \dot{B}^2_{2,1,<}} \lesssim c^{-1} \|E_0\|_{\dot{B}^2_{2,2,<}} + \|B_0\|_{\dot{B}^1_{2,2,<}} + \|u\|_{L^\infty_t L^2_x} \|B\|_{L^2_t \dot{B}^2_{2,1}}.$$

But, due to the coefficient $||u||_{L_t^{\infty}L_x^2}$, such an estimate would eventually lead to a smallness condition, uniform in *c*, on the initial energy \mathcal{E}_0 , which is not desirable. Therefore, instead, we employ (3-20)

and (3-22). More precisely, by setting s = 1 in (3-22) and $s = 2 - \alpha$ in (3-20), for any choice $0 < \alpha < 1$, we obtain the respective estimates

$$\begin{split} \|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} &\lesssim c^{-1} \|E_{0}\|_{\dot{B}^{2}_{2,2,<}} + \|B_{0}\|_{\dot{B}^{1}_{2,2,<}} + \|u\|_{L^{\infty}_{t,x}\cap L^{\infty}_{t}\dot{H}^{1}_{x}} \|B\|_{L^{2}_{t}\dot{B}^{1}_{2,1}}, \\ \|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} &\lesssim c^{-1} \|E_{0}\|_{\dot{B}^{2}_{2,2,<}} + \|B_{0}\|_{\dot{B}^{1}_{2,2,<}} + \|u\|_{L^{\infty}_{t}L^{2}_{x}} \|u\|_{L^{\infty}_{t}\dot{H}^{1}_{x}}^{\alpha} \|B\|_{L^{2}_{t}\dot{B}^{1}_{2,\infty}}^{\alpha} \|B\|_{L^{2}_{t}\dot{B}^{2}_{2,\infty}}^{1-\alpha}. \end{split}$$

In fact, it is possible to straightforwardly adapt the proofs of the above estimates to derive the more useful combined control, where we split the high and low frequencies of B in the right-hand side,

$$\begin{split} \|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} &\lesssim c^{-1} \|E_{0}\|_{\dot{B}^{2}_{2,2,<}} + \|B_{0}\|_{\dot{B}^{1}_{2,2,<}} + \|u\|_{L^{\infty}_{t,x}\cap L^{\infty}_{t}\dot{H}^{1}_{x}} \|\mathbb{1}_{\{|D| \geq \frac{1}{2}\sigma c\}} B\|_{L^{2}_{t}\dot{B}^{1}_{2,1}} \\ &+ \|u\|_{L^{\infty}_{t}L^{2}_{x}}^{1-\alpha} \|u\|_{L^{\infty}_{t}\dot{H}^{1}_{x}}^{\alpha} \|\mathbb{1}_{\{|D| < \frac{1}{2}\sigma c\}} B\|_{L^{2}_{t}\dot{B}^{1}_{2,\infty}}^{\alpha} \|\mathbb{1}_{\{|D| < \frac{1}{2}\sigma c\}} B\|_{L^{2}_{t}\dot{B}^{2}_{2,\infty}}^{1-\alpha} \\ &\lesssim \|(E_{0}, B_{0})\|_{\dot{H}^{1}_{x}} + \|u\|_{L^{\infty}_{t,x}\cap L^{\infty}_{t}\dot{H}^{1}_{x}} (c^{-1}\|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} + c^{-\frac{3}{4}}\|B\|_{\tilde{L}^{2}_{t}\dot{B}^{7/4}_{2,1,>}}) \\ &+ \|u\|_{L^{\infty}_{t}L^{2}_{t}\dot{L}^{2}_{x}}^{1-\alpha} \|u\|_{L^{\infty}_{t}\dot{H}^{1}_{x}}^{\alpha} \|B\|_{L^{2}_{t}\dot{H}^{1}_{x}}^{\alpha} \|B\|_{L^{2}_{t}\dot{H}^{1}_{x}}^{1-\alpha}. \end{split}$$
(3-31)

Further combining the preceding estimate with the energy estimate (3-24) from Lemma 3.10, we obtain

$$\begin{split} \|(E,B)\|_{L^{\infty}_{t}\dot{H}^{1}_{x}} + c\|E\|_{L^{2}_{t}\dot{H}^{1}_{x}} + \|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} \\ \lesssim \|(E_{0},B_{0})\|_{\dot{H}^{1}_{x}} + \|u\|_{L^{\infty}_{t,x}\cap L^{\infty}_{t}\dot{H}^{1}_{x}}(c^{-1}\|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} + c^{-\frac{3}{4}}\|B\|_{\tilde{L}^{2}_{t}\dot{B}^{7/4}_{2,1,>}}) \\ &+ \|u\|_{L^{\infty}_{t}L^{2}_{x}}^{1-\alpha}\|u\|^{\alpha}_{L^{\infty}_{t}\dot{H}^{1}_{x}}\|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}}^{\alpha}. \end{split}$$

Finally, by a classical use of Young's inequality $ab \le \alpha a^{1/\alpha} + (1-\alpha)b^{1/(1-\alpha)}$, with $a, b \ge 0$, aimed at absorbing the term $\|B\|_{L^2_t \dot{B}^2_{2,1,<}}^{1-\alpha}$ with the above left-hand side, we conclude that

$$\begin{split} \|(E,B)\|_{L_{t}^{\infty}\dot{H}_{x}^{1}} + c\|E\|_{L_{t}^{2}\dot{H}_{x}^{1}} + \|B\|_{L_{t}^{2}\dot{B}_{2,1,<}^{2}} \\ \lesssim \|(E_{0},B_{0})\|_{\dot{H}_{x}^{1}} + \|u\|_{L_{t,x}^{\infty}\cap L_{t}^{\infty}\dot{H}_{x}^{1}}(c^{-1}\|B\|_{L_{t}^{2}\dot{B}_{2,1,<}^{2}} + c^{-\frac{3}{4}}\|B\|_{\tilde{L}_{t}^{2}\dot{B}_{2,1,>}^{7/4}}) \\ &+ \|u\|_{L_{t}^{\infty}L_{x}^{2}}^{1/\alpha-1}\|u\|_{L_{t}^{\infty}\dot{H}_{x}^{1}}\|B\|_{L_{t}^{2}\dot{H}_{x}^{1}}. \quad (3-32) \end{split}$$

Parabolic stability of magnetic field. The parabolic stability of the magnetic field comes as a result of the almost-parabolic estimates established in Lemma 3.11, which we conveniently reproduce here:

$$\|B\|_{L^{2}_{t}\dot{H}^{1}_{x}} \lesssim \|B_{0}\|_{L^{2}_{x}} + c^{-1}\|(E_{0}, B_{0})\|_{\dot{H}^{1}_{x}} + c^{-1}\|u\|_{L^{\infty}_{t}L^{2}_{x}}\|\nabla B\|_{L^{2}_{t}L^{\infty}_{x}}.$$
(3-33)

This estimate will serve to control the term $||B||_{L^2_{\tau}\dot{H}^1_{\tau}}$ in (3-32).

Ohm's law estimate. Finally, we need to employ Ohm's law from (1-1) to control the electric current j. More precisely, by Ohm's law and the normal structure (1-2), we have

$$\|j\|_{L^{2}_{t}\dot{H}^{1}_{x}} \lesssim c\|E\|_{L^{2}_{t}\dot{H}^{1}_{x}} + \|P(u \times B)\|_{L^{2}_{t}\dot{H}^{1}_{x}} \lesssim c\|E\|_{L^{2}_{t}\dot{H}^{1}_{x}} + \|u\|_{L^{\infty}_{t}L^{2}_{x}}\|\nabla B\|_{L^{2}_{t}L^{\infty}_{x}},$$
(3-34)

which will be used to control $||j||_{L^2_t \dot{H}^1_x}$ in (3-29).

Nonlinear energy estimate. We are now in a position to derive the global nonlinear energy estimate which will yield a uniform bound on solutions to (1-1). Thus, inspired by the above set of estimates, we introduce the energy $\mathcal{H}(t_1, t_2)$, with $0 \le t_1 \le t_2$, by setting

$$\begin{aligned} \mathcal{H}(t_{1},t_{2}) &:= \|u\|_{L^{\infty}_{t}\dot{H}^{1}_{x}} + \mathcal{E}^{(p-2)/(2(p-1))}_{0} \|u\|_{L^{\infty}_{t}\dot{W}^{1,p}_{x}}^{p/(2(p-1))} + c^{-\frac{3}{4}} \|(E,B)\|_{\tilde{L}^{\infty}_{t}\dot{B}^{7/4}_{2,1,>}} + c^{\frac{1}{4}} \|(E,B)\|_{\tilde{L}^{2}_{t}\dot{B}^{7/4}_{2,1,>}} \\ &+ \|(E,B)\|_{\tilde{L}^{2}_{t}\dot{B}^{1}_{\infty,1,>}} + \|(E,B)\|_{L^{\infty}_{t}\dot{H}^{1}_{x}} + c\|E\|_{L^{2}_{t}\dot{H}^{1}_{x}} + \|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} \end{aligned}$$

where all time-norms are taken over the interval $[t_1, t_2)$. Since (u, E, B) is assumed, without any loss of generality, to be a smooth solution of (1-1), observe that $\mathcal{H}(t_1, t_2)$ is continuous on $\{0 \le t_1 \le t_2\}$. In particular, we can further define the continuous function

$$\mathcal{H}(t) := \mathcal{H}(t,t) = \|u(t)\|_{\dot{H}^{1}_{x}} + \mathcal{E}_{0}^{(p-2)/(2(p-1))} \|u(t)\|_{\dot{W}^{1,p}_{x}}^{p/(2(p-1))} + c^{-\frac{3}{4}} \|(E,B)(t)\|_{\dot{B}^{7/4}_{2,1,>}} + \|(E,B)(t)\|_{\dot{H}^{1}_{x}}$$

for every $t \ge 0$. Note that

$$\mathcal{H}(t) \leq \mathcal{H}(t_1, t_2)$$

for all $t \in [t_1, t_2]$.

It will also be useful to assign the notation

$$\mathcal{J}(t_1, t_2) := \| j \|_{L^2_t}$$

to the dissipation produced by the electric current in (1-3), where the L^2 -norm is taken over the timeinterval $[t_1, t_2)$, as well. In particular, one has the uniform bound

$$\mathcal{J}(t_1, t_2) \le \left(\frac{\sigma}{2}\right)^{\frac{1}{2}} \mathcal{E}_0 \tag{3-35}$$

by virtue of the energy inequality (1-3).

Observe now that all the above estimates, which were stated on the time-interval [0, T), could equally well be written over any other finite time-interval $[t_1, t_2)$, provided one replaces the initial data (u_0, E_0, B_0) by the data at time t_1 . Thus, employing the energy inequality (1-3), the energies $\mathcal{H}(t_1, t_2)$ and $\mathcal{H}(t)$, and the embedding

$$\|\nabla B\|_{L^{2}_{t}L^{\infty}_{x}} \leq \|B\|_{L^{2}_{t}\dot{B}^{1}_{\infty,1}} \lesssim \|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} + \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{1}_{\infty,1,>}} \leq \mathcal{H}(t_{1},t_{2}),$$

one can write the parabolic stability estimate (3-33) as

$$\|B\|_{L^{2}_{t}\dot{H}^{1}_{x}} \lesssim \mathcal{E}_{0} + c^{-1}\mathcal{H}(t_{1}) + c^{-1}\mathcal{E}_{0}\mathcal{H}(t_{1}, t_{2})$$
(3-36)

and the Ohm's law estimate (3-34) as

$$\|j\|_{L^{2}_{t}\dot{H}^{1}_{x}} \lesssim (1+\mathcal{E}_{0})\mathcal{H}(t_{1},t_{2}), \tag{3-37}$$

which are linear in $\mathcal{H}(t_1, t_2)$.

Then, incorporating the linear estimate (3-37) into the velocity control (3-29), we obtain

$$\|u\|_{L^{\infty}_{t}\dot{H}^{1}_{x}} \lesssim \mathcal{H}(t_{1}) + \mathcal{J}(t_{1},t_{2})\mathcal{H}(t_{1},t_{2}),$$

$$\|u\|_{L^{\infty}_{t,x}} \lesssim \mathcal{E}_{0}^{\frac{p-2}{2(p-1)}} \|u\|_{L^{\infty}_{t}\dot{W}^{1,p}_{x}} \lesssim \mathcal{H}(t_{1}) + \mathcal{E}_{0}^{\frac{p-2}{2(p-1)}} (1+\mathcal{E}_{0})^{\frac{p-2}{2(p-1)}} \mathcal{J}(t_{1},t_{2})^{\frac{1}{p-1}} \mathcal{H}(t_{1},t_{2}).$$
(3-38)

Furthermore, the use of (3-36) and (3-38) in the high-frequency control (3-30) leads to

$$c^{-\frac{3}{4}} \| (E,B) \|_{\tilde{L}_{t}^{\infty} \dot{B}_{2,1,>}^{7/4}} + c^{\frac{1}{4}} \| (E,B) \|_{\tilde{L}_{t}^{2} \dot{B}_{2,1,>}^{7/4}} + \| (E,B) \|_{\tilde{L}_{t}^{2} \dot{B}_{\infty,1,>}} \\ \lesssim \mathcal{H}(t_{1}) + c^{-\frac{3}{4}} \mathcal{H}(t_{1},t_{2}) (\mathcal{E}_{0} + c^{-1} \mathcal{H}(t_{1}) + c^{-1} \mathcal{E}_{0} \mathcal{H}(t_{1},t_{2}))^{\frac{1}{4}} \mathcal{H}(t_{1},t_{2})^{\frac{3}{4}} + c^{-1} \mathcal{H}(t_{1},t_{2})^{2} \\ \lesssim \mathcal{H}(t_{1}) + c^{-\frac{3}{4}} \mathcal{E}_{0}^{1/4} \mathcal{H}(t_{1},t_{2})^{\frac{7}{4}} + c^{-1} (1 + \mathcal{E}_{0}^{1/4}) \mathcal{H}(t_{1},t_{2})^{2} \\ \lesssim \mathcal{H}(t_{1}) + \lambda \mathcal{H}(t_{1},t_{2}) + \lambda^{-\frac{1}{3}} c^{-1} \mathcal{E}_{0}^{1/3} \mathcal{H}(t_{1},t_{2})^{2} + c^{-1} (1 + \mathcal{E}_{0}^{1/4}) \mathcal{H}(t_{1},t_{2})^{2}$$

$$(3-39)$$

for any
$$\lambda > 0$$
, whereas a similar procedure applied to the low-frequency estimate (3-32) yields

$$\|(E, B)\|_{L_{t}^{\infty}\dot{H}_{x}^{1}} + c \|E\|_{L_{t}^{2}\dot{H}_{x}^{1}} + \|B\|_{L_{t}^{2}\dot{B}_{2,1,<}^{2}}$$

$$\lesssim \mathcal{H}(t_{1}) + c^{-1}\mathcal{H}(t_{1}, t_{2})^{2} + \mathcal{E}_{0}^{1/\alpha - 1}(\mathcal{H}(t_{1}) + \mathcal{J}(t_{1}, t_{2})\mathcal{H}(t_{1}, t_{2}))(\mathcal{E}_{0} + c^{-1}\mathcal{H}(t_{1}) + c^{-1}\mathcal{E}_{0}\mathcal{H}(t_{1}, t_{2}))$$

$$\lesssim (1 + \mathcal{E}_{0}^{1/\alpha})\mathcal{H}(t_{1}) + \mathcal{E}_{0}^{1/\alpha}\mathcal{J}(t_{1}, t_{2})\mathcal{H}(t_{1}, t_{2}) + (1 + \mathcal{E}_{0}^{1/\alpha + 1})c^{-1}\mathcal{H}(t_{1}, t_{2})^{2}.$$
(3-40)

All in all, summing estimates (3-38), (3-39) and (3-40) together and setting λ in (3-39) small enough that the term $\lambda \mathcal{H}(t_1, t_2)$ can be absorbed by the resulting left-hand side, we finally arrive at the crucial nonlinear energy estimate

$$\begin{aligned} \mathcal{H}(t_1, t_2) &\lesssim (1 + \mathcal{E}_0^{1/\alpha}) \mathcal{H}(t_1) + (1 + \mathcal{E}_0^{1/\alpha}) \mathcal{J}(t_1, t_2) \mathcal{H}(t_1, t_2) \\ &+ \mathcal{E}_0^{(p-2)/(2(p-1))} (1 + \mathcal{E}_0^{(p-2)/(2(p-1))}) \mathcal{J}(t_1, t_2)^{\frac{1}{p-1}} \mathcal{H}(t_1, t_2) + (1 + \mathcal{E}_0^{1/\alpha+1}) c^{-1} \mathcal{H}(t_1, t_2)^2 \end{aligned}$$

for any $0 \le t_1 \le t_2$, which, using (3-35), can be slightly simplified to

$$\mathcal{H}(t_1, t_2) \leq C_* (1 + \mathcal{E}_0^{1/\alpha}) \mathcal{H}(t_1) + C_* (1 + \mathcal{E}_0^{1/\alpha + (p-2)/(p-1)}) \mathcal{J}(t_1, t_2)^{\frac{1}{p-1}} \mathcal{H}(t_1, t_2) + C_* (1 + \mathcal{E}_0^{1/\alpha + 1}) c^{-1} \mathcal{H}(t_1, t_2)^2, \quad (3-41)$$

where $C_* > 0$ only depends on fixed parameters (in particular, it is independent of time, the speed of light *c* and the initial data). We are now going to show that (3-41) leads to a global bound on $\mathcal{H}(t_1, t_2)$.

Conclusion of proof. Let us consider a partition of time

$$0 = t_0 < t_1 < \dots < t_n < t_{n+1} = \infty$$

such that, for every $i = 0, 1, \ldots, n-1$,

$$C_*(1+\mathcal{E}_0^{1/\alpha+(p-2)/(p-1)})\mathcal{J}(t_i,t_{i+1})^{\frac{1}{p-1}} = \frac{1}{2} \quad \text{and} \quad C_*(1+\mathcal{E}_0^{1/\alpha+(p-2)/(p-1)})\mathcal{J}(t_n,t_{n+1})^{\frac{1}{p-1}} \le \frac{1}{2}.$$

In particular, by virtue of (3-35), one has, for any $t \in [t_i, t_{i+1}]$ with i = 0, 1, ..., n, that

$$\frac{i}{(2C_*(1+\mathcal{E}_0^{1/\alpha+(p-2)/(p-1)}))^{2(p-1)}} = \sum_{k=0}^{i-1} \mathcal{J}(t_k, t_{k+1})^2 \le \mathcal{J}(t_0, t)^2 \le \frac{\sigma}{2}\mathcal{E}_0^2.$$
(3-42)

It then follows from (3-41) that

$$\mathcal{H}(t_i, t) \le 2C_*(1 + \mathcal{E}_0^{1/\alpha})\mathcal{H}(t_i) + 2C_*(1 + \mathcal{E}_0^{1/\alpha+1})c^{-1}\mathcal{H}(t_i, t)^2$$
(3-43)

for all i = 0, ..., n and $t \in [t_i, t_{i+1}]$.

Next, for fixed i, we introduce the quadratic polynomial

$$p(X) = 2C_*(1 + \mathcal{E}_0^{1/\alpha + 1})c^{-1}X^2 - X + 2C_*(1 + \mathcal{E}_0^{1/\alpha})\mathcal{H}(t_i),$$

whose roots

$$\lambda_{\pm} = \frac{1 \pm \sqrt{1 - 16C_*^2 (1 + \mathcal{E}_0^{1/\alpha + 1})(1 + \mathcal{E}_0^{1/\alpha})c^{-1}\mathcal{H}(t_i)}}{4C_* (1 + \mathcal{E}_0^{1/\alpha + 1})c^{-1}}$$

are real and distinct, provided

$$\mathcal{H}(t_i) < \frac{c}{16C_*^2(1 + \mathcal{E}_0^{1/\alpha + 1})(1 + \mathcal{E}_0^{1/\alpha})}.$$
(3-44)

Observe that (3-43) can be rewritten as

 $p(\mathcal{H}(t_i, t)) \ge 0$

for all $t \in [t_i, t_{i+1}]$. Therefore, by continuity of $\mathcal{H}(t_i, t)$ and assuming that (3-44) is satisfied, we deduce that $\mathcal{H}(t_i, t) \leq \lambda_-$ for all $t \in [t_i, t_{i+1}]$ if it is true for $t = t_i$, i.e., $\mathcal{H}(t_i) \leq \lambda_-$.

Now, it is readily seen that (3-44) implies $\mathcal{H}(t_i) \leq \lambda_-$ if we assume, without any loss of generality, that $2C_*(1 + \mathcal{E}_0^{1/\alpha}) \geq 1$. Thus, by continuity of $t \mapsto \mathcal{H}(t_i, t)$, we conclude that (3-44) is sufficient to deduce the bound

$$\mathcal{H}(t) \le \mathcal{H}(t_i, t) \le \lambda_- \le 4C_*(1 + \mathcal{E}_0^{1/\alpha})\mathcal{H}(t_i)$$

for all $t \in [t_i, t_{i+1}]$, where we used the elementary inequality $1 - \sqrt{1-z} \le z$ for all $z \in [0, 1]$ in the last step. Then, a straightforward iterative process leads us to the estimate

$$\mathcal{H}(t_i, t) \le 4C_*(1 + \mathcal{E}_0^{1/\alpha})\mathcal{H}(t_i) \le [4C_*(1 + \mathcal{E}_0^{1/\alpha})]^{i+1}\mathcal{H}(t_0)$$

for each i = 0, 1, ..., n and all $t \in [t_i, t_{i+1}]$, if the initial data satisfies

$$\mathcal{H}(t_0) < \frac{c}{4C_*(1 + \mathcal{E}_0^{1/\alpha + 1})[4C_*(1 + \mathcal{E}_0^{1/\alpha})]^{i+1}}$$

Further noticing, for any $t \in [t_i, t_{i+1}]$, that

$$\mathcal{H}(t_0, t) \le \mathcal{H}(t_i, t) + \sum_{k=0}^{l-1} \mathcal{H}(t_k, t_{k+1})$$

we conclude, in view of (3-42), that one has the global bound

$$\mathcal{H}(0,t) \leq \sum_{k=0}^{i} [4C_{*}(1+\mathcal{E}_{0}^{1/\alpha})]^{k+1} \mathcal{H}(0) = 4C_{*}(1+\mathcal{E}_{0}^{1/\alpha}) \frac{[4C_{*}(1+\mathcal{E}_{0}^{1/\alpha})]^{i+1}-1}{4C_{*}(1+\mathcal{E}_{0}^{1/\alpha})-1} \mathcal{H}(0)$$

$$\leq 2[4C_{*}(1+\mathcal{E}_{0}^{1/\alpha})]^{i+1} \mathcal{H}(0) \leq 2[4C_{*}(1+\mathcal{E}_{0}^{1/\alpha})]^{1+[2C_{*}(1+\mathcal{E}_{0}^{1/\alpha+(p-2)/(p-1)})]^{2(p-1)}} \mathcal{J}(0,t)^{2} \mathcal{H}(0).$$

provided

$$\mathcal{H}(0) < \frac{c}{4C_*(1 + \mathcal{E}_0^{1/\alpha + 1})[4C_*(1 + \mathcal{E}_0^{1/\alpha})]^{1 + [2C_*(1 + \mathcal{E}_0^{1/\alpha + (p-2)/(p-1)})]^{2(p-1)}(\sigma/2)\mathcal{E}_0^2}} \le \frac{c}{4C_*(1 + \mathcal{E}_0^{1/\alpha + 1})[4C_*(1 + \mathcal{E}_0^{1/\alpha})]^{1 + n}}$$
tially

holds initially.

Summarizing the preceding developments, we have now established the existence of an independent constant $C_* > 0$ such that, if the smooth solution (u, E, B) has initial data satisfying

$$4C_*(1+\mathcal{E}_0^{1/\alpha+1})[4C_*(1+\mathcal{E}_0^{1/\alpha})]^{1+[2C_*(1+\mathcal{E}_0^{1/\alpha+(p-2)/(p-1)})]^{2(p-1)}(\sigma/2)\mathcal{E}_0^2}\mathcal{H}(0) < c, \qquad (3-45)$$

then the bound

$$\mathcal{H}(0,t) \le 2[4C_*(1+\mathcal{E}_0^{1/\alpha})]^{1+[2C_*(1+\mathcal{E}_0^{1/\alpha+(p-2)/(p-1)})]^{2(p-1)}\mathcal{J}(0,t)^2}\mathcal{H}(0)$$
(3-46)

holds globally for any $t \in [0, \infty)$. In view of the approximation procedure laid out in Section 3.2, this uniform control allows us to complete the construction of solutions claimed in the statement of Theorem 3.1.

Moreover, observe that all global bounds on (u, E, B) stated in (3-2) are a direct consequence of (3-36) and (3-46).

Finally, in order to deduce the simpler initial condition (3-1) from (3-45), there only remains to notice, for any given $\varepsilon > 0$, by taking $0 < \alpha < 1$ and 2 sufficiently close to the values 1 and 2, respectively, that

$$4C_*(1+\mathcal{E}_0^{1/\alpha+1})[4C_*(1+\mathcal{E}_0^{1/\alpha})]^{1+[2C_*(1+\mathcal{E}_0^{1/\alpha+(p-2)/(p-1)})]^{2(p-1)}(\sigma/2)\mathcal{E}_0^2} \le C_{**}e^{C_{**}\mathcal{E}_0^{4+\varepsilon}}$$

for some large independent constant $C_{**} > 0$. Then, since the global bound (3-46) holds for that particular choice of p close to 2, one can use the ensuing uniform controls on the solution (u, E, B) in combination with (3-29) to propagate the L_x^p -norm of the vorticity ω for higher values 2 , which completes the proof of Theorem 3.1.

3.9. *Proof of Theorem 3.2.* The proof of Theorem 3.2 is a continuation of that of Theorem 3.1. Thus, assuming that the solution (u, E, B) produced by Theorem 3.1 is already constructed, we observe that Lemma 3.6 provides us with the additional bound

$$\begin{aligned} \|\omega(t)\|_{L^{\infty}_{x}} &\lesssim \|\omega(0)\|_{L^{\infty}_{x}} + \|j\|_{L^{2}([0,t);L^{\infty}_{x})} \|\nabla B\|_{L^{2}([0,t);L^{\infty}_{x})} \\ &\lesssim \|\omega(0)\|_{L^{\infty}_{x}} + (\|j\|_{L^{2}([0,t);\dot{B}^{0}_{\infty,1,<})} + \|j\|_{L^{2}([0,t);\dot{B}^{0}_{\infty,1,>})}) \|\nabla B\|_{L^{2}([0,t);L^{\infty}_{x})} \end{aligned}$$

for any $t \ge 0$, which, when combined with Ohm's law, the two-dimensional embedding $\dot{B}_{2,1}^1 \subset \dot{B}_{\infty,1}^0$ and the paradifferential product law (3-14), further yields

$$\begin{split} \|\omega(t)\|_{L^{\infty}_{x}} &\lesssim \|\omega(0)\|_{L^{\infty}_{x}} + (\|j\|_{L^{2}_{t}\dot{B}^{1}_{2,1,<}} + c\|E\|_{L^{2}_{t}\dot{B}^{0}_{\infty,1,>}} + \|P(u \times B)\|_{L^{2}_{t}\dot{B}^{1}_{2,1}})\|\nabla B\|_{L^{2}_{t}L^{\infty}_{x}} \\ &\lesssim \|\omega(0)\|_{L^{\infty}_{x}} + (c\|j\|_{L^{2}_{t}\dot{B}^{0}_{2,\infty,<}} + \|E\|_{L^{2}_{t}\dot{B}^{1}_{\infty,1,>}} \\ &+ \|u\|_{L^{\infty}_{t}L^{2}_{x}}\|B\|_{L^{2}_{t}\dot{B}^{1}_{\infty,1}} + \|u\|_{L^{\infty}_{t}\dot{H}^{1}_{x}}\|B\|_{L^{2}_{t}\dot{H}^{1}_{x}})\|\nabla B\|_{L^{2}_{t}L^{\infty}_{x}}. \end{split}$$

Now, recall from (3-3) that ∇B belongs to $L^2(\mathbb{R}^+; \dot{B}^0_{\infty,1}) \subset L^2(\mathbb{R}^+; L^\infty_x)$. Therefore, by virtue of the energy inequality (1-3), the global bounds (3-2) and the assumption that the initial vorticity belongs to L^∞_x , we conclude the pointwise boundedness of $\omega(t)$ for all times.

Observe, though, that the ensuing bound $\omega \in L^{\infty}_{t,x}$ is global in time, but it is not uniform in *c*. Nevertheless, it is possible to derive another global bound on the vorticity in $L^{\infty}_{t,x}$, uniformly in *c*, by employing Ohm's law to control *j* in $L^2_t \dot{B}^1_{2,1,<}$ and requiring the additional initial assumption that $E_0 \in \dot{B}^1_{2,1}$.

Specifically, the use of Ohm's law to expand the low frequencies of j leads to the necessity of controlling cE uniformly in the space $L_t^2 \dot{B}_{2,1,<}^1$, which can be achieved by relying on the low-frequency estimates from Lemma 3.9. More precisely, by combining (3-21), with s = 1, and (3-20), with $s = 2 - \alpha$ and $0 < \alpha < 1$, it is possible to establish, by repeating the steps leading to (3-31), that

$$c \|E\|_{L^{2}_{t}\dot{B}^{1}_{2,1,<}} \lesssim \|E_{0}\|_{\dot{B}^{1}_{2,1}} + \|B_{0}\|_{\dot{H}^{1}_{x}} + \|u\|_{L^{\infty}_{t,x}\cap L^{\infty}_{t}\dot{H}^{1}_{x}} (c^{-1}\|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}} + c^{-\frac{3}{4}}\|B\|_{\tilde{L}^{2}_{t}\dot{B}^{7/4}_{2,1,>}}) \\ + \|u\|_{L^{\infty}_{t}L^{2}_{x}} \|u\|_{L^{\infty}_{t}\dot{H}^{1}_{x}}^{\alpha} \|B\|_{L^{2}_{t}\dot{H}^{1}_{x}}^{\alpha} \|B\|_{L^{2}_{t}\dot{H}^{2}_{x}}^{1/4})$$

All terms in the right-hand side of this estimate are now uniformly controlled (in c) by the bounds (3-2), provided one further assumes that the initial data E_0 belongs to $\dot{B}_{2,1}^1$. This concludes the justification of the bound $\omega \in L_{t,x}^{\infty}$, uniformly in c.

We turn now to the uniqueness of solutions to (1-1), which rests upon Yudovich's fundamental ideas. To that end, suppose that

$$(u_i, E_i, B_i) \in L^{\infty}([0, T); L^2_x),$$

with i = 1, 2, are two weak solutions to the two-dimensional incompressible Euler–Maxwell system (1-1) for the same initial data and for some existence time T > 0. We are going to establish a weak-strong uniqueness principle by requiring a control on the solution (u_2, E_2, B_2) which is stronger than the one on (u_1, E_1, B_1) .

In a natural way, we denote the vorticities and electric currents associated to each solution by ω_i and j_i , respectively. Furthermore, we assume that each solution satisfies its corresponding energy inequality (1-3) and that

$$u_1 \in L^2([0,T); L^\infty_x), \quad \omega_2 \in \bigcap_{2 \le q \le \infty} L^{q'}([0,T); L^q_x), \quad j_2 \in L^1([0,T); L^\infty_x).$$

By virtue of the Gagliardo–Nirenberg inequality (3-17), recall that the above bounds are sufficient to imply that

$$u_2 \in L^2([0,T); L^\infty_x)$$

as well. Note that we are not requiring here that the solutions have the normal structure (1-2).

Next, a straightforward duality argument on (1-1), similar to the computation which gives the energy inequality (1-3), leads to

$$\begin{aligned} \frac{d}{dt} \int_{\mathbb{R}^2} (u_1 \cdot u_2 + E_1 \cdot E_2 + B_1 \cdot B_2) \, dx &+ \frac{2}{\sigma} \int_{\mathbb{R}^2} j_1 \cdot j_2 \, dx \\ &= -\int_{\mathbb{R}^2} (u_2 \otimes (u_1 - u_2)) : \nabla(u_1 - u_2) \, dx \\ &+ \int_{\mathbb{R}^2} (((j_1 - j_2) \times (B_1 - B_2)) \cdot u_2 - (j_2 \times (B_1 - B_2)) \cdot (u_1 - u_2)) \, dx. \end{aligned}$$

It is to be emphasized that the solutions considered here have sufficient regularity and integrability to justify the above computation rigorously.

We introduce now the modulated energy

$$F_{\varepsilon}(t) := \frac{1}{2} (\|\tilde{u}(t)\|_{L^2}^2 + \|\tilde{E}(t)\|_{L^2}^2 + \|\tilde{B}(t)\|_{L^2}^2) + \varepsilon,$$

where

$$\tilde{u} := u_1 - u_2, \quad \tilde{E} := E_1 - E_2, \quad \tilde{B} := B_1 - B_2,$$

and $\varepsilon > 0$ merely ensures the positivity of F_{ε} .

Then, integrating the preceding identity in time and combining the result with the energy inequality (1-3) for each solution, we deduce the estimate

$$\begin{split} F_{\varepsilon}(t) &+ \frac{1}{\sigma} \int_{0}^{t} \|\tilde{j}(\tau)\|_{L_{x}^{2}}^{2} d\tau \\ &\leq \varepsilon - \int_{0}^{t} \int_{\mathbb{R}^{2}} [((\tilde{u} \cdot \nabla)u_{2}) \cdot \tilde{u} + (\tilde{j} \times \tilde{B}) \cdot u_{2} - (j_{2} \times \tilde{B}) \cdot \tilde{u}](\tau) \, dx \, d\tau \\ &\leq \varepsilon + \int_{0}^{t} \|\nabla u_{2}\|_{L_{x}^{q}} \|\tilde{u}\|_{L_{x}^{\infty}}^{2/q} \|\tilde{u}\|_{L_{x}^{2}}^{2/q'} \, d\tau + \int_{0}^{t} \|\tilde{j}\|_{L_{x}^{2}} [\|\tilde{B}\|_{L_{x}^{2}} \|u_{2}\|_{L_{x}^{\infty}} + \|j_{2}\|_{L_{x}^{\infty}} \|\tilde{B}\|_{L_{x}^{2}} \|\tilde{u}\|_{L_{x}^{2}}^{2}] \, d\tau \\ &\leq \varepsilon + \int_{0}^{t} \|\nabla u_{2}\|_{L_{x}^{q}} \|\tilde{u}\|_{L_{x}^{\infty}}^{2/q} \|\tilde{u}\|_{L_{x}^{2}}^{2/q'} \, d\tau + \frac{1}{2\sigma} \int_{0}^{t} \|\tilde{j}(\tau)\|_{L_{x}^{2}}^{2} \, d\tau \\ &\quad + \int_{0}^{t} \Big[\frac{\sigma}{2} \|\tilde{B}\|_{L_{x}^{2}}^{2} \|u_{2}\|_{L_{x}^{\infty}}^{2} + \|j_{2}\|_{L_{x}^{\infty}} \|\tilde{B}\|_{L_{x}^{2}} \|\tilde{u}\|_{L_{x}^{2}}^{2} \Big] \, d\tau \end{split}$$

for all $t \in [0, T)$ and any $2 \le q < \infty$, where we have defined $\tilde{j} := j_1 - j_2$.

The next step relies on a classical sharp estimate on the Biot–Savart law (1-14). More precisely, by exploiting that the map $\omega \mapsto \nabla(-\Delta^{-1}\nabla \times \omega) = \nabla u$ produces a Calderón–Zygmund singular integral operator, it is possible to show that

$$\|\nabla u\|_{L^{a}(\mathbb{R}^{2})} \leq C_{\mathrm{BS}}\frac{a^{2}}{a-1}\|\nabla \times u\|_{L^{a}(\mathbb{R}^{2})}$$

for all $1 < a < \infty$ and any divergence-free vector field u, where $C_{BS} > 0$ is independent of a. We refer to [Bahouri et al. 2011, Section 7.1.1] for more details concerning the Biot–Savart law and to [Grafakos 2014, Section 6.2.3] for a Fourier multiplier theorem which can be used to obtain the correct dependence of the above Biot–Savart estimate in the parameter a.

Thus, we deduce from the previous bound that

$$F_{\varepsilon}(t) \leq \varepsilon + \int_{0}^{t} 2C_{\mathrm{BS}}q \|\omega_{2}\|_{L_{x}^{q}} \|\tilde{u}\|_{L_{x}^{\infty}}^{2/q} F_{\varepsilon}(\tau)^{\frac{1}{q'}} + [\sigma\|u_{2}\|_{L_{x}^{\infty}}^{2} + \|j_{2}\|_{L_{x}^{\infty}}]F_{\varepsilon}(\tau) d\tau.$$

Moreover, denoting the right-hand side of the above inequality by $G_{\varepsilon}(t)$ and observing that it is absolutely continuous on [0, T), we obtain

$$\frac{d}{dt}G_{\varepsilon}(t) \le 2C_{\rm BS}q \|\omega_2\|_{L^q_x} \|\tilde{u}\|_{L^{\infty}_x}^{2/q} G_{\varepsilon}(t)^{\frac{1}{q'}} + [\sigma \|u_2\|_{L^{\infty}_x}^2 + \|j_2\|_{L^{\infty}_x}]G_{\varepsilon}(t)$$

for almost every $t \in [0, T)$. Therefore, further introducing the absolutely continuous functional

$$\Theta_{\varepsilon}(t) := G_{\varepsilon}(t)e^{-\int_0^t [\sigma \|u_2\|_{L_x}^2 + \|j_2\|_{L_x}^\infty](\tau)\,d\tau},$$

we see that

whence

$$\frac{d}{dt}\Theta_{\varepsilon}(t) \leq 2C_{\mathrm{BS}}q \|\omega_{2}\|_{L_{x}^{q}} \|\tilde{u}\|_{L_{x}^{\infty}}^{2/q}\Theta_{\varepsilon}(t)^{\frac{1}{q'}}e^{-\frac{1}{q}\int_{0}^{t}[\sigma\|u_{2}\|_{L_{x}^{\infty}}^{2}+\|j_{2}\|_{L_{x}^{\infty}}](\tau)\,d\tau}$$
$$\frac{d}{dt}\Theta_{\varepsilon}(t)^{\frac{1}{q}} \leq 2C_{\mathrm{BS}}\|\omega_{2}\|_{L_{x}^{q}}\|\tilde{u}\|_{L_{x}^{\infty}}^{2/q}e^{-\frac{1}{q}\int_{0}^{t}[\sigma\|u_{2}\|_{L_{x}^{\infty}}^{2}+\|j_{2}\|_{L_{x}^{\infty}}](\tau)\,d\tau}.$$

Observe that all technical difficulties incurred when dividing by $\Theta_{\varepsilon}(t)^{1/q'}$ in the last step are removed by the use of $\varepsilon > 0$.

Now, integrating in time leads to

$$F_{\varepsilon}(t)^{\frac{1}{q}} \leq G_{\varepsilon}(t)^{\frac{1}{q}} \leq \varepsilon e^{\frac{1}{q} \int_{0}^{t} [\sigma \|u_{2}\|_{L_{x}^{\infty}}^{2} + \|j_{2}\|_{L_{x}^{\infty}}](\tau) d\tau} + 2C_{BS} \int_{0}^{t} \|\omega_{2}(\tau)\|_{L_{x}^{q}}^{2/q} \|\tilde{u}(\tau)\|_{L_{x}^{\infty}}^{2/q} e^{\frac{1}{q} \int_{\tau}^{t} [\sigma \|u_{2}\|_{L_{x}^{\infty}}^{2} + \|j_{2}\|_{L_{x}^{\infty}}](s) ds} d\tau,$$

whereby, letting $\varepsilon \to 0$, we end up with

Finally, considering values
$$t \in [0, T)$$
 such that

$$\|\omega_2\|_{L^1([0,t);L^\infty_x)} < \frac{1}{2C_{\rm BS}}$$

and then letting q tend to infinity, we conclude that $(\tilde{u}, \tilde{E}, \tilde{B})(t) = 0$, thereby establishing the uniqueness of solutions on a time interval [0, t) for some $t \in (0, T)$.

Now, observe that this argument can be reproduced on any time interval $[t_0, T)$, with $t_0 > 0$ and such that $(\tilde{u}, \tilde{E}, \tilde{B})(t_0) = 0$, to prove the uniqueness of solutions on $[t_0, t)$ for some $t \in (t_0, T)$. In other words, we have shown that the set

$$S = \left\{ t \in [0, T) : \int_0^t \| (\tilde{u}, \tilde{E}, \tilde{B})(s) \|_{L^2_x} \, ds = 0 \right\}$$

is open. Since the function $t \mapsto \int_0^t \|(\tilde{u}, \tilde{E}, \tilde{B})(s)\|_{L^2_x} ds$ is continuous, the set S is actually both open and closed. Furthermore, it is nonempty and [0, T) is connected. We conclude that S = [0, T) and, therefore, that both solutions (u_1, E_1, B_1) and (u_2, E_2, B_2) match on the whole interval of existence [0, T). This completes the proof of the weak-strong uniqueness principle and concludes the proof of the theorem. \Box

Remark. In view of the estimates on the electric current *j* established in the preceding proof, it seems also possible to propagate the boundedness of the $\dot{B}_{p,1}^{s}$ -norms of the vorticity, with $p \in (1, \infty)$ and s = 2/p, by making use of the methods developed in [Vishik 1998] to prove the global well-posedness of the two-dimensional incompressible Euler system in critical spaces. We also refer to [Abidi et al. 2010; Hassainia 2022] for well-posedness results of similar models in critical spaces.

Remark. It is possible to propagate the boundedness of L^p -norms of the vorticity for values $1 \le p < 2$. Indeed, a variation of the proof of Lemma 3.6 gives

$$\begin{split} \|\omega(t)\|_{L_x^p} &\leq \|\omega(0)\|_{L_x^p} + \|j\|_{L^2([0,t);L_x^2)} \|\nabla B\|_{L^2([0,t);L_x^{2p/(2-p)})} \\ &\leq \|\omega(0)\|_{L_x^p} + \|j\|_{L^2([0,t);L_x^2)} \|\nabla B\|_{L^2([0,t);L_x^2)}^{2/p-1} \|\nabla B\|_{L^2([0,t);L_x^\infty)}^{2-2/p} \end{split}$$

for any t > 0. It is then readily seen that the terms involving ∇B are controlled by the bounds (3-2) and (3-3), whereas the electric current *j* remains bounded by virtue of the energy inequality (1-3).

3.10. *Proof of Theorem 3.3.* The proof of Theorem 3.3 builds upon the estimates established in Theorem 3.1. Thus, following the proof of that theorem, we assume that we have a smooth solution (u, E, B) of (1-1) for some smooth initial data (u_0, E_0, B_0) , and we only derive the bounds relevant to our argument through formal estimates on (u, E, B), keeping in mind that the full justification of the result is then completed by carrying out the approximation strategy laid out in Section 3.2.

Now, the proof of Theorem 3.1 establishes that the bound (3-45) on the initial data (u_0, E_0, B_0) implies the global uniform bound (3-46) on the solution (u, E, B). In particular, combining the two inequalities (3-45) and (3-46), we see that, for any 0 < A < 1, if the initial data satisfies (3-45) with its right-hand side replaced by Ac, then the solution (u, E, B) satisfies the estimate

$$\mathcal{H}(0,T) < \frac{Ac}{2C_*(1 + \mathcal{E}_0^{1/\alpha + 1})}$$
(3-47)

for all $T \in [0, \infty)$. This global bound will come in handy below, with some small but fixed value for the constant A.

Next, in order to derive a higher-regularity estimate on the field (E, B), we extend the control of high electromagnetic frequencies (3-30) to higher smoothness parameters. Specifically, a direct application of estimate (3-19) from Lemma 3.8 yields

$$c^{1-s} \| (E,B) \|_{\widetilde{L}^{\infty}_{t} \dot{B}^{s}_{2,n,>}} + c^{2-s} \| (E,B) \|_{\widetilde{L}^{2}_{t} \dot{B}^{s}_{2,n,>}} + c^{\frac{7}{4}-s} \| (E,B) \|_{\widetilde{L}^{2}_{t} \dot{B}^{s-3/4}_{\infty,n,>}} \\ \lesssim c^{1-s} \| (E_{0},B_{0}) \|_{\dot{B}^{s}_{2,n,>}} + c^{1-s} \| u \|_{L^{\infty}_{t,x} \cap L^{\infty}_{t} \dot{H}^{1}_{x}} \| B \|_{\widetilde{L}^{2}_{t} \dot{B}^{s}_{2,n,>}}$$

on any time interval [0, T) for any $\frac{7}{4} < s < 2$ and $1 \le n \le \infty$. Then, recalling that the energy $\mathcal{H}(0, T)$ controls the velocity u in $L_{t,x}^{\infty} \cap L_t^{\infty} \dot{H}_x^1$ (thanks to the Gagliardo–Nirenberg interpolation inequality (3-16)) and the magnetic field B in $L_t^2 \dot{B}_{2,1,<}^2$, we infer that the last term above can be bounded by

$$c^{1-s}\mathcal{H}(0,T)(\|B\|_{\tilde{L}^{2}_{t}\dot{B}^{1}_{2,\infty,<}}^{2-s}\|B\|_{\tilde{L}^{2}_{t}\dot{B}^{2}_{2,\infty,<}}^{s-1} + \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{s}_{2,n,>}})$$

$$\lesssim c^{1-s}\mathcal{H}(0,T)(\|B\|_{L^{2}_{t}\dot{H}^{1}_{x}}^{2-s}\|B\|_{L^{2}_{t}\dot{B}^{2}_{2,1,<}}^{s-1} + \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{s}_{2,n,>}})$$

$$\lesssim c^{1-s}\mathcal{H}(0,T)((\mathcal{E}_{0}+c^{-1}\mathcal{H}(0)+c^{-1}\mathcal{E}_{0}\mathcal{H}(0,T))^{2-s}\mathcal{H}(0,T)^{s-1} + \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{s}_{2,n,>}}).$$

where we also employed (3-36) in the last step to control B in $L_t^2 \dot{H}_x^1$.

All in all, combining the preceding inequalities and making use of (3-47), we arrive at

$$c^{1-s} \| (E,B) \|_{\tilde{L}^{\infty}_{t} \dot{B}^{s}_{2,n,>}} + c^{2-s} \| (E,B) \|_{\tilde{L}^{2}_{t} \dot{B}^{s}_{2,n,>}} + c^{\frac{7}{4}-s} \| (E,B) \|_{\tilde{L}^{2}_{t} \dot{B}^{s-3/4}_{\infty,n,>}} \\ \lesssim c^{1-s} \| (E_{0},B_{0}) \|_{\dot{B}^{s}_{2,n,>}} + c^{1-s} \mathcal{E}^{2-s}_{0} \mathcal{H}(0,T)^{s} + c^{-1} (1+\mathcal{E}^{2-s}_{0}) \mathcal{H}(0,T)^{2} + Ac^{2-s} \| B \|_{\tilde{L}^{2}_{t} \dot{B}^{s}_{2,n,>}}$$

Thus, setting the value of the constant A small enough (with respect to fixed parameters only) that the last term above can be absorbed by the left-hand side, the previous estimate can be recast as

$$c^{1-s} \| (E,B) \|_{\widetilde{L}^{\infty}_{t} \dot{B}^{s}_{2,n,>}} + c^{2-s} \| (E,B) \|_{\widetilde{L}^{2}_{t} \dot{B}^{s}_{2,n,>}} + c^{\frac{1}{4}-s} \| (E,B) \|_{\widetilde{L}^{2}_{t} \dot{B}^{s-3/4}_{\infty,n,>}} \\ \lesssim c^{1-s} \| (E_{0},B_{0}) \|_{\dot{B}^{s}_{2,n,>}} + c^{1-s} \mathcal{E}^{2-s}_{0} \mathcal{H}(0,T)^{s} + c^{-1} (1 + \mathcal{E}^{2-s}_{0}) \mathcal{H}(0,T)^{2} .$$

which provides the pursued uniform control of the solution (u, E, B) in higher-regularity spaces.

Finally, it only remains to notice that

$$c^{-\frac{3}{4}} \| (E_0, B_0) \|_{\dot{B}_{2,1}^{7/4}} \le c^{-\frac{3}{4}} \| (E_0, B_0) \|_{\dot{B}_{2,1,<}^{7/4}} + c^{-\frac{3}{4}} \| (E_0, B_0) \|_{\dot{B}_{2,1,>}^{7/4}} \\ \lesssim \| (E_0, B_0) \|_{\dot{H}^1} + c^{1-s} \| (E_0, B_0) \|_{\dot{B}_{2,n}^s},$$

which allows us to deduce that a suitable choice of independent constant $C_{**} > 0$ in the initial assumption (3-4) implies the corresponding initial condition (3-1) in Theorem 3.1, with its right-hand side replaced by Ac (this is necessary to guarantee the validity of (3-47)), thus completing the proof of the theorem. \Box

3.11. *Uniform bounds for fixed initial data.* As previously mentioned, the controls (3-2) and (3-5), from Theorems 3.1 and 3.3, hold for any families of initial data such that the left-hand sides of (3-1) and (3-4) remain respectively bounded. In particular, within such families, the corresponding collection of global solutions only satisfies the respective uniform bounds

$$c^{-\frac{3}{4}}(E,B) \in \tilde{L}^{\infty}(\mathbb{R}^+;\dot{B}^{7/4}_{2,1}), \quad c^{1-s}(E,B) \in \tilde{L}^{\infty}(\mathbb{R}^+;\dot{B}^{s}_{2,n}).$$

Thus, there is, in general, no bound on the size of the electromagnetic field (E, B) in $\tilde{L}^{\infty}(\mathbb{R}^+; \dot{B}_{2,1}^{7/4})$ and $\tilde{L}^{\infty}(\mathbb{R}^+; \dot{B}_{2,n}^s)$, uniformly in *c*, if the corresponding family of initial data (E_0, B_0) only satisfies a uniform control

$$c^{-\frac{3}{4}}(E_0, B_0) \in \dot{B}_{2,1}^{7/4}$$
 and $c^{1-s}(E_0, B_0) \in \dot{B}_{2,n}^s$,

respectively.

For example, such sets of initial electromagnetic fields occur naturally when considering mollifications

$$(u_0^c, E_0^c, B_0^c) = \varphi_c * (u_0, E_0, B_0),$$

where $\varphi_c(x) = c^2 \varphi(cx)$ is a classical approximate identity and (u_0, E_0, B_0) is a given initial data satisfying

$$(\mathcal{E}_0 + \|u_0\|_{\dot{H}^1 \cap \dot{W}^{1,p}} + \|(E_0, B_0)\|_{\dot{H}^1})C_{***}e^{C_{***}\mathcal{E}_0^{4+\varepsilon}} < c,$$
(3-48)

with $C_{***} > 0$. Indeed, it is readily seen that (u_0^c, E_0^c, B_0^c) satisfies the bounds (3-1) and (3-4), provided (3-48) holds for some suitable constant C_{***} .

We are now going to show that the solutions constructed in Theorems 3.1 and 3.3 have, in fact, an electromagnetic field which remains uniformly bounded in

$$\widetilde{L}^{\infty}(\mathbb{R}^+; \dot{B}^{7/4}_{2,1})$$
 and $\widetilde{L}^{\infty}(\mathbb{R}^+; \dot{B}^s_{2,n}),$

provided that their corresponding initial values are selected within a bounded family of $\dot{B}_{2,1}^{7/4}$ and $\dot{B}_{2,n}^s$, respectively. In particular, such uniform bounds hold whenever one considers fixed initial data independent of *c*. This is of special significance in the study of the limiting regime $c \to \infty$ in order to derive sharp asymptotic bounds.

The next result provides a suitable energy estimate on the damped Maxwell system (1-15), and the ensuing corollary clarifies the uniform boundedness properties of the solutions built in Theorems 3.1 and 3.3.

Lemma 3.12. Let d = 2. Assume that (E, B) is a smooth solution to (1-15) for some initial data (E_0, B_0) and some divergence-free vector field u, with the normal structure (1-2).

Then, one has the estimate

Proof. This result follows from a direct energy estimate on the damped Maxwell system (1-15) and is an extension of Lemma 3.10.

In order to show (3-49), we first localize (1-15) in frequencies by applying Δ_j , for $j \in \mathbb{Z}$, and then perform a classical energy estimate on each dyadic frequency component $(\Delta_j E, \Delta_j B)$. This procedure yields the control

$$\begin{aligned} \frac{1}{2c} \| (\Delta_j E, \Delta_j B)(T) \|_{L^2_x}^2 + \sigma c \| \Delta_j E \|_{L^2_t([0,T);L^2_x)}^2 \\ &\leq \frac{1}{2c} \| (\Delta_j E, \Delta_j B)(0) \|_{L^2_x}^2 + \sigma \| \Delta_j P(u \times B) \|_{L^2_t([0,T);L^2_x)} \| \Delta_j E \|_{L^2_t([0,T);L^2_x)} \\ &\leq \frac{1}{2c} \| (\Delta_j E, \Delta_j B)(0) \|_{L^2_x}^2 + \frac{\sigma}{2c} \| \Delta_j P(u \times B) \|_{L^2_t([0,T);L^2_x)}^2 + \frac{\sigma c}{2} \| \Delta_j E \|_{L^2_t([0,T);L^2_x)}^2 \end{aligned}$$

Then, summing over j in ℓ^n , with $1 \le n \le \infty$, we deduce, for any $s \in \mathbb{R}$, that

$$\|(E,B)\|_{\tilde{L}^{\infty}_{t}([0,T);\dot{B}^{s}_{2,n})} + c \|E\|_{\tilde{L}^{2}_{t}([0,T);\dot{B}^{s}_{2,n})} \lesssim \|(E_{0},B_{0})\|_{\dot{B}^{s}_{2,n}} + \|P(u\times B)\|_{\tilde{L}^{2}_{t}([0,T);\dot{B}^{s}_{2,n})}$$

Next, by the paradifferential product law (3-13), we see that

 $\|P(u \times B)\|_{\tilde{L}^{2}_{t}\dot{B}^{s}_{2,n}} \lesssim \|u\|_{L^{\infty}_{t,x} \cap L^{\infty}_{t}\dot{B}^{1}_{2,\infty}} \|B\|_{\tilde{L}^{2}_{t}\dot{B}^{s}_{2,n}}$

for any -1 < s < 2 and $1 \le n \le \infty$. Therefore, combining the preceding inequalities, we conclude that $\|(E, B)\|_{\tilde{L}^{\infty}_{t}\dot{B}^{s}_{2,n}} + c \|E\|_{\tilde{L}^{2}_{t}\dot{B}^{s}_{2,n}}$

$$\lesssim \|(E_0, B_0)\|_{\dot{B}_{2,n}^s} + \|u\|_{L^{\infty}_{t,x} \cap L^{\infty}_t \dot{B}_{2,\infty}^1} (\|B\|_{\tilde{L}^2_t \dot{B}_{2,n,<}^s} + \|B\|_{\tilde{L}^2_t \dot{B}_{2,n,>}^s})$$

$$\lesssim \|(E_0, B_0)\|_{\dot{B}_{2,n}^s} + \|u\|_{L^{\infty}_{t,x} \cap L^{\infty}_t \dot{B}_{2,\infty}^1} (\|B\|_{\tilde{L}^2_t \dot{B}_{2,\infty,<}^1}^{2-s} \|B\|_{\tilde{L}^2_t \dot{B}_{2,\infty,<}^s}^{s-1} + \|B\|_{\tilde{L}^2_t \dot{B}_{2,n,>}^s})$$

for all 1 < s < 2 and $1 \le n \le \infty$.

Finally, combining the previous estimate with straightforward embeddings of Besov spaces proves (3-49), thereby concluding the proof of the lemma.

Corollary 3.13. Consider parameters $p \in (2, \infty)$, $\frac{7}{4} \le s < 2$ and $1 \le n \le \infty$ such that n = 1 if $s = \frac{7}{4}$. For any fixed initial data

$$(u_0, E_0, B_0) \in ((H^1 \cap \dot{W}^{1,p}) \times (H^1 \cap \dot{B}^s_{2,n}) \times (H^1 \cap \dot{B}^s_{2,n}))(\mathbb{R}^2)$$

satisfying the assumptions of Theorem 3.1 (if $s = \frac{7}{4}$ and n = 1) or Theorem 3.3 (if $s > \frac{7}{4}$), the corresponding global solutions (u^c , E^c , B^c) constructed therein satisfy the bounds

$$\begin{split} u^{c} &\in L^{\infty}(\mathbb{R}^{+}; H^{1} \cap \dot{W}^{1,p}), \quad (E^{c}, B^{c}) \in L^{\infty}(\mathbb{R}^{+}; H^{1}), \quad (E^{c}, B^{c}) \in \tilde{L}^{\infty}(\mathbb{R}^{+}; \dot{B}^{s}_{2,n}), \\ &(cE^{c}, B^{c}) \in L^{2}(\mathbb{R}^{+}; \dot{H}^{1}), \quad B^{c} \in L^{2}(\mathbb{R}^{+}; \dot{B}^{2}_{2,1,<}), \\ c^{\frac{7}{4}-s}(E^{c}, B^{c}) \in \tilde{L}^{2}(\mathbb{R}^{+}; \dot{B}^{s-3/4}_{\infty,n,>}), \quad cE^{c} \in \tilde{L}^{2}(\mathbb{R}^{+}; \dot{B}^{s}_{2,n}), \quad c^{\frac{1}{4}}B^{c} \in \tilde{L}^{2}(\mathbb{R}^{+}; \dot{B}^{s}_{2,n,>}), \end{split}$$

uniformly in c.

Proof. This result follows from a straightforward combination of Theorems 3.1 and 3.3 with Lemma 3.12. Indeed, it is readily seen that the uniform bounds (3-2) and (3-5) are sufficient to control the right-hand side of (3-49) for appropriate values of (s, n), thereby showing the improved uniform boundedness of the solutions (u^c, E^c, B^c) .

Appendix A: Littlewood–Paley decompositions and Besov spaces

We recall here the fundamentals of Littlewood–Paley decompositions and introduce a precise definition of Besov spaces.

A.1. Littlewood-Paley decompositions. We are going to use the Fourier transform

$$\mathcal{F}f(\xi) = \hat{f}(\xi) := \int_{\mathbb{R}^d} e^{-i\xi \cdot x} f(x) \, dx$$

and its inverse

$$\mathcal{F}^{-1}g(x) = \tilde{g}(x) := \frac{1}{(2\pi)^d} \int_{\mathbb{R}^d} e^{ix\cdot\xi} g(\xi) \, d\xi$$

in any dimension $d \ge 1$.

Now, consider smooth cutoff functions $\psi(\xi), \varphi(\xi) \in C_c^{\infty}(\mathbb{R}^d)$ satisfying

$$\psi, \varphi \ge 0$$
 are radial, $\sup \psi \subset \{|\xi| \le 1\}, \sup \varphi \subset \{\frac{1}{2} \le |\xi| \le 2\}$

and

$$1 = \psi(\xi) + \sum_{k=0}^{\infty} \varphi(2^{-k}\xi) \quad \text{for all } \xi \in \mathbb{R}^d.$$

Defining the scaled cutoffs

$$\psi_k(\xi) := \psi(2^{-k}\xi), \quad \varphi_k(\xi) := \varphi(2^{-k}\xi)$$

for each $k \in \mathbb{Z}$, so that

supp
$$\psi_k \subset \{|\xi| \le 2^k\}$$
, supp $\varphi_k \subset \{2^{k-1} \le |\xi| \le 2^{k+1}\}$,

it is readily seen that

$$1 \equiv \psi + \sum_{k=0}^{\infty} \varphi_k$$

provides us with a dyadic partition of unity of \mathbb{R}^d . Notice also that

$$1 \equiv \psi_j + \sum_{k=j}^{\infty} \varphi_k$$

for any $j \in \mathbb{Z}$ and

$$1 \equiv \sum_{k=-\infty}^{\infty} \varphi_k$$

away from the origin $\xi = 0$.

Next, denoting the space of tempered distributions by S', we introduce the Fourier multiplier operators

$$S_k, \Delta_k : \mathcal{S}'(\mathbb{R}^d) \to \mathcal{S}'(\mathbb{R}^d),$$

with $k \in \mathbb{Z}$, defined by

$$S_k f := \mathcal{F}^{-1} \psi_k \mathcal{F} f = (\mathcal{F}^{-1} \psi_k) * f \quad \text{and} \quad \Delta_k f := \mathcal{F}^{-1} \varphi_k \mathcal{F} f = (\mathcal{F}^{-1} \varphi_k) * f.$$
(A-1)

The Littlewood–Paley decomposition of $f \in S'$ is then given by

$$S_0 f + \sum_{k=0}^{\infty} \Delta_k f = f$$

where the series is convergent in \mathcal{S}' .

Similarly, one can verify that the homogeneous Littlewood-Paley decomposition

$$\sum_{\substack{k=-\infty\\k\to-\infty}}^{\infty} \Delta_k f = f$$

$$\lim_{\substack{k\to-\infty}} \|S_k f\|_{L^{\infty}} = 0.$$
(A-2)

holds in S' if $f \in S'$ satisfies

Observe that (A-2) holds if \hat{f} is locally integrable around the origin, or whenever $S_0 f$ belongs to $L^p(\mathbb{R}^d)$ for some $1 \le p < \infty$. In particular, note that (A-2) excludes all nonzero polynomials.

A.2. Besov spaces. For any $s \in \mathbb{R}$ and $1 \le p, q \le \infty$, we define now the homogeneous Besov space $\dot{B}_{p,q}^{s}(\mathbb{R}^{d})$ as the subspace of tempered distributions satisfying (A-2) endowed with the norm

$$\|f\|_{\dot{B}^{s}_{p,q}(\mathbb{R}^{d})} = \left(\sum_{k \in \mathbb{Z}} 2^{ksq} \|\Delta_{k} f\|_{L^{p}(\mathbb{R}^{d})}^{q}\right)^{\frac{1}{q}}$$

if $q < \infty$, and

$$\|f\|_{\dot{B}^{s}_{p,q}(\mathbb{R}^d)} = \sup_{k \in \mathbb{Z}} (2^{ks} \|\Delta_k f\|_{L^p(\mathbb{R}^d)})$$

if $q = \infty$. One can show that $\dot{B}_{p,q}^s$ is a Banach space if s < d/p, or if s = d/p and q = 1; see [Bahouri et al. 2011, Theorem 2.25].
We also consider here the homogeneous Sobolev space $\dot{H}^{s}(\mathbb{R}^{d})$ for any real value $s \in \mathbb{R}$, which is defined as the subspace of tempered distributions whose Fourier transform is locally integrable equipped with the norm

$$\|f\|_{\dot{H}^{s}} = \left(\int_{\mathbb{R}^{d}} |\xi|^{2s} |\hat{f}(\xi)|^{2} d\xi\right)^{\frac{1}{2}}.$$

One verifies that \dot{H}^s is a Hilbert space if and only if $s < \frac{1}{2}d$; see [Bahouri et al. 2011, Proposition 1.34]. Moreover, it is possible to show that $\dot{H}^s = \dot{B}_{2,2}^s$ whenever $s < \frac{1}{2}d$.

A.3. *Chemin–Lerner spaces.* For any time T > 0 and any choice of parameters $s \in \mathbb{R}$ and $1 \le p, q, r \le \infty$, with s < d/p (or s = d/p and q = 1), the spaces

$$L^r([0,T); \dot{B}^s_{p,q}(\mathbb{R}^d))$$

are naturally defined as L^r -spaces with values in the Banach spaces $\dot{B}_{p,q}^s$. In addition to these vector-valued Lebesgue spaces, we define the spaces

$$\tilde{L}^r([0,T);\dot{B}^s_{p,q}(\mathbb{R}^d))$$

as the subspaces of tempered distributions such that

$$\lim_{k \to -\infty} \|S_k f\|_{L^r([0,T);L^p(\mathbb{R}^d))} = 0.$$

endowed with the norm

$$\|f\|_{\tilde{L}^{r}([0,T);B^{s}_{p,q}(\mathbb{R}^{d}))} = \left(\sum_{k=-\infty}^{\infty} 2^{ksq} \|\Delta_{k}f\|_{L^{r}([0,T);L^{p}(\mathbb{R}^{d}))}^{q}\right)^{\frac{1}{q}}$$

if $q < \infty$, and with the obvious modifications in the case $q = \infty$.

One can easily check that, if $q \ge r$, then

$$L^{r}([0,T); \dot{B}^{s}_{p,q}(\mathbb{R}^{d})) \subset \widetilde{L}^{r}([0,T); \dot{B}^{s}_{p,q}(\mathbb{R}^{d}))$$

and that, if $q \leq r$, then

$$\widetilde{L}^r([0,T);\dot{B}^s_{p,q}(\mathbb{R}^d)) \subset L^r([0,T);\dot{B}^s_{p,q}(\mathbb{R}^d)).$$

We refer the reader to [Bahouri et al. 2011, Section 2.6.3] for more details on Chemin–Lerner spaces.

A.4. *Embeddings.* We present now a few embeddings and inequalities in Besov spaces which are routinely used throughout this work.

First, a direct application of Young's convolution inequality to (A-1) yields

$$\|\Delta_k f\|_{L^r(\mathbb{R}^d)} \lesssim 2^{kd\left(\frac{1}{p} - \frac{1}{r}\right)} \|\Delta_k f\|_{L^p(\mathbb{R}^d)}$$
(A-3)

for any $1 \le p \le r \le \infty$. A suitable use of (A-3) then leads to the embedding

$$\|f\|_{\dot{B}^{s}_{r,q}(\mathbb{R}^{d})} \lesssim \|f\|_{\dot{B}^{s+d(1/p-1/r)}(\mathbb{R}^{d})}$$
(A-4)

for any $1 \le p \le r \le \infty$, $1 \le q \le \infty$ and $s \in \mathbb{R}$, which can be interpreted as a Sobolev embedding in the framework of Besov spaces.

Moreover, recalling that $\ell^q \subset \ell^r$, for all $1 \le q \le r \le \infty$, one has

$$\dot{B}^{s}_{p,q}(\mathbb{R}^d) \subset \dot{B}^{s}_{p,r}(\mathbb{R}^d)$$

for all $s \in \mathbb{R}$, $1 \le p \le \infty$ and $1 \le q \le r \le \infty$.

Next, observe that

$$\|f\|_{L^{p}(\mathbb{R}^{d})} = \left\|\sum_{k\in\mathbb{Z}}\Delta_{k}f\right\|_{L^{p}(\mathbb{R}^{d})} \le \sum_{k\in\mathbb{Z}}\|\Delta_{k}f\|_{L^{p}(\mathbb{R}^{d})} = \|\Delta_{k}f\|_{\dot{B}^{0}_{p,1}(\mathbb{R}^{d})}$$
(A-5)

for every $1 \le p \le \infty$. Therefore, by combining (A-4) and (A-5), we obtain

$$\|f\|_{L^{\infty}(\mathbb{R}^d)} \lesssim \|f\|_{\dot{B}^{d/2}_{2,1}(\mathbb{R}^d)}$$

This estimate is particularly useful in view of the failure of the embedding of the Sobolev space $\dot{H}^{d/2}(\mathbb{R}^d)$ into $L^{\infty}(\mathbb{R}^d)$.

Further considering any cutoff function $\chi \in C_c^{\infty}(\mathbb{R}^d)$ such that $\mathbb{1}_{\{|\xi| \le 1\}} \le \chi(\xi) \le \mathbb{1}_{\{|\xi| \le 2\}}$, one can show, for any c > 0, $\alpha > 0$ and $1 \le p \le \infty$, that

$$\left\|\chi\left(\frac{D}{c}\right)f\right\|_{\dot{B}^{s+\alpha}_{p,1}(\mathbb{R}^d)} \lesssim c^{\alpha} \|f\|_{\dot{B}^{s}_{p,\infty}(\mathbb{R}^d)}$$

and

$$c^{\alpha}\left\|(1-\chi)\left(\frac{D}{c}\right)f\right\|_{\dot{B}^{s}_{p,1}(\mathbb{R}^{d})} \lesssim \|f\|_{\dot{B}^{s+\alpha}_{p,\infty}(\mathbb{R}^{d})},$$

where the operator m(D) denotes the Fourier multiplier associated with the symbol $m(\xi)$ for any $m \in C_c^{\infty}(\mathbb{R}^d)$.

Finally, we mention another essential inequality in Besov spaces which is related to their interpolation properties. Specifically, one has the interpolation, or convexity, inequality

$$\|f\|_{\dot{B}^{s}_{p,1}} \lesssim \|f\|_{\dot{B}^{s_{0}}_{p,\infty}}^{1-\theta} \|f\|_{\dot{B}^{s_{1}}_{p,\infty}}^{\theta}$$

for any $p \in [1, \infty]$, $s, s_0, s_1 \in \mathbb{R}$ and $\theta \in (0, 1)$ such that $s = (1 - \theta)s_0 + \theta s_1$ and $s_0 \neq s_1$.

Note that the preceding estimates and embeddings can be adapted to the setting of Chemin–Lerner spaces in a straightforward way.

A.5. *Paradifferential product estimates.* Here, we recall the basic principles of paraproduct decompositions and some essential paradifferential product estimates that follow from it.

For any two suitable tempered distributions f and g, introducing the paraproduct

$$T_f g = \sum_{j \in \mathbb{Z}} S_{j-2} f \Delta_j g$$

readily leads to the decomposition

$$fg = T_f g + T_g f + R(f,g),$$

where

$$R(f,g) = \sum_{\substack{j,k \in \mathbb{Z} \\ |j-k| \le 2}} \Delta_j f \Delta_k g$$

is the remainder. For any choice of integrability parameters in $[1, \infty]$ such that

$$\frac{1}{a} = \frac{1}{a_1} + \frac{1}{a_2}, \quad \frac{1}{b} = \frac{1}{b_1} + \frac{1}{b_2}, \quad \frac{1}{c} = \frac{1}{c_1} + \frac{1}{c_2},$$

it can be shown, in the context of Chemin-Lerner spaces, that

$$\|T_{f}g\|_{\tilde{L}_{t}^{a}\dot{B}_{b,c}^{\alpha+\beta}} \lesssim \|f\|_{\tilde{L}_{t}^{a_{1}}\dot{B}_{b_{1},c_{1}}^{\alpha}} \|g\|_{\tilde{L}_{t}^{a_{2}}\dot{B}_{b_{2},c_{2}}^{\beta}}$$
(A-6)

for any $\alpha < 0$ and $\beta \in \mathbb{R}$, and that

$$\|R(f,g)\|_{\tilde{L}^{a}_{t}\dot{B}^{\alpha+\beta}_{b,c}} \lesssim \|f\|_{\tilde{L}^{a_{1}}_{t}\dot{B}^{\alpha}_{b_{1},c_{1}}} \|g\|_{\tilde{L}^{a_{2}}_{t}\dot{B}^{\beta}_{b_{2},c_{2}}}$$
(A-7)

for any $\alpha, \beta \in \mathbb{R}$ with $\alpha + \beta > 0$. Moreover, one has the endpoint estimates

$$\|T_{f}g\|_{\tilde{L}_{t}^{a}\dot{B}_{b,c}^{\beta}} \lesssim \|f\|_{L_{t}^{a_{1}}L_{x}^{b_{1}}}\|g\|_{\tilde{L}_{t}^{a_{2}}\dot{B}_{b_{2},c}^{\beta}}$$
(A-8)

for all $\beta \in \mathbb{R}$ and

$$\|R(f,g)\|_{L^{a}_{t}L^{b}_{x}} \lesssim \|f\|_{\tilde{L}^{a_{1}}_{t}\dot{B}^{\alpha}_{b_{1},c_{1}}}\|g\|_{\tilde{L}^{a_{2}}_{t}\dot{B}^{-\alpha}_{b_{2},c_{2}}}$$

for all $\alpha \in \mathbb{R}$, provided $1/c_1 + 1/c_2 = 1$. Similar bounds hold for Besov spaces, and we refer to [Bahouri et al. 2011, Section 2.6] for more details on such paradifferential estimates.

We finally recall two important product rules of paradifferential calculus in the context of Besov spaces, which are direct consequences of the preceding bounds. First, exploiting (A-6) and (A-7) (for Besov spaces), we have

$$\|fg\|_{\dot{B}^{s+t-d/2}_{2,1}} \lesssim \|f\|_{\dot{H}^s} \|g\|_{\dot{H}^s}$$

for any $-\frac{1}{2}d < s, t < \frac{1}{2}d$, with s + t > 0. Second, we find that

$$\|fg\|_{\dot{H}^{s}} \lesssim \|f\|_{L^{\infty} \cap \dot{B}_{2,\infty}^{d/2}} \|g\|_{\dot{H}^{s}}$$

for all $-\frac{1}{2}d < s < \frac{1}{2}d$, which follows from a suitable combination of (A-6), (A-7) and (A-8) (for Besov spaces, as well).

Appendix B: Oscillatory integrals and dispersion

We give here a justification of the dispersive estimate (2-23) employing the stationary phase method. This method is classical and we will rely on [Bahouri et al. 2011], when needed, to refer the reader to complete details on the technical results pertaining to the method.

Generally speaking, we are considering here oscillatory integrals of the form

$$I_{\psi}(t) = \int_{\mathbb{R}^d} e^{it\phi(\xi)} \psi(\xi) \, d\xi$$

for smooth test functions $\psi \in C_c^{\infty}(\mathbb{R}^d)$ and $t \in \mathbb{R}$, where the smooth phase $\phi(\xi)$ only needs to be defined on the support of ψ . It is readily seen that

$$|I_{\psi}(t)| \le \|\psi\|_{L^1(\mathbb{R}^d)}$$

We seek now to understand the asymptotic behavior of $I_{\psi}(t)$ when |t| is large, which requires us to exploit the cancellations in the integral $I_{\psi}(t)$ due to the oscillatory term $e^{it\phi(\xi)}$. There are two cases to consider: the stationary phase and the nonstationary phase.

The stationary phase. This case analyzes the asymptotic behavior of $I_{\psi}(t)$ near critical points of the phase, i.e., near points in the integration domain where $\nabla \phi(\xi) = 0$. More precisely, we suppose here that

$$|
abla \phi(\xi)| \le \varepsilon_0$$

for some $\varepsilon_0 \in (0, 1]$ and for all $\xi \in \text{supp } \psi$. Under such assumptions, Theorem 8.9 from [Bahouri et al. 2011] establishes that, for any positive integer *N*, there is a constant $C_{N,\phi,\psi} > 0$ such that

$$|I_{\psi}(t)| \le C_{N,\phi,\psi} \int_{\operatorname{supp}\psi} \frac{d\xi}{(1+\varepsilon_0 t |\nabla\phi(\xi)|^2)^N}$$
(B-1)

for all t > 0.

The nonstationary phase. The decay of $I_{\psi}(t)$ is better when $\nabla \phi$ does not vanish on the support of ψ . More precisely, assuming now that

 $|\nabla \phi(\xi)| \ge \varepsilon_0$

for some $\varepsilon_0 \in (0, 1]$ and for all $\xi \in \text{supp } \psi$, Theorem 8.8 from [Bahouri et al. 2011] shows that, for any positive integer *N*, there is a constant $C_{N,\phi,\psi} > 0$ such that

$$|I_{\psi}(t)| \le \frac{C_{N,\phi,\psi}}{(\varepsilon_0 t)^N} \tag{B-2}$$

for all t > 0.

The asymptotic estimate (B-2) always offers a faster decay than (B-1) and, therefore, the oscillatory integral $I_{\psi}(t)$ can be treated as a remainder term wherever the phase $\nabla \phi$ does not vanish. In conclusion, the overall asymptotic behavior of $I_{\psi}(t)$ is, in general, determined by the critical points of the phase.

All in all, as explained in Theorem 8.12 from [Bahouri et al. 2011], it is possible to combine the preceding estimates to show, for all $\psi \in C_c^{\infty}(\mathbb{R}^d)$, $\varepsilon_0 \in (0, 1]$ and any positive numbers N and N', that there are positive constants C_N and $C_{N'}$ such that

$$|I_{\psi}(t)| \leq \frac{C_N}{(\varepsilon_0 t)^N} + C_{N'} \int_{A_{\phi}} \frac{d\xi}{(1 + \varepsilon_0 t |\nabla \phi(\xi)|^2)^{N'}}$$
(B-3)

for all t > 0, where the set A_{ϕ} is defined as

$$A_{\phi} := \{ \xi \in \operatorname{supp} \psi : |\nabla \phi(\xi)| \le \varepsilon_0 \}.$$

We are now in a position to prove (2-23). To be precise, we are going to establish the equivalent estimate (up to a scaling of the variable *x*)

$$\left| \int_{\mathbb{R}^d} e^{it\phi(x,\xi)} \psi(\xi) \, d\xi \right| \le \frac{C_{\psi}}{t^{(d-1)/2}} \tag{B-4}$$

for all t > 0 and $x \in \mathbb{R}^d$, where the phase is defined by

$$\phi(x,\xi) = x \cdot \xi \pm \delta(\xi)$$
, with $\delta(\xi) = \sqrt{|\xi|^2 - \frac{1}{4}\alpha^2}$ and $0 \le \alpha \le \frac{1}{2}$

and the test function $\psi \in C_c^{\infty}(\mathbb{R}^d)$ satisfies supp $\psi \subset \{\frac{1}{4} < |\xi| < R\}$ for some $R > \frac{1}{4}$, while the constant $C_{\psi} > 0$ is independent of t, x and α .

To that end, noting that $\phi(x, \xi)$ is smooth on the support of ψ and setting $\varepsilon_0 = \frac{1}{2}$, $N = \frac{1}{2}(d-1)$ and N' = d in (B-3), we find that

$$\left| \int_{\mathbb{R}^d} e^{it\phi(x,\xi)} \psi(\xi) \, d\xi \right| \lesssim \frac{1}{t^{(d-1)/2}} + \int_A \frac{d\xi}{\left(1 + t \left| x \pm \frac{\xi}{\delta(\xi)} \right|^2 \right)^d}$$

where

$$A := \left\{ \frac{1}{4} < |\xi| < R, \left| x \pm \frac{\xi}{\delta(\xi)} \right| \le \frac{1}{2} \right\}.$$

Now, notice that $x \neq 0$ if A is nonempty. In particular, for any $\xi \in A$, we can write the decomposition

$$\xi = \zeta_1 + \zeta'$$
, with $\zeta_1 := \left(\frac{\xi \cdot x}{|x|^2}\right) x$ and $\zeta' := \xi - \zeta_1$,

whence, since $\zeta' \cdot x = 0$,

$$\left|x \pm \frac{\xi}{\delta(\xi)}\right|^2 = \left|x \pm \frac{\zeta_1}{\delta(\xi)}\right|^2 + \left|\frac{\zeta'}{\delta(\xi)}\right|^2 \ge \frac{|\zeta'|^2}{\delta(\xi)^2} \ge \frac{|\zeta'|^2}{R^2}.$$

We therefore conclude that

$$\left| \int_{\mathbb{R}^d} e^{it\phi(x,\xi)} \psi(\xi) \, d\xi \right| \lesssim \frac{1}{t^{(d-1)/2}} + \int_{\{|\zeta'| < R\} \subset \mathbb{R}^{d-1}} \frac{d\zeta'}{(1+t|\zeta'|^2)^d} \lesssim \frac{1}{t^{(d-1)/2}},$$

which completes the justification of (B-4).

References

- [Abidi et al. 2010] H. Abidi, T. Hmidi, and S. Keraani, "On the global well-posedness for the axisymmetric Euler equations", *Math. Ann.* **347**:1 (2010), 15–41. MR Zbl
- [Arsénio 2019] D. Arsénio, "Recent progress on the mathematical theory of plasmas", *Bol. Soc. Port. Mat.* **77** (2019), 27–38. MR Zbl
- [Arsénio and Gallagher 2020] D. Arsénio and I. Gallagher, "Solutions of Navier–Stokes–Maxwell systems in large energy spaces", *Trans. Amer. Math. Soc.* **373**:6 (2020), 3853–3884. MR Zbl
- [Arsénio and Saint-Raymond 2019] D. Arsénio and L. Saint-Raymond, From the Vlasov-Maxwell-Boltzmann system to incompressible viscous electro-magneto-hydrodynamics, I, Eur. Math. Soc., Zürich, 2019. MR Zbl
- [Arsénio et al. 2015] D. Arsénio, S. Ibrahim, and N. Masmoudi, "A derivation of the magnetohydrodynamic system from Navier–Stokes–Maxwell systems", *Arch. Ration. Mech. Anal.* **216**:3 (2015), 767–812. MR Zbl

- [Bahouri et al. 2011] H. Bahouri, J.-Y. Chemin, and R. Danchin, *Fourier analysis and nonlinear partial differential equations*, Grundl. Math. Wissen. **343**, Springer, 2011. MR Zbl
- [Bergh and Löfström 1976] J. Bergh and J. Löfström, *Interpolation spaces: an introduction*, Grundl. Math. Wissen. 223, Springer, 1976. MR Zbl
- [Biskamp 1993] D. Biskamp, *Nonlinear magnetohydrodynamics*, Cambridge Monogr. Plasma Phys. **1**, Cambridge Univ. Press, 1993. MR
- [Davidson 2001] P. A. Davidson, An introduction to magnetohydrodynamics, Cambridge Univ. Press, 2001. MR Zbl
- [Germain and Masmoudi 2014] P. Germain and N. Masmoudi, "Global existence for the Euler–Maxwell system", *Ann. Sci. École Norm. Sup.* (4) **47**:3 (2014), 469–503. MR Zbl
- [Germain et al. 2014] P. Germain, S. Ibrahim, and N. Masmoudi, "Well-posedness of the Navier–Stokes–Maxwell equations", *Proc. Roy. Soc. Edinburgh Sect. A* 144:1 (2014), 71–86. MR Zbl
- [Grafakos 2014] L. Grafakos, Classical Fourier analysis, 3rd ed., Grad. Texts in Math. 249, Springer, 2014. MR Zbl
- [Guo et al. 2016] Y. Guo, A. D. Ionescu, and B. Pausader, "Global solutions of the Euler-Maxwell two-fluid system in 3D", Ann. of Math. (2) 183:2 (2016), 377–498. MR Zbl
- [Hassainia 2022] Z. Hassainia, "On the global well-posedness of the 3D axisymmetric resistive MHD equations", *Ann. Henri Poincaré* **23**:8 (2022), 2877–2917. MR Zbl
- [Inui 2019] T. Inui, "The Strichartz estimates for the damped wave equation and the behavior of solutions for the energy critical nonlinear equation", *NoDEA Nonlinear Differential Equations Appl.* **26**:6 (2019), art. id. 50. MR Zbl
- [Inui and Wakasugi 2021] T. Inui and Y. Wakasugi, "Unconditional well-posedness for the energy-critical nonlinear damped wave equation", *J. Evol. Equ.* **21**:4 (2021), 5171–5201. MR Zbl
- [Keel and Tao 1998] M. Keel and T. Tao, "Endpoint Strichartz estimates", Amer. J. Math. 120:5 (1998), 955–980. MR Zbl
- [Kunstmann and Weis 2004] P. C. Kunstmann and L. Weis, "Maximal L_p -regularity for parabolic equations, Fourier multiplier theorems and H^{∞} -functional calculus", pp. 65–311 in *Functional analytic methods for evolution equations*, Lecture Notes in Math. **1855**, Springer, 2004. MR Zbl
- [Lemarié-Rieusset 2016] P. G. Lemarié-Rieusset, *The Navier–Stokes problem in the 21st century*, CRC Press, Boca Raton, FL, 2016. MR Zbl
- [Majda and Bertozzi 2002] A. J. Majda and A. L. Bertozzi, *Vorticity and incompressible flow*, Cambridge Texts Appl. Math. 27, Cambridge Univ. Press, 2002. MR Zbl
- [Masmoudi 2010] N. Masmoudi, "Global well posedness for the Maxwell–Navier–Stokes system in 2D", *J. Math. Pures Appl.* (9) **93**:6 (2010), 559–571. MR Zbl
- [Saint-Raymond 2003] L. Saint-Raymond, "Convergence of solutions to the Boltzmann equation in the incompressible Euler limit", *Arch. Ration. Mech. Anal.* **166**:1 (2003), 47–80. MR Zbl
- [Saint-Raymond 2009a] L. Saint-Raymond, *Hydrodynamic limits of the Boltzmann equation*, Lecture Notes in Math. **1971**, Springer, 2009. MR Zbl
- [Saint-Raymond 2009b] L. Saint-Raymond, "Hydrodynamic limits: some improvements of the relative entropy method", *Ann. Inst. H. Poincaré C Anal. Non Linéaire* **26**:3 (2009), 705–744. MR Zbl
- [Simon 1987] J. Simon, "Compact sets in the space L^p(0, T; B)", Ann. Mat. Pura Appl. (4) 146 (1987), 65–96. MR Zbl
- [Vishik 1998] M. Vishik, "Hydrodynamics in Besov spaces", Arch. Ration. Mech. Anal. 145:3 (1998), 197-214. MR Zbl
- Received 7 Sep 2022. Revised 22 Dec 2023. Accepted 2 May 2024.
- DIOGO ARSÉNIO: diogo.arsenio@nyu.edu Division of Science, New York University Abu Dhabi, Abu Dhabi, United Arab Emirates
- HAROUNE HOUAMED: haroune.houamed@nyu.edu Division of Science, New York University Abu Dhabi, Abu Dhabi, United Arab Emirates



EQUIVARIANT PROPERTY GAMMA AND THE TRACIAL LOCAL-TO-GLOBAL PRINCIPLE FOR C*-DYNAMICS

GÁBOR SZABÓ AND LISE WOUTERS

We consider the notion of equivariant uniform property Gamma for actions of countable discrete groups on C*-algebras that admit traces. In case the group is amenable and the C*-algebra has a compact tracial state space, we prove that this property implies a kind of tracial local-to-global principle for the C*-dynamical system, generalizing a recent such principle for C*-algebras exhibited in work of Castillejos et al. For actions on simple nuclear \mathcal{Z} -stable C*-algebras, we use this to prove that equivariant uniform property Gamma is equivalent to equivariant \mathcal{Z} -stability, generalizing a result of Gardella, Hirshberg, and Vaccaro.

Introduction		1385
1.	Preliminaries	1390
2.	Equivariant uniform property Gamma	1398
3.	Dynamical complemented partitions of unity	1402
4.	Dynamical tracial local-to-global principle	1417
5.	Equivariant Jiang–Su stability	1425
Acknowledgements		1429
References		1429

Introduction

This article aims to extend the fine structure theory for actions of amenable groups on finite simple C^* -algebras, in particular those covered by the Elliott program. The classification of such C^* -algebras, which mirrors the celebrated Connes–Haagerup classification of injective factors [Connes 1976; Haagerup 1987], has been nearly completed as a culmination of numerous articles by many researchers over the past decade, such as [Elliott et al. 2024; 2020; Gong and Lin 2020; 2022; Gong et al. 2020a; 2020b; Schafhauser 2020; Tikuisis et al. 2017]. Furthermore, Carrion, Gabe, Schafhauser, Tikuisis, and White have announced an eagerly awaited new conceptual proof of the classification theorem [Carrión et al. 2023b], which does not directly rely on the prior works related to tracial approximation. By now it has been recognized that the next natural step is to understand the underlying symmetries of classifiable C*-algebras, which can be interpreted as the goal to classify group actions on them. This mirrors the work

© 2025 MSP (Mathematical Sciences Publishers). Distributed under the Creative Commons Attribution License 4.0 (CC BY). Open Access made possible by subscribing institutions via Subscribe to Open.

Szabó was supported by BOF project C14/19/088, funded by the research council of KU Leuven, and research project G085020N, funded by the Research Foundation Flanders (FWO). Wouters was supported by Ph.D. grant 11B6620N, funded by the Research Foundation Flanders (FWO).

MSC2020: 46L35, 46L55.

Keywords: C*-dynamics, amenability, Jiang–Su algebra.

of Connes, Jones, Ocneanu and others [Connes 1977; Jones 1980; Katayama et al. 1998; Kawahigashi et al. 1992; Masuda 2007; 2013; Ocneanu 1985; Sutherland and Takesaki 1989]. When it comes to classifying group actions on C*-algebras, a number of researchers have introduced many sophisticated methods over the years to classify specific kinds of group actions utilizing certain Rokhlin-type properties [Evans and Kishimoto 1997; Izumi 2004a; 2004b; Izumi and Matui 2010; 2021a; 2021b; Katsura and Matui 2008; Kishimoto 1995; 1998a; 1998b; Matui 2008; 2010; 2011; Nakamura 2000; Sato 2010; Szabó 2021a]. In direct comparison to the generality achieved for actions on von Neumann algebras, the implementation of the involved methods (in particular the Evans–Kishimoto intertwining argument) for actions on C*-algebras remained challenging beyond some specific classes of actions or acting groups.

To combat these methodological obstacles, the first author introduced a categorical framework in [Szabó 2021c] to open up the classification of C*-dynamics up to cocycle conjugacy to methodology directly inspired by [Elliott 2010]. For actions on classifiable C*-algebras without traces, the so-called Kirchberg algebras [1995], this idea led to the recent breakthrough in [Gabe and Szabó 2022; 2024]. The main result of said work implies that, given any countable amenable group *G*, any outer *G*-action on a Kirchberg algebra is uniquely determined by its KK^G -class up to cocycle conjugacy.

Although one might be tempted to guess that similar breakthrough results ought to be in reach for actions on finite classifiable C*-algebras, one still has a long way to go before such a goal can be achieved. In analogy to the original obstacles to classify all simple nuclear C*-algebras [Rørdam 2003; Toms 2008; Villadsen 1999], there are basic structural questions to be settled before a classification theory such as in [Gabe and Szabó 2024] can be attempted on finite C*-algebras. When concerned with just the underlying C*-algebras, this is already a serious challenge. On the one hand, there is the question whether the C*-algebras under consideration automatically satisfy certain properties predicted by classification. For the purpose of this article we highlight the property of Jiang–Su stability. If \mathcal{Z} is the so-called Jiang–Su algebra from [Jiang and Su 1999], then a C^{*}-algebra A is called Jiang–Su stable or \mathcal{Z} -stable when $A \cong A \otimes \mathcal{Z}$. Although this might seem like a technical property at first glance, it becomes natural with more context: Firstly, \mathcal{Z} behaves (as a C^{*}-algebra) very much like an infinite-dimensional version of the complex numbers \mathbb{C} , for instance at the level of K-theory and traces. Secondly, there is by now a pile of evidence that Z-stability holds automatically for many C*-algebras arising from various constructions like the crossed product [Kerr 2020; Kerr and Naryshkin 2021; Kerr and Szabó 2020; Toms and Winter 2013]. The discovery that \mathcal{Z} -stability does in fact *not* hold automatically for all simple nuclear C*-algebras has, among other things, led to the nearly proven Toms–Winter conjecture, which asserts that \mathcal{Z} -stability can only hold or fail in conjunction with some other, a priori different, regularity conditions.

On the other hand, there is the question about precisely what additional structural consequences (not necessarily equivalent characterizations) are shared by Jiang–Su stable C*-algebras, a good example of which is the recent breakthrough work [Castillejos et al. 2021b] (which was in turn continuing work from [Bosa et al. 2019; Matui and Sato 2014a; Sato et al. 2015]). The most novel technical achievement therein can be identified as the *tracial local-to-global principle* for C*-algebras satisfying the so-called uniform property Gamma, which is a weaker assumption than Jiang–Su stability. Said principle concerns the behavior of elements in a given C*-algebra A with respect to the 2-seminorm $\|\cdot\|_{2,\tau}$ induced by

individual tracial states τ on A on the one hand, and the behavior with respect to the uniform tracial 2-norm $\|\cdot\|_u = \sup_{\tau} \|\cdot\|_{2,\tau}$ on the other hand. While the latter is often of interest in the deeper structure and classification theory for C*-algebras, the former can be understood by studying the tracial von Neumann algebra $\pi_{\tau}(A)''$ arising as the weak closure of A under the GNS representation associated to an individual trace τ . In a nutshell, the tracial local-to-global principle asserts that any suitable behavior that can be observed one trace at a time can also be observed uniformly, which often allows one to *transfer*, so to speak, phenomena from von Neumann algebras to the C*-algebraic context. This became in turn the main technical driving force behind the main results in [Castillejos et al. 2021b; 2022], which can be summarized by saying that the Toms–Winter conjecture holds for all simple nuclear C*-algebras having the uniform property Gamma. Whether the latter property automatically holds for simple nuclear nonelementary C*-algebras is presently unknown but is of high interest as it currently represents the main obstacle towards a full solution to the Toms–Winter conjecture.

When we turn our attention to C*-dynamics instead of C*-algebras, we can (and should) study analogous well-behavedness properties as for C*-algebras, one important example of which is equivariant Jiang-Su stability. An action α : $G \cap A$ on a C*-algebra is called (equivariantly) Jiang–Su stable or \mathcal{Z} -stable, if α is cocycle conjugate to $\alpha \otimes id_{\mathcal{Z}}$: $G \curvearrowright A \otimes \mathcal{Z}$. Assuming G is amenable, it is presently open if this happens automatically when A is simple nuclear and \mathcal{Z} -stable; see [Szabó 2021b, Conjecture A]. We note that the analogous question for nonamenable groups is known to have a negative answer [Gardella and Lupini 2021; Jones 1983], although recent insights as in [Gardella et al. 2024; Suzuki 2021] leave some hope for the class of amenable actions of nonamenable groups, which we shall not investigate in this article. If one stresses the point again that \mathcal{Z} essentially looks like an infinite-dimensional version of \mathbb{C} , it should not come as a surprise that we can only expect a classification of G-actions on classifiable C^* -algebras by reasonable invariants if they are equivariantly \mathcal{Z} -stable. The existing work in this direction seems to indicate that equivariant Jiang-Su stability may indeed hold automatically whenever one can reasonably expect it [Gardella et al. 2022; Matui and Sato 2012; 2014b; Sato 2019; Szabó 2018a; Wouters 2023]. The other possible line of investigation, namely further structural consequences of equivariant Jiang–Su stability for group actions, was initiated in [Gardella et al. 2022] as a direct adaption of techniques in [Castillejos et al. 2021b], albeit under rather restrictive assumptions on the actions. We recall one of the key concepts from both said paper and the present work but restrict ourselves in this introduction to the case of unital simple C*-algebras for convenience, despite actually investigating the concept in broader generality.¹

Definition A. Let *G* be a countable discrete group. Let *A* be a separable unital simple C*-algebra such that all 2-quasitraces on *A* are traces and $T(A) \neq \emptyset$. Given a free ultrafilter ω on \mathbb{N} , form the uniform tracial ultrapower A^{ω} of *A*. An action $\alpha : G \cap A$ is said to have equivariant uniform property Gamma if, for any $k \ge 2$, there exist pairwise orthogonal projections $p_1, \ldots, p_k \in (A^{\omega} \cap A')^{\alpha^{\omega}}$ such that

$$\tau(ap_j) = \frac{1}{k}\tau(a) \quad \text{for all } j = 1, \dots, k, \ a \in A, \ \tau \in T_{\omega}(A).$$

¹The reader might consult Definitions 1.7, 1.8, and 2.1 and compare with [Gardella et al. 2022, Definition 3.1].

One can observe easily (Remark 2.2) that equivariant uniform property Gamma is always implied by equivariant Jiang-Su stability. This is important to note because it means that all nontrivial consequences of this property will also hold for Jiang-Su stable actions, and may in fact turn out to be useful for subsequent applications. The most important technical consequence relevant to the present work is the tracial local-to-global principle for C*-dynamical systems over amenable groups. An ad hoc version of this principle appeared in [Gardella et al. 2022, Lemma 4.5] but is only applicable (see [Gardella et al. 2022, Remark 2.2]) for actions that induce an action on tracial states with finite orbits of uniformly bounded size. This assumption appeared not only as a prerequisite for the theory in [Gardella et al. 2022] but is implicitly crucial for the usefulness of the conclusion of this ad hoc principle, which only involves tracial states that are fixed by the action.² In this article we aim to remove any assumptions about how actions $G \curvearrowright A$ are allowed to act on the traces of A, as well as strengthen the conclusion of the tracial local-to-global principle compared to [Gardella et al. 2022], in such a way as to directly generalize and strengthen the known principle for C*-algebras [Castillejos et al. 2021b, Lemma 4.1]. In addition to formulating our result in the language of *-polynomials as all prior papers did, we would also like to promote the following (formally equivalent) formulation of the tracial local-to-global principle for C*-dynamics, which becomes our main technical result. We not only restrict ourselves for the moment to actions on unital simple nuclear C*-algebras (similarly to before) for convenience but give a slightly weaker version here that is easier to state. We treat a stronger version of the statement in broader generality in the main body of the paper; see Theorems 4.2 and 4.6.

Theorem B. Let G be a countable amenable group and A a separable unital simple nuclear C*-algebra. Let $\alpha : G \cap A$ be an action with equivariant uniform property Gamma. Let $\delta : G \cap D$ be an action on a separable C*-algebra and $B \subseteq D$ a δ -invariant C*-subalgebra. Suppose that $\varphi : (B, \delta) \to (A^{\omega}, \alpha^{\omega})$ is an equivariant *-homomorphism. Then φ extends to an equivariant *-homomorphism $\overline{\varphi} : (D, \delta) \to (A^{\omega}, \alpha^{\omega})$ if and only if, for every trace $\tau \in \overline{T_{\omega}(A)}^{w^*}$, there exists a *-homomorphism $\varphi^{\tau} : (D, \delta) \to (\pi_{\tau}^{\alpha^{\omega}}(A^{\omega})'', \alpha^{\omega})$ with $\varphi^{\tau}|_{B} = \pi_{\tau}^{\alpha^{\omega}} \circ \varphi$.³ Here π_{τ} denotes the GNS representation associated to the trace τ , and $\pi_{\tau}^{\alpha^{\omega}}$ denotes the direct sum representation $\bigoplus_{g \in G} \pi_{\tau} \circ \alpha_{g^{-1}}^{\omega}$.

Not unlike previous approaches, the proof of this main result factors through a kind of dynamical version of CPoU (the existence of so-called *complemented partitions of unity*) that we establish along the way; see Lemma 3.2. There are two things to note about this, however. Firstly, the present version of dynamical CPoU does not generally match the property suggested for this purpose in [Gardella et al. 2022], and based on our work we are in fact uncertain whether that property can be expected to hold even under the validity of the above theorem. Secondly, we would like to propose a slight perspective shift by viewing the above local-to-global principle as the primary conceptual property to be studied and exploited instead of the dynamical CPoU, which we feel — especially compared to previous iterations — to be rather unwieldy by itself due to its elaborate technical nature.

²This uses that if $\alpha : G \curvearrowright A$ is assumed to induce an action $G \curvearrowright T(A)$ with finite orbits of uniformly bounded cardinality M > 0, then one has $\|\cdot\|_{2,u} \le M \|\cdot\|_{2,T(A)^{\alpha}}$ as norms on A.

³In actuality one may even allow φ^{τ} to have range in the tracial ultrapower of this von Neumann algebra, but this requires more cumbersome notation to state rigorously.

In the main body of the paper, we actually prove a stronger version of Theorem B for a class of actions on much more general C*-algebras. We would like to comment that the starting point of our theory presumes that the underlying C*-algebra satisfies a kind of *weak CPoU*, namely the one shown to hold for nuclear C*-algebras in [Castillejos et al. 2021b, Lemma 3.6]. Fortunately, a result in the recent preprint [Carrión et al. 2023a] implies that this kind of weak CPoU in fact holds automatically for all C*-algebras with compact tracial state space, which we can use to our advantage.

As for the rest of the paper, we apply Theorem B (or rather Theorems 4.2 and 4.6) to gain insight on equivariant Jiang–Su stability. A famous argument due to Matui and Sato [2012] and the main result of [Szabó 2021b] allows us to argue (as explained in Section 5) that an action α as above is equivariantly Jiang–Su stable if and only if $A \cong A \otimes Z$ and α is uniformly McDuff, i.e., there exist unital *-homomorphisms

$$M_n \to (A^{\omega} \cap A')^{\alpha^{\omega}}$$
 for all $n \in \mathbb{N}$.

Once we note that the latter property is known to hold one trace at a time as a consequence of Ocneanu's theorem [1985] (in the generality we need it, this is imported from [Szabó and Wouters 2024]), the above result can be applied to the α -equivariant inclusion

$$1_n \otimes \mathrm{id}_A : A \to M_n(A)$$

to deduce the following consequence. As before, we note that we prove this result in greater generality than stated here; see Theorem 5.7.

Corollary C (cf. [Gardella et al. 2022, Theorem 7.6]). Let α : $G \curvearrowright A$ be an action of a countable amenable group on a separable unital simple nuclear \mathcal{Z} -stable C*-algebra. Then α has equivariant uniform property Gamma if and only if α is equivariantly Jiang–Su stable.

We expect the main result of this article to have an impact on subsequent applications of equivariant uniform property Gamma or equivariant Jiang–Su stability, in particular in the context of classifying actions on tracial C*-algebras.

As far as potential further research is concerned, let us point out that, for group actions $\alpha : G \cap A$ that are assumed to be "sufficiently free",⁴ the theory pursued in this article can be seen as an instance where one studies uniform property Gamma for the inclusion of C*-algebras $A \subseteq A \rtimes_{\alpha,r} G$ in such a way as to strengthen uniform property Gamma for *A*. It is a tantalizing issue to determine a common framework encompassing all applications of interest regarding uniform property Gamma for more general inclusions of C*-algebras. For instance, it has been hypothesized in past work [Kerr and Szabó 2020, Remark 9.6] that, for a free minimal action $G \cap X$ of an amenable group on a compact metric space, some desirable dynamical properties ought to follow from a different kind of uniform property Gamma for the inclusion $C(X) \subseteq C(X) \rtimes G$, namely the one that strengthens uniform property Gamma for the crossed product; see also [Liao and Tikuisis 2022].

⁴A priori, this may have several different interpretations.

1. Preliminaries

Notation 1.1. Throughout this paper, we will use the following notation and conventions unless specified otherwise:

- By default, *ω* denotes some free ultrafilter on N. At times it can make it easier to state a claim using two free ultrafilters, in which case we denote a second one by *κ*.
- If F is a finite subset inside another set M, we often denote this by $F \subseteq M$.
- \mathbb{K} denotes the compact operators on the Hilbert space $\ell^2(\mathbb{N})$.
- Let A be a C*-algebra. We denote its positive elements by A₊ and its minimal unitization by Ã. We will also make use of its Pedersen ideal, denoted by P(A). We assume the reader is familiar with the basic properties of this object. Given a positive element a ∈ A and ε > 0, we denote by (a − ε)₊ the positive part of the self-adjoint element a − ε1_Ã.
- The topological cone of lower semicontinuous traces on A_+ will be denoted by $\widetilde{T}(A)$; cf. [Elliott et al. 2011]. We call such a trace τ on *A trivial* if it is $\{0, \infty\}$ -valued. It is well known that trivial traces are in one-to-one correspondence with the ideal lattice of *A* by mapping a trivial trace τ to the linear span of $\tau^{-1}(0)$. The set of nontrivial lower semicontinuous traces on A_+ will be denoted by $T^+(A)$ and the set of tracial states will be denoted by T(A). In this paper, we say that a compact subset $K \subset T^+(A)$ is a *compact generator* for $T^+(A)$ if $\mathbb{R}^{>0}K = T^+(A)$.⁵
- In addition, we denote by $Q\widetilde{T}_2(A)$ the set of lower semicontinuous 2-quasitraces (see [Blanchard and Kirchberg 2004, Definition 2.22]) on A, which contains $\widetilde{T}(A)$. We usually only mention them to assume in appropriate contexts that there are no genuine quasitraces, i.e., $Q\widetilde{T}_2(A) = \widetilde{T}(A)$.

We recall the following existence theorem for traces. This follows from a combination of the work of Blackadar and Cuntz [1982, Theorem 1.5] and Haagerup [2014]; see also [Blanchard and Kirchberg 2004, Remark 2.29 (i)].

Theorem 1.2. Let A be a simple, exact C*-algebra such that $A \otimes \mathbb{K}$ contains no infinite projections. Then $Q\widetilde{T}_2(A) = \widetilde{T}(A)$ and $T^+(A) \neq \emptyset$.

In particular, this implies that each stably finite, simple, separable, nuclear C^* -algebra admits a nontrivial trace.

Definition 1.3 [Kirchberg 2006, Definition 1.1; Kirchberg and Rørdam 2014, Definition 4.3]. Let *A* be a C*-algebra with an action $\alpha : G \curvearrowright A$ of a discrete group.

(1) The *ultrapower* of A is defined as

$$A_{\omega} := \ell^{\infty}(A) / \big\{ (a_n)_{n \in \mathbb{N}} \in \ell^{\infty}(A) : \lim_{n \to \omega} \|a_n\| = 0 \big\}.$$

(2) Pointwise application of α on representing sequences induces an action on the ultrapower, which we will denote by $\alpha_{\omega}: G \curvearrowright A_{\omega}$.

⁵In case A is simple, an example of such a compact generator is given by $\{\tau \in T^+(A) \mid \tau(a) = 1\}$ for some $a \in \mathcal{P}(A)_+ \setminus \{0\}$.

(3) There is a natural inclusion $A \subset A_{\omega}$ by identifying an element of A with its constant sequence. Define

 $A_{\omega} \cap A' := \{x \in A_{\omega} \mid [x, A] = 0\}$ and $A_{\omega} \cap A^{\perp} := \{x \in A_{\omega} \mid xA = Ax = 0\}.$

The quotient

$$F_{\omega}(A) := (A_{\omega} \cap A') / (A_{\omega} \cap A^{\perp})$$

is called the (*corrected*) central sequence algebra. If A is σ -unital, then $F_{\omega}(A)$ is unital, where the unit is represented by a sequential approximate unit $(e_n)_{n \in \mathbb{N}}$.

(4) Since A is α_{ω} -invariant, so are $A_{\omega} \cap A'$ and $A_{\omega} \cap A^{\perp}$. Thus, α_{ω} induces an action on $F_{\omega}(A)$, which we will denote by $\tilde{\alpha}_{\omega} : G \curvearrowright F_{\omega}(A)$.

Definition 1.4. Let A be a C*-algebra. A sequence of tracial states $(\tau_n)_{n \in \mathbb{N}}$ on A defines a trace on A_{ω} via

$$[(a_n)_{n\in\mathbb{N}}]\mapsto \lim_{n\to\omega}\tau_n(a_n).$$

A trace of this form is called a *limit trace*. The set of all limit traces on A_{ω} will be denoted by $T_{\omega}(A)$. More generally, following [Szabó 2021b, Definition 2.1], a sequence $(\tau_n)_{n \in \mathbb{N}}$ in $\widetilde{T}(A)$ defines a lower semicontinuous trace $\tau : \ell^{\infty}(A)_+ \to [0, \infty]$ by

$$\tau((a_n)_{n\in\mathbb{N}}) = \sup_{\varepsilon>0} \lim_{n\to\omega} \tau_n((a_n-\varepsilon)_+).$$

This trace is the lower semicontinuous regularization of the trace given by $\lim_{n\to\omega} \tau_n(a_n)$; see [Elliott et al. 2011, Lemma 3.1]. This regularization ensures that $\tau((a_n)_{n\in\mathbb{N}}) = 0$ if $\lim_{n\to\omega} ||a_n|| = 0$, so τ also induces a lower semicontinuous trace on A_{ω} . A trace of this form on A_{ω} is called a *generalized limit trace*. The set of all generalized limit traces is denoted by $\widetilde{T}_{\omega}(A)$.

For the next part, assume A is separable. For any $a \in A_+$ and $\tau \in \widetilde{T}_{\omega}(A)$, we can define a trace

$$\tau_a: (A_{\omega} \cap A')_+ \to [0,\infty], \quad x \mapsto \tau(ax).$$

We have that $\tau_a(x) \le ||x|| \tau(a)$, so this trace is bounded whenever $\tau(a) < \infty$. Note that this trace also induces a trace on $F_{\omega}(A)$, which by abuse of notation will also be denoted by τ_a . Clearly this yields a tracial state under the assumption $\tau(a) = 1$. Let us say that a given tracial state τ on $F_{\omega}(A)$ is a *canonical trace* if it belongs to the weak-*-closed convex hull of { $\tau_a | \tau \in \widetilde{T}_{\omega}(A), a \in A_+, \tau(a) = 1$ }.

Remark 1.5. We point out that it is not necessary to consider generalized limit traces in an important subcase that often occurs in the literature. Namely, assume *A* is a separable simple C*-algebra with $\emptyset \neq T^+(A) = \mathbb{R}^{>0}T(A)$ such that T(A) is compact.⁶ Then it follows by [Castillejos and Evington 2021, Proposition 2.3] that every generalized limit trace $\tau \in \widetilde{T}_{\omega}(A)$ that is finite on some nonzero positive element of *A* is a multiple of an ordinary limit trace.

Next we recall how various versions of tracial ultrapowers are defined.

⁶For instance, this is automatic when A is separable, simple, nuclear, unital, and stably finite.

Definition 1.6 [Ando and Haagerup 2014, Propositions 3.1 and 3.2]. Suppose \mathcal{M} is a finite von Neumann algebra with faithful normal tracial state τ . Then the *tracial von Neumann algebra ultrapower* is defined as

$$\mathcal{M}^{\omega} := \ell^{\infty}(\mathcal{M}) / \big\{ (x_n)_{n \in \mathbb{N}} \in \ell^{\infty}(\mathcal{M}) \mid \lim_{n \to \infty} \|x_n\|_{2,\tau} = 0 \big\}.$$
(1-1)

This is again a von Neumann algebra with a faithful normal tracial state τ^{ω} that is defined on representative sequences by $\tau^{\omega}((x_n)_{n \in \mathbb{N}}) = \lim_{n \to \omega} \tau(x_n)$.

The notation used for the tracial von Neumann algebra ultrapower is the same as for the uniform tracial ultrapower of a suitable C^* -algebra as defined below. It will be clear from context which of the two notions we use. In the special case that A is a C^* -algebra with unique tracial state τ and no unbounded traces, the uniform tracial ultrapower A^{ω} is naturally isomorphic, by Kaplansky's density theorem, to the von Neumann tracial ultrapower $(\pi_{\tau}(A)'')^{\omega}$, where π_{τ} denotes the GNS representation associated to τ . Note that on a tracial von Neumann algebra (\mathcal{M}, τ) , the topology induced by the $\|\cdot\|_{2,\tau}$ -norm agrees with the *-strong operator topology on bounded subsets. So equivalently, in (1-1) one can quotient out by the sequences that converge to 0 in the *-strong operator topology, which makes the construction equivalent to the Ocneanu ultrapower; cf. [Ando and Haagerup 2014].

Definition 1.7. Let *A* be a C*-algebra. Given a constant $p \ge 1$ and $\tau \in T(A)$, we define a seminorm $\|\cdot\|_{p,\tau}$ on *A* by

$$||a||_{p,\tau} = \tau (|a|^p)^{1/p}, \quad a \in A.$$

We will in particular appeal to the cases p = 1 or p = 2 subsequently. For a nonempty set $X \subset T(A)$, we define a seminorm $\|\cdot\|_{2,X}$ on A by

$$||a||_{2,X} := \sup_{\tau \in X} ||a||_{2,\tau}$$

for all $a \in A$. The seminorm $\|\cdot\|_{2,T(A)}$ is also denoted by $\|\cdot\|_{2,u}$. This is a norm if and only if, for all nonzero $a \in A$, there exists some $\tau \in T(A)$ such that $\tau(a^*a) > 0$, which is in particular the case when A is simple with T(A) nonempty.

We note that in the construction below, we deviate from other sources by making a very explicit distinction in terminology between C^* -algebras that do or do not admit nontrivial unbounded traces.

Definition 1.8 (cf. [Castillejos et al. 2021b, Section 1.3]⁷). Let A be a C*-algebra with $T(A) \neq \emptyset$. Then the *trace-kernel ideal (with respect to bounded traces)* inside A_{ω} is defined by

$$J_A^{\mathsf{b}} := \left\{ [(a_n)_{n \in \mathbb{N}}] \in A_{\omega} \mid \lim_{n \to \omega} ||a_n||_{2, T(A)} = 0 \right\}.$$

The uniform bounded tracial ultrapower is defined as the quotient

$$A^{\omega,\mathrm{b}} := A_{\omega}/J_A^{\mathrm{b}}$$

Whenever $\|\cdot\|_{2,T(A)}$ defines a norm on A, there also exists a canonical embedding of A into $A^{\omega,b}$. Then $A^{\omega,b} \cap A'$ is called the *uniform bounded tracial central sequence algebra*. Whenever we have an action

 $^{^{7}}$ The cited source assumes separability, but we generalize the definition beyond that case.

 $\alpha : G \frown A$ of a discrete group, the ideal J_A^b is α_{ω} -invariant. Hence, there is an induced action on the uniform bounded tracial ultrapower, which we will denote by $\alpha^{\omega} : G \frown A^{\omega,b}$.

Clearly, every limit trace vanishes on J_A^b and hence also induces a tracial state on $A^{\omega,b}$. We will also use $T_{\omega}(A)$ to denote the collection of limit traces on $A^{\omega,b}$. Note that

$$J_A^{\mathsf{b}} = \{ x \in A_{\omega} : \|x\|_{2, T_{\omega}(A)} = 0 \},\$$

so in particular $\|\cdot\|_{2,T_{\omega}(A)}$ defines a norm on $A^{\omega,b}$.

Finally, if we assume A is a simple C*-algebra such that $Q\widetilde{T}_2(A) = \widetilde{T}(A)$ and $\emptyset \neq T^+(A) = \mathbb{R}^{>0} \cdot T(A)$ with T(A) compact, then we simply call $J_A = J_A^b$ the trace-kernel ideal, $A^{\omega} = A^{\omega,b}$ the uniform tracial ultrapower, and $A^{\omega} \cap A'$ the uniform tracial central sequence algebra.

Remark 1.9. Our choice to add the extra "bounded" in the terminology above and the extra letter "b" in the notation, which is usually not included in other sources such as the ones we cite, is deliberate and has the purpose to not overuse the word "uniform", in particular in cases where it becomes rather misleading. This is most apparent for nonsimple C*-algebras; if *B* is any unital simple C*-algebra with $T(B) \neq \emptyset$, then the above construction applied to $A = B \oplus \mathbb{K}$ yields $A^{\omega,b} = B^{\omega}$ by virtue of the fact that the canonical trace on \mathbb{K} is unbounded. Since one of the two tracial direct summands is entirely forgotten in this construction, this object seems unfit to be called "uniform tracial". However, even the case of simple C*-algebras is enough to illustrate why one should not equate $A^{\omega,b}$ with the object capturing all "uniform tracial" data. Namely, the range result [Gong and Lin 2022] combined with a little playing around with invariants allows one to see that, given any metrizable Choquet simplex *S* with $\partial_e S$ admitting some isolated point, there exists a (nonunital) classifiable C*-algebra *A* such that $T^+(A)$ has a Choquet base affinely homeomorphic to *S*, yet *A* has a unique tracial state τ . In this scenario, we have $A^{\omega,b} \cong (\pi_{\tau}(A)'')^{\omega} \cong \mathbb{R}^{\omega}$ as a consequence of Connes' theorem. So despite *A* having a rich tracial structure, the only trace captured by this construction is τ , which compels us to not apply the word "uniform" or the notation " A^{ω} " to such an example.

Note that the phenomenon discussed here is also what motivated us to subsequently revise the definition of (equivariant) uniform property Gamma in the spirit of [Castillejos and Evington 2021], as well as introduce an auxiliary version of it that explicitly only takes into account tracial states, even when the surrounding C*-algebra may have other unbounded traces.

Remark 1.10. Let A be a σ -unital C*-algebra with $T(A) \neq \emptyset$. By [Castillejos et al. 2021b, Proposition 1.11] the uniform bounded tracial ultrapower $A^{\omega,b}$ is unital if and only if T(A) is compact.⁸ Moreover, [Castillejos et al. 2021b, Lemma 1.10] shows that, in that case, the natural map $A_{\omega} \cap A' \to A^{\omega,b} \cap A'$ factors through $F_{\omega}(A)$. In case A is separable, this natural map is surjective by a combination of Propositions 4.5 (iii) and 4.6 in [Kirchberg and Rørdam 2014] (the unitality hypothesis in the second cited proposition is not needed, as it suffices to take a unit in the minimal unitization for the proof).

As we have argued above, there are some issues if one is trying to define the object A^{ω} for a C^{*}-algebra A that may possess many unbounded traces. In fact, trying to find a viable general definition that has the

⁸The cited statement assumes separability of A, but a closer look at the proof shows that σ -unitality is sufficient.

same level of utility as in the case of unital C*-algebras has eluded a number of researchers for years. By introducing the next few definitions and observations, however, we wish to promote the viewpoint that there is a rather natural way to define the object $A^{\omega} \cap A'$ for any separable C*-algebra A, even if we do not know at present how to properly define the object A^{ω} itself. Since we are unsure of the viability of this definition when A admits genuine quasitraces, we wish to be cautious and shall only define the concepts below under the assumption that A does not admit them.⁹

Definition 1.11. Let *A* be a separable C*-algebra with $Q\widetilde{T}_2(A) = \widetilde{T}(A)$ and $T^+(A) \neq \emptyset$. The *trace-kernel ideal* \mathcal{J}_A inside $F_{\omega}(A)$ is defined as the set of elements $x \in F_{\omega}(A)$ such that, for every generalized limit trace $\tau \in \widetilde{T}_{\omega}(A)$ and $a \in A_+$ with $0 < \tau(a) < \infty$, we have $\tau_a(x^*x) = 0$. With some abuse of notation, we denote the quotient by

$$A^{\omega} \cap A' = F_{\omega}(A) / \mathcal{J}_A. \tag{1-2}$$

It is clear from construction that a canonical trace on $F_{\omega}(A)$ vanishes on \mathcal{J}_A , so it descends to a tracial state on $A^{\omega} \cap A'$. As before, we call a given tracial state on $A^{\omega} \cap A'$ a *canonical trace* if it is induced by a canonical trace on $F_{\omega}(A)$, or equivalently if it belongs to the weak-*-closed convex hull of the tracial states τ_a on $A^{\omega} \cap A'$, where $\tau \in \widetilde{T}_{\omega}(A)$ and $a \in A_+$ with $\tau(a) = 1$.

If $\alpha : G \curvearrowright A$ is an action of a discrete group with induced action $\tilde{\alpha}_{\omega} : G \curvearrowright F_{\omega}(A)$, then clearly \mathcal{J}_A is $\tilde{\alpha}_{\omega}$ -invariant, so we obtain an induced action $\alpha^{\omega} : G \curvearrowright A^{\omega} \cap A'$.

Remark 1.12. In the case that A is simple, and $Q\widetilde{T}_2(A) = \widetilde{T}(A)$ and $\emptyset \neq T^+(A) = \mathbb{R}^{>0}T(A)$ with T(A) compact, it follows from Remarks 1.5 and 1.10 that $F_{\omega}(A)/\mathcal{J}_A = A^{\omega,b} \cap A'$, so the notation $A^{\omega} \cap A'$ is consistent with the last part of Definition 1.8.

Remark 1.13 (see remark after [Kirchberg and Rørdam 2014, Definition 4.3]). Let $p \ge 1$ be any constant. Given any element x in a C*-algebra B with a tracial state θ , one has the inequalities

$$\|x\|_{1,\theta} \le \|x\|_{p,\theta} \le \|x\|_{1,\theta}^{1/p} \|x\|^{1-1/p}$$

This implies that an element in either Definition 1.8 or 1.11 belongs to the trace-kernel ideal if and only if its tracial *p*-norms vanish with respect to the appropriately chosen (limit) traces. We will frequently use this without further mention for p = 1.

Remark 1.14. One of Kirchberg's initial observations about $F_{\omega}(A)$, which attests to the naturality of its construction, is that it is a stable invariant. We are about to argue that the same applies to the construction $A \mapsto A^{\omega} \cap A'$. For this purpose, let $\{e_{k,\ell} \mid k, \ell \ge 1\}$ be a set of matrix units generating \mathbb{K} , and let $1_n \in \mathbb{K}$ be the increasing approximate unit given by $1_n = \sum_{j=1}^n e_{j,j}$. We recall (see [Kirchberg 2006, Proposition 1.9, Corollary 1.10]) that there is a canonical isomorphism $\theta : F_{\omega}(A) \to F_{\omega}(A \otimes \mathbb{K})$ defined as follows: given an element $x \in F_{\omega}(A)$ represented by a central sequence $(x_n)_{n \in \mathbb{N}}$ in A, it is sent to the element $\theta(x)$ represented by the central sequence $(x_n \otimes 1_n)_{n \in \mathbb{N}}$.

 $^{^{9}}$ At the same time, we note that the concepts make sense formally anyway, and none of the subsequent arguments hinge on the assumption that *A* does not admit genuine quasitraces.

Proposition 1.15. Let A be a separable C*-algebra with $Q\widetilde{T}_2(A) = \widetilde{T}(A)$ and $T^+(A) \neq \emptyset$. Then the canonical isomorphism $F_{\omega}(A) \cong F_{\omega}(A \otimes \mathbb{K})$ preserves the canonical traces on both sides. Consequently, it descends to a canonical isomorphism

$$A^{\omega} \cap A' \cong (A \otimes \mathbb{K})^{\omega} \cap (A \otimes \mathbb{K})'.$$

Proof. As we set up before the proposition, we denote the canonical isomorphism by θ . It is clear that it induces an affine homeomorphism between all tracial states on $F_{\omega}(A)$ and on $F_{\omega}(A \otimes \mathbb{K})$ via $\tau \mapsto \tau \circ \theta^{-1}$. The claim amounts to showing that the image of the canonical traces on the left is equal to the canonical traces on the right.

Let Tr be the unique lower semicontinuous trace on \mathbb{K} with $\operatorname{Tr}(e_{1,1}) = 1$. We keep in mind that the assignment $\widetilde{T}(A) \to \widetilde{T}(A \otimes \mathbb{K})$ given by $\tau \mapsto \tau \otimes \operatorname{Tr}$ is an affine homeomorphism. Given a generalized limit trace $\tau \in \widetilde{T}_{\omega}(A)$ induced by a sequence $(\tau_n)_{n \in \mathbb{N}}$ in $\widetilde{T}(A)$, let us denote by $\tau^s \in \widetilde{T}_{\omega}(A \otimes \mathbb{K})$ the generalized limit trace induced by the sequence $(\tau_n \otimes \operatorname{Tr})_{n \in \mathbb{N}}$ in $\widetilde{T}(A \otimes \mathbb{K})$. Clearly the assignment $\tau \mapsto \tau^s$ is also a bijection between generalized limit traces. Let such a generalized limit trace τ be given on A_{ω} . Given $a \in (A \otimes \mathbb{K})_+$, we can write $a = \sum_{k,\ell=1}^{\infty} a_{k,\ell} \otimes e_{k,\ell}$ for uniquely determined elements $a_{k,\ell} \in A$. It then follows from [Castillejos and Evington 2021, Proposition 2.9] that we have a norm-convergent sum expression

$$\tau_a^s \circ \theta = \sum_{\ell=1}^{\infty} \tau_{a_{\ell,\ell}}.$$
(1-3)

Applied to $a = b \otimes e_{1,1}$ for some $b \in A_+$ with $\tau(b) = 1$, this gives $\tau_b \circ \theta^{-1} = \tau_a^s$. From this we can infer that canonical traces are mapped to canonical traces. The general expression (1-3) applied to $a \in (A \otimes \mathbb{K})_+$ with $\tau^s(a) = 1$ shows that we have a bijection.

We give two more technical lemmas that will be useful later on.

Lemma 1.16. Let A be a C*-algebra with positive element $a \in A_+$. Let $(\varepsilon_n)_{n \in \mathbb{N}}$ be a sequence of positive constants such that $\lim_{n\to\infty} \varepsilon_n = 0$, and let $(b_n)_{n\in\mathbb{N}}$ be a sequence of positive elements such that $\|b_n - a\| < \varepsilon_n$. For each $\tau \in \widetilde{T}_{\omega}(A)$ and $c \in F_{\omega}(A)$, one has

$$\lim_{n\to\infty}\tau_{(b_n-\varepsilon_n)_+}(c)=\tau_a(c).$$

Proof. For each $n \in \mathbb{N}$, there exists a contraction $d_n \in A$ such that $(b_n - \varepsilon_n)_+ = d_n a d_n^*$ by [Kirchberg and Rørdam 2002, Lemma 2.2]. This implies that $\tau_{(b_n - \varepsilon_n)_+}(c) \leq \tau_a(c)$. Since the sequence $(b_n - \varepsilon_n)_+$ converges to a and τ is lower semicontinuous, this leads to the desired result.

Lemma 1.17. Let A be a simple C*-algebra with $a \in \mathcal{P}(A)_+ \setminus \{0\}$, and let K be a compact generator for $T^+(A) \neq \emptyset$. Take a generalized limit trace $\tau \in \widetilde{T}_{\omega}(A)$ such that $0 < \tau(a) < \infty$. Then there exists a sequence $(\theta_n)_{n \in \mathbb{N}}$ in K such that the associated generalized limit trace θ on A_{ω} is a scalar multiple of τ and such that, for each sequence $(b_n)_{n \in \mathbb{N}}$ representing an element of A_{ω} , we have that

$$\theta((ab_n)_{n\in\mathbb{N}}) = \lim_{n\to\omega} \theta_n(ab_n).$$
(1-4)

Proof. The fact that there exists a sequence $(\theta_n)_{n \in \mathbb{N}}$ in *K* such that the associated generalized limit trace is a multiple of τ follows directly from [Szabó 2021b, Lemma 2.10 and Remark 2.11]. Let $B := \overline{a^{1/2}Aa^{1/2}}$ denote the hereditary subalgebra generated by $a^{1/2}$. Then $B \subseteq \mathcal{P}(A)$; see for example [Pedersen 1979, Proposition 5.6.2]. Since *K* is compact, we claim that $\sup_{\sigma \in K} ||\sigma|_B || < \infty$. Suppose that this would not be the case; then, for all $n \in \mathbb{N}$, we could find a $\sigma_n \in K$ and a positive contraction $d_n \in B$ such that $\sigma_n(d_n) \ge n2^n$. Consider $d := \sum_{n=1}^{\infty} 2^{-n} d_n \in B$ and $\sigma = \lim_{n \to \omega} \sigma_n \in K$ (using the compactness of *K*). Then, for each $n \in \mathbb{N}$, we would get

$$\sigma(d) = \lim_{n \to \omega} \sigma_n(d) \ge \lim_{n \to \omega} \sigma_n(2^{-n}d_n) = \infty$$

but this is a contradiction, since *d* belongs to the Pedersen ideal. As a consequence we get that, when restricted to the hereditary subalgebra $\overline{a^{1/2}\ell^{\infty}(A)a^{1/2}} \subseteq \ell^{\infty}(B)$, the trace formed by $\lim_{n\to\omega} \theta_n$ is already bounded and hence continuous, so formula (1-4) holds.

The following proposition is a useful lifting property in various contexts. The proof relies on the concept of G- σ -ideals; see [Szabó 2018c, Definition 4.1]. Let $\alpha : G \curvearrowright A$ and $\beta : G \curvearrowright B$ be actions on C*-algebras. As in [Kirchberg 2006], we call an equivariant surjective *-homomorphism $\pi : (A, \alpha) \rightarrow (B, \beta)$ strongly locally semisplit, if for every separable β -invariant C*-subalgebra $D \subseteq B$, there exists an equivariant c.p.c. order-zero map $\phi : (D, \beta) \rightarrow (A, \alpha)$ such that $\pi \circ \phi = id_D$.

Proposition 1.18. Let A be a separable simple C*-algebra with $Q\tilde{T}_2(A) = \tilde{T}(A)$ and $T^+(A) \neq \emptyset$. Let $\alpha : G \frown A$ be an action of a countable discrete group. Then the quotient map

$$(F_{\omega}(A), \tilde{\alpha}_{\omega}) \to (A^{\omega} \cap A', \alpha^{\omega})$$

is strongly locally semisplit.

Proof. By [Szabó 2018c, Proposition 4.5 (ii)], it suffices to prove that $\mathcal{J}_A \subset F_\omega(A)$ is a G- σ -ideal. Fix an element $0 \neq a \in \mathcal{P}(A)_+$ and a compact generator $K \subset T^+(A)$.¹⁰ By [Szabó 2021b, Proposition 2.4] and Lemma 1.16, we can conclude that \mathcal{J}_A coincides with the ideal of those elements $x \in F_\omega(A)$ such that $\tau_a(x^*x) = 0$ for all $\tau \in \widetilde{T}_\omega(A)$ induced by any sequence $\tau_n \in K$. Since *a* belongs to the Pedersen ideal and *K* is compact, this further implies that an element $x \in F_\omega(A)$ represented by a sequence $(x_n)_{n \in \mathbb{N}}$ in *A* belongs to \mathcal{J}_A precisely when $\lim_{n \to \omega} \max_{\tau \in K} \tau(a^{1/2}x_n^*x_na^{1/2}) = 0$.

We proceed to show that \mathcal{J}_A is a G- σ -ideal. Let $D \subset F_{\omega}(A)$ be a separable $\tilde{\alpha}_{\omega}$ -invariant C*-subalgebra. Let $(d_{k,n})_{n,k\in\mathbb{N}}$ and $(c_{k,n})_{n,k\in\mathbb{N}}$ be two bounded double sequences in A such that, for each $k \in \mathbb{N}$, the sequences $(d_{k,n})_{n\in\mathbb{N}}$ and $(c_{k,n})_{n\in\mathbb{N}}$ are approximately central, the set $\{d^{(k)} = [(d_{k,n})_{n\in\mathbb{N}}] \mid k \in \mathbb{N}\}$ defines a dense subset in the unit ball of D, and the set $\{c^{(k)} = [(c_{k,n})_{n\in\mathbb{N}}] \mid k \in \mathbb{N}\}$ defines a dense subset in the unit ball of $D \cap \mathcal{J}_A$. By Kasparov's lemma [1988, Lemma 1.4], we can find, for any $\varepsilon > 0$, $F \Subset G$, and $m \in \mathbb{N}$, a positive element $e \in \mathcal{J}_A$ such that

$$\max_{k \le m} \|[e, d^{(k)}]\| \le \varepsilon, \quad \max_{k \le m} \|(1 - e)c^{(k)}\| \le \varepsilon, \quad \text{and} \quad \max_{g \in F} \|e - \tilde{\alpha}_{\omega}(e)\| \le \varepsilon.$$

¹⁰As pointed out in the footnote after defining compact generators in Notation 1.1, this always exists as a consequences of simplicity.

Let $b \in A$ be a strictly positive contraction. If we represent e by an approximately central sequence $(e_n)_{n \in \mathbb{N}}$ of positive contractions in *A*, then it follows that

$$\max_{k \le m} \lim_{n \to \omega} \|[e_n, d_{k,n}]b\| \le \varepsilon, \quad \max_{k \le m} \lim_{n \to \omega} \|(1 - e_n)c_{k,n}b\| \le \varepsilon,$$

and

$$\max_{g \in F} \lim_{n \to \omega} \|(e_n - \alpha_g(e_n))b\| \le \varepsilon, \quad \lim_{n \to \omega} \max_{\tau \in K} \tau(a^{1/2}e_na^{1/2}) = 0.$$

Appealing to Kirchberg's ε -test [Kirchberg and Rørdam 2014, Lemma 3.1], we can find another approximately central sequence $(e_n)_{n \in \mathbb{N}}$ of positive contractions in A satisfying the stronger property

 $\lim_{n \to \omega} \max_{\tau \in K} \tau(a^{1/2} e_n a^{1/2}) = 0$ $\lim_{n \to \infty} (\|[e_n, d_{k,n}]b\| + \|(1 - e_n)c_{k,n}b\| + \|(e_n - \alpha_g(e_n))b\|) = 0 \text{ and }$

for all $k \in \mathbb{N}$ and $g \in G$. This means that this sequence represents a positive contraction $e \in (\mathcal{J}_A \cap D')^{\tilde{\alpha}_{\omega}}$ such that ec = c for all $c \in \mathcal{J}_A \cap D$. This finishes the proof.

To end this preliminary section, we prove the following tracial inequality.

Lemma 1.19. Let B be a C^{*}-algebra with $a, b \in B_+$ and $\tau \in T(B)$. Then

$$||a-b||_{2,\tau}^2 \le ||a^2-b^2||_{1,\tau}$$

Proof. If we replace B by its weak closure of the GNS representation $\pi_{\tau}(B)''$, it is enough to show this in the case that B is a von Neumann algebra with faithful normal tracial state τ .

Historically, this was proved by Powers and Størmer [1970, Lemma 4.1] in the case $B = M_n(\mathbb{C})$ for some $n \in \mathbb{N}$. When B is a von Neumann algebra with faithful normal tracial state τ , this follows from applying [Haagerup 1975, Lemma 2.10], which is formulated for the space $L^2(B, \tau)$, to elements in $B \subset L^2(B, \tau)$. For the reader's convenience we give here a more direct proof using an idea from [Anantharaman and Popa 2014, Theorem 7.3.7].

Given $a, b \in B_+$, let p and q denote the spectral projections of a - b corresponding to $[0, +\infty)$ and $(-\infty, 0)$, respectively. This means that a - b = (p - q)|a - b| and $p \perp q$. First of all, we have

$$\tau((a^2 - b^2)p) - \tau((a - b)^2p) = \tau(b(a - b)p) + \tau((a - b)bp) = 2\tau(b^{1/2}(a - b)pb^{1/2}) \ge 0$$

since (a - b)p > 0. So we get

$$\tau((a-b)^2 p) \le \tau((a^2 - b^2)p), \tag{1-5}$$

and in a similar way we can obtain that

$$\tau((b-a)^2 q) \le \tau((b^2 - a^2)q).$$
(1-6)

Combining (1-5) and (1-6) gives

$$\tau((a-b)^2) = \tau((a-b)^2(p+q)) \le \tau((a^2-b^2)(p-q)).$$

Also,

$$\tau((a^2 - b^2)(p - q)) = \tau(p(a^2 - b^2)p) + \tau(q(b^2 - a^2)q) \le \tau(|a^2 - b^2|(p + q)) \le ||a^2 - b^2||_{1,\tau}$$

we $||p + q|| < 1$. This implies the result.

since $||p+q|| \le 1$. This implies the result.

2. Equivariant uniform property Gamma

The notion of uniform property Gamma was introduced in [Castillejos et al. 2021b] and further studied in [Castillejos et al. 2022], where it served as a uniform C*-algebraic version of property Gamma introduced by Murray and von Neumann for II₁ factors [1943]. Recently, a dynamical version of this property was introduced in the separable unital setting in [Gardella et al. 2022] called the *equivariant uniform property Gamma*. Here we revise the definition to account for separable C*-algebras with possibly unbounded traces, generalizing the concept called "stabilised property Gamma" by Castillejos and Evington [2021, Definition 2.5]. We choose not to adopt that name because one can argue that uniform property Gamma ought to be a stable property in the first place, just like property Gamma is for von Neumann algebras. In light of recent work by Lin [2023] who proposed a more general framework for C*-algebras that admit genuine quasitraces, we shall state the definition only in the absence of such.

For separable unital simple exact C*-algebras, the definition below corresponds to the earlier definition given in [Gardella et al. 2022] (see Proposition 2.4 below) but not in the nonsimple case, as demonstrated by C*-algebras that arise as extensions of unital classifiable C*-algebras by the compacts (see the type of example mentioned in Remark 1.9, for instance).

Definition 2.1. Let *A* be a separable C*-algebra with $Q\widetilde{T}_2(A) = \widetilde{T}(A)$ and $T^+(A) \neq \emptyset$, and let $\alpha : G \frown A$ be an action by a countable discrete group. We say that α has *equivariant uniform property Gamma* (or *equivariant property Gamma* for short) if, for all $n \in \mathbb{N}$, there exist pairwise orthogonal projections $p_1, \ldots, p_n \in (A^{\omega} \cap A')^{\alpha^{\omega}}$ such that, for all $a \in A_+$ and $\tau \in \widetilde{T}_{\omega}(A)$ with $\tau(a) < \infty$,

$$\tau_a(p_i) = \frac{1}{n}\tau(a).$$

Remark 2.2. We can notice immediately from the naturality of the isomorphism in Proposition 1.15 that equivariant uniform property Gamma is preserved under stable cocycle conjugacy. That is, if *A* and *B* are C*-algebras as above and we have actions $\alpha : G \curvearrowright A$ and $\beta : G \curvearrowright B$ such that $\alpha \otimes id_{\mathbb{K}}$ is cocycle conjugate to $\beta \otimes id_{\mathbb{K}}$, then α^{ω} is conjugate to β^{ω} via a map preserving the canonical traces. In particular, equivariant uniform property Gamma holds for α if and only if it holds for β .

Next, we observe (cf. [Castillejos et al. 2021b, Proposition 2.3]) that, whenever $\alpha : G \curvearrowright A$ is an equivariantly \mathcal{Z} -stable action on a separable C*-algebra with $Q\widetilde{T}_2(A) = \widetilde{T}(A)$ and $T^+(A) \neq \emptyset$, it automatically has equivariant property Gamma. Indeed, a cocycle conjugacy between α and $\alpha \otimes id_{\mathcal{Z}}$ is easily seen to give rise to a unital *-homomorphism

$$\mathcal{Z}^{\omega} \cap \mathcal{Z}' \to (A^{\omega} \cap A')^{\alpha^{\omega}}.$$

For this purpose one chooses an approximate unit $e_n \in A$ and considers a sequence of maps $\mathbb{Z} \to A \otimes \mathbb{Z}$, $x \mapsto e_n \otimes x$, composed with such a cocycle conjugacy, which is seen to induce such a homomorphism. It is well known that $\mathbb{Z}^{\omega} \cap \mathbb{Z}'$ admits unital embeddings of matrix algebras of arbitrary size $n \ge 2$. So if we fix *n* and define $p_1, \ldots, p_n \in (A^{\omega} \cap A')^{\alpha^{\omega}}$ as the image of the canonical rank-one projections inside a matrix algebra under the aforementioned *-homomorphism, then they satisfy the necessary requirements for equivariant property Gamma by uniqueness of the trace on the $n \times n$ matrices. The following is a version of equivariant property Gamma for possibly nonseparable C^{*}-algebras that exclusively takes into account the bounded traces. This agrees with [Gardella et al. 2022, Definition 3.1] for separable unital C^{*}-algebras but not with the general definition of equivariant property Gamma given above.

Definition 2.3. Let *A* be a σ -unital C*-algebra with *T*(*A*) nonempty and compact, and let $\alpha : G \cap A$ be an action of a countable discrete group. We say that α has *local equivariant property Gamma with respect to bounded traces* if, for all $n \in \mathbb{N}$ and $\|\cdot\|_{2,T_{\omega}(A)}$ -separable subsets $S \subset A^{\omega,b}$, there exist pairwise orthogonal projections $p_1, \ldots, p_n \in (A^{\omega,b})^{\alpha^{\omega}} \cap S'$ such that $\tau(ap_i) = \tau(a)/n$ for all $a \in S$ and $\tau \in T_{\omega}(A)$.

In the unital separable simple setting, Definitions 2.1 and 2.3 are equivalent. We prove this fact in a slightly more general setting in the proposition below.

Proposition 2.4. Let A be a simple separable C*-algebra with $Q\widetilde{T}_2(A) = \widetilde{T}(A)$, and such that $T(A) \neq \emptyset$ is compact and $T^+(A) = \mathbb{R}^{>0}T(A)$.¹¹ Then an action $\alpha : G \curvearrowright A$ of a countable discrete group has equivariant property Gamma if and only if α has local equivariant property Gamma with respect to bounded traces.

Proof. The assumptions on *A* imply that every generalized limit trace on *A* that is finite on some nonzero positive element of *A* is a multiple of an ordinary limit trace (see Remark 1.5) and that $A^{\omega} \cap A' = A^{\omega,b} \cap A'$ (see Remark 1.12). Therefore, it suffices to show that the existence of pairwise orthogonal projections $p_1, \ldots, p_n \in (A^{\omega} \cap A')^{\alpha^{\omega}}$ such that $\tau(ap_i) = \tau(a)/n$ for all $a \in A$ and $\tau \in T_{\omega}(A)$ implies, for any $\|\cdot\|_{2,T_{\omega}(A)}$ -separable $S \subset A^{\omega}$, the existence of pairwise orthogonal projections $p'_1, \ldots, p'_n \in (A^{\omega} \cap S')^{\alpha^{\omega}}$ such that $\tau(ap'_i) = \tau(a)/n$ for all $a \in S$ and $\tau \in T_{\omega}(A)$. This follows by a standard reindexation argument, which we omit.

The next part of this section is devoted to proving an equivalence between equivariant property Gamma for an action $\alpha : G \curvearrowright A$ and local equivariant property Gamma with respect to bounded traces for its induced action $\alpha^{\omega} : G \curvearrowright A^{\omega} \cap A'$, at least in the setting when A is simple nuclear and has stable rank one.¹² Recall that A is said to have stable rank one if the invertibles of \tilde{A} are dense in \tilde{A} . We start by observing the following description of the tracial state space of $A^{\omega} \cap A'$.

Proposition 2.5. Let A be a separable, simple, nuclear C^{*}-algebra with uniform property Gamma and stable rank one. Then every tracial state on $A^{\omega} \cap A'$ is a canonical trace, i.e., one has

$$T(A^{\omega} \cap A') = \overline{\operatorname{conv}}^{w^*} \{ \tau_a \mid \tau \in \widetilde{T}_{\omega}(A), \, a \in A_+, \, \tau(a) = 1 \}.$$

Proof. Using exactly the same argument as in the proof of [Castillejos and Evington 2021, Lemma 3.3] and modifying it as hinted in the remark stated before [Castillejos and Evington 2021, Theorem 3.4], we may appeal to [Antoine et al. 2022, Theorem 7.13] (since we assume stable rank one) and pick a nonzero hereditary C*-subalgebra $B \subset A \otimes \mathbb{K}$ with $T^+(B) = \mathbb{R}^{>0}T(B)$ and for which T(B) is nonempty and

¹¹We note that this is automatic if one assumes, e.g., that A has continuous scale (see [Lin 1991, Definition 2.5]), which is a rather common assumption in the context of classification.

¹²Although we use it in the proof, it is likely that stable rank one is not so important for the claim to hold, although we take no guess as to pinning down the correct general assumptions. We note, however, that simple finite \mathcal{Z} -stable C*-algebras have stable rank one; see [Fu et al. 2022; Rørdam 2004].

compact. By Brown's theorem, it follows that *A* and *B* are stably isomorphic. Proposition 1.15 implies that we have an isomorphism $A^{\omega} \cap A' \cong B^{\omega} \cap B'$ that induces a bijection between the canonical traces on the left and the right. Hence the claim holds for *A* if and only if it holds for *B*.

Now *B* has uniform property Gamma (see Remark 2.2), so [Castillejos et al. 2021b, Lemma 3.7] implies that *B* has CPoU. By the "no silly trace" theorem [Castillejos et al. 2021a, Proposition 2.5],¹³ one has that $T(B^{\omega})$ is the weak-*-closed convex hull of the limit traces. If *B* is unital, then the claim follows directly from [Castillejos et al. 2021b, Proposition 4.6]. If *B* is nonunital, we can extend the inclusion map $B \subset B^{\omega}$ to a unital inclusion $B^{\dagger} \subset B^{\omega}$. From this point of view, we have a trivial equality of algebras

$$B^{\omega} \cap B' = B^{\omega} \cap (B^{\dagger})' \cap \{1_{B^{\omega}} - 1_{B^{\dagger}}\}^{\perp}.$$

In this case it follows from [Castillejos and Evington 2020, Proposition 5.7] that $T(B^{\omega} \cap B')$ is the closed convex hull of traces of the form τ_a , where $\tau \in T_{\omega}(B)$ is a limit trace and $a \in B^{\dagger}$ is a positive element with $\tau(a) = 1$. If $(e_n)_{n \in \mathbb{N}}$ is an increasing approximate unit in *B*, then $b_n = e_n a e_n \in B$ converges to *a* strictly, and hence $||b_n - a||_{2,\tau} \to 0$. This implies the convergence of tracial states $\tau(b_n)^{-1}\tau_{b_n} \to \tau_a$ in the norm topology, so we observe the equality

$$T(B^{\omega} \cap B') = \overline{\operatorname{conv}}^{w^*} \{ \tau_b \mid \tau \in T_{\omega}(B), \ b \in B_+, \ \tau(b) = 1 \}.$$

Theorem 2.6. Let A be a separable, simple, nuclear C*-algebra with stable rank one. Then $\alpha : G \curvearrowright A$ has equivariant uniform property Gamma if and only if $\alpha^{\omega} : G \curvearrowright A^{\omega} \cap A'$ has local equivariant uniform property Gamma with respect to bounded traces.

Proof. In order to increase readability in this proof, let us specify another free ultrafilter κ on \mathbb{N} (which may or may not be equal to ω).

We shall show the "if" part first, which actually holds for arbitrary separable simple C*-algebras with $Q\widetilde{T}_2(A) = \widetilde{T}(A)$ and $T^+(A) \neq \emptyset$. Let $k \ge 2$. Assuming α^{ω} has local equivariant property Gamma with respect to bounded traces, we can find pairwise orthogonal projections $p_1, \ldots, p_k \in ((A^{\omega} \cap A')^{\kappa, b})^{(\alpha^{\omega})^{\kappa}}$ such that

$$\tau(ap_j) = \frac{1}{k}\tau(a) \quad \text{for } j = 1, \dots, k, \ a \in A, \ \tau \in T_{\kappa}(A^{\omega} \cap A').$$

For each j = 1, ..., k, let p_j be represented by a sequence of positive contractions $(p_{j,n})_{n \in \mathbb{N}}$ in $A^{\omega} \cap A'$. Let in turn each element $p_{j,n}$ be represented by a central sequence $(x_{j,n,\ell})_{\ell \in \mathbb{N}}$ of positive contractions in *A*. Traces in $T_{\kappa}(A^{\omega} \cap A')$ in particular include limit traces associated to sequences of canonical traces. Let $C \subset \mathcal{P}(A)_+ \setminus \{0\}$ be a countable dense subset. Let $K \subset T^+(A)$ be a compact generator. By the conclusion of Lemma 1.17, it follows, for all $a \in C$ and all sequences $(\theta_{\ell})_{\ell \in \mathbb{N}}$ in *K*, that, if τ is the limit trace on A_{ω} induced by $(\theta_{\ell})_{\ell \in \mathbb{N}}$ and τ_a is the induced bounded trace on $A^{\omega} \cap A'$ that we view in a trivial way as a multiple of a (constant) limit trace on $(A^{\omega} \cap A')^{\kappa,b}$, then

$$0 = \lim_{n \to \kappa} \|p_{j,n} - p_{j,n}^2\|_{1,\tau_a} = \lim_{n \to \kappa} \tau(a|p_{j,n} - p_{j,n}^2|) \stackrel{\text{Lemma 1.17}}{=} \lim_{n \to \kappa} \lim_{\ell \to \omega} \theta_\ell(a|x_{j,n,\ell} - x_{j,n,\ell}^2|).$$

¹³Strictly speaking the conclusion is about the reduced tracial product B^{∞} in the reference, but this makes no difference to the argument there.

Since the sequence $(\theta_\ell)_{\ell \in \mathbb{N}}$ in K was arbitrary, we may rewrite this as

$$0 = \lim_{n \to \kappa} \lim_{\ell \to \omega} \max_{\theta \in K} \theta(a | x_{j,n,\ell} - x_{j,n,\ell}^2 |).$$

We may argue in a completely analogous fashion to see that

$$0 = \lim_{n \to \kappa} \lim_{\ell \to \omega} \max_{\theta \in K} \theta(a|x_{j,n,\ell} - \alpha_g(x_{j,n,\ell})|), \quad g \in G,$$

as well as

$$0 = \lim_{n \to \kappa} \lim_{\ell \to \omega} \max_{\theta \in K} \left| \theta(ax_{j,n,\ell}) - \frac{1}{k} \theta(a) \right| = \lim_{n \to \kappa} \lim_{\ell \to \omega} \max_{\theta \in K} \theta(ax_{j,n,\ell} x_{i,n,\ell})$$

for all i, j = 1, ..., k with $i \neq j$. Lastly, we have by definition that $(x_{j,n,\ell})_{\ell \in \mathbb{N}}$ is a central sequence as $\ell \to \omega$. Appealing to Kirchberg's ε -test, we can find central sequences of positive contractions $e_{\ell}^{(j)}$ in A for j = 1, ..., k satisfying, for all $a \in C$, the properties

$$0 = \lim_{\ell \to \omega} \max_{\theta \in K} \left| \theta(ae_{\ell}^{(j)}) - \frac{1}{k}\theta(a) \right| = \lim_{\ell \to \omega} \max_{\theta \in K} \theta(a|e_{\ell}^{(j)} - e_{\ell}^{(j)2}|)$$

and

$$0 = \lim_{\ell \to \omega} \max_{\theta \in K} \theta(ae_{\ell}^{(j)}e_{\ell}^{(i)}) = \lim_{\ell \to \omega} \max_{\theta \in K} \theta(a|e_{\ell}^{(j)} - \alpha_g(e_{\ell}^{(j)})|), \quad g \in G \text{ and } i \neq j$$

We consider the resulting elements $e_j \in A^{\omega} \cap A'$ represented by $(e_{\ell}^{(j)})_{\ell \in \mathbb{N}}$. Given that *C* was dense in A_+ , we may conclude that they are pairwise orthogonal projections belonging to $(A^{\omega} \cap A')^{\alpha^{\omega}}$ satisfying $\tau_a(e_j) = \tau(a)/k$ for all $\tau \in \widetilde{T}_{\omega}(A)$ and $a \in C$ with $\tau(a) < \infty$. In conclusion, this shows that α has equivariant uniform property Gamma.

For the "only if" part, suppose that α has equivariant property Gamma. Given $k \ge 2$, there exist pairwise orthogonal projections $p_1, \ldots, p_k \in (A^{\omega} \cap A')^{\alpha^{\omega}}$ such that, for all $a \in A_+$ and $\tilde{\tau} \in \tilde{T}_{\omega}(A)$ with $\tilde{\tau}(a) < \infty$,

$$\tilde{\tau}_a(p_j) = \frac{1}{k} \tilde{\tau}(a) \quad \text{for } j = 1, \dots, k.$$

As above, choose a compact generator $K \subset T^+(A)$. If we represent each element p_j by a central sequence of positive contractions $(p_{j,n})_{n \in \mathbb{N}}$ in A, then we can argue as before and see that, for all $a \in \mathcal{P}(A)_+ \setminus \{0\}$, $g \in G$, and $i \neq j$, one has the limit properties

$$0 = \lim_{n \to \omega} \max_{\theta \in K} \left| \theta(ap_{j,n}) - \frac{1}{k} \theta(a) \right| = \lim_{n \to \omega} \max_{\theta \in K} \theta(a|p_{j,n} - \alpha_g(p_{j,n})|)$$

and

$$0 = \lim_{n \to \omega} \max_{\theta \in K} \theta(ap_{j,n}p_{i,n}) = \lim_{n \to \omega} \max_{\theta \in K} \theta(a|p_{j,n} - p_{j,n}^2|).$$

Now take a countable subset $S \subset (A^{\omega} \cap A')^{\kappa,b}$ whose closure would represent a separable subset as in Definition 2.3. Without loss of generality, let us assume *S* consists of positive elements. Choose a countable subset $S_0 \subset (A^{\omega} \cap A')_+$ such that every element of *S* is represented by a bounded S_0 -valued sequence. Next, choose an increasing sequence of finite sets $F_n \subset \mathcal{P}(A)_+ \setminus \{0\}$ such that their union is dense in A_+ and every element in S_0 has a representing sequence in $\prod_{n \in \mathbb{N}} F_n$. Appealing to the above stated properties of the sequences $(p_{j,n})_{n \in \mathbb{N}}$ for j = 1, ..., n, we may find an increasing sequence of natural numbers $\ell \mapsto n_{\ell}$ such that the resulting subsequences satisfy

$$0 = \lim_{\ell \to \infty} \max_{a \in F_{\ell}} \|[a, p_{j, n_{\ell}}]\| = \lim_{\ell \to \infty} \max_{a \in F_{\ell}} \max_{\theta \in K} \theta(a|p_{j, n_{\ell}} - \alpha_g(p_{j, n_{\ell}})|),$$
(2-1)

$$0 = \lim_{\ell \to \infty} \max_{a \in F_{\ell}} \max_{\theta \in K} \theta(ap_{j,n_{\ell}}p_{i,n_{\ell}}) = \lim_{\ell \to \infty} \max_{a \in F_{\ell}} \max_{\theta \in K} \theta(a|p_{j,n_{\ell}} - p_{j,n_{\ell}}^2|),$$
(2-2)

and

$$0 = \lim_{\ell \to \infty} \max_{a,b \in F_{\ell}} \max_{\theta \in K} \left| \theta(abp_{j,n_{\ell}}) - \frac{1}{k} \theta(ab) \right|$$
(2-3)

for all i, j = 1, ..., k with $i \neq j$. By the choice of the sets F_{ℓ} , we can see that $(p_{j,n_{\ell}})_{\ell \in \mathbb{N}}$ defines a central sequence in A, and its induced element $e_j \in A^{\omega} \cap A'$ commutes with elements in S_0 . We keep in mind the conclusion of Lemma 1.17. Then conditions (2-1) and (2-2) imply that $e_1, ..., e_k$ are pairwise orthogonal projections in $(A^{\omega} \cap A')^{\alpha^{\omega}}$. Condition (2-3) implies that, for all j = 1, ..., k, $\tau \in \widetilde{T}_{\omega}(A)$, every $a \in A_+$ with $\tau(a) = 1$, and every $b \in S_0$, we have

$$\tau_a(be_j) = \tau(abe_j) = \frac{1}{k}\tau(ab) = \frac{1}{k}\tau_a(b).$$

By Proposition 2.5, the weak-*-closed convex hull of such tracial states τ_a yields the whole tracial state space of $A^{\omega} \cap A'$. In other words, we may conclude

$$\tau(be_j) = \frac{1}{k}\tau(b)$$
 for all $j = 1, \dots, k, b \in S_0$ and $\tau \in T(A^{\omega} \cap A')$.

We may view e_j as constant elements inside $(A^{\omega} \cap A')^{\kappa,b}$. Since every element in *S* was represented by a sequence in S_0 , we may conclude that the elements e_1, \ldots, e_k satisfy the required property from Definition 2.3 applied to the action

$$\alpha^{\omega}:G \curvearrowright A^{\omega} \cap A'.$$

We conclude that α^{ω} has local equivariant property Gamma with respect to bounded traces.

3. Dynamical complemented partitions of unity

This section contains the most involved technical arguments of the article, namely the proof that local equivariant property Gamma implies the existence of a dynamical version of complemented partitions of unity [Castillejos et al. 2021b, Definition 3.1], or dynamical CPoU for short. In the case where the induced action on the tracial state space has the property that all orbits are finite with uniformly bounded cardinality, a different iteration of dynamical CPoU was proved in [Gardella et al. 2022, Theorem 4.3]. However, we note that the general statement we prove is a weaker and more intricate version compared to earlier versions but will nevertheless be sufficient to deduce the tracial local-to-global principle.

The starting point for the approach in this section is the following weaker version of CPoU shown in [Castillejos et al. 2021b, Lemma 3.6] for nuclear C*-algebras, which turns out to hold automatically with the aid of the theory of tracially complete C*-algebras [Carrión et al. 2023a].

Proposition 3.1. Let A be a σ -unital C*-algebra with T(A) nonempty and compact. Then, for every $\|\cdot\|_{2,T_{\omega}(A)}$ -separable subset $S \subset A^{\omega,b}$, every $k \in \mathbb{N}$, every family $a_1, \ldots, a_k \in A_+$, and every

$$\delta > \sup_{\tau \in T(A)} \min_{i=1,\dots,k} \tau(a_i),$$

there exist $e_1, \ldots, e_k \in (A^{\omega, b} \cap S')^1_+$ such that, for all $\tau \in T_{\omega}(A)$,

•
$$\tau\left(\sum_{i=1}^{k} e_i\right) = 1$$

• $\tau(a_i e_i) \leq \delta \tau(e_i)$ for $i = 1, \ldots, k$.

Proof. Let *S*, *k*, a_1, \ldots, a_k , and δ be chosen as in the assumption. Set

$$\delta_0 := \sup_{\tau \in T(A)} \min_{i=1,\dots,k} \tau(a_i) < \delta.$$

Since *S* is $\|\cdot\|_{2,T_{\omega}(A)}$ -separable, it is first of all clear that one may find a nondegenerate separable C*subalgebra $A_0 \subseteq A$ containing the tuple a_1, \ldots, a_k such that every element of *S* can be represented by a bounded sequence in A_0 . As every tracial state on *A* restricts to one on A_0 , the tracial state space of A_0 is still nonempty and compact, and furthermore

$$\sup_{\tau\in T(A_0)}\min_{i=1,\ldots,k}\tau(a_i)\geq \delta_0.$$

Let $\eta > 0$. We claim that there exists a finite set $F_{\eta} \in A$ and $\varepsilon_{\eta} > 0$ such that, if ρ is any state on A with

$$\max_{x\in F_{\eta}}|\rho(x^*x)-\rho(xx^*)|<\varepsilon_{\eta},$$

then $\min_{i=1,...,k} \rho(a_i) < \delta_0 + \eta$. If we suppose for a moment that this were false, then it follows that, for every finite set $F \subseteq A$ and every $\varepsilon > 0$, there exists a state $\rho_{(F,\varepsilon)}$ on A with

$$\max_{x \in F} |\rho_{(F,\varepsilon)}(x^*x) - \rho_{(F,\varepsilon)}(xx^*)| < \varepsilon \quad \text{and} \quad \min_{i=1,\dots,k} \rho_{(F,\varepsilon)}(a_i) \ge \delta_0 + \eta.$$

We can view $\rho_{(F,\varepsilon)}$ as a net of states by equipping the set of pairs (F, ε) with the obvious order. By the Banach–Anaoglu theorem, there exists a subset $(\rho_{\lambda})_{\lambda \in \Lambda}$ that weak-*-converges to a positive functional ρ' with norm at most one on A. By the properties of the net $\rho_{(F,\varepsilon)}$, it is clear that ρ' is tracial. Hence $\min_{i=1,...,k} \rho'(a_i) \leq \delta_0$, while at the same time

$$\min_{i=1,\dots,k} \rho'(a_i) = \lim_{(F,\varepsilon)} \min_{i=1,\dots,k} \rho_{(F,\varepsilon)}(a_i) \ge \delta_0 + \eta,$$

which is a contradiction.

Using this intermediate claim, we choose for each $n \ge 1$ a finite set $F_n \Subset A$ and $\varepsilon_n > 0$ satisfying the above conclusion for $\eta = 1/n$. Let $A_1 \subseteq A$ be the C*-algebra generated by A_0 and all the finite sets F_n , which is clearly still separable. Since A_1 contains all the finite sets F_n , it follows that every tracial state τ on A_1 must satisfy

$$\min_{i=1,\dots,k} \tau(a_i) \le \delta_0 + \frac{1}{n}, \quad n \ge 1,$$

which leads to

$$\sup_{\tau\in T(A_1)}\min_{i=1,\ldots,k}\tau(a_i)=\delta_0<\delta.$$

By all the properties arranged for the subalgebra $A_1 \subseteq A$ so far, it is clear for proving our main claim that we may swap A for the subalgebra A_1 . In other words, we may assume without loss of generality that A is separable.

By [Carrión et al. 2023a, Definition 3.19, Proposition 3.23], the tracial completion $\overline{A}^{T(A)}$ of A yields a factorial tracially complete C*-algebra. Note that as per the ultraproduct construction of tracially complete C*-algebras in [Carrión et al. 2023a], the object $(\overline{A}^{T(A)})^{\omega}$ in that sense becomes canonically isomorphic to the C*-algebra $A^{\omega,b}$ as considered in Definition 1.8. Because A is separable, $\overline{A}^{T(A)}$ is $\|\cdot\|_{2,T(A)}$ -separable. Thus we may directly apply [Carrión et al. 2023a, Theorem 6.15] (inserting the unit in place of the projection q appearing there) and find the elements $e_1, \ldots, e_k \in (A^{\omega,b} \cap S')^1_+$ with the desired properties.

The main achievement of this section is the following technical lemma.

Lemma 3.2. Given $\varepsilon > 0$ and $t \in (0, 1)$, there exists a universal constant $\eta = \eta(\varepsilon, t) > 0$ such that the following holds: Let A be a σ -unital C*-algebra with T(A) nonempty and compact. Let G be a countable discrete group, and let $\alpha : G \curvearrowright A$ be an action with local equivariant property Gamma with respect to bounded traces. Suppose that $F, H \Subset G$ are finite subsets such that

$$|gH\Delta H| < \eta |H|$$
 for all $g \in F$.

Then, for every $\|\cdot\|_{2,T_{\omega}(A)}$ -separable subset $S \subset A^{\omega,b}$, every family $a_1, \ldots, a_k \in (A^{\omega,b})_+$, and every constant $\delta > 0$ with

$$\frac{\delta}{|H|} > \sup_{\tau \in T_{\omega}(A)} \min_{i=1,\dots,k} \tau(a_i), \tag{3-1}$$

there exist pairwise orthogonal projections $p_1, \ldots, p_k \in A^{\omega, b} \cap S'$ such that, for all $\tau \in T_{\omega}(A)$, one has

$$\tau(p_1 + \dots + p_k) > t, \tag{3-2}$$

$$\tau(a_i p_i) \leq \delta \tau(p_i) \quad \text{for } i = 1, \dots, k, \tag{3-3}$$

$$\max_{g \in F} \sum_{i=1}^{n} \|\alpha_{g}^{\omega}(p_{i}) - p_{i}\|_{2,\tau}^{2} < \varepsilon.$$
(3-4)

Remark 3.3. A standard argument shows that the statement in Lemma 3.2 is equivalent to the existence of a universal constant $\eta(\varepsilon, t) > 0$ satisfying the following statement (using approximations instead of the uniform bounded tracial ultrapower):

If $\alpha : G \curvearrowright A$ is an action and $F, H \Subset G$ are all given as in Lemma 3.2, then, for every finite subset $S \Subset A$, every $\xi > 0$, every family $a_1, \ldots, a_k \in A_+$, and every $\delta > 0$ with

~

$$\frac{\delta}{|H|} > \sup_{\tau \in T(A)} \min_{i=1,\dots,k} \tau(a_i), \tag{3-5}$$

there exist pairwise orthogonal contractions $e_1, \ldots, e_k \in A_+$ such that

$$\begin{split} \|[e_i, x]\|_{2,u} &< \xi \quad for \ x \in S, \ i = 1, \dots, k, \\ \|e_i - e_i^2\|_{2,u} &< \xi \quad for \ i = 1, \dots, k, \\ \tau(e_1 + \dots + e_k) > t - \xi \quad for \ \tau \in T(A), \\ \tau(a_i e_i) &< \delta \tau(e_i) + \xi \quad for \ \tau \in T(A), \ i = 1, \dots, k, \\ \max_{g \in F} \sum_{i=1}^k \|\alpha_g(e_i) - e_i\|_{2,\tau} < \varepsilon + \xi \quad for \ \tau \in T(A). \end{split}$$

In particular, this means that it suffices to prove Lemma 3.2 for positive elements a_1, \ldots, a_k taken in A instead of $A^{\omega,b}$. In this case, (3-1) and (3-5) are equivalent.

The proof of Lemma 3.2 is an adapted version of the proof in the nondynamical setting (cf. [Castillejos et al. 2021b, Section 3]) but also incorporates new ideas related to the dynamical structure. Before we delve into the details, we shall give an overview of the strategy. The construction of the pairwise orthogonal projections p_1, \ldots, p_k in the statement of Lemma 3.2 is done in three steps:

(1) Instead of producing pairwise orthogonal projections p_1, \ldots, p_k , we start by producing (not yet pairwise orthogonal) positive contractions $e_1, \ldots, e_k \in A^{\omega, b} \cap S'$ that satisfy

$$\tau(e_1 + \dots + e_k) = 1$$
, $\tau(a_i e_i) \le \delta \tau(e_i)$ for $i = 1, \dots, k, \ \tau \in T_{\omega}(A)$,

and that are approximately invariant under α^{ω} in the right sense. This is done in Lemma 3.4 and is the only part of the proof that makes use of the approximate Følner property that appears in the assumption of the lemma.

(2) Next, we use equivariant property Gamma to turn these contractions into orthogonal projections $p'_1, \ldots, p'_k \in A^{\omega, b} \cap S'$. As a consequence of this procedure we get that

$$\tau(p'_1 + \dots + p'_k) = \frac{1}{k} \quad \text{for } \tau \in T_\omega(A).$$

but they still satisfy (3-3) and are still approximately invariant in the right sense. This is done in Lemma 3.6. (3) In order to enlarge the trace of the sum of the projections, we repeat the above steps underneath the

projection $1_{A^{\omega}} - \sum_{i=1}^{k} p'_{i}$.¹⁴ We continue this procedure inductively until we end up with orthogonal projections p_{1}, \ldots, p_{k} whose sum exceeds t in trace and that still satisfy (3-3). If everything is done carefully from the start and $\eta > 0$ is chosen correctly, we can control the error in the invariance of the projections and make sure they satisfy (3-4) in the end. (We note, informally, that this error grows with the number of times this procedure is repeated, which is the ultimate reason why we cannot simply work with t = 1 in the statement.)

We shall now implement the above strategy. Combining the contractions arising from Proposition 3.1 with an averaging argument over suitable Følner sets allows us to carry out the first step.

 $^{^{14}}$ For this one actually needs a somewhat stronger version of the second step; see Lemma 3.7.

Lemma 3.4. Let A be a σ -unital C*-algebra with T(A) nonempty and compact. Let $\alpha : G \cap A$ be an action by a countable discrete group. Let $\varepsilon > 0$, and let finite subsets $F \Subset G$ and $H \Subset G$ be given such that $|gH\Delta H| < \varepsilon |H|$ for all $g \in F$. Then, for all $\|\cdot\|_{2,T_{\omega}(A)}$ -separable subsets $S \subset A^{\omega,b}$, all $\delta > 0$, and all $a_1, \ldots, a_k \in A_+$ with

$$\frac{\delta}{|H|} > \sup_{\tau \in T(A)} \min_{i=1,\dots,k} \tau(a_i),$$

there exist $e_1, \ldots, e_k \in (A^{\omega, b} \cap S')^1_+$ such that, for $\tau \in T_{\omega}(A)$,

- $\tau\left(\sum_{i=1}^{k} e_i\right) = 1$,
- $\tau(a_i e_i) \leq \delta \tau(e_i)$ for $i = 1, \ldots, k$, and
- $\max_{g \in F} \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) e_{i}\|_{1,\tau} < \varepsilon.$

Proof. Given $S \subset A^{\omega,b}$, $\delta > 0$, and $a_1, \ldots, a_k \in A_+$ as above, we define

$$a'_i := \frac{1}{|H|} \sum_{g \in H} \alpha_{g^{-1}}(a_i), \quad i = 1, \dots, k.$$

Note that, for each $\tau \in T(A)$, the trace $\frac{1}{|H|} \sum_{g \in H} \tau \circ \alpha_{g^{-1}}$ is again an element of T(A), so we see that

$$\frac{\delta}{|H|} > \sup_{\tau \in T(A)} \min_{i=1,\dots,k} \tau(a'_i).$$

By Proposition 3.1, we know that there exist

$$e'_1, \ldots, e'_k \in \left(A^{\omega} \cap \left(\bigcup_{g \in G} \alpha_g^{\omega}(S)\right)'\right)_+^1$$

such that, for $\tau \in T_{\omega}(A)$,

$$\tau\left(\sum_{i=1}^{k} e'_{i}\right) = 1,$$

$$\tau(a'_{i}e'_{i}) \leq \frac{\delta}{|H|}\tau(e'_{i}) \quad \text{for } i = 1, \dots, k.$$
(3-6)

In particular, this last equation implies that

$$\tau(\alpha_{g^{-1}}(a_i)e'_i) \le \delta\tau(e'_i), \quad g \in H, \ \tau \in T_{\omega}(A).$$
(3-7)

Now, for $i = 1, \ldots, k$, define

$$e_i := |H|^{-1} \sum_{g \in H} \alpha_g^{\omega}(e_i').$$

Clearly this still is a positive contraction in $A^{\omega} \cap S'$. Notice that

$$\tau\left(\sum_{i=1}^{k} e_i\right) = |H|^{-1} \sum_{g \in H} (\tau \circ \alpha_g^{\omega}) \left(\sum_{i=1}^{k} e_i'\right) \stackrel{(3-6)}{=} 1 \quad \text{for } \tau \in T_{\omega}(A)$$

For $\tau \in T_{\omega}(A)$ and $i = 1, \ldots, k$, we have

$$\tau(a_i e_i) = |H|^{-1} \tau \left(a_i \sum_{g \in H} \alpha_g^{\omega}(e_i') \right) = |H|^{-1} \sum_{g \in H} (\tau \circ \alpha_g^{\omega}) (\alpha_{g^{-1}}(a_i) e_i')$$

$$\stackrel{(3-7)}{\leq} |H|^{-1} \sum_{g \in H} \delta(\tau \circ \alpha_g^{\omega}) (e_i') = \delta \tau(e_i).$$

Lastly, we see that, for $g \in F$ and $\tau \in T_{\omega}(A)$, we have

$$\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau} \leq |H|^{-1} \sum_{i=1}^{k} \sum_{h \in gH\Delta H} \|\alpha_{h}^{\omega}(e_{i}')\|_{1,\tau} = |H|^{-1} \sum_{h \in gH\Delta H} \sum_{i=1}^{k} \tau(\alpha_{h}^{\omega}(e_{i}'))$$

$$\stackrel{(3-6)}{=} |H|^{-1} |gH\Delta H| < \varepsilon.$$

Analogously as in the nondynamical setting (cf. [Castillejos et al. 2021b, Lemma 2.4]), (local) equivariant property Gamma allows one to replace positive contractions by projections without changing the tracial values in $A^{\omega,b}$. A different generalization of this lemma was proved in [Gardella et al. 2022, Proposition 3.4], but for the purposes of this paper we need a way to control the (tracially) approximate fixedness of the elements for the action.

Lemma 3.5. Let A be a σ -unital C*-algebra with T(A) nonempty and compact, and let $\alpha : G \frown A$ be an action of a countable discrete group. Assume that α has local equivariant property Gamma with respect to bounded traces. Let $S \subset A^{\omega,b}$ be a $\|\cdot\|_{2,T_{\omega}(A)}$ -separable subset, and let $b \in A^{\omega,b} \cap S'$ be a positive contraction. Then there exists a projection $p \in A^{\omega,b} \cap S'$ such that

$$\tau(ap) = \tau(ab) \quad \text{for } a \in S, \ \tau \in T_{\omega}(A), \tag{3-8}$$

and such that, for all $g \in G$ and $\tau \in T_{\omega}(A)$, one has

$$\|\alpha_g^{\omega}(p) - p\|_{2,\tau}^2 \le \|\alpha_g^{\omega}(b)^{1/2} - b^{1/2}\|_{2,\tau} \|\alpha_g^{\omega}(b)^{1/2} + b^{1/2}\|_{2,\tau}.$$
(3-9)

Proof. Fix $n \in \mathbb{N}$. By a common reindexation trick, it suffices to find a positive contraction $p \in A^{\omega,b} \cap S'$ satisfying (3-8), (3-9), and $||p - p^2||_{2,T_{\omega}(A)}^2 \le 1/n$. An element p satisfying all the necessary properties except (3-9) is constructed in the proof of [Castillejos et al. 2021b, Lemma 2.4] with the use of uniform property Gamma. We show that when the construction is carried out using (local) equivariant property Gamma instead, the resulting projection also satisfies the extra condition (3-9).

Just as in [Castillejos et al. 2021b], we define functions $f_1, \ldots, f_n \in C([0, 1])$ such that $f_i|_{[0,(i-1)/n]} = 0$, $f_i|_{[i/n,1]} = 1$, and f_i is linear on [(i - 1)/n, i/n]. Note that not only $(1/n) \sum_{i=1}^n f_i = id_{[0,1]}$, but the monotonicity of each f_i also implies

$$\frac{1}{n}\sum_{i=1}^{n}|f_i(t_1) - f_i(t_2)| = |t_1 - t_2|, \quad t_1, t_2 \in [0, 1].$$
(3-10)

By local equivariant property Gamma with respect to bounded traces, we can find pairwise orthogonal projections $p_1, \ldots, p_n \in (A^{\omega,b})^{\alpha^{\omega}} \cap S' \cap \{b\}'$ such that $\tau(p_i x) = \tau(x)/n$ for $i = 1, \ldots, n, \tau \in T_{\omega}(A)$,

 $x \in C^*(S \cup \{b\})$. Define

$$p := \sum_{i=1}^{n} p_i f_i(b) \in A^{\omega, \mathfrak{b}} \cap S'.$$

By repeating the arguments in the proof of [Castillejos et al. 2021b, Lemma 2.4] verbatim, we may conclude that $||p - p^2||_{2,T_{\omega}(A)}^2 \leq 1/n$ and $\tau(ap) = \tau(ab)$ for all $a \in S$ and $\tau \in T_{\omega}(A)$. We need to show that (3-9) holds as well. Fix $\tau \in T_{\omega}(A)$ and $g \in G$. By [Connes 1976, I.1], there exists a positive Radon measure ν on $[0, 1]^2$ such that, for every pair of functions $h_1, h_2 \in C_0((0, 1])$, the functions $(s, t) \mapsto h_1(s)$ and $(s, t) \mapsto h_2(t)$ are square integrable and

$$\|h_1(\alpha_g^{\omega}(b)) - h_2(b)\|_{2,\tau}^2 = \int_{[0,1]^2} |h_1(s) - h_2(t)|^2 \,\mathrm{d}\nu(s,t).$$

Then we get

$$\begin{split} \|\alpha_g^{\omega}(p) - p\|_{2,\tau}^2 &= \left\|\sum_{i=1}^n p_i(f_i(\alpha_g^{\omega}(b)) - f_i(b))\right\|_{2,\tau}^2 \\ &= \frac{1}{n} \sum_{i=1}^n \|f_i(\alpha_g^{\omega}(b)) - f_i(b)\|_{2,\tau}^2 \\ &= \frac{1}{n} \sum_{i=1}^n \int_{[0,1]^2} |f_i(s) - f_i(t)|^2 \, \mathrm{d}\nu(s,t) \\ &\leq \frac{1}{n} \sum_{i=1}^n \int_{[0,1]^2} |f_i(s) - f_i(t)| \, \mathrm{d}\nu(s,t) \\ &\stackrel{(3-10)}{=} \int_{[0,1]^2} |s - t| \, \mathrm{d}\nu(s,t) \\ &\leq \sqrt{\int_{[0,1]^2} |s^{1/2} - t^{1/2}|^2 \, \mathrm{d}\nu(s,t)} \sqrt{\int_{[0,1]^2} |s^{1/2} + t^{1/2}|^2 \, \mathrm{d}\nu(s,t)} \\ &= \|\alpha_g^{\omega}(b)^{1/2} - b^{1/2}\|_{2,\tau} \|\alpha_g^{\omega}(b)^{1/2} + b^{1/2}\|_{2,\tau}. \end{split}$$

Using Lemma 3.5, we can construct orthogonal projections that play a similar role to the positive elements in Lemma 3.4.

Lemma 3.6. Let A be a σ -unital C*-algebra with T(A) nonempty and compact. Let $\alpha : G \cap A$ be an action by a countable discrete group and assume it has local equivariant property Gamma with respect to bounded traces. Let $\varepsilon > 0$, $F \Subset G$, and $H \Subset G$ be such that $|gH\Delta H| < \varepsilon |H|$ for all $g \in F$. Then, for every $\|\cdot\|_{2,T_{\omega}(A)}$ -separable subset $S \subset A^{\omega,b}$, all $\delta > 0$, and all $a_1, \ldots, a_k \in A_+$ with

$$\frac{\delta}{|H|} > \sup_{\tau \in T(A)} \min_{i=1,\dots,k} \tau(a_i),$$

there exist pairwise orthogonal projections $p_1, \ldots, p_k \in A^{\omega, b} \cap S'$ such that, for all $\tau \in T_{\omega}(A)$,

- $\tau\left(\sum_{i=1}^{k} p_i\right) = 1/k$,
- $\tau(a_i p_i) \leq \delta \tau(p_i)$ for $i = 1, \ldots, k$, and
- $\max_{g \in F} \sum_{i=1}^{k} \|\alpha_g^{\omega}(p_i) p_i\|_{2,\tau}^2 < 2\sqrt{\varepsilon}/k.$

Proof. By Lemma 3.4, we can find $e_1, \ldots, e_k \in (A^{\omega, b} \cap S')^1_+$ such that, for all $\tau \in T_{\omega}(A)$,

$$\tau\left(\sum_{i=1}^{k} e_i\right) = 1,\tag{3-11}$$

$$\tau(a_i e_i) \le \delta \tau(e_i) \quad \text{for } i = 1, \dots, k, \tag{3-12}$$

$$\max_{g \in F} \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau} < \varepsilon.$$
(3-13)

Let $S_0 = S \cup \{1_{A^{\omega,b}}, a_1, \dots, a_k\}$. Apply Lemma 3.5 for each $i \in \{1, \dots, k\}$ and find a projection $p_i \in A^{\omega,b} \cap S'_0$ such that, for all $a \in S_0$, $\tau \in T_{\omega}(A)$, and $g \in G$, we have

$$\tau(ap_i) = \tau(ae_i),\tag{3-14}$$

$$\|\alpha_g^{\omega}(p_i) - p_i\|_{2,\tau}^2 \le \|\alpha_g^{\omega}(e_i)^{1/2} - e_i^{1/2}\|_{2,\tau} \|\alpha_g^{\omega}(e_i)^{1/2} + e_i^{1/2}\|_{2,\tau}.$$
(3-15)

This already implies the two following facts:

$$\tau\left(\sum_{i=1}^{k} p_i\right) \stackrel{(3-14)}{=} \tau\left(\sum_{i=1}^{k} e_i\right) \stackrel{(3-11)}{=} 1 \quad \text{for } \tau \in T_{\omega}(A), \tag{3-16}$$

$$\tau(a_i p_i) \stackrel{(3-14)}{=} \tau(a_i e_i) \stackrel{(3-12)}{\leq} \delta \tau(e_i) \stackrel{(3-14)}{=} \delta \tau(p_i) \quad \text{for } i = 1, \dots, k, \ \tau \in T_{\omega}(A).$$
(3-17)

Furthermore, we get, for $g \in F$ and $\tau \in T_{\omega}(A)$, that

$$\begin{split} \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i}) - p_{i}\|_{2,\tau}^{2} &\leq \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i})^{1/2} - e_{i}^{1/2}\|_{2,\tau} \|\alpha_{g}^{\omega}(e_{i})^{1/2} + e_{i}^{1/2}\|_{2,\tau} \\ &\leq \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i})^{1/2} - e_{i}^{1/2}\|_{2,\tau} (\|\alpha_{g}^{\omega}(e_{i})^{1/2}\|_{2,\tau} + \|e_{i}^{1/2}\|_{2,\tau}) \\ \\ & \overset{\text{Lemma 1.19}}{\leq} \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}^{1/2} (\|\alpha_{g}^{\omega}(e_{i})^{1/2}\|_{2,\tau} + \|e_{i}^{1/2}\|_{2,\tau}) \\ &= \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}^{1/2} \tau (\alpha_{g}^{\omega}(e_{i}))^{1/2} + \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}^{1/2} \tau (e_{i})^{1/2} \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) + \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} \tau (e_{i}) \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) + \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} \tau (e_{i}) \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) + \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} \tau (e_{i}) \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) + \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} \tau (e_{i}) \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) + \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} \tau (e_{i}) \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) + \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} \tau (e_{i}) \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) + \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) + \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i}) \\ &\leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(e_{i}) - e_{i}\|_{1,\tau}}} \sum_{i=1}^{k} (\tau \circ \alpha_{g}^{\omega})(e_{i})$$

Set

$$S_1 = S \cup C^*(\{a_1, \ldots, a_k\} \cup \{\alpha_g^{\omega}(p_j) \mid 1 \le j \le k, g \in G\}) \subset A^{\omega, \mathfrak{b}}.$$

Since *A* has local equivariant property Gamma with respect to bounded traces, we can find pairwise orthogonal projections $r_1, \ldots, r_k \in (A^{\omega,b})^{\alpha^{\omega}} \cap S'_1$ such that

$$\tau(r_i a) = \frac{1}{k} \tau(a) \quad \text{for } \tau \in T_\omega(A), \ a \in S_1, \ i = 1, \dots, k.$$
(3-19)

Set $p'_i := r_i p_i$. Then clearly $p'_1, \ldots, p'_k \in A^{\omega, b} \cap S'$ are pairwise orthogonal projections. We get, for each $\tau \in T_{\omega}(A)$, that

$$\tau\left(\sum_{i=1}^{k} p_i'\right) = \tau\left(\sum_{i=1}^{k} r_i p_i\right) \stackrel{(3-19)}{=} \frac{1}{k} \tau\left(\sum_{i=1}^{k} p_i\right) \stackrel{(3-16)}{=} \frac{1}{k}.$$

For $i = 1, \ldots, k$ and $\tau \in T_{\omega}(A)$, we get

$$\tau(a_i p_i') = \tau(a_i r_i p_i) \stackrel{(3-19)}{=} \frac{1}{k} \tau(a_i p_i) \stackrel{(3-17)}{\leq} \frac{\delta}{k} \tau(p_i) \stackrel{(3-19)}{=} \delta \tau(p_i').$$

Lastly, for $g \in F$ and $\tau \in T_{\omega}(A)$, we get

$$\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i}') - p_{i}'\|_{2,\tau}^{2} \stackrel{(3-19)}{=} \frac{1}{k} \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i}) - p_{i}\|_{2,\tau}^{2} \stackrel{(3-18)}{<} \frac{2\sqrt{\varepsilon}}{k}.$$

Hence the projections p'_i satisfy all the required properties. This finishes the proof.

In order to carry out the inductive argument to enlarge the trace of the sum of the constructed orthogonal projections, we need a stronger version of the previous lemma that allows us to find the orthogonal projections under an arbitrary tracially constant projection q.

Lemma 3.7. Let A be a σ -unital C*-algebra with T(A) nonempty and compact. Let $\alpha : G \cap A$ be an action by a countable discrete group and assume it has local equivariant property Gamma with respect to bounded traces. Let $\varepsilon > 0$, $F \Subset G$, and $H \Subset G$ be such that $|gH\Delta H| < \varepsilon |H|$ for all $g \in F$. Choose $\delta > 0$ and $a_1, \ldots, a_k \in A_+$ such that

$$\frac{\delta}{|H|} > \sup_{\tau \in T(A)} \min_{i=1,\dots,k} \tau(a_i).$$

For every $\mu \in (0, 1]$ and $\|\cdot\|_{2, T_{\omega}(A)}$ -separable subset $S_0 \subset A^{\omega, b}$, there exists a $\|\cdot\|_{2, T_{\omega}(A)}$ -separable subset $S \subset A^{\omega, b}$ such that, if $q \in A^{\omega, b} \cap S'$ is a projection with $\tau(q) = \mu$ for all $\tau \in T_{\omega}(A)$, there exist pairwise orthogonal projections $p_1, \ldots, p_k \in A^{\omega, b} \cap S'_0 \cap \{\alpha_g^{\omega}(q) \mid g \in G\}'$ such that, for all $\tau \in T_{\omega}(A)$,

- $\sum_{i=1}^{k} \tau(p_i q) = \mu/k$,
- $\tau(a_i p_i q) \leq \delta \tau(p_i q)$ for $i = 1, \ldots, k$,
- $\max_{g \in F} \sum_{i=1}^{k} \|q(\alpha_g^{\omega}(p_i) p_i)\|_{2,\tau}^2 \le 2\mu\sqrt{\varepsilon}/k$, and
- $\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i})(\alpha_{g}^{\omega}(q)-q)\|_{2,\tau}^{2} \leq (1/k) \|\alpha_{g}^{\omega}(q)-q\|_{2,T_{\omega}(A)}^{2}$ for all $g \in G$.

Proof. Let $\varepsilon > 0$, $F, H \subseteq G, \delta > 0$, and $a_1, \ldots, a_k \in A_+$ be as in the statement of the lemma. In order to prove the claim, it suffices to prove the following local statement:

For every $\mu \in (0, 1]$, $\zeta > 0$, $T \Subset A$, and $E \Subset G$, there exist $S \Subset A$ and $\xi > 0$ such that, if $q \in A_+^1$ satisfies

$$||q-q^2||_{2,u} < \xi, \quad \sup_{\tau \in T(A)} |\tau(q)-\mu| < \xi, \text{ and } ||[q,s]||_{2,u} < \xi \text{ for } s \in S,$$

then there exist pairwise orthogonal projections $p_1, \ldots, p_k \in A^1_+$ such that

$$\begin{split} \|p_{i} - p_{i}^{2}\|_{2,u} < \zeta, \\ \|[p_{i}, t]\|_{2,u} < \zeta \quad \text{for } t \in T \cup \{\alpha_{g}(q) \mid g \in E\}, \\ \sup_{\tau \in T(A)} \left|\sum_{i=1}^{k} \tau(p_{i}q) - \frac{\mu}{k}\right| < \zeta, \\ \tau(a_{i}p_{i}q) < \delta\tau(p_{i}q) + \zeta \quad \text{for } \tau \in T(A), \\ \sup_{\tau \in T(A)} \max_{g \in F} \sum_{i=1}^{k} \|q(\alpha_{g}(p_{i}) - p_{i}\|_{2,\tau}^{2} < \frac{2\mu\sqrt{\varepsilon}}{k} + \zeta, \\ \sup_{\tau \in T(A)} \max_{g \in E} \sum_{i=1}^{k} \|\alpha_{g}(p_{i})^{1/2}(\alpha_{g}(q) - q)\|_{2,\tau}^{2} < \frac{1}{k} \|\alpha_{g}(q) - q\|_{2,u}^{2} + \zeta. \end{split}$$

We prove this local statement by contradiction. Suppose there exist $\mu \in (0, 1]$, $\zeta > 0$, $T \subseteq A$, and $E \subseteq G$ for which the statement does not hold. In other words, this means that, for every $\emptyset \neq S \subseteq A$, we can find a $q_S \in A_+^1$ such that

$$||q_S - q_S^2||_{2,u} < 1/|S|, \quad \sup_{\tau \in T(A)} |\tau(q_S) - \mu| < 1/|S|, \text{ and } ||[q_S, s]||_{2,u} < 1/|S| \text{ for } s \in S,$$

but there exist no pairwise orthogonal projections $p_1, \ldots, p_k \in A^1_+$ satisfying

$$\|p_i - p_i^2\|_{2,u} < \zeta, \tag{3-20}$$

$$\|[p_i, t]\|_{2,u} < \zeta \quad \text{for } t \in T \cup \{\alpha_g(q_S) \mid g \in E\},$$
(3-21)

$$\sup_{\tau \in T(A)} \left| \sum_{i=1}^{k} \tau(p_i q_S) - \frac{\mu}{k} \right| < \zeta,$$
(3-22)

$$\tau(a_i p_i q_S) < \delta \tau(p_i q_S) + \zeta \quad \text{for } \tau \in T(A), \tag{3-23}$$

$$\sup_{\tau \in T(A)} \max_{g \in F} \sum_{i=1}^{k} \|q_{S}(\alpha_{g}(p_{i}) - p_{i})\|_{2,\tau}^{2} < \frac{2\mu\sqrt{\varepsilon}}{k} + \zeta,$$
(3-24)

$$\sup_{\tau \in T(A)} \max_{g \in E} \sum_{i=1}^{k} \|\alpha_g(p_i)^{1/2} (\alpha_g(q_S) - q_S)\|_{2,\tau}^2 < \frac{1}{k} \|\alpha_g(q_S) - q_S\|_{2,u}^2 + \zeta.$$
(3-25)

In this way we get a net $(q_S)_S$ indexed by the finite subsets of A equipped with inclusion as the natural partial order. We can take a free ultrafilter $\tilde{\omega}$ on this index set of finite subsets of A as follows. For each $I \subseteq A$ consider the set

$$P_I = \{ J \Subset A \mid I \subseteq J \}.$$

As the collection of P_I is closed under finite intersections, they form a filter basis and hence there is a minimal filter on the set of finite subsets of A containing all the sets P_I for $I \subseteq A$. This filter will be free and can be extended to a free ultrafilter $\tilde{\omega}$ by Zorn's lemma.

Similarly as in Definitions 1.3 and 1.8, we can define the norm and bounded tracial ultrapower of A over the ultrafilter $\tilde{\omega}$. We also get a set of limit traces $T_{\tilde{\omega}}(A)$ over $\tilde{\omega}$. Then the net $(q_S)_S$ defines a projection $q \in A^{\tilde{\omega},b} \cap A'$ with value μ on all limit traces on $A^{\tilde{\omega},b}$. By Lemma 3.6, we can find pairwise orthogonal projections $p'_1, \ldots, p'_k \in A^1_+$ such that

$$\|p'_i - {p'_i}^2\|_{2,u} < \zeta \quad \text{for } i = 1, \dots, k,$$
 (3-26)

$$\|[p'_{i}, t]\|_{2,u} < \zeta \quad \text{for } t \in T,$$
(3-27)

$$\sup_{\tau \in T(A)} \left| \tau \left(\sum_{i=1}^{\kappa} p_i' \right) - \frac{1}{k} \right| < \frac{1}{4}\zeta,$$
(3-28)

$$\tau(a_i p'_i) < \delta \tau(p'_i) + \zeta \quad \text{for } i = 1, \dots, k, \ \tau \in T(A),$$
(3-29)

$$\max_{g \in F} \sum_{i=1}^{k} \|\alpha_g(p_i') - p_i'\|_{2,\tau}^2 < \frac{2\sqrt{\varepsilon}}{k} \quad \text{for } \tau \in T(A).$$
(3-30)

Since $q \in A^{\tilde{\omega}, b} \cap A'$, it follows, for each $\tau \in T_{\tilde{\omega}}(A)$ and $g \in G$, that the assignment

$$A \to \mathbb{C} : a \mapsto \tau(a\alpha_g^{\tilde{\omega}}(q))/\tau(\alpha_g^{\tilde{\omega}}(q)) = \tau(a\alpha_g^{\tilde{\omega}}(q))/\mu$$

defines a tracial state on A. In particular, we get that, for $\tau \in T_{\tilde{\omega}}(A)$, the following hold:

$$\left|\tau\left(\sum_{i=1}^{k} p_i'\alpha_g^{\tilde{\omega}}(q)\right) - \frac{\mu}{k}\right| \stackrel{(3-28)}{<} \frac{1}{4}\mu\zeta \le \frac{1}{4}\zeta \quad \text{for } g \in G,$$
(3-31)

$$\tau(a_i p_i' q) \stackrel{(3-29)}{<} \delta \tau(p_i' q) + \mu \zeta \le \delta \tau(p_i' q) + \zeta, \tag{3-32}$$

$$\max_{g \in F} \sum_{i=1}^{k} \|q(\alpha_g(p_i') - p_i')\|_{2,\tau}^2 \stackrel{(3-30)}{<} \frac{2\mu\sqrt{\varepsilon}}{k}.$$
(3-33)

Next we show that, for all $\tau \in T_{\tilde{\omega}}(A)$ and $g \in G$,

$$\sum_{i=1}^{k} \|\alpha_{g}(p_{i}')^{1/2} (\alpha_{g}^{\tilde{\omega}}(q) - q)\|_{2,\tau}^{2} < \frac{1}{k} \|\alpha_{g}^{\tilde{\omega}}(q) - q\|_{2,u}^{2} + \zeta.$$
(3-34)

Fix $\tau \in T_{\tilde{\omega}}(A)$ and $g \in G$. Assume $\tau(q\alpha_g^{\tilde{\omega}}(q)) > 0$. Then the map

$$A \to \mathbb{C}, \quad a \mapsto \tau(aq\alpha_g^{\tilde{\omega}}(q))/\tau(q\alpha_g^{\tilde{\omega}}(q))$$

defines a tracial state on A. This means that

$$\left| \tau \left(\sum_{i=1}^{k} \alpha_g(p_i') q \alpha_g^{\tilde{\omega}}(q) \right) - \frac{1}{k} \tau(q \alpha_g^{\tilde{\omega}}(q)) \right| \stackrel{(3-28)}{<} \frac{1}{4} \zeta \tau(q \alpha_g^{\tilde{\omega}}(q)) \le \frac{1}{4} \zeta.$$
(3-35)

Note that, if $\tau(q\alpha_g^{\tilde{\omega}}(q)) = 0$, then

$$\tau\left(\sum_{i=1}^{k}\alpha_{g}(p_{i}')q\alpha_{g}^{\tilde{\omega}}(q)\right)=0=\frac{1}{k}\tau(q\alpha_{g}^{\tilde{\omega}}(q)),$$

and hence (3-35) also holds in this case. Now

$$\tau\left(\sum_{i=1}^{k} \alpha_g(p_i')(\alpha_g^{\tilde{\omega}}(q)-q)^2\right) = \tau\left(\sum_{i=1}^{k} \alpha_g(p_i')(\alpha_g^{\tilde{\omega}}(q)+q-\alpha_g^{\tilde{\omega}}(q)q-q\alpha_g^{\tilde{\omega}}(q))\right)$$

$$\stackrel{(3-31),(3-35)}{<} \frac{1}{k}\tau(\alpha_g^{\tilde{\omega}}(q)+q-\alpha_g^{\tilde{\omega}}(q)q-q\alpha_g^{\tilde{\omega}}(q))+\zeta$$

$$= \frac{1}{k}\tau((\alpha_g^{\tilde{\omega}}(q)-q)^2)+\zeta.$$

This proves (3-34). If we combine this with (3-26), (3-27), and (3-31)–(3-33), we can conclude that, for some q_S in the net, (3-20)–(3-25) must hold. This gives the desired contradiction.

Proof of Lemma 3.2. Given $\varepsilon > 0$ and $t \in (0, 1)$, choose $\eta > 0$ small enough that

$$4\left\lceil \frac{t}{1-t}\right\rceil \sqrt{\eta} < \varepsilon. \tag{3-36}$$

We show that such a constant η satisfies the required properties. Let $\alpha : G \cap A$, finite sets $F, H \in G$, and $S \subset A^{\omega,b}$ be given as in the statement of the lemma. By Remark 3.3, it suffices to consider $\delta > 0$ and $a_1, \ldots, a_k \in A_+$ such that

$$\frac{\delta}{|H|} > \sup_{\tau \in T(A)} \min_{i=1,\dots,k} \tau(a_i).$$

We construct the pairwise orthogonal projections p_1, \ldots, p_k in $N := k \lceil t/(1-t) \rceil$ steps. Define a sequence (s_n) in [0, 1) inductively by setting $s_0 = 0$ and setting $s_{i+1} = s_i + (1-s_i)/k$. Note that, when s < t, we have s + (1-s)/k > s + (1-t)/k. If we assumed for a moment that $s_N < t$, then this sequence is less than t for all of the first N steps, leading to $s_N > N(1-t)/k \ge t$, which is a contradiction; hence we must have $s_N > t$. Next, we construct separable subsets S_0, \ldots, S_N as follows: Set $S_N := S$. Given $i \in \{1, \ldots, N\}$ such that S_i is defined, let S_{i-1} be the union of S_i and the set determined by Lemma 3.7 with $1 - s_{i-1}$ in place of μ and S_i in place of S_0 .

In the initial step, we set $p_1^{(0)} = \cdots = p_k^{(0)} = 0$. Now suppose that, for some $n \in \mathbb{N}$, we have pairwise orthogonal projections $p_1^{(n)}, \ldots, p_k^{(n)} \in A^{\omega, \mathfrak{b}} \cap S'_n$ such that, for all $\tau \in T_{\omega}(A)$,

$$\tau(p_1^{(n)} + \dots + p_k^{(n)}) = s_n, \tag{3-37}$$

$$\tau(a_i p_i^{(n)}) \le \delta \tau(p_i^{(n)}) \quad \text{for } i = 1, \dots, k,$$
(3-38)

$$\max_{g \in F} \sum_{i=1}^{k} \|\alpha_g^{\omega}(p_i^{(n)}) - p_i^{(n)}\|_{2,\tau}^2 \le \frac{4n\sqrt{\eta}}{k},\tag{3-39}$$

$$\max_{g \in F} \left\| \sum_{i=1}^{k} \alpha_{g}^{\omega}(p_{i}^{(n)}) - \sum_{i=1}^{k} p_{i}^{(n)} \right\|_{2, T_{\omega}(A)}^{2} \le 2s_{n}\sqrt{\eta}.$$
(3-40)

Note that the $p_i^{(0)}$ trivially satisfy (3-37)–(3-40). We show that we can construct pairwise orthogonal projections $p_1^{(n+1)}, \ldots, p_k^{(n+1)} \in A^{\omega,b} \cap S'_{n+1}$ such that, for all $\tau \in T_{\omega}(A)$,

$$\tau(p_1^{(n+1)} + \dots + p_k^{(n+1)}) = s_{n+1}, \tag{3-41}$$

$$\tau(a_i p_i^{(n+1)}) \le \delta \tau(p_i^{(n+1)}) \quad \text{for } i = 1, \dots, k,$$
(3-42)

$$\max_{g \in F} \sum_{i=1}^{k} \|\alpha_g^{\omega}(p_i^{(n+1)}) - p_i^{(n+1)}\|_{2,\tau}^2 \le \frac{4(n+1)\sqrt{\eta}}{k},\tag{3-43}$$

$$\max_{g \in F} \left\| \sum_{i=1}^{k} \alpha_g^{\omega}(p_i^{(n+1)}) - \sum_{i=1}^{k} p_i^{(n+1)} \right\|_{2, T_{\omega}(A)}^2 \le 2s_{n+1}\sqrt{\eta}.$$
(3-44)

Define

$$q := 1_{A^{\omega, \mathbf{b}}} - \sum_{i=1}^{k} p_i^{(n)}$$

Note that q is a projection in $A^{\omega,b} \cap S'_n$ with $\tau(q) = 1 - s_n$ for all $\tau \in T_{\omega}(A)$. By Lemma 3.7 and our choice of S_n , we can find pairwise orthogonal projections

$$r_1, \ldots, r_k \in A^{\omega, \mathfrak{b}} \cap S'_{n+1} \cap \{\alpha_g^{\omega}(q) \mid g \in G\}'$$

such that, for all $\tau \in T_{\omega}(A)$,

$$\tau\left(\sum_{i=1}^{k} r_i q\right) = \frac{1-s_n}{k},\tag{3-45}$$

$$\tau(a_i r_i q) \le \delta \tau(r_i q) \quad \text{for } i = 1, \dots, k,$$
(3-46)

$$\max_{g \in F} \sum_{i=1}^{k} \|q(\alpha_g^{\omega}(r_i) - r_i)\|_{2,\tau}^2 \le \frac{2(1 - s_n)\sqrt{\eta}}{k},\tag{3-47}$$

$$\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(r_{i})(\alpha_{g}^{\omega}(q)-q)\|_{2,\tau}^{2} \leq \frac{1}{k} \|\alpha_{g}^{\omega}(q)-q\|_{2,T_{\omega}(A)}^{2} \quad \text{for } g \in G.$$
(3-48)

Define

$$p_i^{(n+1)} \coloneqq p_i^{(n)} + qr_i.$$

By construction (recall that $S_{n+1} \subset S_n$), the elements $p_1^{(n+1)}, \ldots, p_k^{(n+1)}$ are pairwise orthogonal projections in $A^{\omega,b} \cap S'_{n+1}$. We show that they satisfy (3-41)–(3-44). Fix $\tau \in T_{\omega}(A)$. Note first of all that

$$\tau\left(\sum_{i=1}^{k} p_i^{(n+1)}\right) = \tau\left(\sum_{i=1}^{k} p_i^{(n)}\right) + \tau\left(\sum_{i=1}^{k} qr_i\right) \stackrel{(3-37),(3-45)}{=} s_n + \frac{1}{k}(1-s_n) = s_{n+1}$$

Moreover, for $i = 1, \ldots, k$, we have

$$\tau(a_i p_i^{(n+1)}) = \tau(a_i p_i^{(n)}) + \tau(a_i q r_i) \stackrel{(3-38),(3-46)}{\leq} \delta \tau(p_i^{(n)}) + \delta \tau(q r_i) = \delta \tau(p_i^{(n+1)}).$$
This shows already that the projections satisfy (3-41) and (3-42). Next we prove that they satisfy (3-43). Note that $p_i^{(n)}$ is orthogonal to q for i = 1, ..., k. Hence, we get, for each $g \in G$ and $\tau \in T_{\omega}(A)$, that

$$\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i}^{(n+1)}) - p_{i}^{(n+1)}\|_{2,\tau}^{2}$$

$$= \sum_{i=1}^{k} \tau((\alpha_{g}^{\omega}(p_{i}^{(n+1)}) - p_{i}^{(n+1)})^{2})$$

$$= \sum_{i=1}^{k} \left(\tau((\alpha_{g}^{\omega}(p_{i}^{(n)}) - p_{i}^{(n)})^{2}) + \tau((\alpha_{g}^{\omega}(qr_{i}) - qr_{i})^{2}) - 2\tau(\alpha_{g}^{\omega}(p_{i}^{(n)})qr_{i}) - 2\tau(p_{i}^{(n)}\alpha_{g}^{\omega}(qr_{i})))\right)$$

$$\leq \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i}^{(n)}) - p_{i}^{(n)}\|_{2,\tau}^{2} + \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(qr_{i}) - qr_{i}\|_{2,\tau}^{2}.$$
(3-49)

For $i = 1, \ldots, k$, we have

$$\begin{aligned} \|\alpha_{g}^{\omega}(qr_{i}) - qr_{i}\|_{2,\tau}^{2} &\leq (\|(\alpha_{g}^{\omega}(q) - q)\alpha_{g}^{\omega}(r_{i})\|_{2,\tau} + \|q(\alpha_{g}^{\omega}(r_{i}) - r_{i})\|_{2,\tau})^{2} \\ &\leq 2(\|(\alpha_{g}^{\omega}(q) - q)\alpha_{g}^{\omega}(r_{i})\|_{2,\tau}^{2} + \|q(\alpha_{g}^{\omega}(r_{i}) - r_{i})\|_{2,\tau}^{2}). \end{aligned}$$

Combining this with (3-49), we find that, for $g \in F$,

$$\begin{split} \sum_{i=1}^{n} \|\alpha_{g}^{\omega}(p_{i}^{(n+1)}) - p_{i}^{(n+1)}\|_{2,\tau}^{2} \\ &\leq \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i}^{(n)}) - p_{i}^{(n)}\|_{2,\tau}^{2} + 2\sum_{i=1}^{k} (\|(\alpha_{g}^{\omega}(q) - q)\alpha_{g}^{\omega}(r_{i})\|_{2,\tau}^{2} + \|q(\alpha_{g}^{\omega}(r_{i}) - r_{i})\|_{2,\tau}^{2}) \\ & (3-39),(3-48) - \frac{4n\sqrt{\eta}}{k} + \frac{2}{k} \left\|\sum_{i=1}^{k} \alpha_{g}^{\omega}(p_{i}^{(n)}) - \sum_{i=1}^{k} p_{i}^{(n)}\right\|_{2,T_{\omega}(A)}^{2} + 2\sum_{i=1}^{k} \|q(\alpha_{g}^{\omega}(r_{i}) - r_{i})\|_{2,\tau}^{2} \\ & (3-40),(3-47) - \frac{4n\sqrt{\eta}}{k} + \frac{4s_{n}\sqrt{\eta}}{k} + \frac{4(1-s_{n})\sqrt{\eta}}{k} = \frac{4(n+1)\sqrt{\eta}}{k}. \end{split}$$

Lastly, we show that the elements $p_i^{(n+1)}$ satisfy (3-44). We get, for each $g \in G$, that

$$\begin{split} \sum_{i=1}^{k} (\alpha_{g}^{\omega}(p_{i}^{(n+1)}) - p_{i}^{(n+1)}) &= \sum_{i=1}^{k} (\alpha_{g}^{\omega}(p_{i}^{(n)}) - p_{i}^{(n)}) + \alpha_{g}^{\omega} \left(q\left(\sum_{j=1}^{k} r_{j}\right)\right) - q\left(\sum_{j=1}^{k} r_{j}\right) \\ &= q - \alpha_{g}^{\omega}(q) + \alpha_{g}^{\omega} \left(q\left(\sum_{j=1}^{k} r_{j}\right)\right) - q\left(\sum_{j=1}^{k} r_{j}\right) \\ &= q \left(1 - \sum_{j=1}^{k} r_{j}\right) - \alpha_{g}^{\omega} \left(q\left(1 - \sum_{j=1}^{k} r_{j}\right)\right) \\ &= q \left(\sum_{j=1}^{k} (\alpha_{g}^{\omega}(r_{j}) - r_{j})\right) + q \left(1 - \sum_{j=1}^{k} \alpha_{g}^{\omega}(r_{j})\right) - \alpha_{g}^{\omega} \left(q \left(1 - \sum_{j=1}^{k} r_{j}\right)\right) \\ &= q \left(\sum_{j=1}^{k} (\alpha_{g}^{\omega}(r_{j}) - r_{j})\right) + (q - \alpha_{g}^{\omega}(q)) \left(1 - \sum_{j=1}^{k} \alpha_{g}^{\omega}(r_{j})\right). \end{split}$$

Keeping in mind that q is a projection and the elements r_j are pairwise orthogonal projections, we have, for all $g \in F$ and $\tau \in T_{\omega}(A)$, that

$$\begin{split} & \left\| \sum_{i=1}^{k} \alpha_{g}^{\omega}(p_{i}^{(n+1)}) - \sum_{i=1}^{k} p_{i}^{(n+1)} \right\|_{2,\tau}^{2} \\ &= \tau \left(\left(\sum_{i=1}^{k} \alpha_{g}^{\omega}(p_{i}^{(n+1)}) - \sum_{i=1}^{k} p_{i}^{(n+1)} \right)^{2} \right) = \tau \left(\left(\alpha_{g}^{\omega}(p_{i}^{(n)} + qr_{i}) - \sum_{i=1}^{k} p_{i}^{(n)} + qr_{i} \right)^{2} \right) \\ &= \tau \left(\left(\sum_{i=1}^{k} q(\alpha_{g}^{\omega}(r_{i}) - r_{i}) + (q - \alpha_{g}^{\omega}(q))(1 - \alpha_{g}^{\omega}(r_{i})) \right)^{2} \right) \\ &= \tau \left(q\left(\sum_{i,j=1}^{k} \alpha_{g}^{\omega}(r_{i}r_{j}) + r_{i}r_{j} - r_{i}\alpha_{g}^{\omega}(r_{j}) - \alpha_{g}^{\omega}(r_{i})r_{j} \right) + (q - \alpha_{g}^{\omega}(q))^{2} \left(1 - \sum_{j=1}^{k} \alpha_{g}^{\omega}(r_{j}) \right) \right) \\ &- 2\tau \left(\left(\sum_{j=1}^{k} r_{j} \right) q(1 - \alpha_{g}^{\omega}(q)) \left(1 - \sum_{j=1}^{k} \alpha_{g}^{\omega}(r_{j}) \right) \right) \\ &\leq \tau \left(q\left(\sum_{i,j=1}^{k} \alpha_{g}^{\omega}(r_{i}r_{j}) + r_{i}r_{j} - r_{i}\alpha_{g}^{\omega}(r_{j}) - \alpha_{g}^{\omega}(r_{i})r_{j} \right) + (q - \alpha_{g}^{\omega}(q))^{2} \right) \\ &\leq \tau \left(q\left(\sum_{i,j=1}^{k} \alpha_{g}^{\omega}(r_{i}) + r_{i} - r_{i}\alpha_{g}^{\omega}(r_{j}) - \alpha_{g}^{\omega}(r_{i})r_{j} \right) + (q - \alpha_{g}^{\omega}(q))^{2} \right) \\ &= \tau \left(q\left(\sum_{i=1}^{k} \alpha_{g}^{\omega}(r_{i}) + r_{i} - r_{i}\alpha_{g}^{\omega}(r_{i}) - \alpha_{g}^{\omega}(r_{i})r_{j} \right) + (q - \alpha_{g}^{\omega}(q))^{2} \right) \\ &= \tau \left(q\left(\sum_{i=1}^{k} (\alpha_{g}^{\omega}(r_{i}) - r_{i})^{2} \right) + (q - \alpha_{g}^{\omega}(q))^{2} \right) = \sum_{i=1}^{k} \|q(r_{i} - \alpha_{g}^{\omega}(r_{i}))\|_{2,\tau}^{2} + \|q - \alpha_{g}^{\omega}(q)\|_{2,\tau}^{2} \right) \\ &\leq 2 \left(s_{n} + \frac{1}{k} (1 - s_{n}) \right) \sqrt{\eta} = 2 s_{n+1} \sqrt{\eta}. \end{split}$$

In the first inequality in the above computation, we used the fact that q commutes with all the elements r_i , thus the term $\left(\sum_{j=1}^k r_j\right)q(1-\alpha_g^{\omega}(q))\left(1-\sum_{j=1}^k \alpha_g^{\omega}(r_j)\right)$ is a product of two positive elements, whose trace value must be nonnegative. In the second inequality, we used the pairwise orthogonality of the projections r_i , so the mixed terms in the double sum appearing above contribute the trace value of $-(r_i\alpha_g^{\omega}(r_j)+\alpha_g^{\omega}(r_i)r_j)$, which likewise is nonpositive.

As explained before, one has $s_N > t$. So if we start the inductive procedure with the projections $p_1^{(0)} = \cdots = p_k^{(0)} = 0$, then after N steps we obtain projections $p_1^{(N)}, \ldots, p_k^{(N)}$ satisfying

$$\sum_{i=1}^{k} \tau(p_i^{(N)}) > t \quad \text{for } \tau \in T_{\omega}(A).$$

Moreover, at that point we have

$$\tau(a_i p_i^{(N)}) \stackrel{(3-42)}{\leq} \delta \tau(p_i^{(N)}) \quad \text{for } i = 1, \dots, k, \ \tau \in T_{\omega}(A),$$
$$\max_{g \in F} \sum_{i=1}^k \|\alpha_g^{\omega}(p_i^{(N)}) - p_i^{(N)}\|_{2,\tau}^2 \stackrel{(3-43)}{\leq} \frac{4N\sqrt{\eta}}{k} \stackrel{(3-36)}{<} \varepsilon.$$

We conclude that $p_1^{(N)}, \ldots, p_k^{(N)}$ satisfy all the required properties.

1416

4. Dynamical tracial local-to-global principle

Here we prove our main technical result, namely that equivariant property Gamma implies a *tracial localto-global principle* for actions of amenable groups. Roughly, this means that whenever a *-polynomial identity has (local) approximate solutions one tracial presentation at a time, then it has (global) approximate solutions in the uniform tracial 2-norm. We begin by precisely defining these polynomial identities.

Definition 4.1 (cf. [Gardella et al. 2022, Definition 4.4]). Let *G* be a discrete group, and let *X* be a countable set of noncommutative variables. A *noncommutative G*-*-*polynomial* in the variables *X* is a noncommutative *-polynomial in the variables $\{g \cdot x \mid g \in G, x \in X\}$.

Let *A* be a C*-algebra with action $\alpha : G \cap A$. Suppose that $h(x_1, \ldots, x_r)$ is a *G*-*-polynomial in *r* noncommuting variables. Given a tuple $(a_1, \ldots, a_r) \in A^r$, the evaluation $h(a_1, \ldots, a_r)$ is computed by interpreting $g \cdot x_i$ as $\alpha_g(a_i)$ for $g \in G$ and $i = 1, \ldots, r$.

The main theorem that we prove in this section is the following.

Theorem 4.2. Let A be a σ -unital C*-algebra with T(A) nonempty and compact, and with weak CPoU. Let α : $G \curvearrowright A$ be an action by an amenable countable discrete group and assume it has local equivariant property Gamma with respect to bounded traces. For each $m \in \mathbb{N}$, let

$$h_m(x_1,\ldots,x_{r_m},z_1,\ldots,z_{s_m})$$

be a G-*-polynomial in $r_m + s_m$ noncommuting variables. Let $(a_i)_{i \in \mathbb{N}}$ be a sequence in $A^{\omega,b}$. Suppose, for every $\varepsilon > 0$, $\ell \in \mathbb{N}$, and $\tau \in \overline{T_{\omega}(A)}^{w^*}$, there exist contractions $(y_i^{\tau})_{i \in \mathbb{N}}$ in $A^{\omega,b}$ such that

 $||h_m(a_1,...,a_{r_m},y_1^{\tau},...,y_{s_m}^{\tau})||_{2,\tau} < \varepsilon \text{ for } m = 1,...,\ell.$

Then there exist contractions $(y_i)_{i \in \mathbb{N}}$ in $A^{\omega, b}$ such that

$$h_m(a_1,\ldots,a_{r_m},y_1,\ldots,y_{s_m}) = 0 \quad for \ all \ m \in \mathbb{N}.$$

$$(4-1)$$

We reduce the complexity of the polynomials involved in the proof of this result with the following. **Lemma 4.3.** Let G be a countable discrete group. Consider sets of variables $X = \{x_i \mid i \in \mathbb{N}\}$ and $Z = \{z_i \mid i \in \mathbb{N}\}$. Assume

 $\mathcal{P} = \{h_m(x_1, \ldots, x_{r_m}, z_1, \ldots, z_{s_m}) \mid m \in \mathbb{N}\}$

is a countable set of noncommutative G-*-polynomials in the variables $X \cup Z$. Then there exists another set of variables $Z' = \{z'_i \mid i \in \mathbb{N}\}$ and another countable set of noncommutative G-*-polynomials

$$\mathcal{P}' = \{h'_m(x_1, \dots, x_{r'_m}, z'_1, \dots, z'_{s'_m}) \mid m \in \mathbb{N}\}$$

in the variables $X \cup Z'$ such that every G-*-polynomial h'_m satisfies one of the following properties:

- (1) $h'_m(x_1, \ldots, x_{r'_m}, 0, \ldots, 0) = 0$ (i.e., no terms in the polynomial with variables only in X) and $h'_m(1, \ldots, 1, z'_1, \ldots, z'_{s'_n})$ is an ordinary *-polynomial in the variables Z';
- (2) $h'_m(x_1, \ldots, x_{r'_m}, z'_1, \ldots, z'_{s'_m}) = \|h'_m(x_1, \ldots, x_{r'_m}, 0, \ldots, 0)\|z'_{s'_m} h'_m(x_1, \ldots, x_{r'_m}, 0, \ldots, 0);$
- (3) $h'_m(x_1, ..., x_{r'_m}, z'_1, ..., z'_{s'_m}) = g \cdot z'_i z'_j$ for some $1 \le i, j \le s'_m$ and some $g \in G$.

Moreover we have that, for every action $\beta : G \curvearrowright B$ on any C*-algebra, any sequence $(b_i)_{i \in \mathbb{N}}$ in B, and subset $T \subset T(B)$ the following two statements hold:

(a) There exist contractions $(y_i)_{i \in \mathbb{N}}$ in B such that

$$||h_m(b_1,...,b_{r_m},y_1,...,y_{s_m})||_{2,T} = 0$$
 for all $m \in \mathbb{N}$

if and only if there exist contractions $(y'_i)_{i \in \mathbb{N}}$ in B such that

$$\|h'_m(b_1,\ldots,b_{r'_m},y'_1,\ldots,y'_{s'_m})\|_{2,T} = 0 \text{ for all } m \in \mathbb{N}.$$

(b) For each $\varepsilon > 0$ and each $\ell \in \mathbb{N}$, there exist contractions $(y_i)_{i \in \mathbb{N}}$ in B such that

$$||h_m(b_1,...,b_{r_m},y_1,...,y_{s_m})||_{2,T} < \varepsilon \quad for \ m = 1,...,\ell$$

if and only if, for each $\varepsilon' > 0$ and each $\ell' \in \mathbb{N}$, there exist contractions $(y'_i)_{i \in \mathbb{N}}$ in B such that

$$||h'_m(b_1,\ldots,b_{r'_m},y'_1,\ldots,y'_{s'_m})||_{2,T} < \varepsilon' \text{ for } m = 1,\ldots,\ell'.$$

Proof. Define

$$Z' = \{z_i \mid i \in \mathbb{N}\} \cup \{z_{i,g} \mid i \in \mathbb{N}, g \in G \setminus \{e\}\} \cup \{w_m \mid m \in \mathbb{N}\}.$$

For each $m \in \mathbb{N}$, we can take the *G*-*-polynomial h_m in the variables $X \cup Z$ and define h''_m in the variables $X \cup Z'$ by replacing every instance of a variable $g \cdot z_i$ for some $i \in \mathbb{N}$ and $g \in G \setminus \{e\}$ by $z_{i,g}$, e.g., the polynomial $g \cdot z_1 - z_2$ would be transformed into $z_{1,g} - z_2$. Next, we define a new *G*-*-polynomial h''_m for each $m \in \mathbb{N}$ by setting

$$h_m'''(X \cup Z') = h_m''(X \cup Z') + \|h_m(x_1, \dots, x_{r_m}, 0, \dots, 0)\|w_m - h_m(x_1, \dots, x_{r_m}, 0, \dots, 0).$$
(4-2)

Set $\mathcal{P}'_1 := \{h'''_m(X \cup Z') \mid m \in \mathbb{N}\}$. All the polynomials in this set are of type (1) mentioned in the statement of this lemma. Next, define the sets of *G*-*-polynomials

$$\mathcal{P}'_2 := \{ \|h_m(x_1, \dots, x_{r_m}, 0, \dots, 0)\|w_m - h_m(x_1, \dots, x_{r_m}, 0, \dots, 0) \mid m \in \mathbb{N} \}$$

and

$$\mathcal{P}'_3 := \{g \cdot z_i - z_{i,g} \mid i \in \mathbb{N}, g \in G \setminus \{e\}\}.$$

These sets consist of polynomials of type (2) and (3), respectively.

Consider $\mathcal{P}' = \mathcal{P}'_1 \cup \mathcal{P}'_2 \cup \mathcal{P}'_3$. The *G*-*-polynomials in this set are all of the right form, and we claim that this does the job. It suffices to show part (a). This is because, for a given C*-algebra *B*, action $\beta : G \cap B$, sequence $(b_i)_{i \in \mathbb{N}} \in B$, and subset $T \subset T(B)$, statement (b) immediately follows from statement (a) when applied to the C*-algebra B_{ω} , with action β_{ω} , sequence $(b_i)_{i \in \mathbb{N}}$ in $B \subset B_{\omega}$, and the set of limit traces on B_{ω} arising from sequences of traces in *T*.

To show part (a), fix a C*-algebras *B*, an action $\beta : G \frown B$, a sequence $(b_i)_{i \in \mathbb{N}}$, and subset $T \subset T(B)$. Let $(y_i)_{i \in \mathbb{N}}$ be any sequence of contractions in *B*. For notational brevity, we denote these sequences by $\overline{b} = (b_i)_i$ and $\overline{y} = (y_i)_i$. Furthermore we shall also write (exclusively in this proof), for two elements $x, y \in B$, the expression " $x =_T y$ " as shorthand for $||x - y||_{2,T} = 0$. We set $y_{i,g} = \beta_g(y_i)$ for $i \in \mathbb{N}$ and $g \in G \setminus \{e\}$ and

$$w_m = \begin{cases} 0, & h_m(b_1, \dots, b_{r_m}, 0, \dots, 0) \\ \frac{h_m(b_1, \dots, b_{r_m}, 0, \dots, 0)}{\|h_m(b_1, \dots, b_{r_m}, 0, \dots, 0)\|}, & h_m(b_1, \dots, b_{r_m}, 0, \dots, 0) \neq 0. \end{cases}$$

Then the tuple

$$\bar{z} := (y_i)_{i \in \mathbb{N}} \times (y_{i,g})_{i \in \mathbb{N}, g \in G \setminus \{e\}} \times (w_m)_{m \in \mathbb{N}}$$

represents a choice for the free variables of Z' inside B. By definition we have $p(\bar{b}, \bar{z}) = 0$ for all $p \in \mathcal{P}'_2 \cup \mathcal{P}'_3$. By definition of the polynomials h''_m , we have

$$h_m''(b_1,\ldots,b_{r_m},\bar{z}) = h_m''(b_1,\ldots,b_{r_m},\bar{z}) + q(\bar{b},\bar{z}) = h_m''(b_1,\ldots,b_{r_m},\bar{z})$$

for some *-polynomial $q \in \mathcal{P}'_2$. Due to the vanishing of all the *-polynomials of \mathcal{P}'_3 in \overline{z} and given how the polynomial h''_m arises from the polynomial h_m via substitution of variables, we may finally observe

$$h_m(b_1,\ldots,b_{r_m},y_1,\ldots,y_{s_m})=h_m''(b_1,\ldots,b_{r_m},\bar{z}).$$

This shows immediately that if the sequence $(y_i)_i$ satisfies

$$||h_m(b_1, \ldots, b_{r_m}, y_1, \ldots, y_{s_m})||_{2,T} = 0$$
 for all $m \in \mathbb{N}$.

then we also have $\|p(\bar{b}, \bar{z})\|_{2,T} = 0$ for all $p \in \mathcal{P}'$. In particular, we get the "only if" part in (a).

Conversely, suppose that

$$\bar{z} := (y_i)_{i \in \mathbb{N}} \times (y_{i,g})_{i \in \mathbb{N}, g \in G \setminus \{e\}} \times (w_m)_{m \in \mathbb{N}}$$

is an arbitrary tuple with values in the unit ball of *B* representing a choice for the free variables in Z' such that $p(\bar{b}, \bar{z}) = 0$ for all $p \in \mathcal{P}'$. By doing the above computations in reverse, we can see that $p(\bar{z}) =_T 0$ for $p \in \mathcal{P}'_3$ forces $y_{i,g} =_T \beta_g(y_i)$ for all $g \in G \setminus \{e\}$. Moreover, the vanishing $p(\bar{b}, \bar{z}) =_T 0$ for $p \in \mathcal{P}'_2$ forces the equation

$$w_m = \frac{h_m(b_1, \dots, b_{r_m}, 0, \dots, 0)}{\|h_m(b_1, \dots, b_{r_m}, 0, \dots, 0)\|}$$

when $h_m(b_1, \ldots, b_{r_m}, 0, \ldots, 0) \neq 0$. Similar to how we argued above, this implies, for all $m \ge 1$, that

$$h_m''(b_1,\ldots,b_{r_m},\bar{z}) = h_m''(b_1,\ldots,b_{r_m},\bar{z}) + q(\bar{b},\bar{z}) =_T h_m''(b_1,\ldots,b_{r_m},\bar{z})$$

for some *-polynomial $q \in \mathcal{P}'_2$. Given how the polynomial h''_m arises from the polynomial h_m via substituion of variables, we may finally observe

$$h_m(b_1,\ldots,b_{r_m},y_1,\ldots,y_{s_m}) =_T h''_m(b_1,\ldots,b_{r_m},\bar{z}), \quad m \ge 1.$$

This shows the "if" part of (a) and finishes the proof.

Proof of Theorem 4.2. Equation (4-1) is equivalent to

$$||h_m(a_1, \ldots, a_{r_m}, y_1, \ldots, y_{s_m})||_{2, T_{\omega}(A)} = 0$$
 for all $m \in \mathbb{N}$.

By the previous lemma, we may assume that the *-polynomials h_m are all of one of the following types:

- (1) $h_m(a_1, ..., a_{r_m}, 0, ..., 0) = 0$ and $h_m(1_{A^{\omega, b}}, ..., 1_{A^{\omega, b}}, z_1, ..., z_{s_m})$ is an ordinary *-polynomial.
- (2) h_m is of the form $||h_m(x_1, \ldots, x_{r_m}, 0, \ldots, 0)||z_i h_m(x_1, \ldots, x_{r_m}, 0, \ldots, 0)$ for some $i \in \mathbb{N}$. Equivalently, since this doesn't change the solutions, we may assume that $h_m(a_1, \ldots, a_{r_m}, z_1, \ldots, z_{s_m})$ is of the form $z_i a$ for some $i \in \mathbb{N}$ and $a \in C^*(\{\alpha_g^{\omega}(a_i) \mid i \in \mathbb{N}, g \in G\})$ with ||a|| = 1.
- (3) h_m is of the form $g \cdot z_i z'_i$ for some $i, i' \in \mathbb{N}$ and $g \in G$.

By Kirchberg's ε -test, it suffices to find, for each $\varepsilon > 0$ and $\ell \in \mathbb{N}$, contractions $(y_i)_{i \in \mathbb{N}}$ in $A^{\omega, b}$ such that

$$||h_m(a_1,\ldots,a_{r_m},y_1,\ldots,y_{s_m})||_{2,\tau} \le \varepsilon \quad \text{for } m=1,\ldots,\ell \text{ and } \tau \in T_\omega(A)$$

Choose $\varepsilon > 0$ and $\ell \in \mathbb{N}$ arbitrarily. Denote by $F \subseteq G$ the set of $g \in G$ appearing in h_m for some $m = 1, \ldots, \ell$. Set

$$\delta = \frac{1}{18}\varepsilon^2. \tag{4-3}$$

Choose $t \in (0, 1)$ such that

$$1 - t < \frac{1}{2}\varepsilon^2. \tag{4-4}$$

Let $\eta > 0$ be the universal constant from Lemma 3.2 corresponding to the tuple (δ^2, t) . Since *G* is amenable, we can find $H \subseteq G$ such that $|gH\Delta H| < \eta |H|$ for each $g \in F$. By assumption, for each $\tau \in \overline{T_{\omega}(A)}^{w^*}$, we can find contractions $(y_i^{\tau})_{i \in \mathbb{N}} \in A^{\omega, b}$ such that

$$||h_m(a_1,\ldots,a_{r_m},y_1^{\tau},\ldots,y_{s_m}^{\tau})||_{2,\tau}^2 < \frac{\varepsilon^2}{2\ell|H|} \text{ for } m=1,\ldots,\ell.$$

Define

$$b^{\tau} := \sum_{m=1}^{\ell} |h_m(a_1, \dots, a_{r_m}, y_1^{\tau}, \dots, y_{s_m}^{\tau})|^2 \in A^{\omega, \mathbf{b}}.$$
(4-5)

Then we get

$$\tau(b^{\tau}) = \sum_{m=1}^{\ell} \|h_m(a_1, \dots, a_{r_m}, y_1^{\tau}, \dots, y_{s_m}^{\tau})\|_{2,\tau}^2 < \frac{\varepsilon^2}{2|H|}.$$

By continuity and compactness of $\overline{T_{\omega}(A)}^{w^*}$, we can find finitely many tracial states $\tau_1, \ldots, \tau_k \in \overline{T_{\omega}(A)}^{w^*}$ such that

$$\frac{\varepsilon^2}{2|H|} > \sup_{\tau \in T_{\omega}(A)} \min_{i=1,\dots,k} \tau(b^{\tau_i}).$$

By Lemma 3.2, it follows that we can find pairwise orthogonal projections

$$p_1, \ldots, p_k \in A^{\omega, \mathsf{b}} \cap \left(\bigcup_{g \in G} \alpha_g^{\omega}(\{y_i^{\tau_1}, \ldots, y_i^{\tau_n}, a_i \mid i \in \mathbb{N}\}) \right)'$$

such that, for $\tau \in T_{\omega}(A)$,

$$\tau\left(\sum_{j=1}^{k} p_j\right) > t,\tag{4-6}$$

$$\tau(b^{\tau_j} p_j) \le \frac{1}{2} \varepsilon^2 \tau(p_j) \quad \text{for } j = 1, \dots, k,$$
(4-7)

$$\max_{g \in F} \sum_{j=1}^{\kappa} \|\alpha_g(p_j) - p_j\|_{2,\tau}^2 < \delta^2.$$
(4-8)

1420

Define $y_i = \sum_{j=1}^k p_j y_i^{\tau_j}$ for $i \in \mathbb{N}$. We show that, for $\tau \in T_{\omega}(A)$ and $m = 1, \ldots, \ell$,

$$\left|\tau(|h_m(a_1,\ldots,a_{r_m},y_1,\ldots,y_{s_m})|^2) - \tau\left(\sum_{j=1}^k p_j |h_m(a_1,\ldots,a_{r_m},y_1^{\tau_j},\ldots,y_{s_m}^{\tau_j})|^2\right)\right| < \frac{1}{2}\varepsilon^2.$$
(4-9)

We distinguish three cases. First, assume that $h_m(a_1, \ldots, a_{r_m}, 0, \ldots, 0) = 0$ and that

$$h_m(1_{A^{\omega,b}},\ldots,1_{A^{\omega,b}},z_1,\ldots,z_{s_m})$$

is an ordinary *-polynomial. In this case

$$|h_m(a_1,\ldots,a_{r_m},y_1,\ldots,y_{s_m})|^2 = \sum_{j=1}^k p_j |h_m(a_1,\ldots,a_{r_m},y_1^{\tau_j},\ldots,y_{s_m}^{\tau_j})|^2$$

since p_1, \ldots, p_k are pairwise orthogonal projections. Second, assume h_m is of the form $z_i - a$ for some $i \in \mathbb{N}$ and some element $a \in C^*(\{\alpha_g^{\omega}(a_i) \mid i \in \mathbb{N}, g \in G\})$ with ||a|| = 1. In this case we have

$$\begin{aligned} \left| \tau(|h_m(a_1, \dots, a_{r_m}, y_1, \dots, y_{s_m})|^2) - \tau\left(\sum_{j=1}^k p_j |h_m(a_1, \dots, a_{r_m}, y_1^{\tau_j}, \dots, y_{s_m}^{\tau_j})|^2\right) \right| \\ &= \left| \tau\left(\left|\sum_{j=1}^k p_j y_i^{\tau_j} - a\right|^2\right) - \tau\left(\sum_{j=1}^k p_j |y_i^{\tau_j} - a|^2\right)\right| = \left| \tau(|a|^2) - \tau\left(\sum_{j=1}^k p_j |a|^2\right) \\ &\leq \tau\left(1 - \sum_{j=1}^k p_j\right) \stackrel{(4-6)}{<} 1 - t \stackrel{(4-4)}{<} \frac{1}{2}\varepsilon^2. \end{aligned}$$

Third, assume h_m is of the form $g \cdot z_i - z_{i'}$ for some $i, i' \in \mathbb{N}$ and $g \in F$. Then we have

$$\begin{aligned} \tau(|h_{m}(a_{1},\ldots,a_{r_{m}},y_{1},\ldots,y_{s_{m}})|^{2}) &- \tau\left(\sum_{j=1}^{k} p_{j}|h_{m}(a_{1},\ldots,a_{r_{m}},y_{1}^{\tau_{j}},\ldots,y_{s_{m}}^{\tau_{j}})|^{2}\right) \\ &= \left|\tau\left(\left|\alpha_{g}^{\omega}\left(\sum_{j=1}^{k} p_{j}y_{i}^{\tau_{j}}\right) - \sum_{j'=1}^{k} p_{j'}y_{i'}^{\tau_{j'}}\right|^{2}\right) - \tau\left(\sum_{j=1}^{k} p_{j}|\alpha_{g}^{\omega}(y_{i}^{\tau_{j}}) - y_{i'}^{\tau_{j}}|^{2}\right)\right) \\ &= \left|\tau\left(\sum_{j=1}^{k} (\alpha_{g}^{\omega}(p_{j}) - p_{j})\alpha_{g}^{\omega}(|y_{i}^{\tau_{j}}|^{2})\right) - \tau\left(\sum_{j=1}^{k} (\alpha_{g}^{\omega}(p_{j}) - p_{j})\alpha_{g}^{\omega}(y_{i}^{\tau_{j'}})^{*}\sum_{j'=1}^{k} p_{j'}y_{i'}^{\tau_{j'}}\right) \\ &- \tau\left(\left(\sum_{j'=1}^{k} p_{j'}y_{i'}^{\tau_{j'}}\right)^{*}\sum_{j=1}^{k} (\alpha_{g}^{\omega}(p_{j}) - p_{j})\alpha_{g}^{\omega}(y_{i}^{\tau_{j}})\right)\right) \\ &\leq \left\|\sum_{j=1}^{k} (\alpha_{g}^{\omega}(p_{j}) - p_{j})\alpha_{g}^{\omega}(|y_{i}^{\tau_{j}}|^{2})\right\|_{2,\tau} + 2\left\|\sum_{j=1}^{k} (\alpha_{g}^{\omega}(p_{j}) - p_{j})\alpha_{g}^{\omega}(y_{i}^{\tau_{j}})\right\|_{2,\tau} \\ &\leq \left\|\sum_{j=1}^{k} (\alpha_{g}^{\omega}(p_{j}) - p_{j})\alpha_{g}^{\omega}(|y_{i}^{\tau_{j}}|^{2})\right\|_{2,\tau} + 2\left\|\sum_{j=1}^{k} (\alpha_{g}^{\omega}(p_{j}) - p_{j})\alpha_{g}^{\omega}(y_{i}^{\tau_{j}})\right\|_{2,\tau}. \end{aligned}$$

$$(4-10)$$

Note that, for any $g \in G$ and any positive contractions $c_1, \ldots, c_k \in A^{\omega, b}$ commuting with the p_j and $\alpha_g^{\omega}(p_j)$, one has that

$$\begin{split} \left\| \sum_{i=1}^{k} (\alpha_{g}^{\omega}(p_{i}) - p_{i})c_{i} \right\|_{2,\tau}^{2} &= \sum_{i=1}^{k} \tau ((\alpha_{g}^{\omega}(p_{i}) - p_{i})^{2}c_{i}^{2}) + \sum_{\substack{i,j=1\\i \neq j}}^{k} \tau (c_{i}(\alpha_{g}^{\omega}(p_{i}) - p_{i})(\alpha_{g}^{\omega}(p_{j}) - p_{j})c_{j}) \\ &\leq \sum_{i=1}^{k} \left(\tau ((\alpha_{g}^{\omega}(p_{i}) - p_{i})^{2}) - \sum_{\substack{j=1,\dots,k\\j \neq i}} \tau (c_{i}\alpha_{g}^{\omega}(p_{i})p_{j}c_{j}) - \sum_{\substack{j=1,\dots,k\\j \neq i}} \tau (c_{i}p_{i}\alpha_{g}^{\omega}(p_{j})c_{j}) \right) \\ &\leq \sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i}) - p_{i}\|_{2,\tau}^{2}, \end{split}$$

where in the last inequality we used the tracial property and the fact that the c_i commute with the p_j and $\alpha_g^{\omega}(p_j)$ to show that the last two terms can be rewritten as the negative of the trace of positive elements.

In particular we have, for $g \in F$, that

$$\left\|\sum_{j=1}^{k} (\alpha_{g}^{\omega}(p_{j}) - p_{j}) \alpha_{g}^{\omega}(|y_{i}^{\tau_{j}}|^{2})\right\|_{2,\tau} \leq \sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i}) - p_{i}\|_{2,\tau}^{2}} \leq \delta \stackrel{(4-8)}{\leq} \delta \stackrel{(4-3)}{=} \frac{1}{18} \varepsilon^{2}.$$
 (4-11)

For j = 1, ..., k, we have that $\alpha_g^{\omega}(y_i^{\tau_j})$ is a contraction that can be written as a linear combination of positive contractions commuting with the p_i and $\alpha_g^{\omega}(p_i)$. An application of the triangle inequality yields

$$\left\|\sum_{i=1}^{k} (\alpha_{g}^{\omega}(p_{i}) - p_{i})\alpha_{g}^{\omega}(y_{i}^{\tau_{j}})\right\|_{2,\tau} \leq 4\sqrt{\sum_{i=1}^{k} \|\alpha_{g}^{\omega}(p_{i}) - p_{i}\|_{2,\tau}^{2}} \stackrel{(4-8)}{<} 4\delta \stackrel{(4-3)}{=} \frac{2}{9}\varepsilon^{2}.$$
 (4-12)

Combining (4-10) with (4-11) and (4-12), we get

$$\left|\tau(|h_m(a_1,\ldots,a_{r_m},y_1,\ldots,y_{s_m})|^2)-\tau\left(\sum_{j=1}^k p_j|h_m(a_1,\ldots,a_{r_m},y_1^{\tau_j},\ldots,y_{s_m}^{\tau_j})|^2\right)\right|<\frac{9}{18}\varepsilon^2=\frac{1}{2}\varepsilon^2.$$

Thus, we have indeed shown that (4-9) holds for all $\tau \in T_{\omega}(A)$ and $m = 1, \ldots, \ell$. From (4-5) we see that

$$\sum_{j=1}^{k} p_j |h_m(a_1, \dots, a_{r_m}, y_1^{\tau_j}, \dots, y_{s_m}^{\tau_j})|^2 \le \sum_{j=1}^{k} p_j b^{\tau_j} \quad \text{for } m = 1, \dots, \ell.$$
(4-13)

As a consequence, for $\tau \in T_{\omega}(A)$ and $m = 1, \ldots, \ell$, we get

$$\|h_{m}(a_{1},\ldots,a_{r_{m}},y_{1},\ldots,y_{s_{m}})\|_{2,\tau}^{2} = \tau(|h_{m}(a_{1},\ldots,a_{r_{m}},y_{1},\ldots,y_{s_{m}})|^{2})$$

$$\stackrel{(4-9)}{<} \tau\left(\sum_{j=1}^{k} p_{j}|h_{m}(a_{1},\ldots,a_{r_{m}},y_{1}^{\tau_{j}},\ldots,y_{s_{m}}^{\tau_{j}})|^{2}\right) + \frac{1}{2}\varepsilon^{2}$$

$$\stackrel{(4-13)}{\leq} \sum_{j=1}^{k} \tau(p_{j}b^{\tau_{j}}) + \frac{1}{2}\varepsilon^{2} \stackrel{(4-7)}{\leq} \sum_{j=1}^{k} \frac{1}{2}\varepsilon^{2}\tau(p_{j}) + \frac{1}{2}\varepsilon^{2} \leq \frac{1}{2}\varepsilon^{2} + \frac{1}{2}\varepsilon^{2} = \varepsilon^{2}. \quad \Box$$

The next theorem gives an alternative formulation of the tracial local-to-global principle that is convenient to use in certain applications. Before we state it, we introduce some notation.

Notation 4.4. Let *A* be a C*-algebra with an action $\alpha : G \cap A$ of a countable discrete group. Given a tracial state $\tau \in T(A)$, denote by $\pi_{\tau} : A \to \mathcal{B}(H_{\tau})$ the corresponding GNS representation. Then we define the representation

$$\pi_{\tau}^{\alpha}: A \to \mathcal{B}(\ell^2(G, H_{\tau})), \quad \pi_{\tau}^{\alpha}(x)(\xi)(h) = \pi_{\tau}(\alpha_h^{-1}(x))\xi(h)$$

The left-regular representation $\lambda : G \to \mathcal{U}(\ell^2(G, H_\tau))$, defined by $(g \cdot \xi)(h) = \xi(g^{-1}h)$ for $\xi \in \ell^2(G, H_\tau)$ and $g, h \in G$, implements the action α on $\pi^{\alpha}_{\tau}(A)$, so we get a continuous extension of the action $\alpha : G \curvearrowright \pi^{\alpha}_{\tau}(A)''$ on the weak closure.

Notice that $\pi_{\tau}^{\alpha}(A)'' \subseteq \prod_{g \in G} \pi_{\tau}(A)''$. The trace τ on A extends to a faithful normal trace on $\pi_{\tau}(A)''$ and, by composition with the natural quotient map $q_g : \prod_{g \in G} \pi_{\tau}(A)'' \to \pi_{\tau}(A)''$ onto the summand with index $g \in G$, also to a normal trace on $\prod_{g \in G} \pi_{\tau}(A)''$, which we will denote by $\tilde{\tau}_g$. Notice that $\tilde{\tau}_g \circ \pi_{\tau}^{\alpha} = \tau \circ \alpha_g^{-1}$. Let $(c_g)_{g \in G}$ be a sequence in (0, 1) such that $\sum_{g \in G} c_g = 1$. Then $\tilde{\tau} := \sum_{g \in G} c_g \tilde{\tau}_g$ defines a faithful normal tracial state on $\prod_{g \in G} \pi_{\tau}(A)''$ and hence also on the subalgebra $\pi_{\tau}^{\alpha}(A)''$. In this way we can form the tracial von Neumann algebra ultrapower $(\pi_{\tau}^{\alpha}(A)'')^{\omega}$. Note that, on bounded subsets of $\prod_{g \in G} \pi_{\tau}(A)''$, the strong operator topology is induced by the norm $\|\cdot\|_{2,\tilde{\tau}}$, or equivalently by the seminorms $\{\|\cdot\|_{2,\tilde{\tau}_g} \mid g \in G\}$. Since $\pi_{\tau}^{\alpha}(A)''$ is a von Neumann subalgebra, it follows that on bounded subsets its strong operator topology is also induced by (the restrictions of) these (semi)norms.

Remark 4.5. With the above notation and terminology, the condition in Theorem 4.2 that requires, for every $\varepsilon > 0$, $\ell \in \mathbb{N}$, and $\tau \in \overline{T_{\omega}(A)}^{w^*}$, the existence of contractions $(y_i)_{i \in \mathbb{N}}$ in $A^{\omega,b}$ such that

$$||h_m(a_1,...,a_{r_m},y_1,...,y_{s_m})||_{2,\tau} < \varepsilon \text{ for } m = 1,...,\ell$$

is equivalent to the following statement (by Kaplansky's density theorem): for every $\varepsilon > 0$, $\ell \in \mathbb{N}$, and $\tau \in \overline{T_{\omega}(A)}^{w^*}$, there exist contractions $(y'_i)_{i \in \mathbb{N}}$ in $\pi_{\tau}^{\alpha^{\omega}}(A^{\omega,b})''$ such that

$$||h_m(a_1, \ldots, a_{r_m}, y'_1, \ldots, y'_{s_m})||_{2,\tilde{\tau}} < \varepsilon \text{ for } m = 1, \ldots, \ell.$$

Making use of the tracial von Neumann algebra ultrapowers, this is also equivalent to the following statement: for every $\tau \in \overline{T_{\omega}(A)}^{w^*}$, there are contractions $(y_i'')_{i \in \mathbb{N}}$ in $(\pi_{\tau}^{\alpha^{\omega}}(A^{\omega,b})'')^{\kappa}$ such that $h_m(a_1, \ldots, a_{r_m}, y_1'', \ldots, y_s'') = 0$ for every $m \in \mathbb{N}$.

Theorem 4.6. Let A be a σ -unital C*-algebra with T(A) nonempty and compact. Let $\alpha : G \curvearrowright A$ be an action by an amenable countable discrete group G and assume it has local equivariant property Gamma with respect to bounded traces. Let ω and κ be two free ultrafilters on \mathbb{N} . Let $\delta : G \curvearrowright D$ be an action on a separable C*-algebra and let $B \subset D$ be a separable, δ -invariant C*-subalgebra. Suppose $\varphi : (B, \delta) \rightarrow (A^{\omega, b}, \alpha^{\omega})$ is an equivariant *-homomorphism. Then the following are equivalent:

(1) For every $\tau \in \overline{T_{\omega}(A)}^{w^*}$, there exists an equivariant *-homomorphism

$$\varphi^{\tau}: (D, \delta) \to ((\pi_{\tau}^{\alpha^{\omega}}(A^{\omega, \mathbf{b}})'')^{\kappa}, (\alpha^{\omega})^{\kappa})$$

such that $\varphi^{\tau}|_{B} = \pi_{\tau}^{\alpha^{\omega}} \circ \varphi$.

(2) There is an equivariant *-homomorphism $\bar{\varphi} : (D, \delta) \to (A^{\omega, b}, \alpha^{\omega})$ with $\bar{\varphi}|_B = \varphi$.

Proof. It is clear that (2) implies (1). To prove the other implication, take a countable dense $\mathbb{Q}[i]$ -*-subalgebra $C \subset D$ such that it is δ -invariant and such that $C \cap B$ is also dense in B. By inductively enlarging C we may in addition assume that, for each contraction $x \in C$, one has $1 - \sqrt{1 - x^* x} \in C$. Let \mathcal{P} denote the countable family of G-*-polynomials with coefficients in $A^{\omega, b}$ in the variables $\{X_c\}_{c \in C}$ encoding all relations in C:

- $g \cdot X_c X_{\delta_g(c)}$ for all $c \in C$ and $g \in G$,
- $\lambda X_c + X_{c'} X_{\lambda c + c'}$ for all $c, c' \in C$ and $\lambda \in \mathbb{Q}[i]$,
- $X_c X_{c'} X_{cc'}$ for $c, c' \in C$,
- $X_c^* X_{c^*}$ for $c \in C$,
- $\varphi(b) X_b$ for $b \in B \cap C$.

It follows from (1) that, for every $\tau \in \overline{T_{\omega}(A)}^{w^*}$, the equations in \mathcal{P} have exact solutions in $(\pi_{\tau}^{\alpha^{\omega}}(A^{\omega,b})'')^{\kappa}$. By Remark 4.5 this means precisely that all conditions to apply Theorem 4.2 are fulfilled, and we can find exact solutions to all equations in \mathcal{P} in $A^{\omega,b}$. This is equivalent to the existence of a $\mathbb{Q}[i]$ -linear, *-preserving, multiplicative, equivariant map $\bar{\varphi} : C \to A^{\omega,b}$ with $\bar{\varphi}|_{B\cap C} = \varphi|_{B\cap C}$. We observe that $\bar{\varphi}$ is contractive. Indeed, if $x \in C$ is a contraction, then $y = 1 - \sqrt{1 - x^*x}$ is a self-adjoint element also belonging to C, which satisfies

Hence

$$\bar{\varphi}(x)^*\bar{\varphi}(x) + \bar{\varphi}(y)^2 - 2\bar{\varphi}(y) = 0,$$

 $x^*x + y^2 - 2y = x^*x + (y - 1)^2 - 1 = 0.$

or equivalently,

$$\bar{\varphi}(x)^* \bar{\varphi}(x) + (1 - \bar{\varphi}(y))^2 = 1.$$

We see that $\bar{\varphi}(x)^* \bar{\varphi}(x)$ is a contraction, and hence $\bar{\varphi}(x)$ is as well. In conclusion, φ extends to an equivariant *-homomorphism $\bar{\varphi} : (D, \delta) \to (A^{\omega, b}, \alpha^{\omega})$ with $\bar{\varphi}_B = \varphi$.

In many cases of interest we get the following corollary from Proposition 2.4, which directly generalizes and recovers the technical machinery related to uniform property Gamma from the nondynamical setting; see [Castillejos et al. 2021b, Lemma 4.1]. We note that, upon close inspection of our proof so far, this particular corollary can be obtained based on [Castillejos et al. 2021b, Lemma 3.6] without relying on the preprint [Carrión et al. 2023a].

Corollary 4.7. Let A be a separable, simple, nuclear C*-algebra with T(A) nonempty and compact, and such that $T^+(A) = \mathbb{R}^{>0}T(A)$. Let $\alpha : G \cap A$ be an action by a countable amenable discrete group that has equivariant property Gamma. Then α satisfies the conclusion of Theorems 4.2 and 4.6.

Remark 4.8. For potential subsequent applications of the theory in this article, let us reflect on how we ended up with the main result of this section. It is worthwhile to note that the amenability of the group *G* is used (in the proof of Theorem 4.2) through the Følner condition exclusively for the purpose of having access to a finite set $H \Subset G$ that satisfies the conclusion of Lemma 3.4. At no other point in the whole chain of argument is it necessary to know that *H* is actually a set that is almost invariant with respect

to F, or anything else about H for that matter. This culminates in the following more explicit observation, which we suspect may be, at some point, interesting to consider for certain actions of nonamenable groups.

Let *A* be a σ -unital C*-algebra with T(A) nonempty and compact. Let $\alpha : G \curvearrowright A$ be an action of a countable discrete group. Suppose that, for all $\varepsilon > 0$ and $F \Subset G$, there exists a finite subset $H \Subset G$ that satisfies the same conclusion as in Lemma 3.4. If α has local equivariant property Gamma with respect to bounded traces, then α also satisfies the conclusion of Theorems 4.2 and 4.6.

5. Equivariant Jiang-Su stability

In this section we use the dynamical tracial local-to-global principle derived in the previous section combined with von Neumann algebraic results to conclude that, for actions of countable amenable groups on separable, simple, nuclear, finite, \mathcal{Z} -stable C*-algebras, equivariant property Gamma implies equivariant \mathcal{Z} -stability. Although one can get by with known variations of Ocneanu's theorem [1985], for many applications treated in this section, our most general results here need a more general McDuff-type theorem for actions of amenable groups on von Neumann algebras, which we import from our recent work [Szabó and Wouters 2024].

We begin by reducing the problem of equivariant \mathcal{Z} -stability to the existence of so-called tracially large c.p.c. order-zero maps $M_n \to F_{\omega}(A)^{\tilde{\alpha}_{\omega}}$ for $n \ge 2$. The argument is well known to experts and traces back to the work of Matui and Sato [2012]. It makes use of an equivariant version of their property (SI), for which the general framework needed here was developed in [Szabó 2021b].

Definition 5.1 [Szabó 2021b, Definition 2.5; Castillejos et al. 2023, Definition 1.3]. Let A be a separable, simple C*-algebra with $T^+(A) \neq \emptyset$.

- We say that a positive contraction f ∈ F_ω(A) is *tracially supported at* 1 if the following holds: for every nonzero positive element a ∈ P(A), there exists a constant κ = κ(f, a) > 0 such that, for every τ ∈ T_ω(A) with 0 < τ(a) < ∞, one has inf_{k∈ℕ} τ_a(f^k) ≥ κτ(a).
- (2) A positive element $e \in F_{\omega}(A)$ is called *tracially null* if $e \in \mathcal{J}_A$ in the sense of Definition 1.11.
- (3) Let *B* be a unital C*-algebra. A c.p.c. order-zero map $\phi : B \to F_{\omega}(A)$ is called *tracially large* if $\tau_a \circ \phi(1) = \tau(a)$ for all nonzero positive elements $a \in \mathcal{P}(A)$ and $\tau \in \widetilde{T}_{\omega}(A)$ with $\tau(a) < \infty$.

Remark 5.2. It follows from [Szabó 2021b, Proposition 2.4] that any of the conditions above hold for all nonzero positive elements $a \in \mathcal{P}(A)$ if and only if they hold for just a single such element, so in practice it suffices to check them for a single $a \in \mathcal{P}(A)_+ \setminus \{0\}$.

Definition 5.3 [Szabó 2021b, Definition 2.7]. Let *A* be a separable, simple C*-algebra with $T^+(A) \neq \emptyset$ and an action $\alpha : G \frown A$ of a countable discrete group. We say that α has *equivariant property* (*SI*) if the following holds:

Whenever $e, f \in F_{\omega}(A)^{\tilde{\alpha}_{\omega}}$ are two positive contractions such that f is tracially supported at 1 and e is tracially null, there exists a contraction $s \in F_{\omega}(A)^{\tilde{\alpha}_{\omega}}$ such that fs = s and $s^*s = e$.

It follows from [Szabó 2021b, Corollary 4.3] that all actions of amenable groups on nonelementary, separable, simple, nuclear C*-algebras with strict comparison have property (SI). Combined with the following theorem, it gives a powerful sufficient criterion for equivariant Jiang–Su stability. This is not new to the experts but has never been formally stated in this generality before, so we shall give the proof for the reader's convenience.

Theorem 5.4. Let A be a separable, simple C*-algebra with $T^+(A) \neq \emptyset$, and let $\alpha : G \curvearrowright A$ be an action of a countable discrete group with equivariant property (SI). Then α is equivariantly \mathcal{Z} -stable if and only if, for every $n \in \mathbb{N}$, there exists a unital *-homomorphism $M_n \to (A^{\omega} \cap A')^{\alpha^{\omega}}$.

Proof. Since the "only if" part can be obtained with the standard argument sketched in Remark 2.2, we prove the "if" part. Given $n \in \mathbb{N}$, let $\phi' : M_n \to (A^{\omega} \cap A')^{\alpha^{\omega}}$ be a unital *-homomorphism. By Proposition 1.18, we can find a tracially large c.p.c. order-zero map $\phi : M_n \to F_{\omega}(A)^{\tilde{\alpha}_{\omega}}$ that lifts ϕ' . Set $e := 1_{F_{\omega}(A)} - \phi(1)$, and set $f := \phi(e_{1,1})$. Both are positive contractions in $F_{\omega}(A)^{\tilde{\alpha}_{\omega}}$. Since ϕ is tracially large, we can conclude immediately that e is tracially null. Since $e_{1,1}$ is a projection, it follows that $\phi(e_{1,1}) - \phi(e_{1,1})^m$ is tracially null for any $m \ge 1$. Moreover, for every $\tau \in \widetilde{T}_{\omega}(A)$ and $a \in \mathcal{P}(A)_+ \setminus \{0\}$ such that $\tau(a) < \infty$, the functional $\tau_a \circ \phi$ is a bounded trace and therefore a multiple of the unique tracial state on M_n . So, for every $k \in \mathbb{N}$, we have

$$\tau_a(f^k) = \tau_a(\phi(e_{1,1})^k) = \tau_a(\phi(e_{1,1})) = \frac{1}{n}\tau(a).$$

This proves that f is tracially supported at 1.

Since α has equivariant property (SI), we can find a contraction $s \in F_{\omega}(A)^{\tilde{\alpha}_{\omega}}$ such that fs = s and $s^*s = e$. By [Rørdam and Winter 2010, Theorem 5.1], this implies the existence of a unital *-homomorphism from the dimension drop algebra $Z_{n,n+1}$ into $F_{\omega}(A)^{\tilde{\alpha}_{\omega}}$. As \mathcal{Z} is an inductive limit of those algebras, we find a unital *-homomorphism $\mathcal{Z} \to F_{\omega}(A)^{\tilde{\alpha}_{\omega}}$.¹⁵ This implies equivariant \mathcal{Z} -stability by [Szabó 2018b, Corollary 3.8].

We shall now prove that, for actions of amenable groups on simple nuclear \mathcal{Z} -stable C*-algebras, equivariant uniform property Gamma is equivalent to equivariant \mathcal{Z} -stability. We end up giving two separate arguments to prove this result in two cases. Firstly, we prove this result for actions on C*-algebras that have a compact nonempty tracial state space and no unbounded traces, for which it is sufficient to appeal to Corollary 4.7. Secondly, we prove the result in full generality, but this requires the full power of our theory based on the results from [Carrión et al. 2023a].

Let us proceed in the first case.

Theorem 5.5. Let A be a separable, nuclear, simple \mathbb{Z} -stable C*-algebra with T(A) nonempty and compact, and such that $T^+(A) = \mathbb{R}^{>0}T(A)$. Let $\alpha : G \curvearrowright A$ be an action of a countable discrete amenable group. If α has equivariant property Gamma, then α is equivariantly \mathbb{Z} -stable.

Proof. By Theorem 5.4, it suffices to construct a unital *-homomorphism $M_n \to (A^{\omega} \cap A')^{\alpha^{\omega}}$ for $n \ge 2$. We appeal to Corollary 4.7 and hence know that α obeys the conclusion of Theorem 4.2.

¹⁵This is a standard reindexation trick. Alternatively one can deduce this for example by a combination of Corollary 3.9 and Lemma 4.2 in [Barlak and Szabó 2016].

Let $(a_k)_{k \in \mathbb{N}}$ be a dense sequence in *A*. Then the existence of such a desired *-homomorphism is equivalent to the existence of elements $e_{1,1}, e_{2,1}, \ldots, e_{n,1} \in A^{\omega}$ satisfying the equations

$$e_{i,1}e_{j,1} = 0$$
, $e_{i,1}^*e_{j,1} = \delta_{ij}e_{1,1}$, $e_{1,1}^2 = e_{1,1}$, $\alpha_g^{\omega}(e_{i,1}) = e_{i,1}$, and $a_k e_{i,1} - e_{i,1}a_k = 0$

for all $g \in G$, i, j = 2, ..., n, and $k \in \mathbb{N}$. By Theorem 4.2, it suffices to show that, for each $\varepsilon > 0$, finite subset $F \subseteq G$, $m \in \mathbb{N}$, and every tracial state $\tau \in T(A)$, there exists contractions $f_{1,1}, f_{2,1}, ..., f_{n,1} \in A$ satisfying

$$\begin{split} \|f_{i,1}f_{j,1}\|_{2,\tau} < \varepsilon, \quad \|f_{i,1}^*f_{j,1} - \delta_{ij}f_{1,1}\|_{2,\tau} < \varepsilon, \quad \|f_{1,1}^2 - f_{1,1}\|_{2,\tau} < \varepsilon, \\ \|\alpha_g(f_{i,1}) - f_{i,1}\|_{2,\tau} < \varepsilon, \quad \text{and} \quad \|a_k f_{i,1} - f_{i,1} a_k\|_{2,\tau} < \varepsilon \end{split}$$

for all $g \in F$, i, j = 2, ..., n, and $1 \le k \le m$. This is the case, however, if and only if, for every $\tau \in T(A)$, there exists a unital equivariant *-homomorphism $M_n \to ((\pi_{\tau}^{\alpha}(A)'')^{\omega} \cap A')^{\alpha^{\omega}}$. Since A is nuclear, the tracial von Neumann algebra $N_{\tau} := \pi_{\tau}^{\alpha}(A)''$ is injective; see for example [Blackadar 2006, Theorem IV.2.2.13]. Since it does not have any direct summand of type I, it follows from Connes' theorem [1976] that $N_{\tau} \otimes \mathcal{R} \cong N_{\tau}$. Hence the claim follows directly from [Szabó and Wouters 2024, Theorem A].

Next we proceed in the second and more general case.

Lemma 5.6. Let A be a σ -unital C*-algebra with T(A) nonempty and compact. Let α : $G \curvearrowright A$ be an action by a countable discrete amenable group G and assume it has local equivariant property Gamma with respect to bounded traces. Then the following are equivalent:

- (1) For all $\tau \in T(A)$, there exists a unital *-homomorphisms $\varphi^{\tau} : M_n \to ((\pi^{\alpha}_{\tau}(A)'')^{\omega})^{\alpha^{\omega}}$.
- (2) There exists a unital *-homomorphism $\varphi: M_n \to (A^{\omega,b})^{\alpha^{\omega}}$.

Proof. For each $\tau \in T(A)$, the map π_{τ}^{α} induces a unital *-homomorphism

$$(A^{\omega,\mathbf{b}})^{\alpha^{\omega}} \to ((\pi^{\alpha}_{\tau}(A)'')^{\omega})^{\alpha^{\omega}},$$

so (2) implies (1). We use the fact that α has local equivariant property Gamma to prove the other implication. By Theorem 4.6 the following are equivalent:

(a) For all $\tau \in \overline{T_{\omega}(A)}^{w^*}$, there exists a unital equivariant *-homomorphism

$$\varphi^{\tau}: (M_n, \operatorname{id}_{M_n}) \to ((\pi_{\tau}^{\alpha^{\omega}}(A^{\omega, \mathfrak{b}})'')^{\kappa}, (\alpha^{\omega})^{\kappa}).$$

(b) There exists a unital equivariant *-homomorphism $\varphi : (M_n, id_{M_n}) \to (A^{\omega,b}, \alpha^{\omega}).$

Statement (b) is equivalent to statement (2) above. Hence, in order to prove the implication it suffices to prove that statement (1) implies (a). Take $\tau \in \overline{T_{\omega}(A)}^{w^*}$ and denote its restriction to A by σ . The canonical map $A \to A^{\omega,b}$ induces an equivariant *-homomorphism $(\pi_{\sigma}^{\alpha}(A), \alpha) \to (\pi_{\tau}^{\alpha^{\omega}}(A^{\omega,b})'', \alpha^{\omega})$ that is continuous on the unit ball with respect to the strong operator topology. Hence, it can be extended to a unital equivariant *-homomorphism $\pi_{\sigma}^{\alpha}(A)'' \to \pi_{\tau}^{\alpha^{\omega}}(A^{\omega,b})''$. Combining this with (1), this means we can find a unital *-homomorphism $M_n \to ((\pi_{\tau}^{\alpha^{\omega}}(A^{\omega,b})'')^{\kappa})^{(\alpha^{\omega})^{\kappa}}$. This ends the proof.

Theorem 5.7. Let A be a separable, simple, nuclear \mathbb{Z} -stable C*-algebra such that $T^+(A) \neq \emptyset$. Let $\alpha : G \cap A$ be an action of a countable discrete amenable group. If α has equivariant property Gamma, then α is equivariantly \mathbb{Z} -stable.

Proof. Combining [Szabó 2021b, Corollary 4.3] with Theorem 5.4, we see that, given $n \ge 2$, it suffices to construct a unital *-homomorphism $M_n \to (A^{\omega} \cap A')^{\alpha^{\omega}}$. For convenience, let us specify a (possibly different) free ultrafilter κ on \mathbb{N} . It suffices to show that we can construct a unital *-homomorphism $M_n \to ((A^{\omega} \cap A')^{\kappa, b})^{(\alpha^{\omega})^{\kappa}}$, as a reindexation trick will then yield the required unital *-homomorphism $M_n \to (A^{\omega} \cap A')^{\alpha^{\omega}}$. By Theorem 2.6, we conclude that $\alpha^{\omega} = G \cap A^{\omega} \cap A'$ has local equivariant property Gamma with respect to bounded traces. Thus, by Lemma 5.6, it suffices to prove that, for all $\tau \in T(A^{\omega} \cap A')$, there exists a unital *-homomorphism $\varphi^{\tau} : M_n \to ((\pi_{\tau}^{\alpha^{\omega}}(A^{\omega} \cap A')'')^{\kappa})^{(\alpha^{\omega})^{\kappa}}$. We note that the C*-algebra A has uniform property Gamma. Using the same trick as in the proof of Proposition 2.5, we conclude that $A^{\omega} \cap A' \cong B^{\omega} \cap B'$ for a hereditary subalgebra $B \subset A \otimes \mathbb{K}$ such that $T^+(B) = \mathbb{R}^{>0}T(B)$ and T(B) is compact. Using [Castillejos et al. 2022, Theorem 4.6], we can hence conclude that there exists a unital *-homomorphism $M_2 \to A^{\omega} \cap A'$. By a standard reindexation trick, we can argue that such a *-homomorphism can be chosen to additionally commute with any specified separable subset of $A^{\omega} \cap A'$.

Fix $\tau \in T(A^{\omega} \cap A')$. We show first that $N_{\tau} := \pi_{\tau}^{\alpha^{\omega}} (A^{\omega} \cap A')''$ contains a $\|\cdot\|_{2,\tau}$ -separable, α^{ω} -invariant von Neumann subalgebra that tensorially absorbs the hyperfinite II₁-factor. By the aforementioned property of A, we can find a unital embedding $\phi_1 : M_2 \to \pi_{\tau}^{\alpha^{\omega}} (A^{\omega} \cap A') \subseteq N_{\tau}$. Set

$$B_1 := C^* \bigg(\bigcup_{g \in G} \alpha_g^{\omega}(\phi_1(M_2)) \bigg).$$

Using that B_1 is a separable subquotient of $A^{\omega} \cap A'$, we again use the aforementioned property of A and find a unital embedding

$$\phi_2: M_2 \to \pi_\tau^{\alpha^\omega}(A_\omega \cap A') \cap B_1' \subseteq N_\tau \cap B_1'$$

Set

$$B_2 := C^* \bigg(B_1 \cup \bigcup_{g \in G} \alpha_g^{\omega}(\phi_2(M_2)) \bigg).$$

Carry on with this procedure inductively, i.e., given the C*-algebra $B_i \subset \pi_\tau^{\alpha^{\omega}}(A^{\omega} \cap A')$, find a unital *-homomorphism $\phi_{i+1}: M_2 \to \pi_\tau^{\alpha^{\omega}}(A^{\omega} \cap A') \cap B'_i$ and set

$$B_{i+1} := C^* \bigg(B_i \cup \bigcup_{g \in G} \alpha_g^{\omega}(\phi_{i+1}(M_2)) \bigg).$$

Define $\mathcal{B} := \overline{\bigcup_{i \in \mathbb{N}} B_i} \|\cdot\|_{2,\tau} \subset N_{\tau}$. Then \mathcal{B} is a $\|\cdot\|_{2,\tau}$ -separable, α^{ω} -invariant von Neumann subalgebra of N_{τ} such that additionally $\mathcal{B} \cong \mathcal{B} \otimes \mathcal{R}$ by [Szabó and Wouters 2024, Corollary 3.8] because it satisfies the McDuff-type criterion (existence of a unital *-homomorphism $\mathcal{R} \to \mathcal{B}_{\omega}$) by construction. Denote the restriction of α^{ω} to \mathcal{B} by β . By [Szabó and Wouters 2024, Theorem A], it follows that β is cocycle conjugate to $\beta \otimes id_{\mathcal{R}}$. In particular, we can find a unital *-homomorphism

$$M_n \to (\mathcal{B}^{\kappa})^{\beta^{\kappa}} \subset ((\pi_{\tau}^{\alpha^{\omega}} (A^{\omega} \cap A')'')^{\kappa})^{(\alpha^{\omega})^{\kappa}}.$$

Acknowledgements

We would like to express our sincere gratitude to the anonymous reviewers for their valuable comments that helped improve this article.

References

- [Anantharaman and Popa 2014] C. Anantharaman and S. Popa, "An introduction to II₁ factors", preprint, 2014, available at https://www.math.ucla.edu/~popa/Books/IIunV15.pdf.
- [Ando and Haagerup 2014] H. Ando and U. Haagerup, "Ultraproducts of von Neumann algebras", *J. Funct. Anal.* **266**:12 (2014), 6842–6913. MR Zbl
- [Antoine et al. 2022] R. Antoine, F. Perera, L. Robert, and H. Thiel, "C*-algebras of stable rank one and their Cuntz semigroups", *Duke Math. J.* **171**:1 (2022), 33–99. MR Zbl
- [Barlak and Szabó 2016] S. Barlak and G. Szabó, "Sequentially split *-homomorphisms between C*-algebras", *Int. J. Math.* **27**:13 (2016), art. id. 1650105. MR Zbl
- [Blackadar 2006] B. Blackadar, *Operator algebras: theory of* C*-*algebras and von Neumann algebras*, Encycl. Math. Sci. **122**, Springer, 2006. MR Zbl
- [Blackadar and Cuntz 1982] B. E. Blackadar and J. Cuntz, "The structure of stable algebraically simple C*-algebras", *Amer. J. Math.* **104**:4 (1982), 813–822. MR Zbl
- [Blanchard and Kirchberg 2004] E. Blanchard and E. Kirchberg, "Non-simple purely infinite C*-algebras: the Hausdorff case", *J. Funct. Anal.* **207**:2 (2004), 461–513. MR Zbl
- [Bosa et al. 2019] J. Bosa, N. P. Brown, Y. Sato, A. Tikuisis, S. White, and W. Winter, *Covering dimension of* C*-algebras and 2-coloured classification, Mem. Amer. Math. Soc. **1233**, Amer. Math. Soc., Providence, RI, 2019. MR Zbl
- [Carrión et al. 2023a] J. R. Carrión, J. Castillejos, S. Evington, J. Gabe, C. Schafhauser, A. Tikuisis, and S. White, "Tracially complete C*-algebras", preprint, 2023. arXiv 2310.20594
- [Carrión et al. 2023b] J. R. Carrión, J. Gabe, C. Schafhauser, A. Tikuisis, and S. White, "Classifying *-homomorphisms, I: Unital simple nuclear C*-algebras", preprint, 2023. arXiv 2307.06480
- [Castillejos and Evington 2020] J. Castillejos and S. Evington, "Nuclear dimension of simple stably projectionless C*-algebras", *Anal. PDE* **13**:7 (2020), 2205–2240. MR Zbl
- [Castillejos and Evington 2021] J. Castillejos and S. Evington, "Stabilising uniform property Γ", *Proc. Amer. Math. Soc.* **149**:11 (2021), 4725–4737. MR Zbl
- [Castillejos et al. 2021a] J. Castillejos, S. Evington, A. Tikuisis, and S. White, "Classifying maps into uniform tracial sequence algebras", *Münster J. Math.* 14:2 (2021), 265–281. MR Zbl
- [Castillejos et al. 2021b] J. Castillejos, S. Evington, A. Tikuisis, S. White, and W. Winter, "Nuclear dimension of simple C*-algebras", *Invent. Math.* **224**:1 (2021), 245–290. MR Zbl
- [Castillejos et al. 2022] J. Castillejos, S. Evington, A. Tikuisis, and S. White, "Uniform property Γ", *Int. Math. Res. Not.* **2022**:13 (2022), 9864–9908. MR Zbl
- [Castillejos et al. 2023] J. Castillejos, K. Li, and G. Szabó, "On tracial \mathcal{Z} -stability of simple non-unital C*-algebras", *Canad. J. Math.* (online publication April 2023).
- [Connes 1976] A. Connes, "Classification of injective factors: cases II₁, II_{∞}, III_{λ}, $\lambda \neq 1$ ", Ann. of Math. (2) **104**:1 (1976), 73–115. MR Zbl
- [Connes 1977] A. Connes, "Periodic automorphisms of the hyperfinite factor of type II₁", *Acta Sci. Math. (Szeged)* **39**:1-2 (1977), 39–66. MR Zbl
- [Elliott 2010] G. A. Elliott, "Towards a theory of classification", Adv. Math. 223:1 (2010), 30-48. MR Zbl
- [Elliott et al. 2011] G. A. Elliott, L. Robert, and L. Santiago, "The cone of lower semicontinuous traces on a C*-algebra", *Amer. J. Math.* **133**:4 (2011), 969–1005. MR Zbl
- [Elliott et al. 2020] G. A. Elliott, G. Gong, H. Lin, and Z. Niu, "The classification of simple separable KK-contractible C*-algebras with finite nuclear dimension", *J. Geom. Phys.* **158** (2020), art. id. 103861. MR Zbl

- [Elliott et al. 2024] G. A. Elliott, G. Gong, H. Lin, and Z. Niu, "On the classification of simple C*-algebras with finite decomposition rank, II", *J. Noncommut. Geom.* (online publication June 2024).
- [Evans and Kishimoto 1997] D. E. Evans and A. Kishimoto, "Trace scaling automorphisms of certain stable AF algebras", *Hokkaido Math. J.* **26**:1 (1997), 211–224. MR Zbl
- [Fu et al. 2022] X. Fu, K. Li, and H. Lin, "Tracial approximate divisibility and stable rank one", *J. Lond. Math. Soc.* (2) **106**:4 (2022), 3008–3042. MR Zbl
- [Gabe and Szabó 2022] J. Gabe and G. Szabó, "The stable uniqueness theorem for equivariant Kasparov theory", 2022. To appear in *Amer. J. Math.* arXiv 2202.09809
- [Gabe and Szabó 2024] J. Gabe and G. Szabó, "The dynamical Kirchberg–Phillips theorem", *Acta Math.* 232:1 (2024), 1–77. MR Zbl
- [Gardella and Lupini 2021] E. Gardella and M. Lupini, "Group amenability and actions on *Z*-stable C*-algebras", *Adv. Math.* **389** (2021), art. id. 107931. MR Zbl
- [Gardella et al. 2022] E. Gardella, I. Hirshberg, and A. Vaccaro, "Strongly outer actions of amenable groups on \mathcal{Z} -stable nuclear C*-algebras", *J. Math. Pures Appl.* (9) **162** (2022), 76–123. MR Zbl
- [Gardella et al. 2024] E. Gardella, S. Geffen, J. Kranz, P. Naryshkin, and A. Vaccaro, "Tracially amenable actions and purely infinite crossed products", *Math. Ann.* (online publication March 2024).
- [Gong and Lin 2020] G. Gong and H. Lin, "On classification of non-unital amenable simple C*-algebras, II", *J. Geom. Phys.* **158** (2020), art. id. 103865. MR Zbl
- [Gong and Lin 2022] G. Gong and H. Lin, "On classification of nonunital amenable simple C*-algebras, III : The range and the reduction", *Ann. K-Theory* 7:2 (2022), 279–384. MR Zbl
- [Gong et al. 2020a] G. Gong, H. Lin, and Z. Niu, "A classification of finite simple amenable Z-stable C*-algebras, I: C*-algebras with generalized tracial rank one", C. R. Math. Acad. Sci. Soc. R. Can. 42:3 (2020), 63–450. MR Zbl
- [Gong et al. 2020b] G. Gong, H. Lin, and Z. Niu, "A classification of finite simple amenable \mathcal{Z} -stable C*-algebras, II: C*-algebras with rational generalized tracial rank one", C. R. Math. Acad. Sci. Soc. R. Can. 42:4 (2020), 451–539. MR Zbl
- [Haagerup 1975] U. Haagerup, "The standard form of von Neumann algebras", Math. Scand. 37:2 (1975), 271–283. MR Zbl
- [Haagerup 1987] U. Haagerup, "Connes' bicentralizer problem and uniqueness of the injective factor of type III₁", *Acta Math.* **158**:1-2 (1987), 95–148. MR Zbl
- [Haagerup 2014] U. Haagerup, "Quasitraces on exact C*-algebras are traces", C. R. Math. Acad. Sci. Soc. R. Can. **36**:2-3 (2014), 67–92. MR Zbl
- [Izumi 2004a] M. Izumi, "Finite group actions on C*-algebras with the Rohlin property, I", *Duke Math. J.* **122**:2 (2004), 233–280. MR Zbl
- [Izumi 2004b] M. Izumi, "Finite group actions on C*-algebras with the Rohlin property, II", *Adv. Math.* **184**:1 (2004), 119–160. MR Zbl
- [Izumi and Matui 2010] M. Izumi and H. Matui, " \mathbb{Z}^2 -actions on Kirchberg algebras", *Adv. Math.* **224**:2 (2010), 355–400. MR Zbl
- [Izumi and Matui 2021a] M. Izumi and H. Matui, "Poly-ℤ group actions on Kirchberg algebras, I", *Int. Math. Res. Not.* **2021**:16 (2021), 12077–12154. MR Zbl
- [Izumi and Matui 2021b] M. Izumi and H. Matui, "Poly-ℤ group actions on Kirchberg algebras, II", *Invent. Math.* **224**:3 (2021), 699–766. MR Zbl
- [Jiang and Su 1999] X. Jiang and H. Su, "On a simple unital projectionless C*-algebra", *Amer. J. Math.* **121**:2 (1999), 359–413. MR Zbl
- [Jones 1980] V. F. R. Jones, *Actions of finite groups on the hyperfinite type* II₁ *factor*, Mem. Amer. Math. Soc. 237, Amer. Math. Soc., Providence, RI, 1980. MR Zbl
- [Jones 1983] V. F. R. Jones, "A converse to Ocneanu's theorem", J. Operator Theory 10:1 (1983), 61–63. MR Zbl
- [Kasparov 1988] G. G. Kasparov, "Equivariant *KK*-theory and the Novikov conjecture", *Invent. Math.* **91**:1 (1988), 147–201. MR Zbl
- [Katayama et al. 1998] Y. Katayama, C. E. Sutherland, and M. Takesaki, "The characteristic square of a factor and the cocycle conjugacy of discrete group actions on factors", *Invent. Math.* **132**:2 (1998), 331–380. MR Zbl

EQUIVARIANT PROPERTY GAMMA AND THE TRACIAL LOCAL-TO-GLOBAL PRINCIPLE FOR C*-DYNAMICS 1431

- [Katsura and Matui 2008] T. Katsura and H. Matui, "Classification of uniformly outer actions of \mathbb{Z}^2 on UHF algebras", *Adv. Math.* **218**:3 (2008), 940–968. MR Zbl
- [Kawahigashi et al. 1992] Y. Kawahigashi, C. E. Sutherland, and M. Takesaki, "The structure of the automorphism group of an injective factor and the cocycle conjugacy of discrete abelian group actions", *Acta Math.* **169**:1-2 (1992), 105–130. MR Zbl [Kerr 2020] D. Kerr, "Dimension, comparison, and almost finiteness", *J. Eur. Math. Soc.* **22**:11 (2020), 3697–3745. MR Zbl
- [Kerr and Naryshkin 2021] D. Kerr and P. Naryshkin, "Elementary amenability and almost finiteness", preprint, 2021. arXiv 2107.05273
- [Kerr and Szabó 2020] D. Kerr and G. Szabó, "Almost finiteness and the small boundary property", *Comm. Math. Phys.* **374**:1 (2020), 1–31. MR Zbl
- [Kirchberg 1995] E. Kirchberg, "Exact C*-algebras, tensor products, and the classification of purely infinite algebras", pp. 943–954 in *Proceedings of the International Congress of Mathematicians*, *II* (Zürich, 1994), edited by S. D. Chatterji, Birkhäuser, Basel, 1995. MR Zbl
- [Kirchberg 2006] E. Kirchberg, "Central sequences in C*-algebras and strongly purely infinite algebras", pp. 175–231 in *Operator algebras* (Oslo, 2004), edited by O. Brattelli et al., Abel Symp. **1**, Springer, 2006. MR Zbl
- [Kirchberg and Rørdam 2002] E. Kirchberg and M. Rørdam, "Infinite non-simple C*-algebras: absorbing the Cuntz algebra \mathcal{O}_{∞} ", *Adv. Math.* **167**:2 (2002), 195–264. MR Zbl
- [Kirchberg and Rørdam 2014] E. Kirchberg and M. Rørdam, "Central sequence C*-algebras and tensorial absorption of the Jiang–Su algebra", *J. Reine Angew. Math.* **695** (2014), 175–214. MR Zbl
- [Kishimoto 1995] A. Kishimoto, "The Rohlin property for automorphisms of UHF algebras", *J. Reine Angew. Math.* **465** (1995), 183–196. MR Zbl
- [Kishimoto 1998a] A. Kishimoto, "Automorphisms of AT algebras with the Rohlin property", J. Operator Theory **40**:2 (1998), 277–294. MR Zbl
- [Kishimoto 1998b] A. Kishimoto, "Unbounded derivations in AT algebras", J. Funct. Anal. 160:1 (1998), 270–311. MR Zbl
- [Liao and Tikuisis 2022] H.-C. Liao and A. Tikuisis, "Almost finiteness, comparison, and tracial *Z*-stability", *J. Funct. Anal.* **282**:3 (2022), art. id. 109309. MR Zbl
- [Lin 1991] H. X. Lin, "Simple C*-algebras with continuous scales and simple corona algebras", *Proc. Amer. Math. Soc.* **112**:3 (1991), 871–880. MR Zbl
- [Lin 2023] H. Lin, "Hereditary uniform property Γ", Sci. China Math. 66:8 (2023), 1813–1830. MR Zbl
- [Masuda 2007] T. Masuda, "Evans–Kishimoto type argument for actions of discrete amenable groups on McDuff factors", *Math. Scand.* **101**:1 (2007), 48–64. MR Zbl
- [Masuda 2013] T. Masuda, "Unified approach to the classification of actions of discrete amenable groups on injective factors", *J. Reine Angew. Math.* **683** (2013), 1–47. MR Zbl
- [Matui 2008] H. Matui, "Classification of outer actions of \mathbb{Z}^N on \mathcal{O}_2 ", Adv. Math. 217:6 (2008), 2872–2896. MR Zbl
- [Matui 2010] H. Matui, " \mathbb{Z} -actions on AH algebras and \mathbb{Z}^2 -actions on AF algebras", *Comm. Math. Phys.* **297**:2 (2010), 529–551. MR Zbl
- [Matui 2011] H. Matui, "ℤ^N-actions on UHF algebras of infinite type", J. Reine Angew. Math. 657 (2011), 225–244. MR Zbl
- [Matui and Sato 2012] H. Matui and Y. Sato, "Z-stability of crossed products by strongly outer actions", *Comm. Math. Phys.* **314**:1 (2012), 193–228. MR Zbl
- [Matui and Sato 2014a] H. Matui and Y. Sato, "Decomposition rank of UHF-absorbing C*-algebras", *Duke Math. J.* 163:14 (2014), 2687–2708. MR Zbl
- [Matui and Sato 2014b] H. Matui and Y. Sato, "Z-stability of crossed products by strongly outer actions, II", *Amer. J. Math.* **136**:6 (2014), 1441–1496. MR Zbl
- [Murray and von Neumann 1943] F. J. Murray and J. von Neumann, "On rings of operators, IV", Ann. of Math. (2) 44 (1943), 716–808. MR Zbl
- [Nakamura 2000] H. Nakamura, "Aperiodic automorphisms of nuclear purely infinite simple C*-algebras", *Ergodic Theory Dynam. Systems* **20**:6 (2000), 1749–1765. MR Zbl
- [Ocneanu 1985] A. Ocneanu, *Actions of discrete amenable groups on von Neumann algebras*, Lecture Notes in Math. **1138**, Springer, 1985. MR Zbl

- [Pedersen 1979] G. K. Pedersen, C*-algebras and their automorphism groups, Lond. Math. Soc. Monogr. 14, Academic Press, London, 1979. MR Zbl
- [Powers and Størmer 1970] R. T. Powers and E. Størmer, "Free states of the canonical anticommutation relations", *Comm. Math. Phys.* **16** (1970), 1–33. MR Zbl
- [Rørdam 2003] M. Rørdam, "A simple C*-algebra with a finite and an infinite projection", *Acta Math.* **191**:1 (2003), 109–142. MR Zbl
- [Rørdam 2004] M. Rørdam, "The stable and the real rank of \mathcal{Z} -absorbing C*-algebras", Int. J. Math. 15:10 (2004), 1065–1084. MR Zbl
- [Rørdam and Winter 2010] M. Rørdam and W. Winter, "The Jiang–Su algebra revisited", *J. Reine Angew. Math.* **642** (2010), 129–155. MR Zbl
- [Sato 2010] Y. Sato, "The Rohlin property for automorphisms of the Jiang–Su algebra", *J. Funct. Anal.* **259**:2 (2010), 453–476. MR Zbl
- [Sato 2019] Y. Sato, "Actions of amenable groups and crossed products of Z-absorbing C*-algebras", pp. 189–210 in *Operator algebras and mathematical physics*, edited by M. Izumi et al., Adv. Stud. Pure Math. **80**, Math. Soc. Japan, Tokyo, 2019. MR Zbl
- [Sato et al. 2015] Y. Sato, S. White, and W. Winter, "Nuclear dimension and Z-stability", *Invent. Math.* 202:2 (2015), 893–921. MR Zbl
- [Schafhauser 2020] C. Schafhauser, "Subalgebras of simple AF-algebras", Ann. of Math. (2) 192:2 (2020), 309–352. MR Zbl
- [Sutherland and Takesaki 1989] C. E. Sutherland and M. Takesaki, "Actions of discrete amenable groups on injective factors of type III_{λ}, $\lambda \neq 1$ ", *Pacific J. Math.* **137**:2 (1989), 405–444. MR Zbl
- [Suzuki 2021] Y. Suzuki, "Equivariant O_2 -absorption theorem for exact groups", *Compos. Math.* **157**:7 (2021), 1492–1506. MR Zbl
- [Szabó 2018a] G. Szabó, "Equivariant Kirchberg–Phillips-type absorption for amenable group actions", *Comm. Math. Phys.* **361**:3 (2018), 1115–1154. MR Zbl
- [Szabó 2018b] G. Szabó, "Strongly self-absorbing C*-dynamical systems", *Trans. Amer. Math. Soc.* **370**:1 (2018), 99–130. MR Zbl
- [Szabó 2018c] G. Szabó, "Strongly self-absorbing C*-dynamical systems, II", *J. Noncommut. Geom.* **12**:1 (2018), 369–406. MR Zbl
- [Szabó 2021a] G. Szabó, "The classification of Rokhlin flows on C*-algebras", *Comm. Math. Phys.* **382**:3 (2021), 2015–2070. MR Zbl
- [Szabó 2021b] G. Szabó, "Equivariant property (SI) revisited", Anal. PDE 14:4 (2021), 1199–1232. MR Zbl
- [Szabó 2021c] G. Szabó, "On a categorical framework for classifying C*-dynamics up to cocycle conjugacy", *J. Funct. Anal.* **280**:8 (2021), art. id. 108927. MR Zbl
- [Szabó and Wouters 2024] G. Szabó and L. Wouters, "Dynamical McDuff-type properties for group actions on von Neumann algebras", *J. Inst. Math. Jussieu* 23:6 (2024), 2593–2629. MR Zbl
- [Tikuisis et al. 2017] A. Tikuisis, S. White, and W. Winter, "Quasidiagonality of nuclear C*-algebras", *Ann. of Math.* (2) **185**:1 (2017), 229–284. MR Zbl
- [Toms 2008] A. S. Toms, "On the classification problem for nuclear C*-algebras", Ann. of Math. (2) **167**:3 (2008), 1029–1044. MR Zbl
- [Toms and Winter 2013] A. S. Toms and W. Winter, "Minimal dynamics and *K*-theoretic rigidity: Elliott's conjecture", *Geom. Funct. Anal.* 23:1 (2013), 467–481. MR Zbl
- [Villadsen 1999] J. Villadsen, "On the stable rank of simple C*-algebras", J. Amer. Math. Soc. 12:4 (1999), 1091–1102. MR Zbl
- [Wouters 2023] L. Wouters, "Equivariant Z-stability for single automorphisms on simple C*-algebras with tractable trace simplices", *Math. Z.* **304**:1 (2023), art. id. 22. MR Zbl

Received 28 Feb 2023. Revised 22 Apr 2024. Accepted 26 Jun 2024.

GÁBOR SZABÓ: gabor.szabo@kuleuven.be Department of Mathematics, KU Leuven, Leuven, Belgium

LISE WOUTERS: Leuven, Belgium





MULTIJET BUNDLES AND APPLICATION TO THE FINITENESS OF MOMENTS FOR ZEROS OF GAUSSIAN FIELDS

MICHELE ANCONA AND THOMAS LETENDRE

We define a notion of multijet for functions on \mathbb{R}^n , which extends the classical notion of jets in the sense that the multijet of a function is defined by contact conditions at several points. For all $p \ge 1$ we build a vector bundle of *p*-multijets, defined over a well-chosen compactification of the configuration space of *p* distinct points in \mathbb{R}^n . As an application, we prove that the linear statistics associated with the zero set of a centered Gaussian field on a Riemannian manifold have a finite *p*-th moment as soon as the field is of class C^p and its (p-1)-jet is nowhere degenerate. We prove a similar result for the linear statistics associated with the critical points of a Gaussian field and those associated with the vanishing locus of a holomorphic Gaussian field.

1.	Introduction	1433
2.	Notation: partitions and function spaces	1440
3.	Divided differences and Kergin interpolation	1441
4.	Evaluation maps and their kernels	1443
5.	Definition of the multijet bundles	1445
6.	Application to zeros of Gaussian fields	1454
7.	Multijets adapted to a differential operator	1465
8.	Multijets of holomorphic maps	1469
Acknowledgments		1474
References		1474

1. Introduction

This paper is concerned with two different but related problems. The first one is to define a natural notion of multijet for a C^k function on \mathbb{R}^n , generalizing the usual notion of k-jet. By multijet we mean that we want to consider a collection of jets at different points in \mathbb{R}^n and patch them together in a relevant way. The second one is to find natural conditions on a Gaussian field $f : \mathbb{R}^n \to \mathbb{R}^r$ ensuring that the (n-r)-dimensional volume of $f^{-1}(0) \cap \mathbb{B}$ admits finite higher moments, where \mathbb{B} stands for the unit ball in \mathbb{R}^n . One way to tackle this second problem is by considering the multijet of the random field f. In the following, we give more details about our contributions concerning the previous two problems, as well as some variations on these questions.

MSC2020: 51M15, 51M25, 55R80, 58A20, 60D05, 60G60.

Keywords: compactification of configuration spaces, Gaussian fields, moments, multijets, random submanifolds.

Letendre was partially supported by the ANR grant UniRanDom (ANR-17-CE40-0008) and a PEPS grant from the CNRS INSMI.

^{© 2025} The Authors, under license to MSP (Mathematical Sciences Publishers). Distributed under the Creative Commons Attribution License 4.0 (CC BY). Open Access made possible by subscribing institutions via Subscribe to Open.

1.1. *Multijet bundles.* Let us start by recalling some standard facts about jets. See [Saunders 1989] for background on this matter. Let $x \in \mathbb{R}^n$ and $k \ge 0$. Two smooth functions f and g on \mathbb{R}^n are said to have the same k-jet at x if f - g vanishes at x, as well as all its partial derivatives up to order k. Having the same k-jet at x is an equivalence relation on $C^{\infty}(\mathbb{R}^n)$, and the space $\mathcal{J}_k(\mathbb{R}^n)_x$ of k-jets at x is the set of equivalence classes for this relation. We denote by $j_k(f, x)$ the k-jet of f at x, that is, its class in $\mathcal{J}_k(\mathbb{R}^n)_x$. The map $j_k(\cdot, x)$ is a linear surjection from $C^{\infty}(\mathbb{R}^n)$ onto the finite-dimensional vector space $\mathcal{J}_k(\mathbb{R}^n)_x$. Of course, $j_k(f, x)$ makes sense even if f is only C^k and defined on some neighborhood of x.

Considering the family of *k*-jet spaces for all $x \in \mathbb{R}^n$, the set $\mathcal{J}_k(\mathbb{R}^n) = \bigsqcup_{x \in \mathbb{R}^n} \mathcal{J}_k(\mathbb{R}^n)_x$ is equipped with a natural vector bundle structure over \mathbb{R}^n . Then, if $\Omega \subset \mathbb{R}^n$ is open and $f : \Omega \to \mathbb{R}$ is \mathcal{C}^k , the map $j_k(f, \cdot)$ is a local section of $\mathcal{J}_k(\mathbb{R}^n) \to \mathbb{R}^n$ over Ω . These definitions are well-behaved with respect to smooth changes of coordinates, so one can define similarly the vector bundle of *k*-jets of functions on a manifold *M*. More generally, if $E \to M$ is a vector bundle over *M*, there is a corresponding vector bundle $\mathcal{J}_k(M, E) \to M$ of *k*-jets of sections of $E \to M$.

In this paper, we are interested in defining similarly a notion of multijet and the associated vector bundles. That is, we want to consider smooth functions on \mathbb{R}^n up to an equivalence relation defined by the vanishing of some derivatives at several points.

Let us make this more precise. Let $p \ge 1$ and $\Delta_p = \{(x_1, \ldots, x_p) \in (\mathbb{R}^n)^p \mid \exists i \ne j \text{ such that } x_i = x_j\}$ denote the diagonal in $(\mathbb{R}^n)^p$. Given $\underline{x} = (x_1, \ldots, x_p) \notin \Delta_p$, we say that f and g have the same multijet at \underline{x} if $f(x_i) = g(x_i)$ for all $i \in [[1, p]]$ (here we use the notation $[[a, b]] = [a, b] \cap \mathbb{N}$). This is an equivalence relation on functions, defined by the vanishing of f - g on the set $\{x_1, \ldots, x_p\} \subset \mathbb{R}^n$, that is, by p independent linear conditions. Thus the corresponding set of classes is a vector space of dimension p, which we denote by $\mathcal{MJ}_p(\mathbb{R}^n)_{\underline{x}}$. We also denote by $\mathrm{mj}_p(f, \underline{x})$ the class of f in this space, that is, its multijet at \underline{x} .

As will be explained later, this defines a vector bundle $\mathcal{MJ}_p(\mathbb{R}^n)$ of rank p over $(\mathbb{R}^n)^p \setminus \Delta_p$. Moreover, for all $\underline{x} \notin \Delta_p$ the linear map $\mathrm{mj}_p(\cdot, \underline{x}) : \mathcal{C}^{\infty}(\mathbb{R}^n) \to \mathcal{MJ}_p(\mathbb{R}^n)_{\underline{x}}$ is surjective, and, for all smooth f, its multijet $\mathrm{mj}_p(f, \cdot)$ is smooth. We would like to extend this picture over the whole of $(\mathbb{R}^n)^p$. Note that the surjectivity conditions rule out defining $\mathcal{MJ}_p(\mathbb{R}^n)$ as $\mathcal{J}_0(\mathbb{R}^n)^p$ with $\mathrm{mj}_p(f, \underline{x}) = (j_0(f, x_i))_{1 \leq i \leq p}$. When $\underline{x} \notin \Delta_p$, the previous notion of multijet is defined by p independent linear conditions: vanishing at each of the x_i . The main issue is that, when $\underline{x} \in \Delta_p$, these conditions are no longer independent and we need to replace them by another p-tuple of independent conditions.

A first natural idea is to look at vanishing with multiplicities. In dimension n = 1 this works very well. Let $\underline{x} \in \mathbb{R}^p$ be a permutation of $(y_1, \ldots, y_1, \ldots, y_m, \ldots, y_m)$, where $(y_j)_{1 \le j \le m} \in \mathbb{R}^m \setminus \Delta_m$ and y_j appears exactly $k_j + 1$ times. We say that f and g have the same multijet at \underline{x} if $(f - g)^{(k)}(y_j) = 0$ for all $j \in [\![1, m]\!]$ and $k \in [\![0, k_j]\!]$. In this sense, having the same multijet is equivalent to having the same Hermite interpolating polynomials at \underline{x} . Thus, we can define $\mathcal{MJ}_p(\mathbb{R})$ as the trivial bundle $\mathbb{R}_{p-1}[X] \times \mathbb{R}^p \to \mathbb{R}^p$ and $\mathrm{mj}_p(f, \underline{x})$ as the Hermite interpolating polynomial of f at \underline{x} .

If n > 1, the previous approach fails already for p = 2. Let us consider $x \in \mathbb{R}^n$ and the corresponding $\underline{x} = (x, x) \in \Delta_2$. Asking for the vanishing of f - g and its differential at x gives us n + 1 independent conditions, which define the 1-jet space $\mathcal{J}_1(\mathbb{R}^n)_x$. This space has dimension n + 1 > 2; hence it is too large to be the $\mathcal{MJ}_2(\mathbb{R}^n)_x$ we are looking for. The next natural idea is to ask only for the vanishing

at x of f - g and one of its directional derivatives. But whatever choice of directional derivative we make will lead to $\text{mj}_2(f, \cdot)$ not being continuous at x for most $f \in C^{\infty}(\mathbb{R}^n)$. Actually, one cannot extend $\mathcal{MJ}_2(\mathbb{R}^n)$ nicely over $(\mathbb{R}^n)^2$ if n > 1. However, we can extend it nicely over a larger space: the blow-up $\text{Bl}_{\Delta_2}((\mathbb{R}^n)^2)$ of $(\mathbb{R}^n)^2$ along Δ_2 . The key idea is that $\text{Bl}_{\Delta_2}((\mathbb{R}^n)^2)$ contains $(\mathbb{R}^n)^2 \setminus \Delta_2$ as a dense open subset and that points in the complement of $(\mathbb{R}^n)^2 \setminus \Delta_2$ can be described by the following data: a base point $x \in \mathbb{R}^n$ and a direction $u \in \mathbb{RP}^{n-1}$. This data tells us exactly which directional derivative to consider at the corresponding point in the exceptional locus of $\text{Bl}_{\Delta_2}((\mathbb{R}^n)^2)$. We will come back to this example later; see Example 5.9.

This long discussion shows that there is a natural way to define a multijet bundle $\mathcal{MJ}_p(\mathbb{R}^n)$ over the configuration space $(\mathbb{R}^n)^p \setminus \Delta_p$, but that it does not extend nicely over $(\mathbb{R}^n)^p$ in general. The case p = 2 hints that it might however be possible to define a natural multijet bundle over a slightly larger space, containing a copy of $(\mathbb{R}^n)^p \setminus \Delta_p$ as a dense open subset. Our first main contribution is to define such an object. Its main properties are gathered in the following statement, where $\mathcal{C}^k(\mathbb{R}^n, V)$ denotes the space of \mathcal{C}^k functions from \mathbb{R}^n to V.

Theorem 1.1 (existence of multijet bundles). Let $n \ge 1$ and $p \ge 1$ and let V be a real vector space of finite dimension $r \ge 1$. There exist a smooth manifold $C_p[\mathbb{R}^n]$ of dimension np without boundary and a smooth vector bundle $\mathcal{MJ}_p(\mathbb{R}^n, V) \to C_p[\mathbb{R}^n]$ of rank rp with the following properties:

(1) There exists a smooth proper surjection $\pi : C_p[\mathbb{R}^n] \to (\mathbb{R}^n)^p$ such that $\pi^{-1}((\mathbb{R}^n)^p \setminus \Delta_p)$ is a dense open subset of $C_p[\mathbb{R}^n]$, and π restricted to $\pi^{-1}((\mathbb{R}^n)^p \setminus \Delta_p)$ is a \mathcal{C}^{∞} -diffeomorphism onto $(\mathbb{R}^n)^p \setminus \Delta_p$.

(2) There exists a map $mj_p : \mathcal{C}^{p-1}(\mathbb{R}^n, V) \times C_p[\mathbb{R}^n] \to \mathcal{MJ}_p(\mathbb{R}^n, V)$ such that

- for all $z \in C_p[\mathbb{R}^n]$, the linear map $\operatorname{mj}_p(\cdot, z) : \mathcal{C}^{p-1}(\mathbb{R}^n, V) \to \mathcal{MJ}_p(\mathbb{R}^n, V)_z$ is surjective;
- for all $f \in \mathcal{C}^{l+p-1}(\mathbb{R}^n, V)$, the section $\operatorname{mj}_p(f, \cdot)$ of $\mathcal{MJ}_p(\mathbb{R}^n, V) \to C_p[\mathbb{R}^n]$ is \mathcal{C}^l .

(3) Let $z \in C_p[\mathbb{R}^n]$ be such that $\pi(z) = (x_1, \ldots, x_p) \notin \Delta_p$. Then for all $f \in C^{p-1}(\mathbb{R}^n, V)$ we have

$$\operatorname{mj}_{p}(f, z) = 0 \quad \Longleftrightarrow \quad \forall i \in \llbracket 1, p \rrbracket, \quad f(x_{i}) = 0.$$

(4) Let $z \in C_p[\mathbb{R}^n]$ be such that $\pi(z)$ is obtained as a permutation of $(y_1, \ldots, y_1, \ldots, y_m, \ldots, y_m)$, where y_j appears exactly $k_j + 1$ times and y_1, \ldots, y_m are pairwise distinct vectors in \mathbb{R}^n . Then, there exists a linear surjection $\Theta_z : \prod_{i=1}^m \mathcal{J}_{k_j}(\mathbb{R}^n, V)_{y_j} \to \mathcal{MJ}_p(\mathbb{R}^n, V)_z$ such that

$$\forall f \in \mathcal{C}^{p-1}(\mathbb{R}^n, V), \quad \mathrm{mj}_p(f, z) = \Theta_z(\mathbf{j}_{k_1}(f, y_1), \dots, \mathbf{j}_{k_m}(f, y_m)).$$

Remark 1.2. In Theorem 1.1, the manifold $C_p[\mathbb{R}^n]$ does not depend on *V*. Part (1) means that we can consider $(\mathbb{R}^n)^p \setminus \Delta_p$ as a dense open subset in $C_p[\mathbb{R}^n]$. Part (2) consists of properties that we expect any reasonable notion of multijet to satisfy. Part (3) means that, as in the previous discussions, if $\pi(z) \notin \Delta_p$ then $\mathcal{MJ}_p(\mathbb{R}^n)_z = \mathcal{C}^{p-1}(\mathbb{R}^n, V)/\sim$, where $f \sim g$ if and only if $f(x_i) = g(x_i)$ for all $i \in [[1, p]]$. Part (4) means that, more generally, $\operatorname{mj}_p(f, z)$ only depends on the collection of jets $(j_{k_j}(f, y_j))_{1 \leq j \leq m}$. In particular, $\operatorname{mj}_p(f, z)$ still makes sense if f is only \mathcal{C}^{k_j} on some neighborhood of y_j . This last condition also means that we can think of $\operatorname{mj}_p(f, z)$ intuitively as a family of p independent linear combinations of partial derivatives of f, up to order k_i at y_j . However this family is neither explicit nor unique in general. Let us now introduce some definitions and notation.

Definition 1.3 (multijets). Let $\Omega \subset \mathbb{R}^n$ be open. We let $C_p[\Omega] = \pi^{-1}(\Omega^p)$ and denote by $\mathcal{MJ}_p(\Omega, V) \to C_p[\Omega]$ the restriction of $\mathcal{MJ}_p(\mathbb{R}^n, V)$ to $C_p[\Omega]$. We call $\mathcal{MJ}_p(\Omega, V) \to C_p[\Omega]$ the bundle of *p*-multijets of functions from Ω to *V*. Its fiber $\mathcal{MJ}_p(\mathbb{R}^n, V)_z$ above $z \in C_p[\Omega]$ is the space of *p*-multijets at *z*. If $V = \mathbb{R}$, we drop it from the notation and write $\mathcal{MJ}_p(\Omega) \to C_p[\Omega]$. Let $f : \Omega \to V$ be of class C^{p-1} , we call the section $\mathrm{mj}_p(f, \cdot)$ of $\mathcal{MJ}_p(\Omega, V)$ the *p*-multijet of *f* and its value at $z \in C_p[\Omega]$ the *p*-multijet of *f* at *z*.

The manifold $C_p[\mathbb{R}^n]$ is what is called in the literature a "compactification" of the configuration space $(\mathbb{R}^n)^p \setminus \Delta_p$. We will use this terminology, even though it is ill-chosen in our case since $C_p[\mathbb{R}^n]$ is not compact. However $C_p[\mathbb{R}^n]$ contains a diffeomorphic copy of $(\mathbb{R}^n)^p \setminus \Delta_p$ as a dense open subset and it is equipped with a proper surjection onto $(\mathbb{R}^n)^p$ so that, in a sense, it is locally a compactification of $(\mathbb{R}^n)^p \setminus \Delta_p$.

Compactifications of configuration spaces are built to understand how a configuration (ordered or not) of *p* distinct points can degenerate as these points converge toward one another. They are usually obtained by blowing up various pieces of the diagonal. Points in the exceptional locus then correspond to singular configurations, with some extra data encoding along which paths regular configurations are allowed to degenerate in order to reach this singular configuration. The hope is that the extra data attached to singular configurations is enough to lift the singularities of the problem under consideration. The simplest example of this kind is the blow-up $Bl_{\Delta_2}((\mathbb{R}^n)^2)$ discussed above. More evolved examples are the space defined by Le Barz [1988], the compactification of Fulton and MacPherson [1994] (see also [Axelrod and Singer 1994; Sinha 2004]), Olver's multispace [2001], the polydiagonal compactification of Ulyanov [2002], the construction of Evain [2005] using Hilbert schemes, and many others.

In dimension n = 1, most of the compactifications of configuration spaces that we found in the literature coincide and can be used to define multijets; see for example [Ancona 2021], where Olver's multispace is used. In higher dimensions they are different and none of them exactly suited our needs. Thus to the best of our knowledge, the manifold $C_p[\mathbb{R}^n]$ in Theorem 1.1 is a new addition to the previous list. We define it by resolving the singularities of some real-algebraic variety, using Hironaka's theorem [1964a; 1964b]. In particular, $C_p[\mathbb{R}^n]$ is obtained by a sequence of blow-ups along Δ_p . Note that this sequence of blow-ups is neither explicit nor unique. Actually, $C_p[\mathbb{R}^n]$ itself is not uniquely defined, but this is not an issue for the applications we have in mind.

1.2. *Finiteness of moments for zeros of Gaussian fields.* Let us now describe our contributions concerning zeros of Gaussian fields. Let $n \ge 1$ and let $r \in [[1, n]]$. In the following *n* will always denote the dimension of the ambient space and *r* the codimension of the random objects we are interested in.

Let $\Omega \subset \mathbb{R}^n$ be open and let $f : \Omega \to \mathbb{R}^r$ be a centered Gaussian field of class \mathcal{C}^1 . We will always assume that f is nondegenerate, in the sense that det $\operatorname{Var}(f(x)) > 0$ for all $x \in \Omega$. Under this hypothesis the zero set $Z = f^{-1}(0)$ is almost surely (n-r)-rectifiable; see [Armentano et al. 2023b]. As such, it admits a well-defined (n-r)-dimensional volume measure dVol_Z induced by the Euclidean metric on \mathbb{R}^n .

1436

We denote by v the random Radon measure on Ω defined by

$$\forall \phi \in \mathcal{C}_c^0(\Omega), \quad \langle \nu, \phi \rangle = \int_Z \phi(x) \, \mathrm{dVol}_Z(x), \tag{1-1}$$

where $\mathcal{C}^0_c(\Omega)$ denotes the space of continuous functions on Ω with compact support.

Actually, $\langle v, \phi \rangle$ makes sense as an almost surely defined random variable as soon as $\phi \in L^{\infty}(\Omega)$ and has compact support. This kind of test-function includes $C_c^0(\Omega)$ and indicator functions of bounded Borel subsets, which are the examples we are most interested in. Random variables of the type $\langle v, \phi \rangle$ are called the linear statistics of v (or of f). Understanding the distribution of these linear statistics is one way to understand the distribution of the random measure v, or equivalently of the random set Z. For example, if $B \subset \Omega$ is a bounded Borel set and $\mathbf{1}_B$ denotes its indicator function, then $\langle v, \mathbf{1}_B \rangle$ is the (n-r)-dimensional volume of $Z \cap B$.

In this setting, a classical question is to determine conditions on the field f ensuring that its linear statistics admit finite moments. If n = r = 1, such conditions were first obtained in [Belyaev 1966]. More generally see [Azaïs and Wschebor 2009, Theorem 3.6], which holds even if f is not Gaussian. If $n \ge r = 1$, a similar result is proved in [Armentano et al. 2023a]; see also [Armentano et al. 2019, Theorem 4.4]. For a survey of previous results for hypersurfaces (i.e., r = 1), we refer to [Azaïs and Wschebor 2009, Chapter 3, Section 2.7] in dimension n = 1 and to the introduction of [Armentano et al. 2023a] in dimension $n \ge 1$. Note that [Priya 2020, Theorem 1.2] implies the finiteness of all moments of the nodal length for some Gaussian fields in \mathbb{R}^2 . This problem was also studied in [Malevich and Volodina 1993] for points in \mathbb{R}^2 .

Our second main result gives simple conditions on the field f ensuring the finiteness of the p-th moments of its linear statistics in any dimension and codimension. These conditions are of two kinds: we require the field to be regular enough, and to be nondegenerate in the following sense.

Definition 1.4 (*p*-nondegeneracy). Let $p \ge 1$ and let $f : \Omega \to \mathbb{R}^r$ be a \mathcal{C}^p centered Gaussian field. We say that the field f is *p*-nondegenerate if for all $x \in \Omega$ the centered Gaussian vector

$$(f(x), D_x f, \dots, D_x^p f) \in \bigoplus_{k=0}^p \operatorname{Sym}^k(\mathbb{R}^n) \otimes \mathbb{R}^p$$

is nondegenerate, where $\text{Sym}^k(\mathbb{R}^n)$ denotes the space of symmetric *k*-linear forms on \mathbb{R}^n and $D_x^k f \in \text{Sym}^k(\mathbb{R}^n) \otimes \mathbb{R}^r$ stands the *k*-th differential of *f* at *x*.

Remark 1.5. If $f = (f_1, ..., f_r)$, the *p*-nondegeneracy condition means more concretely that for all $x \in \Omega$ the Gaussian vector $(\partial^{\alpha} f_i(x))_{1 \le i \le r; |\alpha| \le p}$ is nondegenerate, where we used multi-index notation (see Section 2.2). More abstractly, this condition means that the *p*-jet $j_p(f, x)$ of *f* is nondegenerate for all $x \in \Omega$.

Theorem 1.6 (finiteness of moments). Let $n \ge 1$, let $r \in \llbracket 1, n \rrbracket$ and let $p \ge 1$. Let $\Omega \subset \mathbb{R}^n$ be open, let $f : \Omega \to \mathbb{R}^r$ be a centered Gaussian field and let v be defined as in (1-1). If f is \mathcal{C}^p and (p-1)nondegenerate then $\mathbb{E}[|\langle v, \phi \rangle|^p] < +\infty$ for all $\phi \in L^{\infty}(\Omega)$ with compact support. Example 1.7. Let us give some examples of fields satisfying the assumptions of Theorem 1.6.

• The Bargmann–Fock field, i.e., the smooth stationary Gaussian field on \mathbb{R}^n whose covariance function is $x \mapsto e^{-\|x\|^2/2}$, satisfies the hypotheses of Theorem 1.6.

- Let $f : \mathbb{R}^n \to \mathbb{R}$ be a stationary \mathcal{C}^p centered Gaussian field. If the support of its spectral measure has nonempty interior then f is (p-1)-nondegenerate.
- In codimension r, if $(f_i)_{1 \le i \le r}$ are r independent (p-1)-nondegenerate C^p Gaussian fields then so is $f = (f_1, \ldots, f_r)$.

• The Berry field, i.e., the smooth stationary Gaussian field f on \mathbb{R}^n whose spectral measure is the uniform measure on \mathbb{S}^{n-1} , is 1-nondegenerate but not 2-nondegenerate. Indeed it almost surely satisfies $\Delta f + f = 0$, so that $(f(x), D_x f, D_x^2 f)$ is degenerate for all $x \in \mathbb{R}^n$.

We can consider the same question in a more geometric setting. Let (M, g) be a Riemannian manifold of dimension $n \ge 1$ without boundary and let $E \to M$ be a smooth vector bundle of rank $r \in [[1, n]]$ over M. Let s be a centered Gaussian field on M with values in E, in the sense that s is a random section of $E \to M$ such that for all $m \ge 1$ and all $x_1, \ldots, x_m \in M$ the random vector $(s(x_1), \ldots, s(x_m))$ is a centered Gaussian. We assume that s is almost surely C^1 and that det Var(s(x)) > 0 for all $x \in M$.

As in the Euclidean setting, $Z = s^{-1}(0)$ is almost surely (n-r)-rectifiable. As before, we denote by v the random Radon measure on M defined by integrating over Z with respect to the (n-r)-dimensional volume measure $dVol_Z$ induced by g. For all $\phi \in L^{\infty}(M)$ with compact support, we define the linear statistic $\langle v, \phi \rangle$ as in (1-1). In this context Definition 1.4 adapts as follows.

Definition 1.8 (*p*-nondegeneracy for Gaussian sections). Let $p \ge 1$ and let *s* be a C^p centered Gaussian field on *M* with values in *E*. We say that *s* is *p*-nondegenerate if, for all $x \in M$, the centered Gaussian vector $j_p(s, x) \in \mathcal{J}_p(M, E)_x$ is nondegenerate.

Theorem 1.9 (finiteness of moments for zeros of Gaussian sections). Let $p \ge 1$, let s be a centered Gaussian field on M with values in E and let v be defined as in (1-1). If s is C^p and (p-1)-nondegenerate then $\mathbb{E}[|\langle v, \phi \rangle|^p] < +\infty$ for all $\phi \in L^{\infty}(M)$ with compact support.

We are aware of the very recent paper [Gass and Stecconi 2024], in which the authors prove a result similar to Theorem 1.6, as well as its analogue for zeros of Gaussian fields on a Riemannian manifold. Their work and ours are independent, and the proofs are different. Their idea is to compare the Kac–Rice densities (see Section 6.3) of the field f with those of a well-chosen Gaussian polynomial P. Then they deduce the result for f from the result for P, which is a consequence of Bézout's theorem. Our proof follows a different path, as it relies on the multijet bundle that we defined in Theorem 1.1. Our idea is to observe that the zero set of $F : (x_1, \ldots, x_p) \mapsto (f(x_1), \ldots, f(x_p))$ in the configuration space $\Omega^p \setminus \Delta_p$ is exactly the vanishing locus of the multijet $\text{mj}_p(f, \cdot)$ restricted to $\Omega^p \setminus \Delta_p \subset C_p[\Omega]$. Instead of working with F, which degenerates along Δ_p , we work with the field $\text{mj}_p(f, \cdot)$ that we built to be nondegenerate everywhere. Then, we deduce Theorem 1.6 from the Kac–Rice formula for the expectation (see Proposition 6.17) applied to the p-multijet of f and a compactness argument. **1.3.** *Higher-order multijets and holomorphic multijets.* Let us now discuss two important variations on our main results, Theorems 1.1 and 1.9. In Section 1.1, we said that two functions f and g on \mathbb{R}^n have the same p-multijet at a point $\underline{x} = (x_1, \ldots, x_p) \in (\mathbb{R}^n)^p \setminus \Delta_p$ if and only if f and g have the same value, i.e., the same 0-jet, at x_i for all $i \in [[1, p]]$. In a sense, the p-multijet of f at \underline{x} is obtained by patching together the 0-jets of f at each of the x_i in a relevant way. A natural generalization is to define a higher-order multijet of f at \underline{x} by patching together the k-jets of f at each of the x_i . We define such a higher-order multijet in Section 7. More generally, we define a multijet bundle adapted to a differential operator \mathcal{D} . The case of higher-order multijets corresponds to $\mathcal{D} = \mathbf{j}_k$. The analogue of Theorem 1.1 in this framework is Theorem 7.4 below. We use it to prove an analogue of Theorem 1.9 adapted to \mathcal{D} ; see Theorem 7.8 for a general statement. In the special case where $\mathcal{D} = D$ is the standard differential, the statement is the following.

Theorem 1.10 (finiteness of moments for critical points). Let M be a smooth manifold without boundary. Let $f: M \to \mathbb{R}$ be a centered Gaussian field and let v_D denote the counting measure of its critical locus. Let $p \ge 1$, we assume that f is C^{2p} and (2p-1)-nondegenerate. Then, for all $\phi \in L^{\infty}_{c}(M)$, we have $\mathbb{E}[|\langle v_D, \phi \rangle|^p] < +\infty$.

Another variation on Theorem 1.1 is to define holomorphic multijets for holomorphic maps. This is done in Section 8, and more precisely in Theorem 8.2. This is used to prove a holomorphic version of Theorem 1.9. The general statement is given in Theorem 8.13. For a holomorphic Gaussian field on an open subset of \mathbb{C}^n , it takes the following form.

Theorem 1.11 (finiteness of moments for zeros of holomorphic Gaussian fields). Let $\Omega \subset \mathbb{C}^n$ be open and let $f : \Omega \to \mathbb{C}^r$ be a centered holomorphic Gaussian field, where $r \in [\![1, n]\!]$. Let v be as in Definition 6.11. Let $p \ge 1$, we assume that, for all $x \in \Omega$, the complex Gaussian vector

$$(f(x), D_x f, \dots, D_x^{p-1} f) \in \bigoplus_{k=0}^{p-1} \operatorname{Sym}^k(\mathbb{C}^n) \otimes \mathbb{C}^r$$

is nondegenerate. Then, for all $\phi \in L^{\infty}_{c}(\Omega)$ *, we have* $\mathbb{E}[|\langle v, \phi \rangle|^{p}] < +\infty$ *.*

Note that Theorems 1.10 and 1.11 are not consequences of Theorem 1.9. Indeed, if $f: M \to \mathbb{R}$ is a smooth Gaussian field then Df cannot be 1-nondegenerate because $D^2 f$ is symmetric. Similarly, if M is a complex manifold and s is holomorphic, then s is never 1-nondegenerate because it satisfies the Cauchy–Riemann equations.

Gass and Stecconi [2024] proved, independently and by a different method, results analogous to Theorems 1.10 and 1.11. Actually, they prove Theorem 1.10 under the weaker and optimal hypotheses that f is C^{p+1} and p-nondegenerate. The finiteness of the third moment for the number of critical points of a stationary Gaussian field on \mathbb{R}^d was proved in [Beliaev et al. 2024, Theorem 1.6]. For holomorphic Gaussian fields in dimension n = 1, see [Nazarov and Sodin 2012].

1.4. *Organization of the paper.* In Section 2 we gather useful notation that appears in several parts of the paper. In Section 3 we discuss Kergin interpolation, which is a multivariate polynomial interpolation appearing in the definition of multijets. Section 4 is dedicated to evaluations maps on spaces of polynomials,

and more precisely the properties of their kernels. We define our multijet bundles and prove Theorem 1.1 in Section 5. Section 6 is concerned with the application of multijets to the finiteness of moments for the zeros of Gaussian fields and the proofs of Theorems 1.6 and 1.9. Multijets adapted to a differential operator are discussed in Section 7, where we also prove the analogue of Theorem 1.9 for critical points. Finally, holomorphic multijets are defined in Section 8, where we prove the analogue of Theorem 1.9 for holomorphic Gaussian fields.

2. Notation: partitions and function spaces

The goal of this section is to quickly introduce definitions and notation that will appear in different parts of the paper. We gather them here for the reader's convenience.

2.1. Sets, partitions and diagonals. In this paper, we denote by \mathbb{N} the set of nonnegative integers. Let *a* and $b \in \mathbb{N}$, we use the following notation for integer intervals $[[a, b]] = [a, b] \cap \mathbb{N}$.

Let *A* be a nonempty finite set. For simplicity, in all the notation introduced in this section, if $A = \llbracket 1, p \rrbracket$ we allow ourselves to replace *A* by *p* in the indices and exponents. We denote by |A| the cardinality of *A*. Let *M* be any set. We denote by M^A the Cartesian product of |A| copies of *M* indexed by the elements of *A*. A generic element of M^A is usually denoted by $\underline{x} = (x_a)_{a \in A}$. If $\emptyset \neq B \subset A$, we denote by $\underline{x}_B = (x_a)_{a \in B}$.

Definition 2.1 (large diagonal). We denote by Δ_A the large diagonal in M^A , that is,

$$\Delta_A = \{ (x_a)_{a \in A} \in M^A \mid \exists a, b \in A \text{ such that } a \neq b \text{ and } x_a = x_b \}.$$

Definition 2.2 (partitions). Let *A* be a nonempty and finite set, a *partition* of *A* is a family $\mathcal{I} = \{I_1, \ldots, I_m\}$ of nonempty disjoint subsets of *A* such that $\bigsqcup_{i=1}^m I_i = A$. The subsets I_1, \ldots, I_m are called the *cells* of \mathcal{I} . Given $a \in A$, we denote by $[a]_{\mathcal{I}}$ the only cell of \mathcal{I} that contains *a*. Finally, we denote by \mathcal{P}_A the set of partitions of *A*.

Definition 2.3 (clustering partition). Let $\underline{x} = (x_a)_{a \in A} \in M^A$. We denote by $\mathcal{I}(\underline{x}) \in \mathcal{P}_A$ the only partition such that for all *a* and $b \in A$ we have $x_a = x_b$ if and only if $[a]_{\mathcal{I}(x)} = [b]_{\mathcal{I}(x)}$.

Example 2.4. If $\underline{x} = (x, \ldots, x)$ then $\mathcal{I}(\underline{x}) = \{A\}$. If $\underline{x} \in M^A \setminus \Delta_A$ then $\mathcal{I}(\underline{x}) = \{\{a\} \mid a \in A\} = \mathcal{I}_0$.

Definition 2.5 (strata of the diagonal). For all $\mathcal{I} \in \mathcal{P}_A$, we set $\Delta_{A,\mathcal{I}} = \{\underline{x} \in M^A \mid \mathcal{I}(\underline{x}) = \mathcal{I}\}$, so that $\Delta_{A,\mathcal{I}_0} = M^A \setminus \Delta_A$ and $\Delta_A = \bigsqcup_{\mathcal{I} \neq \mathcal{I}_0} \Delta_{A,\mathcal{I}}$.

Definition 2.6 (diagonal inclusions). Let $\mathcal{I} \in \mathcal{P}_A$. We denote by $\iota_{\mathcal{I}} : M^{\mathcal{I}} \setminus \Delta_{\mathcal{I}} \to \Delta_{A,\mathcal{I}}$ the bijection defined by $\iota_{\mathcal{I}}((y_I)_{I \in \mathcal{I}}) = (y_{[a]_{\mathcal{I}}})_{a \in A}$.

2.2. Spaces of functions, sections and jets. We use the following multi-index notation. Let $\alpha = (\alpha_1, \ldots, \alpha_n) \in \mathbb{N}^n$. We denote its length by $|\alpha| = \alpha_1 + \cdots + \alpha_n$. Let ∂_i denote the *i*-th partial derivative in some product space we denote by $\partial^{\alpha} = \partial_1^{\alpha_1} \cdots \partial_n^{\alpha_n}$. Finally, if $X = (X_1, \ldots, X_n)$, we let $X^{\alpha} = X_1^{\alpha_1} \cdots X_n^{\alpha_n}$.

Definition 2.7 (polynomials). We denote by $\mathbb{R}_d[X]$ the space of real polynomials in *n* variables of degree at most *d*, where $d \in \mathbb{N}$ and $X = (X_1, \ldots, X_n)$ is multivariate.

Definition 2.8 (symmetric forms and differentials). Let $k \in \mathbb{N}$. We denote by $\text{Sym}^k(\mathbb{R}^n)$ the space of symmetric *k*-linear forms on \mathbb{R}^n . Let *V* be a finite-dimensional real vector space. Then $\text{Sym}^k(\mathbb{R}^n) \otimes V$ is the space of symmetric *k*-linear maps from \mathbb{R}^n to *V*. Given a \mathcal{C}^k map $f : \mathbb{R}^n \to V$, we denote by $D_x^k f \in \text{Sym}^k(\mathbb{R}^n) \otimes V$ its *k*-th differential at $x \in \mathbb{R}^n$.

Let *M* and *N* be two manifolds without boundary. For all $k \in \mathbb{N} \cup \{\infty\}$, we denote by $\mathcal{C}^k(M, N)$ the space of \mathcal{C}^k maps from *M* to *N*. If $N = \mathbb{R}$, we drop it from the notation and we simply write $\mathcal{C}^k(M)$. We denote by $L^1_{loc}(M)$ the space of locally integrable functions on *M*. We denote by $\mathcal{C}^0_c(M)$ (resp. $L^\infty_c(M)$) the space of continuous (resp. L^∞) functions on *M* with compact support. Finally, for any Borel subset $B \subset M$, we denote by $\mathbf{1}_B : M \to \mathbb{R}$ its indicator function.

Let $E \to M$ be a vector bundle of finite rank over M, we denote by E_x the fiber above $x \in M$. For all $k \in \mathbb{N} \cup \{\infty\}$, we denote by $\Gamma^k(M, E)$ the space of \mathcal{C}^k sections of $E \to M$.

Definition 2.9 (jets). Let $k \in \mathbb{N}$, we denote by $\mathcal{J}_k(M, E) \to M$ the vector bundle of k-jets of sections of $E \to M$. If $E = V \times M$ is trivial with fiber V, we denote its k-jet bundle by $\mathcal{J}_k(M, V) \to M$. If $V = \mathbb{R}$, we simply write $\mathcal{J}_k(M) \to M$. Given $s \in \Gamma^k(M, E)$, we denote by $j_k(s, x) \in \mathcal{J}_k(M, E)_x$ its k-jet at $x \in M$.

3. Divided differences and Kergin interpolation

An important step in our construction of a multijet for C^k functions is to reduce the problem to that of defining a multijet for polynomials. This is done by polynomial interpolation. In several variables, polynomial interpolation is rather ill-behaved, at least compared with the one-variable case. However, a multivariate polynomial interpolation suiting our needs was defined by Kergin [1980]. A constructive version of his proof was then given in [Micchelli and Milman 1980], using a multivariate version of the socalled divided differences. In this section, we give the definitions of these objects and recall their relevant properties. We refer to the survey [Lorentz 2000] for more background on polynomial interpolation in \mathbb{R}^n .

3.1. *Divided differences.* In this section, we recall the definition of multivariate divided differences; see [Micchelli and Milman 1980]. Let $k \in \mathbb{N}$. We denote by σ_k the standard simplex of dimension k, that is,

$$\sigma_k = \left\{ \underline{t} = (t_0, \dots, t_k) \in [0, 1]^{k+1} \mid \sum_{i=0}^k t_i = 1 \right\} \subset \mathbb{R}^{k+1}.$$
 (3-1)

The simplex σ_k is a subset of $\{\underline{t} \in \mathbb{R}^{k+1} | \sum t_i = 1\}$, and we denote by v_k the (*k*-dimensional) Lebesgue measure on this hyperplane, normalized so that $v_k(\sigma_k) = 1/k!$. One can check that its restriction to σ_k satisfies

$$\int_{\sigma_k} \phi(\underline{t}) \, \mathrm{d}\nu_k(\underline{t}) = \int_{\substack{t_1, \dots, t_k \ge 0\\\sum_{i=1}^k t_i \le 1}} \phi\left(1 - \sum_{i=1}^k t_i, t_1, \dots, t_k\right) \mathrm{d}t_1 \cdots \mathrm{d}t_k,\tag{3-2}$$

where $dt_1 \cdots dt_k$ is the Lebesgue measure on \mathbb{R}^k . For any $\underline{x} = (x_0, \ldots, x_k) \in (\mathbb{R}^n)^{k+1}$, we denote by $\sigma(\underline{x})$ the convex hull of the x_i and we define $\upsilon_{\underline{x}} : \underline{t} \mapsto \sum_{i=0}^k t_i x_i$ from σ_k onto $\sigma(\underline{x})$. Recalling Definition 2.8, we have the following.

Definition 3.1 (divided differences). Let $\underline{x} = (x_i)_{0 \le i \le k} \in (\mathbb{R}^n)^{k+1}$ and let f be a \mathcal{C}^k function defined on some open neighborhood of $\sigma(\underline{x})$ in \mathbb{R}^n . We define the *divided difference* of f at \underline{x} by

$$f[x_0,\ldots,x_k] = \int_{\sigma_k} D_{\nu_{\underline{x}}(\underline{t})}^k f \, \mathrm{d}\nu_k(\underline{t}) \in \mathrm{Sym}^k(\mathbb{R}^n),$$

that is, as the average of $D^k f$ over $\sigma(\underline{x})$ with respect to the pushed-forward measure $(v_x)_*(v_k)$.

Remark 3.2. If $\underline{x} = (x, \dots, x)$ for some $x \in \mathbb{R}^n$ then $f[x, \dots, x] = (1/k!)D_x^k f$.

• Definition 3.1 is invariant under permutation of (x_0, \ldots, x_k) .

• When n = 1, Definition 3.1 coincides with the classical definition of divided differences, under the canonical isomorphism $\text{Sym}^k(\mathbb{R}) \simeq \mathbb{R}$. This is known as the Hermite–Genocchi formula [Micchelli and Milman 1980].

Lemma 3.3 (regularity of divided differences). For all $\underline{x} \in (\mathbb{R}^n)^{k+1}$, the map $f \mapsto f[x_0, \ldots, x_k]$ is linear. Moreover, if f is of class \mathcal{C}^{k+l} then $\underline{x} \mapsto f[x_0, \ldots, x_k]$ is of class \mathcal{C}^l .

Proof. The linearity with respect to f is clear. The regularity with respect to \underline{x} is obtained by derivation under the integral, using Definition 3.1 and (3-2).

3.2. *Kergin interpolation.* This section is dedicated to Kergin interpolation. In the following, we recall the construction of Kergin interpolation in [Micchelli and Milman 1980], which relies on the divided differences introduced in Definition 3.1. We will use the notation introduced in Definition 2.7.

Proposition 3.4 (Kergin interpolation). Let $\underline{x} \in (\mathbb{R}^n)^p$ and let f be a function of class C^{p-1} defined on some neighborhood of $\sigma(\underline{x})$ in \mathbb{R}^n . There exists a unique polynomial $K(f, \underline{x}) \in \mathbb{R}_{p-1}[X]$ such that, for all nonempty $I \subset [\![1, p]\!]$, we have $f[\underline{x}_I] = (K(f, \underline{x}))[\underline{x}_I]$. Moreover,

$$K(f, \underline{x}) = \sum_{k=1}^{p} f[x_1, \dots, x_k](X - x_1, \dots, X - x_{k-1}).$$
(3-3)

Proof. This is the content of [Bojanov et al. 1993, Theorem 12.5] for m = 0. See also [Micchelli and Milman 1980].

Remark 3.5. In particular, Proposition 3.4 implies the following:

- The restriction of $K(\cdot, \underline{x})$ to $\mathbb{R}_{p-1}[X]$ is the identity.
- If x appears with multiplicity at least k + 1 in <u>x</u>, then

$$D_x^k f = k! f[\underbrace{x, \dots, x}_{k+1 \text{ times}}] = k! (K(f, \underline{x}))[\underbrace{x, \dots, x}_{k+1 \text{ times}}] = D_x^k (K(f, \underline{x})).$$

• The map $P \mapsto (P[x_1, \ldots, x_j])_{1 \leq j \leq p}$ is an isomorphism from $\mathbb{R}_{p-1}[X]$ to $\bigoplus_{j=0}^{p-1} \operatorname{Sym}^j(\mathbb{R}^n)$ whose inverse map is given by $(S_j)_{0 \leq j \leq p-1} \mapsto \sum_{j=0}^{p-1} S_j(X - x_1, \ldots, X - x_j)$.

Definition 3.6 (Kergin polynomial). The polynomial $K(f, \underline{x})$ from Proposition 3.4 is called the *Kergin interpolating polynomial* of f at \underline{x} .

Example 3.7. If n = 1, then $K(f, \underline{x})$ is the Hermite interpolating polynomial of f at $\underline{x} \in \mathbb{R}^p$. If $\underline{x} = (x, \dots, x)$, then $K(f, \underline{x})$ is the Taylor polynomial of order p - 1 of f at $x \in \mathbb{R}^n$.

Lemma 3.8 (regularity of the Kergin polynomial). For all $\underline{x} \in (\mathbb{R}^n)^p$, the map $K(\cdot, \underline{x})$ is linear. Moreover, if f is \mathcal{C}^{l+p-1} then $K(f, \cdot)$ is of class \mathcal{C}^l .

Proof. This is a consequence of Lemma 3.3 and (3-3).

We need to prove a form of compatibility in Kergin interpolation, when the set of interpolation points is refined. We will use this fact to prove that the multijet bundle we define below satisfies (4) in Theorem 1.1. The following lemma is stated using the clustering partition $\mathcal{I}(x)$ from Definition 2.3.

Lemma 3.9 (compatibility in Kergin interpolation). For all $\underline{x} \in (\mathbb{R}^n)^p$ the linear map from $\mathbb{R}_{p-1}[X]$ to $\prod_{I \in \mathcal{I}(x)} \mathbb{R}_{|I|-1}[X]$ defined by $(K(\cdot, \underline{x}_I))_{I \in \mathcal{I}(\underline{x})} : P \mapsto (K(P, \underline{x}_I))_{I \in \mathcal{I}(\underline{x})}$ is surjective.

Proof. Let $\underline{x} \in (\mathbb{R}^n)^p$ and let us write $\mathcal{I} = \mathcal{I}(\underline{x})$ for simplicity. As explained at the end of Section 2.1, there exists a unique $\underline{y} = (y_I)_{I \in \mathcal{I}} \in (\mathbb{R}^n)^{\mathcal{I}} \setminus \Delta_{\mathcal{I}}$ such that $\underline{x} = \iota_{\mathcal{I}}(\underline{y})$. Let $(\chi_I)_{I \in \mathcal{I}}$ be smooth functions on \mathbb{R}^n with pairwise disjoint compact supports and such that χ_I is equal to 1 in a neighborhood of y_I .

Let $(P_I)_{I \in \mathcal{I}} \in \prod_{I \in \mathcal{I}} \mathbb{R}_{|I|-1}[X]$. We consider the function $f = \sum_{I \in \mathcal{I}} \chi_I P_I \in \mathcal{C}^{\infty}(\mathbb{R}^n)$. Let $P = K(f, \underline{x})$ and let us prove that $K(P, \underline{x}_I) = P_I$ for all $I \in \mathcal{I}$. For all $k \leq |I| - 1$ we have $D_{y_I}^k P = D_{y_I}^k f = D_{y_I}^k P_I$. Indeed y_I appears with multiplicity |I| in \underline{x} (see Remark 3.5) and f is equal to P_I in a neighborhood of y_I . Recalling Example 3.7, we know that $K(P, \underline{x}_I)$ is the Taylor polynomial of order |I| - 1 at y_I of P, and hence of P_I . Since $P_I \in \mathbb{R}_{|I|-1}[X]$, we get $K(P, \underline{x}_I) = P_I$.

4. Evaluation maps and their kernels

The goal of this section is to study evaluation maps on spaces of polynomials and their kernels. Defining multijets is closely related to these objects. Indeed, let $n \ge 1$ and $p \ge 1$ and recall that Δ_p stands for the large diagonal in $(\mathbb{R}^n)^p$; see Definition 2.1. As explained in the Introduction, when $\underline{x} \notin \Delta_p$ we want the multijet of a \mathcal{C}^{p-1} function f at \underline{x} to be the class of f in $\mathcal{C}^{p-1}(\mathbb{R}^n)/\sim$, where $f \sim g$ if and only if $(f(x_i))_{1 \le i \le p} = (g(x_i))_{1 \le i \le p}$. The Kergin interpolation of Section 3.2 shows that any such class can be represented by a polynomial. Hence, the space of p-multijets at \underline{x} is canonically isomorphic to $\mathbb{R}_{p-1}[X]/\operatorname{ker}\operatorname{ev}_x$, where $\operatorname{ev}_x : P \mapsto (P(x_1), \ldots, P(x_p))$.

Definition 4.1 (Grassmannian). Let *V* be a vector space of finite dimension *N* and $k \in [[0, N]]$. We denote by $Gr_k(V)$ the *Grassmannian* of vector subspaces of *V* of *codimension k*.

Remark 4.2. Beware that this notation is slightly unusual, since in most textbooks $Gr_k(V)$ stands for the Grassmannian of subspaces of dimension *k*.

Let us denote by $\mathcal{L}_{reg}(V, \mathbb{R}^k) \subset V^* \otimes \mathbb{R}^k$ the open dense subset of linear surjective maps from V to \mathbb{R}^k . The group $GL_k(\mathbb{R})$ acts on $\mathcal{L}_{reg}(V, \mathbb{R}^k)$ by multiplication on the left. On the other hand, $L \mapsto ker(L)$ defines a surjective map from $\mathcal{L}_{reg}(V, \mathbb{R}^k)$ to $Gr_k(V)$, and $ker(L_1) = ker(L_2)$ if and only if there exists $M \in GL_k(\mathbb{R})$ such that $L_2 = ML_1$. Thus, one can identify $Gr_k(V)$ with the orbit space $\mathcal{L}_{reg}(V, \mathbb{R}^k)/GL_k(\mathbb{R})$ of the previous action. This is one of the many ways to describe $Gr_k(V)$ as a smooth real-algebraic manifold. **Definition 4.3** (evaluation map). Let $\underline{x} \in (\mathbb{R}^n)^p$. We set $ev_{\underline{x}} : f \mapsto (f(x_1), \dots, f(x_p))$ from any space of functions defined at the x_i to \mathbb{R}^p . The source space will always be clear from the context.

Lemma 4.4 (nondegeneracy of $ev_{\underline{x}}$). Let $\underline{x} \notin \Delta_p$. Then $ev_{\underline{x}} : \mathbb{R}_{p-1}[X] \to \mathbb{R}^p$ is surjective.

Proof. Since $\underline{x} \notin \Delta_p$, we have $\mathcal{I}(\underline{x}) = \{\{1\}, \dots, \{p\}\}$ and $ev_{\underline{x}} = (K(\cdot, x_i))_{1 \le i \le p}$ under the canonical identification $\mathbb{R}_0[X] \simeq \mathbb{R}$. Hence this is just a special case of Lemma 3.9. Alternatively, in the right basis, one can extract a Vandermonde matrix from that of ev_x .

Lemma 4.4 shows that the following map is well-defined from $(\mathbb{R}^n)^p \setminus \Delta_p$ to $\operatorname{Gr}_p(\mathbb{R}_{p-1}[X])$:

$$\mathcal{G}: \underline{x} \longmapsto \ker \operatorname{ev}_{\underline{x}} . \tag{4-1}$$

Lemma 4.5 (algebraicity of \mathcal{G}). The map $\mathcal{G} : (\mathbb{R}^n)^p \setminus \Delta_p \to \operatorname{Gr}_p(\mathbb{R}_{p-1}[X])$ is algebraic.

Proof. Recalling the previous discussion, we have $\mathcal{L}_{reg}(\mathbb{R}_{p-1}[X], \mathbb{R}^p) / \operatorname{GL}_p(\mathbb{R}) \simeq \operatorname{Gr}_p(\mathbb{R}_{p-1}[X])$, where the isomorphism is obtained as the quotient map of ker : $L \mapsto \operatorname{ker}(L)$. In particular,

 $\ker : \mathcal{L}_{\mathrm{reg}}(\mathbb{R}_{p-1}[X], \mathbb{R}^p) \longrightarrow \mathrm{Gr}_p(\mathbb{R}_{p-1}[X]) \simeq \mathcal{L}_{\mathrm{reg}}(\mathbb{R}_{p-1}[X], \mathbb{R}^p) / \mathrm{GL}_p(\mathbb{R})$

is just the canonical projection, which is algebraic.

Writing $\operatorname{ev} : \underline{x} \mapsto \operatorname{ev}_{\underline{x}}$, we have $\mathcal{G} = \ker \circ \operatorname{ev}$. Thus it is enough to prove that ev is algebraic from $(\mathbb{R}^n)^p \setminus \Delta_p$ to $\mathcal{L}_{\operatorname{reg}}(\mathbb{R}_{p-1}[X], \mathbb{R}^p)$. In the basis of $\mathbb{R}_{p-1}[X]$ formed by the monomials $(X^{\alpha})_{|\alpha| < p}$, the matrix of $\operatorname{ev}_{\underline{x}}$ is $(x_i^{\alpha})_{1 \leq i \leq p; |\alpha| < p}$, which depends algebraically on \underline{x} .

Let $\underline{x} \in (\mathbb{R}^n)^p \setminus \Delta_p$, we defined $\mathcal{G}(\underline{x}) \in \operatorname{Gr}_p(\mathbb{R}_{p-1}[X])$ by (4-1). For any nonempty $I \subset \llbracket 1, p \rrbracket$, we define similarly

$$\mathcal{G}_{I}(\underline{x}) = \ker \operatorname{ev}_{\underline{x}_{I}} \in \operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X]) \quad \text{and} \quad \widetilde{\mathcal{G}}_{I}(\underline{x}) = \ker \operatorname{ev}_{\underline{x}_{I}} \in \operatorname{Gr}_{|I|}(\mathbb{R}_{p-1}[X]).$$
(4-2)

Because of the interpolation properties of the Kergin polynomials (see Remark 3.5), we have that $ev_{\underline{x}_I} = (ev_{\underline{x}_I})_{|\mathbb{R}|_{I|-1}[X]} \circ K(\cdot, \underline{x}_I)$ on $\mathbb{R}_{p-1}[X]$. Hence $\widetilde{\mathcal{G}}_I(\underline{x}) = K(\cdot, \underline{x}_I)^{-1}(\mathcal{G}_I(\underline{x}))$. Since $K(\cdot, \underline{x}_I)$ is surjective from $\mathbb{R}_{p-1}[X]$ to $\mathbb{R}_{|I|-1}[X]$, this shows that $\widetilde{\mathcal{G}}_I(\underline{x})$ has indeed codimension |I|, like $\mathcal{G}_I(\underline{x})$.

This collection of subspaces satisfies some incidence relations that will be useful in the following. For all nonempty $I \subset \llbracket 1, p \rrbracket$, we have $\mathcal{G}(\underline{x}) \subset \widetilde{\mathcal{G}}_I(\underline{x})$. Actually, we can be more precise: for any $\mathcal{I} \in \mathcal{P}_p$, we have $\mathcal{G}(\underline{x}) = \bigcap_{I \in \mathcal{I}} \widetilde{\mathcal{G}}_I(\underline{x})$, and this intersection is transverse by a codimension argument.

Remark 4.6. The map $\mathcal{G}: (\mathbb{R}^n)^p \setminus \Delta_p \to \operatorname{Gr}_p(\mathbb{R}_{p-1}[X])$ does not admit an extension as a regular map from $(\mathbb{R}^n)^p$ to $\operatorname{Gr}_p(\mathbb{R}_{p-1}[X])$, except if n = 1 or p = 1, that is, if $\operatorname{Gr}_p(\mathbb{R}_{p-1}[X])$ is a point.

For example, when n = 2 = p, the Grassmannian $\operatorname{Gr}_p(\mathbb{R}_{p-1}[X])$ is the set of lines in $\mathbb{R}_1[X_1, X_2]$. Taking $x = R(\cos \theta, \sin \theta) \in \mathbb{R}^2 \setminus \{0\}$, the reader can check that $\mathcal{G}(0, x) = \operatorname{Span}(X_1 \sin \theta - X_2 \cos \theta)$, which does not converge as $R \to 0$. However, in this case, $\mathcal{G}(0, \cdot)$ extends to the blow-up $\operatorname{Bl}_0(\mathbb{R}^2)$ of \mathbb{R}^2 at 0 and similarly \mathcal{G} extends smoothly to $\operatorname{Bl}_{\Delta_2}((\mathbb{R}^2)^2)$. This suggests that, even though \mathcal{G} does not extend smoothly to $(\mathbb{R}^n)^p$, it might extend to a larger space.

1444

5. Definition of the multijet bundles

In this section we define the vector bundle $\mathcal{MJ}_p(\mathbb{R}^n, V) \to C_p[\mathbb{R}^n]$ of *p*-multijets for functions from \mathbb{R}^n to some finite-dimensional vector space *V* and prove Theorem 1.1. The singularity of \mathcal{G} along Δ_p makes it impossible to define such a bundle over $(\mathbb{R}^n)^p$, which is why we define it over a compactification $C_p[\mathbb{R}^n]$ of the configuration space $(\mathbb{R}^n)^p \setminus \Delta_p$.

The manifold $C_p[\mathbb{R}^n]$ does not depend on V. It is defined in Section 5.1. In the next two sections, we work in the case $V = \mathbb{R}$. All important ideas appear in this case but the notation is slightly simpler. In Section 5.2, we define the bundle $\mathcal{MJ}_p(\mathbb{R}^n)$. In Section 5.3, we prove that *p*-multijets are local, in the sense of (4) in Theorem 1.1. Finally, we define the bundle $\mathcal{MJ}_p(\mathbb{R}^n, V)$ of multijets for vector-valued maps and prove Theorem 1.1 in Section 5.4.

5.1. Definition of the basis $C_p[\mathbb{R}^n]$. In this section, we define the basis $C_p[\mathbb{R}^n]$ over which our *p*-multijet bundles are defined. This is a smooth manifold, obtained a compactification of the configuration space $(\mathbb{R}^n)^p \setminus \Delta_p$ such that $(\mathcal{G}_I)_{I \subset [\![1,p]\!]}$ extends smoothly to $C_p[\mathbb{R}^n]$. Let us first introduce some notation. We denote by Π_0 the projection from the product space

$$(\mathbb{R}^n)^p \times \prod_{\varnothing \neq I \subset \llbracket 1, p \rrbracket} \operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X])$$

onto the factor $(\mathbb{R}^n)^p$. Similarly, we denote by Π_I the projection onto $\operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X])$. Then, let

$$\Sigma = \{ (\underline{x}, (\mathcal{G}_{I}(\underline{x}))_{I \subset \llbracket 1, p \rrbracket}) \mid \underline{x} \in (\mathbb{R}^{n})^{p} \setminus \Delta_{p} \} \subset (\mathbb{R}^{n})^{p} \times \prod_{\varnothing \neq I \subset \llbracket 1, p \rrbracket} \operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X])$$
(5-1)

denote the graph of the map $(\mathcal{G}_I)_{I \subset [\![1,p]\!]}$. We denote by $\overline{\Sigma}$ the closure of Σ in the product space on the right-hand side of (5-1).

Lemma 5.1 (surjectivity of $(\Pi_0)_{|\overline{\Sigma}}$). Let $\underline{x} \in (\mathbb{R}^n)^p$. Then there exists $z \in \overline{\Sigma}$ such that $\Pi_0(z) = \underline{x}$.

Proof. Let $(\underline{x}_n)_{n\in\mathbb{N}}$ be a sequence of points in $(\mathbb{R}^n)^p \setminus \Delta_p$ converging to \underline{x} . Since Grassmannians are compact manifolds, up to extracting subsequences finitely many times, we can assume that for all nonempty $I \subset \llbracket 1, p \rrbracket$ there exists $G_I \in \operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X])$ such that $\mathcal{G}_I(\underline{x}_n) \xrightarrow[n \to +\infty]{} G_I$. Then

$$(\underline{x}_n, (\mathcal{G}_I(\underline{x}_n))_{I \subset \llbracket 1, p \rrbracket}) \xrightarrow[n \to +\infty]{} (\underline{x}, (G_I)_{I \subset \llbracket 1, p \rrbracket}) = z \in \overline{\Sigma}.$$

Lemma 5.2 (location of the new points). We have $\overline{\Sigma} \setminus \Sigma \subset \Pi_0^{-1}(\Delta_p)$.

Proof. Since Σ is the graph of a continuous function on $(\mathbb{R}^n)^p \setminus \Delta_p$, it is closed in the open subset $\Pi_0^{-1}((\mathbb{R}^n)^p \setminus \Delta_p)$. Hence $\overline{\Sigma} \cap \Pi_0^{-1}((\mathbb{R}^n)^p \setminus \Delta_p) = \Sigma$ and $\overline{\Sigma} \setminus \Sigma \subset \Pi_0^{-1}(\Delta_p)$.

Lemma 5.3 (algebraicity of Σ and $\overline{\Sigma}$). The graph Σ is a smooth real-algebraic manifold and $(\Pi_0)_{|\Sigma}$: $\Sigma \to (\mathbb{R}^n)^p \setminus \Delta_p$ is an isomorphism. Moreover, $\overline{\Sigma}$ is a real-algebraic variety whose singular locus is contained in $\overline{\Sigma} \setminus \Sigma$.

Proof. By Lemma 4.5, the set Σ is the graph of an algebraic map, hence a smooth real-algebraic manifold. Additionally, Π_0 is algebraic and its restriction to Σ is the inverse of $\underline{x} \mapsto (\underline{x}, (\mathcal{G}_I(\underline{x}))_{I \subset [\![1,p]\!]})$. Thus $(\Pi_0)|_{\Sigma}$ is an algebraic isomorphism from Σ onto $(\mathbb{R}^n)^p \setminus \Delta_p$. Since Σ is real-algebraic, so is its closure $\overline{\Sigma}$. By Lemma 5.2, we know that $\overline{\Sigma} \cap \Pi_0^{-1}((\mathbb{R}^n)^p \setminus \Delta_p) = \Sigma$ is smooth. Hence, the singular locus of $\overline{\Sigma}$ is contained in $\overline{\Sigma} \cap \Pi_0^{-1}(\Delta_p) = \overline{\Sigma} \setminus \Sigma$.

Example 5.4. In simple cases, we understand very well what $\overline{\Sigma}$ is.

- If p = 1 and $n \ge 1$, then $\Delta_p = \emptyset$ and $\operatorname{Gr}_p(\mathbb{R}_{p-1}[X]) = \{\{0\}\}$, so that $\overline{\Sigma} = \Sigma = \mathbb{R}^n$.
- If n = 1 and $p \ge 1$, then $\operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X]) = \{\{0\}\}\$ for all $I \subset \llbracket 1, p \rrbracket$ and $\overline{\Sigma} = \mathbb{R}^p$.

• If p = 2 and $n \ge 2$, then for $x \ne y$ in \mathbb{R}^n we know that $\mathcal{G}(x, y) \subset \mathbb{R}_1[X]$ is the subspace of affine forms on \mathbb{R}^n vanishing at *x* and *y*, i.e., on the affine line through *x* and *y*. Thus $\mathcal{G}(x, y)$ encodes this line. As $y \rightarrow x$, the accumulation points of $\mathcal{G}(x, y)$ correspond to all the affine lines passing through *x*, and they encode "the direction from which *y* converges to *x*". In this case, one can check that $\overline{\Sigma} = Bl_{\Delta_2}((\mathbb{R}^n)^2)$.

In the previous examples the variety $\overline{\Sigma}$ is smooth, hence the following natural question.

Question. Is $\overline{\Sigma}$ smooth for all $n \ge 1$ and $p \ge 1$?

Lacking a positive answer to this question, since we want $C_p[\mathbb{R}^n]$ to be a smooth manifold, we will define it by resolving the singularities of $\overline{\Sigma}$. The existence of a resolution of singularities is given by Hironaka's theorem [1964a; 1964b]. Our references on this matter are [Kollár 2007; Włodarczyk 2005]. See also [Hauser 2003] for a softer introduction to this theory.

Proposition 5.5 (resolution of singularities). There exists a smooth manifold $C_p[\mathbb{R}^n]$ without boundary of dimension np and a smooth proper $\Pi : C_p[\mathbb{R}^n] \to (\mathbb{R}^n)^p \times \prod_{\varnothing \neq I \subset [\![1,p]\!]} \operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X])$ such that

- (1) $\Pi(C_p[\mathbb{R}^n]) = \overline{\Sigma};$
- (2) $\Pi^{-1}(\Sigma)$ is an open dense subset of $C_p[\mathbb{R}^n]$;
- (3) $\Pi_{\Pi^{-1}(\Sigma)}$ is \mathcal{C}^{∞} -diffeomorphism from $\Pi^{-1}(\Sigma)$ onto Σ .

Proof. We apply Hironaka's theorem [Kollár 2007, Theorem 3.27] to resolve the singularities of $\overline{\Sigma}$. Since $\overline{\Sigma}$ is algebraic by Lemma 5.3, there exists a smooth real-algebraic manifold $C_p[\mathbb{R}^n]$ and a projective morphism $\Pi : C_p[\mathbb{R}^n] \to \overline{\Sigma}$ such that Π is an isomorphism over the smooth locus of $\overline{\Sigma}$.

In particular $C_p[\mathbb{R}^n]$ is smooth, the map $\Pi : C_p[\mathbb{R}^n] \to (\mathbb{R}^n)^p \times \prod_{\emptyset \neq I \subset [\![1,p]\!]} \operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X])$ is smooth and proper, and $\Pi(C_p[\mathbb{R}^n]) \subset \overline{\Sigma}$. Since Σ is contained in the smooth locus of $\overline{\Sigma}$, the restriction of Π to $\Pi^{-1}(\Sigma)$ is an isomorphism; in particular (3) is satisfied.

According to [Włodarczyk 2005, Theorem 1.0.2], the manifold $C_p[\mathbb{R}^n]$ and the projection Π are obtained by a sequence of blow-ups along smooth submanifolds that do not intersect the regular locus of $\overline{\Sigma}$, and hence Σ . This ensures that conditions (1) and (2) are satisfied.

The following corollary proves the existence of the manifold $C_p[\mathbb{R}^n]$ and the proper surjection π : $C_p[\mathbb{R}^n] \to (\mathbb{R}^n)^p$ satisfying (1) in Theorem 1.1.

Corollary 5.6 (existence of the basis $C_p[\mathbb{R}^n]$). There exists a smooth manifold $C_p[\mathbb{R}^n]$ without boundary of dimension np and a smooth proper surjection $\pi : C_p[\mathbb{R}^n] \to (\mathbb{R}^n)^p$ such that

- (1) the open subset $\pi^{-1}((\mathbb{R}^n)^p \setminus \Delta_p)$ is dense in $C_p[\mathbb{R}^n]$ and π induces a \mathcal{C}^{∞} -diffeomorphism from this set onto $(\mathbb{R}^n)^p \setminus \Delta_p$;
- (2) for any nonempty $I \subset \llbracket 1, p \rrbracket$, the map $\mathcal{G}_I \circ \pi$ admits a unique smooth extension to $C_p[\mathbb{R}^n]$.

Proof. We consider $\Pi : C_p[\mathbb{R}^n] \to (\mathbb{R}^n)^p \times \prod_{\emptyset \neq I \subset [\![1,p]\!]} \operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X])$ given by Proposition 5.5 and we let $\pi = \Pi_0 \circ \Pi$. Since Grassmannians are compact, Π_0 is proper. Hence π is smooth and proper because Π and Π_0 are. The surjectivity of π is given by Lemma 5.1 and (1) in Proposition 5.5.

Item (1) in Corollary 5.6 is a consequence of Lemmas 5.2 and 5.3 and of conditions (2) and (3) in Proposition 5.5. Let $I \subset \llbracket 1, p \rrbracket$ be nonempty. On the dense open subset $\pi^{-1}((\mathbb{R}^n)^p \setminus \Delta_p)$ we have $\mathcal{G}_I \circ \pi = \prod_I \circ \Pi$ by definition. In the last equality, the right-hand side is well-defined and smooth on $C_p[\mathbb{R}^n]$, which yields the unique extension we are looking for.

Since it is defined using Hironaka's theorem, the manifold $C_p[\mathbb{R}^n]$ is not unique. However, the value of the smooth extension of $\mathcal{G}_I \circ \pi = \prod_I \circ \Pi$ at $z \in C_p[\mathbb{R}^n]$ only depends on $\Pi(z) \in \overline{\Sigma}$. So this extension does not really depend on the choice of a resolution of singularities. In the following we choose once and for all a realization of $\pi : C_p[\mathbb{R}^n] \to (\mathbb{R}^n)^p$ as in Corollary 5.6. Thanks to (1), we can identify the configuration space $(\mathbb{R}^n)^p \setminus \Delta_p$ with its open dense preimage by π . Under this identification, (2) states that the maps $(\mathcal{G}_I)_{I \subset [\![1,p]\!]}$ extend smoothly to $C_p[\mathbb{R}^n]$. So, from now on, we consider \mathcal{G}_I as a smooth map from $C_p[\mathbb{R}^n]$ to $\operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X])$.

5.2. Definition of the bundle $\mathcal{MJ}_p(\mathbb{R}^n)$. Now that we have defined the base space $C_p[\mathbb{R}^n]$ of our multijet bundle, we can define the bundle itself. The purpose of this section is to construct the vector bundle $\mathcal{MJ}_p(\mathbb{R}^n) \to C_p[\mathbb{R}^n]$ of multijets for functions from \mathbb{R}^n to \mathbb{R} , and the associated multijet map. The construction for vector-valued maps, explained in Section 5.4, is basically a fiberwise direct sum of this simpler case.

Recall that we defined the projections

$$C_p[\mathbb{R}^n] \xrightarrow{\Pi} \overline{\Sigma} \xrightarrow{\Pi_0} (\mathbb{R}^n)^p$$

and that $\pi = \Pi_0 \circ \Pi$. Thanks to Corollary 5.6, and under the identification discussed above, the map $\mathcal{G} = \mathcal{G}_{\llbracket 1,p \rrbracket}$ defined by (4-1) extends as a smooth map from $C_p[\mathbb{R}^n]$ to $\operatorname{Gr}_p(\mathbb{R}_{p-1}[X])$. Seen as a collection of subspaces of $\mathbb{R}_{p-1}[X]$ indexed by $C_p[\mathbb{R}^n]$, this means that \mathcal{G} defines a smooth vector sub-bundle of corank *p* in the trivial bundle $\mathbb{R}_{p-1}[X] \times C_p[\mathbb{R}^n] \to C_p[\mathbb{R}^n]$. We define our multijet bundle as the quotient of this trivial bundle by \mathcal{G} .

Definition 5.7 (vector bundle of multijets). Let $n \ge 1$ and $p \ge 1$. The vector bundle of multijets of order p on \mathbb{R}^n is the smooth vector bundle of rank p over $C_p[\mathbb{R}^n]$ defined by

$$\mathcal{MJ}_p(\mathbb{R}^n) = (\mathbb{R}_{p-1}[X] \times C_p[\mathbb{R}^n])/\mathcal{G}.$$

In particular, for any $z \in C_p[\mathbb{R}^n]$, the fiber $\mathcal{MJ}_p(\mathbb{R}^n)_z = \mathbb{R}_{p-1}[X]/\mathcal{G}(z)$ only depends on $\Pi(z) \in \overline{\Sigma}$.

Recalling the definition of Kergin polynomials given in Section 3.2, we can now define the *p*-multijet of a C^{p-1} function on \mathbb{R}^n .

Definition 5.8 (multijet of a function). Let $f \in C^{p-1}(\mathbb{R}^n)$ and $z \in C_p[\mathbb{R}^n]$. The *multijet of* f at z is the element of $\mathcal{MJ}_p(\mathbb{R}^n)_z$ defined as

$$\operatorname{mj}_{p}(f, z) = K(f, \pi(z)) \operatorname{mod} \mathcal{G}(z).$$

In particular, as an element of $\mathbb{R}_{p-1}[X]/\mathcal{G}(z)$, the multijet $\operatorname{mj}_p(f, z)$ only depends on $\Pi(z) \in \overline{\Sigma}$.

Example 5.9. In Example 5.4 we saw that in simple cases $\overline{\Sigma}$ is smooth. In these cases we set $C_p[\mathbb{R}^n] = \overline{\Sigma}$ and we can describe the bundle $\mathcal{MJ}_p(\mathbb{R}^n) \to C_p[\mathbb{R}^n]$ and the map mj_p .

• If p = 1, then $C_1[\mathbb{R}^n] = \mathbb{R}^n$ and $\mathcal{G} : x \mapsto \{0\} \subset \mathbb{R}_0[X] \simeq \mathbb{R}$. Thus $\mathcal{MJ}_1(\mathbb{R}^n)$ is the trivial bundle $\mathbb{R} \times \mathbb{R}^n \to \mathbb{R}^n$. Moreover, if $f \in \mathcal{C}^0(\mathbb{R}^n)$ then $K(f, x) = f(x) \in \mathbb{R}_0[X] \simeq \mathbb{R}$ and $mj_1(f, x) = f(x)$ for all $x \in \mathbb{R}^n$.

• If n = 1, then $C_p[\mathbb{R}] = \mathbb{R}^p$ and $\mathcal{G} : \underline{x} \mapsto \{0\} \subset \mathbb{R}_{p-1}[X]$. Thus $\mathcal{MJ}_p(\mathbb{R})$ is the trivial bundle $\mathbb{R}_{p-1}[X] \times \mathbb{R}^p \to \mathbb{R}^p$. If $f \in \mathcal{C}^{p-1}(\mathbb{R})$ then $\mathrm{mj}_p(f, \underline{x}) = K(f, \underline{x})$ is the Hermite interpolating polynomial of f at \underline{x} ; see Example 3.7.

Given $\underline{x} = (x_1, \ldots, x_p) \notin \Delta_p$, Lemma 4.4 shows that $\operatorname{ev}_{\underline{x}} : \mathbb{R}_{p-1}[X] \to \mathbb{R}^p$ is an isomorphism. We can then consider the Lagrange basis $(L_i(\underline{x}))_{1 \leq i \leq p}$ of $\mathbb{R}_{p-1}[X]$ which is the preimage by $\operatorname{ev}_{\underline{x}}$ of the canonical basis of \mathbb{R}^p . We then have $\operatorname{mj}_p(f, \underline{x}) = K(f, \underline{x}) = \sum_{i=1}^p f(x_i)L_i(\underline{x})$. Geometrically, this means that the map $(P, \underline{x}) \mapsto (\operatorname{ev}_{\underline{x}}(P), \underline{x})$ defines a local trivialization of $\mathcal{MJ}_p(\mathbb{R}) \to C_p[\mathbb{R}]$ over $\mathbb{R}^p \setminus \Delta_p$ and that $\underline{x} \mapsto (L_i(\underline{x}))_{1 \leq i \leq p}$ is the corresponding frame. Moreover, it is tautological that $\operatorname{mj}_p(f, \underline{x})$ reads as $(f(x_1), \ldots, f(x_p))$ in this trivialization.

In this example, one can also define a global trivialization of $\mathcal{MJ}_p(\mathbb{R})$ by considering the global frame of Newton polynomials $\underline{x} \mapsto (N_k(\underline{x}))_{1 \leq k \leq p}$, where $N_k(\underline{x}) = \prod_{1 \leq i < k} (X - x_i)$. By (3-3) we have $K(f, \underline{x}) = \sum_{k=1}^{p} f[x_1, \ldots, x_k] N_k(\underline{x})$, so that $\operatorname{mj}_p(f, \underline{x})$ reads as $(f[x_1, \ldots, x_k])_{1 \leq k \leq p}$ in this trivialization, where the divided differences are the classical ones in dimension 1. In this setting, we used in [Ancona and Letendre 2021] a strategy that can be roughly summarized as replacing $(f(x_i))_{1 \leq i \leq p}$ by $(f[x_1, \ldots, x_k])_{1 \leq k \leq p}$. Our present point of view shows that we were actually considering $\operatorname{mj}_p(f, \underline{x})$ all along, but read in different trivializations.

• If p = 2, we saw that $C_2[\mathbb{R}^n] = Bl_{\Delta_2}((\mathbb{R}^n)^2)$. Given $z \in C_2[\mathbb{R}^n]$, if $\pi(z) = (x_1, x_2) \notin \Delta_2$, we know that $\mathcal{G}(z) \subset \mathbb{R}_1[X]$ is the subspace of affine forms vanishing on the line $L_z \subset \mathbb{R}^n$ through x_1 and x_2 . It is then natural to think of the class of P modulo $\mathcal{G}(z)$ as its restriction to L_z . Parametrizing L_z by

$$t \mapsto x_1 + \frac{x_2 - x_1}{\|x_2 - x_1\|}t,$$

one can check that

$$P \mapsto P\left(x_1 + \frac{x_2 - x_1}{\|x_2 - x_1\|}T\right)$$

induces an isomorphism $\mathcal{MJ}_2(\mathbb{R}^n)_z \simeq \mathbb{R}_1[T] \simeq \mathbb{R}^2$, where *T* is univariate and the second isomorphism is obtained by reading coordinates in the canonical basis (1, *T*) of $\mathbb{R}_1[T]$.

Recalling Definition 3.1, we have

$$P[x_1, x_2] \cdot \frac{x_2 - x_1}{\|x_2 - x_1\|} T = \left(\int_0^1 D_{x_1 + t(x_2 - x_1)} P \cdot (x_2 - x_1) \, \mathrm{d}t \right) \frac{T}{\|x_2 - x_1\|} = \frac{P(x_2) - P(x_1)}{\|x_2 - x_1\|} T,$$

and $P = K(P, x_1, x_2)$ is given by (3-3). Letting $\tilde{P}(z) = P(x_2) - P(x_1)/||x_2 - x_1||$, we have

$$P\left(x_{1} + \frac{x_{2} - x_{1}}{\|x_{2} - x_{1}\|}T\right) = K(P, x_{1}, x_{2})\left(x_{1} + \frac{x_{2} - x_{1}}{\|x_{2} - x_{1}\|}T\right) = P(x_{1}) + \widetilde{P}(z)T$$

Thus, the previous isomorphism $\mathcal{MJ}_2(\mathbb{R}^n)_z \to \mathbb{R}^2$ is $(P \mod \mathcal{G}(z)) \mapsto (P(\pi(z)_1), \widetilde{P}(z))$.

Let us now consider $z \in \pi^{-1}(\Delta_2)$. This exceptional divisor is the projectivized normal bundle of Δ_2 in $(\mathbb{R}^n)^2$. So we can think of z as a point in the diagonal, say $(x, x) \in \Delta_2$, and a line in $(\mathbb{R}^n)^2$ which is orthogonal to Δ_2 , say spanned by (u, -u) with $u \in \mathbb{S}^{n-1}$. Then $z = \lim_{\varepsilon \to 0} (x + \varepsilon u, x - \varepsilon u)$ in $C_2[\mathbb{R}^n]$. By continuity, $\mathcal{G}(z)$ is the space of affine forms vanishing on the line $L_z \subset \mathbb{R}^n$ parametrized by $t \mapsto x + tu$. As above, mapping P to the coefficients of $P(x + Tu) = P(x) + (D_x P \cdot u)T \in \mathbb{R}_1[T]$ induces an isomorphism $\mathcal{MJ}_2(\mathbb{R}^n)_z \to \mathbb{R}^2$. Letting $\widetilde{P}(z) = D_x P \cdot u$, this isomorphism is again $(P \mod \mathcal{G}(z)) \mapsto (P(\pi(z)_1), \widetilde{P}(z))$.

Actually, one can check that everything depends smoothly on the base point $z \in C_2[\mathbb{R}^n]$, so that the bundle map $(P \mod \mathcal{G}(z), z) \mapsto (P(\pi(z)_1), \tilde{P}(z), z)$ defines a global trivialization $\mathcal{MJ}_2(\mathbb{R}^n) \rightarrow \mathbb{R}^2 \times C_2[\mathbb{R}^n]$. If $f \in \mathcal{C}^1(\mathbb{R}^n)$, with the same notation as above, $mj_2(f, z)$ reads in this trivialization as

$$\left(f(x_1), \frac{f(x_2) - f(x_1)}{\|x_2 - x_1\|}\right)$$

if $z \notin \pi^{-1}(\Delta_2)$ and as $(f(x), D_x f \cdot u)$ otherwise.

In these examples, the multijet bundle $\mathcal{MJ}_p(\mathbb{R}^n) \to C_p[\mathbb{R}^n]$ is trivial. This raises the following.

Question. Is $\mathcal{MJ}_p(\mathbb{R}^n) \to C_p[\mathbb{R}^n]$ trivial for all $n \ge 1$ and $p \ge 1$?

The following two lemmas prove that the bundle map $mj_p : C^{p-1}(\mathbb{R}^n) \times C_p[\mathbb{R}^n] \to \mathcal{MJ}_p(\mathbb{R}^n)$ satisfies (2) and (3) in Theorem 1.1.

Lemma 5.10 (regularity of multijets). The map $\operatorname{mj}_p(\cdot, z) : \mathcal{C}^{p-1}(\mathbb{R}^n) \to \mathcal{MJ}_p(\mathbb{R}^n)_z$ is a linear surjection for all $z \in C_p[\mathbb{R}^n]$. Additionally, let $l \ge 0$ and let $f \in \mathcal{C}^{l+p-1}(\mathbb{R}^n)$. Then $\operatorname{mj}_p(f, \cdot)$ is a section of class \mathcal{C}^l of $\mathcal{MJ}_p(\mathbb{R}^n) \to C_p[\mathbb{R}^n]$.

Proof. Let $z \in C_p[\mathbb{R}^n]$. The map $K(\cdot, \pi(z)) : \mathcal{C}^{p-1}(\mathbb{R}^n) \to \mathbb{R}_{p-1}[X]$ is linear by Lemma 3.8. It is also surjective since its restriction to $\mathbb{R}_{p-1}[X]$ is the identity. Since $\operatorname{mj}_p(\cdot, z)$ is the composition of $K(\cdot, \pi(z))$ with the canonical projection from $\mathbb{R}_{p-1}[X]$ onto $\mathcal{MJ}_p(\mathbb{R}^n)_z$, it is a linear surjection.

Let $l \ge 0$ and let $f \in C^{l+p-1}(\mathbb{R}^n)$. By Lemma 3.8, we have $K(f, \cdot) \in C^l((\mathbb{R}^n)^p, \mathbb{R}_{p-1}[X])$. Since π is smooth, we get $K(f, \cdot) \circ \pi \in C^l(C_p[\mathbb{R}^n], \mathbb{R}_{p-1}[X])$. In other words, $K(f, \cdot) \circ \pi$ defines a section of class C^l of the trivial bundle $\mathbb{R}_{p-1}[X] \times C_p[\mathbb{R}^n] \to C_p[\mathbb{R}^n]$. Since \mathcal{G} is a smooth sub-bundle of $\mathbb{R}_{p-1}[X] \times C_p[\mathbb{R}^n]$, projecting onto the quotient bundle $\mathcal{MJ}_p(\mathbb{R}^n)$ does not decrease the regularity. \Box

Lemma 5.11 (multijets and evaluation). Let $z \in C_p[\mathbb{R}^n]$ be such that $\pi(z) = (x_1, \ldots, x_p) \notin \Delta_p$. Then for all $f \in C^{p-1}(\mathbb{R}^n)$ we have $\operatorname{mj}_p(f, z) = 0$ if and only if, for all $i \in [[1, p]], f(x_i) = 0$.

Proof. Let us denote by $\underline{x} = (x_1, \ldots, x_p) = \pi(z) \notin \Delta_p$. For all $f \in \mathcal{C}^{p-1}(\mathbb{R}^n)$, we have

$$\mathrm{mj}_p(f,z) = 0 \quad \Longleftrightarrow \quad K(f,\underline{x}) \in \mathcal{G}(\underline{x}) \quad \Longleftrightarrow \quad \mathrm{ev}_{\underline{x}}(K(f,\underline{x})) = 0 \quad \Longleftrightarrow \quad \mathrm{ev}_{\underline{x}}(f) = 0,$$

since $K(f, \underline{x})$ interpolates the values of f on at x_1, \ldots, x_p (see Remark 3.5).

1449

Actually, we can describe more precisely the relation between multijets and evaluation outside of the diagonal. This will appear in the proof of Theorem 6.26 below. Thanks to Lemma 4.4, the smooth bundle map ev : $(P, \underline{x}) \mapsto (\text{ev}_{\underline{x}}(P), \underline{x})$ from $\mathbb{R}_{p-1}[X] \times ((\mathbb{R}^n)^p \setminus \Delta_p)$ to $\mathbb{R}^p \times ((\mathbb{R}^n)^p \setminus \Delta_p)$ is surjective and its kernel is exactly the sub-bundle ker ev = \mathcal{G} . Thus it induces a smooth bundle map $\tau : \mathcal{M}\mathcal{J}_p(\mathbb{R}^n)_{|(\mathbb{R}^n)^p \setminus \Delta_p} \to \mathbb{R}^p \times ((\mathbb{R}^n)^p \setminus \Delta_p)$ defined by $\tau(P \mod \mathcal{G}(\underline{x})) = (\text{ev}_{\underline{x}}(P), \underline{x})$, which is bijective. Thus τ defines a smooth local trivialization of $\mathcal{M}\mathcal{J}_p(\mathbb{R}^n)$ over $(\mathbb{R}^n)^p \setminus \Delta_p$. Moreover, for all $f \in \mathcal{C}^{p-1}(\mathbb{R}^n)$ and $z \in C_p[\mathbb{R}^n]$ such that $\underline{x} = \pi(z) \notin \Delta_p$ we have

$$\tau(\operatorname{mj}_p(f, z)) = \tau(K(f, \underline{x}) \mod \mathcal{G}(\underline{x})) = (\operatorname{ev}_{\underline{x}}(K(f, \underline{x})), \underline{x}) = (\operatorname{ev}_{\underline{x}}(f), \underline{x})$$

Hence $mj_p(f, z)$ simply reads as $(f(x_1), \ldots, f(x_p))$ in this trivialization.

5.3. Localness of multijets. The goal of this section is to prove that the multijet bundle $\mathcal{MJ}_p(\mathbb{R}^n) \to C_p[\mathbb{R}^n]$ defined in the previous section satisfies (4) in Theorem 1.1. Let $z \in C_p[\mathbb{R}^n]$, let $\underline{x} = \pi(z)$ and let $\mathcal{I} = \mathcal{I}(\underline{x})$ be as in Definition 2.3. As explained in Section 2.1, there is a unique $\underline{y} = (y_I)_{I \in \mathcal{I}} \in \mathbb{R}^{\mathcal{I}} \setminus \Delta_{\mathcal{I}}$ such that $\underline{x} = \iota_{\mathcal{I}}(\underline{y})$. Recalling that we dropped $V = \mathbb{R}$ from the notation in the present case, we can restate (4) in Theorem 1.1 as: there exists $\Theta_z : \prod_{I \in \mathcal{I}} \mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I} \to \mathcal{MJ}_p(\mathbb{R}^n)_z$ a linear surjection such that $\underline{mj}_p(f, z) = \Theta_z((j_{|I|-1}(f, y_I))_{I \in \mathcal{I}})$ for all $f \in C^{p-1}(\mathbb{R}^n)$.

This property is fundamental. First, it shows that $mj_p(f, z)$ is obtained by patching together (part of) the jets of order |I| - 1 of f at y_I , which justifies the name multijet. Second, it shows that $mj_p(f, z)$ only depends on the values of f in arbitrarily small neighborhoods of the y_I . This is not obvious at all since the definition of $mj_p(f, z)$ involves divided differences of f, which are obtained by integrating on the whole convex hull $\sigma(x)$ of the x_i (see Definition 3.1). In particular, it shows that $mj_p(f, z)$ makes sense even if f is only $C^{|I|-1}$ in some neighborhood of y_I for all $I \in \mathcal{I}$. Hence Definition 1.3 makes sense even if Ω is not convex.

In the following, we consider what we think of as the *I*-th part of a multijet, where $I \subset [[1, p]]$. This is just a variation on what we did in Definitions 5.7 and 5.8 and it is defined as follows.

Definition 5.12 (*I*-multijets). Let $n \ge 1$ and $p \ge 1$ and let $I \subset [[1, p]]$ be nonempty, we define the bundle of *I*-multijets as the following smooth bundle of rank |I| over $C_p[\mathbb{R}^n]$:

$$\mathcal{MJ}_{I}(\mathbb{R}^{n}) = (\mathbb{R}_{|I|-1}[X] \times C_{p}[\mathbb{R}^{n}])/\mathcal{G}_{I}.$$

Let $f \in \mathcal{C}^{|I|-1}(\mathbb{R}^n)$ and $z \in C_p[\mathbb{R}^n]$. We define by $mj_I(f, z) = K(f, \pi(z)_I) \mod \mathcal{G}_I(z) \in \mathcal{MJ}_I(\mathbb{R}^n)_z$ the *I-multijet* of f at z.

As explained in Section 5.1, for all $\emptyset \neq I \subset [\![1, p]\!]$ we have a map $\mathcal{G}_I : C_p[\mathbb{R}^n] \to \operatorname{Gr}_{|I|}(\mathbb{R}_{|I|-1}[X])$ extending the one on $(\mathbb{R}^n)^p \setminus \Delta_p$. Let $z \in C_p[\mathbb{R}^n]$ and $\underline{x} = \pi(z)$. As in Section 4 we define $\widetilde{\mathcal{G}}_I(z) = K(\cdot, \underline{x}_I)^{-1}(\mathcal{G}_I(z)) \in \operatorname{Gr}_{|I|}(\mathbb{R}_{p-1}[X])$, where $K(\cdot, \underline{x}_I) : \mathbb{R}_{p-1}[X] \to \mathbb{R}_{|I|-1}[X]$. Note that $\widetilde{\mathcal{G}}_I(z)$ has the same codimension as $\mathcal{G}_I(z)$ since $K(\cdot, \underline{x}_I)$ is surjective.

Lemma 5.13 (compatibility of the \mathcal{G}_I). For all $I \subset \llbracket 1, p \rrbracket$ and $z \in C_p[\mathbb{R}^n]$ we have $\mathcal{G}(z) \subset \widetilde{\mathcal{G}}_I(z)$.

Proof. Recall from Section 4 that $\mathcal{G}(z) \subset \widetilde{\mathcal{G}}_I(z)$ for any $z \in (\mathbb{R}^n)^p \setminus \Delta_p \subset C_p[\mathbb{R}^n]$, that is, $K(\cdot, \pi(z)_I)(\mathcal{G}(z)) \subset \mathcal{G}_I(z)$. This incidence relation is a closed condition. By construction, the subset $(\mathbb{R}^n)^p \setminus \Delta_p$ is dense
in $C_p[\mathbb{R}^n]$ and both terms in the previous inclusion are continuous with respect to z; see Lemma 3.8 and Corollary 5.6. Hence the inclusion actually holds for any $z \in C_p[\mathbb{R}^n]$. Thus $\mathcal{G}(z) \subset \widetilde{\mathcal{G}}_I(z)$ for all $z \in C_p[\mathbb{R}^n].$

Let $\emptyset \neq I \subset [[1, p]]$, let $z \in C_p[\mathbb{R}^n]$ and $\underline{x} = \pi(z)$. We consider $\operatorname{mj}_I(\cdot, z) : \mathbb{R}_{p-1}[X] \to \mathcal{MJ}_I(\mathbb{R}^n)_z$ from Definition 5.12. This linear map is surjective as the composition of $K(\cdot, \underline{x}_I)$ and the projection modulo $\mathcal{G}_I(z)$. Moreover, ker(mj_I(\cdot, z)) = $\widetilde{\mathcal{G}}_I(z)$ contains $\mathcal{G}(z)$ by Lemma 5.13. Hence, mj_I(\cdot, z) induces a surjective linear map from $\mathcal{MJ}_p(\mathbb{R}^n)_z = \mathbb{R}_{p-1}[X]/\mathcal{G}(z)$ onto $\mathcal{MJ}_I(\mathbb{R}^n)_z$ that we still denote by $mj_{I}(\cdot, z)$. This is summarized in the following commutative diagram, where the vertical arrows are the canonical projections and all arrows are surjective:

Note that $(P, z) \mapsto (K(P, \pi(z)_I), z)$ is a smooth bundle map over $C_p[\mathbb{R}^n]$ from $\mathbb{R}_{p-1}[X] \times C_p[\mathbb{R}^n]$ to $\mathbb{R}_{|I|-1}[X] \times C_p[\mathbb{R}^n]$. Hence, the previous diagram (5-2) defines a smooth surjective bundle map $\operatorname{mj}_{I} : (P \mod \mathcal{G}(z)) \mapsto (P \mod \mathcal{G}_{I}(z)) \text{ from } \mathcal{MJ}_{p}(\mathbb{R}^{n}) \text{ to } \mathcal{MJ}_{I}(\mathbb{R}^{n}) \text{ over } C_{p}[\mathbb{R}^{n}].$

Definition 5.14 (partitioned multijet). For all $\mathcal{I} \in \mathcal{P}_p$ and $z \in C_p[\mathbb{R}^n]$, we define a linear map from $\mathcal{MJ}_p(\mathbb{R}^n)_z$ to $\prod_{I \in \mathcal{I}} \mathcal{MJ}_I(\mathbb{R}^n)_z$ by $mj_{\mathcal{I}}(\cdot, z) : \alpha \mapsto (mj_I(\alpha, z))_{I \in \mathcal{I}}$.

As above, $\operatorname{mj}_{\mathcal{I}} : (\alpha, z) \mapsto \operatorname{mj}_{\mathcal{I}}(\alpha, z)$ defines a smooth bundle map over $C_p[\mathbb{R}^n]$ from $\mathcal{MJ}_p(\mathbb{R}^n)$ to $\bigoplus_{I \in \mathcal{I}} \mathcal{MJ}_I(\mathbb{R}^n)$, which is obtained as the quotient of $(P, z) \mapsto ((K(P, \pi(z)_I)_{I \in \mathcal{I}}, z))$. However, $mj_{\mathcal{I}}(\cdot, z)$ is not surjective in general. The following lemma proves its surjectivity in some cases.

Lemma 5.15 (splitting of multijets). Let $z \in C_p[\mathbb{R}^n]$, let $\underline{x} = \pi(z)$ and let $\mathcal{I}(\underline{x})$ be defined as in Definition 2.3. Then $\operatorname{mj}_{\mathcal{I}(x)}(\cdot, z) : \mathcal{MJ}_p(\mathbb{R}^n)_z \to \prod_{I \in \mathcal{I}(x)} \mathcal{MJ}_I(\mathbb{R}^n)_z$ is an isomorphism.

Proof. The map $mj_{\mathcal{I}(x)}(\cdot, z)$ is linear between two spaces of the same dimension $p = \sum_{I \in \mathcal{I}(x)} |I|$, so it is enough to prove its surjectivity. Let $(\alpha_I)_{I \in \mathcal{I}(\underline{x})} \in \prod_{I \in \mathcal{I}(\underline{x})} \mathcal{M}\mathcal{J}_I(\mathbb{R}^n)_z$. For each $I \in \mathcal{I}(\underline{x})$ there exists $P_I \in \mathbb{R}_{|I|-1}[X]$ such that $\alpha_I = P_I \mod \mathcal{G}_I(z)$. By Lemma 3.9, there exists $P \in \mathbb{R}_{p-1}[X]$ such that $K(P, \underline{x}_I) = P_I$ for all $I \in \mathcal{I}(\underline{x})$. Let $\alpha = P \mod \mathcal{G}(z) \in \mathcal{MJ}_p(\mathbb{R}^n)_z$. Then, for all $I \in \mathcal{I}(\underline{x})$, we have

$$\operatorname{mj}_{I}(\alpha, z) = \operatorname{mj}_{I}(P, z) = K(P, \underline{x}_{I}) \mod \mathcal{G}_{I}(z) = P_{I} \mod \mathcal{G}_{I}(z) = \alpha_{I}.$$

Hence $\operatorname{mj}_{\mathcal{I}(x)}(\alpha, z) = (\alpha_I)_{I \in \mathcal{I}(x)}$, and $\operatorname{mj}_{\mathcal{I}(x)}(\cdot, z)$ is indeed surjective.

Let $k \in \mathbb{N}$ and let $x \in \mathbb{R}^n$. By definition, two maps f and $g \in \mathcal{C}^k(\mathbb{R}^n)$ have the same k-jet at x if and only if they have the same Taylor polynomial of order k at x. Let $\underline{x} = (x, ..., x)$. Recalling Example 3.7, the linear map $K(\cdot, x) : \mathcal{C}^k(\mathbb{R}^n) \to \mathbb{R}_k[X]$ is surjective, and it induces an isomorphism $\mathcal{J}_k(\mathbb{R}^n)_x \simeq \mathbb{R}_k[X]$.

Let $z \in C_p[\mathbb{R}^n]$, let $\underline{x} = \pi(z)$, let $\mathcal{I} = \mathcal{I}(\underline{x})$ and let $(y_I)_{I \in \mathcal{I}} = \iota_{\mathcal{I}}^{-1}(\underline{x})$; see Definitions 2.3 and 2.6. For all $I \in \mathcal{I}$, the canonical isomorphism $\mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I} \simeq \mathbb{R}_{|I|-1}[X]$ allows us to see the projection from $\mathbb{R}_{|I|-1}[X]$ onto $\mathcal{M}\mathcal{J}_I(\mathbb{R}^n)_z$ as a canonical linear surjection $\varpi_{z,I}: \mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I} \to \mathcal{M}\mathcal{J}_I(\mathbb{R}^n)_z$.

Definition 5.16 (gluing map). Let $z \in C_p[\mathbb{R}^n]$, let $\underline{x} = \pi(z)$ and let $(y_I)_{I \in \mathcal{I}} = \iota_{\mathcal{I}}^{-1}(\underline{x})$, where $\mathcal{I} = \mathcal{I}(\underline{x})$. We define $\varpi_z : (\alpha_I)_{I \in \mathcal{I}} \mapsto (\varpi_{z,I}(\alpha_I))_{I \in \mathcal{I}}$ from $\prod_{I \in \mathcal{I}} \mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I}$ to $\prod_{I \in \mathcal{I}} \mathcal{M}\mathcal{J}_I(\mathbb{R}^n)_z$. We also define $\Theta_z = \operatorname{mj}_{\mathcal{I}}(\cdot, z)^{-1} \circ \varpi_z$.

We can now check that Θ_z satisfies (4) in Theorem 1.1.

Lemma 5.17 (localness of multijets). For all $z \in C_p[\mathbb{R}^n]$, the map Θ_z is a linear surjection from $\prod_{I \in \mathcal{I}} \mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I}$ to $\mathcal{M}\mathcal{J}_p(\mathbb{R}^n)_z$. Moreover, it is the only map such that

$$\forall f \in \mathcal{C}^{p-1}(\mathbb{R}^n), \quad \Theta_z((\mathbf{j}_{|I|-1}(f, y_I))_{I \in \mathcal{I}}) = \mathrm{mj}_p(f, z).$$

Proof. With the same notation as in Definition 5.16, for all $I \in \mathcal{I}$ the map $\varpi_{z,I}$ is a linear surjection by definition. Hence so is ϖ_z . Since $mj_{\mathcal{I}}(\cdot, z)$ is an isomorphism by Lemma 5.15, the map $\Theta_z = mj_{\mathcal{I}}(\cdot, z)^{-1} \circ \varpi_z$ is also a linear surjection.

Let $f \in C^{p-1}(\mathbb{R}^n)$. For all $I \in \mathcal{I}$, the image of $j_{|I|-1}(f, y_I)$ under the canonical isomorphism $\mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I} \simeq \mathbb{R}_{|I|-1}[X]$ is the Taylor polynomial $K(f, \underline{x}_I)$. Hence

$$\varpi_{z,I}(\mathbf{j}_{|I|-1}(f, \mathbf{y}_I)) = K(f, \underline{x}_I) \mod \mathcal{G}_I(z) = \mathrm{mj}_I(K(f, \underline{x}_I), z).$$

Thus, we have

$$\varpi_{\mathbb{Z}}((\mathbf{j}_{|I|-1}(f, y_I))_{I \in \mathcal{I}}) = (\mathrm{mj}_I(K(f, \underline{x}_I), z))_{I \in \mathcal{I}} = \mathrm{mj}_{\mathcal{I}}(\mathrm{mj}_p(f, z), z),$$

and finally

$$\Theta_{z}((\mathbf{j}_{|I|-1}(f, y_{I}))_{I \in \mathcal{I}}) = \mathrm{mj}_{p}(f, z).$$

Since the y_I are pairwise distinct, every element of $\prod_{I \in \mathcal{I}} \mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I}$ can be realized as $(j_{|I|-1}(f, y_I))_{I \in \mathcal{I}}$ for some $f \in \mathcal{C}^{\infty}(\mathbb{R}^n)$. Hence the previous relation completely defines Θ_z .

5.4. *Multijets of vector-valued maps.* So far we have only defined multijets of real-valued functions. In this section, we extend the previous construction to maps from \mathbb{R}^n to some vector space *V* of dimension $r \ge 1$. Let $\pi : C_p[\mathbb{R}^n] \to (\mathbb{R}^n)^p$ be given by Corollary 5.6 as before. We define $\mathcal{MJ}_p(\mathbb{R}^n, V) \to C_p[\mathbb{R}^n]$ as the tensor product of the bundle $\mathcal{MJ}_p(\mathbb{R}^n) \to C_p[\mathbb{R}^n]$ from Definition 5.7 with the trivial bundle $V \times C_p[\mathbb{R}^n] \to C_p[\mathbb{R}^n]$.

Definition 5.18 (multijet bundle of vector-valued maps). Let $n \ge 1$ and $p \ge 1$; let V be a real vector space of dimension $r \ge 1$. We define the *bundle of p-multijets of V-valued maps on* \mathbb{R}^n as the following smooth bundle of rank pr over $C_p[\mathbb{R}^n]$:

$$\mathcal{MJ}_p(\mathbb{R}^n, V) = \mathcal{MJ}_p(\mathbb{R}^n) \otimes V.$$

Definition 5.19 (multijet of a map). Let (v_1, \ldots, v_r) denote a basis of *V*. Let $z \in C_p[\mathbb{R}^n]$ and let $f = \sum_{i=1}^r f_i v_i \in C^{p-1}(\mathbb{R}^n, V)$. We define by $\text{mj}_p(f, z) = \sum_{i=1}^r \text{mj}_p(f_i, z) \otimes v_i \in \mathcal{MJ}_p(\mathbb{R}^n, V)_z$ the *p*-multijet of *f* at *z*.

Lemma 5.20 (independence from the basis). In Definition 5.19, the vector $mj_p(f, z)$ does not depend on the choice of the basis (v_1, \ldots, v_r) .

Proof. Let (w_1, \ldots, w_r) be another basis of V. There exists a matrix $(a_{ij})_{1 \le i, j \le r} \in \operatorname{GL}_r(\mathbb{R})$ such that $v_i = \sum_{j=1}^r a_{ij} w_j$ for all $i \in \llbracket 1, r \rrbracket$. Letting $g_j = \sum_{i=1}^r a_{ij} f_i$ for all $j \in \llbracket 1, r \rrbracket$, we get

$$f = \sum_{i=1}^{r} f_i v_i = \sum_{1 \le i, j \le r} a_{ij} f_i w_j = \sum_{j=1}^{r} g_j w_j.$$

Then, by linearity of the *p*-multijet for functions, we have

$$\sum_{j=1}^{r} \operatorname{mj}_{p}(g_{j}, z) \otimes w_{j} = \sum_{1 \leq i, j \leq r} a_{ij} \operatorname{mj}_{p}(f_{i}, z) \otimes w_{j} = \sum_{i=1}^{r} \operatorname{mj}_{p}(f_{i}, z) \otimes v_{i}.$$

Example 5.21. If $V = \mathbb{R}^r$, then for all $z \in C_p[\mathbb{R}^n]$ and $f = (f_1, \ldots, f_r) \in \mathcal{C}^{p-1}(\mathbb{R}^n, \mathbb{R}^r)$ we have

$$\mathcal{MJ}_p(\mathbb{R}^n,\mathbb{R}^r)_z = \mathcal{MJ}_p(\mathbb{R}^n)_z \otimes \mathbb{R}^r \simeq (\mathcal{MJ}_p(\mathbb{R}^n)_z)^r,$$

and under this canonical isomorphism $mj_p(f, z) = (mj_p(f_i, z))_{1 \le i \le r}$, as one would expect.

Let $x \in \mathbb{R}^n$ and $k \in \mathbb{N}$. We have a canonical isomorphism $\mathcal{J}_k(\mathbb{R}^n, V)_x \simeq \mathcal{J}_k(\mathbb{R}^n)_x \otimes V$. If (v_1, \ldots, v_r) is a basis of *V*, this isomorphism is totally determined by the fact that the image of $j_k(f, x)$ is $\sum_{i=1}^r j_k(f_i, x) \otimes v_i$ for all $f = \sum_{i=1}^r f_i v_i \in \mathcal{C}^k(\mathbb{R}^n, V)$. As in the proof of Lemma 5.20, this does not depend on the choice of the basis (v_1, \ldots, v_r) .

Definition 5.22 (gluing map for vector-valued multijets). Let $z \in C_p[\mathbb{R}^n]$, let $\underline{x} = \pi(z)$, let $\mathcal{I} = \mathcal{I}(\underline{x})$ and let $(y_I)_{I \in \mathcal{I}} = \iota_{\mathcal{I}}^{-1}(\underline{x})$. Using the previous canonical isomorphisms, we have

$$\prod_{I\in\mathcal{I}}\mathcal{J}_{|I|-1}(\mathbb{R}^n,V)_{y_I}=\prod_{I\in\mathcal{I}}\mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I}\otimes V=\left(\prod_{I\in\mathcal{I}}\mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I}\right)\otimes V.$$

Recalling Definition 5.16, we define a linear map

$$\Theta_{z}: \prod_{I \in \mathcal{I}} \mathcal{J}_{|I|-1}(\mathbb{R}^{n}, V)_{y_{I}} \to \mathcal{M}\mathcal{J}_{p}(\mathbb{R}^{n}, V)_{z}$$

by $\Theta_z(\alpha \otimes v) = \Theta_z(\alpha) \otimes v$ for all $\alpha \in \prod_{I \in \mathcal{I}} \mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I}$ and all $v \in V$.

We now have everything we need to prove Theorem 1.1.

Proof of Theorem 1.1. The base space $C_p[\mathbb{R}^n]$ and the projection π are given by Corollary 5.6. In particular, they satisfy (1) in Theorem 1.1. Definition 5.19 and Lemma 5.10 show that mj_p satisfies (2). Similarly, (3) is satisfied thanks to Lemma 5.11 and the definition of mj_p for V-valued maps.

Let us check that the linear maps Θ_z from Definition 5.22 satisfy (4). Let $z \in C_p[\mathbb{R}^n]$, let $\underline{x} = \pi(z)$, let $\mathcal{I} = \mathcal{I}(\underline{x})$ and let $(y_I)_{I \in \mathcal{I}} = \iota_{\mathcal{I}}^{-1}(\underline{x})$. Let us also denote by (v_1, \ldots, v_r) a basis of *V*. Let $\alpha \in \mathcal{MJ}_p(\mathbb{R}^n, V)_z$. There exists $\alpha_1, \ldots, \alpha_r \in \mathcal{MJ}_p(\mathbb{R}^n)_z$ such that $\alpha = \sum_{i=1}^r \alpha_i \otimes v_i$. By Lemma 5.17, for each $i \in [[1, r]]$, there exists $\beta_i \in \prod_{I \in \mathcal{I}} \mathcal{J}_{|I|-1}(\mathbb{R}^n)_{y_I}$ such that $\alpha_i = \Theta_z(\beta_i)$. Hence,

$$\Theta_z\left(\sum_{i=1}^r \beta_i \otimes v_i\right) = \sum_{i=1}^r \Theta_z(\beta) \otimes v_i = \alpha,$$

and Θ_z is indeed surjective.

Finally, let us consider $f = \sum_{i=1}^{r} f_i v_i \in C^{p-1}(\mathbb{R}^n, V)$. Then, by Lemma 5.17 once again,

$$\Theta_{z}((\mathbf{j}_{|I|-1}(f, y_{I}))_{I \in \mathcal{I}}) = \Theta_{z}\left(\sum_{i=1}^{r} (\mathbf{j}_{|I|-1}(f_{i}, y_{I}))_{I \in \mathcal{I}} \otimes v_{i}\right) = \sum_{i=1}^{r} \Theta_{z}((\mathbf{j}_{|I|-1}(f_{i}, y_{I}))_{I \in \mathcal{I}}) \otimes v_{i}$$
$$= \sum_{i=1}^{r} \mathrm{mj}_{p}(f_{i}, z) \otimes v_{i} = \mathrm{mj}_{p}(f, z).$$

Remark 5.23. Another way to define $\mathcal{MJ}_p(\mathbb{R}^n, V)$ and mj_p is the following. If $f : \mathbb{R}^n \to V$ is regular enough, then the divided differences from Definition 3.1 still make sense, only this time $f[x_0, \ldots, x_k] \in$ $\text{Sym}^k(\mathbb{R}^n) \otimes V$. Then, one can still define $K(f, \underline{x})$ as in Proposition 3.4, and it defines an element of $\mathbb{R}_{p-1}[X] \otimes V$ that interpolates the divided differences of f. Similarly, everything we did from Section 3.1 to Section 5.3 can be adapted to the case of V-valued maps, simply by tensoring each vector space by V, and each linear map by Id_V . One can check that we recover the same objects as in Definitions 5.18 and 5.19, up to canonical isomorphisms.

6. Application to zeros of Gaussian fields

This section is concerned with our application of multijet bundles to Gaussian fields. In Section 6.1, we describe the local model for the Gaussian fields with values in a vector bundle that we consider. In Section 6.2, we prove a Bulinskaya-type lemma and a Kac–Rice formula for the zeros of these fields. Section 6.3 is dedicated to the definition of the Kac–Rice densities of order larger than 2. We also relate the properties of these functions with the moments of the linear statistics associated with our field. Finally, we prove Theorems 1.6 and 1.9 in Section 6.4, using the multijet bundles defined in Theorem 1.1.

6.1. *Gaussian vectors and Gaussian sections.* In this section, we briefly recall some notation and conventions concerning Gaussian vectors. Then we describe the local model for Gaussian fields with values in a vector bundle. We will mostly consider centered random vectors in finite-dimensional vector spaces, so we restrict ourselves to this setting. In the following, V is a finite-dimensional real vector space.

Definition 6.1 (Gaussian vector). We say that a random vector X with values in V is a *centered Gaussian* vector if, for all $\eta \in V^*$, the real random variable $\eta(X)$ is a centered Gaussian in \mathbb{R} .

In particular, a centered Gaussian vector in V has finite moments up to any order. Let us assume that V is endowed with an inner product $\langle \cdot, \cdot \rangle$. Then for all $v \in V$, we define $v^* = \langle v, \cdot \rangle \in V^*$.

Definition 6.2 (variance operator). Let *X* be a centered Gaussian vector in $(V, \langle \cdot, \cdot \rangle)$. Then its *variance operator* is the nonnegative self-adjoint endomorphism $Var(X) = \mathbb{E}[X \otimes X^*]$ of *V*. We say that *X* is *nondegenerate* if Var(X) is invertible.

Recall that a centered Gaussian vector in $(V, \langle \cdot, \cdot \rangle)$ is completely determined by its variance. In the following, we denote by $\mathcal{N}(0, \Lambda)$ the centered Gaussian distribution of variance Λ , and by $X \sim \mathcal{N}(0, \Lambda)$ the fact that X follows this distribution.

Definition 6.3 (Gaussian field). Let $E \to M$ be a vector bundle over some manifold M. We say that a random section s of $E \to M$ is a *centered Gaussian field* if for all $m \ge 1$ and all x_1, \ldots, x_m the

random vector $(s(x_1), \ldots, s(x_m))$ is a centered Gaussian. We say that this field is *nondegenerate* if s(x) is nondegenerate for all $x \in M$.

If the centered Gaussian field *s* is C^p , then its jet $j_k(s, x)$ is a centered Gaussian for all $x \in M$. Thus, the definition of *p*-nondegeneracy of the field makes sense; see Definition 1.8. Note that 0-nondegenerate simply means nondegenerate.

Since this will appear in several places later on, let us describe the local model for Gaussian fields in this context. Let $x_0 \in M$. There exists a chart (U, φ) of M around x_0 . That is $\varphi : U \to \Omega$ is a diffeomorphism between an open neighborhood U of x_0 and an open subset $\Omega \subset \mathbb{R}^n$. Up to reducing U, we can assume that E is trivial over U, i.e., there exists a trivialization $\tau : E_{|U} \to \mathbb{R}^r \times U$. Letting $\tau_{\varphi} = (\mathrm{Id}, \varphi) \circ \tau$, we have the following commutative diagram, where arrows on the top row are bundle maps covering the maps on the bottom row:

Let *s* be a local section of $E_{|U}$. Then $\tau_{\varphi} \circ s \circ \varphi^{-1}$ is a section of the trivial bundle on the right-hand side of (6-1). Hence there exists a map $f : \Omega \to \mathbb{R}^r$ such that $\tau_{\varphi} \circ s \circ \varphi^{-1} = (f, \operatorname{Id})$. For all $x \in \Omega$, the vector f(x)is the image of $s(\varphi^{-1}(x))$ by a linear bijection. Thus, if $s : M \to E$ is a centered Gaussian field, its restriction to *U* corresponds to a centered Gaussian field $f : \Omega \to \mathbb{R}^r$. Moreover, *f* has the same regularity as *s*.

The local trivializations on the diagram (6-1) induce a similar picture for jet bundles so that we have a local trivialization $\mathcal{J}_p(U, E_{|U}) \simeq \mathcal{J}_p(\Omega, \mathbb{R}^r)$, under which $j_p(f, x)$ corresponds to $j_p(s, \varphi^{-1}(x))$. Thus, the Gaussian section *s* is *p*-nondegenerate in the sense of Definition 1.8 if and only if *f* is *p*-nondegenerate in the sense of Definition 1.4. If this is the case, up to replacing Ω by a smaller Ω' such that $\overline{\Omega}' \subset \Omega$ is compact, we can assume that *f* is uniformly *p*-nondegenerate, in the sense that det $Var(j_p(f, x))$ is bounded from below on Ω . This local picture is summarized in the following lemma.

Lemma 6.4 (local model for Gaussian fields). Let $s : M \to E$ be a centered *p*-nondegenerate Gaussian field. For all $x_0 \in M$, there exist an open neighborhood U of x_0 and a local trivialization of the form (6-1) such that *s* reads in local coordinates as a centered Gaussian field $f : \Omega \to \mathbb{R}^r$ of the same regularity as *s* and which is uniformly *p*-nondegenerate on Ω .

6.2. Bulinskaya lemma and Kac–Rice formula for the expectation. Let (M, g) be a Riemannian manifold of dimension $n \ge 1$ without boundary and let $E \to M$ be a vector bundle of rank $r \in [[1, n]]$. We consider a nondegenerate centered Gaussian field $s : M \to E$, in the sense of Definition 6.3. The goal of this section is to state a Bulinskaya-type lemma and a Kac–Rice formula for the expectation of the linear statistics of s.

When *M* is an open subset of \mathbb{R}^n and $E = \mathbb{R}^r \times M$ is trivial, these results are proved by Armentano, Azaïs and Leòn [Armentano et al. 2023b, Proposition 2.1 and Theorem 2.2]. They are extended to fields

on submanifolds of \mathbb{R}^N in [Armentano et al. 2023b, Section 9.1]. In the following we check that the results of that work can be adapted to the case of Gaussian sections.

Remark 6.5. Some readers are most interested in the zeros of Gaussian fields from \mathbb{R}^n to \mathbb{R}^r and the present geometric setting may seem overly complicated to them. Let us stress that, even in the simpler setting of fields from \mathbb{R}^n to \mathbb{R}^r , our proof of Theorem 1.6 uses the Kac–Rice formula in the more general setting we are studying here.

In order to state the Bulinskaya lemma and the Kac-Rice formula, we need the following.

Definition 6.6 (Jacobian determinant). Let $L: V \to V'$ be a linear map between Euclidean spaces and let L^* denote its adjoint map. The *Jacobian* of L is defined as $Jac(L) = det(LL^*)^{1/2}$.

Remark 6.7. We have $Jac(L) \ge 0$, and Jac(L) > 0 if and only if L is surjective. In particular, the fact that Jac(L) = 0 depends only on L and not on the Euclidean structures on V and V'. Thus the condition Jac(L) = 0 makes sense even if no inner product is specified.

Proposition 6.8 (weak Bulinskaya lemma). Let ∇ be a connection on $E \rightarrow M$. If the centered Gaussian field $s: M \rightarrow E$ is C^1 and nondegenerate, the (n-r)-dimensional Hausdorff measure of

$$\{x \in M \mid s(x) = 0 \text{ and } \operatorname{Jac}(\nabla_x s) = 0\}$$

is almost surely 0.

Remark 6.9. If s(x) = 0 then $\nabla_x s$ does not depend on ∇ . Hence the random set we are interested in Proposition 6.8 does not depend on the choice ∇ .

Proof of Proposition 6.8. We can cover M by countably many open trivialization domains of the type described in Lemma 6.4. Then it is enough to prove the result in each of these domains.

Let $U \subset M$ be as Lemma 6.4. In local coordinates, the restriction of *s* reads as a uniformly nondegenerate C^1 centered Gaussian field $f : \Omega \to \mathbb{R}^r$, where $\Omega \subset \mathbb{R}^n$ is open. Moreover, for any $x \in \Omega$ such that f(x) = 0, the covariant derivative of *s* reads as $D_x f$, independently of the choice of ∇ . Thus, we are left with proving that the (n-r)-dimensional Hausdorff measure of

$$\{x \in \Omega \mid f(x) = 0 \text{ and } \operatorname{Jac}(D_x f) = 0\}$$

is almost surely 0, which is given by [Armentano et al. 2023b, Proposition 2.1].

Let us assume from now on that the centered Gaussian field $s: M \to E$ is C^1 and nondegenerate. We denote its zero set by $Z = s^{-1}(0)$. Let us define $Z_{sing} = \{x \in Z \mid Jac(\nabla_x s) = 0\}$ and $Z_{reg} = Z \setminus Z_{sing}$. By Proposition 6.8, the (n-r)-dimensional Hausdorff measure of the singular part Z_{sing} is almost surely 0. On the other hand, the regular part Z_{reg} is the set of points where *s* vanishes transversally. As such, it is a (possibly empty) C^1 submanifold of *M* of codimension *r* without boundary. Thus, *Z* is almost surely the union of an open (in *Z*) regular part Z_{reg} of dimension n - r, and a negligible singular part Z_{sing} that we can think of as a set of lower dimension. Let us mention that, under additional assumptions on the field, the singular part is almost surely empty.

Proposition 6.10 (strong Bulinskaya lemma). If the centered Gaussian field $s : M \to E$ is C^2 and 1-nondegenerate, then $Z_{sing} = \emptyset$ almost surely.

Proof. As in the proof of Proposition 6.8, it is enough to prove the result in local coordinates given by Lemma 6.4. In these coordinates, *s* reads as a C^2 centered Gaussian field $f : \Omega \to \mathbb{R}^r$ which is uniformly 1-nondegenerate, that is, det Var $(f(x), D_x f)$ is bounded from below on Ω . Then the result follows from [Azaïs and Wschebor 2009, Proposition 6.12].

The Riemannian metric g induces a metric on Z_{reg} , which in turn defines an (n-r)-dimensional Riemannian volume measure $dVol_Z$. This measure coincides with the (n-r)-dimensional Hausdorff measure on Z. In the following, we consider this measure as a Radon measure on M defined as follows. Recall that the space of Radon measures is the topological dual of $C_c^0(M)$, and that being a nonnegative Radon is equivalent to being a Borel measure which is finite on compact subsets.

Definition 6.11 (random measure associated with *Z*). We denote by ν the random nonnegative Radon measure on *M* defined by

$$\forall \phi \in \mathcal{C}^0_c(M), \quad \langle v, \phi \rangle = \int_{Z_{\text{reg}}} \phi(x) \, \mathrm{dVol}_Z(x).$$

We define $\langle \nu, \phi \rangle$ similarly if ϕ is nonnegative Borel function (in which case $\langle \nu, \phi \rangle \in [0, +\infty]$) or if ϕ is a Borel function such that $\langle \nu, |\phi| \rangle < +\infty$ almost surely.

Example 6.12. If n = r then Z is almost surely locally finite. In this case $\nu = \sum_{x \in Z} \delta_x$ is the random counting measure of this point process.

Let us go back to the local model of Lemma 6.4. Around any $x_0 \in M$ there exists a chart (U, φ) and a local trivialization of the kind described by (6-1). Since *s* is C^1 and nondegenerate, it corresponds in local coordinates to a C^1 nondegenerate Gaussian field $f : \Omega \to \mathbb{R}^r$. We still denote by *Z* (resp. Z_{reg}) the image of *Z* (resp. Z_{reg}) by φ , which is the zero set of *f* (resp. its regular part). Similarly, we still denote by *g* (resp. dVol_{*Z*}) the push-forward to Ω of the metric *g* (resp. of the measure dVol_{*Z*}), and we identify test-functions on *U* with test-functions on Ω . Thus, if $\phi \in L_c^{\infty}(U)$ we have

$$\langle v, \phi \rangle = \int_{Z_{\text{reg}}} \phi(x) \, \mathrm{dVol}_Z(x),$$

where we think of everything on the right-hand side as defined on Ω . Now, the measure $dVol_Z$ is the (n-r)-dimensional Riemannian volume on $Z \subset \Omega \subset \mathbb{R}^n$ induced by g. In the following, we will need to understand how it compares with the Riemannian volume $dVol_Z^0$ on Z induced by the Euclidean metric. This is the purpose of what comes next.

Definition 6.13 (Riemannian densities). Let $x \in \Omega$ and let *G* be a subspace of \mathbb{R}^n . We denote by $det(g(x)|_G)$ the determinant of the restriction to *G* of the inner product g(x), in any basis of *G* which is orthonormal for the Euclidean inner product of \mathbb{R}^n .

For all $r \in [[0, n]]$, we denote by $\gamma_r : \Omega \times \operatorname{Gr}_r(\mathbb{R}^n) \to (0, +\infty)$ the continuous map defined by $\gamma_r : (x, G) \mapsto \operatorname{det}(g(x)_{|G})^{1/2}$. We also write $\gamma : x \mapsto \gamma_0(x, \mathbb{R}^n)$ for simplicity.

Lemma 6.14 (comparing volumes). Let Z be a submanifold of codimension r of Ω and let $d\operatorname{Vol}_Z$ (resp. $d\operatorname{Vol}_Z^0$) denote the (n-r)-dimensional Riemannian volume on Z induced by g (resp. the Euclidean metric). Then $d\operatorname{Vol}_Z$ admits the density $x \mapsto \gamma_r(x, T_x Z)$ with respect to $d\operatorname{Vol}_Z^0$. In particular $d\operatorname{Vol}_\Omega$ admits the density γ with respect to the Lebesgue measure on Ω .

Proof. This follows directly from the definition of the Riemannian volume measures; see [Lee 2018, Chapter 3] for example. \Box

We can now state and prove the Kac–Rice formula for the expectation of the linear statistics in our setting of Gaussian fields on M with values in a vector bundle E.

Definition 6.15 (Kac–Rice density). Let $\rho_1 : M \to [0, +\infty)$ be defined by

$$\rho_1: x \longmapsto \frac{\mathbb{E}[\operatorname{Jac}(\nabla_x s) \mid s(x) = 0]}{\det(2\pi \operatorname{Var}(s(x)))^{1/2}},$$

where the numerator stands for the conditional expectation of $Jac(\nabla_x s)$ given that s(x) = 0.

Remark 6.16. Since *s* is nondegenerate and C^1 , the function ρ_1 is well-defined and continuous. Moreover, it does not depend on the choice of ∇ , nor on the choice of a metric on *E*.

Proposition 6.17 (Kac–Rice formula for the expectation). Let *s* be a nondegenerate C^1 centered Gaussian field. Then, for any Borel function $\phi : M \to \mathbb{R}$ which is nonnegative or such that $\phi \rho_1 \in L^1(M)$, we have

$$\mathbb{E}[\langle \nu, \phi \rangle] = \int_M \phi(x) \rho_1(x) \, \mathrm{dVol}_M(x),$$

i.e., $\mathbb{E}[v]$ *is the Radon measure on M with density* ρ_1 *with respect to the Riemannian volume* $dVol_M$.

Remark 6.18. For any Borel maps ϕ_1 and ϕ_2 , we have

$$\mathbb{E}[|\langle \nu, \phi_1 \rangle - \langle \nu, \phi_2 \rangle|] \leq \mathbb{E}[\langle \nu, |\phi_1 - \phi_2|\rangle] = \int_M |\phi_1(x) - \phi_2(x)|\rho_1(x) \,\mathrm{dVol}_M(x).$$

Then, if $\phi_1 = \phi_2$ almost everywhere on *M*, we have $\langle \nu, \phi_1 \rangle = \langle \nu, \phi_2 \rangle$ almost surely. Thus, $\langle \nu, \phi \rangle$ makes sense as a random variable even if ϕ is only defined up to modification on a negligible set.

Proof of Proposition 6.17. By a partition of unity argument, it enough to prove the result if ϕ is compactly supported in an open domain U satisfying the same properties as in Lemma 6.4. In this case, in local coordinates, the field s corresponds to a nondegenerate C^1 centered Gaussian field $f : \Omega \to \mathbb{R}^r$ with $\Omega \subset \mathbb{R}^n$ open. Thanks to Remark 6.16, we can assume that ∇ corresponds in this trivialization to the standard derivation for maps from Ω to \mathbb{R}^r and that the metric on E corresponds to the canonical inner product on \mathbb{R}^r . Identifying Z_{reg} , the metric g, the measure $d \operatorname{Vol}_Z$ and the test-function ϕ with their images in the trivialization, we have reduced our problem to proving the result for the vanishing locus of f with the volume measures induced by g.

By Lemma 6.14, we have

$$\mathbb{E}[\langle v, \phi \rangle] = \mathbb{E}\left[\int_{Z_{\text{reg}}} \phi(x) \, \mathrm{d} \operatorname{Vol}_{Z}(x)\right] = \mathbb{E}\left[\int_{Z_{\text{reg}}} \phi(x) \gamma_{r}(x, \ker D_{x}f) \, \mathrm{d} \operatorname{Vol}_{Z}^{0}(x)\right].$$

For all $x \in \Omega$ and $\lambda \in C^0(\Omega, \mathcal{L}(\mathbb{R}^n, \mathbb{R}^r))$ we define $\Psi(x, \lambda) = \phi(x)\gamma_r(x, \ker \lambda(x))\mathbf{1}_O(\lambda(x))$, where $O = \{L \in \mathcal{L}(\mathbb{R}^n, \mathbb{R}^r) \mid \operatorname{Jac}(L) > 0\}$. Since *O* is open and the maps ker : $O \to \operatorname{Gr}_r(\mathbb{R}^n)$ and γ_r are continuous, the map Ψ is lower semicontinuous with respect to each variable, where $C^0(\Omega, \mathcal{L}(\mathbb{R}^n, \mathbb{R}^r))$ is equipped with the weak topology. Thus, we can apply the Euclidean Kac–Rice formula with weight from [Armentano et al. 2023b, Theorem 7.1 and Remark 8] to $\Psi(x, Df)$. This yields

$$\mathbb{E}[\langle \nu, \phi \rangle] = \int_{\Omega} \phi(x) \frac{\mathbb{E}[\gamma_r(x, \ker D_x f) \operatorname{Jac}^0(D_x f) \mid f(x) = 0]}{\det(2\pi \operatorname{Var}(f(x)))^{1/2}} \, \mathrm{d}x,$$

where Jac^0 means that we computed the Jacobian with respect to the Euclidean metric on \mathbb{R}^n .

To conclude, we need to compare Jac^0 with the Jacobian Jac with respect to g. This is the content of Lemma 6.19 below, which yields that $\gamma_r(x, \ker D_x f) Jac^0(D_x f) = \gamma(x) Jac(D_x f)$. Since $\gamma(x)$ is deterministic, by Lemma 6.14 we have

$$\mathbb{E}[\langle v, \phi \rangle] = \int_{\Omega} \phi(x) \frac{\mathbb{E}[\operatorname{Jac}(D_x f) \mid f(x) = 0]}{\det(2\pi \operatorname{Var}(f(x)))^{1/2}} \gamma(x) \, \mathrm{d}x = \int_{\Omega} \phi(x) \rho_1(x) \, \mathrm{d}\operatorname{Vol}_{\Omega}(x)$$

This proves that the result holds locally, that is, for a field $f : \Omega \to \mathbb{R}^r$, with the volume measures induced by any Riemannian metric on Ω , which concludes the proof.

Lemma 6.19 (comparing Jacobians). Let $x \in \Omega$ and let $L : \mathbb{R}^n \to \mathbb{R}^r$ be a surjective linear map. With the same notation as above, we have $\gamma_r(x, \ker L) \operatorname{Jac}^0(L) = \gamma(x) \operatorname{Jac}(L)$.

Proof. We denote by L_g^* (resp. L_0^*) the adjoint of L with respect to the inner product g(x) (resp. the Euclidean inner product). In a Euclidean orthonormal basis adapted to ker $(L)^{\perp} \oplus$ ker(L), the matrix of g(x) is symmetric of the form $\begin{pmatrix} A & ^tB \\ B & C \end{pmatrix}$ with A and C positive-definite, the matrix of L is $(F \ 0)$, that of L_0^* is $\begin{pmatrix} {}^tF \\ 0 \end{pmatrix}$ and that of L_g^* is $\begin{pmatrix} X \\ Y \end{pmatrix}$. We have $\begin{pmatrix} {}^tF \\ B & C \end{pmatrix} \begin{pmatrix} X \\ B & C \end{pmatrix} \begin{pmatrix} X \\ Y \end{pmatrix}$, which leads to ${}^tF = (A - {}^tBC^{-1}B)X$. Hence,

$$\det(LL_0^*) = \det(F^{t}F) = \det(A - {}^{t}BC^{-1}B) \det(FX) = \det(A - {}^{t}BC^{-1}B) \det(LL_g^*),$$

and $\gamma_r(x, G) \operatorname{Jac}^0(L) = \det(C)^{1/2} \det(A - {}^{\mathsf{t}}BC^{-1}B)^{1/2} \operatorname{Jac}(L)$. Since $A - {}^{\mathsf{t}}BC^{-1}B$ is the Schur complement of *C* in the matrix of g(x), we have $\det(C)^{1/2} \det(A - {}^{\mathsf{t}}BC^{-1}B)^{1/2} = \gamma(x)$.

6.3. *Factorial moment measures and Kac–Rice densities.* As in the previous section, we consider a nondegenerate C^1 centered Gaussian field $s : M \to E$ which is a random section of some vector bundle $E \to M$. Recall that $n = \dim(M)$ and $r \in [[1, n]]$ is the rank of E. We are interested in the finiteness of the moments of the linear statistics $\langle v, \phi \rangle$ with $\phi \in L_c^{\infty}(M)$; see Definition 6.11. In this section, we introduce the factorial moment measures and Kac–Rice densities of the fields s, and we relate them to higher moments of the linear statistics.

In the following, p will always denote the order of the moment we are considering. Recall that, under our hypothesis on s, the random measure v is almost surely a nonnegative Radon measure on M; see Remark 6.18.

Definition 6.20 (product measures). Let $p \ge 1$. We denote by $\nu^{\otimes p}$ the product measure of ν with itself *p*-times. We also denote by $\nu^{[p]}$ the restriction of $\nu^{\otimes p}$ to $M^p \setminus \Delta_p$. That is, for any test-function Φ ,

$$\langle v^{\otimes p}, \Phi \rangle = \int_{Z_{\text{reg}}^{p}} \Phi(\underline{x}) \, \mathrm{dVol}_{Z}^{\otimes p}(\underline{x}) \quad \text{and} \quad \langle v^{[p]}, \Phi \rangle = \int_{Z_{\text{reg}}^{p} \setminus \Delta_{p}} \Phi(\underline{x}) \, \mathrm{dVol}_{Z}^{\otimes p}(\underline{x})$$

Almost surely, these measures are Radon measures on M^p . More generally, if we want to consider product spaces indexed by a nonempty finite set A instead of $\llbracket 1, p \rrbracket$, we denote by $\nu^{\otimes A}$ the product measure of ν with itself |A| times on M^A and by $\nu^{[A]}$ its restriction to $M^A \setminus \Delta_A$.

The following lemma describes the relation between $\nu^{\otimes p}$ and $\nu^{[p]}$, using the notation introduced in Section 2.1.

Lemma 6.21 (relation between $\nu^{\otimes p}$ and $\nu^{[p]}$). Let $p \ge 1$. If r < n then $\nu^{\otimes p} = \nu^{[p]}$. If r = n then $\nu^{\otimes p} = \sum_{\mathcal{I} \in \mathcal{P}_p} (\iota_{\mathcal{I}})_* (\nu^{[\mathcal{I}]})$.

Proof. If r < n then Z_{reg} is a C^1 submanifold of positive dimension in M. In particular, the large diagonal in $(Z_{\text{reg}})^p$ has positive codimension, and hence is negligible for $d\text{Vol}_Z^{\otimes p}$. Thus $v^{\otimes p} = v^{[p]}$ in this case.

If r = n then Z_{reg} is a locally finite set and ν is its counting measure. Similarly, $\nu^{\otimes p}$ is the counting measure of the locally finite $(Z_{reg})^p$, and Δ_p is no longer negligible for this measure. If r = n = 1, we proved in [Ancona and Letendre 2021, Lemma 2.7] that $\nu^{\otimes p} = \sum_{\mathcal{I} \in \mathcal{P}_p} (\iota_{\mathcal{I}})_* (\nu^{[\mathcal{I}]})$. The proof is purely combinatorics and it extends immediately to the case $r = n \ge 1$.

Our interest in these measures is that, by the Fubini theorem, for all $\phi \in L_c^{\infty}(M)$ we have $\mathbb{E}[\langle v, \phi \rangle^p] = \mathbb{E}[\langle v^{\otimes p}, \phi^{\otimes p} \rangle] = \langle \mathbb{E}[v^{\otimes p}], \phi^{\otimes p} \rangle$, where $\phi^{\otimes p} : (x_1, \dots, x_p) \mapsto \phi(x_1) \cdots \phi(x_p)$. Thus, the measure $\mathbb{E}[v^{\otimes p}]$ is closely related with the computation of moments of linear statistics. For technical reasons, it is more convenient to consider $\mathbb{E}[v^{[p]}]$ instead.

Definition 6.22 (moment measures). Let $p \ge 1$. The measure $\mathbb{E}[v^{\otimes p}]$ is called the *p*-th moment measure of the field *s* and $\mathbb{E}[v^{[p]}]$ is called its *p*-th factorial moment measure.

Definition 6.23 (Kac–Rice density of order *p*). Let $p \ge 1$ and let us assume that the random vector $(s(x_1), \ldots, s(x_p))$ is nondegenerate for all $(x_1, \ldots, x_p) \in M^p \setminus \Delta_p$. Then we define

$$\rho_p: (x_1, \dots, x_p) \longmapsto \frac{\mathbb{E}\left[\prod_{i=1}^p \operatorname{Jac}(\nabla_{x_i} s) \mid \forall i \in \llbracket 1, p \rrbracket, s(x_i) = 0\right]}{\det(2\pi \operatorname{Var}(s(x_1), \dots, s(x_p)))^{1/2}}$$

from $M^p \setminus \Delta_p$ to $[0, +\infty)$, where the numerator is the conditional expectation of $\prod_{i=1}^p \operatorname{Jac}(\nabla_{x_i} s)$ given that $s(x_i) = 0$ for all $i \in [1, p]$.

Once again, ρ_p is well-defined and continuous on $M^p \setminus \Delta_p$ thanks to our nondegeneracy hypothesis. However, its expression is singular along Δ_p . In particular, ρ_p is in general not bounded, which raises the question of its local integrability near Δ_p . For example, if $f : \mathbb{R}^n \to \mathbb{R}$ is a nondegenerate enough stationary Gaussian field and p = 2, one can check that as $y \to x$, the corresponding Kac–Rice density $\rho_2(x, y)$ behaves like ||y - x|| if n = 1 and like 1/||y - x|| if n > 1.

Proposition 6.24 (Kac–Rice formula for the *p*-th factorial moment). Let s be a C^1 centered Gaussian field such that $(s(x_1), \ldots, s(x_p))$ is nondegenerate for all $(x_1, \ldots, x_p) \in M^p \setminus \Delta_p$. Then, for any Borel function $\Phi: M^p \to \mathbb{R}$ which is nonnegative or such that $\Phi \rho_p \in L^1(M^p)$, we have

$$\mathbb{E}[\langle v^{[p]}, \Phi \rangle] = \int_{M^p} \Phi(\underline{x}) \rho_p(\underline{x}) \,\mathrm{dVol}_M^{\otimes p}(\underline{x}).$$

i.e., $\mathbb{E}[v^{[p]}]$ is the measure on M^p with density ρ_p with respect to the Riemannian volume $dVol_M^{\otimes p}$.

Proof. Let us consider $S: (x_1, \ldots, x_p) \mapsto (s(x_1), \ldots, s(x_p))$ on $M^p \setminus \Delta_p$, which is a random section of the restriction over $M^p \setminus \Delta_p$ of the vector bundle $E^p \to M^p$. This is a nondegenerate \mathcal{C}^1 centered Gaussian field on $M^p \setminus \Delta_p$, and $\nu^{[p]}$ is the measure of integration over its zero set. Bearing in mind that Δ_p is negligible in M^p for $d\operatorname{Vol}_M^{\otimes p}$, the result follows from Proposition 6.17 applied to S. \square

The following proposition relates the properties of the Kac-Rice densities, the moment measures and the moments of linear statistics.

Proposition 6.25 (relation between moments, measures and densities). Let $p \ge 1$ and let s be a C^1 centered Gaussian field such that $(s(x_1), \ldots, s(x_p))$ is nondegenerate for all $(x_1, \ldots, x_p) \notin \Delta_p$. Then the following four properties are equivalent:

- (1) For all $\phi \in L^{\infty}_{c}(M)$, we have $\mathbb{E}[|\langle v, \phi \rangle|^{p}] < +\infty$.
- (2) For all $k \in [1, p]$, the moment measure $\mathbb{E}[v^{\otimes k}]$ is Radon on M^k , i.e., finite on compact sets.
- (3) For all $k \in [[1, p]]$, the factorial moment measure $\mathbb{E}[v^{[k]}]$ is Radon on M^k .
- (4) For all $k \in [[1, p]]$, the Kac–Rice density satisfies $\rho_k \in L^1_{loc}(M^k)$.

Proof. Let us assume (1). Let $k \in [[1, p]]$ and let $K_0 \subset M^k$ be compact. There exists a compact set $K \subset M$ such that $K_0 \subset K^k$. Then,

$$\langle \mathbb{E}[v^{\otimes k}], \mathbf{1}_{K_0} \rangle \leqslant \langle \mathbb{E}[v^{\otimes k}], \mathbf{1}_K^{\otimes k} \rangle = \mathbb{E}[\langle v, \mathbf{1}_K \rangle^k] = \mathbb{E}[|\langle v, \mathbf{1}_K \rangle|^k].$$

Since $\mathbf{1}_K \in L^{\infty}_c(M)$, the *p*-th absolute moment of $\langle v, \mathbf{1}_K \rangle$ is finite; hence so is its *k*-th absolute moment. Thus $\langle \mathbb{E}[\nu^{\otimes k}], \mathbf{1}_{K_0} \rangle < +\infty$ for all compact K_0 and (2) is satisfied.

If (2) is satisfied then so is (3). Indeed, for any $k \in [[1, p]]$, the measure $\nu^{[k]}$ is the restriction of $\nu^{\otimes k}$ to $M^k \setminus \Delta_k$. Thus, for any compact $K \subset M^k$ we have

$$\langle \mathbb{E}[\nu^{[k]}], \mathbf{1}_K \rangle = \mathbb{E}[\langle \nu^{[k]}, \mathbf{1}_K \rangle] \leqslant \mathbb{E}[\langle \nu^{\otimes k}, \mathbf{1}_K \rangle] = \langle \mathbb{E}[\nu^{\otimes k}], \mathbf{1}_K \rangle < +\infty.$$

If (3) is satisfied, let $k \in [1, p]$ and let $K \subset M^k$ be a compact. By Proposition 6.24 we have

$$\int_{K} \rho_{k}(\underline{x}) \, \mathrm{dVol}_{M}^{\otimes p}(\underline{x}) = \int_{M} \mathbf{1}_{K}(\underline{x}) \rho_{k}(\underline{x}) \, \mathrm{dVol}_{M}^{\otimes p}(\underline{x}) = \mathbb{E}[\langle \nu^{[k]}, \mathbf{1}_{K} \rangle] = \langle \mathbb{E}[\nu^{[k]}], \mathbf{1}_{K} \rangle < +\infty.$$

Thus ρ_k is integrable on any compact set, that is, $\rho_k \in L^1_{loc}(M^k)$. This proves (4) in this case.

Finally, let us assume that (4) holds. Let $\phi \in L_c^{\infty}(M)$, and let us denote by $K \subset M$ its compact support. We have $\mathbb{E}[|\langle v, \phi \rangle|^p] \leq \mathbb{E}[\langle v, |\phi| \rangle^p] \leq ||\phi||_{\infty}^p \mathbb{E}[\langle v, \mathbf{1}_K \rangle^p]$, so it is enough to prove that $\mathbb{E}[\langle v, \mathbf{1}_K \rangle^p] =$

 $\mathbb{E}[\langle v^{\otimes p}, \mathbf{1}_{K^p} \rangle]$ is finite. By Lemma 6.21, whether r = n or not, we have

$$\mathbb{E}[\langle \nu^{\otimes p}, \mathbf{1}_{K^p} \rangle] \leq \sum_{\mathcal{I} \in \mathcal{P}_p} \mathbb{E}[\langle \nu^{[\mathcal{I}]}, \mathbf{1}_{K^p} \circ \iota_{\mathcal{I}} \rangle] = \sum_{\mathcal{I} \in \mathcal{P}_p} \langle \mathbb{E}[\nu^{[\mathcal{I}]}], \mathbf{1}_{K^{\mathcal{I}}} \rangle.$$

Then, the Kac–Rice formula for moments and the local integrability of the $(\rho_k)_{1 \leq k \leq p}$ yields

$$\mathbb{E}[\langle \nu, \mathbf{1}_K \rangle^p] \leqslant \sum_{\mathcal{I} \in \mathcal{P}_p} \langle \mathbb{E}[\nu^{[\mathcal{I}]}], \mathbf{1}_{K^{\mathcal{I}}} \rangle = \sum_{\mathcal{I} \in \mathcal{P}_p} \int_{K^{|\mathcal{I}|}} \rho_{|\mathcal{I}|}(\underline{x}) \, \mathrm{dVol}_M^{\otimes |\mathcal{I}|}(\underline{x}) < +\infty,$$

which proves (1) and concludes the proof.

6.4. *Proofs of Theorems 1.6 and 1.9: finiteness of moments.* The goal of this section is to prove Theorems 1.6 and 1.9, which give simple conditions for the finiteness of the moments of the linear statistics of a Gaussian field. We begin by proving a local version of Theorem 1.6, under a nondegeneracy hypothesis for the multijets of the field. This is Theorem 6.26 below. Then we deduce Theorem 1.9 from Theorem 6.26, in the case of Gaussian fields with value in a vector bundle. Finally, Theorem 1.6 is obtained as a special case of Theorem 1.9.

Let $\Omega \subset \mathbb{R}^n$ be open. Recall that $\mathcal{MJ}_p(\Omega, \mathbb{R}^r) \to C_p[\Omega]$ is defined in Definition 1.3 as the restriction over $C_p[\Omega] \subset C_p[\mathbb{R}^n]$ of the vector bundle $\mathcal{MJ}_p(\mathbb{R}^n, \mathbb{R}^r) \to C_p[\mathbb{R}^n]$ from Theorem 1.1.

Theorem 6.26 (finiteness of moments, local version). Let $f : \Omega \to \mathbb{R}^r$ be a centered Gaussian field and v be as in Definition 6.11. Let $p \ge 1$. If f is C^p and for all $k \in \llbracket 1, p \rrbracket$ the Gaussian field $\operatorname{mj}_k(f, \cdot) : C_k[\Omega] \to \mathcal{MJ}_k(\Omega, \mathbb{R}^r)$ is nondegenerate, then the four equivalent statements in Proposition 6.25 hold.

Proof. Let $f : \Omega \to \mathbb{R}^r$ be a \mathcal{C}^p centered Gaussian field such that $mj_k(f, \cdot) : C_k[\Omega] \to \mathcal{MJ}_k(\Omega, \mathbb{R}^r)$ is nondegenerate for all $k \in [[1, p]]$.

<u>Step 1</u>: Gaussianity and nondegeneracy of the multijets. Since f is C^p , for all $k \in \llbracket 1, p \rrbracket$ we have $mj_k(f, \cdot) \in \Gamma^1(C_k[\Omega], \mathcal{MJ}_k(\Omega, \mathbb{R}^r))$ because of (2) in Theorem 1.1. Since f is centered and Gaussian, so is any finite collection of jets of f. Then, for all $m \ge 1$ and all $z_1, \ldots, z_m \in C_k[\Omega]$ we have that $(mj_k(f, z_1), \ldots, mj_k(f, z_m))$ is a centered Gaussian. Indeed, by (4) in Theorem 1.1, this is the image of a centered Gaussian by a linear map. Thus, $mj_k(f, \cdot)$ is a nondegenerate C^1 centered Gaussian field on $C_k[\Omega]$ with values in $\mathcal{MJ}_k(\Omega, \mathbb{R}^r)$.

Let $z \notin \pi^{-1}(\Delta_p)$ and let $\underline{x} = (x_1, \dots, x_p) = \pi(z)$. By (4) in Theorem 1.1, the map Θ_z is a linear surjection. A dimension argument shows that it is actually a bijection. Thus

$$(f(x_1), \ldots, f(x_p)) = (j_0(f, x_1), \ldots, j_0(f, x_p)) = \Theta_z^{-1}(mj_p(f, z)),$$

which proves that $(f(x_1), \ldots, f(x_p))$ is nondegenerate. Thus, the hypotheses of Proposition 6.25 are satisfied, and the four statements appearing in this proposition are indeed equivalent.

<u>Step 2</u>: Comparing zeros of f and $mj_k(f, \cdot)$. Let $k \in [[1, p]]$. In the following we are going to prove that $\mathbb{E}[\nu^{[k]}]$ is a Radon measure on Ω^k , which is enough to conclude the proof. In the following, we say that a subset of $C_k[\Omega]$ (resp. Ω^k) is negligible if its k(n-r)-dimensional Hausdorff measure is 0.

1462

Let us consider the Gaussian field $mj_k(f, \cdot) : C_k[\Omega] \to \mathcal{MJ}_k(\Omega, \mathbb{R}^r)$. We have checked above that it satisfies the hypotheses of Proposition 6.8. Let $X \subset C_k[\Omega]$ denote the zero set of $mj_k(f, \cdot)$. As in Section 6.2, we define $X_{sing} = \{z \in C_k[\Omega] \mid mj_k(f, x) = 0 \text{ and } Jac(\nabla_z mj_k(f, \cdot)) = 0\}$ and $X_{reg} = X \setminus X_{sing}$. Recall that X_{reg} is a \mathcal{C}^1 submanifold of codimension kr and that X_{sing} is almost surely negligible by Proposition 6.8. Let $Y = X \cap \pi^{-1}(\Omega^k \setminus \Delta_k)$. We also let $Y_{sing} = Y \cap X_{sing}$ and $Y_{reg} = Y \cap X_{reg}$.

Recalling that $Z = f^{-1}(0) \subset \Omega$, for all $z \in C_k[\Omega]$ we have $z \in Y$ if and only if $\pi(z) \in Z^k \setminus \Delta_k$; see (3) in Theorem 1.1. By (1) in the same theorem, the restriction of π to $\pi^{-1}(\Omega^k \setminus \Delta_k)$ is a diffeomorphism. Hence $\pi(Y) = Z^k \setminus \Delta_k$, the set $\pi(Y_{\text{reg}})$ is a C^1 submanifold of $\Omega^k \setminus \Delta_k$, and $\pi(Y_{\text{sing}})$ is almost surely negligible. Since $Z^k \setminus (Z_{\text{reg}})^k$ is also almost surely negligible, the submanifolds $Z_{\text{reg}}^k \setminus \Delta_k$ and $\pi(Y_{\text{reg}})$ are almost surely the same, up to a negligible set (actually the reader can check that $\pi(Y_{\text{reg}}) = Z_{\text{reg}}^k \setminus \Delta_k$ using the trivialization τ introduced at the end of Section 5.2). Recalling Definition 6.20, this shows that $\nu^{[k]}$ is the same as the integral over $\pi(Y_{\text{reg}})$ with respect to the Riemannian volume $d\text{Vol}_{\pi(Y)}$ induced by the Euclidean metric on $(\mathbb{R}^n)^k$. At this stage we know that, almost surely,

$$\forall \Phi \in L^{\infty}_{c}(\Omega^{k}), \quad \langle \nu^{[k]}, \Phi \rangle = \int_{\pi(Y_{\text{reg}})} \Phi(\underline{x}) \, \mathrm{dVol}_{\pi(Y)}(\underline{x}). \tag{6-2}$$

<u>Step 3</u>: Comparing volumes. Let us introduce a Riemannian metric g on $C_k[\Omega]$. It induces a volume measure $dVol_X$ on X_{reg} , and hence on Y_{reg} . Additionally, let $z \in C_k[\Omega]$ and let $G \subset T_z(C_k[\Omega])$ be a vector subspace of codimension kr. We define $J(G) = det((D_z\pi_{|G})^*D_z\pi_{|G})^{1/2}$, where the adjoint of $D_z\pi_{|G}$ is computed with respect to g on G and to the Euclidean metric on $(\mathbb{R}^n)^k$. Since π is smooth, this defines a smooth nonnegative function J on the total space of the Grassmannian bundle $Gr_{kr}(T(C_k[\Omega])) \to C_k[\Omega]$ of subspaces of codimension kr in the tangent of $C_k[\Omega]$. Our interest in this map is that if $z \in Y_{reg}$ and $G = T_z Y_{reg}$ then J(G) is the Jacobian determinant of $D_z(\pi_Y)$, where the \mathcal{C}^∞ -diffeomorphism $\pi_Y : Y_{reg} \to \pi(Y_{reg})$ is the restriction of π on both sides.

Let $K \subset \Omega^k$ be compact and let us apply (6-2) to $\mathbf{1}_K$. Using the previous notation, the change of variables π_Y yields

$$\langle v^{[k]}, \mathbf{1}_K \rangle = \int_{Y_{\text{reg}}} \mathbf{1}_K(\pi(z)) J(T_z Y_{\text{reg}}) \, \mathrm{dVol}_X(z) \leqslant \int_{X_{\text{reg}}} \mathbf{1}_{\pi^{-1}(K)}(z) J(T_z X_{\text{reg}}) \, \mathrm{dVol}_X(z).$$

By (1) in Theorem 1.1, the projection π is proper; hence $\widetilde{K} = \pi^{-1}(K)$ is compact. Since the bundle $\operatorname{Gr}_{kr}(T(C_k[\Omega])) \to C_k[\Omega]$ has compact fiber, its restriction over $\widetilde{K} \subset C_k[\Omega]$ is compact. By continuity, the function *J* is bounded on this compact set by some constant C_K . Finally, we have proved that, almost surely,

$$\langle v^{[k]}, \mathbf{1}_K \rangle \leqslant C_K \langle \tilde{v}, \mathbf{1}_{\widetilde{K}} \rangle,$$

where $\tilde{\nu}$ is defined by integrating over X_{reg} with respect to $d\text{Vol}_X$. Taking expectation on both sides we get $\langle \mathbb{E}[\nu^{[k]}], \mathbf{1}_K \rangle \leq C_K \langle \mathbb{E}[\tilde{\nu}], \mathbf{1}_{\widetilde{K}} \rangle$.

<u>Step 4</u>: Applying the Kac–Rice formula to multijets. Now, X_{reg} is the regular part of the zero set X of the Gaussian field $\text{mj}_k(f, \cdot)$. We have checked at the beginning of the proof that $\text{mj}_k(f, \cdot)$ satisfies the hypotheses of Proposition 6.17. This proposition yields that $\mathbb{E}[\tilde{\nu}]$ is a Radon measure on $C_k[\Omega]$. Hence $\langle \mathbb{E}[\tilde{\nu}], \mathbf{1}_{\tilde{K}} \rangle$ is finite, and so is $\langle \mathbb{E}[\nu^{[k]}], \mathbf{1}_{K} \rangle$. Thus $\mathbb{E}[\nu^{[k]}]$ is Radon on Ω^k , which concludes the proof. \Box

We can now prove Theorem 1.9, which gives a criterion for the finiteness of the *p*-th moments of the linear statistics associated with a centered Gaussian field $s: M \to E$, where $E \to M$ is some vector bundle of rank *r* over a Riemannian manifold (M, g) without boundary of dimension $n \ge r$. The idea of the proof is to patch together the local results obtained by applying Theorem 6.26 in nice local trivializations.

Proof of Theorem 1.9. Let $p \ge 1$ and let $s \in \Gamma^p(M, E)$ be a centered Gaussian field which is C^p and (p-1)-nondegenerate.

<u>Step 1</u>: Existence of nice local trivializations. Let $x_0 \in M$. There exists an open neighborhood U of x_0 and a local trivialization of E over U given by Lemma 6.4. In this trivialization, the Gaussian section s corresponds to a centered Gaussian field $f : \Omega \to \mathbb{R}^r$ which is \mathcal{C}^p and (p-1)-nondegenerate. We denote by $x \in \Omega$ the image of x_0 in local coordinates.

Let $k \in \llbracket 1, p \rrbracket$ and let $\underline{x} = (x, ..., x) \in \Omega^k$. Since $j_{p-1}(f, x)$ is nondegenerate so is $j_{k-1}(f, x)$; see Definition 1.4. Then, by (4) in Theorem 1.1, for all $z \in \pi^{-1}(\{\underline{x}\}) \subset C_k[\Omega]$ the Gaussian vector $mj_k(f, z) = \Theta_z(j_{k-1}(f, x)) \in \mathcal{MJ}_k(\Omega, \mathbb{R}^r)_z$ is nondegenerate. By (1) in Theorem 1.1, the map π is proper; hence $\pi^{-1}(\{\underline{x}\})$ is compact. One the other hand, since f is \mathcal{C}^p , we know $mj_k(f, \cdot)$ is at least \mathcal{C}^1 . Thus $z \mapsto \det Var(mj_k(f, z))$ is continuous on $C_k[\Omega]$ and positive on the compact $\pi^{-1}(\{\underline{x}\})$, and hence on some neighborhood V_k of $\pi^{-1}(\{\underline{x}\})$ in $C_k[\Omega]$.

Up to reducing V_k we can assume that $V_k = \pi^{-1}(W_k)$, where W_k is an open neighborhood of \underline{x} in Ω^k . Otherwise, there would exist a sequence $(z_n)_{n \in \mathbb{N}} \in C_k[\Omega] \setminus V_k$ such that $\pi(z_n) \xrightarrow[n \to +\infty]{} \underline{x}$. By properness of π , up to extracting a subsequence, we could assume that $z_n \xrightarrow[n \to +\infty]{} \underline{x}$. By continuity $z \in \pi^{-1}({\underline{x}})$, which would be absurd. Since W_k is a neighborhood of \underline{x} in Ω^k , there exists an open neighborhood Υ_k of x in Ω such that $(\Upsilon_k)^k \subset W_k$.

Let us define $\Upsilon = \bigcap_{k=1}^{p} \Upsilon_k$, which is an open neighborhood of *x*. For all $k \in [[1, p]]$, we have $C_k[\Upsilon] = \pi^{-1}(\Upsilon^k) \subset \pi^{-1}((\Upsilon_k)^k) \subset \pi^{-1}(W_k) = V_k$ and $\operatorname{mj}_k(f, \cdot)$ is nondegenerate on $C_k[\Upsilon]$. Thus, up to replacing Ω by the smaller neighborhood Υ of *x* in Ω and replacing *U* by the corresponding neighborhood of x_0 on *M*, we can assume that the local trivialization given by Lemma 6.4 is such that, for all $k \in [[1, p]]$, the Gaussian field $\operatorname{mj}_k(f, \cdot) : C_k[\Omega] \to \mathcal{MJ}_k(\Omega, \mathbb{R}^r)$ is nondegenerate.

<u>Step 2</u>: Reduction to the local case. Let $\phi \in L_c^{\infty}(M)$ and let *K* denote its support. By compactness, there exists a finite family $(U_i)_{i=1}^m$ of open subsets such that $K \subset \bigcup_{i=1}^m U_i$ and each U_i is the domain of nice trivialization of the type we built in the previous paragraph. Letting $U_0 = M \setminus K$, there exists a smooth partition of unity $(\chi_i)_{i=0}^m$ subordinated to the open covering $(U_i)_{i=0}^m$ of *M*. Then $\phi = \sum_{i=1}^m \chi_i \phi$ by the definition of *K*.

Recall that ν is the measure from Definition 6.11. We have $|\langle \nu, \phi \rangle| \leq \langle \nu, |\phi| \rangle = \sum_{i=1}^{m} \langle \nu, \phi_i \rangle$, where $\phi_i = \chi_i |\phi|$ for all $i \in [[1, m]]$. Let $p \ge 1$, by Hölder's inequality we get

$$\mathbb{E}[|\langle \nu, \phi \rangle|^{p}] \leq \mathbb{E}\left[\left(\sum_{i=1}^{m} \langle \nu, \phi_{i} \rangle\right)^{p}\right] = \sum_{1 \leq i_{1}, \dots, i_{p} \leq m} \mathbb{E}\left[\prod_{j=1}^{p} \langle \nu, \phi_{i_{j}} \rangle\right] \leq \sum_{1 \leq i_{1}, \dots, i_{p} \leq m} \prod_{j=1}^{p} \mathbb{E}[\langle \nu, \phi_{i_{j}} \rangle^{p}]^{1/p} \\ \leq m^{p} \max_{1 \leq i \leq m} \mathbb{E}[\langle \nu, \phi_{i} \rangle^{p}].$$

Thus, in order to prove Theorem 1.9, it is enough to prove that $\mathbb{E}[\langle \nu, \phi \rangle^p] < +\infty$ for any nonnegative $\phi \in L_c^{\infty}(M)$ whose support is included in the domain of a nice trivialization.

<u>Step 3</u>: Local case. Let $U \subset M$ be an open subset over which we have a nice trivialization of E and s. That is, U is as in Lemma 6.4, the section s reads as $f : \Omega \to \mathbb{R}^r$ in local coordinates, and in addition we can assume that for all $k \in \llbracket 1, p \rrbracket$ the field $\operatorname{mj}_k(f, \cdot)$ is nondegenerate on $C_k[\Omega]$. Identifying objects on U with their image in the local trivialization, we reduced our problem to proving that $\mathbb{E}[\langle v, \phi \rangle^p] < +\infty$ for all nonnegative $\phi \in L_c^{\infty}(\Omega)$. Note that v is the measure of integration over Z_{reg} with respect to the Riemannian volume measure $d\operatorname{Vol}_Z$ induced by the metric g. In order to apply Theorem 6.26, we need to compare v with \tilde{v} , which is the measure of integration over Z_{reg} with respect to the Euclidean volume measure $d\operatorname{Vol}_Z^0$.

Let $\phi \in L_c^{\infty}(\Omega)$ be nonnegative and let *K* denote its compact support. Recalling Definition 6.13, Lemma 6.14 shows that

$$\langle v, \phi \rangle = \int_{Z_{\text{reg}}} \phi(x) \, \mathrm{dVol}_Z(x) = \int_{Z_{\text{reg}} \cap K} \phi(x) \gamma_r(x, \ker D_x f) \, \mathrm{dVol}_Z^0(x).$$

Since γ_r is continuous and $K \times \operatorname{Gr}_r(\mathbb{R}^n)$ is compact, the nonnegative function γ_r is bounded by some $C_K > 0$ on this set. Thus $\langle \nu, \phi \rangle \leq C_K \langle \tilde{\nu}, \phi \rangle$. Since $f : \Omega \to \mathbb{R}^r$ satisfies the hypotheses of Theorem 6.26 and $\tilde{\nu}$ is the measure of integration over its zero set induced by the Euclidean metric, we have $\mathbb{E}[\langle \nu, \phi \rangle^p] \leq C_K^p \mathbb{E}[\langle \tilde{\nu}, \phi \rangle^p] < +\infty$.

We conclude this section with the proof of Theorem 1.6, which is a corollary of Theorem 1.9.

Proof of Theorem 1.6. Let $f : \Omega \to \mathbb{R}^r$ be a centered Gaussian field which is \mathcal{C}^p and (p-1)-nondegenerate in the sense of Definition 1.4. Then $s = (f, \operatorname{Id})$ is a random section of the trivial bundle $\mathbb{R}^r \times \Omega \to \Omega$. This *s* is also a \mathcal{C}^p and (p-1)-nondegenerate centered Gaussian field. Its vanishing locus (as a section) is the same as the vanishing locus of *f*. Hence, the result follows from applying Theorem 1.9 to *s*.

7. Multijets adapted to a differential operator

In Theorem 1.1 we defined multijets such that, over the configuration space $(\mathbb{R}^n)^p \setminus \Delta_p \subset C_p[\mathbb{R}^n]$, the *p*-multijet $\operatorname{mj}_p(f, \underline{x})$ reads as $(f(x_1), \ldots, f(x_p))$ in the natural trivialization τ (see the end of Section 5.2). Thus $\operatorname{mj}_p(f, \underline{x})$ is a way to patch together the 0-jets of f at x_i into a smooth object that does not degenerate along Δ_p . In this section, we explain how a similar construction allows us to build a multijet that patches together the *k*-jets of f at x_i , and more generally the values at x_i of $\mathcal{D}f$, where \mathcal{D} is a differential operator. In Section 7.1 we recall the definition of a differential operator. Then we define a multijet adapted to a given differential operator in Section 7.2. Finally, in Section 7.3, we prove Theorem 1.10.

7.1. *Differential operator.* In this section, we recall a few fact about differential operators. In the following, we use the multi-index notation introduced in Section 2.2.

Definition 7.1 (differential operator). Let $\Omega \subset \mathbb{R}^n$ be open, let $q, r \ge 1$ and let $d \ge 0$. We say a *differential* operator of order at most d is a linear map $\mathcal{D} : C^d(\Omega, \mathbb{R}^q) \to C^0(\Omega, \mathbb{R}^r)$ such that there exist continuous

functions $(a_{ij\alpha})_{1 \leq i \leq r; 1 \leq j \leq q; |\alpha| \leq d}$ on Ω such that, for all $f = (f_1, \ldots, f_q) \in \mathcal{C}^d(\Omega, \mathbb{R}^q)$,

$$\mathcal{D}(f): x \longmapsto \left(\sum_{j=1}^{q} \sum_{|\alpha| \leqslant d} a_{ij\alpha}(x) \partial^{\alpha} f_{j}(x)\right)_{1 \leqslant i \leqslant r}.$$
(7-1)

More generally, let *M* be a manifold of dimension *n* and let $E \to M$ and $F \to M$ be two vector bundles of ranks *q* and *r* respectively. We say that $\mathcal{D}: \Gamma^d(M, E) \to \Gamma^0(M, F)$ is a *differential operator of order at most d* if around any point $x \in M$ there exist a chart and local trivializations of *E* and *F* such that \mathcal{D} is of the form (7-1) in the corresponding local coordinates. We say that \mathcal{D} is of *order* $d \in \mathbb{N}$ if it is of order at most *d* and not of order at most d - 1. If $s \in \Gamma^d(M, E)$ and $x \in M$, we write $\mathcal{D}s = \mathcal{D}(s)$ and $\mathcal{D}_x s = \mathcal{D}(s)(x)$ for simplicity.

Remark 7.2. Let us make some important comments.

• If $\mathcal{D}: \Gamma^d(M, E) \to \Gamma^0(M, F)$ is a differential operator of order at most *d*, then it is of the form (7-1) in any set of local coordinates on *M*, *E* and *F*.

• An equivalent definition of a differential operator of order at most *d* is that it factors linearly through the bundle of *d*-jets. That is, there exists $L \in \Gamma^0(M, \mathcal{J}_d(M, E)^* \otimes F)$ such that $\mathcal{D}_x s = L(x) j_d(s, x) \in F_x$ for all $s \in \Gamma^d(M, E)$ and $x \in M$.

In the following we always assume that M, E, F and L are smooth. In particular, the functions $(a_{ij\alpha})$ appearing in the local expression (7-1) of \mathcal{D} are smooth. This implies that if $s \in \Gamma^{d+l}(M, E)$ then $\mathcal{D}s \in \Gamma^{l}(M, F)$.

Example 7.3. The main examples we have in mind are the following.

• The differential $D: \mathcal{C}^1(M) \to \Gamma^0(M, T^*M)$ is a differential operator of order 1.

• For all $k \in \mathbb{N}$, the jet map $j_k : \Gamma^k(M, E) \to \Gamma^0(M, \mathcal{J}_k(M, E))$ is a differential operator of order k corresponding to L(x) being the identity of $\mathcal{J}_k(M, E)_x$ for all $x \in M$.

• If *M* is equipped with a Riemannian metric, the Laplace–Beltrami operator Δ acting on $C^2(M)$ is a differential operator of order 2.

• If ∇ is a connection on $E \to M$ then $\nabla : \Gamma^1(M, E) \to \Gamma^0(M, T^*M \otimes E)$ is a differential operator of order 1. Indeed, in a local frame (e_1, \ldots, e_q) of *E* and local coordinates (x_1, \ldots, x_n) on *M* the covariant derivative of $s = \sum_{j=1}^q f_j e_j \in \Gamma^1(M, E)$ at *x* is given by

$$\nabla_x s = \sum_{i=1}^n \sum_{j=1}^q \left(\partial_i f_j(x) + \sum_{k=1}^q \mu_{ijk}(x) f_k(x) \right) dx_i \otimes e_j(x),$$

where the (μ_{ijk}) are defined by the relations $\nabla e_k = \sum_{i=1}^n \sum_{j=1}^q \mu_{ijk} dx_i \otimes e_j$ for all $k \in [[1, q]]$.

7.2. *Multijets adapted to* \mathcal{D} . The purpose of this section is to explain how to modify the construction of Section 5 in order to define a multijet bundle adapted to a given differential operator.

Let n, q and $r \ge 1$. We consider a differential operator $\mathcal{D} : \mathcal{C}^d(\mathbb{R}^n, \mathbb{R}^q) \to \mathcal{C}^0(\mathbb{R}^n, \mathbb{R}^r)$ of order d. As in Remark 7.2, there exists a section L of $\mathcal{J}_d(\mathbb{R}^n, \mathbb{R}^q)^* \otimes \mathbb{R}^r$ such that for any $f \in \mathcal{C}^d(\mathbb{R}^n, \mathbb{R}^q)$ and $x \in \mathbb{R}^n$

1466

we have $\mathcal{D}_x f = L(x) j_d(f, x)$. We assume that for all $x \in \mathbb{R}^n$ the linear map $\mathcal{D}_x : \mathcal{C}^d(\mathbb{R}^n, \mathbb{R}^q) \to \mathbb{R}^r$ is surjective, which is equivalent to $L(x) : \mathcal{J}_d(\mathbb{R}^n, \mathbb{R}^q)_x \to \mathbb{R}^r$ being surjective. Moreover, we assume that L is smooth. In this context, we have the following analogue of Theorem 1.1. It holds in particular if $\mathcal{D} = j_k$ or $\mathcal{D} = D$ is the differential.

Theorem 7.4 (existence of multijets adapted to \mathcal{D}). Let $\mathcal{D} : \mathcal{C}^d(\mathbb{R}^n, \mathbb{R}^q) \to \mathcal{C}^0(\mathbb{R}^n, \mathbb{R}^r)$ be a differential operator of order d as above. Let $p \ge 1$. There exist a smooth manifold $C_p^{\mathcal{D}}[\mathbb{R}^n]$ of dimension np without boundary and a smooth vector bundle $\mathcal{MJ}_p^{\mathcal{D}}(\mathbb{R}^n) \to C_p^{\mathcal{D}}[\mathbb{R}^n]$ of rank rp with the following properties:

(1) There exists a smooth proper surjection $\pi : C_p^{\mathcal{D}}[\mathbb{R}^n] \to (\mathbb{R}^n)^p$ such that $\pi^{-1}((\mathbb{R}^n)^p \setminus \Delta_p)$ is a dense open subset of $C_p^{\mathcal{D}}[\mathbb{R}^n]$, and π restricted to $\pi^{-1}((\mathbb{R}^n)^p \setminus \Delta_p)$ is a \mathcal{C}^{∞} -diffeomorphism onto $(\mathbb{R}^n)^p \setminus \Delta_p$.

(2) There exists a map $\operatorname{mj}_p^{\mathcal{D}} : \mathcal{C}^{(d+1)p-1}(\mathbb{R}^n, \mathbb{R}^r) \times C_p^{\mathcal{D}}[\mathbb{R}^n] \to \mathcal{MJ}_p^{\mathcal{D}}(\mathbb{R}^n)$ such that

- for all $z \in C_p^{\mathcal{D}}[\mathbb{R}^n]$, the map $\mathrm{mj}_p^{\mathcal{D}}(\cdot, z) : \mathcal{C}^{(d+1)p-1}(\mathbb{R}^n, \mathbb{R}^r) \to \mathcal{MJ}_p^{\mathcal{D}}(\mathbb{R}^n)_z$ is surjective;
- for all $f \in \mathcal{C}^{l+(d+1)p-1}(\mathbb{R}^n, \mathbb{R}^r)$, the section $\operatorname{mj}_p^{\mathcal{D}}(f, \cdot)$ of $\mathcal{MJ}_p^{\mathcal{D}}(\mathbb{R}^n) \to C_p^{\mathcal{D}}[\mathbb{R}^n]$ is \mathcal{C}^l .
- (3) Let $z \in C_p^{\mathcal{D}}[\mathbb{R}^n]$ be such that $\pi(z) = (x_1, \ldots, x_p) \notin \Delta_p$. Then for all $f \in \mathcal{C}^{(d+1)p-1}(\mathbb{R}^n, \mathbb{R}^q)$

$$\operatorname{mj}_p^{\mathcal{D}}(f,z) = 0 \quad \Longleftrightarrow \quad \forall i \in \llbracket 1, p \rrbracket, \ \mathcal{D}_{x_i} f = 0.$$

(4) Let $z \in C_p^{\mathcal{D}}[\mathbb{R}^n]$, let $\mathcal{I} = \mathcal{I}(\pi(z))$ and let $(y_I)_{I \in \mathcal{I}} = \iota_{\mathcal{I}}^{-1}(\pi(z)) \in (\mathbb{R}^n)^{\mathcal{I}} \setminus \Delta_{\mathcal{I}}$. There exists a linear surjection $\Theta_z^{\mathcal{D}} : \prod_{I \in \mathcal{I}} \mathcal{J}_{(d+1)|I|-1}(\mathbb{R}^n, \mathbb{R}^q)_{y_I} \to \mathcal{MJ}_p^{\mathcal{D}}(\mathbb{R}^n)_z$ such that

$$\forall f \in \mathcal{C}^{(d+1)p-1}(\mathbb{R}^n, \mathbb{R}^r), \quad \mathrm{mj}_p^{\mathcal{D}}(f, z) = \Theta_z^{\mathcal{D}}((\mathbf{j}_{(d+1)|I|-1}(f, y_I))_{I \in \mathcal{I}})$$

Proof. The proof follows the same strategy as what we did in Sections 4 and 5 in order to prove Theorem 1.1. Let us sketch its main steps.

Let $\underline{x} = (x_1, \ldots, x_p) \in (\mathbb{R}^n)^p$ and let $\underline{\hat{x}} = (\underline{x}, \ldots, \underline{x}) \in (\mathbb{R}^n)^{(d+1)p}$. Let $f \in \mathcal{C}^{(d+1)p-1}(\mathbb{R}^n, \mathbb{R}^q)$. The polynomial map $K(f, \underline{\hat{x}}) \in \mathbb{R}_{(d+1)p-1}[X] \otimes \mathbb{R}^q$ is defined as in Definition 3.6. For all $i \in \llbracket 1, p \rrbracket$, since x_i appears with multiplicity d + 1 in $\underline{\hat{x}}$, the map $K(f, \underline{\hat{x}})$ has the same d-jet as f at x_i . Hence $\mathcal{D}_{x_i} f = L(x_i) j_d(f, x_i) = L(x_i) j_d(K(f, \underline{\hat{x}}), x_i) = \mathcal{D}_{x_i}(K(f, \underline{\hat{x}}))$.

If $\underline{x} \notin \Delta_p$, let us define $\operatorname{ev}_{\underline{x}}^{\mathcal{D}} : P \mapsto (\mathcal{D}_{x_i}P)_{1 \leq i \leq p}$ from $\mathbb{R}_{(d+1)p-1}[X] \otimes \mathbb{R}^q$ to $(\mathbb{R}^r)^p$. Since we assumed that $L(x_i)$ is surjective for all $i \in [[1, p]]$, the previous interpolation result proves that $\operatorname{ev}_{\underline{x}}^{\mathcal{D}}$ is surjective. Then, as in (4-2), for all nonempty $I \subset [[1, p]]$ we define

$$\mathcal{G}_{I}^{\mathcal{D}}(\underline{x}) = \ker \operatorname{ev}_{\underline{x}_{I}}^{\mathcal{D}} \in \operatorname{Gr}_{r|I|}(\mathbb{R}_{(d+1)|I|-1}[X] \otimes \mathbb{R}^{q}).$$

We also define $\mathcal{G}^{\mathcal{D}}(\underline{x}) = \mathcal{G}^{\mathcal{D}}_{\llbracket 1, p \rrbracket}(\underline{x}).$

Following the same strategy as in Section 5, we denote by $\Sigma_{\mathcal{D}}$ the graph of $(\mathcal{G}_I^{\mathcal{D}})_{I \subset [\![1,p]\!]}$ defined on $(\mathbb{R}^n)^p \setminus \Delta_p$. We define $C_p^{\mathcal{D}}[\mathbb{R}^n]$ as a resolution of the singularities of the algebraic variety

$$\overline{\Sigma}_{\mathcal{D}} \subset (\mathbb{R}^n)^p \times \prod_{\varnothing \neq I \subset \llbracket 1, p \rrbracket} \operatorname{Gr}_{r|I|}(\mathbb{R}_{(d+1)|I|-1}[X] \otimes \mathbb{R}^q).$$

The manifold $C_p^{\mathcal{D}}[\mathbb{R}^n]$ satisfies the analogue of Corollary 5.6. In particular the maps $\mathcal{G}_I^{\mathcal{D}}$ with $I \subset \llbracket 1, p \rrbracket$ extend smoothly to $C_p^{\mathcal{D}}[\mathbb{R}^n]$. Then we define the *p*-multijet bundle adapted to \mathcal{D} as

$$\mathcal{MJ}_p^{\mathcal{D}}(\mathbb{R}^n) = \left((\mathbb{R}_{(d+1)p-1}[X] \otimes \mathbb{R}^q) \times C_p^{\mathcal{D}}[\mathbb{R}^n] \right) / \mathcal{G}^{\mathcal{I}}$$

over $C_p^{\mathcal{D}}[\mathbb{R}^n]$, similarly to Definition 5.7. Given a function $f \in \mathcal{C}^{(d+1)p-1}(\mathbb{R}^n, \mathbb{R}^q)$, we define its *p*-multijet adapted to \mathcal{D} at $z \in C_p^{\mathcal{D}}[\mathbb{R}^n]$ as

$$\operatorname{mj}_p^{\mathcal{D}}(f, z) = K(f, \widehat{\pi(z)}) \operatorname{mod} \mathcal{G}^{\mathcal{D}}(z).$$

Then, following the same steps as in Section 5, one can check that the objects we just defined satisfy the conditions in Theorem 7.4. \Box

As before, thanks to the localness condition in Theorem 7.4(4), the multijet $\text{mj}_p^{\mathcal{D}}(f, z)$ makes sense even if f is only defined and $\mathcal{C}^{(d+1)|I|-1}$ near y_I for all $I \in \mathcal{I}(\pi(z))$. Hence, the following definition makes sense.

Definition 7.5 (multijets adapted to \mathcal{D}). Let $\Omega \subset \mathbb{R}^n$ be open. We define $C_p^{\mathcal{D}}[\Omega] = \pi^{-1}(\Omega^p)$ and denote by $\mathcal{MJ}_p^{\mathcal{D}}(\Omega)$ the restriction of $\mathcal{MJ}_p^{\mathcal{D}}(\mathbb{R}^n)$ to $C_p^{\mathcal{D}}[\Omega]$. We call $\mathcal{MJ}_p^{\mathcal{D}}(\Omega) \to C_p^{\mathcal{D}}[\Omega]$ the *bundle of p-multijets adapted to* \mathcal{D} . Let $f : \Omega \to \mathbb{R}^q$ be of class $\mathcal{C}^{(d+1)p-1}$. We call the section $\operatorname{mj}_p^{\mathcal{D}}(f, \cdot)$ of $\mathcal{MJ}_p^{\mathcal{D}}(\Omega)$ the *p-multijet of f adapted to* \mathcal{D} .

7.3. *Finiteness of moments for critical points.* The purpose of this section is to prove Theorem 1.10. More generally we prove an analogous result for the zero set of Ds, where *s* is a section of a vector bundle $E \rightarrow M$ and D is a differential operator; see Theorem 7.8. This is done by adapting what we did in Section 6 to this framework.

Let (M, g) be Riemannian manifold of dimension $n \ge 1$ without boundary. Let $E \to M$ (resp. $F \to M$) be a smooth vector bundle of rank $q \ge 1$ (resp. $r \in [\![1, n]\!]$). We consider a differential operator \mathcal{D} : $\Gamma^d(M, E) \to \Gamma^0(M, F)$ of order $d \ge 0$, corresponding to a smooth section $L \in \Gamma^{\infty}(M, \mathcal{J}_d(M, E)^* \otimes F)$; see Remark 7.2. Thanks to this smoothness assumption we have $\mathcal{D} : \Gamma^{d+l}(M, E) \to \Gamma^l(M, F)$ for all $l \ge 0$. Finally we assume that $L(x) : \mathcal{J}_d(M, E)_x \to F_x$ (or equivalently $\mathcal{D}_x : \Gamma^d(M, E) \to F_x$) is surjective for all $x \in M$.

Let $s : M \to E$ be a centered Gaussian field on M with values in E in the sense of Definition 6.3. We assume that s is C^{d+1} and d-nondegenerate, so that $j_d(s, \cdot)$ is a centered Gaussian field with values in $\mathcal{J}_d(M, E)$ which is C^1 and nondegenerate. Then $\mathcal{D}s \in \Gamma^1(M, F)$ is a centered Gaussian field with values in F which is nondegenerate because of the surjectivity assumption on L. Everything we did in Sections 6.2 and 6.3 applies to $\mathcal{D}s$. In particular, $\mathcal{D}s$ satisfies the weak Bulinskaya lemma (see Proposition 6.8). Hence its vanishing locus is almost surely the union of a codimension-r submanifold of M and a negligible singular set. We denote by $\nu_{\mathcal{D}}$ the random Radon measure on M defined by integrating over the zero set of $\mathcal{D}s$. The formal definition is similar to Definition 6.11.

Example 7.6. Let us assume that $E = \mathbb{R} \times M$ is trivial. Then we can identify $\Gamma^1(M, E)$ with $\mathcal{C}^1(M)$ and consider the differential $D : \mathcal{C}^1(M) \to \Gamma^0(M, T^*M)$, which is a differential operator of order 1. Let $f : M \to \mathbb{R}$ be a \mathcal{C}^2 and 1-nondegenerate centered Gaussian field. Then Df is a nondegenerate \mathcal{C}^1

1468

centered Gaussian field on M with values in T^*M , the vanishing locus of Df is the set of critical points of f, and v_D is the counting measure this random set.

We can now state the analogue of Theorem 6.26 in this context, bearing in mind that Proposition 6.25 applies to v_{D} .

Theorem 7.7 (finiteness of moments for $v_{\mathcal{D}}$, local version). Let $\Omega \subset \mathbb{R}^n$ be open and $f : \Omega \to \mathbb{R}^q$ be a centered Gaussian field. Let $r \in \llbracket 1, n \rrbracket$. Let $\mathcal{D} : \mathcal{C}^d(\Omega, \mathbb{R}^q) \to \mathcal{C}^0(\Omega, \mathbb{R}^r)$ be a differential operator of order d satisfying the previous hypotheses and $v_{\mathcal{D}}$ denote the measure of integration over the zero set of $\mathcal{D}f$. Let $p \ge 1$. If f is $\mathcal{C}^{(d+1)p}$ and the Gaussian field $\operatorname{mj}_k^{\mathcal{D}}(f, \cdot) : C_k^{\mathcal{D}}[\Omega] \to \mathcal{MJ}_k^{\mathcal{D}}(\Omega)$ is nondegenerate for all $k \in \llbracket 1, p \rrbracket$, then the four equivalent statements in Proposition 6.25 hold for $v_{\mathcal{D}}$.

Proof. Under these hypotheses, for all $k \in [\![1, p]\!]$ the Gaussian field $mj_k^{\mathcal{D}}(f, \cdot)$ is at least \mathcal{C}^1 . Then the proof is the same as that of Theorem 6.26.

Theorem 7.8 (finiteness of moments for zeros of $\mathcal{D}s$). In the setting introduced at the beginning of this section, let $s : M \to E$ be a centered Gaussian field and $v_{\mathcal{D}}$ denote the measure of integration over the zero set of $\mathcal{D}s$. Let $p \ge 1$. If s is $\mathcal{C}^{(d+1)p}$ and ((d+1)p-1)-nondegenerate then $\mathbb{E}[|\langle v_{\mathcal{D}}, \phi \rangle|^p] < +\infty$ for all $\phi \in L_c^{\infty}(M)$.

Proof. We deduce Theorem 7.8 from Theorem 7.7 in the same way that we deduced Theorem 1.9 from Theorem 6.26; see Section 6.4. \Box

8. Multijets of holomorphic maps

The purpose of this section is to explain how to adapt what we did in Sections 3 to 6 to the case of holomorphic maps. Theorem 1.6 asks for the (p-1)-nondegeneracy of the field f, that is, $j_{p-1}(f, x)$ needs to be nondegenerate for all x. If $f : \mathbb{C}^n \to \mathbb{C}$ is a centered holomorphic Gaussian field, then $(f(x), D_x f)$ is always degenerate. Indeed, identifying canonically \mathbb{C} with \mathbb{R}^2 , the differential $D_x f$ takes values in the subspace of $\mathcal{L}(\mathbb{R}^{2n}, \mathbb{R}^2)$ consisting of \mathbb{R} -linear maps that are actually \mathbb{C} -linear. Thus, if we see the holomorphic field f as a smooth field from \mathbb{R}^{2n} to \mathbb{R}^2 , we cannot apply Theorem 1.6. From the point of view of multijets, the multijet $mj_p(f, \cdot)$ of a holomorphic function $f : \mathbb{C}^n \to \mathbb{C}$ takes values in a strict sub-bundle of $\mathcal{MJ}_p(\mathbb{R}^{2n}, \mathbb{R}^2)$, which is similar to what happens for jet bundles. Thus, the field $mj_p(f, \cdot)$ associated with a holomorphic Gaussian field f is necessarily degenerated and Theorem 6.26 does not apply. To remedy this situation, we define in Section 8.1 a multijet bundle adapted to holomorphic maps. Then, in Section 8.2, we use this holomorphic multijet to prove Theorem 1.11.

8.1. *Definition of the holomorphic multijet bundles.* In this section, we define a multijet bundle for holomorphic maps. Our main result is an equivalent of Theorem 1.1 in this context. Let us first introduce some notation.

Definition 8.1 (spaces of holomorphic maps). We define the following spaces.

• We denote by $\mathbb{C}_d[X]$ the space of complex polynomials of degree at most d in n variables, where $X = (X_1, \ldots, X_n)$ is multivariate.

• If *M* and *N* are two complex manifolds, we denote by $\mathcal{O}(M, N)$ the space of holomorphic maps from *M* to *N*. If $N = \mathbb{C}$, we simply write $\mathcal{O}(M)$.

• If $E \to M$ is a holomorphic vector bundle, we denote by $\mathcal{J}_k^{\mathbb{C}}(M, E) \to M$ the holomorphic bundle of *k*-jets of holomorphic sections of *E*. If $E = V \times M$ is trivial with fiber *V*, we denote its holomorphic *k*-jet bundle by $\mathcal{J}_k^{\mathbb{C}}(M, V) \to M$. If $V = \mathbb{C}$, we simply write $\mathcal{J}_k^{\mathbb{C}}(M) \to M$. Given a holomorphic section *s* of *E*, we denote by $j_k^{\mathbb{C}}(s, x)$ its holomorphic *k*-jet at $x \in M$.

Theorem 8.2 (existence of holomorphic multijet bundles). Let $n \ge 1$ and $p \ge 1$ and let V be a complex vector space of dimension $r \ge 1$. There exist a complex manifold $C_p^{\mathbb{C}}[\mathbb{C}^n]$ of dimension np and a holomorphic vector bundle $\mathcal{MJ}_p^{\mathbb{C}}(\mathbb{C}^n, V) \to C_p^{\mathbb{C}}[\mathbb{C}^n]$ of rank rp with the following properties:

(1) There exists a holomorphic proper surjection $\pi : C_p^{\mathbb{C}}[\mathbb{C}^n] \to (\mathbb{C}^n)^p$ such that $\pi^{-1}((\mathbb{C}^n)^p \setminus \Delta_p)$ is a dense open subset of $C_p^{\mathbb{C}}[\mathbb{C}^n]$, and π restricted to $\pi^{-1}((\mathbb{C}^n)^p \setminus \Delta_p)$ is a biholomorphism onto $(\mathbb{C}^n)^p \setminus \Delta_p$.

(2) There exists a map $\operatorname{mj}_p^{\mathbb{C}} : \mathcal{O}(\mathbb{C}^n, V) \times C_p^{\mathbb{C}}[\mathbb{C}^n] \to \mathcal{MJ}_p^{\mathbb{C}}(\mathbb{C}^n, V)$ such that

- for all $z \in C_p^{\mathbb{C}}[\mathbb{C}^n]$, the linear map $\mathrm{mj}_p^{\mathbb{C}}(\cdot, z) : \mathcal{O}(\mathbb{C}^n, V) \to \mathcal{MJ}_p^{\mathbb{C}}(\mathbb{C}^n, V)_z$ is surjective;
- for all $f \in \mathcal{O}(\mathbb{C}^n, V)$, the section $\operatorname{mj}_p^{\mathbb{C}}(f, \cdot)$ of $\mathcal{MJ}_p^{\mathbb{C}}(\mathbb{C}^n, V) \to C_p^{\mathbb{C}}[\mathbb{C}^n]$ is holomorphic.

(3) Let $z \in C_p^{\mathbb{C}}[\mathbb{C}^n]$ be such that $\pi(z) = (x_1, \ldots, x_p) \notin \Delta_p$. Then for all $f \in \mathcal{O}(\mathbb{C}^n, V)$ we have

$$\mathrm{mj}_{p}^{\mathbb{C}}(f,z) = 0 \quad \Longleftrightarrow \quad \forall i \in \llbracket 1, p \rrbracket, \ f(x_{i}) = 0.$$

(4) Let $z \in C_p^{\mathbb{C}}[\mathbb{C}^n]$, let $\mathcal{I} = \mathcal{I}(\pi(z))$ and let $(y_I)_{I \in \mathcal{I}} = \iota_{\mathcal{I}}^{-1}(\pi(z)) \in (\mathbb{C}^n)^{\mathcal{I}} \setminus \Delta_{\mathcal{I}}$. There exists a linear surjection $\Theta_z^{\mathbb{C}} : \prod_{I \in \mathcal{I}} \mathcal{J}_{|I|-1}^{\mathbb{C}}(\mathbb{C}^n, V)_{y_I} \to \mathcal{M}\mathcal{J}_p^{\mathbb{C}}(\mathbb{C}^n, V)_z$ such that

$$\forall f \in \mathcal{O}(\mathbb{C}^n, V), \quad \mathrm{mj}_p^{\mathbb{C}}(f, z) = \Theta_z^{\mathbb{C}}((j_{|I|-1}^{\mathbb{C}}(f, y_I))_{I \in \mathcal{I}}).$$

Proof. The proof follows the same steps as what we did in Sections 3, 4 and 5 to prove Theorem 1.1. In the following, we sketch how the proof of Theorem 1.1 adapts to the holomorphic case.

<u>Step 1</u>: Divided differences and Kergin interpolation. Let us consider $f \in \mathcal{O}(\mathbb{C}^n)$ and $\underline{x} = (x_0, \ldots, x_k) \in (\mathbb{C}^n)^{k+1}$. The divided difference $f[x_0, \ldots, x_k]$ from Definition 3.1 still makes sense. Since f is holomorphic, it is now a symmetric \mathbb{C} -multilinear form on \mathbb{C}^n that depends linearly on f and is holomorphic with respect to \underline{x} . As explained in [Kergin 1980, Proposition 5.1], the Kergin interpolating polynomial is well-behaved with respect to holomorphic maps. Given $f \in \mathcal{O}(\mathbb{C}^n)$ and $\underline{x} \in (\mathbb{C}^n)^p$, (3-3) defines $K(f, \underline{x}) \in \mathbb{C}_{p-1}[X]$ that interpolates the values of $f[\underline{x}_I]$ for all nonempty $I \subset [[1, p]]$. The equivalent of Lemma 3.8 is true, in the sense that $K(\cdot, \underline{x})$ is \mathbb{C} -linear and $K(f, \cdot)$ is holomorphic from $(\mathbb{C}^n)^p$ to $\mathbb{C}_{p-1}[X]$.

In this complex framework, the equivalent of Lemma 3.9 holds, that is: for all $\underline{x} \in (\mathbb{C}^n)^p$ the map $P \mapsto (K(P, \underline{x}_I))_{I \in \mathcal{I}(\underline{x})}$ is surjective from $\mathbb{C}_{p-1}[X]$ to $\prod_{I \in \mathcal{I}(\underline{x})} \mathbb{C}_{|I|-1}[X]$. Note however that the proof we gave of Lemma 3.9 does not adapt to the holomorphic setting since it uses bump functions. Here, we deduce the surjectivity of $(K(\cdot, \underline{x}_I))_{I \in \mathcal{I}(\underline{x})}$ from a general amplitude result in algebraic geometry. Let $\mathcal{I} = \mathcal{I}(\underline{x})$ and $(y_I)_{I \in \mathcal{I}} = \iota_{\mathcal{I}}^{-1}(\underline{x})$. For all $I \in \mathcal{I}$, let $P_I \in \mathbb{C}_{|I|-1}[X]$. Multiplying each monomial in P_I by the right power of X_0 yields a homogeneous polynomial $\widetilde{P}_I \in \mathbb{C}_{p-1}[X_0, \dots, X_n]$, that is, a global holomorphic section of the line bundle $\mathcal{O}(p-1) \to \mathbb{CP}^n$. Recall that $\mathcal{O}(p-1)$ is the (p-1)-th tensor

power of the hyperplane line bundle $\mathcal{O}(1) \to \mathbb{CP}^n$. Since $\mathcal{O}(1)$ is very ample, the bundle $\mathcal{O}(p-1)$ is (p-1)-jet ample; see [Beltrametti and Sommese 1993, Corollary 2.1]. This means that there exists $\widetilde{P} \in \mathbb{C}_{p-1}^{\text{hom}}[X_0, \ldots, X_n]$ with the same (|I|-1)-jet as \widetilde{P}_I at y_I for all $I \in \mathcal{I}$, where we see \mathbb{C}^n as a standard affine chart in \mathbb{CP}^n . Then, for all $I \in \mathcal{I}$, the polynomial $P = \widetilde{P}(1, X_1, \ldots, X_n) \in \mathbb{C}_{p-1}[X]$ has the same (|I|-1)-jet (i.e., the same Taylor polynomial of order |I|-1) as P_I at y_I . Thus $(K(P, \underline{x}_I))_{I \in \mathcal{I}} = (P_I)_{I \in \mathcal{I}}$. Step 2: Kernel of the evaluation and resolution of singularities. As in Definition 4.3, we define a complex evaluation map by $\operatorname{ev}_{\underline{x}}^{\mathbb{C}} : f \mapsto (f(x_1), \ldots, f(x_p))$, where $\underline{x} \in (\mathbb{C}^n)^p$. If $\underline{x} \notin \Delta_p$, this map is surjective from $\mathbb{C}_{p-1}[X]$ to \mathbb{C}^p . Hence we can define $\mathcal{G}_I^{\mathbb{C}}(\underline{x}) = \ker \operatorname{ev}_{\underline{x}_I}^{\mathbb{C}} \in \operatorname{Gr}_{|I|}(\mathbb{C}_{|I|-1}[X])$ for all nonempty $I \subset [\![1, p] \!]$, where the Grassmannian is now the Grassmannian of complex subspaces of codimension |I|. Then, everything we did in Sections 4 and 5 works in the holomorphic setting after replacing \mathbb{R} -linear objects by \mathbb{C} -linear ones.

We define $\Sigma_{\mathbb{C}}$ as the graph of $(\mathcal{G}_{I}^{\mathbb{C}})_{I \subset \llbracket 1, p \rrbracket}$ from $(\mathbb{C}^{n})^{p} \setminus \Delta_{p}$ to $\prod_{\varnothing \neq I \subset \llbracket 1, p \rrbracket} \operatorname{Gr}_{|I|}(\mathbb{C}_{|I|-1}[X])$ and $C_{p}^{\mathbb{C}}[\mathbb{C}^{n}]$ as a resolution of the singularities of its closure $\overline{\Sigma}_{\mathbb{C}}$ in $(\mathbb{C}^{n})^{p} \times \prod_{\varnothing \neq I \subset \llbracket 1, p \rrbracket} \operatorname{Gr}_{|I|}(\mathbb{C}_{|I|-1}[X])$. The resolution of singularities is a result from algebraic geometry which holds over fields of characteristic 0. In particular, in Proposition 5.5 and Corollary 5.6, "smooth" can be replaced by "algebraic" everywhere. The same results hold over \mathbb{C} , in which case algebraic implies holomorphic. Thus, $C_{p}^{\mathbb{C}}[\mathbb{C}^{n}]$ is a complex manifold of dimension np, which satisfies the equivalent of Corollary 5.6 with "smooth" replaced by "holomorphic".

<u>Step 3</u>: Definition of the holomorphic multijet bundles. Everything we did in Sections 5.2, 5.3 and 5.4 adapts to the holomorphic setting. It is enough to replace C^k functions by holomorphic ones and to write the linear arguments over \mathbb{C} instead of \mathbb{R} . We can then define the holomorphic vector bundle of *p*-multijets of holomorphic functions on \mathbb{C}^n by

$$\mathcal{MJ}_p^{\mathbb{C}}(\mathbb{C}^n) = (\mathbb{C}_{p-1}[X] \times C_p^{\mathbb{C}}[\mathbb{C}^n])/\mathcal{G}^{\mathbb{C}}$$
(8-1)

and the *p*-multijet of $f \in \mathcal{O}(\mathbb{C}^n)$ by $\text{mj}_p^{\mathbb{C}}(f, z) = K(f, \pi(z)) \mod \mathcal{G}^{\mathbb{C}}(z)$ for all $z \in C_p^{\mathbb{C}}[\mathbb{C}^n]$. If *V* is a complex vector space of finite dimension, we define as in Definition 5.18

$$\mathcal{MJ}_{p}^{\mathbb{C}}(\mathbb{C}^{n}, V) = \mathcal{MJ}_{p}^{\mathbb{C}}(\mathbb{C}^{n}) \otimes V.$$
(8-2)

Then we define the *p*-multijet of $f \in \mathcal{O}(\mathbb{C}^n, V)$ as in Definition 5.19. If (v_1, \ldots, v_r) is a basis of *V* and $f = \sum_{i=1}^r f_i v_i$ is holomorphic, then $\operatorname{mj}_p^{\mathbb{C}}(f, z) = \sum_{i=1}^r \operatorname{mj}_p^{\mathbb{C}}(f_i, z) \otimes v_i$ for all $z \in C_p^{\mathbb{C}}[\mathbb{C}^n]$. As in Lemma 5.20, this definition does not depend on a choice of basis. This defines the holomorphic *p*-multijet that we are looking for.

As in the real case, thanks to (4) in Theorem 8.2, the holomorphic multijet $\text{mj}_p^{\mathbb{C}}(f, z)$ of f at $z \in C_p^{\mathbb{C}}[\mathbb{C}^n]$ only depends on the germ of f near the x_i , where $(x_i)_{1 \le i \le p} = \pi(z)$. Thus, we can define a holomorphic multijet bundle over any open subset of \mathbb{C}^n .

Definition 8.3 (holomorphic multijets). Let $\Omega \subset \mathbb{C}^n$ be open. We denote by $C_p^{\mathbb{C}}[\Omega] = \pi^{-1}(\Omega^p)$ and by $\mathcal{MJ}_p^{\mathbb{C}}(\Omega, V) \to C_p^{\mathbb{C}}[\Omega]$ the restriction of $\mathcal{MJ}_p^{\mathbb{C}}(\mathbb{C}^n, V)$ to $C_p[\Omega]$. If $V = \mathbb{C}$, we drop it from the notation and write $\mathcal{MJ}_p^{\mathbb{C}}(\Omega) \to C_p^{\mathbb{C}}[\Omega]$. Let $f \in \mathcal{O}(\Omega, V)$, we call the section $\operatorname{mj}_p^{\mathbb{C}}(f, \cdot)$ of $\mathcal{MJ}_p^{\mathbb{C}}(\Omega, V)$ the *holomorphic p-multijet of f*. **8.2.** *Application to the zeros of holomorphic Gaussian fields.* In this section, we explain how the holomorphic multijets defined in Section 8.1 allow us to prove Theorem 1.11, and the analogue of Theorem 6.26 for holomorphic Gaussian fields. We start by recalling a few facts about complex Gaussian vectors; see [Andersen et al. 1995, Chapter 2].

A random variable $X \in \mathbb{C}$ is called a *centered complex Gaussian* if its real and imaginary parts are independent real centered Gaussian variables of the same variance, i.e., there exists $\lambda \ge 0$ such that $X = X_{\Re} + iX_{\Im}$, with $(X_{\Re}, X_{\Im}) \sim \mathcal{N}(0, \lambda \operatorname{Id})$ in \mathbb{R}^2 .

Definition 8.4 (complex Gaussian vector). We say that a random vector X in a finite-dimensional complex vector space V is a *centered Gaussian* if for all $\eta \in V^*$ the complex variable $\eta(X)$ is a centered complex Gaussian.

If *V* is equipped with a Hermitian inner product $\langle \cdot, \cdot \rangle$ and we define $v^* = \langle v, \cdot \rangle$ then the *variance* of *X* is the nonnegative Hermitian operator $\operatorname{Var}_{\mathbb{C}}(X) = \mathbb{E}[X \otimes X^*]$. We say that *X* is *nondegenerate* if $\operatorname{Var}_{\mathbb{C}}(X)$ is positive-definite.

Remark 8.5. As in the real case, the Gaussianity and nondegeneracy of *X* do not depend on $\langle \cdot, \cdot \rangle$, but the variance operator does.

A centered complex Gaussian vector X in $(V, \langle \cdot, \cdot \rangle)$ is completely determined by its variance. For example, if $\operatorname{Var}_{\mathbb{C}}(X) = \Lambda$ is positive-definite, then X admits the density $v \mapsto e^{-\langle v, \Lambda^{-1}v \rangle}/\det(\pi \Lambda)$ with respect to the Lebesgue measure on V. We denote by $\mathcal{N}_{\mathbb{C}}(0, \Lambda)$ the centered complex Gaussian distribution of variance Λ . Then $X \sim \mathcal{N}_{\mathbb{C}}(0, \Lambda)$ in \mathbb{C}^n if and only if its real and imaginary part satisfy

$$(X_{\Re}, X_{\Im}) \sim \mathcal{N}\left(0, \frac{1}{2}\begin{pmatrix} \Re(\Lambda) & \Im(\Lambda) \\ -\Im(\Lambda) & \Re(\Lambda) \end{pmatrix}\right)$$

in \mathbb{R}^{2n} .

Let $E \to M$ be a holomorphic vector bundle over a complex manifold M. We denote by $H^0(M, E)$ the vector space of global holomorphic sections of $E \to M$.

Definition 8.6 (holomorphic Gaussian field). We say that a random section $s \in H^0(M, E)$ is a *centered* holomorphic Gaussian field if for all $m \ge 1$ and all x_1, \ldots, x_m the random vector $(s(x_1), \ldots, s(x_m))$ is a centered complex Gaussian. We say that this field is *nondegenerate* if s(x) is nondegenerate for all $x \in M$.

Note that if $s \in H^0(M, E)$ is a centered holomorphic Gaussian field then, for all $k \in \mathbb{N}$, its holomorphic k-jet $j_k^{\mathbb{C}}(s, \cdot)$ defines a centered holomorphic Gaussian field with values in $\mathcal{J}_k^{\mathbb{C}}(M, E)$.

Definition 8.7 (*p*-nondegeneracy for holomorphic fields). Let $p \ge 1$. We say that the centered holomorphic Gaussian field $s \in H^0(M, E)$ is *p*-nondegenerate if the centered complex Gaussian $j_p^{\mathbb{C}}(s, x) \in \mathcal{J}_p^{\mathbb{C}}(M, E)_x$ is nondegenerate for all $x \in M$.

As in the real framework, we need the following definition.

Definition 8.8 (complex Jacobian). Let $L: V \to V'$ be a \mathbb{C} -linear map between Hermitian spaces and let L^* denote its adjoint map. The *complex Jacobian* of L is defined as $Jac_{\mathbb{C}}(L) = det(LL^*)$.

Remark 8.9. If we see *V*, *V'* and *L* as \mathbb{R} -linear objects and we equip *V* and *V'* with the Euclidean structures induced by their Hermitian inner products, then the real and complex Jacobians are related by $Jac_{\mathbb{C}}(L) = Jac(L)^2$; see [Andersen et al. 1995, Theorem A.5].

Let us consider a complex manifold M of complex dimension n equipped with a Riemannian metric gand a holomorphic vector bundle $E \to M$ of complex rank $r \in [\![1, n]\!]$. In the following, we denote by ∇ a connection on E which is compatible with the complex structure. As in the real case, the choice of this connection will not matter. Let $s \in H^0(M, E)$ be a centered holomorphic Gaussian field on M with values in E and let $Z = s^{-1}(0)$ denote its vanishing locus. We will always assume that s is nondegenerate. In this setting, the random section s satisfies a strong Bulinskaya-type lemma.

Proposition 8.10 (holomorphic Bulinskaya lemma). Almost surely the following set is empty:

 $\{x \in M \mid s(x) = 0 \text{ and } \operatorname{Jac}_{\mathbb{C}}(\nabla_x s) = 0\}.$

In particular, the zero set Z is almost surely a (possibly empty) complex submanifold of complex codimension r in M.

Proof. It is enough to check the result locally. On an open subset $\Omega \subset M$ over which *E* is trivial, we can consider *s* as a nondegenerate smooth centered Gaussian field from Ω to $\mathbb{C}^r \simeq \mathbb{R}^{2r}$. Then, the local result follows from [Lerario and Stecconi 2019, Theorem 7].

Let us consider Z as random submanifold of real codimension 2r in the Riemannian manifold (M, g) of real dimension 2n. The metric g induces a (2n-2r)-dimensional Riemannian volume $dVol_Z$ on Z and we can define ν as in Definition 6.11, bearing in mind that $Z = Z_{reg}$. Then Propositions 6.17, 6.24 and 6.25 hold for the holomorphic field s and the associated linear statistics $\langle \nu, \phi \rangle$ with $\phi \in L_c^{\infty}(M)$. More generally, everything we did in Sections 6.2, 6.3 and 6.4 adapts to the holomorphic setting.

Remark 8.11. In Definition 6.23, the function ρ_p is defined in terms of real Jacobians and the variance of $(s(x_1), \ldots, s(x_p))$ seen as a real Gaussian vector. One can check that another expression of $\rho_p(x_1, \ldots, x_p)$ is the following, which is more natural in our holomorphic framework:

$$\forall (x_1, \dots, x_p) \notin \Delta_p, \quad \rho_p(x_1, \dots, x_p) = \frac{\mathbb{E}\left[\prod_{i=1}^p \operatorname{Jac}_{\mathbb{C}}(\nabla_{x_i} s) \mid \forall i \in \llbracket 1, p \rrbracket, s(x_i) = 0\right]}{\det(\pi \operatorname{Var}_{\mathbb{C}}(s(x_1), \dots, s(x_p)))}$$

We can now state the equivalent of Theorem 6.26 for holomorphic Gaussian fields. Let $\Omega \subset \mathbb{C}^n$ be open. Recall that $\mathcal{MJ}_p^{\mathbb{C}}(\Omega, \mathbb{C}^r) \to C_p^{\mathbb{C}}[\Omega]$ is defined in Definition 8.3 as the restriction over $C_p^{\mathbb{C}}[\Omega] \subset C_p^{\mathbb{C}}[\mathbb{C}^n]$ of the vector bundle $\mathcal{MJ}_p^{\mathbb{C}}(\mathbb{C}^n, \mathbb{C}^r) \to C_p^{\mathbb{C}}[\mathbb{C}^n]$ from Theorem 8.2.

Theorem 8.12 (finiteness of moments for holomorphic fields, local version). Let $f : \Omega \to \mathbb{C}^r$ be a centered holomorphic Gaussian field and v be as in Definition 6.11. Let $p \ge 1$. If for all $k \in [[1, p]]$ the holomorphic Gaussian field $mj_k^{\mathbb{C}}(f, \cdot) \in H^0(C_k^{\mathbb{C}}[\Omega], \mathcal{MJ}_k^{\mathbb{C}}(\Omega, \mathbb{C}^r))$ is nondegenerate, then the four equivalent statements in Proposition 6.25 hold.

Proof. The proof of Theorem 6.26 relies mostly on two facts that are valid for all $k \in [[1, p]]$. First, on the open dense subset $\Omega^k \setminus \Delta_k \subset C_k[\Omega]$, the zero set of $mj_k(f, \cdot)$ is the same as that of $(x_1, \ldots, x_k) \mapsto$

 $(f(x_1), \ldots, f(x_k))$. And second, the field $mj_k(f, \cdot)$ is nondegenerate on $C_k[\Omega]$, so that we can apply the Kac–Rice formula (Proposition 6.17) to the *k*-multijet.

These two facts are still true in the present holomorphic setting. Hence, the same proof as that of Theorem 6.26 yields the result. \Box

We deduce from this result the equivalent of Theorem 1.9 for a centered holomorphic Gaussian field *s* on a complex manifold *M* of dimension *n* with values in a holomorphic vector bundle *E* of rank $r \in [[1, n]]$. Theorem 1.11 is a special case of the following.

Theorem 8.13 (finiteness of moments for zeros of holomorphic Gaussian sections). Let $p \ge 1$, let $s \in H^0(M, E)$ be a centered holomorphic Gaussian field and let v be as in Definition 6.11. If s is (p-1)-nondegenerate then $\mathbb{E}[|\langle v, \phi \rangle|^p] < +\infty$ for all $\phi \in L^{\infty}_{c}(M)$.

Proof. We deduce Theorem 8.13 from Theorem 8.12 in the same way that we deduced Theorem 1.9 from Theorem 6.26; see Section 6.4. \Box

Remark 8.14. In particular, Theorem 8.13 proves the local integrability of the *p*-points correlation functions studied in [Bleher et al. 2000] and their scaling limit.

Acknowledgments

The authors thank Vincent Borrelli, Nicolas Vichery and Gabriel Rivière for independently suggesting to take a look at various compactifications of configuration spaces.

References

- [Ancona 2021] M. Ancona, "Random sections of line bundles over real Riemann surfaces", *Int. Math. Res. Not.* **2021**:9 (2021), 7004–7059. MR Zbl
- [Ancona and Letendre 2021] M. Ancona and T. Letendre, "Zeros of smooth stationary Gaussian processes", *Electron. J. Probab.* **26** (2021), art. id. 68. MR Zbl
- [Andersen et al. 1995] H. H. Andersen, M. Højbjerre, D. Sørensen, and P. S. Eriksen, *Linear and graphical models for the multivariate complex normal distribution*, Lecture Notes in Stat. **101**, Springer, 1995. MR Zbl
- [Armentano et al. 2019] D. Armentano, J.-M. Azaïs, D. Ginsbourger, and J. R. León, "Conditions for the finiteness of the moments of the volume of level sets", *Electron. Commun. Probab.* 24 (2019), art. id. 17. MR Zbl
- [Armentano et al. 2023a] D. Armentano, J. M. Azaïs, F. Dalmao, J. R. León, and E. Mordecki, "On the finiteness of the moments of the measure of level sets of random fields", *Braz. J. Probab. Stat.* **37**:1 (2023), 219–245. MR Zbl
- [Armentano et al. 2023b] D. Armentano, J.-M. Azaïs, and J. R. León, "On a general Kac–Rice formula for the measure of a level set", preprint, 2023. arXiv 2304.07424
- [Axelrod and Singer 1994] S. Axelrod and I. M. Singer, "Chern–Simons perturbation theory, II", *J. Differential Geom.* **39**:1 (1994), 173–213. MR Zbl
- [Azaïs and Wschebor 2009] J.-M. Azaïs and M. Wschebor, *Level sets and extrema of random processes and fields*, Wiley, Hoboken, NJ, 2009. MR Zbl
- [Beliaev et al. 2024] D. Beliaev, M. McAuley, and S. Muirhead, "A central limit theorem for the number of excursion set components of Gaussian fields", *Ann. Probab.* **52**:3 (2024), 882–922. MR Zbl
- [Beltrametti and Sommese 1993] M. C. Beltrametti and A. J. Sommese, "On *k*-jet ampleness", pp. 355–376 in *Complex analysis and geometry*, edited by V. Ancona and A. Silva, Springer, 1993. MR Zbl

- [Belyaev 1966] Y. K. Belyaev, "On the number of intersections of a level by a Gaussian stochastic process, I", *Theory Probab. Appl.* **11**:1 (1966), 106–113. Zbl
- [Bleher et al. 2000] P. Bleher, B. Shiffman, and S. Zelditch, "Universality and scaling of correlations between zeros on complex manifolds", *Invent. Math.* **142**:2 (2000), 351–395. MR Zbl
- [Bojanov et al. 1993] B. D. Bojanov, H. A. Hakopian, and A. A. Sahakian, *Spline functions and multivariate interpolations*, Math. Appl. **248**, Kluwer, Dordrecht, 1993. MR Zbl
- [Evain 2005] L. Evain, "Compactifications des espaces de configuration dans les schémas de Hilbert", *Bull. Soc. Math. France* **133**:4 (2005), 497–539. MR Zbl
- [Fulton and MacPherson 1994] W. Fulton and R. MacPherson, "A compactification of configuration spaces", *Ann. of Math.* (2) **139**:1 (1994), 183–225. MR Zbl
- [Gass and Stecconi 2024] L. Gass and M. Stecconi, "The number of critical points of a Gaussian field: finiteness of moments", *Probab. Theory Related Fields* **190**:3-4 (2024), 1167–1197. MR Zbl
- [Hauser 2003] H. Hauser, "The Hironaka theorem on resolution of singularities (or: A proof we always wanted to understand)", *Bull. Amer. Math. Soc.* (*N.S.*) **40**:3 (2003), 323–403. MR Zbl
- [Hironaka 1964a] H. Hironaka, "Resolution of singularities of an algebraic variety over a field of characteristic zero, I", *Ann. of Math.* (2) **79**:1 (1964), 109–203. MR Zbl
- [Hironaka 1964b] H. Hironaka, "Resolution of singularities of an algebraic variety over a field of characteristic zero, II", *Ann. of Math.* (2) **79**:2 (1964), 205–326. MR Zbl
- [Kergin 1980] P. Kergin, "A natural interpolation of C^K functions", J. Approx. Theory 29:4 (1980), 278–293. MR Zbl
- [Kollár 2007] J. Kollár, Lectures on resolution of singularities, Ann. of Math. Stud. 166, Princeton Univ. Press, 2007. MR Zbl
- [Le Barz 1988] P. Le Barz, "La variété des triplets complets", Duke Math. J. 57:3 (1988), 925-946. MR Zbl
- [Lee 2018] J. M. Lee, Introduction to Riemannian manifolds, 2nd ed., Grad. Texts in Math. 176, Springer, 2018. MR Zbl
- [Lerario and Stecconi 2019] A. Lerario and M. Stecconi, "Differential topology of Gaussian random fields", preprint, 2019. arXiv 1902.03805
- [Lorentz 2000] R. A. Lorentz, "Multivariate Hermite interpolation by algebraic polynomials: a survey", *J. Comput. Appl. Math.* **122**:1-2 (2000), 167–201. MR Zbl
- [Malevich and Volodina 1993] T. L. Malevich and L. N. Volodina, "Some finiteness conditions for factorial moments of the number of zeros of Gaussian field zeros", *Theory Probab. Appl.* **38**:1 (1993), 27–45. Zbl
- [Micchelli and Milman 1980] C. A. Micchelli and P. Milman, "A formula for Kergin interpolation in \mathbb{R}^{k} ", *J. Approx. Theory* **29**:4 (1980), 294–296. MR Zbl
- [Nazarov and Sodin 2012] F. Nazarov and M. Sodin, "Correlation functions for random complex zeroes: strong clustering and local universality", *Comm. Math. Phys.* **310**:1 (2012), 75–98. MR Zbl
- [Olver 2001] P. J. Olver, "Geometric foundations of numerical algorithms and symmetry", *Appl. Algebra Engrg. Comm. Comput.* **11**:5 (2001), 417–436. MR Zbl
- [Priya 2020] L. Priya, "Overcrowding estimates for zero count and nodal length of stationary Gaussian processes", preprint, 2020. arXiv 2012.10857
- [Saunders 1989] D. J. Saunders, *The geometry of jet bundles*, Lond. Math. Soc. Lect. Note Ser. **142**, Cambridge Univ. Press, 1989. MR Zbl
- [Sinha 2004] D. P. Sinha, "Manifold-theoretic compactifications of configuration spaces", *Selecta Math.* (*N.S.*) **10**:3 (2004), 391–428. MR Zbl
- [Ulyanov 2002] A. P. Ulyanov, "Polydiagonal compactification of configuration spaces", *J. Algebraic Geom.* **11**:1 (2002), 129–159. MR Zbl
- [Włodarczyk 2005] J. Włodarczyk, "Simple Hironaka resolution in characteristic zero", J. Amer. Math. Soc. 18:4 (2005), 779–822. MR Zbl

Received 1 Oct 2023. Accepted 24 May 2024.

MICHELE ANCONA: michele.ancona@unice.fr Université Côte d'Azur, CNRS, LJAD, Nice, France

THOMAS LETENDRE: letendre@math.cnrs.fr Université Paris-Saclay, CNRS, LMO, Orsay, France





EXISTENCE OF SOLUTIONS TO A FRACTIONAL SEMILINEAR HEAT EQUATION IN UNIFORMLY LOCAL WEAK ZYGMUND-TYPE SPACES

NORISUKE IOKU, KAZUHIRO ISHIGE AND TATSUKI KAWAKAMI

We introduce uniformly local weak Zygmund-type spaces and obtain an optimal sufficient condition for the existence of solutions to the critical fractional semilinear heat equation.

1. Introduction

Consider the Cauchy problem for the fractional semilinear heat equation

$$\begin{cases} \partial_t u + (-\Delta)^{\frac{\theta}{2}} u = |u|^{p-1} u, & x \in \mathbb{R}^n, \ t > 0, \\ u(x,0) = \varphi(x), & x \in \mathbb{R}^n, \end{cases}$$
(P)

where $n \ge 1$, $\partial_t := \partial/\partial t$, $\theta \in (0, 2]$, p > 1, and φ is a locally integrable function in \mathbb{R}^n . Here $(-\Delta)^{\theta/2}$ denotes the fractional power of the Laplace operator $-\Delta$ in \mathbb{R}^n . In this paper we establish the local-in-time existence of solutions to problem (P) in the critical case

$$p = p_{\theta} := 1 + \frac{\theta}{n}$$

in the framework of uniformly local weak Zygmund-type spaces.

The solvability of the Cauchy problem for semilinear heat equations has fascinated many mathematicians since the pioneering work by Fujita [1966]. The literature is very large, and we refer the reader to the comprehensive monograph [Quittner and Souplet 2007] and the papers [Andreucci and DiBenedetto 1991; Baras and Pierre 1985; Brezis and Cazenave 1996; Fujishima et al. 2023; 2024; Fujishima and Ioku 2021; 2022; Giraudon and Miyamoto 2022; Hisa and Ishige 2018; Hisa et al. 2023; Ishige et al. 2014; 2020; 2022; Kozono and Yamazaki 1994; Laister and Sierżęga 2020; 2021; Laister et al. 2016; Miyamoto 2021; Robinson and Sierżęga 2013; Sugitani 1975; Weissler 1981; Zhanpeisov 2023], some of which are closely related to this paper, while the others include recent developments in this subject. The study of the solvability of problem (P) is divided into the following three cases:

 $1 (subcritical case), <math>p > p_{\theta}$ (supercritical case), $p = p_{\theta}$ (critical case).

We collect some known results on necessary conditions and sufficient conditions for the existence of solutions to problem (P).

MSC2020: 35A01, 35A21, 35K58, 35R11, 46E30.

Keywords: solvability, fractional semilinear heat equation, Zygmund-type spaces.

^{© 2025} MSP (Mathematical Sciences Publishers). Distributed under the Creative Commons Attribution License 4.0 (CC BY). Open Access made possible by subscribing institutions via Subscribe to Open.

- (1) Subcritical case (1 .
- (a) <u>Necessity</u>: There exists $C_1 = C_1(n, \theta, p) > 0$ such that if problem (P) possesses a nonnegative solution in $\mathbb{R}^n \times (0, T)$ for some T > 0, then

$$\sup_{x\in\mathbb{R}^n}\int_{B(x,T^{1/\theta})}\varphi(y)\,dy\leq C_1T^{\frac{n}{\theta}-\frac{1}{p-1}}.$$

See [Andreucci and DiBenedetto 1991; Baras and Pierre 1985] for $\theta = 2$ and [Hisa and Ishige 2018] for $\theta \in (0, 2]$.

(b) <u>Sufficiency</u>: There exists $\epsilon_1 = \epsilon_1(n, \theta, p) > 0$ such that if

$$\sup_{x \in \mathbb{R}^n} \int_{B(x, T^{1/\theta})} |\varphi(y)| \, dy \le \epsilon_1 T^{\frac{n}{\theta} - \frac{1}{p-1}}$$

for some $T \in (0, \infty)$, then problem (P) possesses a solution in $\mathbb{R}^n \times (0, T)$. See, e.g., [Andreucci and DiBenedetto 1991; Hisa and Ishige 2018; Weissler 1981].

The results in (a) and (b) (see also (1-4)) imply that, for any nonnegative measurable initial function φ in \mathbb{R}^n , problem (P) possesses a local-in-time nonnegative solution if and only if

$$\sup_{x\in\mathbb{R}^n}\int_{B(x,1)}\varphi(y)\,dy<\infty.$$

- (2) Supercritical case $(p > p_{\theta})$.
- (a) <u>Necessity</u>: There exists $C_2 = C_2(n, \theta, p) > 0$ such that if problem (P) possesses a nonnegative solution in $\mathbb{R}^n \times (0, T)$ for some T > 0, then

$$\sup_{x\in\mathbb{R}^n}\sup_{\sigma\in(0,T^{1/\theta})}|B(x,\sigma)|^{\frac{\theta}{n(p-1)}-1}\int_{B(x,\sigma)}\varphi(y)\,dy\leq C_2.$$

See [Andreucci and DiBenedetto 1991; Baras and Pierre 1985] for $\theta = 2$ and [Hisa and Ishige 2018] for $\theta \in (0, 2]$.

(b) <u>Sufficiency</u>: For any $r \in (1, \infty)$, there exists $\epsilon_2 = \epsilon_2(n, \theta, p, r) > 0$ such that if

$$\sup_{x \in \mathbb{R}^n} \sup_{\sigma \in (0, T^{1/\theta})} |B(x, \sigma)|^{\frac{\theta}{n(p-1)} - \frac{1}{r}} \left[\int_{B(x, \sigma)} |\varphi(y)|^r \, dy \right]^{\frac{1}{r}} \le \epsilon_2$$

for some $T \in (0, \infty]$, then problem (P) possesses a solution in $\mathbb{R}^n \times (0, T)$. See [Kozono and Yamazaki 1994; Robinson and Sierżęga 2013] for $\theta = 2$ and [Hisa and Ishige 2018; Ishige et al. 2020; 2022; Zhanpeisov 2023] for $\theta \in (0, 2]$. See, e.g., [Andreucci and DiBenedetto 1991; Ishige et al. 2014; Weissler 1981] for related results.

- (3) Critical case $(p = p_{\theta})$.
- (a) <u>Necessity</u>: There exists $C_3 = C_3(n, \theta) > 0$ such that if problem (P) possesses a nonnegative solution in $\mathbb{R}^n \times (0, T)$ for some T > 0, then

$$\sup_{x \in \mathbb{R}^n} \int_{B(x,\sigma)} \varphi(y) \, dy \le C_3 \left[\log \left(e + \frac{T^{1/\theta}}{\sigma} \right) \right]^{-\frac{n}{\theta}}, \quad \sigma \in (0, T^{\frac{1}{\theta}}).$$

See [Baras and Pierre 1985] for $\theta = 2$ and [Hisa and Ishige 2018] for $\theta \in (0, 2]$.

1478

(b) <u>Sufficiency</u>: For any $\alpha > 0$, there exists $\epsilon_3 = \epsilon_3(n, \theta, \alpha) > 0$ such that if

$$\sup_{x\in\mathbb{R}^n}\Psi_{\alpha}^{-1}\left[\frac{1}{|B(x,\sigma)|}\int_{B(x,\sigma)}\Psi_{\alpha}(T^{\frac{1}{p-1}}|\varphi(y)|)\,dy\right] \leq \epsilon_{3}\rho(\sigma T^{-\frac{1}{\theta}}), \quad \sigma\in(0,T^{\frac{1}{\theta}}),$$

for some T > 0, then problem (P) possesses a solution in $\mathbb{R}^n \times (0, T)$, where

$$\Psi_{\alpha}(s) := s[\log(e+s)]^{\alpha}, \quad \rho(s) := s^{-n} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\frac{n}{\theta}}$$

See [Hisa and Ishige 2018; Ishige et al. 2020; 2022].

Furthermore, the results in (2) and (3) imply the following results.

(4) Let $p \ge p_{\theta}$, and set

$$\varphi_{c}(x) := \begin{cases} |x|^{-n} \left[\log\left(e + \frac{1}{|x|}\right) \right]^{-\frac{n}{\theta} - 1} & \text{if } p = p_{\theta}, \\ |x|^{-\frac{\theta}{p-1}} & \text{if } p > p_{\theta} \end{cases} \quad \text{for } x \in \mathbb{R}^{n}.$$

$$(1-1)$$

(a) There exists $C_4 = C_4(n, \theta, p) > 0$ such that if

$$\varphi(x) \ge C_4 \varphi_c(x)$$

12

for almost all x in a neighborhood of the origin, then problem (P) possesses no local-in-time nonnegative solutions.

(b) There exists $\epsilon_4 = \epsilon_4(n, \theta, p) > 0$ such that if

$$|\varphi(x)| \le \epsilon_4 \varphi_c(x) + K$$
, a.a. $x \in \mathbb{R}^n$,

for some $K \ge 0$, then problem (P) possesses a local-in-time solution.

The results in (4) show that the "strength" of the singularity at the origin of the function φ_c is the critical threshold for the local-in-time solvability of problem (P). The function φ_c is quite useful for identifying optimal function spaces to which initial functions belong from the view of the solvability of problem (P). We remark that assertion (2b) with r = 1 and assertion (3b) with $\alpha = 0$ do not hold. (See [Takahashi 2016, Theorem 1 and Proposition 1], which treat only the case of $\theta = 2$ but which is also applicable to the case of $\theta \in (0, 2)$. See also [Kan and Takahashi 2017, Section 4].)

There are (at least) two useful strategies for the proof of the existence of solutions to problem (P). One is the supersolution method (SSM) and the other is the contraction mapping theorem (CMT). SSM depends on the following principle: if there exists a nonnegative supersolution v to problem (P) in $\mathbb{R}^n \times (0, T)$ for some T > 0, then problem (P) possesses a nonnegative solution u in $\mathbb{R}^n \times (0, T)$ such that $u \le v$ in $\mathbb{R}^n \times (0, T)$. In our problem (P) with nonnegative initial function φ , the following functions have been used as supersolutions in $\mathbb{R}^n \times (0, T)$ for some T > 0:

$$2S_{\theta}(t)\varphi \ (1 p_{\theta}), \quad 2\Psi_{\alpha}^{-1}(S_{\theta}(t)\Psi_{\alpha}(\varphi)) \ (p = p_{\theta}),$$

where $S_{\theta}(t)\varphi$ is a solution to the fractional heat equation (see (1-5)), r > 1, and Ψ_{α} is as in assertion (3b). (See, e.g., [Weissler 1981] for 1 ; [Hisa and Ishige 2018; Robinson and Sierżęga 2013] for $p > p_{\theta}$; [Hisa and Ishige 2018] for $p = p_{\theta}$.) Furthermore, thanks to the arguments in [Tayachi and Weissler 2014], SSM is also applicable to the study of the existence of sign-changing solutions to problem (P) (see [Ishige et al. 2020; 2022]), however we require additional arguments in which sense the solution converges to the initial function. On the other hand, CMT is widely used in the proof of the existence of solutions in various evolution equations, and the choice of function spaces is crucial. For our problem (P) with $p > p_{\theta}$, the existence of solutions has been proved by CMT in the framework of weak Lebesgue spaces (see [Fujishima and Ioku 2021; Ishige et al. 2014]) and Morrey spaces (see [Kozono and Yamazaki 1994; Zhanpeisov 2023]). The results in [Fujishima and Ioku 2021; Ishige et al. 2014; Kozono and Yamazaki 1994; Zhanpeisov 2023] cover the result in (4b) with $p > p_{\theta}$. However, in the critical case $p = p_{\theta}$, the arguments in [Fujishima and Ioku 2021; Ishige et al. 2014; Kozono and Yamazaki 1994; Zhanpeisov 2023] are not applicable to the proof of assertion (4b) by the logarithmic singularity of φ_c .

	supersolution method (SSM)	weak spaces (CMT)	Morrey spaces (CMT)
$p > p_{\theta}$	[Hisa and Ishige 2018] [Robinson and Sierżęga 2013]	[Fujishima and Ioku 2021] [Ishige et al. 2014]	[Kozono and Yamazaki 1994] [Zhanpeisov 2023]
$p = p_{\theta}$	[Hisa and Ishige 2018]	open	not applicable (see [Takahashi 2016])

The aim of this paper is to establish a sharp sufficient condition on the existence of solutions to problem (P) in the critical case $p = p_{\theta}$ in the framework of Banach spaces. For the critical case $p = p_{\theta}$, the weak Zygmund space $L^{1,\infty}(\log L)^{1+n/\theta}$ seems a reasonable Banach space since $\varphi_c \in L^{1,\infty}(\log L)^{1+n/\theta}$. (See Remark 4.4 (i) for the definition of the weak Zygmund spaces $L^{q,\infty}(\log L)^{\alpha}$, where $1 \le q < \infty$ and $\alpha \ge 0$.) Then we require sharp decay estimates of solutions to the fractional heat equation in the weak Zygmund spaces $L^{q,\infty}(\log L)^{\alpha}$; however, by the peculiarity of $L^{1,\infty}(\log L)^{\alpha}$, it seems difficult to obtain our desired sharp decay estimates. (See Remark 4.4 (ii) for further details.)

In this paper we introduce new weak Zygmund-type spaces $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ and uniformly local weak Zygmund-type spaces $\mathfrak{L}^{q,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$. Then we establish sharp decay estimates of solutions to the fractional heat equation in the spaces $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ and $\mathfrak{L}^{q,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$, and obtain a sufficient condition on the existence of solutions to problem (P) with $p = p_{\theta}$ in the framework of the space $\mathfrak{L}^{q,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$. Our sufficient condition is simpler than that of assertion (3b) and covers assertion (4b) with $p = p_{\theta}$.

We introduce some notation and define the weak Zygmund-type spaces $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ and the uniformly local weak Zygmund-type spaces $\mathfrak{L}^{q,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$. We also formulate the definition of solutions to problem (P). Let \mathcal{M} be the set of Lebesgue measurable sets in \mathbb{R}^n . For any $E \in \mathcal{M}$, we denote by |E|and χ_E the *n*-dimensional Lebesgue measure of E and the characteristic function of E, respectively. Let L^1_{loc} be the set of locally integrable functions in \mathbb{R}^n . For any $q \in [1, \infty]$, we denote by L^q and $\|\cdot\|_{L^q}$ the usual L^q -space on \mathbb{R}^n and its norm, respectively.

Let $q \in [1, \infty]$ and $\alpha \in [0, \infty)$. We define the weak Zygmund-type space $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ by

$$\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha} := \{ f \in L^1_{\mathrm{loc}} : \|f\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} < \infty \},\$$

where

$$\|f\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} := \begin{cases} \sup_{s>0} \{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \sup_{|E|=s} \int_{E} |f(x)|^{q} dx \}^{\frac{1}{q}} & \text{if } q \in [1,\infty), \\ \|f\|_{L^{\infty}} & \text{if } q = \infty. \end{cases}$$
(1-2)

Then $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ is a Banach space equipped with the norm $\|\cdot\|_{\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}}$ (see Lemma 2.1). See (2-9) for different expressions of the norm $\|\cdot\|_{\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}}$. We remark that

$$L^{q} = \mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{0} \supset \mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha} \quad \text{for } \alpha \ge 0.$$
(1-3)

Next, we introduce the uniformly local weak Zygmund-type space $\mathfrak{L}_{ul}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ by

$$\mathfrak{L}^{q,\infty}_{\mathrm{ul}}(\log\mathfrak{L})^{\alpha} := \{ f \in L^1_{\mathrm{loc}} : \| f \|_{\mathfrak{L}^{q,\infty}_{\mathrm{ul}}(\log\mathfrak{L})^{\alpha}} < \infty \},\$$

where

$$\|f\|_{\mathfrak{L}^{q,\infty}_{\mathrm{ul}}(\log\mathfrak{L})^{\alpha}} := \sup_{z \in \mathbb{R}^n} \|f\chi_{B(z,1)}\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}}$$

Then $\mathcal{L}_{ul}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ is also a Banach space equipped with the norm $\|\cdot\|_{\mathfrak{L}_{ul}^{q,\infty}(\log \mathfrak{L})^{\alpha}}$. We often write, for any $f \in \mathfrak{L}_{ul}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ and $\rho > 0$,

$$|||f|||_{q,\alpha;\rho} := \sup_{z \in \mathbb{R}^n} ||f\chi_{B(z,\rho)}||_{\mathcal{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}}$$

for simplicity. We remark that $\mathfrak{L}_{ul}^{\infty,\infty}(\log \mathfrak{L})^{\alpha} = L^{\infty}$ and $\|\|\cdot\|\|_{\infty,\alpha;\rho} = \|\cdot\|_{L^{\infty}}$ for all $\alpha \in [0,\infty)$. Notice that, for any $k \ge 1$, there exists C = C(n,k) > 0 such that

$$|||f|||_{q,\alpha;k\rho} \le C \,|||f|||_{q,\alpha;\rho} \tag{1-4}$$

for $f \in \mathfrak{L}^{q,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\alpha}$ and $\rho > 0$.

We formulate the definition of solutions to problem (P). Let $\theta \in (0, 2]$. Let G_{θ} be the fundamental solution to the fractional heat equation

$$\partial_t v + (-\Delta)^{\frac{\theta}{2}} v = 0 \quad \text{in } \mathbb{R}^n \times (0, \infty).$$

For any φ in L^1_{loc} , we write

$$(S_{\theta}(t)\varphi)(x) := \int_{\mathbb{R}^n} G_{\theta}(x-y,t)\varphi(y) \, dy, \quad (x,t) \in \mathbb{R}^n \times (0,\infty), \tag{1-5}$$

for simplicity.

Definition 1.1. Let $\theta \in (0, 2]$, p > 1, and T > 0. Set $F_p(s) := |s|^{p-1}s$ for $s \in \mathbb{R}$. Let u be a measurable and finite almost everywhere function in $\mathbb{R}^n \times (0, T)$. We say that u is a solution to problem (P) in $\mathbb{R}^n \times (0, T)$ if, for almost all $(x, t) \in \mathbb{R}^n \times (0, T)$,

- $G_{\theta}(x-y,t)\varphi(y)$ is integrable in \mathbb{R}^n with respect to $y \in \mathbb{R}^n$,
- $G_{\theta}(x-y,t-s)F_{p}(u(y,s))$ is integrable in $\mathbb{R}^{n} \times (0,t)$ with respect to $(y,s) \in \mathbb{R}^{n} \times (0,t)$,
- *u* satisfies

$$u(x,t) = [S_{\theta}(t)\varphi](x) + \int_0^t [S_{\theta}(t-s)F_p(u(s))](x) \, ds.$$

We are ready to state our main results.

Theorem 1.2. Let $\theta \in (0, 2]$, $p = p_{\theta} = 1 + \theta/n$, and $T_* \in (0, \infty)$. Then there exists $\epsilon > 0$ such that if $\varphi \in \mathfrak{L}^{1,\infty}_{ul}(\log \mathfrak{L})^{n/\theta}$ satisfies

 $\||\varphi\||_{1,\frac{n}{\alpha};T^{1/\theta}} \leq \epsilon \quad for some \ T \in (0, T_*],$

then problem (P) possesses a solution $u \in C((0,T) : \mathfrak{L}^{1,\infty}_{ul}(\log \mathfrak{L})^{n/\theta}) \cap L^{\infty}_{loc}((0,T) : L^{\infty})$ in $\mathbb{R}^n \times (0,T)$, with u satisfying

$$\sup_{t \in (0,T)} \| u(t) \|_{1,\frac{n}{\theta};T^{1/\theta}} + \sup_{t \in (0,T)} t^{\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t}\right) \right]^{\overline{\theta}} \| u(t) \|_{L^{\infty}} < \infty.$$
(1-6)

Furthermore, the solution u satisfies

$$\lim_{t \to +0} \|u(t) - S_{\theta}(t)\varphi\|_{\mathfrak{L}^{1,\infty}_{ul}(\log \mathfrak{L})^{\gamma}} = 0 \quad \text{for any } \gamma \in [0, n/\theta),$$

$$\lim_{t \to +0} u(t) = \varphi \quad \text{in the sense of distributions.}$$
(1-7)

We remark that Theorem 1.2 with $T_* = \infty$ does not hold since problem (P) possesses no global-in-time positive solutions (see [Sugitani 1975]). As a direct consequence of Theorem 1.2, we obtain assertion (4b).

Corollary 1.3. Let $\theta \in (0, 2]$ and $p = p_{\theta}$. Let φ_c be as in (1-1). Then there exists $\epsilon > 0$ such that if

$$|\varphi(x)| \le \epsilon \varphi_c(x) + K$$
, a.a. $x \in \mathbb{R}^n$,

for some $K \ge 0$, then problem (P) possesses a local-in-time solution.

Furthermore, as a consequence of Theorem 1.2, we have the following.

Theorem 1.4. Let $\theta \in (0, 2]$ and $p = p_{\theta}$. If $\varphi \in \mathfrak{L}_{ul}^{1,\infty}(\log \mathfrak{L})^{\alpha}$ for some $\alpha > n/\theta$, then problem (P) possesses a solution u in $\mathbb{R}^n \times (0, T)$ for some T > 0, with u satisfying (1-6) and (1-7).

The rest of this paper is organized as follows. In Section 2 we collect some properties of nonincreasing rearrangements of measurable functions and prove some lemmas in $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ and $\mathfrak{L}^{q,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$. Furthermore, we recall Hardy's inequalities and some properties of $S_{\theta}(t)\varphi$. In Section 3 we establish decay estimates of $S_{\theta}(t)\varphi$ in weak Zygmund-type spaces (see Proposition 3.1). Furthermore, we obtain decay estimates of $S_{\theta}(t)\varphi$ in $\mathfrak{L}^{q,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$ using Besicovitch's covering lemma. In Section 4 we apply the contraction mapping theorem in $\mathfrak{L}^{q,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$ to prove Theorems 1.2 and 1.4. In the Appendix we give two propositions on relations among the weak Zygmund-type spaces $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$, the Zygmund spaces $L^{q}(\log L)^{\alpha}$, and the weak Zygmund spaces $L^{q,\infty}(\log L)^{\alpha}$.

2. Preliminaries

In this section we introduce some notation and give some lemmas on our weak Zygmund-type spaces. Furthermore, we recall some lemmas on Hardy's inequalities. In all that follows, we will use C to denote generic positive constants and point out that C may take different values within a calculation. For any positive functions f_1 and f_2 in $(0, \infty)$, we write

$$f_1 \asymp f_2$$
 for $s > 0$ if $C^{-1} f_2(s) \le f_1(s) \le C f_2(s)$ for $s > 0$.

2.1. Weak Zygmund-type spaces. For any (Lebesgue) measurable function f in \mathbb{R}^n , we denote by μ_f the distribution function of f, that is,

$$\mu_f(\lambda) := |\{x : |f(x)| > \lambda\}| \quad \text{for } \lambda > 0.$$

We define the nonincreasing rearrangement f^* of f by

$$f^*(s) := \inf\{\lambda > 0 : \mu_f(\lambda) \le s\} \quad \text{for } s \in [0, \infty).$$

Here we adopt the convention $\inf \emptyset = \infty$. Then f^* is nonincreasing and right-continuous in $[0, \infty)$, and it has the following properties (see [Grafakos 2008, Proposition 1.4.5]):

$$(kf)^* = |k|f^*, \quad (|f|^q)^* = (f^*)^q, \quad \int_{\mathbb{R}^n} |f(x)|^q \, dx = \int_0^\infty f^*(s)^q \, ds, \quad f^*(0) = ||f||_{L^\infty}, \quad (2-1)$$

where $q \in (0, \infty)$ and $k \in \mathbb{R}$. We remark that if $E \in \mathcal{M}$ with $|E| < \infty$, then

$$(\chi_E)^*(s) = \chi_{[0,|E|)}(s) \text{ for } s \ge 0.$$
 (2-2)

Define

$$f^{**}(s) := \frac{1}{s} \int_0^s f^*(\tau) \, d\tau \quad \text{for } s \in (0, \infty).$$
(2-3)

Here we collect properties of f^* and f^{**} used in the paper.

(a) Since f^* is nonincreasing in $(0, \infty)$, it follows that

$$f^{**}(s) \ge f^{*}(s) \quad \text{for } s \in (0, \infty).$$
 (2-4)

(b) For any $q \in [1, \infty)$, Jensen's inequality together with (2-1) yields

$$(f^{**}(s))^q \le \frac{1}{s} \int_0^s f^*(\tau)^q \, d\tau = \frac{1}{s} \int_0^s (|f|^q)^*(\tau) \, d\tau = (|f|^q)^{**}(s) \quad \text{for } s \in (0,\infty).$$
(2-5)

(c) It follows from [Bennett and Sharpley 1988, Chapter 2, Proposition 3.3] that

$$f^{**}(s) = \frac{1}{s} \int_0^s f^*(\tau) \, d\tau = \frac{1}{s} \sup_{|E|=s} \int_E |f(x)| \, dx \quad \text{for } s \in (0,\infty).$$
(2-6)

(d) (O'Neil's inequality) For any $f, g \in L^1$, it follows from [O'Neil 1963, Lemma 1.6] that

$$(f * g)^{**}(s) \le \int_{s}^{\infty} f^{**}(\tau) g^{**}(\tau) d\tau \quad \text{for } s \in (0, \infty),$$
(2-7)

where

$$(f * g)(x) = \int_{\mathbb{R}^n} f(x - y)g(y) \, dy.$$

(e) For any $f_1, f_2 \in L^1_{loc}$, it follows from [O'Neil 1963, Theorem 3.3] that

$$(f_1 f_2)^{**}(s) \le \frac{1}{s} \int_0^s f_1^*(\tau) f_2^*(\tau) \, d\tau \quad \text{for } s \in (0, \infty).$$
(2-8)

Let $q \in [1, \infty)$ and $\alpha \ge 0$. For any L^1_{loc} -function f, by (1-2), (2-1), and (2-6), we have

$$\|f\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} = \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} s(|f|^{q})^{**}(s) \right\}^{\frac{1}{q}} = \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \int_{0}^{s} (|f|^{q})^{*}(\tau) \, d\tau \right\}^{\frac{1}{q}} = \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \int_{0}^{s} f^{*}(\tau)^{q} \, d\tau \right\}^{\frac{1}{q}}.$$
(2-9)

Furthermore, for any $E \in \mathcal{M}$ with $|E| < \infty$, it follows from (2-2) and (2-8) that

$$(f\chi_E)^{**}(s) = (f\chi_E\chi_E)^{**}(s) \le \frac{1}{s} \int_0^s (f\chi_E)^*(\tau)(\chi_E)^*(\tau) \, d\tau = \frac{1}{s} \int_0^{\min\{s,|E|\}} (f\chi_E)^*(\tau) \, d\tau.$$
(2-10)

For any $\beta \in [\alpha, \infty)$, since

the function
$$(0, \infty) \ni \tau \mapsto \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\alpha - \beta} \in \mathbb{R}$$
 is nondecreasing, (2-11)

by (2-1), (2-9), and (2-10), we have

$$\begin{split} \|f\chi_E\|_{\mathfrak{L}^{1,\infty}(\log\mathfrak{L})^{\alpha}} &= \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} s(f\chi_E)^{**}(s) \right\} \\ &\leq \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \int_{0}^{\min\{s,|E|\}} (f\chi_E)^{*}(\tau) \, d\tau \right\} \\ &= \sup_{00} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta} \int_{0}^{s} (f\chi_E)^{*}(\tau) \, d\tau \right\} \\ &= \left[\log\left(e + \frac{1}{|E|}\right) \right]^{\alpha-\beta} \|f\chi_E\|_{\mathfrak{L}^{1,\infty}(\log\mathfrak{L})^{\beta}}. \end{split}$$

In particular,

$$|||f|||_{1,\alpha;\rho} \le C \left[\log\left(e + \frac{1}{\rho}\right) \right]^{\alpha - \beta} |||f|||_{1,\beta;\rho}$$
(2-12)

for $f \in \mathfrak{L}^{1,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\beta}$, $0 \le \alpha \le \beta$, and $\rho > 0$. Here we show that $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ and $\mathfrak{L}^{q,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\alpha}$ are Banach spaces.

Lemma 2.1. For any $1 \le q < \infty$ and $\alpha \ge 0$, the weak Zygmund-type space $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ and the uniformly local weak Zygmund-type space $\mathfrak{L}^{q,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\alpha}$ are Banach spaces.

Proof. Let $1 \le q < \infty$ and $\alpha \ge 0$. It suffices to prove that $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ (resp. $\mathfrak{L}^{q,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$) is a complete metric space with the norm $\|\cdot\|_{\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}}$ (resp. $\|\cdot\|_{\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}}$). Let $\{f_n\}$ be a Cauchy sequence in $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$. It follows from (1-3) that $\{f_n\}$ is a Cauchy sequence in L^q , and hence there exists $f \in L^q$

1484

such that $f_n \to f$ as $n \to \infty$ in L^q . Since the Cauchy sequence $\{f_n\}$ is bounded in $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$, we observe from (1-2) that $f \in \mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$. It remains to prove that $f_n \to f$ as $n \to \infty$ in $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$. For this aim, we take a subsequence $\{f_{n_k}\}$ such that $\{f_{n_k}\}$ converges almost everywhere to f. Then Fatou's lemma gives us that

$$\|f - f_{n_j}\|_{\mathcal{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}} \leq \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \sup_{|E|=s} \liminf_{k \to \infty} \int_{E} |f_{n_k}(x) - f_{n_j}(x)|^q dx \right\}^{\frac{1}{q}} \\ \leq \liminf_{k \to \infty} \|f_{n_k} - f_{n_j}\|_{\mathcal{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}}.$$

This implies that f_{n_j} converges to f in $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$. Thus $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ is a complete metric space. Similarly, we see that $\mathfrak{L}^{q,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$ is a complete metric space.

Next, we prove two lemmas on our weak Zygmund-type spaces and uniformly local weak Zygmund-type spaces.

Lemma 2.2. Let $q_1, q_2 \in [1, \infty]$ and $\alpha_1, \alpha_2 \ge 0$ be such that

$$1 = \frac{1}{q_1} + \frac{1}{q_2}, \quad \alpha = \frac{\alpha_1}{q_1} + \frac{\alpha_2}{q_2}.$$
 (2-13)

Then

$$\|f_1 f_2\|_{\mathfrak{L}^{1,\infty}(\log \mathfrak{L})^{\alpha}} \le \|f_1\|_{\mathfrak{L}^{q_1,\infty}(\log \mathfrak{L})^{\alpha_1}} \|f_2\|_{\mathfrak{L}^{q_2,\infty}(\log \mathfrak{L})^{\alpha_2}}$$
(2-14)

for $f_1 \in \mathfrak{L}^{q_1,\infty}(\log \mathfrak{L})^{\alpha_1}$ and $f_2 \in \mathfrak{L}^{q_2,\infty}(\log \mathfrak{L})^{\alpha_2}$. Furthermore,

$$\|\tilde{f}_{1}\tilde{f}_{2}\|_{1,\alpha;\rho} \le \|\tilde{f}_{1}\|_{q_{1},\alpha_{1};\rho} \|\tilde{f}_{2}\|_{q_{2},\alpha_{2};\rho}$$
(2-15)

for $\tilde{f}_1 \in \mathfrak{L}^{q_1,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\alpha_1}, \ \tilde{f}_2 \in \mathfrak{L}^{q_2,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\alpha_2}, and \ \rho > 0.$

Proof. Let $q_1, q_2 \in [1, \infty)$ and $\alpha_1, \alpha_2 \ge 0$ satisfy (2-13). Let

$$f_1 \in \mathfrak{L}^{q_1,\infty}(\log \mathfrak{L})^{\alpha_1}$$
 and $f_2 \in \mathfrak{L}^{q_2,\infty}(\log \mathfrak{L})^{\alpha_2}$.

It follows from Hölder's inequality, (2-8), and (2-9) that

$$\begin{split} \|f_{1}f_{2}\|_{\mathfrak{L}^{1,\infty}(\log\mathfrak{L})^{\alpha}} &= \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} s(f_{1}f_{2})^{**}(s) \right\} \\ &\leq \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \int_{0}^{s} f_{1}^{*}(\tau) f_{2}^{*}(\tau) d\tau \right\} \\ &\leq \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \left(\int_{0}^{s} f_{1}^{*}(\tau)^{q_{1}} d\tau \right)^{\frac{1}{q_{1}}} \left(\int_{0}^{s} f_{2}^{*}(\tau)^{q_{2}} d\tau \right)^{\frac{1}{q_{2}}} \right\} \\ &\leq \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha_{1}} \int_{0}^{s} f_{1}^{*}(\tau)^{q_{1}} d\tau \right\}^{\frac{1}{q_{1}}} \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha_{2}} \int_{0}^{s} f_{2}^{*}(\tau)^{q_{2}} d\tau \right\}^{\frac{1}{q_{2}}} \\ &= \|f_{1}\|_{\mathfrak{L}^{q_{1},\infty}(\log\mathfrak{L})^{\alpha_{1}}} \|f_{2}\|_{\mathfrak{L}^{q_{2},\infty}(\log\mathfrak{L})^{\alpha_{2}}}. \end{split}$$

Thus (2-14) holds. Furthermore, for any $\tilde{f}_1 \in \mathfrak{L}^{q_1,\infty}_{ul}(\log \mathfrak{L})^{\alpha_1}$, $\tilde{f}_2 \in \mathfrak{L}^{q_2,\infty}_{ul}(\log \mathfrak{L})^{\alpha_2}$, and $\rho > 0$, by (2-14) we have

$$\begin{split} \|\|f_1 f_2\|\|_{1,\alpha;\rho} &= \sup_{x \in \mathbb{R}^n} \|f_1 f_2 \chi_{B(x,\rho)}\|_{\mathfrak{L}^{1,\infty}(\log \mathfrak{L})^{\alpha}} \\ &\leq \sup_{x \in \mathbb{R}^n} \{\|\tilde{f}_1 \chi_{B(x,\rho)}\|_{\mathfrak{L}^{q_1,\infty}(\log \mathfrak{L})^{\alpha_1}} \|\tilde{f}_2 \chi_{B(x,\rho)}\|_{\mathfrak{L}^{q_2,\infty}(\log \mathfrak{L})^{\alpha_2}} \} \\ &\leq \sup_{x \in \mathbb{R}^n} \|\tilde{f}_1 \chi_{B(x,\rho)}\|_{\mathfrak{L}^{q_1,\infty}(\log \mathfrak{L})^{\alpha_1}} \cdot \sup_{x \in \mathbb{R}^n} \|\tilde{f}_2 \chi_{B(x,\rho)}\|_{\mathfrak{L}^{q_2,\infty}(\log \mathfrak{L})^{\alpha_2}} \\ &= \|\|\tilde{f}_1\|_{q_1,\alpha_1;\rho} \|\|\tilde{f}_2\|_{q_2,\alpha_2;\rho}. \end{split}$$

Thus (2-15) holds, and Lemma 2.2 follows for $q_1, q_2 \in [1, \infty)$. If $q_1 = \infty$ or $q_2 = \infty$, the conclusion follows from (1-2).

Lemma 2.3. Let $q \in [1, \infty)$ and $\alpha \ge 0$. Then, for any r > 0 with $rq \ge 1$,

$$\begin{aligned} \||f|^r\|_{\mathcal{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}} &= \|f\|_{\mathcal{L}^{rq,\infty}(\log \mathfrak{L})^{\alpha}}^r \quad for \ f \in \mathcal{L}^{rq,\infty}(\log \mathfrak{L})^{\alpha}, \\ \|\||\tilde{f}|^r\|_{q,\alpha;\rho} &= \|\|\tilde{f}\||_{rq,\alpha;\rho}^r \quad for \ \tilde{f} \in \mathcal{L}^{q,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\alpha} \ and \ \rho > 0. \end{aligned}$$

Proof. It follows from (2-9) that

$$\begin{aligned} \||f|^r\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} &= \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \int_0^s ((|f|^r)^q)^*(\tau) \, d\tau \right\}^{\frac{1}{q}} \\ &= \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \int_0^s (|f|^{rq})^*(\tau) \, d\tau \right\}^{\frac{r}{rq}} \\ &= \|f\|_{\mathfrak{L}^{rq,\infty}(\log\mathfrak{L})^{\alpha}}^r \end{aligned}$$

for $f \in \mathcal{L}^{rq,\infty}(\log \mathcal{L})^{\alpha}$. Then

$$\|\||\tilde{f}|^{r}\|\|_{q,\alpha;\rho} = \sup_{x \in \mathbb{R}^{n}} \||\tilde{f}|^{r} \chi_{B(x,\rho)}\|_{\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}}$$
$$= \sup_{x \in \mathbb{R}^{n}} \||\tilde{f}|\chi_{B(x,\rho)}\|_{\mathfrak{L}^{rq,\infty}(\log \mathfrak{L})^{\alpha}}^{r}$$
$$= \||\tilde{f}\||_{rq,\alpha;\rho}^{r}$$

for $\tilde{f} \in \mathfrak{L}^{rq,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\alpha}$ and $\rho > 0$. Thus Lemma 2.3 follows.

2.2. *Hardy's inequalities.* We recall the following two lemmas on Hardy's inequality. (See [Muckenhoupt 1972, Theorems 1 and 2].) Throughout this paper, for any $q \in [1, \infty]$, we denote by q' the Hölder conjugate of q, that is, q' = q/(q-1) if $q \in (1, \infty)$, $q' = \infty$ if q = 1, and q' = 1 if $q = \infty$.

Lemma 2.4. Let $q \in [1, \infty]$. Let U and V be locally integrable functions in $[0, \infty)$. Then there exists C > 0 such that

$$\|U\widetilde{F}\|_{L^{q}((0,\infty))} \le C \|Vf\|_{L^{q}((0,\infty))}, \quad \text{with } \widetilde{F}(s) := \int_{0}^{s} f(\tau) \, d\tau,$$

holds for all locally integrable functions f in $[0, \infty)$ if and only if

$$\sup_{s>0} \{ \|U\|_{L^q((s,\infty))} \|V^{-1}\|_{L^{q'}((0,s))} \} < \infty.$$

1486
Lemma 2.5. Let $q \in [1, \infty]$. Let U and V be locally integrable functions in $[0, \infty)$. Then there exists C > 0 such that

$$\|U\widehat{F}\|_{L^{q}((0,\infty))} \le C \|Vf\|_{L^{q}((0,\infty))}, \quad \text{with } \widehat{F}(s) := \int_{s}^{\infty} f(\tau) \, d\tau,$$

holds for all locally integrable functions f in $(0, \infty)$, with $f \in L^1((1, \infty))$, if and only if

$$\sup_{s>0}\{\|U\|_{L^{q}((0,s))}\|V^{-1}\|_{L^{q'}((s,\infty))}\}<\infty.$$

2.3. *Fundamental solutions.* Let $\theta \in (0, 2]$. Let G_{θ} be the fundamental solution to the fractional heat equation

$$\partial_t v + (-\Delta)^{\frac{\theta}{2}} v = 0 \quad \text{in } \mathbb{R}^n \times (0, \infty)$$

The function G_{θ} is positive and smooth in $\mathbb{R}^n \times (0, \infty)$, and it satisfies

$$G_{\theta}(x,t) = (4\pi t)^{-\frac{n}{2}} \exp\left(-\frac{|x|^2}{4t}\right) \le Ch_{\theta,t}(x) \quad \text{if } \theta = 2,$$

$$G_{\theta}(x,t) \asymp h_{\theta,t}(x) \qquad \text{if } 0 < \theta < 2$$
(2-16)

for $(x, t) \in \mathbb{R}^n \times (0, \infty)$, where

$$h_{\theta,t}(x) := t^{-\frac{n}{\theta}} (1 + t^{-\frac{1}{\theta}} |x|)^{-n-\theta}.$$
(2-17)

Furthermore,

- $G_{\theta}(x,t) = t^{-\frac{n}{\theta}} G_{\theta}(t^{-\frac{1}{\theta}}x,1), \ \int_{\mathbb{R}^n} G_{\theta}(x,t) \, dx = 1,$
- $G_{\theta}(\cdot, 1)$ is radially symmetric and $G_{\theta}(x, 1) \leq G_{\theta}(y, 1)$ if $|x| \geq |y|$,
- $G_{\theta}(x,t) = \int_{\mathbb{R}^n} G_{\theta}(x-y,t-s) G_{\theta}(y,s) \, dy$

for $x, y \in \mathbb{R}^n$ and 0 < s < t (see, e.g., [Bogdan and Jakubowski 2007; Brandolese and Karch 2008; Sugitani 1975]), and

$$\lim_{t \to +0} \|S_{\theta}(t)\eta - \eta\|_{L^{\infty}} = 0 \quad \text{for } \eta \in C_0(\mathbb{R}^n).$$
(2-18)

In addition, it follows from Young's inequality that

$$\|S_{\theta}(t)\eta\|_{L^{q}} \le Ct^{-\frac{n}{\theta}\left(\frac{1}{r} - \frac{1}{q}\right)} \|\eta\|_{L^{r}}$$
(2-19)

for $\eta \in L^r$, $1 \le r \le q \le \infty$, and t > 0.

3. Decay estimates of $S_{\theta}(t)\varphi$

In this section we obtain decay estimates of $S_{\theta}(t)\varphi$ in our weak Zygmund-type spaces and uniformly local weak Zygmund-type spaces. For simplicity we write $g_t := G_{\theta}(\cdot, t)$ and $h_t := h_{\theta,t}$.

Proposition 3.1. Let $\theta \in (0, 2]$, $1 \le r \le q \le \infty$, and α , $\beta \ge 0$. Assume that $\alpha \le \beta$ if r = q. Then there exists C > 0 such that

$$\|S_{\theta}(t)\varphi\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}} \leq Ct^{-\frac{n}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right)\right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} \|\varphi\|_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}}$$

for $\varphi \in \mathfrak{L}^{r,\infty}(\log \mathfrak{L})^{\alpha}$ and t > 0.

Before starting the proof, we recall the following relations on logarithmic functions: for any fixed L > 1 and k > 0,

$$\log\left(e + \frac{1}{s}\right) \asymp \log\left(L + \frac{1}{s}\right) \asymp \log\left(e + \frac{k}{s}\right) \asymp \log\left(e + \frac{1}{s^k}\right) \quad \text{for } s > 0.$$
(3-1)

Furthermore, we have the following results.

Lemma 3.2. (1) Let q > -1 and $\alpha \in \mathbb{R}$. Then there exists $C_1 > 0$ such that

$$\int_0^s \tau^q \left[\log\left(e + \frac{1}{\tau}\right) \right]^\alpha d\tau \le C_1 s^{q+1} \left[\log\left(e + \frac{1}{s}\right) \right]^\alpha \quad \text{for } s > 0.$$

(2) Let S > 0 and $\alpha < -1$. Then there exists $C_2 > 0$ such that

$$\int_0^s \tau^{-1} \left[\log\left(e + \frac{1}{\tau}\right) \right]^\alpha d\tau \le C_2 \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha+1} \quad \text{for } s \in (0, S).$$

(3) Let q < -1 and $\alpha \in \mathbb{R}$. Then there exists $C_3 > 0$ such that

$$\int_{s}^{\infty} \tau^{q} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\alpha} d\tau \leq C_{3} s^{q+1} \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \quad \text{for } s > 0.$$

Proof. We prove assertion (1). Let $\delta > 0$ be such that $q - \delta > -1$. Then there exists $L \in [e, \infty)$ such that

the function
$$(0, \infty) \ni \tau \mapsto \tau^{\delta} \left[\log \left(L + \frac{1}{\tau} \right) \right]^{\alpha}$$
 is nondecreasing. (3-2)

This together with (3-1) implies that

$$\begin{split} \int_0^s \tau^q \bigg[\log \bigg(e + \frac{1}{\tau} \bigg) \bigg]^{\alpha} d\tau &\leq C \int_0^s \tau^{q-\delta} \cdot \tau^{\delta} \bigg[\log \bigg(L + \frac{1}{\tau} \bigg) \bigg]^{\alpha} d\tau \\ &\leq C s^{\delta} \bigg[\log \bigg(L + \frac{1}{s} \bigg) \bigg]^{\alpha} \int_0^s \tau^{q-\delta} d\tau \leq C s^{q+1} \bigg[\log \bigg(e + \frac{1}{s} \bigg) \bigg]^{\alpha} \end{split}$$

for s > 0. Thus assertion (1) follows.

We prove assertion (2). Let S > 0. It follows that

$$\int_{0}^{s} \tau^{-1} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\alpha} d\tau \le C \int_{0}^{s} \tau^{-1} |\log \tau|^{\alpha} d\tau \le C |\log s|^{\alpha+1} \le C \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha+1}$$

for $s \in (0, \frac{1}{2})$. If $S \ge \frac{1}{2}$, then

$$\int_0^s \tau^{-1} \left[\log\left(e + \frac{1}{\tau}\right) \right]^\alpha d\tau \le \int_{\frac{1}{4}}^s \tau^{-1} \left[\log\left(e + \frac{1}{\tau}\right) \right]^\alpha d\tau + C \le C \le C \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha+1}$$

for $s \in \left[\frac{1}{2}, S\right)$. Thus assertion (2) follows.

It remains to prove assertion (3). Let $\delta' > 0$ be such that $q + \delta' < -1$. Then there exists $L' \in [e, \infty)$ such that

the function
$$(0, \infty) \ni \tau \mapsto \tau^{-\delta'} \left[\log \left(L' + \frac{1}{\tau} \right) \right]^{\alpha}$$
 is nonincreasing

This together with (3-1) implies that

$$\int_{s}^{\infty} \tau^{q} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\alpha} d\tau \leq C \int_{s}^{\infty} \tau^{q+\delta'} \cdot \tau^{-\delta'} \left[\log\left(L' + \frac{1}{\tau}\right) \right]^{\alpha} d\tau$$
$$\leq C s^{-\delta'} \left[\log\left(L' + \frac{1}{s}\right) \right]^{\alpha} \int_{s}^{\infty} \tau^{q+\delta'} d\tau \leq C s^{q+1} \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha}$$
$$s > 0, \text{ Thus assertion (3) follows.} \qquad \Box$$

for s > 0. Thus assertion (3) follows.

Next, we prepare the following lemma on h_t^* , where $h_t = h_{\theta,t}$ is as in (2-17).

Lemma 3.3. Let $1 \le r \le q < \infty$ and $\gamma \in \mathbb{R}$. Assume that $\gamma \ge 0$ if r = q. Then there exists C > 0 such that

$$\int_0^\infty \tau^{q\left(1-\frac{1}{r}\right)} \left[\log\left(e+\frac{1}{\tau}\right) \right]^\gamma (h_t^*(\tau))^q \, d\,\tau \le C t^{-\frac{nq}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right) \right]^\gamma \tag{3-3}$$

for t > 0*.*

Proof. It follows from (2-17) that

$$(h_t)^*(s) = h_t((\omega_n^{-1}s)^{\frac{1}{n}}e_1) \le Ct^{-\frac{n}{\theta}}(1 + t^{-\frac{1}{\theta}}s^{\frac{1}{n}})^{-n-\theta}$$

for $s \in [0, \infty)$ and $t \in (0, \infty)$, where ω_n is the volume of the *n*-dimensional unit ball B(0, 1) and $e_1 := (1, 0, \dots, 0) \in \mathbb{R}^n$. Then

$$I := \int_{0}^{\infty} \tau^{q(1-\frac{1}{r})} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\gamma} (h_{t}^{*}(\tau))^{q} d\tau$$

$$\leq Ct^{-\frac{nq}{\theta}} \int_{0}^{\infty} \tau^{q(1-\frac{1}{r})} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\gamma} (1 + t^{-\frac{1}{\theta}} \tau^{\frac{1}{n}})^{-q(n+\theta)} d\tau$$

$$\leq Ct^{-\frac{nq}{\theta}(\frac{1}{r} - \frac{1}{q})} \int_{0}^{\infty} \xi^{nq(1-\frac{1}{r})+n-1} (1+\xi)^{-q(n+\theta)} \left[\log\left(e + \frac{1}{(t^{1/\theta}\xi)^{n}}\right) \right]^{\gamma} d\xi \qquad (3-4)$$

for t > 0.

We first consider the case of $\gamma \ge 0$. It follows from (3-1) that

$$\left[\log\left(e + \frac{1}{(t^{1/\theta}\xi)^n}\right)\right]^{\gamma} \le C \left[\log\left(e + \frac{1}{t^{1/\theta}\xi}\right)\right]^{\gamma}$$
$$\le C \left[\log\left(e + \frac{1}{t^{1/\theta}}\right) + \log\left(e + \frac{1}{\xi}\right)\right]^{\gamma}$$
$$\le C \left[\log\left(e + \frac{1}{t}\right)\right]^{\gamma} + C \left[\log\left(e + \frac{1}{\xi}\right)\right]^{\gamma}$$
(3-5)

for t > 0 and $\xi \in (0, \frac{1}{2})$. Similarly, by (3-1), we have

$$\left[\log\left(e + \frac{1}{(t^{1/\theta}\xi)^n}\right)\right]^{\gamma} \le \left[\log\left(e + \frac{2^n}{(t^{1/\theta})^n}\right)\right]^{\gamma} \le C\left[\log\left(e + \frac{1}{t}\right)\right]^{\gamma}$$
(3-6)

for t > 0 and $\xi \in \left[\frac{1}{2}, \infty\right)$. Since

$$nq\left(1-\frac{1}{r}\right) + n - 1 - q(n+\theta) = -\frac{nq}{r} + n - 1 - q\theta = -nq\left(\frac{1}{r} - \frac{1}{q}\right) - 1 - q\theta < -1,$$
(3-7)

by Lemma 3.2, (3-4), (3-5), and (3-6), we obtain

$$\begin{split} I &\leq Ct^{-\frac{nq}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \int_{0}^{\frac{1}{2}} \xi^{nq\left(1-\frac{1}{r}\right)+n-1} \left(\left[\log\left(e+\frac{1}{t}\right) \right]^{\gamma} + \left[\log\left(e+\frac{1}{\xi}\right) \right]^{\gamma} \right) d\xi \\ &\quad + Ct^{-\frac{nq}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \int_{\frac{1}{2}}^{\infty} \xi^{nq\left(1-\frac{1}{r}\right)+n-1} (1+\xi)^{-q\left(n+\theta\right)} \left[\log\left(e+\frac{1}{t}\right) \right]^{\gamma} d\xi \\ &\leq Ct^{-\frac{nq}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left(1 + \left[\log\left(e+\frac{1}{t}\right) \right]^{\gamma} \right) \\ &\leq Ct^{-\frac{nq}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right) \right]^{\gamma} \end{split}$$

for t > 0. This implies (3-3) in the case of $\gamma \ge 0$.

Consider the case of $\gamma < 0$. Then, by (3-1), we have

$$\left[\log\left(e + \frac{1}{(t^{1/\theta}\xi)^n}\right)\right]^{\gamma} \le \left[\log\left(e + \frac{2^n}{(t^{1/\theta})^n}\right)\right]^{\gamma} \le C\left[\log\left(e + \frac{1}{t}\right)\right]^{\gamma}$$
(3-8)

for t > 0 and $\xi \in (0, \frac{1}{2})$. Let $0 < \delta < \theta q/|\gamma|$. We find $L \in [e, \infty)$ such that the function f in $(0, \infty)$ defined by

$$f(z) := z^{\delta} \log \left(L + \frac{1}{z^n} \right)$$

is nondecreasing in $(0, \infty)$. Since $\gamma < 0$, by (3-1), we obtain

$$\begin{split} \left[\log \left(e + \frac{1}{(t^{1/\theta}\xi)^n} \right) \right]^{\gamma} &\leq C \left[\log \left(L + \frac{1}{(t^{1/\theta}\xi)^n} \right) \right]^{\gamma} = C \left[z^{-\delta\gamma} f(z)^{\gamma} \right]_{z=t^{1/\theta}\xi} \\ &\leq C \left(t^{\frac{1}{\theta}}\xi \right)^{-\delta\gamma} f(z)^{\gamma} |_{z=\frac{t^{1/\theta}}{2}} \leq C \xi^{-\delta\gamma} \left[\log \left(e + \frac{2^n}{t^{n/\theta}} \right) \right]^{\gamma} \\ &\leq C \xi^{-\delta\gamma} \left[\log \left(e + \frac{1}{t} \right) \right]^{\gamma} \end{split}$$

for t > 0 and $\xi \in \left[\frac{1}{2}, \infty\right)$. This together with (3-7) and (3-8) implies that

$$\begin{split} I &\leq Ct^{-\frac{nq}{\theta}\left(\frac{1}{r} - \frac{1}{q}\right)} \int_{0}^{\frac{1}{2}} \xi^{nq\left(1 - \frac{1}{r}\right) + n - 1} \left[\log\left(e + \frac{1}{t}\right) \right]^{\gamma} d\xi \\ &+ Ct^{-\frac{nq}{\theta}\left(\frac{1}{r} - \frac{1}{q}\right)} \int_{\frac{1}{2}}^{\infty} \xi^{nq\left(1 - \frac{1}{r}\right) + n - 1 - \delta\gamma} (1 + \xi)^{-q(n+\theta)} \left[\log\left(e + \frac{1}{t}\right) \right]^{\gamma} d\xi \\ &\leq Ct^{-\frac{nq}{\theta}\left(\frac{1}{r} - \frac{1}{q}\right)} \left[\log\left(e + \frac{1}{t}\right) \right]^{\gamma} \end{split}$$

for t > 0. This implies (3-3) in the case of $\gamma < 0$. Thus Lemma 3.3 follows.

Proof of Proposition 3.1. The proof is divided into the following three cases:

$$1 \le r < q < \infty, \quad 1 \le r = q < \infty, \quad 1 \le r \le q = \infty.$$

<u>Step 1</u>. Consider the case of $1 \le r < q < \infty$. It follows from (2-4), (2-7), (2-9), and (2-16) that

$$\begin{split} \|S_{\theta}(t)\varphi\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}}^{q} &= \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta} \int_{0}^{s} \left((S_{\theta}(t)\varphi)^{*}(\tau) \right)^{q} d\tau \right\} \\ &\leq \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta} \int_{0}^{s} \left((S_{\theta}(t)\varphi)^{**}(\tau) \right)^{q} d\tau \right\} \\ &\leq \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta} \int_{0}^{s} \left(\int_{\tau}^{\infty} g_{t}^{**}(\eta)\varphi^{**}(\eta) d\eta \right)^{q} d\tau \right\} \\ &\leq C \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta} \int_{0}^{s} \left(\int_{\tau}^{\infty} h_{t}^{**}(\eta)\varphi^{**}(\eta) d\eta \right)^{q} d\tau \right\} \end{split}$$

for t > 0. Furthermore, thanks to (2-11), we have

$$\|S_{\theta}(t)\varphi\|^{q}_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}} \leq C \int_{0}^{\infty} \left(\left[\log\left(e + \frac{1}{\tau}\right) \right]^{\frac{\beta}{q}} \int_{\tau}^{\infty} h_{t}^{**}(\eta)\varphi^{**}(\eta) \, d\eta \right)^{q} d\tau$$
(3-9)

for t > 0. On the other hand, set

$$U(\tau) := \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\frac{\beta}{q}}, \quad V(\tau) := \tau \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\frac{\beta}{q}}$$

for $\tau > 0$. It follows from Lemma 3.2(1) and (3) that

$$\begin{split} \sup_{s>0} \left(\int_0^s |U(\tau)|^q \, d\,\tau \right)^{\frac{1}{q}} \left(\int_s^\infty |V(\tau)|^{-q'} \, d\,\tau \right)^{\frac{1}{q'}} \\ & \leq \sup_{s>0} \left\{ C \, s^{\frac{1}{q}} \left[\log\left(e + \frac{1}{s}\right) \right]^{\frac{\beta}{q}} \cdot C \, s^{-1 + \frac{1}{q'}} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\frac{\beta}{q}} \right\} < \infty. \end{split}$$

Then, by Lemma 2.5, (2-3), and (3-9), we have

$$\begin{split} \|S_{\theta}(t)\varphi\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}}^{q} &\leq C \int_{0}^{\infty} \left(\tau \left[\log\left(e+\frac{1}{\tau}\right)\right]^{\frac{\beta}{q}} h_{t}^{**}(\tau)\varphi^{**}(\tau)\right)^{q} d\tau \\ &\leq C \sup_{s>0} \left\{ \left[\log\left(e+\frac{1}{s}\right)\right]^{\alpha} s(\varphi^{**}(s))^{r} \right\}^{\frac{q}{r}} \int_{0}^{\infty} \left(\tau^{1-\frac{1}{r}} \left[\log\left(e+\frac{1}{\tau}\right)\right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} h_{t}^{**}(\tau)\right)^{q} d\tau \end{split}$$

for t > 0. This together with (2-5) and (2-9) implies that

$$\|S_{\theta}(t)\varphi\|^{q}_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}} \leq C \|\varphi\|^{q}_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}} \int_{0}^{\infty} \left(\tau^{-\frac{1}{r}} \left[\log\left(e+\frac{1}{\tau}\right)\right]^{\gamma} \int_{0}^{\tau} h^{*}_{t}(\xi) \, d\xi\right)^{q} d\tau \qquad (3-10)$$

for t > 0, where

$$\gamma := -\frac{\alpha}{r} + \frac{\beta}{q}.$$

Set

$$\widetilde{U}(\tau) = \tau^{-\frac{1}{r}} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\gamma}, \quad \widetilde{V}(\tau) = \tau^{1-\frac{1}{r}} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\gamma}.$$

Since q > r and q' < r', by Lemma 3.2(1) and (3), we have

$$\begin{split} \sup_{s>0} \left(\int_{s}^{\infty} |\tilde{U}(\tau)|^{q} d\tau \right)^{\frac{1}{q}} \left(\int_{0}^{s} |\tilde{V}(\tau)|^{-q'} d\tau \right)^{\frac{1}{q'}} \\ &= \sup_{s>0} \left(\int_{s}^{\infty} \tau^{-\frac{q}{r}} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{q\gamma} d\tau \right)^{\frac{1}{q}} \left(\int_{0}^{s} \tau^{-\frac{q'}{r'}} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{-q'\gamma} d\tau \right)^{\frac{1}{q'}} \\ &\leq \sup_{s>0} \left\{ C s^{\frac{1}{q} - \frac{1}{r}} \left[\log\left(e + \frac{1}{s}\right) \right]^{\gamma} \cdot C s^{\frac{1}{q'} - \frac{1}{r'}} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\gamma} \right\} < \infty. \end{split}$$
(3-11)

Applying Lemma 2.4 to (3-10), by (3-11), we obtain

$$\|S_{\theta}(t)\varphi\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}}^{q} \leq C \|\varphi\|_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}}^{q} \int_{0}^{\infty} \left(\tau^{1-\frac{1}{r}} \left[\log\left(e+\frac{1}{\tau}\right)\right]^{\gamma} h_{t}^{*}(\tau)\right)^{q} d\tau$$

for t > 0. This together with Lemma 3.3 implies that

$$\|S_{\theta}(t)\varphi\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}}^{q} \leq Ct^{-\frac{nq}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right)\right]^{q\gamma} \|\varphi\|_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}}^{q}$$

for t > 0. Thus Proposition 3.1 follows in the case of $1 \le r < q < \infty$.

<u>Step 2</u>. Consider the case of $1 \le r = q < \infty$. It follows from Hölder's inequality and (2-16) that

$$\begin{split} |[S_{\theta}(t)\varphi](x)|^{r} &\leq C \left(\int_{\mathbb{R}^{n}} |h_{t}(x-y)| |\varphi(y)| \, dy \right)^{r} \\ &\leq C \left(\int_{\mathbb{R}^{n}} |h_{t}(x-y)| \, dy \right)^{r-1} \int_{\mathbb{R}^{n}} |h_{t}(x-y)| |\varphi(y)|^{r} \, dy \\ &\leq C \int_{\mathbb{R}^{n}} |h_{t}(x-y)| |\varphi(y)|^{r} \, dy. \end{split}$$

Then it follows from (2-7) and (2-9) that

$$\|S_{\theta}(t)\varphi\|_{\mathcal{L}^{r,\infty}(\log \mathfrak{L})^{\beta}}^{r} = \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta} s(|S_{\theta}(t)\varphi|^{r})^{**}(s) \right\}$$
$$\leq C \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta} s \int_{s}^{\infty} (h_{t})^{**}(\tau) (|\varphi|^{r})^{**}(\tau) \, d\tau \right\}$$
(3-12)

for t > 0. Set

$$\widehat{U}(r) = r \left[\log \left(e + \frac{1}{r} \right) \right]^{\beta}, \quad \widehat{V}(r) = r^2 \left[\log \left(e + \frac{1}{r} \right) \right]^{\beta}.$$

Similarly to (3-2), we find $L \in [e, \infty)$ such that

the function
$$(0, \infty) \ni r \mapsto r \left[\log \left(L + \frac{1}{r} \right) \right]^{\beta}$$
 is nondecreasing.

Then, by (3-1), we have

$$\begin{split} \|\widehat{U}\|_{L^{\infty}(0,s)} &\leq C \sup_{r \in (0,s)} \left\{ r \left[\log \left(L + \frac{1}{r} \right) \right]^{\beta} \right\} \\ &\leq C s \left[\log \left(L + \frac{1}{s} \right) \right]^{\beta} \\ &\leq C s \left[\log \left(e + \frac{1}{s} \right) \right]^{\beta} \end{split}$$

for s > 0. This together with Lemma 3.2(3) implies that

$$\sup_{s>0} \left\{ \|\hat{U}\|_{L^{\infty}((0,s))} \int_{s}^{\infty} |\hat{V}(\tau)|^{-1} d\tau \right\} \le \sup_{s>0} \left\{ Cs \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta} \cdot Cs^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\beta} \right\} < \infty.$$
(3-13)

Applying Lemma 2.5 with $q = \infty$, by (2-9), (3-12) and (3-13), we obtain

$$\begin{split} \|S_{\theta}(t)\varphi\|_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\beta}}^{r} &\leq C \sup_{s>0} \left\{ s^{2} \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta} (h_{t})^{**}(s) (|\varphi|^{r})^{**}(s) \right\} \\ &\leq C \sup_{s>0} \left\{ s \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta-\alpha} (h_{t})^{**}(s) \right\} \cdot \sup_{s>0} \left\{ s \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} (|\varphi|^{r})^{**}(s) \right\} \\ &= C \|h_{t}\|_{\mathfrak{L}^{1,\infty}(\log\mathfrak{L})^{\beta-\alpha}} \|\varphi\|_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}}^{r}$$
(3-14)

for t > 0. Furthermore, since $\alpha \le \beta$, by Lemma 3.3, (2-9), and (2-11), we have

$$\|h_t\|_{\mathfrak{L}^{1,\infty}(\log\mathfrak{L})^{\beta-\alpha}} = \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\beta-\alpha} \int_0^s (h_t)^*(\tau) \, d\tau \right\}$$
$$\leq \int_0^\infty \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\beta-\alpha} (h_t)^*(\tau) \, d\tau$$
$$\leq C \left[\log\left(e + \frac{1}{t}\right) \right]^{\beta-\alpha}$$

for t > 0. This together with (3-14) implies that

$$\|S_{\theta}(t)\varphi\|_{\mathcal{L}^{r,\infty}(\log \mathfrak{L})^{\beta}}^{r} \leq C \left[\log\left(e+\frac{1}{t}\right)\right]^{\beta-\alpha} \|\varphi\|_{\mathcal{L}^{r,\infty}(\log \mathfrak{L})^{\alpha}}^{r} \quad \text{for } t > 0$$

Thus Proposition 3.1 follows in the case of $1 \le r = q < \infty$.

<u>Step 3</u>. It remains to consider the case of $1 \le r \le q = \infty$. If $r = q = \infty$, then it follows from (2-16) that

$$\|S_{\theta}(t)\varphi\|_{L^{\infty}} \leq C \|\varphi\|_{L^{\infty}} \int_{\mathbb{R}^n} h_t(y) \, dy \leq C \|\varphi\|_{L^{\infty}}$$

for t > 0, and Proposition 3.1 follows. On the other hand, in the case of $1 \le r < q = \infty$, let $\tilde{q} \in (r, \infty)$. Then, by Proposition 3.1 with $q = \tilde{q} > r$, (1-3), (2-19), and (3-1), we have

$$\begin{split} \|S_{\theta}(t)\varphi\|_{L^{\infty}} &= \left\|S_{\theta}\left(\frac{t}{2}\right)S_{\theta}\left(\frac{t}{2}\right)\varphi\right\|_{L^{\infty}} \leq Ct^{-\frac{n}{\theta\bar{q}}} \left\|S_{\theta}\left(\frac{t}{2}\right)\varphi\right\|_{L^{\tilde{q}}} = C^{-\frac{n}{\theta\bar{q}}} \left\|S_{\theta}\left(\frac{t}{2}\right)\varphi\right\|_{\mathfrak{L}^{\tilde{q},\infty}(\log\mathfrak{L})^{0}} \\ &\leq Ct^{-\frac{n}{\theta\bar{q}}} \cdot Ct^{-\frac{n}{\theta}\left(\frac{1}{r}-\frac{1}{\bar{q}}\right)} \left[\log\left(e+\frac{2}{t}\right)\right]^{-\frac{\alpha}{r}} \|\varphi\|_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}} \\ &= Ct^{-\frac{n}{\theta r}} \left[\log\left(e+\frac{1}{t}\right)\right]^{-\frac{\alpha}{r}} \|\varphi\|_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}} \end{split}$$

for t > 0. Thus Proposition 3.1 follows in the case of $1 \le r < q = \infty$. The proof of Proposition 3.1 is complete.

Furthermore, by Proposition 3.1, we employ the arguments in the proof of [Hisa and Ishige 2018, Lemma 2.1] to obtain decay estimates of $S_{\theta}(t)\varphi$ in uniformly local weak Zygmund-type spaces.

Proposition 3.4. Let $\theta \in (0, 2]$, $1 \le r \le q \le \infty$, and $\alpha, \beta \ge 0$. Assume that $\alpha \le \beta$ if r = q. There exists C > 0 such that, for any T > 0,

$$\|S_{\theta}(t)\varphi\|\|_{q,\beta;T^{1/\theta}} \le Ct^{-\frac{n}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right)\right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} \|\|\varphi\|\|_{r,\alpha;T^{1/\theta}}$$
(3-15)

for $\varphi \in \mathfrak{L}^{r,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\alpha}$ and $t \in (0, T]$.

Proof. We first consider the case of $\theta \in (0, 2)$. It suffices to prove

$$t^{\frac{n}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right)\right]^{\frac{\alpha}{r}-\frac{\beta}{q}} \|\chi_{B(z,T^{1/\theta})}S_{\theta}(t)\varphi\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}} \le C \, \|\varphi\|\|_{r,\alpha;T^{1/\theta}} \tag{3-16}$$

for $z \in \mathbb{R}^n$ and $0 < t \le T$. For the proof, by translating if necessary, we have only to consider the case of z = 0.

By Besicovitch's covering lemma, we can find an integer m depending only on n and a set

$$\{x_{k,i}\}_{k=1,\ldots,m,\,i\in\mathbb{N}}\subset\mathbb{R}^n\setminus B(0,\,10T^{\frac{1}{\theta}})$$

such that

$$B_{k,i} \cap B_{k,j} = \emptyset \text{ if } i \neq j \text{ and } \mathbb{R}^n \setminus B(0, 10T^{\frac{1}{\theta}}) \subset \bigcup_{k=1}^m \bigcup_{i=1}^\infty B_{k,i},$$
(3-17)

where $B_{k,i} := \overline{B(x_{k,i}, T^{1/\theta})}$. Then

$$|[S_{\theta}(t)\varphi](x)| \le |u_0(x,t)| + \sum_{k=1}^m \sum_{i=1}^\infty |u_{k,i}(x,t)|, \quad (x,t) \in \mathbb{R}^n \times (0,T),$$
(3-18)

where

$$u_0(x,t) := [S_{\theta}(t)(\varphi \chi_{B(0,10T^{1/\theta})})](x), \quad u_{k,i}(x,t) := [S_{\theta}(t)(\varphi \chi_{B_{k,i}})](x).$$

By Proposition 3.1 and (1-4), we have

 $\|u_0(t)\chi_{\boldsymbol{B}(0,T^{1/\theta})}\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}} \leq \|u_0(t)\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}}$

$$\leq Ct^{-\frac{n}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right) \right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} \|\varphi\chi_{B(0,10T^{1/\theta})}\|_{\mathcal{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}}$$

$$\leq Ct^{-\frac{n}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right) \right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} \|\varphi\|_{r,\alpha;10T^{1/\theta}}$$

$$\leq Ct^{-\frac{n}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right) \right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} \|\varphi\|_{r,\alpha;T^{1/\theta}}$$

$$(3-19)$$

for $t \in (0, T]$.

Let k = 1, ..., m and $i \in \mathbb{N}$. By (2-16), we have

$$|u_{k,i}(x,t)| \le C \int_{B(x_{k,i},T^{1/\theta})} h_t(x-y) |\varphi(y)| \, dy = C \int_{\mathbb{R}^n} h_t(x-z-x_{k,i}) \varphi_{k,i}(z) \, dz \tag{3-20}$$

for $(x, t) \in \mathbb{R}^n \times (0, \infty)$, where $\varphi_{k,i}(x) = |\varphi(x + x_{k,i})| \chi_{B(0,T^{1/\theta})}$. Since $|x_{k,i}| \ge 10T^{1/\theta}$, it follows that

$$(1+T^{-\frac{1}{\theta}}|x_{k,i}|)(1+t^{-\frac{1}{\theta}}|x-z|) = 1+T^{-\frac{1}{\theta}}|x_{k,i}|+t^{-\frac{1}{\theta}}|x-z|+t^{-\frac{1}{\theta}}T^{-\frac{1}{\theta}}|x_{k,i}||x-z|$$

$$\leq 1+3t^{-\frac{1}{\theta}}|x_{k,i}|+t^{-\frac{1}{\theta}}|x-z|$$

$$= 1+4t^{-\frac{1}{\theta}}(|x_{k,i}|-|x-z|)-t^{-\frac{1}{\theta}}|x_{k,i}|+5t^{-\frac{1}{\theta}}|x-z|$$

$$\leq 4(1+t^{-\frac{1}{\theta}}(|x_{k,i}|-|x-z|)) \leq 4(1+t^{-\frac{1}{\theta}}|x-z-x_{k,i}|)$$

for $x, z \in B(0, T^{1/\theta})$ and $t \in (0, T)$. This together with (2-16) implies that

$$h_t(x - z - x_{k,i}) \le C t^{-\frac{n}{\theta}} (1 + T^{-\frac{1}{\theta}} |x_{k,i}|)^{-n-\theta} (1 + t^{-\frac{1}{\theta}} |x - z|)^{-n-\theta} \le C (1 + T^{-\frac{1}{\theta}} |x_{k,i}|)^{-n-\theta} g_t(x - z)$$
(3-21)

for $x, z \in B(0, T^{1/\theta})$ and $t \in (0, T)$. We observe from (3-20) and (3-21) that

$$|u_{k,i}(x,t)| \le C(1+T^{-\frac{1}{\theta}}|x_{k,i}|)^{-n-\theta}[S_{\theta}(t)\varphi_{k,i}](x)$$

for $x \in B(0, T^{1/\theta})$ and $t \in (0, T)$. Then, by Proposition 3.1, we obtain $\|u_{k,i}(t)\chi_{B(0,T^{1/\theta})}\|_{\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\beta}}$

$$\leq C(1+T^{-\frac{1}{\theta}}|x_{k,i}|)^{-n-\theta} \|S_{\theta}(t)\varphi_{k,i}\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}}$$

$$\leq C(1+T^{-\frac{1}{\theta}}|x_{k,i}|)^{-n-\theta}t^{-\frac{n}{\theta}(\frac{1}{r}-\frac{1}{q})} \left[\log\left(e+\frac{1}{t}\right)\right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} \|\varphi_{k,i}\|_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}}$$

$$= C(1+T^{-\frac{1}{\theta}}|x_{k,i}|)^{-n-\theta}t^{-\frac{n}{\theta}(\frac{1}{r}-\frac{1}{q})} \left[\log\left(e+\frac{1}{t}\right)\right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} \|\varphi\chi_{B(x_{k,i},T^{1/\theta})}\|_{\mathfrak{L}^{r,\infty}(\log\mathfrak{L})^{\alpha}}$$

$$\leq C(1+T^{-\frac{1}{\theta}}|x_{k,i}|)^{-n-\theta}t^{-\frac{n}{\theta}(\frac{1}{r}-\frac{1}{q})} \left[\log\left(e+\frac{1}{t}\right)\right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} \|\varphi\|_{r,\alpha;T^{1/\theta}}$$

$$\leq C(1+T^{-\frac{1}{\theta}}|x_{k,i}|)^{-n-\theta}t^{-\frac{n}{\theta}(\frac{1}{r}-\frac{1}{q})} \left[\log\left(e+\frac{1}{t}\right)\right]^{-\frac{\alpha}{r}+\frac{\beta}{q}} \|\varphi\|_{r,\alpha;T^{1/\theta}}$$

$$(3-22)$$

for $t \in (0, T)$.

On the other hand, since

$$\frac{1}{2}|y| \le \frac{1}{2}(|x_{k,i}| + T^{\frac{1}{\theta}}) \le |x_{k,i}| \quad \text{for } y \in B_{k,i},$$

we have

$$\frac{1}{|B_{k,i}|} \int_{B_{k,i}} \left(1 + \frac{1}{2} T^{-\frac{1}{\theta}} |y| \right)^{-n-\theta} dy \ge (1 + T^{-\frac{1}{\theta}} |x_{k,i}|)^{-n-\theta}.$$

Then, by (3-17), we see that

$$\sum_{i=1}^{\infty} (1+T^{-\frac{1}{\theta}}|x_{k,i}|)^{-n-\theta} \le CT^{-\frac{n}{\theta}} \sum_{i=1}^{\infty} \int_{B_{k,i}} \left(1+\frac{1}{2}T^{-\frac{1}{\theta}}|y|\right)^{-n-\theta} dy$$
$$\le CT^{-\frac{n}{\theta}} \int_{\mathbb{R}^n} \left(1+\frac{1}{2}T^{-\frac{1}{\theta}}|y|\right)^{-n-\theta} dy \le C$$
(3-23)

for T > 0. Combining (3-18), (3-19), (3-22), and (3-23), we obtain

$$t^{\frac{n}{\theta}\left(\frac{1}{r}-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right)\right]^{\frac{\alpha}{r}-\frac{\beta}{q}} \|\chi_{B(0,T^{1/\theta})}S_{\theta}(t)\varphi\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\beta}}$$

$$\leq C \|\|\varphi\|\|_{r,\alpha;T^{1/\theta}} + C \|\|\varphi\|\|_{r,\alpha;T^{1/\theta}} \sum_{k=1}^{m} \sum_{i=1}^{\infty} (1+T^{-\frac{1}{\theta}}|x_{k,i}|)^{-n-\theta}$$

$$\leq C \|\|\varphi\|\|_{r,\alpha;T^{1/\theta}}$$

for $t \in (0, T)$. This implies (3-16) with z = 0; that is, (3-15) holds. Thus Proposition 3.4 follows in the case of $0 < \theta < 2$.

Consider the case of $\theta = 2$; that is, $S_{\theta}(t) = e^{t\Delta}$. Let $\tau = t^{1/2}$. It follows from (2-16) that

$$|[e^{t\Delta}\varphi](x)| \le C \int_{\mathbb{R}^n} t^{-\frac{n}{2}} (1 + t^{-\frac{1}{2}} |x - y|)^{-n-1} |\varphi(y)| \, dy$$

$$\le C [S_1(\tau)|\varphi|](x)$$

for $(x, t) \in \mathbb{R}^n \times (0, \infty)$. This together with Proposition 3.4 in the case of $\theta = 1$ implies that

$$\begin{split} \| e^{t\Delta} \varphi \| \|_{q,\beta;T^{1/2}} &\leq C \, \| S_1(\tau) |\varphi| \| \|_{q,\beta;T^{1/2}} \\ &\leq C \, \tau^{-n\left(\frac{1}{r} - \frac{1}{q}\right)} \bigg[\log \bigg(e + \frac{1}{\tau} \bigg) \bigg]^{-\frac{\alpha}{r} + \frac{\beta}{q}} \| \varphi \| \|_{r,\alpha;T^{1/2}} \\ &\leq C \, t^{-\frac{n}{2}\left(\frac{1}{r} - \frac{1}{q}\right)} \bigg[\log \bigg(e + \frac{1}{t} \bigg) \bigg]^{-\frac{\alpha}{r} + \frac{\beta}{q}} \| \varphi \| \|_{r,\alpha;T^{1/2}} \quad \text{for } t \in (0,T). \end{split}$$

Thus Proposition 3.4 follows in the case of $\theta = 2$. The proof of Proposition 3.4 is complete.

4. Proof of Theorems 1.2 and 1.4

We apply the contraction mapping theorem to problem (P) in uniformly local weak Zygmund-type spaces and prove Theorems 1.2 and 1.4. We also prove Corollary 1.3. We first prove the following proposition.

Proposition 4.1. Let $p = p_{\theta}$, $T_* \in (0, \infty)$, and $\gamma \in [0, n/\theta)$. Then there exists $\epsilon > 0$ such that if $\varphi \in \mathfrak{L}^{1,\infty}_{ul}(\log \mathfrak{L})^{n/\theta}$ satisfies

$$\|\varphi\|_{1,\frac{n}{\theta};T^{1/\theta}} \le \epsilon \quad \text{for some } T \in (0,T_*], \tag{4-1}$$

then problem (P) possesses a solution

$$u \in C((0,T) : \mathfrak{L}^{1,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{n/\theta}) \cap L^{\infty}_{\mathrm{loc}}(0,T : L^{\infty}) \quad in \ \mathbb{R}^n \times (0,T),$$

with u satisfying

$$\begin{aligned} \| u(t) \|_{1,\frac{n}{\theta};T^{1/\theta}} &\leq C \, \| \varphi \| \|_{1,\frac{n}{\theta};T^{1/\theta}}, \\ \| u(t) \|_{p,\gamma;T^{1/\theta}} &\leq C \, t^{-\frac{n}{\theta} \left(1 - \frac{1}{p}\right)} \left[\log \left(e + \frac{1}{t} \right) \right]^{-\frac{n}{\theta} + \frac{\gamma}{p}} \| \varphi \|_{1,\frac{n}{\theta};T^{1/\theta}}, \\ \| u(t) \|_{L^{\infty}} &\leq C \, t^{-\frac{n}{\theta}} \left[\log \left(e + \frac{1}{t} \right) \right]^{-\frac{n}{\theta}} \| \varphi \|_{1,\frac{n}{\theta};T^{1/\theta}} \end{aligned}$$
(4-2)

for $t \in (0, T)$. Here C is a positive constant depending only on T_* , n, θ , and γ .

Throughout this section, we set

$$T_* \in (0,\infty), \quad T \in (0,T_*], \quad p := p_{\theta} = 1 + \frac{\theta}{n}, \quad \alpha := \frac{n}{\theta}, \quad 0 \le \gamma < \alpha, \quad \varphi \in \mathfrak{L}^{1,\infty}_{\mathrm{ul}}(\log \mathfrak{L})^{\alpha}.$$

Let $\epsilon > 0$, and assume (4-1). By Proposition 3.4, we find $C_* > 0$ such that

$$\sup_{\substack{0 < t < T}} \|S_{\theta}(t)\varphi\|\|_{1,\alpha;T^{1/\theta}} \leq C_{*}\|\|\varphi\|\|_{1,\alpha;T^{1/\theta}} \leq C_{*}\epsilon,$$

$$\sup_{\substack{0 < t < T}} t^{\frac{n}{\theta}(1-\frac{1}{p})} \left[\log\left(e+\frac{1}{t}\right)\right]^{-\frac{\gamma}{p}+\alpha} \|S_{\theta}(t)\varphi\|\|_{p,\gamma;T^{1/\theta}} \leq C_{*}\|\|\varphi\|\|_{1,\alpha;T^{1/\theta}} \leq C_{*}\epsilon, \qquad (4-3)$$

$$\sup_{\substack{0 < t < T}} t^{\frac{n}{\theta}} \left[\log\left(e+\frac{1}{t}\right)\right]^{\alpha} \|S_{\theta}(t)\varphi\|_{L^{\infty}} \leq C_{*}\|\|\varphi\|\|_{1,\alpha;T^{1/\theta}} \leq C_{*}\epsilon.$$

Define

$$X_T := C((0,T) : \mathfrak{L}^{1,\infty}_{ul}(\log \mathfrak{L})^{\alpha}) \cap L^{\infty}_{loc}((0,T) : \mathfrak{L}^{p,\infty}_{ul}(\log \mathfrak{L})^{\gamma}) \cap L^{\infty}_{loc}((0,T) : L^{\infty})$$

Setting $C^* = 2C_*$, for any $u \in X_T$, we say that $u \in X_T(C^*\epsilon)$ if u satisfies

$$\sup_{0 < t < T} \|\|u(t)\|\|_{1,\alpha;T^{1/\theta}} + \sup_{0 < t < T} t^{\frac{n}{\theta}(1-\frac{1}{p})} \left[\log\left(e+\frac{1}{t}\right) \right]^{-\frac{\gamma}{p}+\alpha} \|\|u(t)\|\|_{p,\gamma;T^{1/\theta}} + \sup_{0 < t < T} t^{\frac{n}{\theta}} \left[\log\left(e+\frac{1}{t}\right) \right]^{\alpha} \|u(t)\|_{L^{\infty}} \le C^* \epsilon.$$
(4-4)

For any $u, v \in X_T(C^*\epsilon)$, set

$$d_X(u, v) := d_X^1(u, v) + d_X^2(u, v) + d_X^3(u, v),$$

where

$$d_X^1(u,v) := \sup_{0 < t < T} |||u(t) - v(t)|||_{1,\alpha;T^{1/\theta}},$$

$$d_X^2(u,v) := \sup_{0 < t < T} t^{\frac{n}{\theta} \left(1 - \frac{1}{p}\right)} \left[\log\left(e + \frac{1}{t}\right) \right]^{-\frac{\gamma}{p} + \alpha} |||u(t) - v(t)|||_{p,\gamma;T^{1/\theta}},$$

$$d_X^3(u,v) := \sup_{0 < t < T} t^{\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t}\right) \right]^{\alpha} ||u(t) - v(t)||_{L^{\infty}}.$$

Then (X_T, d_X) is a Banach space and $X_T(C^*\epsilon)$ is closed in (X_T, d_X) . Define

$$\Phi(u) := S_{\theta}(t)\varphi + \int_0^t S_{\theta}(t-s)F_p(u(s))\,ds \quad \text{for } u \in X_T(C^*\epsilon),$$

where $F_p(s) = |s|^{p-1}s$ for $s \in \mathbb{R}$. For the proof of Proposition 4.1 we prepare the following two lemmas. Lemma 4.2. Let $\epsilon > 0$, and assume that (4-1) holds for some $T \in (0, T_*]$. Then there exists $C = C(n, \theta, C_*, T_*) > 0$ such that

$$d_X^1(\Phi(u), \Phi(v)) + d_X^2(\Phi(u), \Phi(v)) \le C\epsilon^{p-1} d_X^2(u, v) \text{ for } u, v \in X_T(C^*\epsilon).$$

Proof. Let $u, v \in X_T(C^*\epsilon)$. Let 0 < s < t < T. It follows that

$$|F_p(u(x,s)) - F_p(v(x,s))| \le w(x,s)|u(x,s) - v(x,s)| \quad \text{for } x \in \mathbb{R}^n,$$
(4-5)

where $w(x,s) := p(|u(x,s)|^{p-1} + |v(x,s)|^{p-1})$. Then, by Lemmas 2.2 and 2.3, we have

$$\|F_{p}(u(s)) - F_{p}(v(s))\|_{1,\gamma;T^{1/\theta}} \leq \|w(s)\|_{\frac{p}{(p-1)},\gamma;T^{1/\theta}} \|u(s) - v(s)\|_{p,\gamma;T^{1/\theta}}$$

$$\leq p(\|u(s)\|_{p,\gamma;T^{1/\theta}}^{p-1} + \|v(s)\|_{p,\gamma;T^{1/\theta}}^{p-1})\|u(s) - v(s)\|_{p,\gamma;T^{1/\theta}}.$$
(4-6)

Since $u, v \in X_T(C^*\epsilon)$, by (4-4), we obtain

$$\begin{aligned} \||F_{p}(u(s)) - F_{p}(v(s))||_{1,\gamma;T^{1/\theta}} \\ &\leq Cs^{-\frac{n(p-1)}{\theta}} (1 - \frac{1}{p}) \bigg[\log \bigg(e + \frac{1}{s} \bigg) \bigg]^{\frac{\gamma(p-1)}{p} - \alpha(p-1)} (C^{*}\epsilon)^{p-1} Cs^{-\frac{n}{\theta}} (1 - \frac{1}{p}) \bigg[\log \bigg(e + \frac{1}{s} \bigg) \bigg]^{\frac{\gamma}{p} - \alpha} d_{X}^{2}(u, v) \\ &= C\epsilon^{p-1} s^{-\frac{n(p-1)}{\theta}} \bigg[\log \bigg(e + \frac{1}{s} \bigg) \bigg]^{\gamma - \alpha p} d_{X}^{2}(u, v). \end{aligned}$$

$$(4-7)$$

This together with Proposition 3.4 implies that

$$\begin{split} \left\| \int_{0}^{t} S_{\theta}(t-s) [F_{p}(u(s)) - F_{p}(v(s))] \, ds \right\|_{q,\beta;T^{1/\theta}} \\ &\leq \int_{0}^{t} \left\| S_{\theta}(t-s) [F_{p}(u(s)) - F_{p}(v(s))] \right\|_{q,\beta;T^{1/\theta}} \, ds \\ &\leq C \int_{0}^{t} (t-s)^{-\frac{n}{\theta}(1-\frac{1}{q})} \left[\log\left(e + \frac{1}{t-s}\right) \right]^{-\gamma + \frac{\beta}{q}} \left\| F_{p}(u(s)) - F_{p}(v(s)) \right\|_{1,\gamma;T^{1/\theta}} \, ds \\ &\leq C \epsilon^{p-1} d_{X}^{2}(u,v) \int_{0}^{t} (t-s)^{-\frac{n}{\theta}(1-\frac{1}{q})} \left[\log\left(e + \frac{1}{t-s}\right) \right]^{-\gamma + \frac{\beta}{q}} s^{-\frac{n}{\theta}(p-1)} \left[\log\left(e + \frac{1}{s}\right) \right]^{\gamma - \alpha p} \, ds \quad (4-8) \\ &\text{for } q \in [1, p] \text{ and } \beta \in [\gamma, \alpha]. \end{split}$$

On the other hand, since

$$\gamma - \alpha p = \gamma - \frac{n}{\theta} \left(1 + \frac{\theta}{n} \right) = \gamma - \frac{n}{\theta} - 1 = \gamma - \alpha - 1 < -1, \tag{4-9}$$

by Lemma 3.2(2) and (3-1), we have

$$\int_{0}^{\frac{t}{2}} (t-s)^{-\frac{n}{\theta}\left(1-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t-s}\right) \right]^{-\gamma+\frac{\beta}{q}} s^{-\frac{n}{\theta}\left(p-1\right)} \left[\log\left(e+\frac{1}{s}\right) \right]^{\gamma-\alpha p} ds$$

$$\leq Ct^{-\frac{n}{\theta}\left(1-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right) \right]^{-\gamma+\frac{\beta}{q}} \int_{0}^{\frac{t}{2}} s^{-1} \left[\log\left(e+\frac{1}{s}\right) \right]^{\gamma-\alpha p} ds$$

$$\leq Ct^{-\frac{n}{\theta}\left(1-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right) \right]^{-\gamma+\frac{\beta}{q}} \cdot C \left[\log\left(e+\frac{1}{t}\right) \right]^{\gamma-\frac{n}{\theta}} = Ct^{-\frac{n}{\theta}\left(1-\frac{1}{q}\right)} \left[\log\left(e+\frac{1}{t}\right) \right]^{\frac{\beta}{q}-\alpha}$$
(4-10)

for $t \in (0, T)$. Similarly, since

$$-\frac{n}{\theta}\left(1-\frac{1}{q}\right) \ge -\frac{n(p-1)}{\theta p} = -\frac{1}{p} > -1,$$

by Lemma 3.2(1) and (3-1), we obtain

$$\begin{split} \int_{\frac{t}{2}}^{t} (t-s)^{-\frac{n}{\theta}(1-\frac{1}{q})} \bigg[\log\bigg(e+\frac{1}{t-s}\bigg) \bigg]^{-\gamma+\frac{\beta}{q}} s^{-\frac{n}{\theta}(p-1)} \bigg[\log\bigg(e+\frac{1}{s}\bigg) \bigg]^{\gamma-\alpha p} ds \\ &\leq Ct^{-\frac{n}{\theta}(p-1)} \bigg[\log\bigg(e+\frac{1}{t}\bigg) \bigg]^{\gamma-\alpha p} \int_{\frac{t}{2}}^{t} (t-s)^{-\frac{n}{\theta}(1-\frac{1}{q})} \bigg[\log\bigg(e+\frac{1}{t-s}\bigg) \bigg]^{-\gamma+\frac{\beta}{q}} ds \\ &\leq Ct^{-1} \bigg[\log\bigg(e+\frac{1}{t}\bigg) \bigg]^{\gamma-\alpha p} \cdot Ct^{-\frac{n}{\theta}(1-\frac{1}{q})+1} \bigg[\log\bigg(e+\frac{1}{t}\bigg) \bigg]^{-\gamma+\frac{\beta}{q}} \\ &= Ct^{-\frac{n}{\theta}(1-\frac{1}{q})} \bigg[\log\bigg(e+\frac{1}{t}\bigg) \bigg]^{\frac{\beta}{q}-\alpha p} \leq Ct^{-\frac{n}{\theta}(1-\frac{1}{q})} \bigg[\log\bigg(e+\frac{1}{t}\bigg) \bigg]^{\frac{\beta}{q}-\alpha} \tag{4-11}$$

for $t \in (0, T)$. Combining (4-8), (4-10), and (4-11) with $(q, \beta) = (1, \alpha)$ and (p, γ) , we deduce that

$$\begin{aligned} d_X^1(\Phi(u), \Phi(v)) + d_X^2(\Phi(u), \Phi(v)) \\ &= \sup_{0 < t < T} \left\| \int_0^t S_\theta(t-s) [F_p(u(s)) - F_p(v(s))] \, ds \right\|_{1,\alpha;T^{1/\theta}} \\ &+ \sup_{0 < t < T} t^{\frac{n}{\theta}(1-\frac{1}{p})} \left[\log\left(e + \frac{1}{t}\right) \right]^{-\frac{\nu}{p} + \alpha} \left\| \int_0^t S_\theta(t-s) [F_p(u(s)) - F_p(v(s))] \, ds \right\|_{p,\gamma;T^{1/\theta}} \\ &\leq C \epsilon^{p-1} d_X^2(u, v) \end{aligned}$$

for $u, v \in X_T(C^*\epsilon)$. Thus Lemma 4.2 follows.

Lemma 4.3. Let $\epsilon > 0$, and assume that (4-1) holds for some $T \in (0, T_*]$. Then there exists $C = C(n, \theta, C_*, T_*) > 0$ such that

$$d_X^3(\Phi(u), \Phi(v)) \le C\epsilon^{p-1}(d_X^2(u, v) + d_X^3(u, v)) \text{ for } u, v \in X_T(C^*\epsilon).$$

Proof. Let $u, v \in X_T(C^*\epsilon)$. Let 0 < s < t < T. Similarly to (4-6), we have

$$\|F_p(u(s)) - F_p(v(s))\|_{L^{\infty}} \le \|w(s)\|_{L^{\infty}} \|u(s) - v(s)\|_{L^{\infty}} \le p(\|u(s)\|_{L^{\infty}}^{p-1} + \|v(s)\|_{L^{\infty}}^{p-1}) \|u(s) - v(s)\|_{L^{\infty}}.$$

Since $u, v \in X_T(C^*\epsilon)$, by (4-4), we obtain

$$\begin{split} \|F_p(u(s)) - F_p(v(s))\|_{L^{\infty}} &\leq C s^{-\frac{n(p-1)}{\theta}} \bigg[\log \bigg(e + \frac{1}{s} \bigg) \bigg]^{-\alpha(p-1)} (C^* \epsilon)^{p-1} \cdot s^{-\frac{n}{\theta}} \bigg[\log \bigg(e + \frac{1}{s} \bigg) \bigg]^{-\alpha} d_X^3(u, v) \\ &= C \epsilon^{p-1} s^{-\frac{np}{\theta}} \bigg[\log \bigg(e + \frac{1}{s} \bigg) \bigg]^{-\alpha p} d_X^3(u, v). \end{split}$$

This together with Proposition 3.4 and (4-7) implies that

$$\begin{split} \left\| \int_{0}^{t} S_{\theta}(t-s) [F_{p}(u(s)) - F_{p}(v(s))] ds \right\|_{L^{\infty}} \\ &\leq \int_{0}^{t} \| S_{\theta}(t-s) [F_{p}(u(s)) - F_{p}(v(s))] \|_{L^{\infty}} ds \\ &\leq C \int_{0}^{\frac{t}{2}} (t-s)^{-\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t-s}\right) \right]^{-\gamma} \| F_{p}(u(s)) - F_{p}(v(s)) \|_{1,\gamma;T^{1/\theta}} ds \\ &+ C \int_{\frac{t}{2}}^{t} \| F_{p}(u(s)) - F_{p}(v(s)) \|_{L^{\infty}} ds \\ &\leq C \epsilon^{p-1} d_{X}^{2}(u,v) \int_{0}^{\frac{t}{2}} (t-s)^{-\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t-s}\right) \right]^{-\gamma} s^{-\frac{n}{\theta}(p-1)} \left[\log\left(e + \frac{1}{s}\right) \right]^{\gamma-\alpha p} ds \\ &+ C \epsilon^{p-1} d_{X}^{3}(u,v) \int_{\frac{t}{2}}^{t} s^{-\frac{np}{\theta}} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\alpha p} ds \\ &\leq C \epsilon^{p-1} t^{-\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t}\right) \right]^{-\gamma} d_{X}^{2}(u,v) \int_{0}^{\frac{t}{2}} s^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{\gamma-\alpha p} ds \\ &+ C \epsilon^{p-1} t^{-\frac{np}{\theta}+1} \left[\log\left(e + \frac{1}{t}\right) \right]^{-\alpha p} d_{X}^{3}(u,v). \end{split}$$

Since $np = n + \theta$ and $\alpha p > \alpha$, we have

$$\begin{split} \left\| \int_{0}^{t} S_{\theta}(t-s) [F_{p}(u(s)) - F_{p}(v(s))] \, ds \right\|_{L^{\infty}} \\ &\leq C \epsilon^{p-1} t^{-\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t}\right) \right]^{-\gamma} d_{X}^{2}(u,v) \int_{0}^{\frac{t}{2}} s^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{\gamma-\alpha p} \, ds \\ &+ C \epsilon^{p-1} t^{-\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t}\right) \right]^{-\alpha} d_{X}^{3}(u,v). \quad (4-12) \end{split}$$

Furthermore, by Lemma 3.2(2) and (4-9) we see that

$$\int_{0}^{\frac{t}{2}} s^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{\gamma - \alpha p} ds \le C \left[\log\left(e + \frac{1}{t}\right) \right]^{\gamma - \alpha}$$
(4-13)

for $t \in (0, T)$. Combining (4-12) and (4-13), we deduce that

$$d_X^3(\Phi(u), \Phi(v)) = \sup_{0 < t < T} t^{\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t}\right) \right]^{\alpha} \left\| \int_0^t S_{\theta}(t-s) [F_p(u(s)) - F_p(v(s))] \, ds \right\|_{L^{\infty}} \le C \epsilon^{p-1} (d_X^2(u, v) + d_X^3(u, v))$$

for $u, v \in X_T(C^*\epsilon)$. Thus Lemma 4.3 follows.

Proof of Proposition 4.1. Let $T_* > 0$. Let $\epsilon > 0$ be small enough. Let $\varphi \in \mathfrak{L}^{1,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$ be such that $\|\|\varphi\|\|_{1,\alpha;T^{1/\theta}} < \epsilon$ for some $T \in (0, T_*]$. By (4-3), (4-4), and Lemma 4.2, we have

$$\sup_{t \in (0,T)} \|\|\Phi(u(t))\|\|_{1,\alpha;T^{1/\theta}} + \sup_{0 < t < T} \left\{ t^{\frac{n}{\theta}(1-\frac{1}{p})} \left[\log\left(e + \frac{1}{t}\right) \right]^{-\frac{\nu}{p}+\alpha} \|\|\Phi(u(t))\|\|_{p,\gamma;T^{1/\theta}} \right\} \\
\leq \|\|S_{\theta}(t)\varphi\|\|_{1,\alpha,T^{1/\theta}} + \sup_{0 < t < T} \left\{ t^{\frac{n}{\theta}(1-\frac{1}{p})} \left[\log\left(e + \frac{1}{t}\right) \right]^{-\frac{\nu}{p}+\alpha} \|\|S_{\theta}(t)\varphi\|\|_{p,\gamma;T^{1/\theta}} \right\} \\
+ d_{X}^{1}(\Phi(u), \Phi(0)) + d_{X}^{2}(\Phi(u), \Phi(0)) \\
\leq C_{*}\epsilon + C\epsilon^{p-1} d_{X}^{2}(u, 0) \\
\leq C^{*}\epsilon \\
\leq C^{*}\epsilon \qquad (4-14)$$

for $u \in X_T(C^*\epsilon)$. Similarly, we observe from Lemma 4.3, (4-3), and (4-4) that

$$\sup_{0 < t < T} \left\{ t^{\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t}\right) \right]^{\alpha} \|\Phi(u(t))\|_{L^{\infty}} \right\} \leq \sup_{0 < t < T} \left\{ t^{\frac{n}{\theta}} \left[\log\left(e + \frac{1}{t}\right) \right]^{\alpha} \|S_{\theta}(t)\varphi\|_{L^{\infty}} \right\} + d_X^3(\Phi(u), \Phi(0))$$
$$\leq C_* \epsilon + C \epsilon^{p-1} (d_X^2(u, 0) + d_X^3(u, 0))$$
$$\leq C_* \epsilon + C \epsilon^{p-1} \cdot 2C^* \epsilon$$
$$\leq C^* \epsilon \tag{4-15}$$

for $u \in X_T(C^*\epsilon)$. By (4-14) and (4-15), we see that $\Phi(u) \in X_T(C^*\epsilon)$ for $u \in X_T(C^*\epsilon)$. Furthermore, taking small enough $\epsilon > 0$ if necessary, by Lemmas 4.2 and 4.3, we have

$$d_X(\Phi(u), \Phi(v)) = d_X^1(\Phi(u), \Phi(v)) + d_X^2(\Phi(u), \Phi(v)) + d_X^3(\Phi(u), \Phi(v))$$

$$\leq C\epsilon^{p-1}(d_X^2(u, v) + d_X^3(u, v))$$

$$\leq \frac{1}{2}d_X(u, v)$$

for $u, v \in X_T(C^*\epsilon)$. Then we apply the contraction mapping theorem to find a unique $u_* \in X_T(C^*\epsilon)$ such that $\Phi(u_*) = u_*$ in $X_T(C^*\epsilon)$. The function u_* is a solution to problem (P) in $\mathbb{R}^n \times (0, T)$, with u_* satisfying (4-2). Thus Proposition 4.1 follows.

Proof of Theorem 1.2. Let T > 0. Let $\varphi \in \mathfrak{L}^{1,\infty}_{ul}(\log \mathfrak{L})^{\alpha}$ be such that $|||\varphi|||_{1,\alpha;T^{1/\theta}}$ is small enough. Then, by Proposition 4.1, we find a solution u to problem (P) in $\mathbb{R}^n \times (0, T)$, with u satisfying (4-2). Let $\beta \in (\gamma, n/\theta)$. Then, by Proposition 3.4, Lemma 2.3, and (4-2), we obtain

$$\begin{split} \|\|u(t) - S_{\theta}(t)\varphi\|\|_{1,\beta;T^{1/\theta}} &\leq \int_{0}^{t} \||S_{\theta}(t-s)F_{p}(u(s))\|\|_{1,\beta;T^{1/\theta}} \, ds \\ &\leq C \int_{0}^{t} \left[\log\left(e + \frac{1}{t-s}\right)\right]^{-\gamma+\beta} \||F_{p}(u(s))\|\|_{1,\gamma;T^{1/\theta}} \, ds \\ &= C \int_{0}^{t} \left[\log\left(e + \frac{1}{t-s}\right)\right]^{-\gamma+\beta} \||u(s)\||_{p,\gamma;T^{1/\theta}}^{p} \, ds \\ &\leq C \||\varphi\|\|_{1,\alpha;T^{1/\theta}}^{p} \int_{0}^{t} \left[\log\left(e + \frac{1}{t-s}\right)\right]^{-\gamma+\beta} s^{-1} \left[\log\left(e + \frac{1}{s}\right)\right]^{\gamma-\alpha p} \, ds \quad (4-16) \end{split}$$

for $t \in (0, T)$. On the other hand, since $\beta < \theta/n$, by Lemma 3.2 (2) and (4-9), we have

$$\int_{0}^{\frac{t}{2}} \left[\log\left(e + \frac{1}{t-s}\right) \right]^{-\gamma+\beta} s^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{\gamma-\alpha p} ds$$

$$\leq C \left[\log\left(e + \frac{1}{t}\right) \right]^{-\gamma+\beta} \int_{0}^{\frac{t}{2}} s^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{\gamma-\alpha p} ds$$

$$\leq C \left[\log\left(e + \frac{1}{t}\right) \right]^{-\gamma+\beta} \cdot C \left[\log\left(e + \frac{1}{t}\right) \right]^{\gamma-\frac{n}{\theta}} \to 0 \qquad (4-17)$$
and

and

$$\int_{\frac{t}{2}}^{t} \left[\log\left(e + \frac{1}{t-s}\right) \right]^{-\gamma+\beta} s^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{\gamma-\alpha\rho} ds$$

$$\leq Ct^{-1} \left[\log\left(e + \frac{1}{t}\right) \right]^{\gamma-\frac{n}{\theta}-1} \int_{\frac{t}{2}}^{t} \left[\log\left(e + \frac{1}{t-s}\right) \right]^{-\gamma+\beta} ds$$

$$\leq Ct^{-1} \left[\log\left(e + \frac{1}{t}\right) \right]^{\gamma-\frac{n}{\theta}-1} \cdot Ct \left[\log\left(e + \frac{1}{t}\right) \right]^{-\gamma+\beta} \to 0 \quad (4-18)$$

as $t \rightarrow +0$. Combining (4-16), (4-17), and (4-18), we see that

$$\lim_{t \to +0} \| u(t) - S_{\theta}(t)\varphi \|_{1,\beta;T^{1/\theta}} = 0 \quad \text{for } \beta \in (\gamma, n/\theta).$$

This together with (2-12) implies that

$$\lim_{t \to +0} |||u(t) - S_{\theta}(t)\varphi|||_{1,\beta;T^{1/\theta}} = 0 \quad \text{for } \beta \in [0, n/\theta).$$
(4-19)

It remains to prove that $u \to \varphi$ in the sense of distributions. Let $\eta \in C_0(\mathbb{R}^n)$. Let R > 0 be such that supp $\eta \subset B(0, R)$. By (1-3), (1-4), and (4-19), we have

$$\left| \int_{\mathbb{R}^{n}} (u(x,t) - [S_{\theta}(t)\varphi](x))\eta(x) \, dx \right| \leq C \, \|\eta\|_{L^{\infty}} \int_{B(0,R)} |u(x,t) - [S_{\theta}(t)\varphi](x)| \, dx$$
$$\leq C \, \|\eta\|_{L^{\infty}} \|\|u(t) - S_{\theta}(t)\varphi\|_{1,0;T^{1/\theta}} \to 0 \tag{4-20}$$

as $t \to +0$. Set

ſ

$$\eta(x,t) := \int_{\mathbb{R}^n} G_{\theta}(x-y,t)\eta(y) \, dy \quad \text{for } (x,t) \in \mathbb{R}^n \times (0,\infty).$$

It follows from (2-18) that

$$\lim_{t \to +0} \|\eta(\cdot, t) - \eta\|_{L^{\infty}} = 0.$$
(4-21)

On the other hand, by (2-16), we have

$$|\eta(x,t)| \le Ct^{-\frac{n}{\theta}} \int_{B(0,R)} (1+t^{-\frac{1}{\theta}}|x-y|)^{-n-\theta} |\eta(y)| \, dy \le C \, \|\eta\|_{L^{\infty}} t^{-\frac{n}{\theta}} \cdot C(t^{-\frac{1}{\theta}}|x|)^{-n-\theta} \le T|x|^{-n-\theta}$$

for $x \in \mathbb{R}^n \setminus B(0, 2R)$ and $t \in (0, T)$. Since $\|\eta(\cdot, t)\|_{L^{\infty}} \le \|\eta\|_{L^{\infty}}$ for t > 0, we obtain

$$|\eta(x,t)| \le C(1+|x|)^{-n-\theta}$$
 for $(x,t) \in \mathbb{R}^n \times (0,T)$. (4-22)

Furthermore, it follows from Proposition 3.4 with $q = \infty$ that

$$[S_{\theta}(1)|\varphi|](0) = \int_{\mathbb{R}^n} G_{\theta}(y,1)|\varphi(y)| \, dy < \infty.$$

This together with (2-16) implies that

$$\int_{\mathbb{R}^n} (1+|y|)^{-n-\theta} |\varphi(y)| \, dy < \infty.$$
(4-23)

Therefore, by (4-21), (4-22), and (4-23), we apply the Fubini theorem and the Lebesgue convergence theorem to obtain

$$\int_{\mathbb{R}^n} [S_{\theta}(t)\varphi](x)\eta(x) \, dx$$

= $\int_{\mathbb{R}^n} \left(\int_{\mathbb{R}^n} G_{\theta}(x-y,t)\varphi(y) \, dy \right) \eta(x) \, dx = \int_{\mathbb{R}^n} \left(\int_{\mathbb{R}^n} G_{\theta}(x-y,t)\eta(x) \, dx \right) \varphi(y) \, dy$
= $\int_{\mathbb{R}^n} \eta(y,t)\varphi(y) \, dy \to \int_{\mathbb{R}^n} \eta(y)\varphi(y) \, dy$

as $t \to +0$. Then we deduce from (4-20) that

$$\lim_{t \to +0} \int_{\mathbb{R}^n} u(x,t)\eta(x) \, dx = \int_{\mathbb{R}^n} \varphi(x)\eta(x) \, dx \quad \text{for } \eta \in C_0(\mathbb{R}^n);$$

that is, $u(t) \rightarrow \varphi$ in the sense of distributions. The proof of Theorem 1.2 is complete.

Proof of Corollary 1.3. Let φ_c be as in (1-1) with $p = p_{\theta}$. It follows from the definition of the nonincreasing rearrangements that

$$(\varphi_c)^*(s) \le C s^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\frac{n}{\theta} - 1} \quad \text{for } s \in (0, \infty).$$

$$(4-24)$$

Let S > 0. Then, by Lemma 3.2 (2), (2-3), and (4-24), we see that

$$(\varphi_c)^{**}(s) \le C s^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\frac{n}{\theta}} \quad \text{for } s \in (0, S).$$

This implies that $\varphi_c \in \mathfrak{L}^{1,\infty}_{ul}(\log \mathfrak{L})^{n/\theta}$. Then Corollary 1.3 follows from Theorem 1.2.

 \square

Proof of Theorem 1.4. Since $\alpha > n/\theta$, it follows from (2-12) that

$$\||\varphi\||_{1,\frac{n}{\theta};T^{1/\theta}} \le C \left[\log \left(e + \frac{1}{T^{1/\theta}} \right) \right]^{\frac{n}{\theta} - \alpha} \||\varphi\||_{1,\alpha;T^{1/\theta}} \to 0 \quad \text{as } T \to +0.$$

Then, by Theorem 1.2, we find a solution u to problem (P) in $\mathbb{R}^n \times (0, T)$ for some small enough T > 0, with u satisfying (1-6) and (1-7). Thus Theorem 1.4 follows.

At the end of this paper we recall the definitions of the usual Zygmund space and the usual weak Zygmund space, and explain the advantage of our weak Zygmund-type spaces.

Remark 4.4. (i) We recall the Zygmund space $L^q (\log L)^{\alpha}$ and the weak Zygmund space $L^{q,\infty} (\log L)^{\alpha}$. For any $q \in [1,\infty]$ and $\alpha \ge 0$, set

$$L^{q}(\log L)^{\alpha} := \{ f \in L^{1}_{loc}(\mathbb{R}^{n}) : \| f \|_{L^{q}(\log L)^{\alpha}} < \infty \},\$$
$$L^{q,\infty}(\log L)^{\alpha} := \{ f \in L^{1}_{loc}(\mathbb{R}^{n}) : \| f \|_{L^{q,\infty}(\log L)^{\alpha}} < \infty \},\$$

where

$$\|f\|_{L^{q}(\log L)^{\alpha}} := \left(\int_{0}^{\infty} \left[\log\left(e + \frac{1}{s}\right)\right]^{\alpha} f^{*}(s)^{q} \, ds\right)^{\frac{1}{q}},\tag{4-25}$$

$$\|f\|_{L^{q,\infty}(\log L)^{\alpha}} := \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} sf^*(s)^q \right\}^{\frac{1}{q}}.$$
(4-26)

See, e.g., [Bennett and Sharpley 1988, Chapter 4, Section 6] and [Wadade 2014]. For the case q > 1, as in the Lorentz space (see, e.g., [Grafakos 2008, Chapter 1, Exercises 1.4.3]), applying Hardy's inequality (see Lemma 2.4) and Lemma 3.2 with (2-3), for any $f \in L^1_{loc}$, we see that $f \in L^q (\log L)^{\alpha}$ if and only if

$$[f]_{L^q(\log L)^{\alpha}} := \left(\int_0^\infty \left[\log\left(e + \frac{1}{s}\right)\right]^{\alpha} f^{**}(s)^q \, ds\right)^{\frac{1}{q}} < \infty.$$

In contrast, the above relation does not hold for the case q = 1. In fact, applying integration by parts, we see that

$$\int_0^\infty \left[\log\left(e + \frac{1}{s}\right) \right]^\alpha f^*(s) \, ds = \alpha \int_0^\infty \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha - 1} f^{**}(s) \frac{ds}{es + 1} + \|f\|_{L^1}.$$

(ii) By O'Neil's inequality (2-7), we have the inequality

$$(G_{\theta}(\cdot,t)*\varphi)^{**}(s) \leq \int_{s}^{\infty} (G_{\theta}(\cdot,t))^{**}(\tau)\varphi^{**}(\tau) d\tau, \quad s>0,$$

which is crucial in the proof of our sharp decay estimates of $S_{\theta}(t)\varphi$. Our Zygmund-type spaces are defined by the average of the nonincreasing rearrangement, and they are effectively used in the proof of our sharp decay estimates of $S_{\theta}(t)\varphi$ (see the proof of Proposition 3.1). These sharp decay estimates of $S_{\theta}(t)\varphi$ in the spaces $\mathcal{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$ enable us to obtain Theorem 1.2.

On the other hand, since the weak Zygmund space $L^{q,\infty}(\log L)^{\alpha}$ is defined by the nonincreasing rearrangement, the inequality

$$(G_{\theta}(\cdot,t)*\varphi)^{*}(s) \le (G_{\theta}(\cdot,t)*\varphi)^{**}(s) \le \int_{s}^{\infty} (G_{\theta}(\cdot,t))^{**}(\tau)\varphi^{**}(\tau) \, d\tau, \quad s > 0,$$
(4-27)

seems useful for the study of decay estimates of $S_{\theta}(t)\varphi$ in the space $L^{q,\infty}(\log L)^{\alpha}$. The first inequality in (4-27) follows from inequality (2-4). However, in general, inequality (2-4) is not sharp in $L^{1,\infty}(\log L)^{\alpha}$, where $\alpha > 1$. Indeed, let $f \in L^{1}_{loc}$ be such that

$$f^*(s) = s^{-1} \left[\log \left(e + \frac{1}{s} \right) \right]^{-\alpha}, \quad s > 0,$$

where $\alpha > 1$. Then $f \in L^{1,\infty}(\log L)^{\alpha}$ and

$$f^{**}(s) \asymp s^{-1} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\alpha + 1}$$

for small enough s > 0. Then $f^*(s)/f^{**}(s) \to 0$ as $s \to +0$, and we see that inequality (2-4) is not sharp. This suggests that it is difficult to obtain sharp decay estimates of $S_{\theta}(t)\varphi$ in the usual weak Zygmund spaces.

(iii) In order to overcome the disadvantage of the usual weak Zygmund spaces, one might consider the weak Zygmund-type spaces

$$\mathbb{L}^{q,\infty}(\log \mathbb{L})^{\alpha} := \{ f \in L^1_{\operatorname{loc}} : \| f \|_{\mathbb{L}^{q,\infty}(\log \mathbb{L})^{\alpha}} < \infty \},\$$

where $1 \le q < \infty$, $\alpha \ge 0$, and

$$\|f\|_{\mathbb{L}^{q,\infty}(\log \mathbb{L})^{\alpha}} := \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} sf^{**}(s)^{q} \right\}^{\frac{1}{q}}.$$

Indeed, applying the arguments to those in the proof of Proposition 3.4, we can obtain similar sharp decay estimates of $S_{\theta}(t)\varphi$ in the weak Zygmund-type space $\mathbb{L}^{q,\infty}(\log \mathbb{L})^{\alpha}$ to those in Proposition 3.4.

On the other hand, in the proof of Theorem 1.2, we used the inequality

$$\||f|^{p}\|_{\mathfrak{L}^{1,\infty}(\log\mathfrak{L})^{\alpha}} \le C \|f\|_{\mathfrak{L}^{p,\infty}(\log\mathfrak{L})^{\alpha}}^{p} \quad \text{for } f \in \mathfrak{L}^{p,\infty}(\log\mathfrak{L})^{\alpha} \tag{4-28}$$

in order to estimate the nonlinear term $|u|^{p-1}u$, where p > 1 and $\alpha \ge 0$. Actually, (4-28) holds with C = 1 and " \le " replace by "=" (see Lemma 2.3). In the case of $\mathbb{L}^{q,\infty}(\log \mathbb{L})^{\alpha}$, it follows from (2-5) that

$$\||f|^p\|_{\mathbb{L}^{1,\infty}(\log \mathbb{L})^{\alpha}} = \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} s(|f|^p)^{**}(s) \right\}$$
$$\geq \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} s(f^{**}(s))^p \right\} = \|f\|_{\mathbb{L}^{p,\infty}(\log \mathbb{L})^{\alpha}}^p$$

for $f \in \mathbb{L}^{p,\infty}(\log \mathbb{L})^{\alpha}$; that is, the reverse to the desired inequality holds. This suggests that it is difficult to obtain a similar result to that of Theorem 1.2 in the framework of weak Zygmund-type spaces $\mathbb{L}^{q,\infty}(\log \mathbb{L})^{\alpha}$.

Appendix

Here we prove two propositions on relations between $L^q (\log L)^{\alpha}$, $L^{q,\infty} (\log L)^{\alpha}$, and $\mathfrak{L}^{q,\infty} (\log \mathfrak{L})^{\alpha}$. We remark that the following relations hold for $\alpha = 0$:

$$L^{q} = L^{q} (\log L)^{0} = \mathfrak{L}^{q,\infty} (\log \mathfrak{L})^{0} \subsetneq L^{q,\infty} = L^{q,\infty} (\log L)^{0} \quad \text{if } q \in [1,\infty).$$

Proposition A.1. Let $1 \le q < \infty$ and $\alpha \ge 0$. Then

$$\|f\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} \leq \|f\|_{L^{q}(\log L)^{\alpha}} \quad \text{for } f \in L^{q}(\log L)^{\alpha},$$
$$\|f\|_{L^{q,\infty}(\log L)^{\alpha}} \leq \|f\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} \quad \text{for } f \in \mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}.$$

Furthermore,

$$L^{q}(\log L)^{\alpha} \subsetneq \mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha} \subsetneq L^{q,\infty}(\log L)^{\alpha}, \quad \alpha > 0.$$

Proof. By (2-9), (2-11), and (4-25), we see that

$$\|f\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} = \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} \int_{0}^{s} (f^{*}(\tau))^{q} d\tau \right\}^{\frac{1}{q}} \\ \leq \sup_{s>0} \left(\int_{0}^{s} \left[\log\left(e + \frac{1}{\tau}\right) \right]^{\alpha} (f^{*}(\tau))^{q} d\tau \right)^{\frac{1}{q}} = \|f\|_{L^{q}(\log L)^{\alpha}}$$

for $f \in L^q (\log L)^{\alpha}$. This implies that $L^q (\log L)^{\alpha} \subset \mathfrak{L}^{q,\infty} (\log \mathfrak{L})^{\alpha}$. Let g be a function in \mathbb{R}^n such that

$$g^*(s) = \frac{d}{ds} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{-\alpha} \right\} \chi_{(0,\delta)}(s) = \frac{\alpha}{es^2 + s} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\alpha - 1} \chi_{(0,\delta)}(s),$$

where $\delta > 0$ is chosen so that g^* is decreasing. Set $f(x) := |g(x)|^{1/q}$. It follows from (2-1) that $f^*(s)^q = g^*(s)$. Furthermore,

$$\|f\|_{\mathcal{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}}^{q} = \sup_{s>0} \left[\log\left(e+\frac{1}{s}\right)\right]^{\alpha} \int_{0}^{s} g^{*}(\eta) \, d\eta = 1$$

and

$$\|f\|_{L^q(\log L)^{\alpha}}^q = \int_0^\infty \left[\log\left(e+\frac{1}{\eta}\right)\right]^{\alpha} g^*(\eta) \, d\eta = \int_0^\delta \frac{\alpha}{e\eta^2 + \eta} \left[\log\left(e+\frac{1}{\eta}\right)\right]^{-1} d\eta = \infty$$

Thus $L^q (\log L)^{\alpha} \subsetneq \mathfrak{L}^{q,\infty} (\log \mathfrak{L})^{\alpha}$.

On the other hand, it follows from (2-1), (2-4), (2-9), and (4-26) that

$$\|f\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} = \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} s(|f|^{q})^{**}(s) \right\}^{\frac{1}{q}}$$

$$\geq \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} s(|f|^{q})^{*}(s) \right\}^{\frac{1}{q}}$$

$$= \sup_{s>0} \left\{ \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} s(f^{*}(s))^{q} \right\}^{\frac{1}{q}} = \|f\|_{L^{q,\infty}(\log L)^{\alpha}}$$

for $f \in \mathcal{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$, and hence $\mathcal{L}^{q,\infty}(\log \mathfrak{L})^{\alpha} \subset L^{q,\infty}(\log L)^{\alpha}$. We finally show that the inclusion is strict. Let f be a function such that

$$f^*(s) = s^{-\frac{1}{q}} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\frac{\alpha}{q}} \chi_{(0,\delta)}(s)$$

where $\delta > 0$ is chosen so that f^* is decreasing. Then $||f||_{L^{q,\infty}(\log L)^{\alpha}} = 1$. On the other hand, for the case $\alpha \leq 1$, we see that

$$s(|f|^{q})^{**}(s) = \int_{0}^{s} \eta^{-1} \left[\log\left(e + \frac{1}{\eta}\right) \right]^{-\alpha} d\eta \ge \int_{0}^{s} \eta^{-1} \left[\log\left(e + \frac{1}{\eta}\right) \right]^{-1} d\eta = \infty$$

for $s \in (0, \delta)$. This implies that $f \notin \mathcal{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$. Furthermore, for the case $\alpha > 1$, there exists C > 0 such that

$$s(|f|^{q})^{**}(s) = \int_{0}^{s} \eta^{-1} \left[\log\left(e + \frac{1}{\eta}\right) \right]^{-\alpha} d\eta$$
$$\geq C \int_{0}^{s} (e\eta^{2} + \eta)^{-1} \left[\log\left(e + \frac{1}{\eta}\right) \right]^{-\alpha} d\eta = \frac{C}{\alpha - 1} \left[\log\left(e + \frac{1}{s}\right) \right]^{1-\alpha}$$

for $s \in (0, \delta)$. In conclusion, there exists C > 0 such that

$$\|f\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} \ge C \sup_{0 < s < \delta} \left[\log\left(e + \frac{1}{s}\right)\right]^{\frac{1}{q}} = \infty$$

Thus $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha} \subsetneq L^{q,\infty}(\log L)^{\alpha}$. The proof of Proposition A.1 is complete.

Let f be a locally integrable function in \mathbb{R}^n such that

$$\left[\log\left(e + \frac{1}{s}\right)\right]^{\alpha} s(|f|^{q})^{**}(s) = 1, \quad s > 0,$$
(A-1)

which is a typical function in $\mathfrak{L}^{q,\infty}(\log \mathfrak{L})^{\alpha}$. By (A-1), we see

$$f^*(s)^q = \frac{d}{ds}(s(|f|^q)^{**}(s)) = \frac{\alpha}{es^2 + s} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\alpha - 1}, \quad s > 0.$$

Since

$$f^*(s) \asymp s^{-\frac{1}{q}} \left[\log\left(e + \frac{1}{s}\right) \right]^{-\frac{\alpha+1}{q}}$$
 for small enough $s > 0$,

we see that f also has a typical singularity of functions in $L^{q,\infty}(\log L)^{\alpha+1}$. These arguments suggest that $\mathcal{L}^{q,\infty}(\log \mathcal{L})^{\alpha}$ is closely related to $L^{q,\infty}(\log L)^{\alpha+1}$.

Proposition A.2. Let $1 \le q < \infty$ and $\alpha > 0$. Then there exists C > 0 such that

$$\|f\|_{\mathfrak{L}^{q,\infty}(\log\mathfrak{L})^{\alpha}} \le C \|f\|_{L^{q,\infty}(\log L)^{\alpha+1}}$$
(A-2)

for $f \in L^{q,\infty}(\log L)^{\alpha+1}$. Furthermore,

$$\inf\left\{\frac{\|f\|_{\mathcal{L}^{q,\infty}(\log \mathcal{L})^{\alpha}}}{\|f\|_{L^{q,\infty}(\log L)^{\alpha+1}}}: f \in L^{q,\infty}(\log L)^{\alpha+1}\right\} = 0.$$
(A-3)

Proof. By Lemma 2.3 and (2-1), it suffices to consider the case q = 1. We first prove (A-2) with q = 1. Let $f \in L^{1,\infty}(\log L)^{\alpha+1}$, where $\alpha > 0$. By Lemma 3.2 (2), for any R > 0, we have

$$sf^{**}(s) = \int_0^s f^*(\eta) \, d\eta \le \left(\int_0^s \eta^{-1} \left[\log\left(e + \frac{1}{\eta}\right)\right]^{-\alpha - 1} \, d\eta\right) \left(\sup_{\eta > 0} \left[\log\left(e + \frac{1}{\eta}\right)\right]^{\alpha + 1} \eta f^*(\eta)\right)$$
$$\le C \left[\log\left(e + \frac{1}{s}\right)\right]^{-\alpha} \left(\sup_{\eta > 0} \left[\log\left(e + \frac{1}{\eta}\right)\right]^{\alpha + 1} \eta f^*(\eta)\right)$$

for $s \in (0, R)$. This together with (2-9) implies that

$$\|f\|_{\mathfrak{L}^{1,\infty}(\log\mathfrak{L})^{\alpha}} = \sup_{0 < s < R} \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha} sf^{**}(s)$$
$$\leq C \sup_{\eta > 0} \left[\log\left(e + \frac{1}{\eta}\right) \right]^{\alpha+1} \eta f^{*}(\eta) = C \|f\|_{L^{1,\infty}(\log L)^{\alpha+1}}.$$

Thus (A-2) holds for q = 1, and the proof of (A-2) is complete.

Next, we prove (A-3) with q = 1. Let $\{f_n\}$ be a sequence in L^1_{loc} such that

$$f_n^*(s) = n [\log(e+n)]^{-\alpha-1} \chi_{(0,\frac{1}{n})}(s)$$

Since

$$f_n^{**}(s) = \begin{cases} n[\log(e+n)]^{-\alpha-1} & \text{for } s \in (0, \frac{1}{n}), \\ s^{-1}[\log(e+n)]^{-\alpha-1} & \text{for } s \in [\frac{1}{n}, \infty), \end{cases}$$
(A-4)

П

we have

$$\|f_n\|_{\mathfrak{L}^{1,\infty}(\log\mathfrak{L})^{\alpha}} = \sup_{s>0} \left[\log\left(e+\frac{1}{s}\right)\right]^{\alpha} sf_n^{**}(s) = \left[\log\left(e+\frac{1}{s}\right)\right]^{\alpha} sf_n^{**}(s)\Big|_{s=\frac{1}{n}} = \left[\log(e+n)\right]^{-1}$$

for n = 1, 2, ... On the other hand, similarly to (3-2), we find $L \in [e, \infty)$ such that

the function
$$(0, \infty) \ni \tau \mapsto \tau \left[\log \left(L + \frac{1}{\tau} \right) \right]^{\alpha + 1}$$
 is nondecreasing.

Then, by (3-1) and (A-4), we have

$$\|f_n\|_{L^{1,\infty}(\log L)^{\alpha+1}} = \sup_{s>0} \left[\log\left(e + \frac{1}{s}\right) \right]^{\alpha+1} s f_n^*(s)$$

$$\geq C \sup_{s>0} \left[\log\left(L + \frac{1}{s}\right) \right]^{\alpha+1} s f_n^*(s) = C \left[\log\left(L + \frac{1}{s}\right) \right]^{\alpha+1} s f_n^*(s) \Big|_{s=\frac{1}{n}} \geq C$$

for n = 1, 2, ... These imply (A-3). Thus Proposition A.2 follows.

Acknowledgements

Ishige and Kawakami were supported in part by JSPS KAKENHI grant number JP 19H05599. Kawakami was also supported in part by JSPS KAKENHI grant number JP 20K03689 and JP 22KK0035. Ioku was supported in part by JSPS KAKENHI grant number JP 21K18582.

References

- [Andreucci and DiBenedetto 1991] D. Andreucci and E. DiBenedetto, "On the Cauchy problem and initial traces for a class of evolution equations with strongly nonlinear sources", *Ann. Scuola Norm. Sup. Pisa Cl. Sci.* (4) **18**:3 (1991), 363–441. MR Zbl
- [Baras and Pierre 1985] P. Baras and M. Pierre, "Critère d'existence de solutions positives pour des équations semi-linéaires non monotones", *Ann. Inst. H. Poincaré Anal. Non Linéaire* **2**:3 (1985), 185–212. MR Zbl
- [Bennett and Sharpley 1988] C. Bennett and R. Sharpley, *Interpolation of operators*, Pure Appl. Math. **129**, Academic Press, Boston, MA, 1988. MR Zbl
- [Bogdan and Jakubowski 2007] K. Bogdan and T. Jakubowski, "Estimates of heat kernel of fractional Laplacian perturbed by gradient operators", *Comm. Math. Phys.* **271**:1 (2007), 179–198. MR Zbl
- [Brandolese and Karch 2008] L. Brandolese and G. Karch, "Far field asymptotics of solutions to convection equation with anomalous diffusion", J. Evol. Equ. 8:2 (2008), 307–326. MR Zbl
- [Brezis and Cazenave 1996] H. Brezis and T. Cazenave, "A nonlinear heat equation with singular initial data", *J. Anal. Math.* **68** (1996), 277–304. MR Zbl
- [Fujishima and Ioku 2021] Y. Fujishima and N. Ioku, "Solvability of a semilinear heat equation via a quasi scale invariance", pp. 79–101 in *Geometric properties for parabolic and elliptic PDEs*, Springer INdAM Ser. **47**, Springer, 2021. MR Zbl
- [Fujishima and Ioku 2022] Y. Fujishima and N. Ioku, "Global in time solvability for a semilinear heat equation without the self-similar structure", *Partial Differ. Equ. Appl.* **3**:2 (2022), art. id. 23. MR Zbl
- [Fujishima et al. 2023] Y. Fujishima, K. Hisa, K. Ishige, and R. Laister, "Solvability of superlinear fractional parabolic equations", *J. Evol. Equ.* **23**:1 (2023), art. id. 4. MR Zbl
- [Fujishima et al. 2024] Y. Fujishima, K. Hisa, K. Ishige, and R. Laister, "Local solvability and dilation-critical singularities of supercritical fractional heat equations", *J. Math. Pures Appl.* (9) **186** (2024), 150–175. MR Zbl
- [Fujita 1966] H. Fujita, "On the blowing up of solutions of the Cauchy problem for $u_t = \Delta u + u^{1+\alpha}$ ", J. Fac. Sci. Univ. Tokyo Sect. I **13** (1966), 109–124. MR Zbl
- [Giraudon and Miyamoto 2022] T. Giraudon and Y. Miyamoto, "Fractional semilinear heat equations with singular and nondecaying initial data", *Rev. Mat. Complut.* **35**:2 (2022), 415–445. MR Zbl
- [Grafakos 2008] L. Grafakos, Classical Fourier analysis, 2nd ed., Grad. Texts in Math. 249, Springer, 2008. MR Zbl
- [Hisa and Ishige 2018] K. Hisa and K. Ishige, "Existence of solutions for a fractional semilinear parabolic equation with singular initial data", *Nonlinear Anal.* **175** (2018), 108–132. MR Zbl
- [Hisa et al. 2023] K. Hisa, K. Ishige, and J. Takahashi, "Initial traces and solvability for a semilinear heat equation on a half space of \mathbb{R}^{N} ", *Trans. Amer. Math. Soc.* **376**:8 (2023), 5731–5773. MR Zbl
- [Ishige et al. 2014] K. Ishige, T. Kawakami, and K. Kobayashi, "Global solutions for a nonlinear integral equation with a generalized heat kernel", *Discrete Contin. Dyn. Syst. Ser. S* **7**:4 (2014), 767–783. MR Zbl
- [Ishige et al. 2020] K. Ishige, T. Kawakami, and S. Okabe, "Existence of solutions for a higher-order semilinear parabolic equation with singular initial data", *Ann. Inst. H. Poincaré C Anal. Non Linéaire* **37**:5 (2020), 1185–1209. MR Zbl
- [Ishige et al. 2022] K. Ishige, T. Kawakami, and S. Okabe, "Existence of solutions to nonlinear parabolic equations via majorant integral kernel", *Nonlinear Anal.* 223 (2022), art. id. 113025. MR Zbl
- [Kan and Takahashi 2017] T. Kan and J. Takahashi, "Time-dependent singularities in semilinear parabolic equations: existence of solutions", *J. Differential Equations* **263**:10 (2017), 6384–6426. MR Zbl
- [Kozono and Yamazaki 1994] H. Kozono and M. Yamazaki, "Semilinear heat equations and the Navier–Stokes equation with distributions in new function spaces as initial data", Comm. Partial Differential Equations 19:5-6 (1994), 959–1014. MR Zbl
- [Laister and Sierżęga 2020] R. Laister and M. Sierżęga, "Well-posedness of semilinear heat equations in L^1 ", Ann. Inst. H. Poincaré C Anal. Non Linéaire **37**:3 (2020), 709–725. MR Zbl
- [Laister and Sierżęga 2021] R. Laister and M. Sierżęga, "A blow-up dichotomy for semilinear fractional heat equations", *Math. Ann.* **381**:1-2 (2021), 75–90. MR Zbl
- [Laister et al. 2016] R. Laister, J. C. Robinson, M. Sierżęga, and A. Vidal-López, "A complete characterisation of local existence for semilinear heat equations in Lebesgue spaces", *Ann. Inst. H. Poincaré C Anal. Non Linéaire* **33**:6 (2016), 1519–1538. MR Zbl

- [Miyamoto 2021] Y. Miyamoto, "A doubly critical semilinear heat equation in the L^1 space", J. Evol. Equ. 21:1 (2021), 151–166. MR Zbl
- [Muckenhoupt 1972] B. Muckenhoupt, "Hardy's inequality with weights", Studia Math. 44 (1972), 31–38. MR Zbl
- [O'Neil 1963] R. O'Neil, "Convolution operators and L(p, q) spaces", Duke Math. J. 30 (1963), 129–142. MR Zbl
- [Quittner and Souplet 2007] P. Quittner and P. Souplet, *Superlinear parabolic problems: blow-up, global existence and steady states*, Birkhäuser, Basel, 2007. MR Zbl
- [Robinson and Sierżęga 2013] J. C. Robinson and M. Sierżęga, "Supersolutions for a class of semilinear heat equations", *Rev. Mat. Complut.* **26**:2 (2013), 341–360. MR Zbl
- [Sugitani 1975] S. Sugitani, "On nonexistence of global solutions for some nonlinear integral equations", *Osaka Math. J.* **12** (1975), 45–51. MR Zbl
- [Takahashi 2016] J. Takahashi, "Solvability of a semilinear parabolic equation with measures as initial data", pp. 257–276 in *Geometric properties for parabolic and elliptic PDE's*, Springer Proc. Math. Stat. **176**, Springer, 2016. MR Zbl
- [Tayachi and Weissler 2014] S. Tayachi and F. B. Weissler, "The nonlinear heat equation with high order mixed derivatives of the Dirac delta as initial values", *Trans. Amer. Math. Soc.* **366**:1 (2014), 505–530. MR Zbl
- [Wadade 2014] H. Wadade, "Optimal embeddings of critical Sobolev–Lorentz–Zygmund spaces", *Studia Math.* 223:1 (2014), 77–96. MR Zbl
- [Weissler 1981] F. B. Weissler, "Existence and nonexistence of global solutions for a semilinear heat equation", *Israel J. Math.* **38**:1-2 (1981), 29–40. MR Zbl
- [Zhanpeisov 2023] E. Zhanpeisov, "Existence of solutions to fractional semilinear parabolic equations in Besov–Morrey spaces", *Discrete Contin. Dyn. Syst.* **43**:11 (2023), 3969–3986. MR Zbl

Received 25 Oct 2023. Revised 21 Dec 2023. Accepted 8 Jun 2024.

NORISUKE IOKU: ioku@tohoku.ac.jp Mathematical Institute, Tohoku University, Sendai, Japan

KAZUHIRO ISHIGE: ishige@ms.u-tokyo.ac.jp Graduate School of Mathematical Sciences, University of Tokyo, Tokyo, Japan

TATSUKI KAWAKAMI: kawakami@math.ryukoku.ac.jp

Applied Mathematics and Informatics Course, Faculty of Advanced Science and Technology, Ryukoku University, Otsu, Japan





A MARCINKIEWICZ MULTIPLIER THEORY FOR SCHUR MULTIPLIERS

CHIAN YEONG CHUAH, ZHEN-CHUAN LIU AND TAO MEI

We prove a Marcinkiewicz-type multiplier theory for the boundedness of Schur multipliers on the Schatten *p*-classes. This generalizes a previous result of J. Bourgain for Toeplitz-type Schur multipliers and complements a recent result by J. Conde-Alonso et al. (*Ann. of Math.* (2) **198**:3 (2023), 1229–1260). As a corollary, we obtain a new unconditional decomposition for the Schatten *p*-classes, $1 . We extend our main result to the <math>\mathbb{Z}^d$ and \mathbb{R}^d cases, and include an operator-valued version of it using Pisier's noncommutative $L^{\infty}(\ell_1)$ -norm.

1. Introduction

Let $A \in B(H)$ be a bounded operator on a (separable) Hilbert space *H*. We can write *A* in its matrix representation

$$A = (a_{k,j})_{k,j \in \mathbb{Z}}$$

with $a_{k,j} = \langle Ae_k, e_j \rangle$ for a given orthonormal basis $\{e_k\}_{k \in \mathbb{Z}}$ of *H*. Given a bounded function *m* on $\mathbb{Z} \times \mathbb{Z}$, we call the map

$$M_m: (a_{k,j}) \mapsto (m(k,j)a_{k,j}) \tag{1-1}$$

a Schur multiplier with symbol *m*. The study of the boundedness of Schur multipliers with respect to the Schatten *p*-norms has a rich history [Bennett 1977; Arazy 1982; Bożejko and Fendler 1984; Berkson and Gillespie 1994; Pisier 1998; 2001; Harcharras 1999; Clément et al. 2000; Aleksandrov and Peller 2002; Doust and Gillespie 2005]. The recent study of noncommutative analysis on the approximation properties of operator functions and operator algebras [Haagerup et al. 2010; Neuwirth and Ricard 2011; Caspers and de la Salle 2015; Potapov et al. 2015; 2017; de Laat and de la Salle 2018; Caspers et al. 2019; Parcet et al. 2022; Mei et al. 2022; Conde-Alonso et al. 2023], especially the work [Lafforgue and de la Salle 2011] on the approximation property of higher-rank Lie groups, draws a lot of attention to the boundedness of Schur multipliers for the case where $1 . Conde-Alonso, González–Pérez, Parcet and Tablate [Conde-Alonso et al. 2023] recently proved a Hörmander–Mikhlin-type Schur multiplier theory for <math>S^p$, 1 , in their remarkable work.

In this article, we prove a Marcinkiewicz-type Schur multiplier theory. Hörmander–Mikhlin-type multipliers and Marcinkiewicz-type multipliers are rooted in classical Fourier analysis. Like their counterpart in Fourier analysis, Marcinkiewicz-type Schur multipliers are a larger class of multipliers

MSC2020: primary 46B28, 46L52; secondary 42A45.

Keywords: Schur multipliers, Schatten *p*-classes, Littlewood–Paley theory, Marcinkiewicz multiplier theory, noncommutative L^p spaces.

^{© 2025} The Authors, under license to MSP (Mathematical Sciences Publishers). Distributed under the Creative Commons Attribution License 4.0 (CC BY). Open Access made possible by subscribing institutions via Subscribe to Open.

and their *p*-boundedness is more subtle; it was shown in [Tao and Wright 2001] that the L^p -bounds of Marcinkiewicz Fourier multipliers are of order $p^{3/2}$ as $p \to \infty$.

When *m* is of Toeplitz-type, i.e., $m(k, j) = \dot{m}(k - j)$ for some function $\dot{m} : \mathbb{Z} \to \mathbb{C}$, one may apply a well-known transference method and obtain bounded Schur multipliers from the classical Fourier multiplier theory. J. Bourgain's work [1986, Theorem 4, Corollary 20] on scalar-valued Fourier multipliers acting on Schatten *p*-valued functions implies that the following Marcinkiewicz-type condition is sufficient for the boundedness of M_m on the Schatten *p*-classes for all 1 :

$$\sum_{2^{n-1} \le |k| < 2^n} |\dot{m}(k+1) - \dot{m}(k)| < C$$
(1-2)

for all $n \in \mathbb{N}$. Let $\dot{m}_{\varepsilon}(k) = \varepsilon_n$ for $2^{n-1} \le |k| < 2^n$. Then the associated multiplier $M_{m_{\varepsilon}}$ is bounded for any sequence $\varepsilon_n = \pm 1$.

To extend Bourgain's result to general non-Toeplitz-type Schur multipliers, one may ask whether the condition that

$$\sum_{2^{n-1} \le |k| < 2^n} |m(k+j+1,j) - m(k+j,j)| < C$$
(1-3)

for all $n \in \mathbb{N}$, $j \in \mathbb{Z}$ implies the S^p boundedness of general Schur multipliers. The answer is yes if m is Toeplitz since condition (1-3) reduces to Bourgain's condition (1-2) in that case. The answer would be yes for general Schur multipliers as well if the family $M_{m_{\varepsilon}}$, defined after condition (1-2), is R-bounded for any family of sequences $\varepsilon = (\varepsilon_k)_k$ valued in $\{\pm 1\}$. This implication was proved in the works of Berkson and Gillespie [1994], Doust and Gillespie [2005] and Clément, de Pagter, Sukochev and Witvliet [Clément et al. 2000], in which they studied the connection between vector-valued Littlewood–Paley theory and Marcinkiewicz multiplier theory. We show in Section 5.1 that this is not true in general and the condition (1-3) is not sufficient for the S^p -boundedness of the associated Schur multiplier.¹

The main result of this article is the following.

Theorem 1.1. M_m defined in (1-1) extends to a bounded map on the Schatten *p*-classes S^p for all $1 with bounds <math>\leq (p^2/(p-1))^3$ if *m* is bounded and there exists a constant *C* such that

$$\sum_{2^{n-1} < |k| < 2^n} |m(k+j+1,j) - m(k+j,j)| < C,$$
(1-4)

$$\sum_{2^{n-1} \le |k| < 2^n} |m(j, k+j+1) - m(j, k+j)| < C$$
(1-5)

for all $n \in \mathbb{N}$, $j \in \mathbb{Z}$.

The writing of this article was motivated by the recent article [Conde-Alonso et al. 2023], although the third author had known Theorem 1.1 previously. The authors of [loc. cit.] further studied Schatten-p-classes indexed in d-dimensional Euclidean spaces, aiming for possible applications to the approximation properties of higher-rank Lie groups. Following this trend, we extend Theorem 1.1 to the higher-dimensional

¹This also shows that the family of "Littlewood–Paley operators" $M_{m_{\varepsilon}}$ mentioned above is *not R*-bounded.

cases as well. The proof of Theorem 1.1 relies on the crucial property that a Schur multiplier is an operatorvalued Fourier multiplying from the left, and is simultaneously an operator-valued Fourier multiplier multiplying from the right. This property was already used by the authors of [Conde-Alonso et al. 2023] in proving a Hörmander–Mikhlin-type criterion for the boundedness of Schur multipliers.

Theorem 1.1 implies a new unconditional decomposition for Schatten *p* classes. For $(n, \ell) \in \mathbb{Z} \times \mathbb{Z}$, let $E_{0,\ell} = \{(\ell, \ell)\} \subset \mathbb{Z} \times \mathbb{Z}$. Let

$$E_{n,\ell} = \{ (k, j) \in \mathbb{Z} \times \mathbb{Z} : 2^{n-1} \le k - j < 2^n, \ \ell 2^n \le k < (\ell+1)2^n \}$$

for n > 0, and

$$E_{n,\ell} = \{(k, j) \in \mathbb{Z} \times \mathbb{Z} : -2^{|n|} < k - j \le -2^{|n|-1}, \ \ell 2^{|n|} \le k < (\ell+1)2^{|n|}\}$$

for n < 0. We then have the decomposition

$$\mathbb{Z} \times \mathbb{Z} = \bigcup_{(n,\ell) \in \mathbb{Z} \times \mathbb{Z}} E_{n,\ell}.$$

Let $P_{n,\ell}$ be the projection onto span $\{e_{k,j}, (k, j) \in E_{n,\ell}\}$. It is easy to see that $\sum_{n,\ell} \varepsilon_{n,\ell} P_{n,\ell}$ is a Schur multiplier satisfying the assumptions of Theorem 1.1 for any bounded sequence $\varepsilon_{n,\ell}$. We then obtain an unconditional decomposition of S^p .

Corollary 1.2. $\sum_{n,\ell} \varepsilon_{n,\ell} P_{n,\ell}$ extends to a bounded map on S^p for all $1 for any bounded sequence <math>\varepsilon_{n,\ell}$.

We will prove Theorem 1.1 in Section 3. We will explain how to extend Theorem 1.1 to the higherdimensional case in Section 4 and explain that the ball-type Schur multipliers remain bounded on $S^p(\ell^2(\mathbb{Z}^d))$ for d > 1 (Example 4.4), contrary to the behavior of Fourier multipliers. We will show that the condition (1-4) alone is not sufficient in Section 5.1 and explain an operator-valued version of Theorem 1.1 in Section 5.2.

2. Preliminaries

Given $d \in \mathbb{N}$, denote by $\mathcal{B}(\ell^2(\mathbb{Z}^d))$ the set of bounded linear operators on $\ell^2(\mathbb{Z}^d)$. We represent the operator $A \in \mathcal{B}(\ell^2(\mathbb{Z}^d))$ as $A = (a_{i,j})_{(i,j) \in \mathbb{Z}^d \times \mathbb{Z}^d}$ with $a_{i,j} = \langle Ae_i, e_j \rangle$ for the canonical basis $\{e_i\}$ of $\ell^2(\mathbb{Z}^d)$. Given a bounded function *m* on $\mathbb{Z}^d \times \mathbb{Z}^d$, the associated Schur multiplier

$$M_m(A) = (m(i, j)a_{ij})$$

extends to a bounded operator on the Hilbert–Schmidt class $S^2(\ell^2(\mathbb{Z}^d))$. We call *m* the symbol of M_m . Recall that the Schatten *p*-class S^p , $1 \le p < \infty$, is the collection of all compact operators *A* with a finite Schatten-*p* norm, which is defined as

$$||A||_{p} = (\operatorname{tr}[(A^{*}A)^{\frac{p}{2}}])^{\frac{1}{p}} = \left(\sum_{i} s_{i}^{p}\right)^{\frac{1}{p}}$$
(2-1)

for $1 \le p < \infty$, where s_i is the *i*-th singular value of *A*. The Schatten *p* norm is unitary invariant and does not depend on the choice of the orthonormal basis. The Schatten-class S^p , $1 \le p < \infty$, and

 $B(\ell^2(\mathbb{Z}^d))$ share many similar properties with ℓ^p , $1 \le p \le \infty$. In particular, the dual space of S^1 (resp. S^p , $1) is isomorphic to <math>B(\ell^2)$ (resp. $S^{p/(p-1)}$). The family forms an interpolation scale

$$[B(\ell^2), S^1]_{\frac{1}{p}} = S^p$$

for $1 . However, <math>S^p$ does not admit an unconditional basis whenever $p \neq 2$. We will prove that, for *m* satisfying additional conditions (1-4) and (1-5), M_m extends to a bounded map on S^p for all $1 , which immediately implies the unconditional decomposition for <math>S^p$ as stated in Corollary 1.2.

Given $d \in \mathbb{N}$, we denote by $L^p(\mathbb{T}^d; S^p)$ the space of S^p -valued Bochner integrable functions f such that

$$||f||_{L^p} = \left(\operatorname{tr} \left[\int_{[0,1)^d} |f|^p(z) \, d\theta \right] \right)^{\frac{1}{p}} < \infty.$$

Here we let $z = e^{i2\pi\theta}$, with $\theta \in [0, 1)^d$, and $|f|^p = (f^*f)^{p/2}$ is defined via the functional calculus. For $f \in L^p(\mathbb{T}^d; S^p)$ with 1 , we have the Fourier expansion

$$f(z) \sim \sum_{k \in \mathbb{Z}^d} \hat{f}(k) z^k,$$

with $\hat{f}(k) = \int_{[0,1]^d} f(z) \bar{z}^k d\theta \in S^p$. Given *R* a finite subset of \mathbb{Z}^d , denote by $S_R f$ the partial Fourier sum

$$S_R f(z) = \sum_{k \in \mathbb{R}} \hat{f}(k) z^k.$$
(2-2)

Choose $\delta \in C^{\infty}(\mathbb{R})$ such that $0 \leq \delta \leq 1$, $\operatorname{supp}(\delta) \subset \left[-2\sqrt{d}, -\frac{1}{4}\right] \cup \left[\frac{1}{4}, 2\sqrt{d}\right]$ and $\delta(x) = 1$ when $\frac{1}{2} \leq |x| \leq \sqrt{d}$. For $j \geq 0$, define $\delta_j(x) := \delta(2^{-j}x)$. For $f \in L^2(\mathbb{T}^d; S^2)$, we define

$$S_j f(z) = \sum_{k \in \mathbb{Z}^d} \delta_j(|k|_2) \hat{f}(k) z^k$$

We denote by $|k|_2$ the ℓ_2 norm of $k \in \mathbb{Z}^d$ in the formula above and will denote by $|k|_{\infty}$ the ℓ_{∞} norm of k. Let $(E_j)_{j\geq 0}$ be the cubes with squared holes in \mathbb{Z}^d given by

$$E_{j} = \begin{cases} \{k \in \mathbb{Z}^{d} : 2^{j-1} \le |k|_{\infty} < 2^{j}\}, & j > 0, \\ \{0\}, & j = 0. \end{cases}$$
(2-3)

Note our construction implies $S_{E_i}S_j = S_{E_i}$, which we will need later.

For a sequence (f_k) in $L^p(\mathbb{T}^d; S^p)$, we use the classical notation

$$\|(f_k)\|_{L^p(\ell_2^c)} = \left\|\left(\sum_k |f_k|^2\right)^{\frac{1}{2}}\right\|_{L^p(\mathbb{T}^d;S^p)}, \quad \|(f_k)\|_{L^p(\ell_2^c)} = \left\|\left(\sum_k |f_k^*|^2\right)^{\frac{1}{2}}\right\|_{L^p(\mathbb{T}^d;S^p)},$$

and

$$\|(f_k)\|_{L^p(\ell_2)} = \begin{cases} \max\{\|(f_k)\|_{L^p(\ell_2^c)}, \|(f_k^*)\|_{L^p(\ell_2^c)}\} & \text{if } 2 \le p \le \infty, \\ \inf_{y_k + z_k = f_k} \|(y_k)\|_{L^p(\ell_2^c)} + \|(z_k)\|_{L^p(\ell_2^c)} & \text{if } 0$$

The above definition is justified by the following noncommutative Khintchine inequality:

Lemma 2.1 [Lust-Piquard 1986; Lust-Piquard and Pisier 1991]. Let (ε_k) be a sequence of independent Rademacher random variables. Then, for $1 \le p < \infty$,

$$\alpha_p^{-1} E_{\varepsilon} \left\| \sum_k \varepsilon_k \otimes f_k \right\|_{L^p(\mathbb{T}^d; S^p)} \le \|(f_k)\|_{L^p(\ell_2)} \le \beta_p E_{\varepsilon} \left\| \sum_k \varepsilon_k \otimes f_k \right\|_{L^p(\mathbb{T}^d; S^p)}.$$
(2-4)

The optimal constant β_p is no greater than $\sqrt{3}$ for $1 \le p \le 2$ and is 1 for $p \ge 2$ (see [Haagerup and Musat 2007]); α_p is 1 for $1 \le p \le 2$ and is of order \sqrt{p} as $p \to \infty$. Inequality (2-4) was pushed further to the case where 0 (see [Pisier and Ricard 2017]).

We will need the following noncommutative Littlewood–Paley theorem on \mathbb{Z}^d .

Lemma 2.2. There is a constant $C_d > 0$ that depends only on d such that

$$\|(S_j f)_{j\geq 0}\|_{L^p(\ell_2)} \le C_d \frac{p^2}{p-1} \|f\|_{L^p(\mathbb{T}^d; S^p)},$$
(2-5)

$$\|f\|_{L^{p}(\mathbb{T}^{d};S^{p})} \leq C_{d} \frac{p^{2}}{p-1} \|(S_{E_{j}}f)_{j\geq 0}\|_{L^{p}(\ell_{2})}$$
(2-6)

for all $f \in L^p(\mathbb{T}^d; S^p)$ and 1 .

Proof. This lemma is well known. We explain here that the dependence of the constants on p is in the order of $p^2/(p-1)$. Given $\varepsilon_j = \pm 1$, let $M_{\varepsilon} = \sum_{j \ge 0} \varepsilon_j S_j$. Our choice of S_j 's makes M_{ε} a so-called Hörmander–Mikhlin multiplier, which in particular is a Calderón–Zygmund operator. So it is bounded from the classical Hardy space H^1 to L^1 . Moreover, it is from H^1 to H^1 since it commutes with the classical Hilbert transform. By [Mei 2007, Theorem 6.4], it extends to a bounded operator on the semicommutative BMO space $BMO_{cr}(L^{\infty}(\mathbb{T}^d) \otimes \mathcal{B}(\ell^2(\mathbb{Z}^d)))$. Inequality (2-5) then follows from the interpolation result [Mei 2007, Theorem 6.2] and the Khintchine inequality (2-4). Inequality (2-6) follows from (2-5) by duality because of the identity $\langle f, g \rangle = \sum_j \langle S_{E_j} f, S_{E_j} g \rangle = \sum_j \langle S_j f, S_{E_j} g \rangle$.

Lemma 2.3. Suppose R_j is a family of boxes with sides parallel to the axes in \mathbb{R}^d . Then there is a constant $C_d > 0$ that depends only on d such that, for all $1 and for all families of measurable functions <math>f_j$ on \mathbb{R}^d , we have

$$\left\| \left(\sum_{j} |S_{R_{j}}(f_{j})|^{2} \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T}^{d};S^{p})} \leq C_{d} \left(\frac{p^{2}}{p-1} \right)^{d} \left\| \left(\sum_{j} |f_{j}|^{2} \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T}^{d};S^{p})}.$$
(2-7)

Proof. Assume d = 1 and $R_j = [a_j, b_j]$. Let T_a to be the operator that sends $f(\cdot)$ to $f(\cdot)e^{i2\pi a_j(\cdot)}$. Let P_+ be the analytic projection. Then

$$S_{R_j} = T_{a_j} P_+ T_{-a_j} - T_{b_j} P_+ T_{-b_j}.$$

Note that $|T_{a_j}f_j|^2 = |f_j|^2$ and we obtain the inequality for d = 1 by the boundedness of P_+ . The case d > 1 holds due to Fubini's theorem.

Lemma 2.4. For sequences $(a_n), (c_n) \in B(H)$, we have

$$\left|\sum_{n} a_n^* c_n\right|^2 \le \left\|\sum_{n} a_n^* a_n\right\| \left(\sum_{n} c_n^* c_n\right).$$
(2-8)

Proof. Given any $v \in S^2$, we have by the Cauchy–Schwarz inequality that

$$\operatorname{tr}\left(v^* \left|\sum_n a_n^* c_n\right|^2 v\right) = \operatorname{tr}\left(\left|\sum_n a_n^* c_n v\right|^2\right) \le \left\|\sum_n a_n^* a_n\right\| \operatorname{tr}\left[v^* \left(\sum_n c_n^* c_n\right) v\right].$$

Since v is arbitrary, we obtain (2-8).

3. Proof of Theorem 1.1

For $z \in \mathbb{T}^d$ given, let Π_z be the *-homomorphism on $\mathcal{B}(\ell^2(\mathbb{Z}^d))$ defined as

$$\Pi_z(A) = U_z A U_z^*$$

with U_z the unitary sending e_k to $z^k e_k$. It is easy to see that Π_z has the presentation

$$\Pi_z : A = (a_{k,j}) \longmapsto (a_{k,j} z^{k-j}), \tag{3-1}$$

with $k, j \in \mathbb{Z}^d$. Π_z defines an isometric isomorphism on $\mathcal{B}(\ell^2(\mathbb{Z}^d))$ and $S^p(\ell^2(\mathbb{Z}^d))$ for all $1 \le p < \infty$ because all these norms are unitary invariant. Considering *z* as a variable on \mathbb{T}^d , define $\Pi : \mathcal{B}(\ell^2(\mathbb{Z}^d)) \to L^{\infty}(\mathbb{T}^d) \otimes \mathcal{B}(\ell^2(\mathbb{Z}^d))$ as

$$\Pi(A)(z) = \Pi_z(A). \tag{3-2}$$

Then Π is an isometric isomorphism from $\mathcal{B}(\ell^2(\mathbb{Z}^d))$ to $L^{\infty}(\mathbb{T}^d) \otimes \mathcal{B}(\ell^2(\mathbb{Z}^d))$ and from $S^p(\ell^2(\mathbb{Z}^d))$ to $L^p(\mathbb{T}^d; S^p(\ell^2(\mathbb{Z}^d)))$ for all $1 \le p < \infty$.

Given a symbol $m = (m(i, j))_{i,j \in \mathbb{Z}^d}$ and $A = (a_{i,j})_{i,j \in \mathbb{Z}^d} \in S^p(\ell^2(\mathbb{Z}^d))$, set

$$M_{l}(n) = \sum_{s \in \mathbb{Z}^{d}} m(s, s - n) e_{s,s}, \quad M_{r}(n) = \sum_{s \in \mathbb{Z}^{d}} m(s + n, s) e_{s,s},$$

$$A(n) = \sum_{s \in \mathbb{Z}^{d}} a_{s,s-n} e_{s,s-n} = \sum_{s \in \mathbb{Z}^{d}} a_{s+n,s} e_{s+n,s}$$
(3-3)

for $n \in \mathbb{Z}^d$. Here $e_{s,t}$ denotes the operator on $\ell^2(\mathbb{Z}^d)$ sending e_t to e_s . Then $M_l(n), M_r(n) \in \mathcal{B}(\ell^2(\mathbb{Z}^d))$ with norm bounded by $C, A(n) \in S^p(\ell^2(\mathbb{Z}^d))$ for all $n \in \mathbb{Z}^d$, and

$$\Pi(A)(z) = \sum_{n} A(n)z^{n}, \quad M_{m}(A) = \sum_{n} M_{l}(n)A(n) = \sum_{n} A(n)M_{r}(n).$$

Here $M_l(n)A(n)$ and $A(n)M_r(n)$ denote the products of operators in $\mathcal{B}(\ell^2(\mathbb{Z}^d))$. Let $f = \Pi(A)$, i.e.,

$$f(z) = \sum_{n \in \mathbb{Z}^d} A(n) z^n.$$
(3-4)

Denote by $\Pi(S^p)$ the image of Π , i.e., the subspace of $L_p(\mathbb{T}^d; S^p)$ consisting of all f in the form of (3-4). Define the operator-valued Fourier multiplier T_M on $\Pi(S^p)$ as

$$T_M f(z) = \sum_{n \in \mathbb{Z}^d} M_l(n) A(n) z^n.$$
(3-5)

1516

Note that $M_l(n)A(n) = A(n)M_r(n)$ for all $n \in \mathbb{Z}^d$; we can represent T_M as a multiplier from the right:

$$T_M f(z) = \sum_{n \in \mathbb{Z}^d} A(n) M_r(n) z^n.$$
(3-6)

 T_M is defined so that the following identity holds:

$$T_M f = \Pi(M_m(A)).$$

Since Π is a trace preserving *-homomorphism, we have

$$\|A\|_{S^{p}} = \|f\|_{L^{p}(\mathbb{T}^{d};S^{p})}, \quad \|M_{m}A\|_{S^{p}} = \|T_{M}(f)\|_{L^{p}(\mathbb{T}^{d};S^{p})}.$$
(3-7)

In order to prove M_m 's boundedness on S^p , we only need to prove that T_M is bounded on $\Pi(S^p)$, i.e., the subspace of $L_p(\mathbb{T}^d; S^p)$ consisting all f in the form of (3-4). By Lemma 2.2 and the transference relation (3-7), it is sufficient to show the inequality

$$\left\| \left(\sum_{j \ge 0} |S_{E_j}(T_M f)|^2 \right)^{\frac{1}{2}} \right\|_{L_p(\mathbb{T}^d; S^p)} \le C_d \left(\frac{p^2}{p-1} \right)^d \left\| \left(\sum_{j \ge 0} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L_p(\mathbb{T}^d; S^p)}$$
(3-8)

and its adjoint form

$$\left\| \left(\sum_{j \ge 0} |(S_{E_j}(T_M f))^*|^2 \right)^{\frac{1}{2}} \right\|_{L_p(\mathbb{T}^d; S^p)} \le C_d \left(\frac{p^2}{p-1} \right)^d \left\| \left(\sum_{j \ge 0} |(S_j f)^*|^2 \right)^{\frac{1}{2}} \right\|_{L_p(\mathbb{T}^d; S^p)}$$
(3-9)

for $p \ge 2$. By duality, we will obtain M_m 's boundedness on S^p for 1 as well.

We will use (3-5) as the presentation of T_M to prove (3-8) and will use the presentation (3-6) to prove (3-9). Note that E_j is symmetric so $(S_{E_j}(T_M f))^* = S_{E_j}(T_M f)^*$ and

$$(T_M f)^* = \left(\sum_{n \in \mathbb{Z}^d} A(n) M_r(n) z^n\right)^* = \sum_{n \in \mathbb{Z}^d} (M_r(-n))^* (A(-n))^* z^n.$$

So, both $S_{E_j}(T_M f)$ and $(S_{E_j}(T_m f))^*$ have the multiplier symbols on the left. This allows us to write the corresponding squares in the forms with M_r or M_l sitting in the middle for both $S_{E_j}(T_M f)$ and $(S_{E_j}(T_m f))^*$ and avoid the usual trouble caused by the noncommutativity of the operator products. After noting these facts, the argument for the case d = 1 is rather standard, which we record below.

Proof of Theorem 1.1. We now set d = 1. By the notation in (3-3), the conditions (1-4) and (1-5) are equivalent to

$$\sup_{j\geq 0} \left\| \sum_{n\in E_j} |M_l(n+1) - M_l(n)| \right\|_{\infty} < C, \quad \sup_{j\geq 0} \left\| \sum_{n\in E_j} |M_r(n+1) - M_r(n)| \right\|_{\infty} < C, \tag{3-10}$$

where E_j is defined as in (2-3). Following the definition (2-2), we define $S_{(a,b)}$ as

$$S_{(a,b)}g(z) = \sum_{a < n < b} \hat{g}(n)z^n$$
(3-11)

for $g(z) = \sum_{n \in \mathbb{Z}} \hat{g}(n) z^n \in L^p(\mathbb{T}; S^p)$. We will deliberately extend the use of this notation and set

$$S_{(a,b)}g = -S_{(b,a)}g$$

when a > b. For $j \in \mathbb{N}$, write $E_{j,1} = (-2^j, -2^{j-1}]$, $E_{j,2} = [2^{j-1}, 2^j)$. Let $2_1^{(j)} = -2^j, 2_2^{(j)} = 2^j$ and

$$\Delta M_l(n) = \begin{cases} M_l(n) - M_l(n-1), & n < 0, \\ M_l(n+1) - M_l(n), & n > 0. \end{cases}$$
(3-12)

By applying summation by parts and the presentation of $T_M f$ in (3-5) and (3-6), we obtain

$$S_{E_j}(T_M f) = \sum_{i=1,2} S_{E_{j,i}}(T_M f) = \sum_{i=1,2} \left(M_l(2_i^{(j-1)})(S_{E_{j,i}}f) + \sum_{n \in E_{j,i}} \Delta M_l(n)(S_{(n,2_i^{(j)})}f) \right)$$
(3-13)
$$\sum_{i=1,2} \left((f_i - f_i) M_i(2_i^{(j-1)}) + \sum_{n \in E_{j,i}} \Delta M_l(n)(S_{(n,2_i^{(j)})}f) \right)$$
(3-14)

$$= \sum_{i=1,2} \left((S_{E_{j,i}}f) M_r(2_i^{(j-1)}) + \sum_{n \in E_{j,i}} (S_{(n,2_i^{(j)})}f) \Delta M_r(n) \right), \quad (3-14)$$

with $\Delta M_r(n)$ defined similarly. We will use the presentation (3-13) to prove (3-8) and will use (3-14) to prove (3-9). The arguments are similar. So we will only give the argument for (3-8). We will ignore the term j = 0 in (3-8) because $\|S_{E_0}(T_M f)\|_{L^p(\mathbb{T};S^p)} \leq C \|f\|_{L^p(\mathbb{T};S^p)}$.

Note $\Delta M_l(n)$ is a diagonal operator; we can write $\Delta M_l(n) = a_n^* b_n$, with a_n, b_n diagonal operators and $|a_n|^2 = |b_n|^2 = |\Delta M_l(n)|$. Then by Lemma 2.4, we have, for i = 1, 2,

$$\left|\sum_{n \in E_{j,i}} \Delta M_{\ell}(n) S_{(n,2_{i}^{(j)})} f\right|^{2} \leq \left\|\sum_{n \in E_{j,i}} |\Delta M_{l}(n)|\right\|_{\infty} \left(\sum_{n \in E_{j,i}} |b_{n} S_{(n,2_{i}^{(j)})} f|^{2}\right)$$
$$\leq C \left(\sum_{n \in E_{j,i}} |b_{n} S_{(n,2_{i}^{(j)})} f|^{2}\right)$$
(3-15)

$$= C\left(\sum_{n \in E_{j,i}} |S_{(n,2_i^{(j)})}(b_n S_j f)|^2\right).$$
(3-16)

Thus,

$$\left\| \left(\sum_{j \in \mathbb{N}} \left| \sum_{n \in E_{j,i}} \Delta M_{\ell}(n) S_{(n,2_i^{(j)})} f \right|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T};S^p)} \leq C^{\frac{1}{2}} \left\| \left(\sum_{j \in \mathbb{N}} \sum_{n \in E_{j,i}} |S_{(n,2_i^{(j)})}(b_n S_j f)|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T};S^p)}.$$

By Lemma 2.3, we get

$$\begin{split} \left\| \left(\sum_{j \in \mathbb{N}} \left| \sum_{n \in E_{j,i}} \Delta M_{\ell}(n) S_{(n,2_{i}^{(j)})} f \right|^{2} \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T};S^{p})} \\ &\leq C_{2} \frac{p^{2}}{p-1} C^{\frac{1}{2}} \left\| \left(\sum_{j \in \mathbb{N}} \sum_{n \in E_{j,i}} |(b_{n}S_{j}f)|^{2} \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T};S^{p})} \\ &= C_{2} \frac{p^{2}}{p-1} C^{\frac{1}{2}} \left\| \left(\sum_{j \in \mathbb{N}} (S_{j}f)^{*} \left(\sum_{n \in E_{j,i}} |b_{n}|^{2} \right) S_{j}f \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T};S^{p})} \\ &= C_{2} \frac{p^{2}}{p-1} C^{\frac{1}{2}} \left\| \left(\sum_{j \in \mathbb{N}} (S_{j}f)^{*} \left(\sum_{n \in E_{j,i}} |\Delta M_{l}(n)| \right) S_{j}f \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T};S^{p})} \\ &\leq C_{2} \frac{p^{2}}{p-1} C \left\| \left(\sum_{j \in \mathbb{N}} |S_{j}f|^{2} \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T};S^{p})}. \end{split}$$
(3-17)

Hence, by (3-13), (3-17), and Lemma 2.3

$$\begin{split} \left\| \left(\sum_{j \in \mathbb{N}} |S_{E_{j,i}}(T_M f)|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}; S^p)} \\ &\leq \left\| \left(\sum_{j \in \mathbb{N}} |M_l(2_i^{(j-1)})(S_{E_{j,i}}f)|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}; S^p)} + C \frac{p^2}{p-1} C_2 \left\| \left(\sum_{j \in \mathbb{N}} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}; S^p)} \\ &\leq \left\| C \left(\sum_{j \in \mathbb{N}} |(S_{E_{j,i}}f)|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}; S^p)} + C \frac{p^2}{p-1} C_2 \left\| \left(\sum_{j \in \mathbb{N}} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}; S^p)} \\ &= \left\| C \left(\sum_{j \in \mathbb{N}} |(S_{E_{j,i}}S_j f)|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}; S^p)} + C \frac{p^2}{p-1} C_2 \left\| \left(\sum_{j \in \mathbb{N}} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}; S^p)} \\ &\leq C \frac{p^2}{p-1} \left\| \left(\sum_{j \in \mathbb{N}} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}; S^p)} \end{split}$$

for i = 1, 2. Therefore we finish the proof of (3-8). The arguments for the adjoint version of (3-9) are similar. We then complete the proof of Theorem 1.1.

Corollary 3.1 [Conde-Alonso et al. 2023, Corollary 3.5]. *The following Mikhlin conditions imply the boundedness of* M_m *on* S^p *for all* 1 :

$$|m(s, s+k) - m(s, s+k+1)| \le \frac{C}{|k|},\tag{3-18}$$

$$|m(s+k,s) - m(s+k+1,s)| \le \frac{C}{|k|}.$$
(3-19)

Proof. It is clear that the Mikhlin conditions (3-18), (3-19) imply the Marcinkiewicz-type conditions (1-4), (1-5).

4. The case d > 1

In this part, we generalize Theorem 1.1 to the d-dimensional case. Before we proceed to the main statement of the theorem, we need to borrow some notation from the calculus of finite differences.

Definition 4.1. Let $\sigma : \mathbb{Z}^d \to \mathbb{C}$ and $j = (j_1, \ldots, j_d) \in \mathbb{Z}^d$. Let $\{e_j\}_{j=1}^d$ be standard basis of \mathbb{Z}^d , i.e., the *j*-th entry of e_j is 1 and all other entries are 0 for $j = 1, \ldots, d$. We define the forward partial difference operators Δ_{t_j} by

$$\Delta_{t_i}\sigma(t) := \sigma(t+e_j) - \sigma(t), \tag{4-1}$$

and for $\alpha \in \{0, 1\}^d$, define

$$\Delta_t^{\alpha} := \Delta_{t_1}^{\alpha_1} \cdots \Delta_{t_d}^{\alpha_d}$$

For $\alpha = (1, ..., 1) \in \{0, 1\}^d$, we simplify the notation Δ_t^{α} as Δ_t . Readers can find more information on the calculus of finite differences in Chapter 3 of [Ruzhansky and Turunen 2010].

4.1. The case d = 2. Recall that we have the partition $\mathbb{Z}^2 = \bigcup_{j \ge 0} E_j$ with E_j defined as

$$E_{j} = \begin{cases} \{(0,0)\}, & j = 0, \\ \{(n_{1},n_{2}) \in \mathbb{Z}^{2} : 2^{j-1} \le |(n_{1},n_{2})|_{\infty} < 2^{j}\}, & j \ge 1. \end{cases}$$
(4-2)

Theorem 4.2. Given $m = (m_{s,t})_{s,t \in \mathbb{Z}^2} \in \mathcal{B}(\ell^2(\mathbb{Z}^2))$, suppose *m* satisfies:

- (i) $\sup_{s,t\in\mathbb{Z}^2} |m_{s,t}| < C_1$.
- (ii) For any $k \in \mathbb{N}$, $s \in \mathbb{Z}^2$, there are constants C_2 , C_3 such that

$$\left(\sum_{t=(t_1,\pm 2^{k-1})\in E_k} |\Delta_{t_1} m_{s,s+t}| + \sum_{t=(\pm 2^{k-1},t_2)\in E_k} |\Delta_{t_2} m_{s,s+t}|\right) < C_2,$$
(4-3)

$$\sum_{t=(t_1,t_2)\in E_k} |\Delta_t m_{s,s+t}| < C_3, \tag{4-4}$$

and

$$\left(\sum_{t=(t_1,\pm 2^{k-1})\in E_k} |\Delta_{t_1}m_{s+t,s}| + \sum_{t=(\pm 2^{k-1},t_2)\in E_k} |\Delta_{t_2}m_{s+t,s}|\right) < C_2,$$
(4-5)

$$\sum_{e(t_1,t_2)\in E_k} |\Delta_t m_{s+t,s}| < C_3.$$
(4-6)

Then M_m is a bounded Schur multiplier on $S^p(\ell^2(\mathbb{Z}^2))$ for $p \in (1, \infty)$ with an upper bound $\leq (p^2/(p-1))^4$. Here C_1, C_2 and C_3 are positive absolute constants.

Now we come to the proof of Theorem 4.2. As explained at the beginning of Section 3, we only need to prove (3-8) and its adjoint version. Recall that S_{E_j} is the partial sum projection on $L^p(\mathbb{T}^2; S^p(\mathbb{Z}^2))$ given by $S_{E_j} f(z) = \sum_{n \in E_j} \hat{f}(n) z^n$, where $z \in \mathbb{T}^2$.

Applying the definition of M_l (3-3), we see that (4-3) and (4-4) imply

$$\left\|\sum_{n=(n_1,\pm 2^{j-1})\in E_j} |\Delta_{n_1} M_l(n)|\right\|_{\infty} + \left\|\sum_{n=(\pm 2^{j-1},n_2)\in E_j} |\Delta_{n_2} M_l(n)|\right\|_{\infty} < C_2,$$
(4-7)

$$\left\|\sum_{n=(n_1,n_2)\in E_j} |\Delta_n M_l(n)|\right\|_{\infty} < C_3.$$
(4-8)

To prove (3-8), we will cut E_j into four rectangles $E_{j,k}$, k = 1, ..., 4, for $j \ge 1$. Let $I_j = [2^{j-1}, 2^j) \cap \mathbb{Z}$, $J_j = [-2^{j-1}, 2^j) \cap \mathbb{Z}$, and set

$$E_{j,1} = J_j \times I_j,$$
 $E_{j,2} = (-I_j) \times J_j,$
 $E_{j,3} = I_j \times (-J_j),$ $E_{j,4} = (-J_j) \times (-I_j).$

Thus, we have

$$S_{E_j} T_M f = \sum_{i=1}^4 S_{E_{j,i}} T_M f.$$
(4-9)

To prove (3-8), it is sufficient to prove

$$\left\| \left(\sum_{j=0}^{\infty} |S_{E_{j,i}} T_M f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p)} \le C' \left(\frac{p^2}{p-1} \right)^2 \left\| \left(\sum_{j=0}^{\infty} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p)}$$
(4-10)

for $p \ge 2$, i = 1, 2, 3, 4. The arguments for i = 1, 2, 3, 4 are similar. We will give the argument for i = 1 only. By the fundamental theorem of calculus,

$$S_{E_{j,1}}T_M f = M_l(-2^{j-1}, 2^{j-1})S_{E_{j,1}}f + \sum_{n_1 \in J_j} \Delta_{n_1}M_l(n_1, 2^{j-1})S_{(n_1, 2^j) \times I_j}f + \sum_{n_2 \in E_j} \Delta_{n_2}M_l(-2^{j-1}, n_2)S_{J_j \times (n_2, 2^j)}f + \sum_{n=(n_1, n_2) \in E_{j,1}} \Delta_n M_l(n_1, n_2)S_{(n_1, 2^j) \times (n_2, 2^j)}f =: P_j^1 + P_j^2 + P_j^3 + P_j^4.$$
(4-11)

By the operator inequality $\left|\sum_{k=1}^{n} a_k\right|^2 \le n \sum_{k=1}^{n} |a_k|^2$, we have

$$|S_{E_{j,1}}T_M f|^2 = |P_j^1 + P_j^2 + P_j^3 + P_j^4|^2 \le 4(|P_j^1|^2 + |P_j^2|^2 + |P_j^3|^2 + |P_j^4|^2).$$
(4-12)

For part P_i^1 , by assumption (i) of Theorem 4.2, we have

$$|P_j^1|^2 = |M_l(-2^{j-1}, 2^{j-1}) S_{E_{j,1}}f|^2 \le C_1^2 |S_{E_{j,1}}f|^2 = C_1^2 |S_{E_{j,1}}S_jf|^2.$$
(4-13)

By Lemma 2.3,

$$\left\| \left(\sum_{j \ge 0} |P_j^1|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p(\mathbb{Z}^2))} \le C \left(\frac{p^2}{p-1} \right)^2 \left\| \left(\sum_{j \ge 0} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p(\mathbb{Z}^2))}.$$
(4-14)

For part P_j^2 , we follow the arguments similar to (3-16) and (3-17) in the one-dimensional case and write $\Delta_{n_1} M_l(n_1, 2^{j-1}) = a_n^* b_n$, with $|a_n|^2 = |b_n|^2 = |\Delta_{n_1} M_l(n_1, 2^{j-1})|$. Letting $R_{n_1,j} = (n_1, 2^j) \times I_j$, we have

$$|P_{j}^{2}|^{2} = \left| \sum_{n_{1} \in J_{j}} \Delta_{n_{1}} M_{l}(n_{1}, 2^{j-1}) S_{(n_{1}, 2^{j}) \times I_{j}} f \right|^{2} \le \left\| \sum_{n_{1} \in J_{j}} |\Delta_{n_{1}} M_{l}(n_{1}, 2^{j-1})| \right\|_{\infty} \left(\sum_{n_{1} \in J_{j}} |S_{R_{n_{1}, j}}(b_{n} S_{j} f)|^{2} \right).$$
(4-15)

Thus, by (4-15), (4-7) and Lemma 2.3 and following the arguments similar to the case d = 1, we get

$$\left\| \left(\sum_{j \ge 0} |P_j^2|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p(\mathbb{Z}^2))} \le C_2 \left(\frac{p^2}{p-1} \right)^2 \left\| \left(\sum_{j \ge 0} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p)}.$$
(4-16)

Similarly, we have

$$\left\| \left(\sum_{j \ge 0} |P_j^3|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p)} \le C_2 \left(\frac{p^2}{p-1} \right)^2 \left\| \left(\sum_{j \ge 0} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p)}.$$
(4-17)

Now we come to the estimate of part P_j^4 . Define $R_{n,j} = (n_1, 2^j) \times (n_2, 2^j)$. Similarly,

$$\begin{split} \left\| \left(\sum_{j \ge 0} |P_{j}^{4}|^{2} \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T}^{2}; S^{p})} &\leq C_{3}^{\frac{1}{2}} \left\| \left(\sum_{j \ge 0} \sum_{n = (n_{1}, n_{2}) \in E_{j, 1}} |S_{R_{n, j}} S_{j}| \Delta_{n} M_{l}(n)|^{\frac{1}{2}} f|^{2} \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T}^{2}; S^{p})} \\ &\leq C_{3}^{\frac{1}{2}} \left(\frac{p^{2}}{p-1} \right)^{2} \left\| \left(\sum_{j \ge 0} \sum_{n = (n_{1}, n_{2}) \in E_{j, 1}} |S_{j}| \Delta_{n} M_{l}(n)|^{\frac{1}{2}} f|^{2} \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T}^{2}; S^{p})} \\ &\leq C_{3} \left(\frac{p^{2}}{p-1} \right)^{2} \left\| \left(\sum_{j \ge 0} |S_{j}f|^{2} \right)^{\frac{1}{2}} \right\|_{L^{p}(\mathbb{T}^{2}; S^{p})}. \end{split}$$
(4-18)

Therefore, by (4-9), (4-11) and (4-15)–(4-18), we have

$$\left\| \left(\sum_{j=0}^{\infty} |S_{E_{j,1}} T_M f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p)} \le C' \left(\frac{p^2}{p-1} \right)^2 \left\| \left(\sum_{j=0}^{\infty} |S_j f|^2 \right)^{\frac{1}{2}} \right\|_{L^p(\mathbb{T}^2; S^p)}.$$
(4-19)

Thus, (4-10) is proved. Hence, we finish the proof of Theorem 4.2.

4.2. *Higher-dimensional case.* We need some additional notation to deal with the case d > 2. Borrowing the notation from [Hytönen et al. 2016], we denote by \mathbb{Z}^{α} the space

$$\mathbb{Z}^{\alpha} := \{ (n_i)_{i:\alpha_i=1} : n_i \in \mathbb{Z} \}$$

for $\alpha \in \{0, 1\}^d$. For any $n \in \mathbb{Z}^d$ and $E = I_1 \times \cdots \times I_d \subseteq \mathbb{Z}^d$, let

$$n_{\alpha} := (n_i)_{i:\alpha_i=1} \in \mathbb{Z}^{\alpha}, \quad E_{\alpha} := \prod_{i:\alpha_i=1} I_i \subseteq \mathbb{Z}^{\alpha}$$

be their natural projections onto \mathbb{Z}^{α} . In particular, we will use the splittings $n = (n_{\alpha}, n_{1-\alpha}) \in \mathbb{Z}^{\alpha} \times \mathbb{Z}^{1-\alpha}$ and $E = E_{\alpha} \times E_{1-\alpha}$, where $\mathbf{1} = (1, ..., 1)$. Suppose $s, t \in \mathbb{Z}^d$ and we abbreviate the interval notation $[s, t) \cap \mathbb{Z}^d$ as [s, t).

Similarly, denote by \mathbb{J}^d the partition $\mathbb{J}^d := \{E_j : j \ge 0\}$ of \mathbb{Z}^d , where

$$E_{j} = \begin{cases} \{(0, \dots, 0)\}, & j = 0, \\ \{(n_{1}, \dots, n_{d}) \in \mathbb{Z}^{d} : 2^{j-1} \le |(n_{1}, \dots, n_{d})|_{\infty} < 2^{j}\}, & j \ge 1. \end{cases}$$
(4-20)

Each E_j can be further decomposed into $2^d(2^d - 1)$ subsets and each of the subsets can be obtained by translation of the cube $F_j = [2^{j-1}, 2^j) \times \cdots \times [2^{j-1}, 2^j)$. Following similar procedures to those in the two-dimensional case and using the discrete fundamental theorem formula,

$$\chi_{[s,t)}(n)m(n) = \chi_{[s,t)} \sum_{\alpha \in \{0,1\}^d} \sum_{k_\alpha \in [s,n)_\alpha} \Delta^\alpha m(s_{1-\alpha}, k_\alpha)$$
$$= \sum_{\alpha \in \{0,1\}^d} \sum_{k_\alpha \in [s,t)_\alpha} \chi_{[k,t)_\alpha \times [s,t)_{1-\alpha}}(n) \Delta^\alpha m(s_{1-\alpha}, k_\alpha), \tag{4-21}$$

we can obtain the following theorem. The details are left to the interested reader.

Theorem 4.3. Given $m = (m_{s,t})_{s,t \in \mathbb{Z}^d} \in \mathcal{B}(\ell^2(\mathbb{Z}^d))$. Suppose *m* satisfies that, for some C > 0:

- (i) $\sup_{s,t\in\mathbb{Z}^d} |m_{s,t}| < C.$
- (ii) For any $n \in \mathbb{N}$, $s \in \mathbb{Z}^d$, $\alpha \in \{0, 1\}^d$, $\alpha \neq 0$, and any $r^{(n)} \in \mathbb{Z}^d$ satisfying $|r_i^{(n)}| = 2^{n-1}$ for all i = 1, ..., d,

$$\sum_{t=(t_{\alpha},r_{1-\alpha}^{(n)})\in E_n} |\Delta_t^{\alpha} m_{s,s+t}| < C, \qquad \sum_{t=(t_{\alpha},r_{1-\alpha}^{(n)})\in E_n} |\Delta_t^{\alpha} m_{s+t,s}| < C.$$
(4-22)

Then M_m extends to a bounded Schur multiplier on $S^p(\ell^2(\mathbb{Z}^d))$ for $p \in (1, \infty)$ with an upper bound $C_d(p^2/(p-1))^{d+2}$. Here C_d is a constant dependent only on the dimension d.
Note that we cannot hope for an analogue of Theorem 4.3 with E_n defined by the ℓ^2 -metric instead of the ℓ^{∞} metric because the ball-type Fourier multipliers are not uniformly bounded on $L^p(\mathbb{T}^d)$ for any d > 1. Doust and Gillespie [2005, Theorem 6.2] gave an example of ball-type Schur multipliers for the case d = 1. Their argument does not seem to extend to the case d > 1.

Example 4.4 (ball Schur multipliers). Let $X_0 = \{(0, 0)\} \subset \mathbb{Z}^d \times \mathbb{Z}^d$. For $i \in \mathbb{N}$, let

$$X_i = \{(k, j) \in \mathbb{Z}^d \times \mathbb{Z}^d : 2^{i-1} \le |(k, j)|_2 < 2^i\}.$$

Let $m_X = \sum_i \varepsilon_i \mathbb{1}_{X_i}$, with $|\varepsilon_i| \le 1$. Then $|\Delta_t^{\alpha} m| \le 2^{|\alpha|}$. Note that

$$|(k, j)|_2 \simeq |k|_2 + |j|_2 \simeq |k - j|_2 + |j|_2 \simeq |k - j|_{\infty} + |j|_{\infty} \simeq |k - j|_{\infty} + |k|_{\infty}.$$

We can find a constant K_d which only depends on d such that the set

$$\{(s_0, s_0 + t) : t \in E_n\} \cup \{(s_0 + t, s_0) : t \in E_n\}$$

intersects with at most K_d many X_i 's for any fixed s_0 . Since $\bigcup_{0 \le i \le n} X_i$ is convex for all n, we conclude that there are at most $2^d K_d$ many nonzero terms in the two summations in (4-22), and the summations are bounded by $2^d K_d$. So (4-22) is satisfied and $M_m = \sum_i \varepsilon_i P_{X_i}$ is bounded on S^p for any 1 .

4.3. *The case of continuous indices.* We explain in this section that Theorems 1.1 and 4.3 extend to the continuous case by approximation. Let $S^p(\mathbb{R}^d)$ be the space of Schatten *p*-class operators acting on the Hilbert space $L^2(\mathbb{R}^d)$. We identify $S^2(\mathbb{R}^d)$ as $L^2(\mathbb{R}^d \times \mathbb{R}^d)$, so for $A \in S^2(\mathbb{R}^d)$ we can talk about its pointwise value $a_{s,t}$. For $m \in L^{\infty}(\mathbb{R}^d \times \mathbb{R}^d)$, we consider the Schur-multiplier-type map

$$M_m(A) = (m(s, t)a_{s,t})_{s,t \in \mathbb{R}^d}$$

Motivated by the work of [Lafforgue and de la Salle 2011; Conde-Alonso et al. 2023], we wish to find sufficient conditions on *m* so that M_m extends to a bounded map with respect to the S^p -norm for 1 .

Theorem 4.5. For $p \in (1, \infty)$, consider the Schur multiplier M_m on $S^p(\mathbb{R})$ with symbol $m(\cdot, \cdot)$ in $L^{\infty}(\mathbb{R}^2)$ whose partial derivatives are continuous on $(-2^{j+1}, -2^j) \cup (2^j, 2^{j+1})$ for all $j \in \mathbb{Z}$. Suppose there exists an absolute constant C such that, for all $j \in \mathbb{Z}$ and $x, y \in \mathbb{R}$,

$$\int_{-2^{j+1}}^{-2^{j}} |\partial_1 m(y+t, y)| \, dt + \int_{2^{j}}^{2^{j+1}} |\partial_1 m(y+t, y)| \, dt \le C, \tag{4-23}$$

$$\int_{-2^{j+1}}^{-2^j} |\partial_2 m(x, x+t)| \, dt + \int_{2^j}^{2^{j+1}} |\partial_2 m(x, x+t)| \, dt \le C. \tag{4-24}$$

Then, the Schur multiplier M_m extends to a bounded map on $S^p(\mathbb{R})$ with $||M_m|| \le C \max\{p^3, 1/(p-1)^3\}$. *Proof.* Let \mathcal{D}_k be the σ -algebra generated by dyadic cubes

$$Q_{k,s,t} = \left(\frac{s}{2^k}, \frac{s+1}{2^k}\right] \times \left(\frac{t}{2^k}, \frac{t+1}{2^k}\right], \quad s, t \in \mathbb{Z}.$$

Then $(\mathcal{D}_k)_{k=1}^{\infty}$ is the usual dyadic filtration for \mathbb{R}^2 . Given $m \in L^{\infty}(\mathbb{R}^2)$, let $m_k = \mathbb{E}_k(m)$ be the conditional expectation of *m* with respect to the σ -algebra \mathcal{D}_k . That is to say

$$m_k(x) = \sum_{Q \in \mathcal{D}_k} \frac{1}{|Q|} \left[\int_Q m(y) \, dy \right] \chi_Q(x) \quad \text{for all } x \in \mathbb{R}^2.$$

Let $L^2(\mathbb{R}, \mathcal{D}_k)$ be the L^2 space of all \mathcal{D}_k -measurable functions. Let $\tilde{m}_k(s, t) = m(s/2^k, t/2^k)$ for $s, t \in \mathbb{Z}$. Note that $S^p(L^2(\mathbb{R}, \mathcal{D}_k))$ is isometrically isomorphic to $S^p(\ell^2(\mathbb{Z}))$. We see that $M_{\tilde{m}_k}$ extends to a bounded Schur multiplier on $S^p(\ell^2(\mathbb{Z}))$ with the same norm if M_{m_k} extends to a bounded Schur multiplier on $S^p(L^2(\mathbb{R}, \mathcal{D}_k))$ and vice versa. By Lemma 1.11 of [Lafforgue and de la Salle 2011],

$$\|M_m\| = \overline{\lim}_{k\to\infty} \|M_{m_k}\| = \overline{\lim}_{k\to\infty} \|M_{\widetilde{m}_k}\|.$$

So, we need to show that $M_{\tilde{m}_k}$ satisfies conditions (1-4), (1-5). First, we verify condition (1-5). For each $j \in \mathbb{N}$ and $s \in \mathbb{Z}$,

$$\begin{split} \sum_{\ell=0}^{2^{j}-2} |m_{k}(s,s+2^{j}+\ell+1) - m_{k}(s,s+2^{j}+\ell)| \\ &= 2^{2k} \sum_{\ell=0}^{2^{j}-2} \left| \int_{\frac{s}{2^{k}}}^{\frac{s+1}{2^{k}}} \int_{\frac{2^{j}+\ell+1}{2^{k}}}^{\frac{2^{j}+\ell+1}{2^{k}}} M\left(y,y+x+\frac{1}{2^{k}}\right) dx \, dy - \int_{\frac{s}{2^{k}}}^{\frac{s+1}{2^{k}}} \int_{\frac{2^{j}+\ell+1}{2^{k}}}^{\frac{2^{j}+\ell+1}{2^{k}}} M(y,y+x) \, dx \, dy \right| \\ &= 2^{2k} \sum_{\ell=0}^{2^{j}-2} \left| \int_{\frac{s}{2^{k}}}^{\frac{s+1}{2^{k}}} \int_{\frac{2^{j}+\ell}{2^{k}}}^{\frac{2^{j}+\ell+1}{2^{k}}} \int_{y+x}^{y+x+\frac{1}{2^{k}}} \partial_{2}M(y,t) \, dt \, dx \, dy \right| \\ &\leq 2^{2k} \sum_{\ell=0}^{2^{j}-2} \int_{\frac{s}{2^{k}}}^{\frac{s+1}{2^{k}}} \int_{\frac{2^{j+\ell}}{2^{k}}}^{\frac{2^{j}+\ell+1}{2^{k}}} \int_{y+x}^{y+x+\frac{1}{2^{k}}} |\partial_{2}M(y,t)| \, dt \, dx \, dy \\ &= 2^{2k} \int_{\frac{s}{2^{k}}}^{\frac{s+1}{2^{k}}} \int_{\frac{2^{j+1}-1}{2^{k}}}^{\frac{2^{j+1}+1}{2^{k}}} \int_{y+x}^{y+x+\frac{1}{2^{k}}} |\partial_{2}(M)|(y,t)| \, dt \, dx \, dy \\ &= 2^{2k} \int_{\frac{s}{2^{k}}}^{\frac{s+1}{2^{k}}} \int_{\frac{2^{j}}{2^{j}}}^{\frac{2^{j+1}-1}{2^{k}}} \int_{y+\frac{2^{j}}{2^{k}}}^{y+x+\frac{1}{2^{k}}} |\partial_{2}(M)|(y,t)| \, dt \, dx \, dy \\ &= 2^{2k} \int_{\frac{s}{2^{k}}}^{\frac{s+1}{2^{k}}} \int_{\frac{y+2^{j+1}-1}{2^{k}}}^{\frac{y+2^{j+1}-1}{2^{k}}} \int_{y+\frac{2^{j}}{2^{k}}}^{\frac{y+1+1}{2^{k}}} \chi_{(x,x+\frac{1}{2^{k}})}(t)|[\partial_{2}(M)](y,t)| \, dt \, dx \, dy \\ &= 2^{2k} \int_{\frac{s}{2^{k}}}^{\frac{s+1}{2^{k}}} \int_{y+\frac{2^{j}}{2^{k}}}^{\frac{y+2^{j+1}-1}{2^{k}}} \int_{\frac{2^{j}}{2^{k}}}^{\frac{y+1+1}{2^{k}}} \chi_{(t-\frac{1}{2^{k},t})}(x) \, dx|[\partial_{2}(M)](y,t)| \, dt \, dy \\ &\leq 2^{k} \int_{\frac{s}{2^{k}}}^{\frac{s+1}{2^{k}}} \int_{y+\frac{2^{j}}{2^{k}}}^{\frac{y+2^{j+1}-1}{2^{k}}} |\partial_{2}M(y,t)| \, dt \, dy \leq \sup_{y\in\mathbb{R}} \int_{\frac{2^{j}}{2^{k}}}^{\frac{2^{j+1}}{2^{k}}} |[\partial_{2}(M)](y,y+t)| \, dt \leq A. \end{split}$$
(4-25)

The last inequality follows from the assumption in (4-24). So, condition (1-5) is verified. Applying the same argument, we utilize the assumption in (4-23) to prove condition (1-4). Therefore, by Theorem 1.1, $||M_m|| = \overline{\lim}_{k \to \infty} ||M_{\widetilde{m}_k}|| \le C \max\{p^3, 1/(p-1)^3\}.$

Similarly, Theorem 4.3 and Example 4.4 have analogues in the continuous case as well.

Theorem 4.6. Define $E_j := \{t \in \mathbb{R}^d : 2^{j-1} \le |t|_{\infty} < 2^j\}$ for $j \in \mathbb{Z}$. For $p \in (1, \infty)$, consider the Schur multiplier $m \in L^{\infty}(\mathbb{R}^{2d})$ whose partial derivatives are continuous up to the boundary of E_k for all $k \in \mathbb{Z}$. Assume there exists a constant C such that

$$\int_{(t_{\alpha}, r_{1-\alpha}^{(j)}) \in E_j} |\partial^{\alpha} m(s, s+t)| dt_{\alpha} \le C,$$
(4-26)

$$\int_{(t_{\alpha}, r_{1-\alpha}^{(j)}) \in E_j} |\partial^{\alpha} m(s+t, s)| dt_{\alpha} \le C$$
(4-27)

for any $j \in \mathbb{Z}$, $s \in \mathbb{R}^d$ and any $r^{(j)} \in \mathbb{R}^d$ with $|r_i^{(j)}| = 2^{j-1}$ for all $1 \le i \le d$. Then, the Schur multiplier M_m extends to a bounded operator on $S^p(\mathbb{R}^d)$ for all $1 with <math>||M_m|| \le C_d \max\{p^{d+2}, 1/(p-1)^{d+2}\}$. Here t_{α} is defined as in Section 4.2.

Remark 4.7. The Schur multipliers in all theorems of this article are also completely bounded on S^p for 1 ; the arguments are exactly the same.

5. Discussions

5.1. Counterexamples.

(1) We show in the following that (1-4) alone is not sufficient for the boundedness of M_m .

Choose a large $K \in \mathbb{N}$. Let $m(s, t) = \exp(i2\pi kj/K)$ if $s = 2^k$, $t = 2^j$ for some $j, k \in \mathbb{N}$ satisfying $1 \le j < k \le K$, and m(s, t) = 0 for other $s, t \in \mathbb{N}$. Let $\tilde{m}(k, j) = \exp(i2\pi kj/K)$ if $1 \le j < k \le K$, and $\tilde{m}(k, j) = 0$ for other $k, j \in \mathbb{N}$. Let U be the partial isometry on $\ell_2(\mathbb{N})$ sending e_k to e_{2^k} . Then, we have $M_{\tilde{m}}(A) = U^*M_m(UAU^*)U$ for any $A \in S^p(\ell_2(\mathbb{N}))$ and $||M_{\tilde{m}}|| \le ||M_m||$.

Note that, for any N, j given, there exists at most one k (actually k = N) satisfying k > j and

$$2^{N-1} - 1 \le |2^k - 2^j| < 2^N.$$

Using the fact that $|m(s, t)| \le 1$, for any *N*, *t*, we get

$$\sum_{2^{N-1} \le |r| < 2^N} |m(t+r+1,t) - m(t+r,t)| \le 2,$$

because there are at most two nonzero terms in the sum above. This means $(m(s, t))_{s,t}$ satisfies the row condition (1-4). On the other hand, if $s = 2^N$, then |m(s, t) - m(s, t+1)| does not vanish if t or t + 1 has the form of 2^j , j = 1, ..., N - 1, by the definition of m(s, t). Hence we have

$$\sum_{2^{N-1} \le |t-s| < 2^N} |m(s, t+1) - m(s, t)| = \sum_{2^{N-1} \le s-t < 2^N} |m(s, t+1) - m(s, t)| = 2(N-1),$$

which shows that $(m(s, t))_{s,t}$ fails the column condition (1-5).

Let *A* be the $K \times K$ matrix $(\exp(-i2\pi kj/K))_{1 \le k,j \le K}$. Then *A* has S^p norm $K^{1/2+1/p}$. $M_{\tilde{m}}(A)$ is the lower triangular matrix with all nonzero coefficients being 1 which has S^p norm $\simeq K$ for any given *p*, $1 . This shows that <math>K^{1/2-1/p} \le ||M_{\tilde{m}}|| \le ||M_m||$. We then conclude that (1-4) alone is not sufficient for the boundedness of M_m . By symmetry, (1-5) alone is not sufficient for the boundedness of M_m either.

(2) A smooth version of the example above implies that neither the assumption (3-18) nor the assumption (3-19) is removable in Corollary 3.1. Indeed, fix a large K > 0, let

$$m_1(s,t) = \exp\left(\frac{i2\pi\log_2 s\log_2 t}{K}\right)$$

for $1 \le s \le t \le 2^K$, $s, t \in \mathbb{N}$,

$$m_1(s,t) = \frac{2^{K+1} - t}{2^K} \exp(i2\pi \log_2 s)$$

for $1 \le s \le t$, $2^K < t \le 2^{K+1}$, $s, t \in \mathbb{N}$ and $m_1(s, t) = 0$ otherwise. Then m_1 satisfies (3-18) because $\left|\frac{\partial}{\partial t} \exp\left(\frac{i2\pi \log_2 s \log_2 t}{K}\right)\right| \lesssim \frac{1}{t} \le \frac{1}{t-s}$

whenever $s < t \le 2^K$. Assuming the sufficiency of (3-18) would imply the uniform boundedness of M_{m_1} for all $1 , which is wrong because <math>M_m(A) = M_{m_1}(VAV)$ for $A \in S^p(\ell^2(\mathbb{N}))$ and m, V defined above. We conclude that neither the assumption (3-18) nor the assumption (3-19) is removable.

(3) Let

$$F_{N,t} = \{(s,t) \in \mathbb{N} \times \mathbb{N} : 2^{N-1} \le |s-t| < 2^N\}$$

for $N, t \in \mathbb{N}$. Let $Q_{N,t}$ be the projection from $S^2(\ell_2(\mathbb{N}))$ onto the span of $\{e_{s,t} : (s,t) \in F_{N,t}\}$. One may wonder whether Corollary 1.2 can be improved so that the Schur multiplier $S_{\varepsilon} = \sum_{N,t \in \mathbb{N}} \varepsilon(N,t) Q_{N,t}$ is bounded for any sequence $|\varepsilon(N,t)| \leq 1$. This is impossible as well.² To see this, let $\varepsilon(N,t) = \exp(i2\pi Nj/K)$ if $t = 2^j$ for some $j \in \mathbb{N}$ and $j < N \leq K$. Let $\varepsilon(N,t) = 0$ otherwise. Let V be the projection on $\ell^2(\mathbb{N})$ such that $V(e_i) = e_i$ if $i = 2^k$ for some $k \in \mathbb{N}$, and $V(e_i) = 0$ otherwise. Then, for M_m defined in the first example and $A \in S^p(\ell_2(\mathbb{N}))$, we have

$$M_m(A) = S_{\varepsilon}(VAV).$$

Therefore, $K^{1/2-1/p} \lesssim ||S_{\varepsilon}||$.

5.2. *Operator-valued symbol.* The Schatten *p* class has a natural operator space structure inherited from the operator space complex interpolation $S^p = (S^{\infty}, S^1)_{1/p}$, $1 . Pisier [1998, Lemma 1.7] proved that, with respect to <math>S^p$'s natural operator space structure, a map *M* on $S^p(\ell^2)$ is completely bounded if and only if $M \otimes id_{S^p(H)}$ is bounded on $S^p(S^p) = S^p(\ell^2 \otimes H)$ for any separable Hilbert space *H*. We will explain an operator-valued version of Theorem 1.1 which particularly implies the complete boundedness of the Schur multipliers considered in Theorem 1.1. We will assume the readers are familiar with the terminology of operator spaces in this subsection.

We will consider $A \in S^2(\ell^2(\mathbb{Z}) \otimes H)$ with H a separable Hilbert space. We present A in its matrix form $(a_{i,j})_{i,j \in \mathbb{Z}}$ with $a_{i,j} \in S^2(H)$. More precisely, denote by e_i the canonical basis of ℓ_2 , let $e_{j,i}$ be the rank-1 operator on ℓ^2 sending e_i to e_j . Denote by tr (resp. τ) the canonical trace on $B(\ell^2(\mathbb{Z}))$ (resp. B(H)). We set

$$a_{i,i} = (\operatorname{tr} \otimes \operatorname{id})(A(e_{i,i} \otimes \operatorname{id}_H)).$$

²We can get the same conclusion for sequences $\varepsilon_k = \pm 1$ by choosing Hadamard orthogonal matrices instead of the matrices $(\exp((-i2\pi kj)/K))_{1 \le k, j \le K}$.

Let \mathcal{M} be a finite von Neumann algebra with a normal faithful tracial state τ . Given an \mathcal{M} -valued bounded function m on $\mathbb{Z} \times \mathbb{Z}$ and $A \in S^2(\ell^2(\mathbb{Z}) \otimes H)$ in its matrix form $(a_{i,j})_{i,j \in \mathbb{Z}}$, we define $M_m(A)$ as the matrix

$$M_m(A) = (m_{i,j} \otimes a_{i,j})_{i,j}.$$
(5-1)

We will show that an analogue of Theorem 1.1 holds, that is, there exists $C_p \simeq (p^2/(p-1))^3$ for 1 such that

$$\|M_m(A)\|_{L^p(\mathcal{M}\otimes B(\ell^2(\mathbb{Z})\otimes H))} \le C_p \|A\|_{S^p(\ell^2\otimes H)}$$

for all $A \in S^2 \cap S^p$. By the density of $S^2 \cap S^p$, M_m extends to a bounded operator from $S^p(\ell^2(\mathbb{Z}) \otimes H)$ to $L^p(\mathcal{M} \otimes B(\ell^2(\mathbb{Z}) \otimes H))$ when *m* satisfies Marcinkiewicz-type conditions. When $\mathcal{M} = \mathbb{C}$, this implies the complete boundedness of M_m in Theorem 1.1 by Pisier's result. We will need Pisier's $L_{\infty}(\ell_1)$ norm to express this Marcinkiewicz-type condition.

Definition 5.1 (Pisier's $L^{\infty}(\ell_1)$ norm). Given *N*-tuples (x_1, \ldots, x_N) in \mathcal{M} , set

$$\|x\|_{L^{\infty}(\mathcal{M};\ell_{1})} = \inf\left\{ \left\| \left(\sum a_{j}a_{j}^{*}\right)^{\frac{1}{2}} \right\| \cdot \left\| \left(\sum b_{j}^{*}b_{j}\right)^{\frac{1}{2}} \right\| \right\},$$
(5-2)

where the infimum runs over all possible factorizations $x_j = a_j b_j$, with $a_j, b_j \in \mathcal{M}$.

When $x_k \ge 0$, we have $||x||_{L^{\infty}(\mathcal{M};\ell_1)} = ||\sum_k |x_k|||$ but the two quantities are not comparable in general. Pisier showed that $||x||_{L^{\infty}(\mathcal{M};\ell_1)} < \infty$ if and only if there is a decomposition $x_k = x_{k,1} - x_{k,2} + ix_{k,3} - ix_{k,4}$ such that $x_{k,\ell} \ge 0$ and $||(x_{k,\ell})_k||_{L^{\infty}(\mathcal{M};\ell_1)} < \infty$ for all $\ell = 1, 2, 3, 4$.

Given M_m defined as in (5-1), let

$$\Delta_{s}m(s,t) = m(s+1,t) - m(s,t), \quad \Delta_{t}m(s,t) = m(s,t+1) - m(s,t)$$

for $s, t \in \mathbb{Z}$.

Theorem 5.2. M_m defined as in (5-1) extends to a bounded map from Schatten *p*-classes $S^p(\ell^2 \otimes H)$ to $L^p(\mathcal{M} \otimes B(\ell^2(\mathbb{Z}) \otimes H))$ for all $1 with bounds <math>\leq (p^2/(p-1))^3$ if *m* is bounded in \mathcal{M} and there is a constant *C* such that,

(i) for any $n \in \mathbb{N}$, $t \in \mathbb{Z}$,

$$\|(\Delta_s m(s+t,t))_{2^{n-1} \le |s| < 2^n}\|_{L^{\infty}(\mathcal{M};\ell_1)} < C,$$
(5-3)

(ii) for any $n \in \mathbb{N}$, $s \in \mathbb{Z}$,

$$\|(\Delta_t m(s, s+t))_{2^{n-1} \le |t| < 2^n}\|_{L^{\infty}(\mathcal{M};\ell_1)} < C.$$
(5-4)

Sketch of proof. Define $\tilde{m}(s, t) = m(s, t) \otimes 1_{\ell_2 \otimes H}$ and $\tilde{a}_{s,t} = 1_{\mathcal{M}} \otimes a_{s,t}$. Then $\tilde{m}(s, t)$ commutes with $a_{s',t'}$ for any $s, t, s', t' \in \mathbb{Z}$. Let

$$\widetilde{M}_{l}(j) = \sum_{s \in \mathbb{Z}} \widetilde{m}(s, s - j) \otimes e_{s,s}, \quad \widetilde{M}_{r}(j) = \sum_{s \in \mathbb{Z}} \widetilde{m}(s + j, s) \otimes e_{s,s},$$

$$\widetilde{A}(j) = \sum_{s \in \mathbb{Z}} \widetilde{a}_{s,s-j} \otimes e_{s,s-j} = \sum_{s \in \mathbb{Z}} \widetilde{a}_{s+j,s} \otimes e_{s+j,s},$$
(5-5)

with $e_{s,t}$ the canonical basis of $S^2(\ell_2(\mathbb{Z}))$. Let $f(z) = \sum_{j \in \mathbb{Z}} \tilde{A}(j) z^j$ and

$$T_{\widetilde{M}}f(z) = \sum_{j \in \mathbb{Z}} \widetilde{M}_l(j)\widetilde{A}(j)z^j.$$
(5-6)

We still have

$$T_{\widetilde{M}}f(z) = \sum_{j \in \mathbb{Z}} \widetilde{A}(j)\widetilde{M}_r(j)z^j$$
(5-7)

and the identities

$$\|f\|_{L^{p}(\mathcal{M}\otimes B(\ell^{2}(\mathbb{Z})\otimes H))} = \|A\|_{S^{p}(\ell_{2}\otimes H)},$$

$$\|T_{\widetilde{M}}f\|_{L^{p}(\mathcal{M}\otimes B(\ell^{2}(\mathbb{Z})\otimes H))} = \|M_{m}(A)\|_{S^{p}(\ell_{2}\otimes H)}$$

Moreover, the conditions (5-3) and (5-4) imply that

$$\|\Delta_{l}\widetilde{M}(j)_{2^{n-1} < |j| \le 2^{n}}\|_{L^{\infty}(\mathcal{M} \otimes B(\ell^{2}(\mathbb{Z})), \ell_{1})}, \|\Delta_{r}\widetilde{M}(j)_{2^{n-1} < |j| \le 2^{n}}\|_{L^{\infty}(\mathcal{M} \otimes B(\ell^{2}(\mathbb{Z})), \ell_{1})} < C$$

for

$$\Delta_l \widetilde{M}(j) = \widetilde{M}_l(j+1) - \widetilde{M}_l(j), \quad \Delta_r \widetilde{M}(j) = \widetilde{M}_r(j+1) - \widetilde{M}_r(j).$$

After these, it is not hard to check that the arguments for the proof of Theorem 1.1 work as well for the tensor case. \Box

Corollary 5.3. The Schur multipliers considered in Theorem 1.1 are completely bounded on the Schutten classes S^p , $1 , with bounds <math>\leq (p^2/(p-1))^3$ with respect to their natural operator space structure.

Remark 5.4. The optimal constant for the L^p bounds of the classical Marcinkiewicz Fourier multipliers is $p^{3/2}$ as $p \to \infty$ [Tao and Wright 2001]. It is unclear what is the optimal asymptotic order for the S^p -bounds of the Schur multipliers in Theorem 1.1.

Open Question. Assume *m* is a bounded map on $\mathbb{Z} \times \mathbb{Z}$ such that

$$\sum_{s} |m(k, j_{s}) - m(k, j_{s+1})|^{2} < C$$

for all possible increasing sequences $j_s \in \mathbb{Z}$. Does M_m extend to a bounded map on S^p for all 1 ?

Remark 5.5. The authors heard this question from Potapov and Sukochev. They told the authors that it stems from the work of Birman and Solomyak on double operator integrals. The third author noticed Theorem 1.1 during his effort of attacking this question.

Acknowledgments

The authors are grateful to the referee for a very careful reading of the article. Liu and Mei are partially supported by NSF grants DMS 2247123.

References

- [Aleksandrov and Peller 2002] A. B. Aleksandrov and V. V. Peller, "Hankel and Toeplitz-Schur multipliers", *Math. Ann.* **324**:2 (2002), 277–327. MR Zbl
- [Arazy 1982] J. Arazy, "Certain Schur–Hadamard multipliers in the spaces C_p ", *Proc. Amer. Math. Soc.* **86**:1 (1982), 59–64. MR Zbl
- [Bennett 1977] G. Bennett, "Schur multipliers", Duke Math. J. 44:3 (1977), 603-639. MR Zbl
- [Berkson and Gillespie 1994] E. Berkson and T. A. Gillespie, "Spectral decompositions and harmonic analysis on UMD spaces", *Studia Math.* **112**:1 (1994), 13–49. MR Zbl
- [Bourgain 1986] J. Bourgain, "Vector-valued singular integrals and the *H*¹-BMO duality", pp. 1–19 in *Probability theory and harmonic analysis* (Cleveland, Ohio, 1983), edited by J.-A. Chao and W. A. Woyczyński, Monogr. Textbooks Pure Appl. Math. **98**, Dekker, New York, 1986. MR Zbl
- [Bożejko and Fendler 1984] M. Bożejko and G. Fendler, "Herz–Schur multipliers and completely bounded multipliers of the Fourier algebra of a locally compact group", *Boll. Un. Mat. Ital. A* (6) **3**:2 (1984), 297–302. MR Zbl
- [Caspers and de la Salle 2015] M. Caspers and M. de la Salle, "Schur and Fourier multipliers of an amenable group acting on non-commutative L^p-spaces", *Trans. Amer. Math. Soc.* **367**:10 (2015), 6997–7013. MR Zbl
- [Caspers et al. 2019] M. Caspers, D. Potapov, F. Sukochev, and D. Zanin, "Weak type commutator and Lipschitz estimates: resolution of the Nazarov–Peller conjecture", *Amer. J. Math.* **141**:3 (2019), 593–610. MR Zbl
- [Clément et al. 2000] P. Clément, B. de Pagter, F. A. Sukochev, and H. Witvliet, "Schauder decomposition and multiplier theorems", *Studia Math.* **138**:2 (2000), 135–163. MR Zbl
- [Conde-Alonso et al. 2023] J. M. Conde-Alonso, A. M. González-Pérez, J. Parcet, and E. Tablate, "Schur multipliers in Schatten–von Neumann classes", *Ann. of Math.* (2) **198**:3 (2023), 1229–1260. MR
- [Doust and Gillespie 2005] I. Doust and T. A. Gillespie, "Schur multiplier projections on the von Neumann–Schatten classes", *J. Operator Theory* **53**:2 (2005), 251–272. MR Zbl
- [Haagerup and Musat 2007] U. Haagerup and M. Musat, "On the best constants in noncommutative Khintchine-type inequalities", *J. Funct. Anal.* **250**:2 (2007), 588–624. MR Zbl
- [Haagerup et al. 2010] U. Haagerup, T. Steenstrup, and R. Szwarc, "Schur multipliers and spherical functions on homogeneous trees", *Internat. J. Math.* **21**:10 (2010), 1337–1382. MR Zbl
- [Harcharras 1999] A. Harcharras, "Fourier analysis, Schur multipliers on S^p and non-commutative $\Lambda(p)$ -sets", *Studia Math.* **137**:3 (1999), 203–260. MR Zbl
- [Hytönen et al. 2016] T. Hytönen, J. van Neerven, M. Veraar, and L. Weis, *Analysis in Banach spaces, Vol. 1: Martingales and Littlewood–Paley theory*, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) **63**, Springer, 2016. MR Zbl
- [de Laat and de la Salle 2018] T. de Laat and M. de la Salle, "Approximation properties for noncommutative L^p -spaces of high rank lattices and nonembeddability of expanders", *J. Reine Angew. Math.* **737** (2018), 49–69. MR Zbl
- [Lafforgue and de la Salle 2011] V. Lafforgue and M. de la Salle, "Noncommutative L^p -spaces without the completely bounded approximation property", *Duke Math. J.* 160:1 (2011), 71–116. MR Zbl
- [Lust-Piquard 1986] F. Lust-Piquard, "Inégalités de Khintchine dans C_p (1)", C. R. Acad. Sci. Paris Sér. I Math. 303:7 (1986), 289–292. MR Zbl
- [Lust-Piquard and Pisier 1991] F. Lust-Piquard and G. Pisier, "Noncommutative Khintchine and Paley inequalities", *Ark. Mat.* **29**:2 (1991), 241–260. MR Zbl
- [Mei 2007] T. Mei, *Operator valued Hardy spaces*, Mem. Amer. Math. Soc. **881**, Amer. Math. Soc., Providence, RI, 2007. MR Zbl
- [Mei et al. 2022] T. Mei, E. Ricard, and Q. Xu, "A Mikhlin multiplier theory for free groups and amalgamated free products of von Neumann algebras", *Adv. Math.* **403** (2022), art. id. 108394. MR Zbl
- [Neuwirth and Ricard 2011] S. Neuwirth and E. Ricard, "Transfer of Fourier multipliers into Schur multipliers and sumsets in a discrete group", *Canad. J. Math.* **63**:5 (2011), 1161–1187. MR Zbl

- [Parcet et al. 2022] J. Parcet, E. Ricard, and M. de la Salle, "Fourier multipliers in $SL_n(\mathbb{R})$ ", *Duke Math. J.* **171**:6 (2022), 1235–1297. MR Zbl
- [Pisier 1998] G. Pisier, *Non-commutative vector valued* L_p -spaces and completely *p*-summing maps, Astérisque **247**, Société Mathématique de France, Paris, 1998. MR Zbl
- [Pisier 2001] G. Pisier, *Similarity problems and completely bounded maps*, 2nd ed., Lecture Notes in Mathematics **1618**, Springer, 2001. MR Zbl
- [Pisier and Ricard 2017] G. Pisier and E. Ricard, "The non-commutative Khintchine inequalities for 0 ", J. Inst. Math. Jussieu 16:5 (2017), 1103–1123. MR Zbl
- [Potapov et al. 2015] D. Potapov, F. Sukochev, and A. Tomskova, "On the Arazy conjecture concerning Schur multipliers on Schatten ideals", *Adv. Math.* **268** (2015), 404–422. MR Zbl
- [Potapov et al. 2017] D. Potapov, A. Skripka, F. Sukochev, and A. Tomskova, "Multilinear Schur multipliers and Schatten properties of operator Taylor remainders", *Adv. Math.* **320** (2017), 1063–1098. MR Zbl
- [Ruzhansky and Turunen 2010] M. Ruzhansky and V. Turunen, *Pseudo-differential operators and symmetries*, Pseudo-Differential Operators. Theory and Applications 2, Birkhäuser, Basel, 2010. Background analysis and advanced topics. MR Zbl
- [Tao and Wright 2001] T. Tao and J. Wright, "Endpoint multiplier theorems of Marcinkiewicz type", *Rev. Mat. Iberoamericana* **17**:3 (2001), 521–558. MR Zbl

Received 9 Nov 2023. Revised 24 Apr 2024. Accepted 24 May 2024.

CHIAN YEONG CHUAH: chuah.21@osu.edu Department of Mathematics, The Ohio State University, Columbus, OH, United States

ZHEN-CHUAN LIU: liu.zhenchuan@uam.es Departamento de Matemáticas, Universidad Autónoma de Madrid, Madrid, Spain

TAO MEI: tao_mei@baylor.edu Department of Mathematics, Baylor University, Waco, TX, United States and

Department of Mathematics, Texas A&M University, College Station, TX, United States





DOUBLE DUALS AND HILBERT MODULES

HUAXIN LIN

Let A be a C^* -algebra, H be a Hilbert A-module and K(H) be the closure of the set of finite-rank module maps. We show that the W^* -algebra of all bounded A^{**} -module maps on the smallest self-dual Hilbert A^{**} -module containing H is isomorphic to $K(H)^{**}$ as W^* -algebras. We also show that the unit ball of H is closed in H^{\sharp} , the dual of H in an A-weak topology of H^{\sharp} , and the unit ball of H is also dense in the unit ball of H^{\sharp} in a weak* topology. Some versions of the Kaplansky density theorem for Hilbert C^* -modules are also presented.

1. Introduction

Hilbert C^* -modules as a generalization of Hilbert spaces were first introduced by I. Kaplansky [1953] in special cases and later by W. Paschke [1973] for general C^* -algebras. Hilbert C^* -modules are crucial to Kasparov's formulation of *KK*-theory [1980]. Early applications also include C^* -algebraic quantum group theory; see [Baaj and Skandalis 1993]. Later, in the study of Cuntz semigroups in connection with the classification of amenable C^* -algebras, Hilbert C^* -modules play an important role; see, for example, [Brown and Ciuperca 2009; Brown and Lin 2025; Coward et al. 2008; Ortega et al. 2011].

Let *A* be a C^* -algebra. Unlike Hilbert spaces, bounded module maps on a Hilbert *A*-module *H* may not have adjoints and the dual module H^{\sharp} , i.e., the Banach *A*-module of all bounded module maps from *H* to *A*, may not be identified as elements in *H*. Moreover, the C^* -algebra L(H) of all bounded module maps with adjoints may not be a W^* -algebra. If $H_0 \subset H$ is a Hilbert *A*-submodule, a bounded module map $\varphi : H_0 \to A$ may not be extended to a bounded module map from *H* to *A*. In general, one should not expect that *H* can be decomposed into an orthogonal direct sum of H_0 and its orthogonal complement. In fact, H_0 may not even have an orthogonal complement. Study of these phenomena may be found, for example, in [Lin 1991a; 1992] and more recently in [Brown and Lin 2025].

However, Paschke [1973] found that, if A is a W^* -algebra, then the dual module H^{\sharp} of a Hilbert A-module H can be made into a Hilbert A-module in a natural way which extends H, and H^{\sharp} is a self-dual Hilbert A-module. Even if A is not a W^* -algebra, one can extend H into an A^{**} -module $H \cdot A^{**}$ naturally. Then its dual $H^{\sim} := (H \cdot A^{**})^{\sharp}$ becomes a self-dual Hilbert A^{**} -module containing H. In fact, H^{\sim} is the smallest self-dual Hilbert A^{**} -module containing H as a Hilbert A-submodule; see Proposition 3.2. Paschke showed that the Banach algebra of all bounded module maps on H^{\sim} becomes a W^* -algebra.

MSC2020: primary 46L05, 46L08; secondary 46L35.

Keywords: Hilbert C^* -modules, double duals.

^{© 2025} MSP (Mathematical Sciences Publishers). Distributed under the Creative Commons Attribution License 4.0 (CC BY). Open Access made possible by subscribing institutions via Subscribe to Open.

HUAXIN LIN

For a Hilbert A-module H, the rank-1 module maps are the module maps T of the form $T(h) = x \langle y, h \rangle$ for all $h \in H$ (and fixed $x, y \in H$, where $\langle \cdot, \cdot \rangle$ is the A-valued inner product). Denote by F(H) the linear span of rank-1 module maps and denote by K(H) the norm closure of F(H). K(H) is a C^* -algebra and an important algebra related to the Hilbert module H. It was proved by Kasparov [1980, Theorem 1] that the C^* -algebra L(H) may be identified with M(K(H)), the multiplier algebra of K(H), and it was proved in [Lin 1991a] that the Banach algebra of all bounded module maps on H is identified with the left multipliers of K(H). (All Hilbert A-modules considered in this paper are right A-modules.) Over the decades, we eventually realized that it is rather convenient to work in $B(H^{\sim})$ in many occasions as we study module maps on a Hilbert module H. It is not difficult to establish a natural normal homomorphism $\Psi : K(H)^{**} \to B(H^{\sim})$ which extends beyond M(K(H)) and LM(K(H)). It remained unknown for many years whether Ψ is an isomorphism. The original motivation of this paper is to show that indeed Ψ is an isomorphism between W^* -algebras $K(H)^{**}$ and $B(H^{\sim})$.

As we study the relation among Hilbert modules H, $H \bullet A^{**}$ and H^{\sim} , naturally we ask: how dense is H in $H \bullet A^{**}$ and in H^{\sim} ? Since $H^{\sim} = (H \bullet A^{**})^{\sharp}$, the dual of $H \bullet A^{**}$, one may also ask about the density of H in H^{\sharp} in general.

We first note that it was shown (Theorem 6.1 of [Brown and Lin 2025]) that *H* is dense in H^{\sharp} in an *A*-weak topology. More precisely, for any $\xi \in H^{\sharp}$, there is a net $\{x_{\alpha}\}$ in *H* with $||x_{\alpha}|| \leq ||\xi||$ for all α such that $\lim_{\alpha} ||\xi(x) - \langle x_{\alpha}, x \rangle|| = 0$ for all $x \in H$. However, we show here that the unit ball of *H* is closed in H^{\sharp} in the topology where $x_{\alpha} \to \xi$ if and only if $\lim_{\alpha} ||\langle \xi - x_{\alpha}, \zeta \rangle|| = 0$ for all $\zeta \in H^{\sharp}$, and where the inner product is extended to H^{\sim} .

On the other hand, it is easy to see that, for any $\xi \in H \cdot A^{**}$, there is a net $\{x_{\lambda}\}$ in H such that $\lim_{\lambda} \pi_U(\langle x_{\lambda}, y \rangle)(v) = \pi_U(\langle \xi, y \rangle)(v)$ for all $y \in H \cdot A^{**}$ and $v \in H_U$, where H_U is the Hilbert space corresponding to the universal representation π_U of A. To be a more useful approximation, one may ask whether the net can be chosen to be bounded (by $||\xi||$). We will present a Kaplansky-style density theorem. Perhaps a more interesting question is: how dense is H in $H^{\sim} = (H \cdot A^{**})^{\sharp}$? Since H^{\sim} is the dual of $H \cdot A^{**}$, it is relatively easy to show that, for any $\zeta \in H^{\sim}$, there is a net $\{z_{\alpha}\}$ in H such that

$$\lim_{\lambda} f(\langle z_{\alpha}, y \rangle) = f(\langle \zeta, y \rangle) \quad \text{for all } y \in H \bullet A^{**} \text{ and } f \in A^{*}.$$

It is more challenging to show that y can be replaced by any element in $H^{\sim} = (H \bullet A^{**})^{\sharp}$. We show that the unit ball of H is actually dense in the unit ball of H^{\sim} in the weak* topology (as H^{\sim} is a conjugate space), another Kaplansky-style density theorem. In fact, we show a stronger density theorem that, for any $\xi \in H^{\sim}$, there is a net $\{x_{\alpha}\}$ in H with $||x_{\alpha}|| \leq ||\xi||$ such that

$$\lim_{\alpha} f(\langle \xi - x_{\alpha}, \xi - x_{\alpha} \rangle) = 0 \quad \text{for all } f \in A^*.$$

2. Self-duals

Definition 2.1. Let *A* be a *C*^{*}-algebra. Denote by \tilde{A} the minimum unitization of *A*. We use the following convention: if *A* is a *C*^{*}-subalgebra of a unital *C*^{*}-algebra *B*, we write $1_{\tilde{A}} = 1_B$ if either *A* is unital and $1_A = 1_{\tilde{A}} = 1_B$, or $A^{\perp} = \{b \in B : ba = ab = 0\} = \{0\}$, and we unitize *A* by adjoining 1_B to form $\tilde{A} \subset B$.

Definition 2.2. Let *X* be a Hilbert space and B(X) be the C^* -algebra of all bounded linear operators on *X*. Suppose that $A \subset B(X)$. Then $\overline{A}^{\text{SOT}}$ is the closure of *A* in the strong operator topology. Note that if $\{e_{\alpha}\}$ is an approximate identity for *A*, then $e_{\alpha} \nearrow 1_M$, i.e., e_{α} increasingly converges to the identity of $M = \overline{A}^{\text{SOT}}$ in the strong operator topology as well as in the weak* topology (of *M*). In particular, we may write $1_{\widetilde{A}} = 1_M$.

This works particularly for the pair A and A^{**} (where X is H_u , the Hilbert space corresponding to the universal representation of A).

In general, if M is a W^* -algebra, we denote by M_* the predual of M.

Definition 2.3. Let *A* be a *C*^{*}-algebra. In this paper, we use the formal definition of Hilbert modules in [Paschke 1973] and consider only right *A*-modules. Recall that a linear space *H* is a pre-Hilbert module if it is also a right *A*-module with an inner product $H \times H \rightarrow A$ satisfying the following properties: for any $x, y, z \in H$, $a \in A$ and $\lambda \in \mathbb{C}$,

- (1) $\langle x, \lambda y + z \rangle = \langle x, y \rangle + \lambda \langle x, y \rangle$,
- (2) $\langle x, ya \rangle = \langle x, y \rangle a$,
- (3) $\langle x, y \rangle^* = \langle y, x \rangle$,
- (4) $\langle x, x \rangle \ge 0$; if $\langle x, x \rangle = 0$, then x = 0.

Define $||x|| = ||\langle x, x \rangle||^{1/2}$ for $x \in H$. Then *H* becomes a normed space. *H* is a Hilbert *A*-module if *H* is complete with this norm.

Denote by H^{\sharp} the Banach space of all bounded module maps from H into A. A Hilbert A-module is said to be self-dual if, for every $f \in H^{\sharp}$, there is $x \in H$ such that

$$f(y) = \langle x, y \rangle$$
 for all $y \in H$.

Denote by B(H) the Banach algebra of all bounded module maps from H into itself, and by L(H) the C^* -algebra of all those bounded module maps T with an adjoint T^* in L(H) defined by

$$\langle T(x), y \rangle = \langle x, T^*(y) \rangle$$
 for all $x, y \in H$.

Let F(H) be the algebra of all finite-rank module maps, i.e., the linear span of all bounded module maps of the form $\theta_{x,y} : H \to H$ defined by

$$\theta_{x,y}(\xi) = x \langle y, \xi \rangle$$

for all $\xi \in H$ and $x, y \in H$. Denote by K(H) the norm closure of F(H), which is a C^{*}-algebra.

By Theorem 1 of [Kasparov 1980], we identify L(H) with M(K(H)), the multiplier algebra of K(H) and, by Theorem 1.5 of [Lin 1991a], B(H) with LM(K(H)), the Banach algebra of left multipliers of K(H) (in $K(H)^{**}$). If H is self-dual, then B(H) = L(H).

We refer to [Kasparov 1980; Lin 1991a; 1992; Paschke 1973] for common terminologies related to Hilbert C^* -modules.

Definition 2.4. Let A be a C^* -algebra and H a Hilbert A-module. Let us give the definition of a self-dual Hilbert A^{**} -module H^{\sim} ; see Definition 1.3 of [Lin 1991a].

We may view *H* as a Hilbert \tilde{A} -module. Let *B* be a unital C^* -algebra containing *A* and $1_{\tilde{A}} = 1_B$ (see the convention in Definition 2.1). The algebraical tensor product $H \otimes B$ becomes a right *B*-module if we set $(h \otimes a) \cdot b = h \otimes ab$ for any $h \in H$ and $a, b \in B$. Define $\langle -, - \rangle : H \otimes B \times H \otimes B \to B$ by

$$\left\langle \sum_{i} h_{i} \otimes a_{i}, \sum_{j} x_{j} \otimes b_{j} \right\rangle = \sum_{i,j} a_{i}^{*} \langle h_{i}, x_{j} \rangle b_{j}$$

and $N = \{z \in H \otimes A^{**} : \langle z, z \rangle = 0\}$. Then $(H \otimes B)/N$ becomes a pre-Hilbert *B*-module (see Section 4 of [Paschke 1973], but exchange *B* with *A*). Denote by $H \cdot B := ((H \otimes B)/N)^-$ (the completion of) the Hilbert *B*-module.

We are particularly interested in the case that $B = A^{**}$. We view \tilde{A} as a C^* -subalgebra of A^{**} . Then $H^{\sim} := (H \bullet A^{**})^{\sharp}$ is a self-dual Hilbert A^{**} -module.

Note that \tilde{A} is ultraweakly dense in A^{**} (since A is). By applying the result [Paschke 1973, Theorem 4.2] to the pair A^{**} (as A in that result) and \tilde{A} (as B in that result, see also the remark right after the proof of that result), we obtain an isometric (surjective) isomorphism $\iota : H^{\sim} := (H \bullet A^{**})^{\sharp} \to B(H, A^{**})$, with $B(H, A^{**})$ the Banach space of all bounded A-module maps from H to A^{**} (written as $M(H, A^{**})$ in that same result).

Let $x \in H$ and $b \in B$. Then

$$||(x \otimes b)/N||^2 = ||b^*\langle x, x\rangle b|| \le ||x||^2 ||b^*b||.$$

Hence

$$||(x \otimes b)/N|| \le ||x|| ||b||.$$

In what follows, for $x \in H$ and $b \in B$, we write $x \bullet b := (x \otimes b)/N$.

In general, if *E* is a self-dual Hilbert module, then B(E) = L(E); see [Paschke 1973, Corollary 3.5]. If in addition *A* is a *W*^{*}-algebra, *B*(*E*) is also a *W*^{*} -algebra; see [Paschke 1973, Proposition 3.11].

Let us recall the description of the predual of B(E) in this case. Denote by E_{\sim} the linear space E with the "twisted" scalar multiplication (i.e., $\lambda x = \overline{\lambda} x$ for $x \in E$ and $\lambda \in \mathbb{C}$) and consider $E_{\sim} \otimes E \otimes A_*$ with the greatest cross-norm, where A_* is the usual predual of the W^* -algebra A. For each $T \in B(E)$, define a linear functional \check{T} on $E_{\sim} \otimes E \otimes A_*$ by

$$\check{T}\left(\sum_{j=1}^{n} x_j \otimes y_j \otimes g_j\right) = \sum_{j=1}^{n} g_j(\langle T(x_j), y_j \rangle)$$

for $x_j, y_j \in E$ and $g_j \in A_*$, $1 \le j \le n$. The map $T \to \check{T}$ is a linear isometry of B(E) = L(E) into $(E_{\sim} \otimes E \otimes A_*)^*$. It was shown [Paschke 1973, Proposition 3.10] that B(E) is weak*-closed in $E_{\sim} \otimes E \otimes A_*$. A bounded net $\{T_{\alpha}\}$ in B(E) converges to $T \in B(E)$ in the weak* topology if and only if

$$f(\langle T_{\alpha}(x), y \rangle) \to f(\langle T(x), y \rangle)$$
 for all $x, y \in E$ and $f \in A_*$

[Paschke 1973, Remark 3.9 and Proposition 3.10]. In particular, $B(H^{\sim})$ is a W*-algebra.

Definition 2.5. Keep the notation in Definition 2.4. Recall that *H* is a Hilbert *A*-module and $\tilde{A} \subset B$. Then $\iota: H \to H \bullet B$ defined by $x \to x \otimes 1$ is an injective map. Note that, for all $a \in A$,

$$\langle (x \cdot a) \otimes 1 - x \otimes a, (x \cdot a) \otimes 1 - x \otimes a \rangle = \langle x \cdot a, x \cdot a \rangle - \langle x \cdot a, x \rangle a - a^* \langle x, x \cdot a \rangle + a^* \langle x, x \rangle a \\ = a^* \langle x, x \rangle a - a^* \langle x, x \rangle a - a^* \langle x, x \rangle a + a^* \langle x, x \rangle a = 0.$$

Hence $\iota(x \cdot a) = x \otimes a/N$ for all $a \in \tilde{A}$. In the case $B = A^{**}$, we then extend ι from H^{\sharp} to $(H \bullet A^{**})^{\sharp}$ by

$$\iota(f)(x \bullet b) = f(x)b$$
 for all $x \in H$ and $b \in A^{**}$

and $f \in H^{\sharp}$. Note that the map is a module map from H^{\sharp} to $(H^{\sim})^{\sharp}$, which is conjugate module isomorphic to H^{\sim} .

From now on, we may view H as a submodule of H^{\sim} and, sometimes, omit the map ι .

The following result provides a convenient and easy fact that $H \bullet B$ is the smallest Hilbert *B*-module containing *H* as a Hilbert *A*-module.

Proposition 2.6. Let A and B be a pair of C^* -algebras such that $A \subset B$, B is unital and $1_{\tilde{A}} = 1_B$. Suppose that H is a Hilbert A-module, H_1 is a Hilbert B-module and there is an embedding $\iota : H \to H_1$ as Hilbert modules, i.e., ι is a linear and A-module map such that

$$\langle \iota(x), \iota(y) \rangle = \langle x, y \rangle$$
 for all $x, y \in H$.

Then there is a unique B-module embedding $\tilde{\iota}: H \bullet B \to H_1$ such that

$$\tilde{\iota}(x \bullet b) = \iota(x)b$$
 for all $x \in H$ and $b \in B$, $\langle \tilde{\iota}(\xi), \tilde{\iota}(\zeta) \rangle = \langle \xi, \zeta \rangle$ for all $\xi, \zeta \in H \bullet B$.

Proof. For any $\xi = \sum_{i=1}^{n} x_i \bullet a_i$, where $x_i \in H$ and $a_i \in B$ $(1 \le i \le n)$, define

$$\tilde{\iota}(\xi) = \sum_{i=1}^n \iota(x_i) a_i.$$

Then, for $\zeta = \sum_{i=1}^{n} y_i \bullet b_i$,

$$\langle \tilde{\iota}(\xi), \tilde{\iota}(\zeta) \rangle = \sum_{i,j}^{n} a_{i}^{*} \langle x_{i}, y_{j} \rangle b_{j} = \langle \xi, \zeta \rangle.$$

In particular,

$$\|\tilde{\iota}(\xi),\tilde{\iota}(\xi)\rangle\| = \left\|\sum_{i,j}^n a_i^* \langle x_i, x_j \rangle b_j\right\| = \|\xi\|^2.$$

Therefore $\|\tilde{\iota}\| \le 1$ on $(H \otimes B)/N$. So $\tilde{\iota}$ is uniquely extended to a contractive linear map from $H \bullet B$ into H_1 . It is a *B*-module map. Since $(H \otimes B)/N$ is dense in $H \bullet B$,

$$\langle \tilde{\iota}(x), \tilde{\iota}(y) \rangle = \langle x, y \rangle$$
 for all $x, y \in H \bullet B$.

To see this embedding is unique, let $\tilde{\iota}_1$ be another such embedding. Then $(\tilde{\iota} - \tilde{\iota}_1)|_H = 0$. For any $\xi = \sum_{i=1}^n x_i \bullet a_i$, where $x_i \in H$ and $a_i \in B$,

$$(\tilde{\iota} - \tilde{\iota}_1)(\xi) = \sum_{i=1}^n (\iota(x_i) - \iota(x_i)) \bullet a_i = 0$$

In other words, $\tilde{\iota}_1 = \tilde{\iota}$.

Definition 2.7. Keep the notation in Definitions 2.3, 2.4 and 2.5. Recall that F(H) is the algebra of all finite-rank module maps. Define $\Psi_0 : F(H) \to F(H \bullet B) \subset B(H \bullet B)$ by

$$\Psi_0(\theta_{x,y})(\zeta) = \iota(x) \langle \iota(y), \zeta \rangle$$

for all $\zeta \in H \bullet B$, $x, y \in H$. Ψ is a *-preserving homomorphism from the *-algebra F(H) into $F(H \bullet B)$. Moreover, Ψ_0 is an isometry on F(H). In particular, $||\Psi_0|| = 1$. Therefore it extends uniquely to a C^* -algebra homomorphism from K(H) to $K(H \bullet B)$, which preserves the norm. It has to be an isometry as F(H) is dense in K(H).

In the case that $B = A^{**}$, we may define $\widetilde{\Psi}_0 : F(H) \to F(H^{\sim}) \subset B(H^{\sim})$ by

$$\Psi_0(\theta_{x,y})(\zeta) = \iota(x)\langle \iota(y), \zeta \rangle$$

for all $\zeta \in H^{\sim}$, $x, y \in H$. Then $\widetilde{\Psi}_0$ is a *-preserving homomorphism from the *-algebra F(H) into $F(H^{\sim})$ and it extends uniquely to a C^* -algebra homomorphism $\widetilde{\Psi}_0$ from K(H) to $K(H^{\sim})$, which preserves the norm. Recall that $\iota(H^{\sharp}) \subset H^{\sim}$.

Proposition 2.8. Let $A \subset B$ be a pair of C^* -algebras, where B is unital and $1_B = 1_{\tilde{A}}$. Let $T \in K(H)$. Then $\Psi_0(T)(x \bullet b) = T(x) \bullet b$ for all $x \in H$ and $b \in B$.

Proof. From the definition, for any $S \in F(H)$, any $x \in H$ and any $b \in B$,

$$\Psi_0(S)(x \otimes b) = S(x) \otimes b \pmod{\mathbb{N}}.$$

Fix $T \in K(H)$, and let $\epsilon > 0$. There exists $S \in F(H)$ such that

$$||T - S|| < \frac{1}{4}\epsilon(1 + ||x \bullet b|| + ||x|| ||b||).$$

Then

$$\|\Psi_0(T) - \Psi_0(S)\| < \frac{1}{4}\epsilon(1 + \|x \otimes b\| + \|x\|\|b\|) \text{ and } \|T(x) \cdot b - S(x) \cdot b\| \le \frac{1}{2}\epsilon.$$

Hence

 $\|\Psi_0(T)(x \bullet b) - T(x) \bullet b\| < \epsilon.$

Since this holds for all $\epsilon > 0$, we conclude that

$$\Psi_0(x \bullet b) = T(x) \bullet b.$$

Lemma 2.9. Let A and B be as in Proposition 2.8 and H be a Hilbert A-module. Suppose that $\{E_{\lambda}\}$ is an approximate identity for K(H). Then $\{\Psi_0(E_{\lambda})\}$ forms an approximate identity for $K(H \bullet B)$. Moreover

$$\lim \|\Psi_0(E_\lambda)(x) - x\| = 0 \quad \text{for all } x \in H \bullet B.$$

Proof. By Lemma 3.1 of [Brown and Lin 2025],

$$\lim_{\lambda \to 0} \|E_{\lambda}(x) - x\| = 0 \quad \text{for all } x \in H.$$
(2-1)

Let $S = \sum_{i=1}^{n} \theta_{x_i, y_i}$, where $x_i, y_i \in (H \otimes B)/N$, $1 \le i \le n$. Write $x_i = \sum_{j=1}^{k(i)} \xi_{j,i} \bullet b_{j,i}$, where $\xi_{j,i} \in H$ and $b_{j,i} \in B$, j = 1, 2, ..., k(i), i = 1, 2, ..., n. By Proposition 2.8,

$$\Psi_0(E_\lambda)(\xi_{j,i} \bullet b_{j,i}) = E_\lambda(\xi_{j,i}) \bullet b_{j,i}.$$

By (2-1),

$$\lim_{\lambda} \|\Psi_0(E_{\lambda})(\xi_{j,i} \bullet b_{j,i}) - (\xi_{j,i} \bullet b_{j,i})\| = 0$$
(2-2)

for j = 1, 2, ..., k(i), i = 1, 2, ..., n. It follows that

$$\lim_{k \to \infty} \|\Psi_0(E_{\lambda})(x_i) - x_i\| = 0, \quad i = 1, 2, \dots, n.$$

For any $z \in H \bullet B$,

$$\Psi_0(E_{\lambda})\theta_{x_i,y_i}(z) = (\Psi_0(E_{\lambda})x_i)\langle y_i, z \rangle = E_{\lambda}(x_i)\langle y_i, z \rangle.$$

It follows that, for $1 \le i \le n$,

$$\lim_{\lambda} \|\Psi_0(E_{\lambda})\theta_{x_i,y_i} - \theta_{x_i,y_i}\| = 0$$

Hence

$$\lim_{\lambda} \|\Psi_0(E_{\lambda})S - S\| = 0.$$

The set of those module maps with the form of S is norm-dense in $K(H \bullet B)$. Therefore we conclude that

 $\lim_{\lambda \to 0} \|\Psi_0(E_{\lambda})S - S\| = 0 \quad \text{for all } S \in K(H \bullet B).$

It follows that $\{\Psi_0(E_\lambda)\}$ forms an approximate identity for $K(H \bullet B)$.

2.10. Let *A* be a *C*^{*}-algebra and *H* be a Hilbert *A*-module. Then H^{\sharp} is a Banach *A*-module in general. Recall that, for each $T \in B(H)$, one may define a bounded conjugate module map $T^* : H \to H^{\sharp}$ as follows: for $x, y \in H$, define

$$T^*(x)(y) = \langle x, T(y) \rangle.$$

So, for a fixed x, we have that $T^*(x)$ gives an element in H^{\sharp} . Moreover, T^* is a bounded conjugate module map from H to H^{\sharp} with $||T^*|| = ||T||$. However, if we view H as a submodule of H^{\sharp} , then T^* is a bounded module map. Note that, if $T \in L(H)$, then $T^* \in L(H)$ and $T^*(H) \subset H$.

If *A* is a *W*^{*}-algebra, by Theorem 3.2 of [Paschke 1973], H^{\sharp} becomes a Hilbert *A* module in a natural way. For $T \in B(H)$ and $f \in H^{\sharp}$, define, for each $x \in H$,

$$T(f)(x) = \langle f, T^*(x) \rangle, \tag{2-3}$$

where T^* is defined above. Thus $\widetilde{T}(f)$ is a bounded linear module map from H to A with $\|\widetilde{T}(f)\| \le \|T\| \|f\|$. Hence we extend T to a bounded (conjugate) module map from H^{\sharp} to H^{\sharp} . As we view H^{\sharp} as a Hilbert A-submodule in this case, T is in fact a bounded module map on H^{\sharp} (we will take the conjugate as Hilbert space cases). By Corollary 3.7 of [Paschke 1973], such an extension is unique.

By Lemma 3.7 of [Lin 1992], one may ease the assumption that A is a W^* -algebra to the assumption that A is a monotone complete C^* -algebra.

Proposition 2.11. Let A and B be as in Proposition 2.8, H be a Hilbert A-module and $\{E_{\lambda}\}$ an approximate identity for K(H). Then

$$\lim_{\lambda \to 0} (\sup\{\|\tilde{\Psi}_0(E_{\lambda})(f)(x) - f(x)\| : f \in (H \bullet B)^{\sharp}, \|f\| \le 1\}) = 0 \quad \text{for all } x \in H \bullet B.$$
(2-4)

Moreover, suppose that $(H \bullet B)^{\sharp}$ extends $H \bullet B$ as a Hilbert B-module, then, for any $T \in B((H \bullet B)^{\sharp})$,

$$\lim_{\lambda} \|\langle \widetilde{\Psi}_0(E_{\lambda})T\widetilde{\Psi}_0(E_{\lambda})(x), y \rangle - \langle T(x), y \rangle \| = 0 \quad \text{for all } x, y \in H \bullet B.$$

Proof. Fix $f \in H^{\sharp}$. For any $x \in H \bullet B$, by Lemma 2.9,

$$\|\widetilde{\Psi}_0(E_{\lambda})(f)(x) - f(x)\| = \|f(E_{\lambda}(x)) - f(x)\| \le \|f\| \|E_{\lambda}(x) - x\| \to 0$$

Hence (2-4) holds.

To see the "moreover" part of the lemma, let $T \in B((H \bullet B)^{\sharp})$. Then, for any $x, y \in H \bullet B$,

$$\begin{aligned} \|\langle \widetilde{\Psi}_{0}(E_{\lambda})T\widetilde{\Psi}_{0}(E_{\lambda})(x), y\rangle - \langle T(x), y\rangle \| \\ & \leq \|\langle T\widetilde{\Psi}_{0}(E_{\lambda})(x), \widetilde{\Psi}_{0}(E_{\lambda})(y)\rangle - \langle T(x), \widetilde{\Psi}_{0}(E_{\lambda})(y)\rangle \| + \|\langle T(x), \widetilde{\Psi}_{0}(E_{\lambda})(y)\rangle - \langle T(x), y\rangle \| \\ & \leq \|y\| \|T\| \|\Psi_{0}(E_{\lambda})(x) - x\| + \|T\| \|x\| \|\Psi_{0}(E_{\lambda})(y) - y\| \end{aligned}$$

By applying Lemma 2.9 to the two terms of the last inequality above, we conclude that

$$\lim_{\lambda} \|\langle \widetilde{\Psi}_0(E_{\lambda})T\widetilde{\Psi}_0(E_{\lambda})(x), y \rangle - \langle T(x), y \rangle \| = 0 \quad \text{for all } x, y \in H \bullet B.$$

Definition 2.12. Let *A* be a C^* -algebra and *H* be a Hilbert *A*-module. Recall [Lin 1991a, Theorem 1.5] that we identify B(H) with LM(K(H)), the Banach algebra of left multipliers of K(H) (in $K(H)^{**}$).

By Lemma 2.9, Ψ_0 maps K(H) into $K(H \bullet B)$ which maps approximate identities to approximate identities. We may then extend a homomorphism $\Psi_0: B(H) = LM(K(H)) \to LM(K(H \bullet B)) = B(H \bullet B)$ by

$$\Psi_0(T) = \lim_{\lambda \to 0} \Psi_0(TE_{\lambda}),$$

where the convergence is in the left strict topology of $LM(K(H \bullet B))$. Since $\Psi_0|_{K(H)}$ is an isometry, so is Ψ_0 .

We are mostly interested in the case that $B = A^{**}$. By Theorem 3.2 of [Paschke 1973], $(H \bullet A^{**})^{\sharp}$ is a self-dual Hilbert A^{**} -module. Therefore, by Section 2.10, for each $T \in B(H)$, the extension $\widetilde{\Psi}_0(T)$ is unique. Hence Ψ_0 may be extended to a Banach algebra isomorphism $\widetilde{\Psi}_0$ from B(H) into $B(H^{\sim})$ such that

$$\Psi_0(T)|_{H \bullet A^{**}} = \Psi_0(T) \quad \text{for all } T \in B(H).$$
 (2-5)

We will visualize the map Ψ_0 a bit more.

Proposition 2.13. Let A and B be a pair of C^* -algebras as in Proposition 2.8 and H be a Hilbert A-module. Then, for any $T \in B(H)$,

$$\lim_{\lambda} \|\Psi_0(T)\Psi_0(E_{\lambda})(x) - \Psi_0(T)(x)\| = 0 \quad \text{for all } x \in H \bullet B.$$
(2-6)

Moreover

 $\Psi_0(T)(x \bullet b) = T(x) \bullet b$ for all $x \in H$ and $b \in B$.

Consequently, $\widetilde{\Psi}_0(\mathrm{id}_H) = \mathrm{id}_{H^{\sim}}$.

Proof. The identity (2-6) follows immediately from Lemma 2.9.

Since

 $\Psi_0(TE_{\lambda})(x \bullet b) = TE_{\lambda}(x) \bullet b,$

by (2-6) and by Lemma 3.1 of [Brown and Lin 2025],

$$\Psi_0(T)(x \bullet b) = T(x) \bullet b$$

for all $x \in H$ and $b \in B$.

- -

For the last part of the proposition, we note that, by considering the pair *A* and *A*^{**}, and by the "moreover" part of the proposition, $\Psi_0(\mathrm{id}_H) = \mathrm{id}_{H \bullet A^{**}}$. Therefore, since the extension $\widetilde{\Psi}_0(\mathrm{id}_{H \bullet A^{**}})$ is unique (Corollary 3.7 of [Paschke 1973], see Section 2.10 for convenience), we must have that $\widetilde{\Psi}_0(\mathrm{id}_H) = \mathrm{id}_{H^{\sim}}$.

The following is a slightly strengthened restatement of [Brown and Lin 2025, Proposition 2.3].

Proposition 2.14. Let A be a C*-algebra and H a Hilbert A-module. Then there is a homomorphism Ψ from $K(H)^{**}$ into $B(H^{\sim})$ such that $\Psi|_{B(H)} = \widetilde{\Psi}_0$. Moreover, if $T \in K(H)^{**}$ and $T_{\lambda} \in K(H)^{**}$ such that $T_{\lambda} \to T$ in the weak* topology, then

$$\lim f(\langle \Psi(T_{\lambda})(x), y \rangle) = f(\langle \Psi(T)(x), y \rangle) \text{ for all } x, y \in H^{\sim} \text{ and } f \in A^*.$$

Proof. By Definition 2.4, $B(H^{\sim}) = L(H^{\sim})$ is a W^* -algebra; see [Paschke 1973, Proposition 3.11]. Let $\pi : B(H^{\sim}) \to B(H_{\pi})$ be a faithful normal representation such that $\pi(B(H^{\sim}))$ is weakly closed in $B(H_{\pi})$. Then, by, for example, [Pedersen 1979, Theorem 3.7.7] and [Conway 2000, Corollary 46.5], there is a normal homomorphism $\Phi : K(H)^{**} \to B(H_{\pi})$ such that $\Phi|_{K(H)} = \pi \circ \widetilde{\Psi}_0|_{K(H)}$ and $\pi \circ \widetilde{\Psi}_0(K(H))$ is weakly dense in $\Phi(K(H)^{**})$. Since $\pi(B(H^{\sim}))$ is a von Neumann algebra, $\Phi(K(H)^{**}) \subset \pi(B(H^{\sim}))$. Since π is injective, we may define $\Psi = \pi^{-1} \circ \Phi$. Recall that π^{-1} is an isomorphism between W^* -algebras $\pi(B(H^{\sim}))$ and $B(H^{\sim})$. It follows that Ψ is weak*-continuous. Then, $\Psi|_{K(H)} = \pi^{-1} \circ \pi \circ \widetilde{\Psi}_0|_{K(H)} = \widetilde{\Psi}_0|_{K(H)}$.

Let $V = B(H^{\sim})_*$ be the predual (as Banach spaces). Then Ψ induces a map $\Psi^* : V \to K(H)^*$, the predual of $K(H)^{**}$, by $L(\Psi^*(v)) = \Psi(L)(v)$ for all $L \in (K(H)^*)^*$ and $v \in V$. Thus if $T_{\lambda} \in K(H)^{**}$ such that $T_{\lambda} \to T$ in the weak* topology in $K(H)^{**}$, then $\Psi(T_{\lambda})(v) = T_{\lambda}(\Psi^*(v))$ converges to $T(\Psi^*(v)) = \Psi(T)(v)$ for all $v \in V$. In other words, $\Psi(T_{\lambda}) \to \Psi(T)$ in the weak* topology in $V^* = B(H^{\sim})$. By Definition 2.4 (see Remark 3.9 and proof of Theorem 3.10 of [Paschke 1973]), this implies, in particular, for any $f \in A^*$, $x, y \in H^{\sim}$, that $f(\langle \Psi(T_{\lambda})(x), y \rangle) \to f(\langle \Psi(T)(x), y \rangle)$.

By Theorem 1.5 of [Lin 1991a], B(H) = LM(K(H)). Let $\{E_{\lambda}\}$ be an approximate identity for K(H). Then $TE_{\lambda} \in K(H)$ for all $T \in B(H)$. It follows from Proposition 2.13 that, for $T \in B(H)$,

$$\lim_{\lambda} \|\Psi(TE_{\lambda})(f)(x) - \Psi(T)(f)(x)\| = \lim_{\lambda} \|\Psi(T)\Psi(E_{\lambda})(f)(x) - \Psi(T)(f)(x)\| = 0$$

for all $x \in H \bullet A^{**}$ and $f \in (H \bullet A^{**})^{\sharp}$. On the other hand, by Lemma 2.9,

$$\lim_{\lambda} \|\Psi_0(TE_{\lambda})(x) - \Psi_0(T)(x)\| = 0 \text{ for all } x \in H \bullet A^{**}.$$

However, we have shown that $\Psi(TE_{\lambda})(y) = \widetilde{\Psi}_0(TE_{\lambda})(y) = \Psi_0(TE_{\lambda})(y)$ for all $y \in H \bullet A^{**}$ (see also Definition 2.12). Therefore, combining these three facts, for $x, y \in H \bullet A^{**}$, we obtain

$$\langle \Psi(T)(x), y \rangle = \langle \Psi_0(T)(x), y \rangle.$$

It follows that $\Psi(T)|_{H \bullet A^{**}} = \Psi_0(T)$. Since the extension of $\Psi_0(T)$ to a bounded module map on $(H \bullet A^{**})^{\sharp}$ is unique (see the end of Section 2.10 and [Lin 1992, Lemma 3.5]), we have $\Psi(T) = \widetilde{\Psi}_0(T)$ for all $T \in B(H)$. Hence

$$\Psi|_{B(H)} = \Psi_0.$$

Definition 2.15. Let *M* be a *W*^{*}-algebra and *H* be a Hilbert *M*-module. Then, H^{\sharp} is a self-dual Hilbert *M*-module by [Paschke 1973, Theorem 3.2]. Let $F_0 : F(H) \to F(H^{\sharp})$ be the homomorphism defined by

$$F_0(\theta_{x,y})(z) = x \langle y, z \rangle$$
 for all $z \in H^{\sharp}$ and $x, y \in H$

Clearly F_0 is an isometry. It extends uniquely to a homomorphism $F_0: K(H) \to K(H^{\sharp})$. We further extend $F: \widetilde{K(H)} \to \widetilde{K(H^{\sharp})}$ by $F(\mathrm{id}_H) = \mathrm{id}_{H^{\sharp}}$.

Proposition 2.16. Let M be a W^* -algebra and H a Hilbert M-module. Then there exists a unital normal homomorphism $F : K(H)^{**} \to B(H^{\sharp})$ such that $F|_{K(H)} = F_0$ and, if $T_{\lambda} \to T$ in the weak* topology of $K(H)^{**}$, then

$$\lim_{\lambda} f(\langle F(T_{\lambda})(x), y \rangle) = f(\langle F(T)(x), y \rangle)$$

for all $x, y \in H^{\sharp}$ and $f \in M_*$, the predual of M. Moreover, $F(T) = \widetilde{T}$ for all $T \in B(H)$ as defined by (2-3).

Proof. Recall that $B(H^{\sharp})$ is a W^* -algebra. We may assume that $B(H^{\sharp})$ acts on a Hilbert space X as a von Neumann algebra with $1_{B(H^{\sharp})} = \operatorname{id}_X$. Then, by [Lin 2001, Theorem 1.8.2] (see also [Pedersen 1979, Theorem 3.7.7]), there is a unital normal homomorphism $F : K(H)^{**} \to \overline{F_0(K(H))}^{\text{SOT}} \subset B(H^{\sharp})$ such that $F|_{K(H)} = F_0$. So F is weak*-continuous (see, for example, [Conway 2000, Corollary 46.5]).

Suppose that $T_{\lambda} \to T$ in the weak* topology of $K(H)^{**}$. Then $F(T_{\lambda}) \to F(T)$ in the weak* topology of $B(H^{\sharp})$. Therefore (see the later part of Definition 2.4, also, Remark 3.9 and the proof of Proposition 3.9 of [Paschke 1973]),

$$f(\langle F(T_{\lambda})(x), y \rangle) \to f(\langle F(T)(x), y \rangle)$$
 for all $x, y \in H^{\sharp}$ and $f \in M_{*}$.

Let $\{E_{\lambda}\}$ be an approximate identity for K(H). Then, for any $T \in B(H)$, by Lemma 3.1 of [Brown and Lin 2025],

$$\lim_{\lambda} \|F(T)F(E_{\lambda})(x) - F(T)(x)\| = \lim_{\lambda} \|F(T)E_{\lambda}(x) - F(T)(x)\| = 0 \quad \text{for all } x \in H.$$

On the other hand, since $F(T)F(E_{\lambda})|_{H} = F(TE_{\lambda})|_{H} = TE_{\lambda}$ and (by [Brown and Lin 2025, Lemma 3.1])

$$\lim_{\lambda} \|TE_{\lambda}(x) - T(x)\| = 0,$$

we conclude that

$$T(x) = F(T)(x)$$
 for all $x \in H$.

Since the extension of T to H^{\sharp} is unique (by Proposition 3.6 of [Paschke 1973], see also Lemma 3.5 of [Lin 1992]), $\tilde{T} = F(T)$.

3. Isomorphism of $B(H^{\sim})$ and $K(H)^{**}$

Let *A* be a monotone complete *C**-algebra and *H* be a Hilbert *A*-module. Then, by Lemma 3.7 of [Lin 1992], H^{\sharp} becomes a self-dual Hilbert *A*-module such that $\langle \tau, x \rangle = \tau(x)$ for all $x \in H$ and $\tau \in H^{\sharp}$. Note that, if *E* is self-dual, we conjugate map E^{\sharp} onto *E* just as in the case of Hilbert spaces.

We will apply the following lemma several times.

Proposition 3.1. Let A be a monotone complete C^* -algebra and $H_1 \subset H_2$ be Hilbert A-modules such that H_2 is self-dual. Then H_1^{\sharp} is an orthogonal summand of H_2^{\sharp} and the embedding $H_1^{\sharp} \to H_2^{\sharp}$ extends the embedding $H_1 \subset H_2$.

Proof. Define $P_0: H_2 \to H_1^{\sharp}$ by

$$P_0(y)(x) = \langle y, x \rangle \quad \text{for all } y \in H_2 \text{ and } x \in H_1.$$
(3-1)

It is a bounded module map (by viewing H_1^{\sharp} as a Hilbert module instead of the dual to avoid the conjugation) with $||P_0|| = 1$. Note that $P_0|_{H_1} = id_{H_1}$.

Let $\tau \in H_1^{\sharp}$. Since A is monotone complete, by Theorem 3.8 of [Lin 1992], there is $\tilde{\tau} \in H_2^{\sharp} = H_2$ such that $\tilde{\tau}|_{H_1} = \tau$ and $\|\tilde{\tau}\| = \|\tau\|$. This implies that P_0 is surjective.

Define $j: H_1^{\sharp} \to H_2^{\sharp} = H_2$ by

$$j(x)(y) = \langle x, P_0(y) \rangle$$
 for all $x \in H_1^{\sharp}$ and $y \in H_2$. (3-2)

Then j extends the embedding $H_1 \hookrightarrow H_2$. Now, for $x \in H_1^{\sharp}$ and $y \in H_2$, by (3-1) and (3-2),

$$P_0 \circ j(x)(y) = P_0(j(x))(y) = \langle j(x), y \rangle = \langle x, P_0(y) \rangle = \langle P_0(y), x \rangle^* = (P_0(y)(x))^* = \langle y, x \rangle^* = \langle x, y \rangle.$$

It follows that $P_0 \circ j = \operatorname{id}|_{H_1^{\sharp}}$, and thus $j : H_1^{\sharp} \to H_2$ is an embedding. With the identification of H_1^{\sharp} and $j(H_1^{\sharp})$, $P_0|_{H_1^{\sharp}} = \operatorname{id}|_{H_1^{\sharp}}$. It follows that P_0 is a projection and H_1^{\sharp} is an orthogonal summand of H_2 . \Box

Applying Propositions 3.1 and 2.6, we obtain the following characterization of H^{\sim} .

Proposition 3.2. Let A be a C^{*}-algebra and H be a Hilbert A-module. Then H^{\sim} is the smallest self-dual Hilbert A^{**}-module containing H as a Hilbert A-submodule.

Proof. Let H_1 be a self-dual Hilbert A^{**} -module containing H as a Hilbert A-submodule. Then, by Proposition 2.6,

$$H \subset H \bullet A^{**} \subset H_1.$$

Applying Proposition 3.1, since H_1 is self-dual,

$$H^{\sim} = (H \bullet A^{**})^{\sharp} \subset H_1^{\sharp} = H_1.$$

The proposition follows.

3.3. In the next proposition, let *A* be a *C**-algebra, and let $H_1 \subset H$ be Hilbert *A*-modules. Then, by Proposition 2.6, $H_1 \bullet A^{**} \subset H \bullet A^{**}$. Since A^{**} is monotone complete and $(H \bullet A^{**})^{\sharp} = H^{\sim}$ and $(H_1 \bullet A^{**})^{\sharp} = H_1^{\sim}$, by Proposition 3.1, we may write $H^{\sim} = H_1^{\sim} \oplus (H_1^{\sim})^{\perp}$. Denote by $P : H^{\sim} \to H_1^{\sim}$ the projection. Note that $P \in L(H^{\sim})$. By Lemma 3.2 of [Lin 1992], $K(H_1)$ is a hereditary *C**-subalgebra of K(H). Let $\Psi_H : K(H)^{**} \to B(H^{\sim})$ and $\Psi_1 : K(H_1)^{**} \to B(H_1^{\sim})$ be the homomorphisms given by Proposition 2.14, respectively.

Proposition 3.4. Using the notation above, we have that

$$\Psi_1 = \Psi_H|_{K(H_1)^{**}} = P\Psi_H P|_{K(H_1)^{**}};$$

in particular, $\Psi_1(T) = \Psi_H(T)|_{H_1^{\sim}} = P\Psi_H(T)P|_{H_1^{\sim}}$ for $T \in K(H_1)^{**}$. Moreover,

$$P\Psi_H(L)P|_{K(H_1)^{**}} \subset \Psi_1(K(H_1)^{**})$$
 for all $L \in K(H)^{**}$.

Furthermore, $\Psi(Q) = P$, where Q is the open projection in $K(H)^{**}$ corresponding to the hereditary C^* -subalgebra $K(H_1)$.

Proof. Denote by $\Psi_{K(H),0}$ the injective homomorphism from K(H) into $K(H \bullet A^{**})$ and by $\Psi_{K(H_1),0}$ the injective homomorphism from $K(H_1)$ into $K(H_1 \bullet A^{**})$ described in Definition 2.7, respectively.

Fix $S \in K(H_1)$. For each $x \in H_1$ and $b \in A^{**}$, by Proposition 2.8,

$$\Psi_{K(H),0}(S)(x \bullet b) = S(x) \bullet b,$$

$$P\Psi_{K(H),0}(S)P(x \bullet b) = P(S(x \bullet b)) = S(x) \bullet b = \Psi_{K(H_1),0}(S)(x \bullet b).$$

It follows that

$$\Psi_{K(H),0}(S)|_{H_1 \bullet A^{**}} = P\Psi_{K(H),0}(S)P|_{H_1 \bullet A^{**}} = \Psi_{K(H_1),0}(S)$$

Since the extensions of $\Psi_{K(H),0}(S)|_{H_1 \bullet A^{**}}$ and $\Psi_{K(H_1),0}(S)$ to bounded module maps on H_1^{\sim} are unique, and $\Psi(S)|_{H_1^{\sim}}$ and $\Psi(S)$ are corresponding extensions, by Corollary 3.7 of [Paschke 1973], we conclude that $\Psi(S)|_{H_1^{\sim}} = P\Psi_H(S)P|_{H_1^{\sim}} = \Psi_1(S)$.

Let $T \in K(H_1)^{**}$ and $\{T_{\lambda}\} \subset K(H_1)$ be a net such that $T_{\lambda} \to T$ in the weak* topology. By Proposition 2.14, for any $g \in A^*$,

$$\lim_{\lambda} |g(\langle \Psi_H(T_\lambda)(x), y \rangle) - g(\langle \Psi_H(T)(x), y \rangle)| = 0,$$
(3-3)

$$\lim_{\lambda} |g(\langle \Psi_1(T_{\lambda})(x), y \rangle) - g(\langle \Psi_1(T)(x), y \rangle)| = 0$$
(3-4)

for all $x, y \in H_1^{\sim}$. Since we have shown that $\Psi_H(T_{\lambda})|_{H_1^{\sim}} = P\Psi_H(T_{\lambda})P|_{H_1^{\sim}} = \Psi_1(T_{\lambda})$, we conclude that

$$\Psi_H(T)|_{H_1^{\sim}} = P\Psi_H(T)P|_{H_1^{\sim}} = \Psi_1(T).$$
(3-5)

Hence

$$\Psi_1 = P \Psi_H P|_{K(H_1)^{**}} = \Psi_H|_{K(H_1)^{**}}.$$

Let $\{q_{\lambda}\}$ be an approximate identity for $K(H_1)$. Then $q_{\lambda} \nearrow \operatorname{id}_{H_1} \in K(H_1)^{**}$. It follows from Proposition 2.14 that

$$\lim_{\lambda} f(\langle \Psi_1(q_{\lambda}(y)), z \rangle) = f(\langle y, z \rangle) \quad \text{for all } y, z \in H_1^{\sim} \text{ and } f \in A^*.$$

On the other hand, we also have that $q_{\lambda} \nearrow Q$ in $K(H)^{**}$. By Proposition 3.1, $H^{\sim} = H_1^{\sim} \oplus (H_1^{\sim})^{\perp}$. Note that $q_{\lambda}(x) \in H_1$ for all $x \in H$. Then, for $x \in H$, $b \in A^{**}$ and $g \in (H_1^{\sim})^{\perp}$, by Proposition 2.8,

$$\langle \Psi_H(q_\lambda)(x \bullet b), g \rangle = g(q_\lambda(x \bullet b))^* = g(q_\lambda(x) \bullet b)^* = 0.$$

It follows that, for any $y \in H \bullet A^{**}$ and $g \in (H_1^{\sim})^{\perp}$,

$$\langle \Psi_H(q_\lambda)(y), g \rangle = 0.$$

Hence, for $g \in (H_1^{\sim})^{\perp}$,

 $\langle y, \Psi_H(q_\lambda)(g) \rangle = 0$ for all $y \in H \bullet A^{**}$.

It follows that $\Psi_H(q_\lambda)(g) = 0$ and

$$\langle \Psi_H(q_\lambda)(z), g \rangle = \langle z, \Psi_H(q_\lambda)(g) \rangle = 0 \text{ for all } z \in H^{\sim}.$$

In other words, $\Psi_H(q_\lambda)(z) \in H_1^{\sim}$ for all $z \in H$ and λ . Therefore

$$P\Psi_H(q_{\lambda}) = \Psi_H(q_{\lambda}) = \Psi_H(q_{\lambda})P.$$

Note that $Pz \in H_1^{\sim}$ for any $z \in H^{\sim}$. Thus, by (3-5) and (3-4),

$$\lim_{\lambda} f(\langle \Psi(q_{\lambda})(y), z \rangle) = \lim_{\lambda} f(\langle \Psi(q_{\lambda})(P(y)), P(z) \rangle) = \lim_{\lambda} f(\langle \Psi_{1}(q_{\lambda})(P(y)), P(z) \rangle)$$
$$= f(\langle P(y), P(z) \rangle) = f(\langle P(y), z \rangle).$$

By (3-3) and (3-5), $\lim_{\lambda} f(\langle \Psi(q_{\lambda})(y), z \rangle) = f(\langle \Psi(Q)(y), z \rangle)$. Therefore

$$\Psi(Q) = P$$

This proves the "furthermore" part. In what follows we will identify Q with P as well as $\Psi(Q)$ and $\Psi(P)$.

Now let $L \in K(H)^{**}$ and $\{L_{\lambda}\} \subset K(H)$ be a net such that $L_{\lambda} \to L$ in the weak* topology. By Proposition 2.14, for any $g \in A^*$, $x, y \in H_1^{\sim}$,

$$\lim_{\lambda} |g(\langle \Psi_H(T_{\lambda})(x), y \rangle) - g(\langle \Psi_H(T)(x), y \rangle)| = 0,$$

$$\lim_{\lambda} |g(\langle \Psi_1(T_{\lambda})(x), y \rangle) - g(\langle \Psi_1(T)(x), y \rangle)| = 0$$

(note that $\Psi_1(T_{\lambda}) = P \Psi_1(T_{\lambda}) P$). We also have, for any $x, y \in H_1^{\sim}$,

$$\langle \Psi_H(PT_\lambda P)(x), y \rangle = \langle \Psi_H(T_\lambda)(x), y \rangle, \langle P\Psi_H(T)P(x), y \rangle = \langle \Psi_H(T)(x), y \rangle.$$

Since $PT_{\lambda}P \in K(H_1)^{**}$, by the first part of the lemma, $\Psi_H(PT_{\lambda}P)(x) = \Psi_1(PT_{\lambda}P)(x)$ for $x \in H_1^{\sim}$. It follows that $P\Psi_H(T)P(x) = \Psi_1(PTP)(x)$ for all $x \in H_1^{\sim}$. Then

$$P\Psi_H(T)P = \Psi_1(PTP) \in \Psi_1(K(H_1)^{**}).$$

3.5. Let *A* be a C^* -algebra and let, for $n \in \mathbb{N}$,

$$H_n = A^{(n)} = \{(a_1, a_2, \dots, a_n) : a_j \in A, 1 \le j \le n\},\$$

the direct sum of *n* copies of *A*, where $\langle a, b \rangle = \sum_{j=1}^{n} a_j^* b_j$ if $a = (a_1, a_2, \dots, a_n)$ and $b = (b_1, b_2, \dots, b_n)$. Let

$$H_A = \left\{ \{a_n\} : a_n \in A \text{ and } \sum_{i=1}^n a_k^* a_k \text{ converges in norm} \right\}$$

be the standard countably generated Hilbert (right) A-module. Note that

$$\langle \{a_n\}, \{b_n\} \rangle = \sum_{n=1}^{\infty} a_n^* b_n$$

We note that H_A is the closure of $\bigcup_n A^{(n)}$. We may also view $H_n = A^{(n)}$ as an orthogonal summand of H_A . Then

$$H_A^{\sharp} = \left\{ \{a_n\} : \left\{ \left\| \sum_{k=1}^n a_k^* a_k \right\| \right\} \text{ is bounded} \right\}.$$

If $g = \{a_n\} \in H^{\sharp}$, then

$$g(x) = \sum_{n=1}^{\infty} a_n^* b_n \quad \text{for all } x = \{b_n\} \in H_A,$$

where the sum converges in norm. Moreover $||g|| = \lim_{n \to \infty} \left\| \sum_{k=1}^{n} a_k^* a_k \right\|$.

If A is a W*-algebra, as mentioned earlier, H_A^{\sharp} becomes a Hilbert A-module in a natural way (see Theorem 3.2 of [Paschke 1973]). In fact, we may define

$$\langle x, y \rangle = \sum_{n=1}^{\infty} a_n^* b_n \text{ for all } x = \{a_n\}, y = \{b_n\} \in H_A^{\sharp}.$$
 (3-6)

To see the right side converges in the weak* topology, we first let $f \in A^*$. Note that, if $\{a_n\} \in H_A^{\sharp}$,

$$\left|\sum_{k=1}^{N} f(a_{k}^{*}a_{k})\right| = \left|f\left(\sum_{k=1}^{N} a_{k}^{*}a_{k}\right)\right| \le \|f\| \left\|\sum_{k=1}^{N} a_{k}^{*}a_{k}\right\|$$

for any integer N. Hence $\{\sum_{k=1}^{n} f(a_k^*a_k)\}$ is bounded, is increasing and converges for any positive linear functional f. Hence, for any m > n,

$$\sum_{k=n}^{m} f(a_k^* a_k) \to 0 \quad \text{as } n \to \infty \quad \text{for all } f \in A^*.$$
(3-7)

For any positive linear functional f of A and for any m > n in \mathbb{N} ,

$$\begin{split} \left| f\left(\sum_{k=n}^{m} a_{k}^{*} b_{k}\right) \right| &= \left| \sum_{k=n}^{m} f(a_{k}^{*} b_{k}) \right| \leq \sum_{k=n}^{m} |f(a_{k}^{*} b_{k})| \\ &\leq \sum_{k=n}^{m} |f(a_{k}^{*} a_{k})|^{1/2} |f(b_{k}^{*} b_{k})|^{1/2} \\ &\leq \left(\left(\sum_{k=n}^{m} |f(a_{k}^{*} a_{k})| \right) \left(\sum_{k=n}^{m} |f(b_{k}^{*} b_{k})| \right) \right)^{1/2} \\ &\leq \|f\|^{1/2} \|\{b_{k}\}\| \left(\sum_{k=n}^{m} |f(a_{k}^{*} a_{k})| \right)^{1/2} \to 0 \quad \text{as } n \to \infty. \end{split}$$

It follows that $f(\sum_{k=1}^{n} a_k^* b_k)$ converges for all $f \in A^*$ as $n \to \infty$. Let us write the limit as $f(\sum_{k=1}^{\infty} a_k^* b_k)$. Then, by the above inequalities (with n = 1), we also have

$$\left| f\left(\sum_{k=1}^{\infty} a_k^* b_k\right) \right| \le \|f\| M_b M_a,$$

where

$$M_a = \sup \left\{ \left\| \sum_{k=1}^n a_k^* a_k \right\| \right\}^{1/2}$$
 and $M_b = \sup \left\{ \left\| \sum_{k=1}^n b_k^* b_k \right\| \right\}^{1/2}$.

Thus $\sum_{k=1}^{\infty} a_k^* b_k$ defines a bounded linear functional on A^* . Its restriction on A_* gives an element in A (recall that A is assumed to be a W^* -algebra). This shows the infinite series in the right side of (3-6) converges in the weak* topology. It is then standard to verify that (3-6) defines an inner product which extends the inner product on H_A .

Let *A* act on a Hilbert space *X* (as a *W**-algebra). Consider $l^2(X)$, the Hilbert space direct sum of countably many copies of *X*. Suppose that $b = \{b_n\} \in H_A^{\sharp}$. Then the infinite matrix $\bar{b} = (b_{i,j})$, with $b_{i,1} = b_i$, $i \in \mathbb{N}$ and $b_{i,j} = 0$ if $j \ge 2$, defines a bounded linear operator on $l^2(X)$, by $\bar{b}(v) = (b_1(v_1), b_2(v_1), \ldots, b_n(v_1), \ldots)$, where $v = (v_1, v_2, \ldots, v_n, \ldots) \in l^2(X)$. Moreover

$$\|\bar{b}\|^{2} = \|\bar{b}^{*}\bar{b}\| = \left\|\sum_{i=1}^{\infty} b_{i}^{*}b_{i}\right\| = \sup\left\{\left\|\sum_{i=1}^{n} b_{i}^{*}b_{i}\right\| : n \in \mathbb{N}\right\}$$
(3-8)

(some of these details in this subsection may be found in [Lin 1991b]).

Proposition 3.6. Let C be a unital C*-algebra and $A \subset C$ be a C*-subalgebra such that $1_{\tilde{A}} = 1_C$. Denote by $R = \overline{AC}$ the closed right ideal of C generated by A. Then:

- (1) $H_A \bullet C = \{\{b_n\} \in H_C : b_n \in R\}.$
- (2) If C is a W*-algebra and $e_{\alpha} \nearrow 1_{C}$, where $\{e_{\alpha}\}$ is an approximate identity for A, then

$$(H_A \bullet C)^{\sharp} = H_C^{\sharp}.$$

Proof. To see (1), we first note that $A \bullet C = R$ as Hilbert *C*-modules. Hence $A^{(n)} \bullet C = R^{(n)}$. Clearly, $H_A \bullet C \subset H_C$. We note that $\{\{r_n\} \in H_C : r_n \in R\}$ is closed in H_C . Since both $\bigcup_n A^{(n)} \bullet C$ and $\bigcup_{n=1}^{\infty} R^{(n)}$ are dense in $\{\{r_n\} \in H_C : r_n \in R\}$, and $\bigcup_n A^{(n)} \bullet C$ is dense $H_A \bullet C$, we obtain

$$\{\{r_n\}\in H_C:r_n\in R\}=H_A\bullet C.$$

This proves (1).

For (2) we may assume that $A \subset C \subset B(X)$, where X is a Hilbert space, $1_C = id_X$, and the range C(X) equals X. Otherwise, we replace X by $1_C(X)$.

<u>Claim 1</u>: $\overline{R(X)} = C(X) = X$. Since $e_{\alpha} \nearrow 1_C = id_X$, for any $v \in X$, $e_{\alpha}(v) \to v$. This proves the claim.

<u>Claim 2</u>: $R^{\sharp} = C$, where R^{\sharp} is the dual of the Hilbert *C*-module *R* (as we assume that *C* is a *W*^{*}-algebra).

Let $f \in R^{\sharp}$. Then $f(e_{\alpha})r = f(e_{\alpha}r) \to f(r)$ for all $r \in R$ in norm as $e_{\alpha}r \to r$ in norm. Hence $f(e_{\alpha})r(v) \to f(r)(v)$ for all $r \in R$ and $v \in X$. Define T on R(X) by $T(r(v)) = \lim_{\alpha} f(e_{\alpha})r(v)$ for all $v \in X$ and $r \in R$. One checks that T is a well-defined linear map on R(X). Moreover, we have

 $||T|| \le \sup\{||f(e_{\alpha})|| : \alpha\} \le ||f||$. Since, by Claim 1, $\overline{R(X)} = X$, we have that *T* extends uniquely to a bounded linear operator (denote by *T* again) on *X*. Moreover, $f(e_{\alpha})$ converges to *T* on *X*. Since *C* is closed in the weak operator topology, $T \in C$. Moreover, Tr(v) = f(r)(v) for all $v \in X$. It follows that Tr = f(r) for all $r \in R$.

For each $c \in C$, define $f_c \in R^{\sharp}$ by

$$f_c(r) = c^* r$$
 for all $r \in R$.

For the above *T*, we note that $f_{T^*}(r) = Tr$ for all $r \in R$. Hence the map $c \to f_c$ is surjective. To see it is injective, suppose that $c^*r = 0$ for all $r \in R$. Then

$$c^* e_{\alpha} c = 0$$
 for all α .

Since $c^*e_{\alpha}c \nearrow c^*c$, this implies that $c^*c = 0$. Thus the map $c \mapsto f_c$ is injective, which extends the identity map on R. It follows that $R^{\sharp} = C$, and Claim 2 is proved.

By Claim 2, we obtain that $((A^{(n)}) \bullet C)^{\sharp} = C^{(n)}$. By (1), $(A^{(n)}) \bullet C$ is a direct summand of $H_A \bullet C$. Hence we may write $((A^{(n)}) \bullet C)^{\sharp} \subset (H_A \bullet C)^{\sharp}$. Together with (1), we obtain that

$$H_A \bullet C \subset H_C \subset (H_A \bullet C)^{\sharp}.$$

Note H_C is a Hilbert *C*-submodule of the self-dual Hilbert *C* module $(H_A \bullet C)^{\sharp}$. It follows from Proposition 3.1 that

$$(H_A \bullet C)^{\sharp} \subset H_C^{\sharp} \subset (H_A \bullet C)^{\sharp}.$$

Consequently, $H_C^{\sharp} = (H_A \bullet C)^{\sharp}$.

3.7. Note that, if A is unital, $H_A \bullet C = H_C$.

From the above discussion, we obtain the following result.

Lemma 3.8. Let A be a C^* -algebra, $H_n = (A^{**})^{(n)}$ and $P_n : H_{A^{**}}^{\sharp} \to H_n$ be the projection. (1) Let $S \subset H_{A^{**}}^{\sharp}$ be a bounded subset. Then, for any $f \in A^*$ and $x \in H_{A^{**}}^{\sharp}$,

$$\lim_{n \to \infty} \sup\{|f(\langle P_n(x), y \rangle) - f(\langle x, y \rangle)| : y \in S\} = 0,$$
$$\lim_{n \to \infty} \sup\{|f(\langle y, P_n(x) \rangle) - f(\langle y, x \rangle)| : y \in S\} = 0.$$

(2) Moreover,

$$\lim_{n \to \infty} |f(\langle P_n(x), P_n(x) \rangle) - f(\langle x, x \rangle)| = 0 \quad \text{for all } x \in H_{A^{**}}^{\sharp} \text{ and } f \in A^*.$$

Proof. Set $M = \sup\{||y|| : y \in S\} + 1$. Let f be a positive linear functional in A^* and $x = \{a_n\} \in H_{A^{**}}^{\sharp}$. For each $y = \{b_n\} \in S$,

$$|f(\langle P_n(x), y \rangle) - f(\langle x, y \rangle)| = \left| \sum_{k=n+1}^{\infty} f(a_k^* b_k) \right| \le \left(\sum_{k=n+1}^{\infty} f(a_k^* a_k) \right)^{1/2} \left(\sum_{k=n+1}^{\infty} f(b_k^* b_k) \right)^{1/2} \le \|f\| \|y\| \left(\sum_{k=n+1}^{\infty} f(a_k^* a_k) \right)^{1/2} \le M \|f\| \left(\sum_{k=n+1}^{\infty} f(a_k^* a_k) \right)^{1/2}.$$

By what has been discussed in Section 3.5,

$$\lim_{n \to \infty} \left(\sum_{k=n+1}^{\infty} f(a_k^* a_n) \right)^{1/2} = 0.$$

Thus, for this f and x, we have that $|f(\langle P_n(x), y \rangle) - f(\langle x, y \rangle)|$ converges uniformly on S. Almost identical estimates show that $|f(\langle y, P_n(x) \rangle) - f(\langle y, x \rangle)|$ converges uniformly on S.

Since any $f \in A^*$ can be written as a linear combination of four positive linear functionals in A^* , the first part of the statement holds.

For the second part, we note that, for any $f \in A^*$ and $x \in H_{A^{**}}^{\sharp}$, by the first part of the lemma (since $||P_n(x)|| \le ||x||$),

$$\lim_{n\to\infty} |f(\langle P_n(x), P_n(x)\rangle) - f(\langle x, P_n(x)\rangle)| = 0.$$

We also have

$$\lim_{n \to \infty} |f(\langle P_n(x), x \rangle) - f(\langle x, x \rangle)| = 0.$$

Hence the second part of the lemma also follows.

The following are two easy facts which we present here for convenience.

Lemma 3.9. Let A be a C*-algebra.

(1) Let *H* be a Hilbert A-module and $\{E_{\lambda}\}$ be an approximate identity for K(H). Suppose $T \in K(H)^{**}$ is a nonzero positive element. Then there is λ_0 such that

$$E_{\lambda}TE_{\lambda} \neq 0 \quad for \ all \ \lambda \geq \lambda_0.$$

(2) Let $T \in K(H_A)^{**}$ be a nonzero positive element and $P_n : H_A \to H_n = A^{(n)}$ be the projection $(n \in \mathbb{N})$. Then, there exists $n_0 \in \mathbb{N}$ such that

$$P_nTP_n \neq 0$$
 for all $n \geq n_0$.

Proof. Let $f \in K(H)^*$ be a positive linear functional. Then

$$|f(T^{1/2}(1-E_{\lambda}))|^{2} \le f(T)f((1-E_{\lambda})^{2}) \le f(T)f(1-E_{\lambda}) \to 0.$$

It follows that $f(T^{1/2}E_{\lambda}) \to f(T^{1/2})$ for all positive linear functionals in $K(H)^*$, whence for all $f \in K(H)^*$. Since $T^{1/2} \neq 0$ for some λ_0 , we have that $T^{1/2}E_{\lambda} \neq 0$ for all $\lambda \ge \lambda_0$. It follows that

$$E_{\lambda}TE_{\lambda}\neq 0$$

for all $\lambda \ge \lambda_0$. This proves (1).

There are several easy proofs for (2). Let us use part (1). Choose an approximate identity $\{e_{\alpha}\}$ for *A*. Let $\lambda = (\alpha, n)$ and $\lambda_1 = (\beta_1, n) \le \lambda_2 = (\beta_2, m)$ if $\beta_1 \le \beta_2$ and $n \le m$. Define

$$E_{\beta,n} = \operatorname{diag}(\overbrace{e_{\beta}, e_{\beta}, \dots, e_{\beta}}^{n}, 0, \dots).$$

Then $\{E_{\beta,n}\}$ forms an approximate identity for $K(H_A) \cong A \otimes \mathcal{K}$. Let $T \in K(H_A)^{**}_+$ be a nonzero positive element. By (1), there is β_0 and $n_0 \in \mathbb{N}$ such that

$$E_{\beta,n}TE_{\beta,n} \neq 0$$
 for all $(\beta, n) \ge (\beta_0, n_0)$.

Hence $||T^{1/2}E_{\beta,n}^2T^{1/2}|| = ||E_{\beta,n}TE_{\beta,n}|| \neq 0$ for all $(\beta, n) \ge (\beta_0, n_0)$. Since

$$T^{1/2}P_nT^{1/2} \ge T^{1/2}E_{\beta,n}^2T^{1/2} \neq 0$$

we have $T^{1/2}P_nT^{1/2} \neq 0$. It follows that

$$P_n T P_n \neq 0$$
 for all $n \ge n_0$.

Lemma 3.10. Let A be a C*-algebra and H be a countably generated Hilbert A-module. Then the homomorphism Ψ from $K(H)^{**}$ into $B(H^{\sim})$ (given by Proposition 2.16) is injective.

Proof. Let $H_n = A^{(n)} = \{(a_1, a_2, \dots, a_n) : a_j \in A\}$ be the Hilbert *A*-module whose inner product is defined by $\langle x, y \rangle = \sum_{j=1}^n a_j^* b_j$, where $x = \{a_j\}_{1 \le j \le n}$ and $y = \{b_j\}_{1 \le j \le n}$. One identifies $K(H_n)$ with $M_n(A)$.

<u>Claim</u>: The map $\Psi: K(H_n)^{**} \to B(H_n^{\sim})$ is a W^* -isomorphism.

Since we identify $K(H_n)$ with $M_n(A)$, we have $K(H_n)^{**} = M_n(A^{**})$.

By Proposition 3.6(2) (and Claim 2 of the proof), $(H_n \bullet A^{**})^{\sharp} = (A^{**})^{(n)}$. So $H_n^{\sim} = (A^{**})^{(n)}$. Note that $B(H_n^{\sim}) = M_n(A^{**})$. One then easily checks that $\Psi : K(H_n)^{**} \to B(H_n^{\sim})$ is a W^* -isomorphism. This proves the claim.

Let us consider the homomorphism $\Psi_{H_A} : K(H_A)^{**} \to B(H_A^{\sim})$ given by Proposition 2.14. Put $T \in K(H_A)^{**}_+ \setminus \{0\}.$

By Lemma 3.9 (2), there exists $n_0 \in \mathbb{N}$ such that $P_n T P_n \neq 0$ for all $n \ge n_0$. Recall that H_n is a direct summand of H_A . Hence by the claim and applying Proposition 3.4, we conclude that $\Psi_{H_A}(P_n T P_n) \neq 0$ for all $n \ge n_0$. There must be an element $x \in H_n$ such that

$$\langle \Psi_{H_A}(P_nTP_n)(x), x \rangle \neq 0.$$

It follows that $\langle \Psi_{H_A}(T)x, x \rangle \neq 0$. Hence $\Psi_{H_A}(T) \neq 0$. This implies that ker $\Psi_{H_A} = \{0\}$.

In general, since *H* is countably generated, by Kasparov's absorbing theorem [1980, Theorem 2], we may write $H_A = H \oplus H^{\perp}$. To show Ψ is injective, let $T \in B(H)^{**}$ be a nonzero element. Then $K(H)^{**} = PK(H_A)^{**}P$, where $P : H_A \to H$ is the projection. Hence PTP = T in $K(H_A)^{**}$. We have shown that $\Psi_{H_A}(PTP) \neq 0$. By Proposition 3.4, we have that $\Psi(T) = P\Psi_{H_A}(T)P|_{H^{\sim}} \neq 0$. This implies that Ψ is injective.

Lemma 3.11. Let A be a C^{*}-algebra and H be a countably generated Hilbert A-module. Then there is an isomorphism Ψ from $K(H)^{**}$ onto $B(H^{\sim})$ as W^{*}-algebras.

Proof. By Lemma 3.10 (and by Proposition 2.14), it suffices to show that Ψ is surjective. Let us first consider the case $H = H_A$ (even though H_A is not countably generated when A is not σ -unital). By the end of Definition 2.4 (see also Remark 3.9 (and Proposition 3.10) of [Paschke 1973]), to show that

 $T \in B(H^{\sim}) = B(H_{A^{**}}^{\sharp})$ is in $\Psi(K(H_A)^{**})$, it suffices to show that, for any $\epsilon > 0$, any finite subsets $X \subset H_{A^{**}}^{\sharp}$ and a finite subset $\mathcal{F} \subset A^*$, there exists $S \in K(H)^{**}$ such that

$$f(\langle \Psi(S)(x), y \rangle) - f(\langle T(x), y \rangle) | < \epsilon \text{ for all } x, y \in X \text{ and } f \in \mathcal{F}.$$

We now fix ϵ , X and \mathcal{F} .

For any $T \in B(H_A^{\sim}) = B(H_A^{\sharp})$,

$$|f(\langle P_n T P_n(x), y \rangle - f(\langle T(x), y \rangle)|$$

$$\leq |f(T P_n(x), P_n(y)\rangle) - f(\langle T(x), P_n(y) \rangle)| + |f(\langle T(x), P_n(y) \rangle) - f(\langle T(x), y \rangle)| \quad (3-9)$$

for any $x, y \in H_{A^{**}}^{\sharp}$ and $f \in A^*$. However, $||P_n(y)|| \le ||y||$ for all $n \in \mathbb{N}$. By Lemma 3.8(1),

$$|f(TP_n(x), P_n(y)\rangle) - f(\langle T(x), P_n(y)\rangle)| \to 0,$$

and by Lemma 3.8 (2),

$$|f(\langle T(x), P_n(y)\rangle) - f(\langle T(x), y\rangle)| \to 0$$

It follows that (by (3-9))

$$\lim_{n \to \infty} |f(\langle P_n T P_n(x), y \rangle - f(\langle T(x), y \rangle)| = 0$$

for all $x, y \in H_A^{\sharp}$ and $f \in A^*$.

We then choose $n_0 \in \mathbb{N}$ such that, for all $n \ge n_0$ (recall P_n is a projection),

$$f(\langle P_n T P_n(x), P_n(y) \rangle) - f(T(x), y) | < \epsilon$$
(3-10)

for all $x, y \in X$ and $f \in \mathcal{F}$.

Now fix $n \ge n_0$. Then we have $P_n(x)$, $P_n(y) \in (H_n)^{\sim}$ for all $x, y \in X$, and $P_nTP_n \subset B(H_n^{\sim})$. By the claim for H_n in the proof of Lemma 3.10, we obtain an element $S \in K(H_n)^{**}$ such that $\Psi_n(S) = (P_nTP_n)|_{H_n^{\sim}}$, where

$$\Psi_n: K(H_n)^{**} \cong M_n(A^{**}) \to B(H_n^{\sim}) = M_n(A^{**})$$

is the isomorphism given by the claim. Note, by Proposition 3.4, that $\Psi(S) = P_n \Psi(S) = \Psi(S)P_n = \Psi_n(S)$. Hence it follows that, for all $x, y \in X$ and $f \in \mathcal{F}$ (and $n \ge n_0$), applying (3-10),

$$\begin{aligned} |f(\langle \Psi(S)(x), y \rangle) &- f(\langle T(x), y \rangle)| \\ &= |f(\langle P_n \Psi(S) P_n(x), y \rangle) - f(T(x), y \rangle)| \\ &= |(f(\langle P_n \Psi(S) P_n(x), P_n(y) \rangle) - f(\langle P_n T P_n(x), P_n(y) \rangle) + |f(\langle P_n T P_n(x), P_n(y) \rangle) - f(T(x), y \rangle)| \\ &< 0 + \epsilon = \epsilon. \end{aligned}$$

As mentioned above, this implies that Ψ is surjective.

For a general countably generated Hilbert *A*-module *H*, by Kasparov's absorbing theorem [1980, Theorem 2], we may write $H_A = H \oplus H^{\perp}$. By Proposition 3.4, $H_A^{\sim} = H^{\sim} \oplus (H^{\perp})^{\sim}$. Let $S \in B(H^{\sim}) \setminus \{0\}$. Define $T \in B(H_A^{\sim})$ by $T|_{H^{\sim}} = S$ and $S|_{(H^{\perp})^{\sim})} = \{0\}$. We have shown that there is $L \in B(H_A)^{**}$ such that $\Psi_{H_A}(L) = S$. Then PSP = S, and, by Proposition 3.4, $\Psi(L) = P\Psi_{H_A}(L)P|_{H^{\sim}} = T$. Hence Ψ is surjective. **Theorem 3.12.** Let A be a C^{*}-algebra and H be a Hilbert A-module. Then there is an isomorphism Ψ (given by Proposition 2.14) from $K(H)^{**}$ onto $B(H^{\sim})$ as W^{*}-algebras. Moreover,

$$\Psi|_{B(H)} = \widetilde{\Psi}_0.$$

Proof. By Proposition 2.14, it suffices to show that Ψ is bijective. If K(H) is unital, by Proposition 2.8 of [Brown and Lin 2025], *H* is finitely generated. The theorem then follows from Lemma 3.11. So we will assume that K(H) is not unital.

Let $\{E_{\lambda}\}$ be an approximate identity for K(H) and $H_{\lambda} = \overline{E_{\lambda}(H)}$. Then $K(H_{\lambda}) = \overline{E_{\lambda}K(H)E_{\lambda}}$ is σ -unital. By Proposition 3.2 of [Brown and Lin 2025], H_{λ} is countably generated.

Denote by $P_{\lambda}: H^{\sim} \to H_{\lambda}^{\sim}$ the projection given by Proposition 3.1 and let $\Psi_{\lambda}: K(H_{\lambda})^{**} \to B(H_{\lambda}^{\sim})$ be the map given by Proposition 2.14.

To see Ψ is injective, let $T \in K(H)^{**}_+ \setminus \{0\}$. It follows from Lemma 3.9 that $E_{\lambda}TE_{\lambda} \neq 0$ for all $\lambda \geq \lambda_0$ and some λ_0 . Since H_{λ} is countably generated, by Lemma 3.11, $\Psi_{\lambda}(E_{\lambda}TE_{\lambda}) \neq 0$ (for $\lambda \geq \lambda_0$). By Proposition 3.4,

$$\Psi(E_{\lambda}TE_{\lambda})|_{H_{\lambda}^{\sim}} = \Psi_{\lambda}(E_{\lambda}TE_{\lambda}).$$

It follows that $\Psi(E_{\lambda}TE_{\lambda})|_{H_{\lambda}^{\sim}} \neq 0$ for all $\lambda \geq \lambda_0$. For $\lambda \geq \lambda_0$, there are $x, y \in H_{\lambda}$ such that

$$\langle \Psi(T)(E_{\lambda}(x)), E_{\lambda}(y) \rangle = \langle \Psi(E_{\lambda}TE_{\lambda})(x), y \rangle \neq 0.$$

Hence $\Psi(T) \neq 0$. This shows that Ψ is injective.

To see that Ψ is surjective, let $L \in B(H^{\sim})$. Since, by Proposition 2.14, $\Psi(K(H)^{**})$ is weak*-closed in the W*-algebra $B(H^{\sim})$, it suffices to show the following: for any $\epsilon > 0$, any finite subsets $X, Y \subset H^{\sim}$ and finite subset $\mathcal{F} \subset A^*$, there exists $T \in K(H)^{**}$ such that

$$|f(\langle \Psi(T)(x), y \rangle) - f(\langle L(x), y \rangle)| < \epsilon \quad \text{for all } x \in X, \ y \in Y, \ f \in \mathcal{F}$$
(3-11)

(see the last part of Definition 2.4). We now fix ϵ , X, Y and \mathcal{F} . By Proposition 2.14 (since $E_{\lambda} \nearrow 1_{K(H)^{**}}$),

$$\lim_{\lambda} f(\langle x, \Psi(E_{\lambda})(y) \rangle) = \lim_{\lambda} f(\langle \Psi(E_{\lambda})(x), y \rangle) = f(\langle x, y \rangle)$$

for all $x, y \in H^{\sim}$ and $f \in A^*$. It follows that there is λ_0 such that, for all $\lambda \ge \lambda_0$,

or

$$\begin{aligned} & \left| f(\langle \Psi(E_{\lambda})(x), L^{*}(y) \rangle) - f(\langle x, L^{*}(y) \rangle) \right| < \frac{1}{2}\epsilon \\ & \left| f(\langle L\Psi(E_{\lambda})(x), y \rangle) - f(\langle L(x), y \rangle) \right| < \frac{1}{2}\epsilon \end{aligned}$$

for all $x \in X$, $y \in Y$ and $f \in \mathcal{F}$. We note that the proof would be shorter if we knew

$$\lim_{\lambda} f(\langle \Psi(E_{\lambda})L\Psi(E_{\lambda})(x), y \rangle) = f(\langle L(x), y \rangle)$$

However, we may also assume that, for fixed λ_0 , there is $\lambda_1 \ge \lambda_0$ such that

or

$$\frac{|f(\langle L\Psi(E_{\lambda_0})(x), \Psi(E_{\lambda})(y)\rangle) - f(\langle L\Psi(E_{\lambda_0})(x), y\rangle)| < \frac{1}{2}\epsilon}{|f(\langle \Psi(E_{\lambda})L\Psi(E_{\lambda_0})(x), y\rangle) - f(\langle L\Psi(E_{\lambda_0})(x), y\rangle)| < \frac{1}{2}\epsilon}$$

 \square

for all $x \in X$, $y \in Y$ and $f \in \mathcal{F}$, and $\lambda \ge \lambda_1$. It follows that, for all $x \in X$, $y \in Y$ and $f \in \mathcal{F}$, if $\lambda \ge \lambda_1$, $|f(\langle L(x), y \rangle) - f(\langle \Psi(E_{\lambda})L\Psi(E_{\lambda_0})(x), y \rangle)|$ $\le |f(\langle L(x), y \rangle) - f(\langle L\Psi(E_{\lambda_0})(x), y \rangle)| + |f(\langle L\Psi(E_{\lambda_0})(x), y \rangle) - f(\langle \Psi(E_{\lambda})L\Psi(E_{\lambda_0})(x), y \rangle)|$ $< \frac{1}{2}\epsilon + \frac{1}{2}\epsilon = \epsilon.$ (3-12)

Fix $\lambda \geq \lambda_1 \geq \lambda_0$. Then $H_{\lambda} = \overline{E_{\lambda}(H)} \supset H_{\lambda_0}$. Hence

$$P_{\lambda}\Psi(E_{\lambda}) = \Psi(E_{\lambda}) \text{ and } \Psi(E_{\lambda_0})P_{\lambda} = \Psi(E_{\lambda_0}).$$
 (3-13)

We also note that $\Psi(E_{\lambda})L\Psi(E_{\lambda_0})|_{H_{\lambda}^{\sim}} \in B(H_{\lambda})$. Since H_{λ} is countably generated, by Lemma 3.11, there is $T_{\lambda} \in K(H_{\lambda})^{**}$ such that

$$\Psi_{\lambda}(T_{\lambda}) = \Psi(E_{\lambda})L\Psi(E_{\lambda_0})|_{H_{\lambda}^{\sim}}.$$
(3-14)

However, by Proposition 3.4,

$$P_{\lambda}\Psi(T_{\lambda})P_{\lambda}|_{H_{\lambda}^{\sim}} = \Psi(T_{\lambda})|_{H_{\lambda}^{\sim}} = \Psi_{\lambda}(T_{\lambda}).$$
(3-15)

Fix $\lambda \ge \lambda_1 \ge \lambda_0$. Then, for any $x \in X$, $y \in Y$ and $f \in A^{**}$, by (3-15), (3-14), (3-13) and (3-12),

$$\begin{split} |f(\langle \Psi(T_{\lambda})(x), (y) \rangle) - f(\langle L(x), y \rangle)| &= |f(\langle \Psi(T_{\lambda})P_{\lambda}(x), P_{\lambda}(y) \rangle) - f(\langle L(x), y \rangle)| \\ &= |f(\langle \Psi(E_{\lambda})L\Psi(E_{\lambda_0})P_{\lambda}(x), P_{\lambda}(y) \rangle) - f(\langle L(x), y \rangle)| \\ &= |f(\langle P_{\lambda}\Psi(E_{\lambda})L\Psi(E_{\lambda_0})P_{\lambda}(x), y \rangle) - f(\langle L(x), y \rangle)| \\ &= |f(\langle \Psi(E_{\lambda})L\Psi(E_{\lambda_0})(x), y \rangle) - f(\langle L(x), y \rangle)| < \epsilon. \end{split}$$

As mentioned above, this implies that Ψ is surjective.

Corollary 3.13. Let A be a W^{*}-algebra and H be a Hilbert A-module. Then $F : K(H)^{**} \to B(H^{\sharp})$, the map given by Proposition 2.16, is a surjective map.

Proof. Consider the pair A and A^{**} and $H^{\sim} = (H \bullet A^{**})^{\sharp}$. By Corollary 4.3 of [Paschke 1973], $H^{\sim} = B(H, A^{**})$, the A^{**} -module of all bounded A^{**} -valued A-module maps from H into A^{**} . It follows that $H^{\sharp} \subset H^{\sim}$ as an A-submodule. It then follows from Proposition 2.6 that $H^{\sharp} \bullet A^{**} \subset H^{\sim}$ as Hilbert A^{**} -modules. Then, by applying Proposition 3.1,

$$(H^{\sharp} \bullet A^{**})^{\sharp} \subset H^{\sim}.$$

However, $H \bullet A^{**} \subset H^{\sharp} \bullet A^{**}$. By applying Proposition 3.1 again, we obtain

$$H^{\sim} = (H \bullet A^{**})^{\sharp} \subset (H^{\sharp} \bullet A^{**})^{\sharp} \subset H^{\sim}$$

Hence $(H^{\sharp} \bullet A^{**})^{\sharp} = H^{\sim}$. Denote by $\widetilde{\Psi} : K(H)^{**} \to B(H^{\sim})$ the isomorphism given by Theorem 3.12 and by $\widetilde{\Psi}_{H^{\sharp}} : B(H^{\sharp}) \to B((H^{\sharp} \bullet A^{**})^{\sharp}) = B(H^{\sim})$ the map given by Theorem 3.12.

Now let $T \in B(H^{\sharp})$. Then, by applying Theorem 3.12, we obtain $a \in K(H)^{**}$ such that $\widetilde{\Psi}(a) = \widetilde{\Psi}_{H^{\sharp}}(T)$. It follows that (viewing $H^{\sharp} \subset H^{\sim}$)

$$\Psi(a)|_{H^{\sharp}} = T.$$

Since $a \in K(H)^{**}$, there exists a net $\{a_{\alpha}\}$ in K(H) such that $a_{\alpha} \to a$ in the weak* topology. Therefore, by Proposition 2.14, for any $f \in A^*$ and any $\xi, \zeta \in H^{\sim}$,

$$\lim_{\alpha} f(\langle (\widetilde{\Psi}(a) - \widetilde{\Psi}(a_{\alpha}))(\xi), \zeta \rangle) = 0.$$

Note, by Theorem 3.12, $\widetilde{\Psi}(a_{\alpha}) = \widetilde{\Psi}_0(a_{\alpha})$. On the other hand, by Proposition 2.16, for any $g \in A_*$ and any $x, y \in H$,

$$\lim g(\langle (F(a) - a_{\alpha})(x), y \rangle) = 0.$$

Hence (since $\widetilde{\Psi}_0(a_\alpha)x = a_\alpha(x)$ for all $x \in H$, see Definition 2.12)

$$g(\langle (F(a) - \Psi(a))(x), y \rangle) = 0$$
 for all $x, y \in H$ and $g \in A_*$.

Since $\widetilde{\Psi}(a)|_{H^{\sharp}} = T$, we actually have

$$g(\langle (F(a) - T)(x), y \rangle) = 0 \quad \text{for all } x, y \in H \text{ and } g \in A_*.$$
(3-16)

Note that $F(a), T \in B(H^{\sharp})$. So $F(a)(x), T(x) \in H^{\sharp}$ for all $x \in H$. It follows that

 $\langle (F(a) - T)(x), y \rangle \in A$ for all $x, y \in H$.

Then, by (3-16),

 $\langle (F(a) - T)(x), y \rangle = 0$ for all $x, y \in H$.

Hence F(a) = T. In other words, F is surjective.

4. A Kaplansky density theorem in Hilbert modules

As mentioned in the introduction, in this section we study the density of H in $H \bullet A^{**}$.

Definition 4.1. Let *X* be a Hilbert space and $A \subset B(X)$ be a *C**-subalgebra of B(X). Let $M = \overline{A}^{SOT}$, the strong operator closure of *A*, and let *H* be a Hilbert *A*-module. Recall, by Proposition 2.6, $H \bullet M$ is the smallest Hilbert *M*-module containing *H* as a Hilbert *A*-module. We consider the question of how large *H* is in $H \bullet M$ as a submodule.

Let $\epsilon > 0$ and V be a finite subset of X. For each $\xi \in H \bullet M$, define

$$N_{\xi,\epsilon,V} = \{ z \in H \bullet M : \|\langle \xi - z, \xi - z \rangle(v) \| < \epsilon, v \in V \}.$$

Let \mathcal{T}_s be the topology generated by $N_{\xi,\epsilon,V}$ for all $\xi \in H \bullet M$, $\epsilon \in \mathbb{R}_+ \setminus \{0\}$, and any finite subset $V \subset X$. In other words, in \mathcal{T}_s , a net $\{z_\alpha\}$ converges to ξ in $H \bullet M$ if and only if

$$\lim_{\alpha \to \infty} \|\langle \xi - z_{\alpha}, \xi - z_{\alpha} \rangle(v)\| = 0 \quad \text{for all } v \in X.$$

In the special case that $X = H_U$ is the Hilbert space corresponding to the universal representation π_U of A and $M = A^{**}$, we use \mathcal{T}_{su} for the topology generated by $N_{\xi,\epsilon,V}$ for all $\xi \in H \bullet A^{**}$, $\epsilon \in \mathbb{R}_+ \setminus \{0\}$, and any finite subset $V \subset H_U$.

We note that *H* is dense in $H \bullet M$ in the topology \mathcal{T}_s , but to be more useful, we will show in Theorem 4.4 that the unit ball of *H* is dense in the unit ball of $H \bullet M$ in \mathcal{T}_s , a Kaplansky-style density theorem.

Lemma 4.2. Suppose that $x \in H \bullet M$ and $\{x_{\alpha}\} \subset H \bullet M$ is a bounded net. Then $x_{\alpha} \to x$ in \mathcal{T}_s if and only if, for any $v \in X$,

$$\limsup\{\|\langle y, x_{\alpha} - x \rangle(v)\| : y \in H \bullet M, \|y\| \le 1\} = 0.$$

Moreover, if $x_{\alpha} \to x$ in \mathcal{T}_s , then, for any $f \in M_*$,

$$\limsup\{|f(\langle y, x_{\alpha} - x \rangle)| : y \in H \bullet M, ||y|| \le 1\} = 0.$$

Proof. Suppose that $x_{\alpha} \to x$ in \mathcal{T}_s . We have (see Proposition 2.3 (ii) of [Paschke 1973]), for any $y \in H \bullet M$,

$$\langle x_{\alpha} - x, y \rangle \cdot \langle y, x_{\alpha} - x \rangle \leq ||y||^{2} \langle x_{\alpha} - x, x_{\alpha} - x \rangle.$$

Then, for any $v \in X$ and any $y \in H \bullet M$ with $||y|| \le 1$,

$$\|\langle y, x_{\alpha} - x \rangle \langle v \rangle \|^{2} = \langle \langle x_{\alpha} - x, y \rangle \cdot \langle y, x_{\alpha} - x \rangle v, v \rangle_{X}$$

$$\leq \|y\|^{2} \langle \langle x_{\alpha} - x, x_{\alpha} - x \rangle v, v \rangle_{X} \leq \|\langle x_{\alpha} - x, x_{\alpha} - x \rangle v \| \|v\| \to 0$$

(where $\langle \cdot, \cdot \rangle_X$ is the inner product in the Hilbert space *X*). Conversely, let $K = \sup_{\alpha} \{ ||x_{\alpha}|| + ||x|| \} + 1$. Then

$$\|\langle x_{\alpha} - x, x_{\alpha} - x \rangle(v)\| \le K \sup\{\|\langle y, x_{\alpha} - x \rangle(v)\| : y \in H \bullet M, \|y\| \le 1\} \to 0$$

For the "moreover" part of the lemma, suppose that $\langle x_{\alpha} - x, x_{\alpha} - x \rangle \rightarrow 0$ in the strong operator topology. Then it converges in the weak operator topology. However, since $\{\langle x_{\alpha} - x, x_{\alpha} - x \rangle\}$ is bounded, this also implies that it converges to zero in the σ -weak topology and in the weak* topology. Hence

$$\lim_{\alpha} f(\langle x_{\alpha} - x, x_{\alpha} - x \rangle) = 0 \quad \text{for all } f \in M_*.$$

Let $f \in M_*$ be a positive normal functional. Then, $f(\langle \cdot, \cdot \rangle)$ defines a pseudo inner product on $H \bullet M$. Hence, for any $y \in H \bullet M$, we have, by the Cauchy–Bunyakovsky–Schwarz inequality,

$$|f(\langle y, x_{\alpha} - x \rangle)|^{2} \leq f(\langle y, y \rangle) f(\langle x_{\alpha} - x, x_{\alpha} - x \rangle) \leq ||f||^{2} ||y||^{2} f(\langle x_{\alpha} - x, x_{\alpha} - x \rangle).$$

Thus

$$\lim_{\alpha} \sup\{|f(\langle y, x_{\alpha} - x \rangle)| : y \in H \bullet M, \|y\| \le 1\} = 0.$$

Lemma 4.3. Let X be a Hilbert space, $A \subset B(X)$ be a C*-subalgebra and $M = \overline{A}^{SOT}$ such that $id_X \in M$. Then the unit ball of H_A is dense in the unit ball of $H_A \bullet M$ in \mathcal{T}_s .

Proof. Let $\xi \in H_A \bullet M$ with $\|\xi\| \le 1$. We will show that there is a net $\{x_\alpha\} \in H$ such that $\|x_\alpha\| \le \|\xi\|$ and $\lim_\alpha \|\langle x_\alpha - \xi, x_\alpha - \xi \rangle(v)\| = 0$ for all $v \in X$. From the inequality

$$\|\langle x_{\alpha} - \xi, x_{\alpha} - \xi \rangle(v)\| \le \|\langle x_{\alpha} - \xi, x_{\alpha} - \xi \rangle^{1/2}\| \|\langle x_{\alpha} - \xi, x_{\alpha} - \xi \rangle^{1/2}(v)\| \le 2\|\langle x_{\alpha} - \xi, x_{\alpha} - \xi \rangle^{1/2}(v)\|,$$

we conclude that it is enough to show that there is a net $\{x_{\alpha}\} \in H$ such that

$$||x_{\alpha}|| \le ||\xi||$$
 and $\lim_{\alpha} ||\langle x_{\alpha} - \xi, x_{\alpha} - \xi\rangle|^{1/2}v|| = 0$

for all $v \in X$. Therefore it suffices to show that, for any $\epsilon > 0$ and any finite subset $V \subset X$, there exists $z \in H$ with $||z|| \le 1$ such that

$$\|(\langle \xi - z, \xi - z \rangle)^{1/2}(v)\| < \epsilon \quad \text{for all } v \in V.$$

To simplify notation, we may also assume that $||v|| \le 1$ for all $v \in V$.

Denote by $R = \overline{AM}$, the closed right ideal of M generated by A. Note, by Proposition 3.6,

$$H_A \bullet M = \{\{b_n\} \in H_B : b_n \in R\}.$$

We write $\xi = \{b_n\} \in H_A \bullet M$. There exists $n_0 \in \mathbb{N}$ such that, for all $n \ge n_0$,

$$\left\|\sum_{k=n}^{\infty}b_n^*b_n\right\| < \frac{1}{2}\epsilon.$$

Fix an integer $n \ge n_0$. Let $P_n : H_A \bullet M \to R^{(n)} = \{(c_1, c_2, \dots, c_n) : c_i \in R\}$ be the projection. Put

$$S = \begin{pmatrix} b_1 & 0 & 0 & \cdots & 0 \\ b_2 & 0 & 0 & \cdots & 0 \\ \vdots & \vdots & \vdots & & \vdots \\ b_n & 0 & 0 & \cdots & 0 \end{pmatrix}.$$

For any $v \in V$, put

$$u_v = \begin{pmatrix} v \\ 0 \\ \vdots \\ 0 \end{pmatrix}.$$

By the Kaplansky density theorem, there is $L \in M_n(A)$ such that

$$||L|| \le ||S||$$
 and $||L(u_v) - S(u_v)|| < \frac{1}{2}\epsilon$ (4-1)

for all $v \in V$. Hence, denoting by $\langle \cdot, \cdot \rangle_X$ the inner product in X,

$$\langle (L-S)^*(L-S)u_v, u_v \rangle_X < \frac{1}{2}\epsilon \quad \text{for all } v \in V.$$
(4-2)

Define $q = \text{diag}(1, 0, \dots, 0) \in M_n(M)$. Then S = Sq. Replacing L by Lq, we may write

$$L = \begin{pmatrix} a_1 & 0 & 0 & \cdots & 0 \\ a_2 & 0 & 0 & \cdots & 0 \\ \vdots & \vdots & \vdots & & \vdots \\ a_n & 0 & 0 & \cdots & 0 \end{pmatrix},$$

where $a_i \in A$, i = 1, 2, ..., n. Then

$$\left\|\sum_{i=1}^{n} a_{i}^{*} a_{i}\right\| = \|L^{*}L\| = \|L\|^{2} \le \|S\|^{2} = \left\|\sum_{i=1}^{n} b_{i}^{*} b_{i}\right\| \le \|\xi\|^{2}.$$
(4-3)

It follows from (4-2) that

$$\left\langle \sum_{i=1}^{n} (b_i - a_i)^* (b_i - a_i)(v), v \right\rangle_X < \frac{1}{2}\epsilon$$

Put $x = (a_1, a_2, \dots, a_n, 0, 0, \dots) \in H_A$. Then, by (4-3), we have $||x|| \le ||\xi||$ and

$$\langle \langle \xi - x, \xi - x \rangle \langle v \rangle, v \rangle_X = \left\langle \sum_{i=1}^n (b_i - a_i)^* (b_i - a_i) \langle v \rangle, v \right\rangle_X + \left\langle \sum_{i=n+1}^\infty b_i^* b_i \langle v \rangle, v \right\rangle_X$$
$$< \frac{1}{2}\epsilon + \left\| \sum_{i=n+1}^\infty b_i^* b_i \right\| < \frac{1}{2}\epsilon + \frac{1}{2}\epsilon = \epsilon.$$

In other words, for any $v \in V$,

$$\|(\langle \xi - x_{\alpha}, \xi - x_{\alpha} \rangle^{1/2}(v)\| = \langle \langle \xi - x, \xi - x \rangle(v), v \rangle_{X} < \epsilon.$$

The lemma then follows.

Theorem 4.4. Let X be a Hilbert space, $A \subset B(X)$ be a C*-subalgebra and $M = \overline{A}^{SOT}$, with $id_X \in M$. Let H be a Hilbert A-module. Then the unit ball of H is dense in the unit ball of $H \bullet M$ in \mathcal{T}_s .

Proof. Let $\xi \in H \bullet M$, with $\|\xi\| \le 1$.

Let us first assume that *H* is a countably generated *A*-module. By Lemma 4.2, it suffices to show that, for any $\epsilon > 0$ and any finite subset $V \subset X$, there exists $z \in H$ with $||z|| \le 1$ such that

$$\|\langle y, \xi - z \rangle(v)\| < \epsilon$$
 for all $y \in H$, $\|y\| \le 1$ and $v \in V$.

To simplify notation, we may also assume that $||v|| \le 1$ for all $v \in V$.

By Kasparov's absorbing theorem [1980, Theorem 2], we may write $H_A = H \oplus H^{\perp}$. It follows that

$$H_A \bullet M = H \bullet M \oplus H^{\perp} \bullet M.$$

Define $Q: H_A \to H$ to be the projection. Then $Q \in L(H_A) = M(K(H_A))$. We identify Q with $\Psi_0(Q)$ in the sense that $Q \in L(H_M)$ which extends $Q|_{H_A}$. In particular, $H \bullet M = Q(H_A \bullet M)$.

By applying Lemmas 4.3 and 4.2, we obtain $z \in H_A$ with $||z|| \le ||\xi||$ such that

$$\|\langle y, \xi - z \rangle(v)\| < \epsilon$$
 for all $y \in H \bullet M$, $\|y\| \le 1$, and $v \in V$.

Note $Q(\xi) = \xi$ and Q(y) = y for all $y \in H$. Put $x = Q(z) \in H$. We have

$$\|\langle y,\xi-x\rangle(v)\| = \|\langle y,Q(\xi)-Q(z)\rangle\rangle(v)\| = \|\langle Q(y),\xi-z\rangle(v)\| = \|\langle y,\xi-z\rangle(v)\| < \epsilon.$$

This proves the case that H is countably generated.

Next we let *H* be a general Hilbert *A*-module. We will show that, for any $\epsilon > 0$ and any finite subset $V \subset X$, there exists $z \in H$ with $||z|| \le 1$ such that

$$\|\langle \xi - z, \xi - z \rangle (v)\| < \epsilon$$
 for all $v \in V$.

Again, we may also assume that $||v|| \le 1$ for all $v \in V$.

Let $\{E_{\lambda}\}$ be an approximate identity for K(H). Then, as in the proof of Theorem 3.12, $H_{\lambda} = \overline{E_{\lambda}(H)}$ is countably generated for each λ . It follows from Lemma 2.9 that there is λ such that

$$\|\Psi_0(E_{\lambda})(\xi) - \xi\| < \frac{1}{4}\epsilon.$$
(4-4)

Fix such a λ . Note that, by Proposition 2.8, $\Psi_0(E_\lambda)(\xi) \in H_\lambda \bullet M \subset H \bullet M$. Since H_λ is countably generated, by the first part of the proof, we obtain $x \in H_\lambda$ with $||x|| \le ||\Psi_0(E_\lambda)(\xi)|| \le ||\xi||$ such that

$$\sup\{\|\langle y, \Psi_0(E_{\lambda})(\xi) - x\rangle(v)\| : y \in H \bullet M, \|y\| \le 1\} < \frac{1}{4}\epsilon.$$
(4-5)

Then, applying (4-4) and then (4-5), for any $v \in V$,

$$\begin{aligned} \|\langle \xi - x, \xi - x \rangle(v)\| &\leq \|\langle \xi - x, \xi - \Psi_0(E_{\lambda})(\xi) \rangle(v)\| + \|\langle \xi - x, \Psi_0(E_{\lambda})(\xi) - x \rangle(v)\| \\ &< 2\|\xi - \Psi_0(E_{\lambda})(\xi)\| + 2\|\langle \frac{1}{2}(\xi - x), \Psi_0(E_{\lambda})(\xi) - x \rangle(v)\| < \frac{1}{2}\epsilon + \frac{1}{2}\epsilon = \epsilon. \quad \Box \end{aligned}$$

We then obtain the following corollary as a Kaplansky density theorem.

Theorem 4.5. Let A be a C^{*}-algebra and H be a Hilbert A-module. Then the unit ball of H is dense in the unit ball of $H \bullet A^{**}$ in \mathcal{T}_{su} .

5. Closeness of H

Let *H* be a Hilbert *A*-module., Then, by Theorem 6.1 of [Brown and Lin 2025], the unit ball of *H* is *A*-weakly dense (see Definition 3.3 of [Brown and Lin 2025]) in the unit ball of H^{\sharp} , i.e., for any $f \in H^{\sharp}$, there is a net $\{x_{\alpha}\}$ in *H* with $||x_{\alpha}|| \leq ||f||$ such that $\lim_{\alpha} ||\langle f - x_{\alpha}, y \rangle|| = 0$ for all $y \in H$. In the case that *A* is a *W**-algebra, H^{\sharp} is a Hilbert *A*-module. One may ask: can one find the net $\{x_{\alpha}\} \in H$ with $||x_{\alpha}|| \leq ||f||$ such that $\lim_{\alpha} ||\langle f - x_{\alpha}, \xi \rangle|| = 0$ for all $\xi \in H^{\sharp}$?

We begin with the following example.

Example 5.1. Let *M* be a *W*^{*}-algebra which contains a self-adjoint element *a* with infinite spectrum. Then, by the spectral theory, one obtains a sequence of mutually orthogonal nonzero projections $p_1, p_2, \ldots, p_n, \ldots$ Let $H = H_M$, and let $\xi = \{p_n\} \in H_M^{\sharp}$. Note that $\|\xi\| = \|\sum_{n=1}^{\infty} p_n\| = 1$ (the convergence is in the strong operator topology and weak* topology of *M*). We claim that there is *no* net $\{x_{\alpha}\}$ in H_M such that

$$\lim_{\alpha} \|\langle \xi - x_{\alpha}, \xi \rangle\| = 0.$$

Otherwise, there would be $x \in H_M$ such that

$$\|\langle \xi - x, \xi \rangle\| < \frac{1}{4}.\tag{5-1}$$

Since $x = \{a_n\} \in H_M$, there is $N \in \mathbb{N}$ such that

$$\left\|\sum_{N+1} a_n^* a_n\right\| < \left(\frac{1}{16}(1+\|x\|)\right)^2.$$

Choose $q = \sum_{n=N+1}^{\infty} p_n \in M$. Define $P_N : H_M^{\sharp} \to M^{(N)} = \{(b_1, b_2, \dots, b_N) : b_i \in M\}$ to be the projection. Then

$$\begin{aligned} \|\langle \xi - P_N(x), \xi \rangle\| &\leq \|\langle \xi - x, \xi \rangle\| + \|\langle (1 - P_N)(x), \xi \rangle\| \\ &< \frac{1}{4} + \|(1 - P_N)(x)\| \|\xi\| < \frac{1}{4} + \frac{1}{16} = \frac{5}{16}. \end{aligned}$$

On the other hand,

$$\frac{5}{16} \ge \|\langle \xi - P_N(x), \xi \rangle\| \ge \|\langle \xi - P_N(x), \xi \rangle q\| = \left\| \left(\sum_{N+1}^{\infty} p_n - \sum_{i=1}^{N} (p_i - a_i)^* p_i \right) q \right\| = \left\| \sum_{N+1}^{\infty} p_n q \right\| = 1.$$

A contradiction. In other words, the question at the beginning of this section is negative. This also follows from Corollary 5.7 below. However, we think that the example above might also be helpful.

Lemma 5.2. Let A be a C^{*}-algebra. Suppose that $\xi \in H_A^{\sharp}$ and $\{x_{\alpha}\}$ is a bounded net in H_A such that

$$\lim_{\alpha} \|\xi(x) - x_{\alpha}(x)\| = 0 \quad \text{for all } x \in H_A$$

and $\xi(x_{\alpha}) := \langle \xi, x_{\alpha} \rangle$ converges in norm. Then $\xi \in H_A$ and $\langle \xi, \xi \rangle = \lim_{\alpha} \langle \xi, x_{\alpha} \rangle$.

Proof. Write $\xi = \{b_n\}$ and $x_{\alpha} = \{a_{\alpha,n}\}$, where $\{b_n\} \in H_A^{\sharp}$, $a_{\alpha,n} \in A$ and, for each α , $\{a_{\alpha,n}\} \in H_A$. Put

$$M = 1 + \sup\{\|x_{\alpha}\| : \alpha\} + \|\xi\| < \infty \text{ and } a = \lim_{\alpha} \langle x_{\alpha}, \xi \rangle.$$

Note $\xi(x_{\alpha}) = \langle \xi, x_{\alpha} \rangle \in A$ for all α . Hence $a \in A$.

Let $P_n : H_A^{\sharp} \to H_n := A^{(n)}$ be the projection to the first *n* copies of *A*, $n \in \mathbb{N}$. Then $P_n \xi \in H_n \subset H_A$. It follows that, for each $n \in \mathbb{N}$,

$$\lim_{\alpha} \langle x_{\alpha}, P_n(\xi) \rangle = \langle \xi, P_n(\xi) \rangle = \sum_{j=1}^n b_j^* b_j.$$
(5-2)

Fix $f \in A^*$. Let $\epsilon > 0$. By Lemma 3.8, since $\{x_{\alpha}\}$ is bounded, there is an integer $N \in \mathbb{N}$ such that, for all $n \ge N$,

$$|f(\langle x_{\alpha},\xi\rangle) - f(\langle x_{\alpha},P_n(\xi)\rangle)| < \frac{1}{3}\epsilon \quad \text{for all } \alpha.$$
(5-3)

Fix any $n \ge N$. By (5-2), choose α_0 such that, for all $\alpha \ge \alpha_0$,

$$\left\| \langle x_{\alpha}, P_{n}(\xi) \rangle - \sum_{j=1}^{n} b_{j}^{*} b_{j} \right\| < \frac{1}{3} \epsilon (1 + \|f\|),$$
(5-4)

$$\|\langle x_{\alpha},\xi\rangle - a\| < \frac{1}{3}\epsilon(1 + \|f\|).$$
(5-5)

It follows that, for all $n \ge N$, by (5-5), (5-3) and (5-4),

$$\begin{aligned} \left| f(a) - f\left(\sum_{j=1}^{n} b_{j}^{*} b_{j}\right) \right| \\ & \leq \left| f(a - \langle x_{\alpha_{0}}, \xi \rangle) \right| + \left| f(\langle x_{\alpha_{0}}, \xi \rangle) - f(x_{\alpha_{0}}, P_{n}(\xi)) \right| + \left\| f \right\| \left\| \langle x_{\alpha_{0}}, P_{n}(\xi) \rangle - \sum_{j=1}^{n} b_{j}^{*} b_{j} \right\| \\ & < \frac{1}{3}\epsilon + \frac{1}{3}\epsilon + \frac{1}{3}\epsilon = \epsilon. \end{aligned}$$

Hence, on the state space S(A) of A,

$$\lim_{n \to \infty} f\left(\sum_{j=1}^{n} b_j^* b_j\right) = f(a).$$
(5-6)

HUAXIN LIN

On the compact space S(A) (in the weak* topology), $\hat{a}(f) = f(a)$ is a continuous function for all $f \in S(A)$, and $\sum_{j=1}^{n} b_j^* b_j$ is increasing. By the Dini theorem, $\sum_{j=1}^{n} b_j^* b_j$ converges uniformly to \hat{a} on S(A). It follows that

$$\sum_{j=1}^{n} b_j^* b_j \to a$$

in norm. This implies that $\xi = \{b_n\} \in H_A$ and $\langle \xi, \xi \rangle = a = \lim_{\alpha} \langle \xi, x_{\alpha} \rangle$.

Proposition 5.3. Let A be a C^* -algebra and H be a Hilbert A-module. Then, for any $T \in K(H)$, one has $\Psi_0(T)(H^{\sharp}) \subset H$, where Ψ_0 is given in Definition 2.7.

Proof. Suppose that $T \in F(H)$ and $T = \sum_{i=1}^{m} \theta_{x_i, y_i}$ for some $x_i, y_i \in H, i = 1, 2, ..., m$. Then, for any $\xi \in H^{\sharp}$,

$$\Psi_0(T)(\xi) = \sum_{i=1}^m x_i \langle y_i, \xi \rangle = \sum_{i=1}^m x_i(\xi(y_i)^*) \in H.$$

Since F(H) is dense in K(H), this implies that $\Psi_0(T)(H^{\sharp}) \subset H$.

Lemma 5.4. Let A be a C^{*}-algebra, H be a Hilbert A-module and $\{E_{\lambda}\}$ be an approximate identity for K(H). Then, for any $\xi \in H^{\sim}$ and any $f \in A^*$,

$$\lim_{\alpha} \sup\{f(\langle \xi - \Psi_0(E_{\lambda})(\xi), y\rangle) \mid : y \in H^{\sim}, \|y\| \le 1\} = 0.$$

Proof. By Lemma 2.9, $\{\Psi_0(E_\lambda)\}\$ is an approximate identity for $K(H \bullet A^{**})$. In the universal representation of $K(H \bullet A^{**})$, $1 - \Psi_0(E_\lambda)$ converges to zero in the strong operator topology. Note that $||1 - \Psi_0(E_\lambda)|| \le 1$. Therefore $(1 - \Psi_0(E_\lambda))(1 - \Psi_0(E_\lambda))$ also converges to zero in the strong operator topology. Hence it converges to zero in the weak operator topology. Since $\{(1 - \Psi_0(E_\lambda))^2\}$ is bounded, it also converges to zero in the weak* topology of $K(H \bullet A^{**})$. Recall that $(H \bullet A^{**})^{\sharp} = H^{\sim}$. It follows from Proposition 2.16, for any $\xi \in H^{\sim}$, that

$$\lim_{\alpha} |f(\langle \xi - \Psi_0(E_{\lambda})(\xi), \xi - \Psi_0(E_{\lambda})(\xi)\rangle)| = \lim_{\alpha} |f(\langle \xi - F \circ \Psi_0(E_{\lambda})(\xi), \xi - F \circ \Psi_0(E_{\lambda})(\xi)\rangle)|$$
$$= \lim_{\alpha} |f(\langle (1 - F \circ \Psi_0(E_{\lambda}))^2(\xi), \xi\rangle)| = 0,$$

where $F : K(H)^{**} \to B(H^{\sharp})$ is the homomorphism given by Proposition 2.16. Suppose that $y \in H^{\sim}$ and $||y|| \le 1$. Then, for any positive linear functional $f \in A^*$,

$$f(\langle \xi - \Psi_0(E_\lambda)(\xi), y \rangle)^2 \le f(\langle \xi - \Psi_0(E_\lambda)(\xi), \xi - \Psi_0(E_\lambda)(\xi) \rangle) f(\langle y, y \rangle)$$

$$\le \|f\| f(\langle \xi - \Psi_0(E_\lambda)(\xi), \xi - \Psi_0(E_\lambda)(\xi) \rangle).$$

It follows that, for any $f \in A^*$,

$$\lim_{\alpha} \sup\{f(\langle \xi - \Psi_0(E_{\lambda})(\xi), y\rangle) | : y \in H^{\sim}, \|y\| \le 1\} = 0.$$

Theorem 5.5. Let A be a C^{*}-algebra and H be a Hilbert A-module. Suppose that $\xi \in H^{\sharp}$ and there is a bounded net $\{x_{\alpha}\}$ in H such that

$$\lim_{\alpha} \|\xi(x) - \langle x_{\alpha}, x \rangle\| = 0 \quad for \ all \ x \in H$$

and $\xi(x_{\alpha}) := \langle \xi, x_{\alpha} \rangle$ converges in norm. Then $\xi \in H$ and $\langle \xi, \xi \rangle = \lim_{\alpha} \langle \xi, x_{\alpha} \rangle \in A$.

1558
Proof. First let us assume H is countably generated. Then, by Kasparov's absorbing theorem [1980, Theorem 2], we may write $H_A = H \oplus H^{\perp}$. Then $\xi \in H^{\sharp} \subset H_A^{\sharp}$. By applying Lemma 5.2, we obtain that

$$\xi \in H_A$$
 and $\langle \xi, \xi \rangle = \lim_{\alpha} \langle \xi, x_{\alpha} \rangle.$

Since $\xi(x_{\alpha}) \in A$, we have $a = \langle \xi, \xi \rangle \in A$. Let $P : H_A \to H$ be the projection. Then $P \in L(H_A)$. Put $\eta = P(\xi) \in H$. Note that $\langle P(\xi) - \xi, x \rangle = 0$ for all $x \in H$. Hence $\xi = \eta$. Therefore this case follows.

In what follows we will work in H^{\sim} and use the inner product in H^{\sim} whenever it is convenient.

In general, let $a = \lim_{\alpha} \langle \xi, x_{\alpha} \rangle$. Since $\langle \xi, x_{\alpha} \rangle = \xi(x_{\alpha}) \in A$ for all α , we have $a \in A$.

<u>Claim</u>: $a = \langle \xi, \xi \rangle$ (in the inner product of H^{\sim}).

Let $\{E_{\lambda}\}$ be an approximate identity for K(H). Let $\epsilon > 0$ and $f \in A^*$, with $||f|| \le 1$. By applying Lemma 5.4, we have (since $\{||\xi - x_{\alpha}||\}$ is bounded)

$$\lim_{\lambda} \left(\sup_{\alpha} \{ |f(\langle \xi - \Psi_0(E_{\lambda})(\xi), \xi - x_{\alpha} \rangle)| \} \right) = 0.$$
(5-7)

Thus, by applying Lemma 5.4 and (5-7), we obtain λ_0 such that, for all $\lambda \ge \lambda_0$,

$$|f(\langle \xi - \Psi_0(E_{\lambda})(\xi), \xi \rangle)| < \frac{1}{3}\epsilon,$$

$$|f(\langle \xi - \Psi_0(E_{\lambda})(\xi), \xi - x_{\alpha} \rangle)| < \frac{1}{3}\epsilon \quad \text{for all } \alpha.$$
(5-8)

Recall that, by Proposition 5.3, $\Psi_0(E_\lambda)(\xi) \in H$. Fix any $\lambda \ge \lambda_0$. Choose α_0 such that, for any $\alpha \ge \alpha_0$,

$$\|\langle \xi, x_{\alpha} \rangle - a\| < \frac{1}{3}\epsilon \quad \text{and} \quad |f(\langle \Psi_0(E_{\lambda})(\xi), \xi - x_{\alpha} \rangle)| < \frac{1}{3}\epsilon.$$
(5-9)

Now, by the first inequality of (5-9), (5-8) and then the second inequality of (5-9),

$$\begin{aligned} |f(\langle \xi, \xi \rangle - a)| &< |f(\langle \xi, \xi \rangle - \langle \xi, x_{\alpha} \rangle)| + \frac{1}{3}\epsilon = |f(\langle \xi, \xi - x_{\alpha} \rangle)| + \frac{1}{3}\epsilon \\ &\leq |f(\langle \xi - \Psi_0(E_{\lambda})(\xi), \xi - x_{\alpha} \rangle)| + |f(\langle \Psi_0(E_{\lambda})(\xi), \xi - x_{\alpha} \rangle)| + \frac{1}{3}\epsilon < \frac{1}{3}\epsilon + \frac{1}{3}\epsilon + \frac{1}{3}\epsilon = \epsilon. \end{aligned}$$

Since this holds for any ϵ , we conclude that

$$f(\langle \xi, \xi \rangle) = f(a)$$
 for all $f \in A^*$.

By the Hahn–Banach theorem, we obtain $\langle \xi, \xi \rangle = a$. The claim is proved.

There exists $x_1 \in \{x_\alpha\}$ and then $x_2 \in \{x_\alpha\}$ such that

$$\|\langle x_1, \xi \rangle - a\| < \frac{1}{2}, \quad \|\langle \xi - x_2, x_1 \rangle\| < \frac{1}{4} \text{ and } \|\langle x_2, \xi \rangle - a\| < \frac{1}{4}.$$

Suppose that we have found x_1, x_2, \ldots, x_n such that

$$\|\langle \xi - x_j, x_i \rangle\| < 1/2^j$$
 and $\|\langle x_j, \xi \rangle - a\| < 1/2^j$, $i = 1, 2, ..., j - 1$,

and j = 1, 2, ..., n. Then choose $x_{n+1} \in \{x_{\alpha}\}$ such that

$$\|\langle \xi - x_{n+1}, x_i \rangle\| < 1/2^{n+1}$$
 and $\|\langle x_{n+1}, \xi \rangle - a\| < 1/2^{n+1}$, $i = 1, 2, ..., n$.

Thus, by induction, we obtain a subsequence $\{x_n\}$ in $\{x_\alpha\}$ such that

$$\lim_{n \to \infty} \|\langle x_n, \xi \rangle - a\| = 0 \quad \text{and} \quad \lim_{n \to \infty} \|\langle \xi - x_n, x_i \rangle\| = 0 \text{ for } i \in \mathbb{N}.$$

Denote by H_0 the Hilbert A-submodule generated by $\{x_1, x_2, \ldots, x_n, \ldots\}$. In particular, $x_n \in H_0$ and $n \in \mathbb{N}$. Let $\eta = \xi|_{H_0}$.

Now H_0 is countably generated and $x_n \in H_0$, so we have

$$\lim_{n \to \infty} \|\eta(x_n) - a\| = \lim_{n \to \infty} \|\xi(x_n) - a\| = 0.$$

Moreover, if $y = \sum_{i=1}^{m} x_i \cdot a_i$, where $a_i \in A$, then

$$\lim_{n\to\infty} \|\eta(y) - \langle x_n, y \rangle\| = 0.$$

Since $\{x_n\}$ is bounded (since $\{x_\alpha\}$ is bounded), this implies that

$$\lim_{n \to \infty} \|\eta(y) - \langle x_n, y \rangle\| = 0 \quad \text{for all } y \in H_0.$$

Applying what has been proved, we conclude that $\eta \in H_0$ and $\lim_{n\to\infty} \langle \eta, x_n \rangle = \langle \eta, \eta \rangle = a$.

We now consider Hilbert A^{**} -modules $H_0 \bullet A^{**} \subset H \bullet A^{**}$. By Proposition 3.1, we obtain a projection $P: H^{\sim} \to H_0^{\sim}$ such that $P|_{H_0 \bullet A^{**}} = \operatorname{id}_{H_0 \bullet A^{**}}$. Then $\eta = P(\xi)$. Hence, by the claim,

$$\|(1-P)\xi\|^{2} = \|\langle (1-P)(\xi), (1-P)(\xi) \rangle\| \le \|\langle (1-P)(\xi), \xi \rangle\| + \|\langle (1-P)(\xi), P(\xi) \rangle\|$$
$$= \|\langle \xi, \xi \rangle - \langle P(\xi), \xi \rangle\| + 0 = \|a - \langle P(\xi), P(\xi) \rangle\| = \|a - \langle \eta, \eta \rangle\| = 0.$$

In other words, $P(\xi) = \eta = \xi$. The theorem follows.

Definition 5.6. Let A be a C^{*}-algebra and H be a Hilbert A-module. Then $H^{\sharp} \subset H^{\sim}$.

For each $\xi \in H^{\sharp}$, $\epsilon > 0$ and a finite subset $Y \subset H^{\sharp}$, define

$$O_{\xi,\epsilon,Y} = \{ \zeta \in H^{\sharp} : \| \langle \xi - \zeta, y \rangle \| < \epsilon, y \in Y \},\$$

where the inner product is taken from H^{\sharp} if H^{\sharp} is a Hilbert A-module, or from H^{\sim} (with values in A^{**}).

Denote by \mathcal{T}_{NW} the topology in H^{\sharp} generated by $O_{\xi,\epsilon,Y}$ for all $\xi \in H^{\sharp}$, $\epsilon \in \mathbb{R}_+ \setminus \{0\}$ and finite subsets $Y \subset H^{\sharp}$. Note that a net $\{\zeta_{\alpha}\}$ converges to ξ in H^{\sharp} in \mathcal{T}_{NW} if and only if

$$\lim_{\alpha} \|\langle \xi - \zeta_{\alpha}, y \rangle\| = 0$$

for all $y \in H^{\sharp}$, where the inner product is the one defined above.

Corollary 5.7. Let A be a C^{*}-algebra and H be a Hilbert A-module. Then, with \mathcal{T}_{NW} , the unit ball of H is closed in H^{\sharp} .

Proof. Let $\xi \in H^{\sharp}$. Suppose that there is a net $\{x_{\alpha}\}$ in H with $||x_{\alpha}|| \leq 1$ such that

$$\lim_{\alpha} \|\langle \xi - x_{\alpha}, \eta \rangle\| = 0 \quad \text{for all } \eta \in H^{\sharp},$$

where the inner product is in H^{\sim} . Then, for each $x \in H$, $\lim_{\alpha} ||\langle \xi - x_{\alpha}, x \rangle|| = 0$ and (by choosing $\eta = \xi$) $\{\xi(x_{\alpha})\} = \{\langle \xi, x_{\alpha} \rangle\}$ converges in norm to $\langle \xi, \xi \rangle$. By Theorem 5.5, $\xi \in H$.

1560

Corollary 5.8. Let A be a monotone complete C^* -algebra and H be a Hilbert A-module. Then the unit ball of H is closed in H^{\sharp} in the topology \mathcal{T}_{NW} , where we view H^{\sharp} as a self-dual Hilbert A-module.

Lemma 5.9. Let X be a Hilbert space, $A \subset B(X)$ be a C^* -subalgebra and $M = \overline{A}^{SOT}$, with $id_X \in M$. Let H be a Hilbert A-module. Suppose that $\xi \in H \bullet M$ and $\langle \xi, x \rangle \in A$ for all $x \in H$. Then $\xi \in H$.

Proof. First let us consider the case that $H = H_A$. Then, by Proposition 3.6,

$$H_A \bullet M = \left\{ \{a_n\} : a_n \in \overline{AM} \text{ and } \sum_{k=1}^n a_k^* a_k \text{ converges in norm} \right\}.$$

Write $\xi = \{b_n\} \in H_A \bullet M$. The condition that $\langle \xi, x \rangle \in A$ for all $x \in H_A$ implies that $\xi \in H_A^{\sharp}$. It follows that $b_n \in A$. Hence $\xi \in H_A$.

Next, let us assume that *H* is countably generated. Let $\xi \in H \bullet M$ and $\langle \xi, x \rangle \in A$ for all $x \in H$. By Kasparov's absorbing theorem, we may write $H_A = H \oplus H^{\perp}$. It follows from what has been proved that $\xi \in H_A$. Let $P : H_A \to H$ be the projection. Then $P(\xi) \in H$. However, $\langle \xi - P(\xi), x \rangle = 0$ for all $x \in H$. For any $y \in H^{\perp}$, since $\xi \in H \bullet M$, we have $\langle \xi, y \rangle = 0$ for all $y \in H$. Hence $\xi = P(\xi) \in H$.

In general, since $\xi \in H \bullet M$, there are $x_{n,i} \in H$, i = 1, 2, ..., k(n), $b_{n,i} \in M$, i = 1, 2, ..., k(n), $n \in \mathbb{N}$, such that

$$\lim_{n\to\infty}\left\|\xi-\sum_{i=1}^{k(n)}x_{n,i}\bullet b_{n,i}\right\|=0.$$

Let H_0 be the Hilbert A-submodule generated by $\{x_{n,i} : 1 \le i \le k(n), n \in \mathbb{N}\}$. Then $\xi \in H_0 \bullet M$ and $\xi|_{H_0} \in H_0^{\sharp}$, as $\langle \xi, h \rangle \in A$ for all $h \in H_0 \subset H$. From what has just been proved, $\xi \in H_0 \subset H$.

We end this section with the following result.

Theorem 5.10. Let A be a C^{*}-algebra and H be a Hilbert A-module. Then the unit ball of H is closed in H^{\sim} in the topology \mathcal{T}_{NW} of $H^{\sim} = (H \bullet A^{**})^{\sharp}$.

Proof. Let $\{x_{\alpha}\}$ be a net in the unit ball of H and $\xi \in H^{\sim}$ such that

$$\lim \|\langle \xi - x_{\alpha}, \zeta \rangle\| = 0 \quad \text{for all } \zeta \in H^{\sim}.$$

Since $H^{\sim} = (H \bullet A^{**})^{\sharp}$ and $H \subset H \bullet A^{**}$, by applying Corollary 5.8, we conclude that $\xi \in H \bullet A^{**}$. We also have, for all $y \in H$,

$$\lim \|\langle \xi - x_{\alpha}, y \rangle\| = 0.$$

Since $\langle x_{\alpha}, y \rangle \in A$, it follows that $\langle \xi, y \rangle \in A$. By Lemma 5.9, $\xi \in H$.

6. A Kaplansky-style density theorem in the self-dual Hilbert modules

In the last section, we show that H is closed in H^{\sharp} and H^{\sim} in the topology \mathcal{T}_{NW} of H^{\sharp} and that of H^{\sim} , respectively. In this section, however, we will show that H is dense in H^{\sim} in a weaker topology. In fact, by Theorem 4.5, it is easy to show that H is dense in H^{\sharp} in \mathcal{T}_0 , the topology defined below (see Definition 6.1). A similar question is whether one can replace x in (6-1) by any element in H^{\sharp} .

Definition 6.1. Let A be a W^* -algebra and H be a Hilbert A-module.

Let $\epsilon > 0$, and let $Y \subset H$ and $\mathcal{F} \subset A_*$ be finite subsets. Let $\xi \in H^{\sharp}$. Define

$$O_{\xi,\epsilon,Y,\mathcal{F}} = \{\zeta \in H^{\sharp} : |f(\langle \xi - \zeta, x \rangle)| < \epsilon, \ x \in Y, \ f \in \mathcal{F}\} \subset H^{\sharp}.$$
(6-1)

Let \mathcal{T}_0 be the topology of H^{\sharp} generated by the subsets $O_{\xi,\epsilon,Y,\mathcal{F}}$.

Let $\epsilon > 0$, and let $Z \subset H^{\sharp}$ and $\mathcal{F} \subset A_*$ be finite subsets. Let $\xi \in H^{\sharp}$. Define

$$O_{\xi,\epsilon,Z,\mathcal{F}} = \{\zeta \in H^{\sharp} : |f(\langle \xi - \zeta, x \rangle)| < \epsilon, x \in Z, f \in \mathcal{F}\} \subset H^{\sharp}.$$

Let \mathcal{T}_w be the topology of H^{\sharp} generated by the subsets $O_{\xi,\epsilon,Z,\mathcal{F}}$.

In fact, by [Paschke 1973, Proposition 3.8] and the definition before it, \mathcal{T}_w is the weak* topology of H^{\sharp} as a conjugate space. So a natural question is whether H is dense in H^{\sharp} in \mathcal{T}_w . To be more useful (but perhaps not useful enough to be used twice on Sundays — see [Pedersen 1979, 2.3.4]), we will also prove a Kaplansky-style density theorem in Theorem 6.4.

Let us also consider another topology. Let $\epsilon > 0$, $\xi \in H^{\sharp}$, and let $\mathcal{F} \subset A_*$ be a finite subset. Define

$$O_{\epsilon,\xi,\mathcal{F}} = \{ \zeta \in H^{\sharp} : |f(\langle \xi - \zeta, \xi - \zeta \rangle)| < \epsilon, \ f \in \mathcal{F} \}.$$

Let \mathcal{T}_{ws} be the topology generated by $O_{\epsilon,\xi,\mathcal{F}}$ for all $\epsilon > 0$, $\xi \in H^{\sharp}$ and finite subsets $\mathcal{F} \subset A_*$. Note that \mathcal{T}_{ws} is stronger than \mathcal{T}_w , which is stronger than \mathcal{T}_0 .

Lemma 6.2. Let X be a Hilbert space and $A \subset B(X)$ be a C*-subalgebra. Suppose that $M = \overline{A}^{SOT}$, with $id_X \in M$ and $b = \{b_k\} \in H_M^{\sharp}$. There is a net $a_{\alpha} = \{(a_{1,\alpha}, a_{2,\alpha}, \ldots, a_{n,\alpha}, \ldots)\} \in H_A$ such that

$$\left\|\sum_{j=1}^{\infty} a_{j,\alpha}^* a_{j,\alpha}\right\|^{1/2} \le \|b\|,\tag{6-2}$$

$$\lim_{\alpha} f\left(\sum_{j=1}^{\infty} (b_j - a_{j,\alpha})^* (b_j - a_{j,\alpha})\right) = 0$$
(6-3)

for all $f \in M_*$.

Proof. Let $Y = l^2(X)$, the Hilbert space direct sum of countably many copies of X. Let $\bar{b} = (c_{i,j}) \in B(Y)$, where $c_{i,1} = b_i$, $i \in \mathbb{N}$, and $c_{i,j} = 0$ if $j \ge 2$ (see (3-8)). Denote by $P_n : Y \to X^{(n)}$ the projection, where $X^{(n)}$ is the direct sum of (first) n copies of X. Let $\epsilon > 0$ and $V \in L^2(X)$ be a finite subset. Then there is $n_0 \in \mathbb{N}$ such that

$$||(1 - P_{n_0})(v)|| < \frac{1}{2}\epsilon(1 + ||b||)$$
 for all $v \in V$.

There is $d \in M_{n_0}(A)$ such that

$$\|(\overline{b}-d)(P_{n_0}(v))\| < \frac{1}{4}\epsilon$$
 for all $v \in V$.

We have

$$\begin{aligned} \|(\bar{b} - dP_{n_0})(v)\| &\leq \|(\bar{b} - dP_{n_0})(1 - P_{n_0})(v)\| + \|(\bar{b} - d)P_{n_0}(v)\| \\ &= \|\bar{b}(1 - P_{n_0})(v)\| + \frac{1}{4}\epsilon < \epsilon \qquad \text{for all } v \in V. \end{aligned}$$

Let *B* be the self-adjoint algebra of those bounded operators on *Y* which can be expressed as infinite matrices with entries in *A*, where all are zero except finitely many of them. Then, by what has been proved, we conclude that, in the strong operator topology (of B(Y)), operator \bar{b} is in the closure of operators in *B* in the strong operator topology.

Then, by the Kaplansky density theorem, there is a net $\{d_{\alpha}\} \in B$ with $||d_{\alpha}|| \le ||\bar{b}||$ such that

$$\lim_{\alpha} \|(\bar{b} - d_{\alpha})v\| = 0 \quad \text{for all } v \in Y.$$

Since $\{\|\bar{b} - d_{\alpha}\|\}$ is bounded, we also have

$$\lim_{\alpha} \|(\bar{b} - d_{\alpha})^* (\bar{b} - d_{\alpha})v\| = 0 \quad \text{for all } v \in Y.$$

We further note that

$$\|\bar{b}\|^{2} = \|(\bar{b})^{*}\bar{b}\| = \left\|\sum_{j=1}^{\infty} b_{j}^{*}b_{j}\right\| \le \|b\|.$$

Then

$$\lim_{\alpha} \|(\bar{b} - d_{\alpha})^* (\bar{b} - d_{\alpha}) P_1 v\| = 0 \quad \text{for all } v \in Y.$$
(6-4)

Note $\bar{b}P_1 = \bar{b}$. Let $d'_{\alpha} = d_{\alpha}P_1 = (d_{i,j,\alpha})$, where $d_{i,j,\alpha} = 0$ if $j \ge 2$. Put $a_{j,\alpha} = d_{1,j,\alpha}$, $j \in \mathbb{N}$. Then, for all $n \in \mathbb{N}$,

$$\left\|\sum_{j=1}^{n} a_{j,\alpha}^{*} a_{j,\alpha}\right\| \leq \|(d_{\alpha}')^{*} d_{\alpha}'\| = \|d_{\alpha}'\|^{2} \leq \|d_{\alpha}\|^{2} \leq \|\bar{b}\|^{2} \leq \|b\|^{2}$$

Put $a_{\alpha} = \{a_{j,\alpha}\}$. Since $d_{\alpha} \in B$, for each α , there are only finitely many $a_{j,\alpha}$ which are not zero. Hence $a_{\alpha} \in H_A$. Then $||a_{\alpha}|| \le ||b||$. Thus (6-2) holds. On the other hand, by (6-4),

$$\lim_{\alpha} \|(\bar{b} - d'_{\alpha})^* (\bar{b} - d'_{\alpha}) P_1 v\| = 0.$$
(6-5)

Let $h \in X$. By (6-5),

$$\lim_{\alpha} \left\| \sum_{j=1}^{\infty} (b_j - a_{j,\alpha})^* (b_j - a_{j,\alpha}) h \right\| = 0.$$

In other words, $\sum_{i=1}^{\infty} (b_j - a_{j,\alpha})^* (b_j - a_{j,\alpha}) = \langle b - a_\alpha, b - a_\alpha \rangle \to 0$ in the strong operator topology. However,

$$\left\|\sum_{j=1}^{n} (b_j - a_{j,\alpha})^* (b_j - a_{j,\alpha})\right\| = \|(\bar{b} - d'_{\alpha})\|^2 \le (\|\bar{b}\| + \|d_{\alpha}\|)^2 \le 4\|b\|^2.$$

Therefore $\sum_{i=1}^{n} (b_j - a_{j,\alpha})^* (b_j - a_{j,\alpha}) \to 0$ in the σ -weak operator topology and hence in the weak* topology (see, for example, 4.6.13 of [Pedersen 1989]). Therefore (6-3) holds.

Lemma 6.3. Let $A \subset B(X)$ be a C^* -subalgebra, and let $M = \overline{A}^{SOT}$, with $1_X \in M$. Suppose that H is a countably generated Hilbert A-module. Then H is dense in $(H \bullet M)^{\sharp}$ in the following sense: for any $\xi \in (H \bullet M)^{\sharp}$, there is a net $x_{\alpha} \in H$ with $||x_{\alpha}|| \leq ||\xi||$ such that

$$\lim_{\alpha} \sup\{|f(\langle \xi - x_{\alpha}, \zeta \rangle)| : \zeta \in (H \bullet M)^{\sharp}, \|\zeta\| \le 1\} = 0 \quad \text{for all } f \in M_{*}.$$
(6-6)

HUAXIN LIN

Proof. Let us first prove this for $H = H_A$, even though when A is not σ -unital, H_A is not countably generated. Lemma 6.2 provides a net $\{x_\alpha\}$ in H_A with $||x_\alpha|| \le ||\xi||$ such that

$$\lim_{\alpha} f(\langle \xi - x_{\alpha}, \xi - x_{\alpha} \rangle) = 0 \quad \text{for all } f \in M_*.$$

Recall that, for any positive linear functional f, the map $H_M^{\sharp} \times H_M^{\sharp} \to \mathbb{R}$ defined by $[x, y]_f = f(\langle x, y \rangle)$ (for all $x, y \in H_M^{\sharp}$) is a pseudo inner product. Therefore, by the Cauchy–Bunyakovsky–Schwarz inequality,

 $f(\langle x, y \rangle)^2 \le f(\langle x, x \rangle) f(\langle y, y \rangle) \text{ for all } x, y \in H_M^{\sharp}.$

It follows that, for any positive normal linear functional f,

$$\sup\{|f(\langle \xi - x_{\alpha}, \zeta \rangle)| : \zeta \in H_{M}^{\sharp}, \|\zeta\| \le 1\}^{2} \le \sup\{f(\langle \zeta, \zeta \rangle)f(\langle \xi - x_{\alpha}, \xi - x_{\alpha} \rangle) : \zeta \in H_{M}^{\sharp}, \|\zeta\| \le 1\}$$
$$= \|f\|f(\langle \xi - x_{\alpha}, \xi - x_{\alpha} \rangle) \to 0.$$

Thus we proved (6-6) holds for $H = H_A$.

Now let *H* be a countably generated Hilbert *A*-module. Then, by Kasparov's absorbing theorem, we may write $H_A = H \oplus H^{\perp}$. Hence $H_A \bullet M = H \bullet M \oplus (H^{\perp} \bullet M)$. It follows that

$$H_M^{\sharp} = (H_A \bullet M)^{\sharp} = (H \bullet M)^{\sharp} \oplus (H^{\perp} \bullet M)^{\sharp}.$$

Let $P: H_M^{\sharp} \to (H \bullet M)^{\sharp}$ be the projection such that $P|_H = id_H$. Then, by what has been proved for H_A , there is a net $y_{\alpha} \in H_A$ such that $||y_{\alpha}|| \le ||\xi||$ and, for any $f \in M_*$,

$$\lim_{\alpha} \sup\{f(\langle \xi - y_{\alpha}, \zeta \rangle) : \zeta \in H_M^{\sharp}, \, \|\zeta\| \le 1\} = 0.$$

Put $x_{\alpha} = P(y_{\alpha}) \in (H \bullet M)^{\sharp}$. Note that $P(\xi) = \xi$. Then, for any $f \in M_*$,

$$\begin{split} \lim_{\alpha} \sup\{f(\langle \xi - x_{\alpha}, \zeta \rangle) : \zeta \in (H \bullet M)^{\sharp}, \, \|\zeta\| \le 1\} &= \lim_{\alpha} \sup\{f(\langle \xi - x_{\alpha}, P(\zeta) \rangle) : \zeta \in (H \bullet M)^{\sharp}, \, \|\zeta\| \le 1\} \\ &= \lim_{\alpha} \sup\{f(\langle \xi - y_{\alpha}, \zeta \rangle) : \zeta \in (H \bullet M)^{\sharp}, \, \|\zeta\| \le 1\} \\ &\le \lim_{\alpha} \sup\{f(\langle \xi - y_{\alpha}, \zeta \rangle) : \zeta \in H_{M}^{\sharp}, \, \|\zeta\| \le 1\} = 0. \quad \Box \end{split}$$

Theorem 6.4. Let X be a Hilbert space, $A \subset B(X)$ a C*-subalgebra and $M = \overline{A}^{SOT}$, with $1_M = id_X$, and let H be a Hilbert A-module. Then the unit ball of H is dense in the unit ball of $(H \bullet M)^{\sharp}$ in \mathcal{T}_{ws} (the topology on $(H \bullet M)^{\sharp}$).

Proof. Let $\xi \in (H \bullet M)^{\sharp}$ with $\|\xi\| \le 1$. It suffices to show that, for any $\epsilon > 0$, any finite subset $Y \subset (H \bullet M)^{\sharp}$ and any finite subset $\mathcal{F} \subset M_*$, there is $x \in H$ such that

$$||x|| \le ||\xi||$$
 and $|f(\langle \xi - x, y \rangle)| < \epsilon$ for all $y \in (H \bullet M)^{\sharp}$, $||y|| \le 1$, and $f \in \mathcal{F}$.

Let us fix ϵ and \mathcal{F} . Choose an approximate identity $\{E_{\lambda}\}$ for K(H). It follows that $E_{\lambda} \nearrow \operatorname{id}_{H}$. Note that $\operatorname{id}_{H} \in M(K(H))$. By the last part of Proposition 2.13, $\Psi_{0}(\operatorname{id}_{H}) = \operatorname{id}_{H \bullet M}$. By [Paschke 1973, Corollary 3.7], $F \circ \Psi_{0}(\operatorname{id}_{H}) = \operatorname{id}_{(H \bullet M)^{\sharp}}$, where $F : K(H \bullet M)^{**} \to B(H \bullet M)^{\sharp}$ is the map given by Proposition 2.16. Note also that, by Lemma 2.9, $\{\Psi_{0}(E_{\lambda})\}$ is an approximate identity for $K(H \bullet M)$. In the universal representation of $K(H \bullet M)$, we have that $1 - \Psi_{0}(E_{\lambda})$ converges to zero in the strong operator topology. Note that $||1 - \Psi_0(E_\lambda)|| \le 1$. Therefore $(1 - \Psi_0(E_\lambda))^*(1 - \Psi_0(E_\lambda))$ also converges to zero in the strong operator topology. Hence (since $\{||(1 - \Psi_0(E_\lambda))^*(1 - \Psi_0(E_\lambda))||\}$ is bounded), it converges to zero in the weak* topology. By Proposition 2.16, we have, for all $f \in M_*$,

 $f(\langle \xi - F \circ \Psi_0(E_{\lambda})(\xi), \xi - F \circ \Psi_0(E_{\lambda})(\xi) \rangle)$

$$= f(\langle (1 - F \circ \Psi_0(E_\lambda))^* (1 - F \circ \Psi_0(E_\lambda))(\xi), \xi \rangle) \to 0.$$
 (6-7)

Next let g be a positive normal linear functional in M_* . Then, for any $y \in (H \bullet M)^{\sharp}$ with $||y|| \le 1$,

$$\begin{aligned} |g(\langle \xi - F \circ \Psi_0(E_{\lambda})(\xi), y \rangle)|^2 &\leq g(\langle \xi - F \circ \Psi_0(E_{\lambda})(\xi), \xi - F \circ \Psi_0(E_{\lambda})(\xi) \rangle)g(\langle y, y \rangle) \\ &\leq ||g|| ||y||^2 g(\langle \xi - F \circ \Psi_0(E_{\lambda})(\xi), \xi - F \circ \Psi_0(E_{\lambda})(\xi) \rangle). \end{aligned}$$

Hence, by (6-7),

$$\lim_{\alpha} \left(\sup\{ |g(\langle \xi - F \circ \Psi_0(E_{\lambda})(\xi), y \rangle)| : y \in (H \bullet M)^{\sharp}, \|y\| \le 1 \} \right) = 0.$$

It follows that, for any $f \in M_*$,

$$\lim_{\alpha} \left(\sup\{ |f(\langle \xi - F \circ \Psi_0(E_\lambda)(\xi), y \rangle)| : y \in (H \bullet M)^{\sharp}, \|y\| \le 1 \} \right) = 0.$$

Put $\Phi := F \circ \Psi_0$. We obtain λ_0 such that, for all $\lambda \ge \lambda_0$,

$$|f(\langle \xi - \Phi(E_{\lambda})(\xi), y \rangle)| < \frac{1}{2}\epsilon \quad \text{for all } y \in (H \bullet M)^{\sharp}, \ \|y\| \le 1, \ \text{and} \ f \in \mathcal{F}.$$
(6-8)

Let $H_{\lambda} = \overline{E_{\lambda}(H)}$. As in the proof of Theorem 3.12, we have that H_{λ} is countably generated. Moreover, by Proposition 3.1,

$$(H \bullet M)^{\sharp} = (H_{\lambda} \bullet M)^{\sharp} \oplus ((H_{\lambda} \bullet M)^{\sharp})^{\perp}.$$

Let $P_{\lambda} : (H \bullet M)^{\sharp} \to (H_{\lambda} \bullet M)^{\sharp}$ be the projection. Note that

$$\Phi(E_{\lambda})(\xi), \, \Phi(E_{\lambda})(y) \in P_{\lambda}((H \bullet M)^{\sharp}) = (H_{\lambda} \bullet M)^{\sharp}$$

for all $y \in (H \bullet M)^{\sharp}$.

It follows from Lemma 6.3 that there is $x \in H_{\lambda}$ with $||x|| \le ||\Phi(E_{\lambda})(\xi)|| \le ||\xi||$ such that

$$|f(\langle \Psi(E_{\lambda})(\xi) - x, P_{\lambda}(y) \rangle)| < \frac{1}{2}\epsilon$$
 for all $y \in (H_{\lambda} \bullet M)^{\sharp}, ||y|| \le 1$.

Since $P_{\lambda}\Phi(E_{\lambda}) = \Phi(E_{\lambda})$ and $x \in H_{\lambda}$, we have, for all $y \in (H_{\lambda} \bullet M)^{\sharp}$, $||y|| \le 1$,

$$|f(\langle \Phi(E_{\lambda})(\xi) - x, y \rangle)| = |f(\langle P_{\lambda}\Phi(E_{\lambda})(\xi) - P_{\lambda}(x), y \rangle)| = |f(\langle \Phi(E_{\lambda})(\xi) - x, P_{\lambda}(y) \rangle)| < \frac{1}{2}\epsilon.$$

Thus (also applying (6-8)) for all $y \in (H \bullet M)^{\sharp}$ with $||y|| \le 1$ and $f \in \mathcal{F}$,

$$|f(\langle \xi - x, y \rangle)| \le |f(\langle \xi - \Phi(E_{\lambda})(\xi), y \rangle)| + |f(\langle \Phi(E_{\lambda})(\xi) - x, y \rangle)| < \frac{1}{2}\epsilon + \frac{1}{2}\epsilon = \epsilon.$$

The next two statements are the main results of this section.

Corollary 6.5. Let A be a W*-algebra and H be a Hilbert A-module. Then the unit ball of H is dense in H^{\ddagger} in \mathcal{T}_{ws} .

Proof. Let M = A and then apply Theorem 6.4.

HUAXIN LIN

Theorem 6.6. Let A be a C^{*}-algebra and H be a Hilbert A-module. Then the unit ball of H is dense in H^{\sim} in \mathcal{T}_{ws} (as $H^{\sim} = (H \bullet A^{**})^{\sharp}$).

Proof. We choose the universal representation π_U and its strong operator closure $A'' = A^{**}$, then apply Theorem 6.4.

References

- [Baaj and Skandalis 1993] S. Baaj and G. Skandalis, "Unitaires multiplicatifs et dualité pour les produits croisés de *C**-algèbres", *Ann. Sci. École Norm. Sup.* (4) **26**:4 (1993), 425–488. MR Zbl
- [Brown and Ciuperca 2009] N. P. Brown and A. Ciuperca, "Isomorphism of Hilbert modules over stably finite *C**-algebras", *J. Funct. Anal.* **257**:1 (2009), 332–339. MR Zbl
- [Brown and Lin 2025] L. G. Brown and H. Lin, "Projective Hilbert modules and sequential approximations", *Sci. China Math.* **68**:1 (2025), 137–168. MR Zbl
- [Conway 2000] J. B. Conway, *A course in operator theory*, Grad. Stud. Math. **21**, Amer. Math. Soc., Providence, RI, 2000. MR Zbl
- [Coward et al. 2008] K. T. Coward, G. A. Elliott, and C. Ivanescu, "The Cuntz semigroup as an invariant for *C**-algebras", *J. Reine Angew. Math.* **623** (2008), 161–193. MR Zbl
- [Kaplansky 1953] I. Kaplansky, "Modules over operator algebras", Amer. J. Math. 75 (1953), 839–858. MR Zbl
- [Kasparov 1980] G. G. Kasparov, "Hilbert C*-modules: theorems of Stinespring and Voiculescu", J. Operator Theory 4:1 (1980), 133–150. MR Zbl
- [Lin 1991a] H. X. Lin, "Bounded module maps and pure completely positive maps", *J. Operator Theory* **26**:1 (1991), 121–138. MR Zbl
- [Lin 1991b] H. X. Lin, "Hilbert C*-modules and their bounded module maps", *Sci. China Ser. A* **34**:12 (1991), 1409–1421. MR Zbl
- [Lin 1992] H. X. Lin, "Injective Hilbert C*-modules", Pacific J. Math. 154:1 (1992), 131–164. MR Zbl
- [Lin 2001] H. Lin, An introduction to the classification of amenable C*-algebras, World Sci., River Edge, NJ, 2001. MR Zbl
- [Ortega et al. 2011] E. Ortega, M. Rørdam, and H. Thiel, "The Cuntz semigroup and comparison of open projections", *J. Funct. Anal.* **260**:12 (2011), 3474–3493. MR Zbl
- [Paschke 1973] W. L. Paschke, "Inner product modules over *B**-algebras", *Trans. Amer. Math. Soc.* **182** (1973), 443–468. MR Zbl
- [Pedersen 1979] G. K. Pedersen, C*-algebras and their automorphism groups, Lond. Math. Soc. Monogr. 14, Academic Press, London, 1979. MR Zbl
- [Pedersen 1989] G. K. Pedersen, Analysis now, Grad. Texts in Math. 118, Springer, 1989. MR Zbl

Received 17 Dec 2023. Revised 5 Apr 2024. Accepted 8 Jun 2024.

HUAXIN LIN: hlin@uoregon.edu

Shanghai Institute for Mathematics and Interdisciplinary Sciences, Shanghai, China



Guidelines for Authors

Authors may submit manuscripts in PDF format on-line at the Submission page at msp.org/apde.

Originality. Submission of a manuscript acknowledges that the manuscript is original and and is not, in whole or in part, published or under consideration for publication elsewhere. It is understood also that the manuscript will not be submitted elsewhere while under consideration for publication in this journal.

Language. Articles in APDE are usually in English, but articles written in other languages are welcome.

Required items. A brief abstract of about 150 words or less must be included. It should be self-contained and not make any reference to the bibliography. If the article is not in English, two versions of the abstract must be included, one in the language of the article and one in English. Also required are keywords and subject classifications for the article, and, for each author, postal address, affiliation (if appropriate), and email address.

Format. Authors are encouraged to use $\mathbb{E}T_{EX}$ but submissions in other varieties of T_{EX} , and exceptionally in other formats, are acceptable. Initial uploads should be in PDF format; after the refereeing process we will ask you to submit all source material.

References. Bibliographical references should be complete, including article titles and page ranges. All references in the bibliography should be cited in the text. The use of BibT_EX is preferred but not required. Tags will be converted to the house format, however, for submission you may use the format of your choice. Links will be provided to all literature with known web locations and authors are encouraged to provide their own links in addition to those supplied in the editorial process.

Figures. Figures must be of publication quality. After acceptance, you will need to submit the original source files in vector graphics format for all diagrams in your manuscript: vector EPS or vector PDF files are the most useful.

Most drawing and graphing packages (Mathematica, Adobe Illustrator, Corel Draw, MATLAB, etc.) allow the user to save files in one of these formats. Make sure that what you are saving is vector graphics and not a bitmap. If you need help, please write to graphics@msp.org with details about how your graphics were generated.

White space. Forced line breaks or page breaks should not be inserted in the document. There is no point in your trying to optimize line and page breaks in the original manuscript. The manuscript will be reformatted to use the journal's preferred fonts and layout.

Proofs. Page proofs will be made available to authors (or to the designated corresponding author) at a Web site in PDF format. Failure to acknowledge the receipt of proofs or to return corrections within the requested deadline may cause publication to be postponed.

ANALYSIS & PDE

Volume 18 No. 6 2025

Damped Strichartz estimates and the incompressible Euler–Maxwell system DIOGO ARSÉNIO and HAROUNE HOUAMED	1309
Equivariant property Gamma and the tracial local-to-global principle for C*-dynamics GÁBOR SZABÓ and LISE WOUTERS	1385
Multijet bundles and application to the finiteness of moments for zeros of Gaussian fields MICHELE ANCONA and THOMAS LETENDRE	1433
Existence of solutions to a fractional semilinear heat equation in uniformly local weak Zygmund-type spaces NORISUKE IOKU, KAZUHIRO ISHIGE and TATSUKI KAWAKAMI	1477
A Marcinkiewicz multiplier theory for Schur multipliers CHIAN YEONG CHUAH, ZHEN-CHUAN LIU and TAO MEI	1511
Double duals and Hilbert modules	1531